UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

Amendment No. 1

FORM S-1

REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933



Aesther Healthcare Acquisition Corp.

(Exact name of registrant as specified in its charter)

Delaware	6770	87-1309280
(State or other jurisdiction of	(Primary Standard Industrial	(I.R.S. Employer
incorporation or organization)	Classification Code Number)	Identification Number)

515 Madison Avenue, Suite 8078 New York, New York 10022 **Telephone:** (646) 908-2658

(Address, Including Zip Code, and Telephone Number, Including Area Code, of Registrant's Principal Executive Offices)

Suren Ajjarapu **Chairman and Chief Executive Officer** 515 Madison Avenue, Suite 8078 New York, New York 10022 **Telephone:** (646) 908-2658

(Name, Address, Including Zip Code, and Telephone Number, Including Area Code, of Agent For Service)

Copies to:

David M. Loev John S. Gillies The Loev Law Firm, PC 6300 West Loop South **Suite 280** Bellaire, Texas 77401

Telephone: (713) 524-4110 Facsimile: (713) 524-4122

Cavas Pavri, Esq. **Schiff Hardin LLP** 100 N. 18th Street, Suite 300 Philadelphia, PA 19103 Telephone: (202) 778-6400 Facsimile: (202) 778-6460

Approximate date of commencement of proposed sale to the public: As soon as practicable after the effective date of this registration statement.

If any of the securities being registered on this Form are to be offered on a delayed or continuous basis pursuant to Rule 415 under the Securities Ac of 1933 check the following box. \Box
If this Form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, please check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. \Box

registration statement number of the earlier effective registration statement for the same offering. \Box

If this Form is a post-effective amendment filed pursuant to Rule 462(c) under the Securities Act, check the following box and list the Securities Act

If this Form is a post-effective amendment filed pursuant to Rule 462(d) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. \Box

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company" and "emerging growth company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer		Accelerated filer	
-------------------------	--	-------------------	--

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 7(a)(2)(B) of the Securities Act.

 \boxtimes

CALCULATION OF REGISTRATION FEE

Title of Each Class of Security Being Registered	Amount Being Registered	 Proposed Maximum Offering Price per Security ⁽¹⁾	Proposed Maximum Aggregate Offering Price ⁽¹⁾		 Amount of Registration Fee	
Units, each consisting of one share of Class A common		_		_		
stock, \$0.0001 par value, and one-half of one redeemable						
warrant ⁽²⁾	11,500,000 Units	\$ 10.00	\$	115,000,000	\$ 12,546.50	
Shares of Class A common stock included as part of the						
units ⁽³⁾	11,500,000 Shares	_		_	—(4)	
	5,750,000					
Redeemable warrants included as part of the units ⁽³⁾	Warrants	_		_	(4)	
Representative Shares	100,000 Shares	\$ 10.00	\$	1,000,000	\$ 109.10	
Total			\$	116,000,000	\$ 12,655.60 ⁽⁵⁾	

- (1) Estimated solely for the purpose of calculating the registration fee pursuant to Rule 457(o) under the Securities Act of 1933, as amended (the "Securities Act").
- (2) Includes 1,500,000 units, consisting of 1,500,000 shares of Class A common stock and 750,000 redeemable warrants, which may be issued upon exercise of a 45-day option granted to the underwriters to cover over-allotments, if any.
- (3) Pursuant to Rule 416, there are also being registered an indeterminable number of additional securities as may be issued to prevent dilution resulting from stock splits, stock dividends or similar transactions.
- (4) No fee pursuant to Rule 457(g).
- (5) A total of \$12,546.50 has previously been paid.

The Registrant hereby amends this Registration Statement on such date or dates as may be necessary to delay its effective date until the registrant shall file a further amendment which specifically states that this Registration Statement shall thereafter become effective in accordance with Section 8(a) of the Securities Act or until the Registration Statement shall become effective on such date as the Securities and Exchange Commission, acting pursuant to said Section 8(a), may determine.

The information in this preliminary prospectus is not complete and may be changed. We may not sell these securities until the registration statement filed with the Securities and Exchange Commission is effective. This preliminary prospectus is not an offer to sell these securities and it is not soliciting an offer to buy these securities in any jurisdiction where the offer or sale is not permitted.

PRELIMINARY PROSPECTUS

SUBJECT TO COMPLETION, DATED SEPTEMBER 2, 2021

\$100,000,000 Aesther Healthcare Acquisition Corp. 10,000,000 Units

Aesther Healthcare Acquisition Corp. is a newly organized blank check company formed for the purpose of effecting a merger, capital stock exchange, asset acquisition, stock purchase, reorganization or similar business combination with one or more businesses, which we refer to as our initial business combination throughout this prospectus. We have not selected any specific business combination target and we have not, nor has anyone on our behalf, initiated any substantive discussions, directly or indirectly, with any business combination target. While we may pursue an initial business combination target in any business or industry, we intend to focus our search on industries that complement our management team's background and to capitalize on the ability of our management team to identify and acquire a business focusing in the pharmaceutical and medical devices sectors, where our management team has extensive experience.

This is an initial public offering of our securities. Each unit has an offering price of \$10.00 and consists of one share of our Class A common stock and one-half of one redeemable warrant as described in more detail in this prospectus. Only whole warrants are exercisable. Each whole warrant entitles the holder thereof to purchase one share of our Class A common stock at a price of \$11.50 per share, subject to adjustment as described herein. No fractional warrants will be issued upon separation of the units and only whole warrants will trade. The underwriters have a 45-day option from the date of this prospectus to purchase up to an additional 1,500,000 units to cover over-allotments, if any.

We will provide our public stockholders with the opportunity to redeem all or a portion of their shares of our Class A common stock upon the completion of our initial business combination, subject to the limitations described herein. We have 12 months from the effective date of this registration statement to consummate our initial business combination. However, if we anticipate that we may not be able to consummate our initial business combination within 12 months, we may, but are not obligated to, extend the period of time to consummate a business combination two times by an additional three months each time (for a total of up to 18 months to complete a business combination). If we are unable to complete our initial business combination within the 12-month period (or up to 18 months, if we extend the time to complete a business combination as described in this prospectus), we will distribute our assets, including the aggregate amount then on deposit in the trust account described below and interest (net of taxes payable), pro rata to our public stockholders, by way of redemption of their shares and thereafter cease all operations except for the purposes of winding up of our affairs, as further described herein. If we are unable to complete our initial business combination within 18 months from the closing of this offering, we will redeem 100% of the public shares for cash, subject to applicable law and certain conditions as further described herein.

Our sponsor, Aesther Healthcare Sponsor, LLC, has agreed to purchase an aggregate of 5,261,000 warrants (or 5,711,000 warrants if the overallotment option is exercised in full) at a price of \$1.00 per warrant (approximately \$5,261,000 in the aggregate, or \$5,711,000 if the overallotment option is exercised in full) in a private placement that will close simultaneously with the closing of this offering. We refer to these warrants as the placement warrants throughout this prospectus. Each placement warrant will be identical to the warrants sold as part of the units sold in this offering, except as described in this prospectus. Each placement warrant is exercisable to purchase one share of our Class A common stock at \$11.50 per share.

Our sponsor owns an aggregate of 2,875,000 shares of our Class B common stock (up to 375,000 shares of which are subject to forfeiture depending on the extent to which the underwriters' over-allotment option is exercised), which will automatically convert into shares of Class A common stock at the time of the consummation of our initial business combination, on a one-for-one basis, subject to adjustment as described herein.

Currently, there is no public market for our units, Class A common stock or warrants. We have applied to list our units on The Nasdaq Global Market, or Nasdaq, under the symbol "AEHAU". We expect that our units will be listed on The Nasdaq Global Market on or promptly after the date of this prospectus. We cannot guarantee that our securities will be approved for listing on Nasdaq. We expect the Class A common stock and warrants comprising the units will begin separate trading on the 52nd day following the date of this prospectus unless the representative informs us of its decision to allow earlier separate trading, subject to our satisfaction of certain conditions. Once the securities comprising the units begin separate trading, we expect that the Class A common stock and warrants will be listed on Nasdaq under the symbols "AEHA" and "AEHAW", respectively.

We are an "emerging growth company" under applicable federal securities laws and will be subject to reduced public company reporting requirements. Investing in our securities involves a high degree of risk. See "Risk Factors" beginning on page 29 for a discussion of information that should be considered in connection with an investment in our securities. Investors will not be entitled to protections normally afforded to investors in Rule 419 blank check offerings.

Neither the U.S. Securities and Exchange Commission nor any state securities commission has approved or disapproved of these securities or determined if this prospectus is truthful or complete. Any representation to the contrary is a criminal offense.

	Pe	r Unit	Total
Public offering price	\$	10.00	\$ 100,000,000
Underwriting discounts and commissions ⁽¹⁾	\$	0.40	\$ 4,000,000
Proceeds, before expenses, to Aesther Healthcare Acquisition Corp.	\$	9.60	\$ 96,000,000

⁽¹⁾ Includes \$0.30 per unit, or \$3,000,000 (or \$3,450,000 if the over-allotment option is exercised in full) in the aggregate, payable to the underwriters for deferred underwriting commissions to be placed in a trust account located in the United States as described herein. The deferred commissions will be released to the representative of the underwriters only on completion of an initial business combination, as described in this prospectus. Does not include certain fees and expenses payable to the underwriters in connection with this offering. We will also issue to the representative of the underwriters and/or its designees 100,000 shares of common stock, which we refer to herein as the "representative's shares" as underwriter compensation in connection with this offering. See the section of this prospectus entitled "Underwriting" beginning on page 152 for a description of compensation and other items of value payable to the underwriters.

Of the proceeds we receive from this offering and the sale of the placement warrants described in this prospectus, \$102.00 million or \$117.3 million if the underwriters' over-allotment option is exercised in full (\$10.20 per unit in either case) will be deposited into a trust account in the United States with Continental Stock Transfer & Trust Company, LLC acting as trustee and JP Morgan Chase Bank, N.A. acting as investment manager.

The underwriters are offering the units for sale on a firm commitment basis. The underwriters expect to deliver the units to the purchasers on or about , 2021.

EF HUTTON

division of Benchmark Investments, LLC

The date of this Prospectus is , 2021

TABLE OF CONTENTS

	Page
Summary	1
Risk Factors	29
Cautionary Note Regarding Forward-Looking Statements	68
<u>Use of Proceeds</u>	70
<u>Dividend Policy</u>	74
<u>Dilution</u>	74
<u>Capitalization</u>	76
Management's Discussion and Analysis of Financial Condition and Results of Operations	77
<u>Proposed Business</u>	83
<u>Management</u>	113
<u>Principal Stockholders</u>	122
Certain Relationships and Related Party Transactions	125
Description of Securities	127
<u>United States Federal Income Tax Considerations</u>	142
<u>Underwriting</u>	152
Legal Matters	159
<u>Experts</u>	159
Where You Can Find Additional Information	159
Index to Financial Statements	160

We are responsible for the information contained in this prospectus. We have not authorized anyone to provide you with different information, and we take no responsibility for any other information others may give to you. We are not, and the underwriters are not, making an offer to sell securities in any jurisdiction where the offer or sale is not permitted. You should not assume that the information contained in this prospectus is accurate as of any date other than the date on the front of this prospectus.

This prospectus includes estimates regarding market and industry data and forecasts which are based on publicly available information, industry reports and publications, reports from government agencies and management's estimates based on third-party data. Third-party industry publications and forecasts generally state that the information contained therein has been obtained from sources generally believed to be reliable. The third-party industry sources referenced in this prospectus include, among others, the IQVIA Institute for Human Data Science, Statista and Fortune Business Insights. We have not independently verified such third-party information, nor have we ascertained the underlying economic assumptions relied upon in those sources, and we cannot assure you of the accuracy or completeness of such information contained in this prospectus. Such data involve risks and uncertainties and is subject to change based on various factors, see "Risk Factors".

This prospectus contains trademarks, service marks and trade names of other companies which are the property of their respective owners. We do not intend our use or display of such names or marks to imply relationships with, or endorsements of us by, any other company.

SUMMARY

This summary only highlights the more detailed information appearing elsewhere in this prospectus. You should read this entire prospectus carefully, including the information under the section of this prospectus entitled "Risk Factors" and our financial statements and the related notes included elsewhere in this prospectus, before investing.

Unless otherwise stated in this prospectus, or the context otherwise requires, references to:

- "common stock" are to our Class A common stock and our Class B common stock, collectively;
- "Exchange Act" are to the Exchange Act of 1934, as amended.
- "equity-linked securities" are to any debt or equity securities that are convertible, exercisable or exchangeable for shares of our Class A common stock issued in connection with our initial business combination including but not limited to a private placement of equity or debt;
- "founder shares" are to shares of our Class B common stock initially purchased by our sponsor in a private placement prior to this offering, and the shares of our Class A common stock issuable upon the conversion thereof as provided herein;
- "initial stockholders" are to our sponsor and any other holders of our founder shares prior to this offering (or their permitted transferees);
- "management" or our "management team" are to our officers and directors;
- "placement warrants" are to the warrants being purchased by our sponsor in the private placement;
- "private placement" are to the private placement of 5,261,000 redeemable warrants (or 5,711,000 warrants if the over-allotment option is exercised in full) at a price of \$1.00 per warrant (approximately \$5,261,000 in the aggregate, or \$5,711,000 if the over-allotment option is exercised in full) that will close simultaneously with the closing of this offering;
- "public shares" are to shares of our Class A common stock sold as part of the units in this offering (whether they are purchased in this offering or thereafter in the open market);
- "public stockholders" are to the holders of our public shares, including our initial stockholders and management team to the extent our initial stockholders and/or members of our management team purchase public shares, provided that each initial stockholder's and member of our management team's status as a "public stockholder" shall only exist with respect to such public shares;
- "public warrants" are to our redeemable warrants sold as part of the units in this offering (whether they are purchased in this offering or thereafter in the open market, including warrants that may be acquired by our sponsor or its affiliates in this offering or thereafter in the open market) and to any placement warrants or warrants issued upon conversion of working capital loans in each case that are sold to third parties that are not initial purchasers or executive officers or directors (or permitted transferees) following the consummation of our initial business combination;
- "representative" are to EF Hutton, division of Benchmark Investments, LLC, who is the representative of the underwriters in this offering;
- "sponsor" are to Aesther Healthcare Sponsor, LLC, a Delaware limited liability company, an affiliate of Suren Ajjarapu, our Chief Executive officer;

- "SEC" are to the U.S. Securities and Exchange Commission;
- "Securities Act" are to the Securities Act of 1933, as amended.
- "underwriters" are to the underwriters of this offering, for which the representative is acting as representative;
- "warrants" are to our redeemable warrants, which includes the public warrants as well as the placement warrants and any warrants issued upon conversion of working capital loans to the extent they are no longer held by the initial holders or their permitted transferees; and
- "we," "us," "company" or "our company" are to Aesther Healthcare Acquisition Corp.

Unless we tell you otherwise, the information in this prospectus assumes that the underwriters will not exercise their over-allotment option.

General

We are a newly-organized blank check company incorporated in June 2021 as a Delaware corporation whose business purpose is to effect a merger, capital stock exchange, asset acquisition, stock purchase, reorganization or similar business combination with one or more businesses, which we refer to as our initial business combination. To date, our efforts have been limited to organizational activities as well as activities related to this offering. We have not selected any specific business combination target and we have not, nor has anyone on our behalf, engaged in any substantive discussions, directly or indirectly, with any business combination target with respect to an initial business combination with us.

While we may pursue an initial business combination opportunity in any business, industry, sector or geographical location, we intend to focus on industries that complement our management team's background, and to capitalize on the ability of our management team to identify and acquire a business and it is our current intention to pursue prospective targets that are focused in the pharmaceutical and medical devices sectors. We anticipate focusing our investment effort broadly across the Pharmaceutical and medical devices sectors. We believe that our investment and operating expertise across multiple industry verticals will give us a large, addressable universe of potential targets. The diversity of the target universe and the number of largely uncorrelated subsectors maximizes that likelihood that our management team will be able to identify and execute an attractive transaction. At the time of preparing this prospectus, we have not identified any specific business combination, nor has anyone on our behalf initiated or engaged in any substantive discussions, formal or otherwise, related to such a transaction. Our efforts to date are limited to organizational activities related to this offering.

Industries Overview

Our objectives are to generate compelling attractive returns and to enhance value through top line growth and hands-on operational improvement for our potential target companies. We believe our management team's personnel, network and relationships combined with their unique and diversified experiences in investing, operating and transforming businesses will uniquely position our team to identify and execute attractive business combination opportunities.

Our Management Team

Our Founder and Chief Executive Officer, Surendra Ajjarapu, serves as CEO of TRxADE HEALTH, INC., an online market place for health traded on Nasdaq under the symbol "MEDS" as well as a director of Oceantech Acquisition I Corp., traded on Nasdaq under the symbol "OTECU". Mr. Ajjarapu has over 25 years of specific experience in growing novel technology-based companies, raising capital, mergers and acquisitions and building superior management teams. We are confident that his entrepreneurial experience and expertise in screening high quality target companies will be extremely beneficial in sourcing a target with strong growth potential. In addition, we believe we can capitalize on his previous experiences in advising and expanding startups to help guide and prepare the target for the business combination.

Mr. Howard A. Doss, our Chief Financial Officer, is a seasoned chief financial officer and accountant. He has served as chief financial officer of TRxADE HEALTH, INC., an online market place for health traded on Nasdaq under the symbol "MEDS" and previously served as a chief financial officer of Sansur Renewable Energy, an alternative energy development company. He has proven himself to be an adaptable and strong leader as he served as a top executive at a wide range of companies. We believe Mr. Doss's experiences will be valuable throughout the entire SPAC IPO process and in selecting target companies for acquisition.

Our management team has been actively involved in operating, advising, expanding and generating high returns for many companies. Their executive leadership, operational oversight, strategic management will boost investor confidence in the team's ability to complete a successful business combination. As a result, we believe our management team is well-positioned to take advantage of growing acquisition opportunities in the pharmaceutical and medical devices sectors.

We have also recruited an accomplished and well-recognized group of leaders to be our board of directors. Our board of directors comes from a plethora of industries where they serve as vice presidents, advisors, directors, presidents, vice chairman for public companies, private companies and venture capital firms in Asia and the United States. The following persons will become members of our board of directors as of the effective date of the registration statement of which this prospectus forms a part:

Mr. Michael L. Peterson, one of our directors, was previously the CEO of PEDEVCO Corp. (NYSE American:PED), a public company engaged primarily in the acquisition, exploration, development and production of oil and natural gas shale plays in the US, formerly served as Interim President and CEO and as director of Pacific Energy Development, as a director of Aemetis, Inc. (formerly AE Biofuels Inc.), a Cupertino, California-based global advanced biofuels and renewable commodity chemicals company (AMTX.OB), and as Chairman and Chief Executive Officer of Nevo Energy, Inc. (NEVE) (formerly Solargen Energy, Inc.), a Cupertino, California-based developer of utility-scale solar farms which he helped form. Mr. Peterson served as a managing partner of American Institutional Partners, a venture investment fund based in Salt Lake City. He served as a First Vice President at Merrill Lynch, where he helped establish a new private client services division to work exclusively with high net worth investors. Mr. Peterson was employed by Goldman Sachs & Co. in a variety of positions and roles, including as a Vice President. His skills in managing businesses in public corporations, financial planning and strategic management will be a great asset for the target company.

Mr. Venkatesh Srinivasan has a tremendous amount of experience in the pharmaceutical industry and currently serves as President of Micro Labs USA and previously served as President of Rising Pharma, USA and as President and CEO of Ascend Laboratories, USA where he grew the business, building a new team and strengthening processes and systems. In addition, Mr. Srinivasan served as a Director at Pfizer India. His skillset will be extremely useful as we conduct due diligence works on target companies.

Mr. Donald G. Fell, brings along a wealth of experience in the field of economics and business to the Company, which is instrumental in executing the Company's business strategy. For over 7 years, Mr. Fell has served as an independent director of TRxADE HEALTH, INC. He is presently Professor and Institute Director for the Davis, California-based Foundation for Teaching Economics and adjunct professor of economics for the University of Colorado, Colorado Springs. Mr. Fell held positions with the University of South Florida as a member of the Executive MBA faculty, Director of Executive and Professional Education and Senior Fellow of the Public Policy Institute from 1995 to 2012. Mr. Fell was also a visiting professor at the University of LaRochelle, France, and an adjunct professor of economics at both Illinois State University and The Ohio State University. Mr. Fell holds undergraduate and graduate degrees in economics from Indiana State University and is all but dissertation (ABD) in economics from Illinois State University. Through his work with the Foundation for Teaching Economics and the University of Colorado, Colorado Springs he has overseen graduate institutes on economic policy and environmental economics in 44 states, throughout Canada, the Islands and Eastern Europe.

Mr. Siva Saravanan has more than 20 years of experience steering digital strategies and technology solutions for businesses. Mr. Saravanan is the Chief Digital Officer at Wavestone US, helping Fortune 1000 business and technology leaders accelerate digital transformation. Prior to joining Wavestone US, Siva was Chief Information Officer and SVP of Business Operations at Reviver, an exciting IoT start-up that creates connected digital license plates to enable true autonomous driving. He designed customer digital experiences, unified commerce, supply chain, field service operations and the digital agenda for Reviver. Siva was also VP for IT Digital Transformation and Program Delivery at Aristocrat Technologies. While at Aristocrat Technologies, he led the transformation of business systems for a leading high-tech gaming manufacturer. Siva spent many years at Verifone as a Senior Director supporting technology operations in 40+ countries and also taking on delivery responsibilities. At VeriFone, he built a world-class global integrated supply chain network for agility and efficiency. Mr. Saravanan holds a M.S. in Systems Engineering from Tennessee State University and B.S in Mechanical Engineering from Annamalai University in Chidambaram, India. He is also on the Advisory Board of NishTech Inc., a digital commerce company and the Advisory Council of George Washington University School of Business Digital Program. Siva is a member of Forbes Technology Council contributing regularly.

We believe our management team is well positioned to take advantage of the growing set of acquisition opportunities focused on the pharmaceutical and medical devices sectors and that our contacts and relationships, ranging from owners and management teams of private and public companies, private equity funds, investment bankers, attorneys, to accountants and business brokers will allow us to generate an attractive transaction for our stockholders.

In addition, we may engage the services of a financial advisory services to us in connection with this offering.

The past performance of the members of our management team, our financial advisor or their affiliates is not a guarantee that we will be able to identify a suitable candidate for our initial business combination or of success with respect to any business combination we may consummate. You should not rely on the historical record of the performance of our management team or any of its affiliates' performance as indicative of our future performance.

Business Strategy and Target Industries

While we may pursue an acquisition in any business industry or sector, we intend to focus our efforts identifying businesses primarily in the pharmaceutical and medical devices sectors. We will seek to enter into a business combination with a business that is well positioned to leverage the industry expertise, insights and relationships of the sponsor and our board of directors to create significant opportunities for value creation and improved financial performance.

The pharmaceutical and healthcare sectors have emerged into the public limelight and focus during the COVID-19 pandemic. It is, broadly speaking, a space defined by intense capital requirements, regulatory challenges and a delayed entrance to the market. While institutional investors may see the above features as a barrier to entry and success on returns, our team is confident that they are instead strong arguments for investment into the industry. Highly notable exits in the space have featured many successful companies developing novel treatments in rare disease, orphan drug populations and gene therapy. We will be focusing our efforts on the pharmaceutical sector (both innovative and generic drugs) and the medtech sector.

1. Pharmaceutical Sector:

The global pharmaceutical market value of \$1.3 trillion in 2018 is expected to cross \$1.5 trillion by 2023 with a 4-5% annual growth globally, according to a January 2019 report by the IQVIA Institute for Human Data Science. Despite the pandemic's impact on global business, we believe the industry is very resilient and robust. The spectrum for drug consumption and pursuant development remains wide - aging populations globally are expected to drive high volume consumption generic drugs, while it is estimated that only 5% of the over 7,000 recognized rare diseases have U.S. Food and Drug Administration (FDA) approved treatments (according to the National Organization for Rare Diseases and the FDA.

There is still some uncertainty about how further regulation may play out as the U.S. government grapples with other high priority issues following the 2020 election. So far, significant drug pricing reform has been excluded from recent bills advancing through the House and Senate. The uncertainty around policy presents challenges for this sector as it seeks to optimize deployment of capital and other resources. The second half of 2021 is expected to continue the return to strong levels of merger and acquisition (M&A) activity seen in recent years. While headwinds and tailwinds exist, we believe the need to optimize portfolios and innovate will lead sector participants to deploy capital through M&A.

2. Medtech Sector:

Medtech is a \$450+ billion market with projected growth of $\sim 5.5\%$ per year (according to an October 2020 report by Statista), split across diverse segments:

- Medtech is split into many sub-segments, including: equipment, devices, consumables, in vitro diagnostics (IVD) and life sciences
- Key segments are typically driven by demographics i.e., the ageing population across the world
- Minimally invasive surgery is an important trend for equipment and device segments and prevention is driving the IVD segment

We believe some of the most important segments are - IVD, imaging, cardiology, orthopedics, dental, endoscopy, urology, drug delivery, and radiotherapy.

The North American market is by far largest market with an estimated 39% of the total market share as of 2020 (according to a May 2021 report published by Fortune Business Insights); however, we believe that it is a highly fragmented market.

According to PwC, regulatory uncertainty has recently presented concerns for the pharmaceuticals and medical devices sector in the form of more involved antitrust reviews and in March 2021, the Federal Trade Commission (FTC) announced that it and its counterpart competition enforcement agencies in the U.S. and abroad are launching a working group to update their approach to analyzing the effects of pharmaceutical mergers. While we believe this could potentially slow down any potential transaction, or increase the costs of such a transaction, we do not expect recent regulatory matters to prevent us from completing a potential transaction in the Medtech space.

3. Healthcare IT

Healthcare information technology (IT) is a field of IT that involves the design, creation, development, use, and maintenance of information systems for the healthcare industry. Automatic and interoperable healthcare information systems offer a host of advantages such as low costs, minimal errors, improved medical care and public health, enhanced efficiency, and better patient satisfaction.

The U.S. Health Care Information Technology (HCIT) market was evaluated at \$61,018 million in 2017, and is estimated to reach \$149,178 million by 2025, at a CAGR of 11.7% during the forecast period.

We anticipate that the increasing digitization in the healthcare industry, demand for value-based care and rising costs of healthcare services, coupled with the increasing number of healthcare facilities will accelerate the growth of the healthcare IT market in the near term.

The U.S. healthcare IT market is driven by growing demand for improved patient safety and patient care as well as increases in government initiatives to promote HCIT. However, the U.S. HCIT is a well-established market and thus, there is a large supply and demand gap with many local as well as international key players operating in this segment, which is expected to hamper market growth. Moreover, numerous key players have collaborated with local players for developing efficient products to create opportunities in the market.

* * * * *

The following factors summarize our selection of this space and the advantages we believe we have over other similar companies:

- **Broad Universe of Potential Targets**. We intend to focus our investment effort broadly across the pharmaceutical industry. We believe that our investment and operating expertise across multiple industry verticals will give us a large, addressable universe of potential targets. The diversity of the target universe and the number of largely uncorrelated subsectors maximizes that likelihood that our management team will be able to identify and execute an attractive transaction.
- Strong Access to Transformative Technology. Our team possesses the analytical and technological capabilities to sift through a multitude of emerging private companies across disease state and drug delivery.
- Leverage our Extensive Network of Relationships to Create a Unique Pipeline of Acquisition Opportunities. We believe the combination of Mr. Ajjarapu's and our Board members' broad investment and operating experience in addition to our ability to access a deep network of public and private enterprises, experienced operators, restructuring advisors, attorneys, accountants, family offices, hedge funds, and private equity firms will enable us to identify and evaluate compelling target businesses.
- Employ a Rigorous Systematic Process of Identifying Target Companies and Acquiring a Business that will be Well-Received by the Public Markets. We believe that our directors' and officers' strong merger and acquisition (M&A) and investment track record in both private and public markets, combined with extensive public market trading experience, will provide a distinct advantage for identifying, valuing and completing a business combination that will meet our investors' expectations.
- Providing an Alternative Path to Becoming Public. We believe our structure will make us an attractive business combination partner to prospective target businesses that desire to become a publicly listed company. A merger with us will offer a target business an alternative process to a public listing rather than the traditional initial public offering process. We believe that target businesses may favor this alternative, which we believe is less expensive, while offering greater certainty of execution than the traditional initial public offering. Furthermore, once a proposed business combination is approved by our stockholders and the transaction is consummated, the target business will have effectively become public, whereas an initial public offering is always subject to the underwriters' ability to complete the offering, as well as general market conditions that could prevent the offering from occurring. Once public, we believe the target business would have greater access to capital and additional means of creating management incentives that are better aligned with stockholders' interests than it would as a private company. A public company can offer further benefits by augmenting a company's profile among potential new customers and vendors and aid in attracting talented management. With public company corporate governance standards, a target business may become attractive to public investors.

• Strong and Stable Financial Position with Flexibility. With funds in the trust account of \$102.0 million (or approximately \$117.3 million if the over-allotment option is exercised in full, including the deferred underwriting fees) available to use for a business combination assuming the successful completion of this offering, we believe that we offer a target business a variety of options such as providing the owners of a target business with shares in a public company and a public means to sell such shares, providing capital for the potential growth and expansion of its operations or strengthening its balance sheet by reducing its debt ratio. Because we will be able to consummate our initial business combination using our cash, debt or equity securities, or a combination of the foregoing, we will have the flexibility to use the most efficient combination that will allow us to tailor the consideration to be paid to the target business to fit its needs and desires. However, since we have no specific business combination under consideration, we have not taken any steps to secure third party financing and there can be no assurance that it will be available to us.

While we may pursue an initial business combination with any business in any industry, commercial sector or location, our initial focus will be on identifying acquisition opportunities in the pharmaceutical and medical devices sectors in the US. Trends supporting the secular growth in the healthcare industry include population aging, rise in chronic diseases and a growing consumer role in care options.

Life sciences including pharma and biopharma:

Trends that fuel growth in these sectors include a rise in chronic diseases, regulatory changes, shift towards preventative care and demand for medicine in emerging economies. Further transforming these sectors are artificial intelligence (AI) and machine learning integration, accelerating the pace of drug discovery and development in a cost-effective manner.

Diagnostic and therapeutic devices, and related medical technologies:

The need for accurate diagnostic testing and monitoring continues to drive growth within innovative medical devices. Additionally, consumers are demanding a superior experience, as smart devices and internet connected devices find new means to solve patient problems. New healthcare technologies are reshaping the diagnosis, care, and management of illness and chronic disease.

It is important to note that the COVID-19 pandemic has had an impact on each of our sectors of interest, leading to an acceleration of several existing trends as well as a creation of new opportunities.

Acquisition Criteria

Consistent with our strategy, we have identified the following general criteria and guidelines which we believe are important in evaluating prospective target businesses. We will use these criteria and guidelines in evaluating acquisition opportunities and will initially target businesses with enterprise values of approximately \$250.0 million to \$1.0 billion, but we may decide to enter into our initial business combination with a target business that does not meet these criteria and guidelines. We currently intend to acquire one or more businesses that we believe has the following characteristics:

- Benefits from a Public Currency and Access to Public Equity Markets. Access to the public equity markets could allow the target company to utilize additional forms of capital, enhancing its ability to pursue accretive acquisitions, high-return capital projects, and/or strengthen its balance sheet and recruit and retain key employees through the use of publicly-traded equity compensation.
- *Has a Strong Competitive Position and Growing Platform.* We will seek to invest in companies that we believe possess not only established business models and sustainable competitive advantages, but also a growing platform for equity investors.
- Operated by a Talented and Incentivized Management Team. We will focus on companies with strong and experienced management teams that desire a significant equity stake in the post-business combination company. We will seek to partner with a management team and/or seller who is well-incentivized and aligned in an effort to create stockholder value.

• Benefits from Our Ability to Uniquely Structure Transaction to Unlock and Maximize Value. We will look for situations where our extensive experience and creativity can architect a win-win solution for both sides of the transaction.

These criteria are not intended to be exhaustive. Any evaluation relating to the merits of a particular initial business combination may be based, to the extent relevant, on these general guidelines as well as other considerations, factors and criteria that from time to time our management may deem relevant. In the event that we decide to enter into our initial business combination with a target business that does not meet the above criteria and guidelines, we will disclose that the target business does not meet the above criteria in our stockholder communications related to our initial business combination, which, as discussed in this prospectus, would be in the form of tender offer documents or proxy solicitation materials that we would file with the SEC.

Initial Business Combination

Nasdaq rules require that we must complete one or more business combinations having an aggregate fair market value of at least 80% of the value of the assets held in the trust account (excluding the deferred underwriting commissions and taxes payable on the interest earned on the trust account) at the time of our signing a definitive agreement in connection with our initial business combination. Our board of directors will make the determination as to the fair market value of our initial business combination. If our board of directors is not able to independently determine the fair market value of our initial business combination, we will obtain an opinion from an independent investment banking firm or another independent entity that commonly renders valuation opinions with respect to the satisfaction of such criteria. While we consider it unlikely that our board of directors will not be able to make an independent determination of the fair market value of our initial business combination, it may be unable to do so if it is less familiar or experienced with the business of a particular target or if there is a significant amount of uncertainty as to the value of a target's assets or prospects. Additionally, pursuant to Nasdaq rules, any initial business combination must be approved by a majority of our independent directors.

We anticipate structuring our initial business combination so that the post-transaction company in which our public stockholders own shares will own or acquire 100% of the equity interests or assets of the target business or businesses. We may, however, structure our initial business combination such that the post-transaction company owns or acquires less than 100% of such interests or assets of the target business in order to meet certain objectives of the prior owners of the target business, the target management team or stockholders or for other reasons, but we will only complete such business combination if the post-transaction company owns or acquires 50% or more of the outstanding voting securities of the target or otherwise acquires a controlling interest in the target sufficient for it not to be required to register as an investment company under the Investment Company Act of 1940, as amended (the "Investment Company Act"). Even if the post-transaction company owns or acquires 50% or more of the voting securities of the target, our stockholders prior to the business combination may collectively own a minority interest in the post-transaction company, depending on valuations ascribed to the target and us in the business combination transaction. For example, we could pursue a transaction in which we issue a substantial number of new shares in exchange for all of the outstanding capital stock, shares or other equity interests of a target. In this case, we would acquire a 100% controlling interest in the target. However, as a result of the issuance of a substantial number of new shares, our stockholders immediately prior to our initial business combination could own less than a majority of our issued and outstanding shares subsequent to our initial business combination. If less than 100% of the equity interests or assets of a target business or businesses are owned or acquired by the post-transaction company, the portion of such business or businesses that is owned or acquired is what will be valued for purposes of the 80% of net assets test. If the business combination involves more than one target business, the 80% of net assets test will be based on the aggregate value of all of the target businesses and we will treat the target businesses together as our initial business combination for purposes of a tender offer or for seeking stockholder approval, as applicable.

To the extent we effect our initial business combination with a company or business that may be financially unstable or in its early stages of development or growth, we may be affected by numerous risks inherent in such company or business. Although our management will endeavor to evaluate the risks inherent in a particular target business, we cannot assure you that we will properly ascertain or assess all significant risk factors.

In evaluating a prospective target business, we expect to conduct a thorough due diligence review which will encompass, among other things, meetings with incumbent management and employees, document reviews, inspection of facilities, as well as a review of financial, operational, legal and other information which will be made available to us.

The time required to select and evaluate a target business and to structure and complete our initial business combination, and the costs associated with this process, are not currently ascertainable with any degree of certainty. Any costs incurred with respect to the identification and evaluation of a prospective target business with which our initial business combination is not ultimately completed will result in our incurring losses and will reduce the funds we can use to complete another business combination.

Sourcing of Potential Initial Business Combination Targets

Certain members of our management team have spent significant portions of their careers working with businesses in the pharmaceutical industry and have developed a wide network of professional services contacts and business relationships in that industry. The members of our board of directors also have significant executive management and public company experience with pharmaceutical and medical devices sectors related companies and bring additional relationships that further broaden our industry network.

We believe that the network of contacts and relationships of our management team will provide us with an important source of acquisition opportunities. In addition, we anticipate that target business candidates will be brought to our attention from various unaffiliated sources, including investment market participants, private equity groups, investment banks, consultants, accounting firms, law firms and large business enterprises.

Members of our management team and our independent directors will directly or indirectly own founder shares and/or placement warrants following this offering and, accordingly, may have a conflict of interest in determining whether a particular target business is an appropriate business with which to effectuate our initial business combination. In particular, Additionally, Mr. Suren Ajjarapu, our Chief Executive Officer and Chairman currently a director of Oceantech Acquisition I Corp., traded on Nasdaq under the symbol "OTECU", a special purpose acquisition company with a class of securities registered under the Exchange Act that focuses its search for a business combination in the leisure marine, yachting and superyachting industries. As of the date of this prospectus, Oceantech Acquisition I Corp. has not entered into a business combination. Further, each of our officers and directors may have a conflict of interest with respect to evaluating a particular business combination if the retention or resignation of any such officers and directors was included by a target business as a condition to any agreement with respect to our initial business combination.

In addition, each of our officers and directors presently has, and any of them in the future may have additional, fiduciary or contractual obligations to other entities pursuant to which such officer or director is or will be required to present a business combination opportunity to such entity. We do not believe, however, that the fiduciary duties or contractual obligations of our officers or directors will materially affect our ability to complete our initial business combination.

In addition, our sponsor and our officers and directors may sponsor or form other special purpose acquisition companies similar to ours or may pursue other business or investment ventures during the period in which we are seeking an initial business combination, and some of such persons are already associated with current special purpose acquisition companies looking for acquisition targets. Similarly, any future financial advisor we engage may work with other special purpose acquisition companies. Such companies, businesses or investments may present additional conflicts of interest in pursuing an initial business combination. However, we do not believe that any such potential conflicts would materially affect our ability to complete our initial business combination.

Corporate Information

Our executive offices are located at 515 Madison Avenue, Suite 8078, New York, New York 10022, and our telephone number is (646) 908-2658.

We are an "emerging growth company," as defined in Section 2(a) of the Securities Act, as modified by the Jumpstart Our Business Startups Act of 2012, or the JOBS Act. As such, we are eligible to take advantage of certain exemptions from various reporting requirements that are applicable to other public companies that are not emerging growth companies including, but not limited to, not being required to comply with the auditor attestation requirements of Section 404 of the Sarbanes-Oxley Act of 2002, or the Sarbanes-Oxley Act, reduced disclosure obligations regarding executive compensation in our periodic reports and proxy statements, and exemptions from the requirements of holding a non-binding advisory vote on executive compensation and stockholder approval of any golden parachute payments not previously approved. If some investors find our securities less attractive as a result, there may be a less active trading market for our securities and the prices of our securities may be more volatile.

In addition, Section 107 of the JOBS Act also provides that an emerging growth company can take advantage of the extended transition period provided in Section 7(a)(2)(B) of the Securities Act for complying with new or revised accounting standards. In other words, an emerging growth company can delay the adoption of certain accounting standards until those standards would otherwise apply to private companies. We intend to take advantage of the benefits of this extended transition period.

We will remain an emerging growth company until the earlier of (1) the last day of the fiscal year (a) following the fifth anniversary of the completion of this offering, (b) in which we have total annual gross revenue of at least \$1.07 billion, or (c) in which we are deemed to be a large accelerated filer, which means the market value of our Class A common stock that is held by non-affiliates exceeds \$700 million as of the prior June 30, and (2) the date on which we have issued more than \$1.0 billion in non-convertible debt securities during the prior three-year period. References herein to emerging growth company will have the meaning associated with it in the JOBS Act. Additionally, we are a "smaller reporting company" as defined in Rule 10(f)(1) of Regulation S-K. Smaller reporting companies may take advantage of certain reduced disclosure obligations, including, among other things, providing only two years of audited financial statements. We will remain a smaller reporting company until the last day of the fiscal year in which (1) the market value of our common stock held by non-affiliates exceeds \$250 million as of the end of the prior June 30th, or (2) our annual revenues exceeded \$100 million during such completed fiscal year and the market value of our common stock held by non-affiliates exceeds \$700 million as of the prior June 30th.

THE OFFERING

In deciding whether to invest in our securities, you should take into account not only the backgrounds of the members of our management team, but also the special risks we face as a blank check company and the fact that this offering is not being conducted in compliance with Rule 419 promulgated under the Securities Act. You will not be entitled to protections normally afforded to investors in Rule 419 blank check offerings. You should carefully consider these and the other risks set forth in the section of this prospectus entitled "Risk Factors."

Securities offered

10,000,000 units (or 11,500,000 units if the underwriters' over-allotment option is exercised in full), at \$10.00 per unit, each unit consisting of:

- one share of Class A common stock; and
- one-half of one redeemable warrant.

Proposed Nasdaq symbols

Units: "AEHAU"

Warrants: "AEHAW"

Class A Common Stock: "AEHA"

Trading commencement and separation of Class A common stock and warrants

The units will begin trading on or promptly after the date of this prospectus. We expect the Class A common stock and warrants comprising the units will begin separate trading on the 52nd day following the date of this prospectus unless the representative informs us of its decision to allow earlier separate trading, subject to our having filed the Current Report on Form 8-K described below and having issued a press release announcing when such separate trading will begin. Once the shares of Class A common stock and warrants commence separate trading, holders will have the option to continue to hold units or separate their units into the component securities. Holders will need to have their brokers contact our transfer agent in order to separate the units into shares of Class A common stock and warrants. No fractional warrants will be issued upon separation of the units and only whole warrants will trade. Accordingly, unless you purchase at least two units, you will not be able to receive or trade a whole warrant.

Separate trading of the Class A common stock and warrants is prohibited until we have filed a Current Report on Form 8-K In no event will the Class A common stock and warrants be traded separately until we have filed a Current Report on Form 8-K with the SEC containing an audited balance sheet reflecting our receipt of the gross proceeds at the closing of this offering. We will file the Current Report on Form 8-K promptly after the closing of this offering, which is anticipated to take place three business days from the date of this prospectus. If the underwriters' over-allotment option is exercised following the initial filing of such Current Report on Form 8-K, a second or amended Current Report on Form 8-K will be filed to provide updated financial information to reflect the exercise of the underwriters' over-allotment option.

Units:

Number outstanding before this offering

and the private placement

Number outstanding after this offering and the private placement

 $10.000.000^{(1)}$

Common stock:

Number outstanding before this offering and the private placement

2,875,000 shares of Class B common stock(2)

Number outstanding after this offering and the private placement

12,600,000 shares of Class A common stock and Class B common $stock^{(1)(3)}$ (or 14,475,000 shares if the underwriters exercise the over-allotment option in full)

Warrants:

Number outstanding before this offering and the private placement

Number of placement warrants to be sold in a private placement simultaneously with this offering

 $5.261.000^{(1)}$

Number of warrants to be outstanding after this offering and the private placement

 $10,261,000^{(1)}$

Exercisability

Each whole warrant is exercisable to purchase one share of our Class A common stock, subject to adjustment as provided herein.

We structured each unit to contain one-half of one whole warrant, with each whole warrant exercisable for one share of Class A common stock, as compared to units issued by some other similar special purpose acquisition companies which contain whole warrants exercisable for one share, in order to reduce the dilutive effect of the warrants upon completion of an initial business combination as compared to units that each contain a whole warrant to purchase one whole share, thus making us, we believe, a more attractive initial business combination partner for target businesses.

- (1) Assumes no exercise of the underwriters' over-allotment option and the forfeiture by our sponsor of 375,000 founder shares.
- (2) Represents 2,875,000 founders' shares classified as Class B common stock, which is convertible into Class A common stock on a one-for-one basis, subject to adjustment as described below adjacent to the caption "Founder shares conversion and anti-dilution rights." The 2,875,000 founders' shares include an aggregate of up to 375,000 founders' shares that are subject to forfeiture by our sponsor depending on the extent to which the underwriters' over-allotment option is exercised.

(3) Comprised of 10,000,000 shares of Class A common stock and 2,500,000 shares of Class B common stock (or founder shares), assuming 375,000 founder shares have been forfeited and 100,000 shares of common stock issuable to the representative of the underwriters and/or its designees. The Class B common stock is convertible into shares of our Class A common stock on a one-for-one basis, subject to adjustment as described below adjacent to the caption "Founder shares conversion and anti-dilution rights."

Exercise price

\$11.50 per share, subject to adjustment as described herein. In addition, if (x) we issue additional shares of Class A common stock or equity-linked securities for capital raising purposes in connection with the closing of our initial business combination at an issue price or effective issue price of less than \$9.20 per share of Class A common stock (with such issue price or effective issue price to be determined in good faith by our board of directors and, in the case of any such issuance to our sponsor or its affiliates, without taking into account any founder shares held by our sponsor or such affiliates, as applicable, prior to such issuance) (the "Newly Issued Price"), (y) the aggregate gross proceeds from such issuances represent more than 60% of the total equity proceeds, and interest thereon, available for the funding of our initial business combination on the date of the consummation of our initial business combination (net of redemptions), and (z) the volume weighted average trading price of our Class A common stock during the 20 trading day period starting on the trading day prior to the day on which we consummate our initial business combination (such price, the "Market Value") is below \$9.20 per share, then the exercise price of the warrants will be adjusted (to the nearest cent) to be equal to 115% of the greater of the Market Value and the Newly Issued Price, and the \$18.00 per share redemption trigger price described below under "Redemption of warrants" will be adjusted (to the nearest cent) to be equal to 180% of the greater of the Market Value and the Newly Issued Price.

Exercise period

The warrants will become exercisable on the later of:

- 30 days after the completion of our initial business combination, and
- 12 months from the closing of this offering;

provided in each case that we have an effective registration statement under the Securities Act covering the shares of Class A common stock issuable upon exercise of the warrants and a current prospectus relating to them is available (or we permit holders to exercise their warrants on a cashless basis under the circumstances specified in the warrant agreement).

We are not registering the shares of Class A common stock issuable upon exercise of the warrants at this time. However, we have agreed that as soon as practicable after the closing of our initial business combination, we will use our best efforts to file with the SEC a registration statement covering the issuance of the shares of Class A common stock issuable upon exercise of the warrants, to cause such registration statement to become effective within 90 days following our initial business combination and to maintain a current prospectus relating to those shares of Class A common stock until the warrants expire or are redeemed, as specified in the warrant agreement. If a registration statement covering the issuance of the shares of Class A common stock issuable upon exercise of the warrants is not effective by the 90th day after the closing of our initial business combination, warrant holders may, until such time as there is an effective registration statement and during any period when we will have failed to maintain an effective registration statement, exercise warrants on a "cashless basis" in accordance with Section 3(a) (9) of the Securities Act or another exemption. If that exemption, or another exemption, is not available, holders will not be able to exercise their warrants on a cashless basis.

The warrants will expire at 5:00 p.m., New York City time, five years after the completion of our initial business combination or earlier upon redemption or liquidation. On the exercise of any warrant, the warrant exercise price will be paid directly to us and not placed in the trust account.

Redemption of warrants

Once the warrants become exercisable, we may redeem the outstanding warrants:

- in whole and not in part;
- at a price of \$0.01 per warrant;
- upon not less than 30 days' prior written notice of redemption given after the warrants become exercisable (the "30-day redemption period") to each warrant holder; and
- if, and only if, the reported last sale price of the Class A common stock equals or exceeds \$18.00 per share (as adjusted for stock splits, stock dividends, rights issuances, reorganizations, recapitalizations and the like) for any 20 trading days within a 30-trading day period commencing once the warrants become exercisable and ending three business days before we send the notice of redemption to the warrant holders.

We will not redeem the warrants as described above unless a registration statement under the Securities Act covering the issuance of the shares of Class A common stock issuable upon exercise of the warrants is then effective and a current prospectus relating to those shares of Class A common stock is available throughout the 30-day redemption period, except if the warrants may be exercised on a cashless basis and such cashless exercise is exempt from registration under the Securities Act. If and when the warrants become redeemable by us, we may not exercise our redemption right if the issuance of shares of common stock upon exercise of the warrants is not exempt from registration or qualification under applicable state blue sky laws or we are unable to effect such registration or qualification. We will use our best efforts to register or qualify such shares of common stock under the blue sky laws of the state of residence in those states in which the warrants were offered by us in this offering.

If we call the warrants for redemption as described above, our management will have the option to require all holders that wish to exercise warrants to do so on a "cashless basis." In determining whether to require all holders to exercise their warrants on a "cashless basis," our management will consider, among other factors, our cash position, the number of warrants that are outstanding and the dilutive effect on our stockholders of issuing the maximum number of shares of Class A common stock issuable upon the exercise of our warrants. In such event, each holder would pay the exercise price by surrendering the warrants for that number of shares of Class A common stock equal to the quotient obtained by dividing (x) the product of the number of shares of Class A common stock underlying the warrants, multiplied by the excess of the "fair market value" (defined below) over the exercise price of the warrants by (y) the fair market value. The "fair market value" for this purpose shall mean the average reported last sale price of the Class A common stock for the 10 trading days ending on the third trading day prior to the date on which the notice of redemption is sent to the holders of warrants.

Please see the section of this prospectus entitled "<u>Description of Securities</u> — Redeemable Warrants — Public Stockholders' Warrants" for additional information.

Founder shares

On June 30, 2021, our sponsor purchased 2,875,000 founder shares for an aggregate purchase price of \$25,000, or approximately \$0.009 per share. Prior to the initial investment in the Company of \$25,000 by our sponsor, the company had no assets, tangible or intangible. The per share purchase price of the founder shares was determined by dividing the amount of cash contributed to the company by the aggregate number of founder shares issued. The number of founder shares issued was determined based on the expectation that the founder shares would represent 20% of the outstanding shares after this offering (excluding the placement warrants and underlying securities, and the representative's shares). As such, our initial stockholders will collectively own 20% of our issued and outstanding shares after this offering. Neither our sponsor nor any of our officers or directors have expressed an intention to purchase any units in this offering. Up to 375,000 founder shares held by our sponsor will be subject to forfeiture by our sponsor depending on the extent to which the underwriters' over-allotment option is exercised so that our initial stockholders will maintain ownership of 20% of our common stock after this offering (excluding the placement warrants and underlying securities, and the representative's shares). We will effect a stock dividend or share contribution prior to this offering should the size of the offering change, in order to maintain such ownership percentage.

The founder shares are identical to the shares of Class A common stock included in the units being sold in this offering, except that:

- the founder shares are shares of Class B common stock that automatically convert into shares of our Class A common stock at the time of the consummation of our initial business combination, on a one-for-one basis, subject to adjustment pursuant to certain anti-dilution rights, as described herein;
- the founder shares are subject to certain transfer restrictions, as described in more detail below;
- our sponsor, officers and directors have entered into, and our director nominees are expected to enter into, a letter agreement with us, pursuant to which they have agreed to (i) waive their redemption rights with respect to any founder shares and public shares held by them in connection with the completion of our initial business combination, (ii) waive their redemption rights with respect to any founder shares and public shares held by them in connection with a stockholder vote to approve an amendment to our amended and restated certificate of incorporation (A) to modify the substance or timing of our obligation to allow redemption in connection with our initial business combination or certain amendments to our charter prior thereto or to redeem 100% of our public shares if we do not complete our initial business combination within 12 months (or up to 18 months, if we extend the time to complete a business combination as described in this prospectus) from the closing of this offering (an "Extension Period") or (B) with respect to any other provision relating to stockholders' rights or pre-initial business combination activity and (iii) waive their rights to liquidating distributions from the trust account with respect to any founder shares held by them if we fail to complete our initial business combination within 12 months (or up to 18 months from the closing of this offering if we extend the period of time to consummate a business combination) from the closing of this offering, although they will be entitled to liquidating distributions from the trust account with respect to any public shares they hold if we fail to complete our initial business combination within the prescribed time frame;

- pursuant to the letter agreement, our sponsor, officers and directors have agreed to vote and our director nominees will agree to vote, any founder shares held by them and any public shares they may acquire during or after this offering (including in open market and privately negotiated transactions) in favor of our initial business combination. If we submit our initial business combination to our public stockholders for a vote, we will complete our initial business combination only if a majority of the then outstanding shares of common stock present and entitled to vote at the meeting to approve the initial business combination are voted in favor of the initial business combination. As a result, in the event that only the minimum number of shares representing a quorum is present at a stockholders' meeting held to vote on our initial business combination, in addition to our initial stockholders' founder shares, we would need only 550,001 or 5.5%, of the 10,000,000 public shares sold in this offering to be voted in favor of an initial business combination (assuming only a quorum is present at the meeting and all shares to be issued to the representative of the underwriters and/or its designees are issued and outstanding and voted in favor of the business combination) in order to have our initial business combination approved (assuming the over-allotment option is not exercised, and that the initial stockholders do not purchase any units in this offering or units or shares in the after-market). In the event that all shares of our outstanding common stock are voted, we would need 3,700,001 or 37.0%, of the 10,000,000 public shares sold in this offering to be voted in favor of an initial business combination (assuming all shares to be issued to the representative of the underwriters and/or its designees are issued and outstanding and voted in favor of the business combination) in order to have our initial business combination approved (assuming the over-allotment option is not exercised, and that the initial stockholders do not purchase any units in this offering or units or shares in the after-market); and
- the founder shares are entitled to registration rights.

Transfer restrictions on founder shares

Our initial stockholders have agreed not to transfer, assign or sell any of their founder shares (or shares of common stock issuable upon conversion thereof) until the earlier to occur of: (A) one year after the completion of our initial business combination and (B) subsequent to our initial business combination, (x) if the reported last sale price of our Class A common stock equals or exceeds \$12.00 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like) for any 20 trading days within any 30-trading day period commencing at least 150 days after our initial business combination, or (y) the date on which we complete a liquidation, merger, capital stock exchange, reorganization or other similar transaction that results in all of our stockholders having the right to exchange their shares of common stock for cash, securities or other property (except as described herein under the section of this prospectus entitled "Principal Stockholders — Restrictions on Transfers of Founder Shares and Placement Warrants"). Any permitted transferees will be subject to the same restrictions and other agreements of our initial stockholders with respect to any founder shares. We refer to such transfer restrictions throughout this prospectus as the lock-up.

Founder shares conversion and anti-dilution rights

The shares of Class B common stock will automatically convert into shares of our Class A common stock at the time of the consummation of our initial business combination on a one-for-one basis, subject to adjustment for stock splits, stock dividends, reorganizations, recapitalizations and the like, and subject to further adjustment as provided herein. In the case that additional shares of Class A common stock, or equity-linked securities, are issued or deemed issued in excess of the amounts offered in this prospectus and related to the closing of the initial business combination, the ratio at which shares of Class B common stock shall convert into shares of Class A common stock will be adjusted (unless the holders of a majority of the outstanding shares of Class B common stock agree to waive such adjustment with respect to any such issuance or deemed issuance) so that the number of shares of Class A common stock issuable upon conversion of all shares of Class B common stock will equal, in the aggregate, on an asconverted basis, 20% of the sum of the total number of all shares of common stock outstanding upon the completion of this offering (excluding the placement warrants and underlying securities, and the representative's shares) plus all shares of Class A common stock and equity-linked securities issued or deemed issued in connection with the initial business combination (excluding any shares or equity-linked securities issued, or to be issued, to any seller in the initial business combination or any private placement-equivalent warrants issued to our sponsor or its affiliates upon conversion of working capital loans made to us). The term "equity-linked securities" refers to any debt or equity securities that are convertible, exercisable or exchangeable for shares of Class A common stock issued in a financing transaction in connection with our initial business combination, including but not limited to a private placement of equity or debt. Securities could be "deemed issued" for purposes of the conversion rate adjustment if such shares are issuable upon the conversion or exercise of convertible securities, warrants or similar securities.

Voting Rights

Placement warrants

Holders of record of the Class A common stock and holders of record of the Class B common stock will vote together as a single class on all matters submitted to a vote of our stockholders, with each share of common stock entitling the holder to one vote, except as required by law.

Our sponsor has committed, pursuant to a written agreement, to purchase an aggregate of 5,261,000 placement warrants (or 5,711,000 if the underwriters' over-allotment option is exercised in full), each exercisable to purchase one share of our Class A common stock at \$11.50 per share, at a price of \$1.00 per warrant (approximately \$5,261,000 in the aggregate or approximately \$5,711,000 in the aggregate if the underwriters' over-allotment option is exercised in full) in a private placement that will occur simultaneously with the closing of this offering. The placement warrants are identical to the warrants sold as part of the units in this offering, subject to limited exceptions. Each placement warrant is exercisable for one share of our Class A common stock at \$11.50 per share.

A portion of the purchase price of the placement warrants will be added to the proceeds from this offering to be held in the trust account such that at the time of closing \$102,000,000 (or \$117,300,000 if the underwriters exercise their over-allotment option in full) will be held in the trust account. There will be no redemption rights or liquidating distributions from the trust account with respect to the founder shares or placement warrants, which will expire worthless if we do not consummate a business combination within 12 months from the closing of this offering or during any Extension Period. Our initial stockholders have agreed to waive their redemption rights with respect to any founder shares (i) in connection with the consummation of a business combination, (ii) in connection with a stockholder vote to amend our amended and restated certificate of incorporation to modify the substance or timing of our obligation to allow redemption in connection with our initial business combination or certain amendments to our amended and restated certificate of incorporation prior thereto, to redeem 100% of our public shares if we do not complete our initial business combination within 12 months from the completion of this offering or during any Extension Period, or with respect to any other provision relating to stockholders' rights or pre-initial business combination activity and (iii) if we fail to consummate a business combination within 12 months from the completion of this offering or during any Extension Period, or if we liquidate prior to the expiration of the 12-month period, or during any Extension Period. However, our initial stockholders will be entitled to redemption rights with respect to any public shares held by them if we fail to consummate a business combination or liquidate within the 18-month period.

Transfer restrictions on placement warrants and underlying securities

Proceeds to be held in trust account

The placement warrants (including the shares of Class A common stock issuable upon exercise of the private placement warrants) will not be transferable, assignable or saleable until 30 days after the consummation of our initial business combination except to permitted transferees. Following such period, the private placement warrants (including the shares of Class A common stock issuable upon exercise of the private placement warrants) will be transferable, assignable or salable, except that the private placement warrants will not trade.

Nasdaq rules provide that at least 90% of the gross proceeds from this offering and the sale of the placement warrants be deposited in a trust account. Of the net proceeds of this offering and the sale of the placement warrants, \$102,000,000, or \$10.20 per unit (\$117,300,000, or \$10.20 per unit, if the underwriters' over-allotment option is exercised in full) will be placed into a trust account in the United States with Continental Stock Transfer & Trust Company, LLC acting as trustee and JP Morgan Chase Bank, N.A. acting as investment manager.

These proceeds include \$3,000,000 (or \$3,450,000 if the underwriters' over-allotment option is exercised in full) in deferred underwriting commissions.

Except with respect to interest earned on the funds held in the trust account that may be released to us to pay our tax obligations and up to \$100,000 of interest that may be used for our dissolution expenses, the proceeds from this offering and the sale of the placement warrants held in the trust account will not be released from the trust account until the earliest to occur of: (a) the completion of our initial business combination, (b) the redemption of any public shares properly submitted in connection with a stockholder vote to amend our amended and restated certificate of incorporation (i) to modify the substance or timing of our obligation to allow redemption in connection with our initial business combination or certain amendments to our amended and restated certificate of incorporation prior thereto or to redeem 100% of our public shares if we do not complete our initial business combination within 12 months from the closing of this offering or during any Extension Period or (ii) with respect to any other provision relating to stockholders' rights or pre-business combination activity, and (c) the redemption of our public shares if we are unable to complete our initial business combination within 12 months from the closing of this offering or during any Extension Period, subject to applicable law. The proceeds deposited in the trust account could become subject to the claims of our creditors, if any, which could have priority over the claims of our public stockholders.

Anticipated expenses and funding sources

Except as described above with respect to the payment of taxes, unless and until we complete our initial business combination, no proceeds held in the trust account will be available for our use. The proceeds held in the trust account will be invested only in U.S. government securities with a maturity of 185 days or less or in money market funds meeting certain conditions under Rule 2a-7 under the Investment Company Act which invest only in direct U.S. government treasury obligations. We will disclose in each quarterly and annual report filed with the SEC prior to our initial business combination whether the proceeds deposited in the trust account are invested in U.S. government treasury obligations or money market funds or a combination thereof. Based upon current interest rates, we expect the trust account to generate approximately \$91,800 of pre-tax interest annually assuming an interest rate of 0.09% per year; however, we can provide no assurances regarding this amount.

Unless and until we complete our initial business combination, we may pay our expenses only from:

- the net proceeds of this offering and the sale of the placement warrants not held in the trust account, which will be approximately \$1,000,000 in working capital after the payment of approximately \$1,261,000 in expenses relating to this offering; and
- any loans or additional investments from our sponsor, members of our management team or their affiliates or other third parties, although they are under no obligation to advance funds or invest in us, and provided that any such loans will not have any claim on the proceeds held in the trust account unless such proceeds are released to us upon completion of an initial business combination.

Conditions to completing our initial business combination

Nasdaq rules require that we must complete one or more business combinations having an aggregate fair market value of at least 80% of the value of the assets held in the trust account (excluding the deferred underwriting commissions and taxes payable on the interest earned on the trust account) at the time of our signing a definitive agreement in connection with our initial business combination. Our board of directors will make the determination as to the fair market value of our initial business combination. If our board of directors is not able to independently determine the fair market value of our initial business combination target, we will obtain an opinion from an independent investment banking firm or another independent entity that commonly renders valuation opinions with respect to the satisfaction of such criteria. While we consider it unlikely that our board of directors will not be able to make an independent determination of the fair market value of our initial business combination target, it may be unable to do so if it is less familiar or experienced with the business of a particular target or if there is a significant amount of uncertainty as to the value of a target's assets or prospects. There is no limitation on our ability to raise funds privately, or through loans in connection with our initial business combination. Additionally, pursuant to Nasdaq rules, any initial business combination must be approved by a majority of our independent directors.

We anticipate structuring our initial business combination either (i) in such a way so that the post-transaction company in which our public stockholders own shares will own or acquire 100% of the equity interests or assets of the target business or businesses, or (ii) in such a way so that the post-transaction company owns or acquires less than 100% of such interests or assets of the target business in order to meet certain objectives of the target management team or stockholders, or for other reasons. However, we will only complete an initial business combination if the post-transaction company owns or acquires 50% or more of the outstanding voting securities of the target or otherwise acquires a controlling interest in the target sufficient for it not to be required to register as an investment company under the Investment Company Act.

Even if the post-transaction company owns or acquires 50% or more of the voting securities of the target, our stockholders prior to the initial business combination may collectively own a minority interest in the post-transaction company, depending on valuations ascribed to the target and us in the initial business combination. For example, we could pursue a transaction in which we issue a substantial number of new shares in exchange for all of the outstanding capital stock of a target.

In this case, we would acquire a 100% controlling interest in the target. However, as a result of the issuance of a substantial number of new shares, our stockholders immediately prior to our initial business combination could own less than a majority of our outstanding shares subsequent to our initial business combination. If less than 100% of the equity interests or assets of a target business or businesses are owned or acquired by the post-transaction company, the portion of such business or businesses that is owned or acquired is what will be taken into account for purposes of Nasdaq's 80% fair market value test. If the initial business combination involves more than one target business, the 80% fair market value test will be based on the aggregate value of all of the transactions and we will treat the target businesses together as the initial business combination for purposes of a tender offer or for seeking stockholder approval, as applicable.

Permitted purchases of public shares and public warrants by our affiliates

If we seek stockholder approval of our initial business combination and we do not conduct redemptions in connection with our initial business combination pursuant to the tender offer rules, our sponsor, initial stockholders, directors, officers, advisors or their affiliates may purchase public shares or public warrants in privately negotiated transactions or in the open market either prior to or following the completion of our initial business combination. There is no limit on the number of shares or warrants our sponsor, initial stockholders, directors, officers, advisors or their affiliates may purchase in such transactions, subject to compliance with applicable law and Nasdaq rules. However, apart from the purchase of the placement warrants, they have no current commitments, plans or intentions to engage in such transactions and have not formulated any terms or conditions for any such transactions. If they engage in such transactions, they will not make any such purchases when they are in possession of any material nonpublic information not disclosed to the seller or if such purchases are prohibited by Regulation M under the Exchange Act. We do not currently anticipate that such purchases, if any, would constitute a tender offer subject to the tender offer rules under the Exchange Act or a going-private transaction subject to the going-private rules under the Exchange Act; however, if the purchasers determine at the time of any such purchases that the purchases are subject to such rules, the purchasers will comply with such rules. We expect that any such purchases will be reported pursuant to Section 13 and Section 16 of the Exchange Act to the extent such purchasers are subject to such reporting requirements.

None of the funds held in the trust account will be used to purchase shares or public warrants in such transactions prior to completion of our initial business combination. See "Proposed Business — Permitted purchases of our securities" for a description of how our sponsor, initial stockholders, directors, officers or any of their affiliates will select which stockholders to purchase securities from in any private transaction

The purpose of any such purchases of shares could be to vote such shares in favor of the initial business combination and thereby increase the likelihood of obtaining stockholder approval of the initial business combination or to satisfy a closing condition in an agreement with a target that requires us to have a minimum net worth or a certain amount of cash at the closing of our initial business combination, where it appears that such requirement would otherwise not be met. The purpose of any such purchases of public warrants could be to reduce the number of public warrants outstanding or to vote such warrants on any matters submitted to the warrant holders for approval in connection with our initial business combination. Any such purchases of our securities may result in the completion of our initial business combination that may not otherwise have been possible. In addition, if such purchases are made, the public "float" of our shares of Class A common stock or warrants may be reduced and the number of beneficial holders of our securities may be reduced, which may make it difficult to maintain or obtain the quotation, listing or trading of our securities on a national securities exchange.

Redemption rights for public stockholders upon completion of our initial business combination We will provide our public stockholders with the opportunity to redeem all or a portion of their public shares upon the completion of our initial business combination at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the trust account as of two business days prior to the consummation of our initial business combination, including interest earned on the funds held in the trust account and not previously released to us to pay our taxes, divided by the number of then outstanding public shares, subject to the limitations described herein. The amount in the trust account is initially anticipated to be \$10.20 per public share, however, there is no guarantee that investors will receive \$10.20 per share upon redemption. The per-share amount we will distribute to investors who properly redeem their shares will not be reduced by the deferred underwriting commissions we will pay to the underwriters. There will be no redemption rights upon the completion of our initial business combination with respect to our warrants. Our sponsor, officers and directors have entered into, and our director nominees are expected to enter into, a letter agreement with us, pursuant to which they have agreed to waive their redemption rights with respect to any founder shares held by them and any public shares they may acquire during or after this offering in connection with the completion of our initial business combination or otherwise.

Manner of conducting redemptions

We will provide our public stockholders with the opportunity to redeem all or a portion of their public shares upon the completion of our initial business combination either (i) in connection with a stockholder meeting called to approve the initial business combination or (ii) by means of a tender offer. The decision as to whether we will seek stockholder approval of a proposed initial business combination or conduct a tender offer will be made by us, solely in our discretion, and will be based on a variety of factors such as the timing of the transaction and whether the terms of the transaction would require us to seek stockholder approval under applicable law or stock exchange listing requirements. Under Nasdaq rules, asset acquisitions and stock purchases would not typically require stockholder approval while direct mergers with our company where we do not survive and any transactions where we issue more than 20% of our outstanding common stock or seek to amend our amended and restated certificate of incorporation would typically require stockholder approval. We may conduct redemptions without a stockholder vote pursuant to the tender offer rules of the SEC unless stockholder approval is required by law or stock exchange listing requirements or we choose to seek stockholder approval for business or other legal reasons. So long as we obtain and maintain a listing for our securities on Nasdaq, we will be required to comply with Nasdaq's stockholder approval rules.

If stockholder approval of the transaction is required by law or stock exchange listing requirements, or we decide to obtain stockholder approval for business or other legal reasons, we will:

- conduct the redemptions in conjunction with a proxy solicitation pursuant to Regulation 14A of
 the Exchange Act, which regulates the solicitation of proxies, and not pursuant to the tender
 offer rules, and
- file proxy materials with the SEC.

If we seek stockholder approval, we will complete our initial business combination only if a majority of the then outstanding shares of common stock present and entitled to vote at the meeting to approve the initial business combination are voted in favor of the initial business combination. A quorum for such meeting will consist of the holders present in person or by proxy of shares of outstanding capital stock of the Company representing a majority of the voting power of all outstanding shares of capital stock of the Company entitled to vote at such meeting. Our initial stockholders will count towards this quorum and, pursuant to the letter agreement, our sponsor, officers and directors have agreed to vote and our director nominees will agree to vote, any founder shares held by them and any public shares acquired by them during or after this offering (including in open market and privately negotiated transactions) in favor of our initial business combination. For purposes of seeking approval of the majority of our outstanding shares of common stock voted, non-votes will have no effect on the approval of our initial business combination once a quorum is obtained. As a result, in addition to our initial stockholders' founder shares, we would need only 550,001 or 5.5%, of the 10,000,000 public shares sold in this offering to be voted in favor of an initial business combination (assuming only the minimum number of shares representing a quorum are voted and all shares to be issued to the representative of the underwriters and/or its designees are issued and outstanding and voted in favor of the business combination) in order to have our initial business combination approved (assuming the over-allotment option is not exercised and that the initial stockholders do not purchase any units in this offering or units or shares in the after). In the event that all shares of our outstanding common stock are voted, we would need 3,700,001 or 37.0%, of the 10,000,000 public shares sold in this offering to be voted in favor of an initial business combination (assuming all shares to be issued to the representative of the underwriters and/or its designees are issued and outstanding and voted in favor of the business combination) in order to have our initial business combination approved (assuming the over-allotment option is not exercised, and that the initial stockholders do not purchase any units in this offering or units or shares in the after-market).

We intend to give approximately 30 days (but not less than 10 days nor more than 60 days) prior written notice of any such meeting, if required, at which a vote shall be taken to approve our initial business combination. These quorum and voting thresholds, and the voting agreements of our initial stockholders, may make it more likely that we will consummate our initial business combination. Each public stockholder may elect to redeem its public shares irrespective of whether they vote for or against the proposed transaction (or abstain from voting).

If a stockholder vote is not required and we do not decide to hold a stockholder vote for business or other legal reasons, we will, pursuant to our amended and restated certificate of incorporation:

- conduct the redemptions pursuant to Rule 13e-4 and Regulation 14E of the Exchange Act, which regulate issuer tender offers, and
- file tender offer documents with the SEC prior to completing our initial business combination which contain substantially the same financial and other information about the initial business combination and the redemption rights as is required under Regulation 14A of the Exchange Act, which regulates the solicitation of proxies.

Such provisions may be amended if approved by holders of at least 65% of our common stock entitled to vote thereon.

Whether or not we maintain our registration under the Exchange Act or our listing on Nasdaq, we will provide our public stockholders with the opportunity to redeem their public shares by one of the two methods listed above. Upon the public announcement of our initial business combination, if we elect to conduct redemptions pursuant to the tender offer rules, we or our sponsor will terminate any plan established in accordance with Rule 10b5-1 to purchase shares of our Class A common stock in the open market, in order to comply with Rule 14e-5 under the Exchange Act.

In the event we conduct redemptions pursuant to the tender offer rules, our offer to redeem will remain open for at least 20 business days, in accordance with Rule 14e-1(a) under the Exchange Act, and we will not be permitted to complete our initial business combination until the expiration of the tender offer period. In addition, we will not redeem any public shares unless our net tangible assets will be at least \$5,000,001 either immediately prior to or upon consummation of our initial business combination and after payment of underwriters' fees and commissions (so that we are not subject to the SEC's "penny stock" rules) or any greater net tangible asset or cash requirement which may be contained in the agreement relating to our initial business combination. If public stockholders tender more shares than we have offered to purchase, we will withdraw the tender offer and not complete the initial business combination.

We may require our public stockholders seeking to exercise their redemption rights, whether they are record holders or hold their shares in "street name," to either tender their certificates to our transfer agent prior to the date set forth in the tender offer documents mailed to such holders, or up to two business days prior to the vote on the proposal to approve the initial business combination in the event we distribute proxy materials, or to deliver their shares to the transfer agent electronically. We believe that this will allow our transfer agent to efficiently process any redemptions without the need for further communication or action from the redeeming public stockholders, which could delay redemptions and result in additional administrative cost. If the proposed initial business combination is not approved and we continue to search for a target company, we will promptly return any certificates delivered, or shares tendered electronically, by public stockholders who elected to redeem their shares.

Our amended and restated certificate of incorporation will provide that in no event will we redeem our public shares unless our net tangible assets are at least \$5,000,001 either immediately prior to or upon consummation of our initial business combination and after payment of underwriters' fees and commissions (so that we are not subject to the SEC's "penny stock" rules) or any greater net tangible asset or cash requirement which may be contained in the agreement relating to our initial business combination. For example, the proposed initial business combination may require: (i) cash consideration to be paid to the target or its owners, (ii) cash to be transferred to the target for working capital or other general corporate purposes or (iii) the retention of cash to satisfy other conditions in accordance with the terms of the proposed initial business combination. In the event the aggregate cash consideration we would be required to pay for all shares of Class A common stock that are validly submitted for redemption plus any amount required to satisfy cash conditions pursuant to the terms of the proposed initial business combination exceed the aggregate amount of cash available to us, we will not complete the initial business combination or redeem any shares, and all shares of Class A common stock submitted for redemption will be returned to the holders thereof.

Redemption rights with respect to founder shares

Our initial stockholders have agreed to waive their redemption rights with respect to any founder shares (i) in connection with the consummation of a business combination, (ii) in connection with a stockholder vote to amend our amended and restated certificate of incorporation to modify the substance or timing of our obligation to allow redemption in connection with our initial business combination or certain amendments to our charter prior thereto, to redeem 100% of our public shares if we do not complete our initial business combination within 12 months from the closing of this offering or during any Extension Period, or with respect to any other provision relating to stockholders' rights or pre-initial business combination activity and (iii) if we fail to consummate a business combination within 12 months from the closing of this offering or during any Extension Period, or if we liquidate prior to the expiration of the 18 month period from the closing of this offering or during any Extension Period. However, our initial stockholders will be entitled to redemption rights with respect to any public shares held by them if we fail to consummate a business combination or liquidate within the 12 months from the closing of this offering or during any Extension Period.

Limitation on redemption rights of stockholders holding 15% or more of the shares sold in this offering if we hold stockholder vote Notwithstanding the foregoing redemption rights, if we seek stockholder approval of our initial business combination and we do not conduct redemptions in connection with our initial business combination pursuant to the tender offer rules, our amended and restated certificate of incorporation will provide that a public stockholder, together with any affiliate of such stockholder or any other person with whom such stockholder is acting in concert or as a "group" (as defined under Section 13 of the Exchange Act), will be restricted from redeeming its shares with respect to more than an aggregate of 15% of the shares sold in this offering, without our prior consent. We believe the restriction described above will discourage stockholders from accumulating large blocks of shares, and subsequent attempts by such holders to use their ability to redeem their shares as a means to force us or our management to purchase their shares at a significant premium to the then-current market price or on other undesirable terms. By limiting our stockholders' ability to redeem to no more than 15% of the shares sold in this offering, we believe we will limit the ability of a small group of stockholders to unreasonably attempt to block our ability to complete our initial business combination, particularly in connection with an initial business combination with a target that requires as a closing condition that we have a minimum net worth or a certain amount of cash. However, we would not be restricting our stockholders' ability to vote all of their shares (including all shares held by those stockholders that hold more than 15% of the shares sold in this offering) for or against our initial business combination.

Redemption rights in connection with proposed amendments to our certificate of incorporation

Our amended and restated certificate of incorporation will provide that any of its provisions related to pre-initial business combination activity (including the requirement to deposit proceeds of this offering and the private placement of units into the trust account and not release such amounts except in specified circumstances, and to provide redemption rights to public stockholders as described herein) may be amended if approved by holders of at least 65% of our common stock entitled to vote thereon, and corresponding provisions of the trust agreement governing the release of funds from our trust account may be amended if approved by holders of at least 65% of our common stock entitled to vote thereon. In all other instances, our amended and restated certificate of incorporation may be amended by holders of a majority of our outstanding common stock entitled to vote thereon, subject to applicable provisions of the Delaware General Corporation Law, or DGCL, or applicable stock exchange rules.

Under our amended and restated certificate of incorporation, we may not issue additional securities that can vote on amendments to our amended and restated certificate of incorporation or on our initial business combination or that would entitle holders thereof to receive funds from the trust account. Our initial stockholders, who will collectively beneficially own 20% of our common stock upon the closing of this offering (assuming they do not purchase any units in this offering and excluding the representative's shares), will participate in any vote to amend our amended and restated certificate of incorporation and/or trust agreement and will have the discretion to vote in any manner they choose. Our sponsor, executive officers, and directors have agreed, and we anticipate that our director nominees will agree, pursuant to a written agreement with us, that they will not propose any amendment to our amended and restated certificate of incorporation (i) to modify the substance or timing of our obligation to allow holders to redeem their shares in connection with an initial business combination or to redeem 100% of our public shares if we do not complete our initial business combination within 12 months from the closing of this offering or during any Extension Period, or (ii) with respect to any other provision relating to stockholders' rights or pre-business combination activity, unless we provide our public stockholders with the opportunity to redeem their shares of Class A common stock upon approval of any such amendment at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the trust account, including interest (which interest shall be net of taxes payable) divided by the number of then outstanding public shares. Our sponsor, officers and directors have entered into a letter agreement with us and we anticipate that our director nominees will enter into a letter agreement with us, pursuant to which they have agreed to waive their redemption rights with respect to any founder shares and any public shares held by them in connection with the completion of our initial business combination.

Release of funds in trust account on closing of our initial business combination

On the completion of our initial business combination, the funds held in the trust account will be used to pay amounts due to any public stockholders who exercise their redemption rights as described above under "Redemption rights for public stockholders upon completion of our initial business combination." We will use the remaining funds to pay the underwriters their deferred underwriting commissions, to pay all or a portion of the consideration payable to the target or owners of the target of our initial business combination and to pay other expenses associated with our initial business combination. If our initial business combination is paid for using equity or debt securities, or not all of the funds released from the trust account are used for payment of the consideration in connection with our initial business combination, we may apply the balance of the cash released to us from the trust account for general corporate purposes, including for maintenance or expansion of operations of post-transaction businesses, the payment of principal or interest due on indebtedness incurred in completing our initial business combination, to fund the purchase of other companies or for working capital.

Ability to Extend Time to Complete Business Combination

We will have until 12 months from the closing of this offering to consummate our initial business combination. However, if we anticipate that we may not be able to consummate our initial business combination within 12 months, we may, by resolution of our board if requested by our sponsor, extend the period of time to consummate a business combination up to two times, each by an additional three months (for a total of up to 18 months to complete a business combination), subject to the sponsor depositing additional funds into the trust account as set out below. Pursuant to the terms of the trust agreement to be entered into between us and Continental Stock Transfer & Trust Company, LLC on the date of this prospectus, in order to extend the time available for us to consummate our initial business combination, our initial shareholders or their affiliates or designees, upon five days advance notice prior to the applicable deadline, must deposit into the trust account for each three-month extension, \$1,000,000, or \$1,150,000 if the underwriters' over-allotment option is exercised in full (\$0.10 per share in either case) on or prior to the date of the applicable deadline, up to an aggregate of \$2,000,000 (or \$2,300,000 if the underwriters' over-allotment option is exercised in full), or approximately \$0.20 per share. Any such payments would be made in the form of a loan. Any such loans will be non-interest bearing and payable upon the consummation of our initial business combination. If we complete our initial business combination, we would repay such loaned amounts out of the proceeds of the trust account released to us. If we do not complete a business combination, we will not repay such loans. Furthermore, the letter agreement with our initial stockholders contains a provision pursuant to which our sponsor has agreed to waive its right to be repaid for such loans out of the funds held in the trust account in the event that we do not complete a business combination. In the event that we receive notice from our sponsor five days prior to the applicable deadline of its wish for us to effect an extension, we intend to issue a press release announcing such intention at least three days prior to the applicable deadline. In addition, we intend to issue a press release the day after the applicable deadline announcing whether or not the funds had been timely deposited. Our sponsor and its affiliates or designees are not obligated to fund the trust account to extend the time for us to complete our initial business combination.

Redemption of public shares and distribution and liquidation if no initial business combination Our amended and restated certificate of incorporation will provide that we will have only 12 months from the closing of this offering, subject to any Extension Period, to complete our initial business combination. We may seek stockholder approval of the amendments to our amended and restated certificate of incorporation and the trust agreement to be entered into between us and Continental Stock Transfer & Trust Company, LLC for any extension beyond 18 months at a meeting called for such purpose. Public stockholders will be offered the opportunity to vote on or redeem their shares in connection with any such extension. If we are unable to complete our initial business combination within such 12-month period, subject to any Extension Period, we will: (i) cease all operations except for the purpose of winding up, (ii) as promptly as reasonably possible but not more than ten business days thereafter, redeem the public shares, at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the trust account including interest earned on the funds held in the trust account and not previously released to us to pay our taxes (less up to \$100,000 of interest to pay dissolution expenses), divided by the number of then outstanding public shares, which redemption will completely extinguish public stockholders' rights as stockholders (including the right to receive further liquidating distributions, if any), subject to applicable law, and (iii) as promptly as reasonably possible following such redemption, subject to the approval of our remaining stockholders and our board of directors, dissolve and liquidate, subject in the case of clauses (ii) and (iii) above to our obligations under Delaware law to provide for claims of creditors and the requirements of other applicable law. There will be no redemption rights or liquidating distributions with respect to our warrants, which will expire worthless if we fail to complete our initial business combination within the 18-month period, subject to any Extension Period.

Our sponsor, officers and directors have entered into a letter agreement with us and we anticipate that our director nominees will agree to enter into with us, a letter agreement, pursuant to which they have agreed to waive their redemption rights with respect to any founder shares if we are forced to liquidate. The underwriters have agreed to waive their rights to their deferred underwriting commission in the event we do not complete our initial business combination and subsequently liquidate and, in such event, the deferred fees will be included with the funds held in the trust account that will be available to fund the redemption of our public shares.

Limited payments to insiders

There will be no finder's fees, reimbursement, consulting fee, non-cash payments, monies in respect of any payment of a loan or other compensation paid by us to our sponsor, officers or directors or any affiliate of our sponsor, officers or directors prior to, or in connection with any services rendered in order to effectuate, the consummation of our initial business combination (regardless of the type of transaction that it is). However, the following payments will be made to our sponsor, officers or directors, or our or their affiliates, none of which will be made from the proceeds of this offering and

the sale of the placement warrants held in the trust account prior to the completion of our initial business combination:

- Repayment of up to an aggregate of \$300,000 in loans made to us by our sponsor to cover offering-related and organizational expenses;
- Payment to Aesther Healthcare Sponsor, LLC, our sponsor, of \$10,000 per month, for up to 18 months, for office space, utilities and secretarial and administrative support;
- Reimbursement for any out-of-pocket expenses related to identifying, investigating and completing an initial business combination; and
- Repayment of non-interest bearing loans which may be made by our sponsor or an affiliate of our sponsor or certain of our officers and directors to finance transaction costs in connection with an intended initial business combination, the terms of which (other than as described above) have not been determined nor have any written agreements been executed with respect thereto. Up to \$1,500,000 of such loans may be convertible into warrants, at a price of \$1.00 per warrant at the option of the lender, upon consummation of our initial business combination. These warrants would be identical to the placement warrants.

Our audit committee will review on a quarterly basis all payments that were made to our sponsor, officers or directors or our or their affiliates.

We will establish and maintain an audit committee, which will be composed entirely of independent directors to, among other things, monitor compliance with the terms described above and the other terms relating to this offering. If any noncompliance is identified, then the audit committee will be charged with the responsibility to immediately take all action necessary to rectify such noncompliance or otherwise to cause compliance with the terms of this offering. For more information, see the section of this prospectus entitled "Management — Committees of the Board of Directors — Audit Committee."

Our sponsor has agreed that it will be liable to us if and to the extent any claims by a third party (other than the independent public accounting firm) for services rendered or products sold to us, or a prospective target business with which we have entered into a written letter of intent, confidentiality or similar agreement or business combination agreement, reduce the amount of funds in the trust account to below the lesser of (i) \$10.20 per public share and (ii) the actual amount per public share held in the trust account as of the date of the liquidation of the trust account, if less than \$10.20 per public share due to reductions in the value of the trust assets, less taxes payable, provided that such liability will not apply to any claims by a third party or prospective target business who executed a waiver of any and all rights to the monies held in the trust account (whether or not such waiver is enforceable) nor will it apply to any claims under our indemnity of the underwriters of this offering against certain liabilities, including liabilities under the Securities Act. However, we have not asked our sponsor to reserve for such indemnification obligations, nor have we independently verified whether our sponsor has sufficient funds to satisfy its indemnity obligations and believe that our sponsor's only assets are securities of our company. Therefore, we cannot assure you that our sponsor would be able to satisfy those obligations. None of our officers or directors will indemnify us for claims by third parties including, without limitation, claims by vendors and prospective target businesses.

Indemnity

Summary of Risk Factors

Our business is subject to numerous risks and uncertainties, including those highlighted in the section titled "Risk Factors," that represent challenges that we face in connection with the successful implementation of our strategy. The occurrence of one or more of the events or circumstances described in the section titled "Risk Factors," alone or in combination with other events or circumstances, may adversely affect our ability to effect a business combination, and may have an adverse effect on our business, cash flows, financial condition and results of operations. Such risks include, but are not limited to:

- we are a newly formed company with no operating history and no revenues;
- our ability to continue as a "going concern";
- we may not be able to complete our initial business combination within the prescribed time frame;
- you will not have any rights or interests in funds from the trust account, except under certain limited circumstances;
- negative interest rate for securities in which we invest the funds held in the trust account;
- our stockholders may be held liable for claims by third parties against us;
- if third parties bring claims against us, the proceeds held in the trust account could be reduced and the per-share redemption amount received by stockholders may be less than \$10.20 per share;
- subsequent to completion of our initial business combination, we may be required to take write-downs or write-offs, restructuring and impairment or other charges;
- conflicts of interest of our sponsor, officers and directors;
- we may have a limited ability to assess the management of a prospective target business;
- our public stockholders may not be afforded an opportunity to vote on our proposed business combination;
- the absence of a redemption threshold may make it possible for us to complete a business combination with which a substantial majority of our stockholders do not agree;
- we may redeem your unexpired warrants prior to their exercise at a time that is disadvantageous to you;
- we may amend the terms of the public warrants in a manner that may be adverse to holders of public warrants with the approval by the holders of at least 65% of the then outstanding public warrants;
- our competitors have advantages over us in seeking business combinations;

- we may be unable to obtain additional financing;
- our warrants may have an adverse effect on the market price of our common stock;
- we may issue additional equity and/or debt securities to complete our initial business combination;
- our sponsor controls a substantial interest in us;
- if we seek stockholder approval of our initial business combination, our sponsor, who controls a substantial interest in us, has agreed to vote in favor of such initial business combination, regardless of how our public stockholders vote;
- the ability of our public stockholders to redeem their shares for cash may make our financial condition unattractive to potential business combination targets, may not allow us to complete the most desirable business combination or optimize our capital structure, and will increase the probability that our initial business combination would be unsuccessful;
- lack of protections normally afforded to investors of blank check companies;
- possibility of losing the ability to redeem all shares equal to or in excess of 15% of our common stock if we seek stockholder approval of our initial business combination and we do not conduct redemptions pursuant to the tender offer rules;
- Nasdaq may delist our securities from trading on its exchange;
- we will likely only be able to complete one business combination with the proceeds of this offering and the sale of the private placement warrants, which will cause us to be solely dependent on a single business which may have a limited number of products or services;
- we are not registering the shares of common stock issuable upon exercise of the warrants sold as part of the units in this offering at this time, and such registration may not be in place when an investor desires to exercise such warrants;
- shares being redeemed and warrants becoming worthless;
- events which may result in the per-share amount held in our trust account dropping below \$10.20 per public share;
- our directors may decide not to enforce the indemnification obligations of our sponsor;
- if, before distributing the proceeds in the trust account to our public stockholders, we file a bankruptcy petition or an involuntary bankruptcy petition is filed against us that is not dismissed, the claims of creditors in such proceeding may have priority over the claims of our stockholders and the per-share amount that would otherwise be received by our stockholders in connection with our liquidation may be reduced;
- because we are not limited to a particular industry or any specific target businesses with which to pursue our initial business combination, you will be unable to ascertain the merits or risks of any particular target business's operations;
- we may seek acquisition opportunities in companies that may be outside of our management's areas of expertise;
- impact of COVID-19 and related risks;
- if we effect our initial business combination with a company with operations or opportunities outside of the United States, we would be subject to a variety of additional risks that may negatively impact our operations; and
- changes in laws or regulations, or a failure to comply with any laws and regulations, tax consequences to business combinations may adversely affect our business, investments and results of operations.

SUMMARY FINANCIAL DATA

The following table summarizes the relevant financial data for our business and should be read with our financial statements, which are included in this prospectus. We have not had any significant operations to date, so only balance sheet data is presented.

	July 12, 2021
Balance Sheet Data:	
Working capital deficiency	\$ (39,798)
Total assets	\$ 89,716
Total liabilities	\$ 64,798
Stockholders' equity	\$ 24,918

RISK FACTORS

An investment in our securities involves a high degree of risk. You should consider carefully all of the risks described below, together with the other information contained in this prospectus, before making a decision to invest in our units. If any of the following events occur, our business, financial condition and operating results may be materially adversely affected. In that event, the trading price of our securities could decline, and you could lose all or part of your investment.

Risks Relating to our Search for, Consummation of, or Inability to Consummate, a Business Combination and Post Business Combination Risks

Our search for a business combination, and any target business with which we ultimately consummate a business combination, may be materially adversely affected by the ongoing coronavirus (COVID-19) pandemic.

The COVID-19 pandemic has resulted in a widespread health crisis that has adversely affected the economies and financial markets worldwide, and the business of any potential target business with which we consummate a business combination could be materially and adversely affected. Furthermore, we may be unable to complete a business combination if continued concerns relating to COVID-19 restrict travel, limit the ability to have meetings with potential investors or the target company's personnel, vendors and services providers are unavailable to negotiate and consummate a transaction in a timely manner. The extent to which COVID-19 impacts our search for a business combination will depend on future developments, which are highly uncertain and cannot be predicted, including new information which may emerge concerning the severity of COVID-19 and the actions to contain COVID-19 or treat its impact, among others. If the disruptions posed by COVID-19 or other matters of global concern continue for an extensive period of time, our ability to consummate a business combination, or the operations of a target business with which we ultimately consummate a business combination, may be materially adversely affected.

Our public stockholders may not be afforded an opportunity to vote on our proposed initial business combination, which means we may complete our initial business combination even though a majority of our public stockholders do not support such a combination.

We may choose not to hold a stockholder vote to approve our initial business combination unless the initial business combination would require stockholder approval under applicable law or stock exchange listing requirements or if we decide to hold a stockholder vote for business or other legal reasons. Except as required by applicable law or stock exchange requirements, the decision as to whether we will seek stockholder approval of a proposed initial business combination or will allow stockholders to sell their shares to us in a tender offer will be made by us, solely in our discretion, and will be based on a variety of factors, such as the timing of the transaction and whether the terms of the transaction would otherwise require us to seek stockholder approval. Accordingly, we may complete our initial business combination even if holders of a majority of our public shares do not approve of the initial business combination we complete. Please see the section of this prospectus entitled "Proposed Business — Stockholders May Not Have the Ability to Approve Our Initial Business Combination" for additional information.

If we seek stockholder approval of our initial business combination, our initial stockholders have agreed to vote and our director nominees will agree to vote, in favor of such initial business combination, regardless of how our public stockholders vote.

Pursuant to the letter agreement, our sponsor, officers and directors have agreed to vote and our director nominees will agree to vote, any founder shares held by them, as well as any public shares they may acquire during or after this offering (including in open market and privately negotiated transactions), in favor of our initial business combination. As a result, in addition to our initial stockholders' founder shares, we would need only 550,001 or 5.5%, of the 10,000,000 public shares sold in this offering to be voted in favor of an initial business combination (assuming only the minimum number of shares representing a quorum are voted) in order to have our initial business combination approved (assuming the over-allotment option is not exercised). In the event that all shares of our outstanding common stock are voted, we would need 3,700,001 or 37.0%, of the 10,000,000 public shares sold in this offering to be voted in favor of an initial business combination (assuming only a quorum is present at the meeting and all shares to be issued to the representative of the underwriters and/or its designees are issued and outstanding and voted in favor of the business combination) in order to have our initial business combination approved (assuming the over-allotment option is not exercised, and that the initial stockholders do not purchase any units in this offering or units or shares in the after-market). Our initial stockholders will own shares representing 20% of our outstanding shares of common stock immediately following the completion of this offering and the private placement (assuming they do not purchase any units in this offering and excluding the representative's shares). Accordingly, if we seek stockholder approval of our initial business combination, the agreement by our initial stockholders to vote in favor of our initial business combination will increase the likelihood that we will receive the requisite stockholder approval for such initial business combination.

Your only opportunity to affect the investment decision regarding a potential business combination will be limited to the exercise of your right to redeem your shares from us for cash, unless we seek stockholder approval of the initial business combination.

At the time of your investment in us, you will not be provided with an opportunity to evaluate the specific merits or risks of our initial business combination. Since our board of directors may complete an initial business combination without seeking stockholder approval, public stockholders may not have the right or opportunity to vote on the initial business combination, unless we seek such stockholder vote. Accordingly, if we do not seek stockholder approval, your only opportunity to affect the investment decision regarding a potential business combination may be limited to exercising your redemption rights within the period of time (which will be at least 20 business days) set forth in our tender offer documents mailed to our public stockholders in which we describe our initial business combination.

The ability of our public stockholders to redeem their shares for cash may make our financial condition unattractive to potential business combination targets, which may make it difficult for us to enter into an initial business combination with a target.

We may seek to enter into an initial business combination agreement with a prospective target that requires as a closing condition that we have a minimum net worth or a certain amount of cash. If too many public stockholders exercise their redemption rights, we would not be able to meet such closing condition and, as a result, would not be able to proceed with the initial business combination. Furthermore, in no event will we redeem our public shares unless our net tangible assets are at least \$5,000,001 either immediately prior to or upon consummation of our initial business combination and after payment of underwriters' fees and commissions (so that we are not subject to the SEC's "penny stock" rules) or any greater net tangible asset or cash requirement which may be contained in the agreement relating to our initial business combination. Consequently, if accepting all properly submitted redemption requests would cause our net tangible assets to be less than \$5,000,001 upon consummation of our initial business combination and after payment of underwriters' fees and commissions or such greater amount necessary to satisfy a closing condition, each as described above, we would not proceed with such redemption and the related business combination and may instead search for an alternate business combination. Prospective targets will be aware of these risks and, thus, may be reluctant to enter into an initial business combination with us.

The ability of our public stockholders to exercise redemption rights with respect to a large number of our shares may not allow us to complete the most desirable business combination or optimize our capital structure.

At the time we enter into an agreement for our initial business combination, we will not know how many stockholders may exercise their redemption rights, and therefore will need to structure the transaction based on our expectations as to the number of shares that will be submitted for redemption. If our initial business combination agreement requires us to use a portion of the cash in the trust account to pay the purchase price, or requires us to have a minimum amount of cash at closing, we will need to reserve a portion of the cash in the trust account to meet such requirements, or arrange for third party financing. In addition, if a larger number of shares are submitted for redemption than we initially expected, we may need to restructure the transaction to reserve a greater portion of the cash in the trust account or arrange for third party financing. Raising additional third-party financing may involve dilutive equity issuances or the incurrence of indebtedness at higher than desirable levels. Furthermore, this dilution would increase to the extent that the anti-dilution provision of the Class B common stock result in the issuance of Class A shares on a greater than one-to-one basis upon conversion of the Class B common stock at the time of the consummation of our business combination. The above considerations may limit our ability to complete the most desirable business combination available to us or optimize our capital structure. The amount of the deferred underwriting commissions payable to the underwriters will not be adjusted for any shares that are redeemed in connection with an initial business combination. The per-share amount we will distribute to stockholders who properly exercise their redemption rights will not be reduced by the deferred underwriting commission and after such redemptions, the per-share value of shares held by non-redeeming stockholders will reflect our obligation to pay the deferred underwriting commissions.

The ability of our public stockholders to exercise redemption rights with respect to a large number of our shares could increase the probability that our initial business combination would be unsuccessful and that you would have to wait for liquidation in order to redeem your stock.

If our initial business combination agreement requires us to use a portion of the cash in the trust account to pay the purchase price, or requires us to have a minimum amount of cash at closing, the probability that our initial business combination would be unsuccessful is increased. If our initial business combination is unsuccessful, you would not receive your pro rata portion of the trust account until we liquidate the trust account.

If you are in need of immediate liquidity, you could attempt to sell your stock in the open market; however, at such time our stock may trade at a discount to the pro rata amount per share in the trust account. In either situation, you may suffer a material loss on your investment or lose the benefit of funds expected in connection with our redemption until we liquidate or you are able to sell your stock in the open market.

Our sponsor paid an aggregate of \$25,000 for the founder shares, or approximately \$0.009 per founder share. As a result of this low initial price, our sponsor, its affiliates and our management team stand to make a substantial profit even if an initial business combination subsequently declines in value or is unprofitable for our public stockholders.

As a result of the low acquisition cost of our founder shares, our sponsor, its affiliates and our management team could make a substantial profit even if we select and consummate an initial business combination with an acquisition target that subsequently declines in value or is unprofitable for our public stockholders. Thus, such parties may have more of an economic incentive for us to enter into an initial business combination with a riskier, weaker-performing or financially unstable business, or an entity lacking an established record of revenues or earnings, than would be the case if such parties had paid the full offering price for their founder shares.

The value of the founder shares and private placement shares following completion of our initial business combination is likely to be substantially higher than the price paid for them, even if the trading price of our shares at such time is substantially less than \$10.00 per share.

The SPAC model may not fully align the interests of our sponsor and management with those of our public stockholders. Upon the closing of this offering, assuming no exercise of the underwriters' over-allotment option, our sponsor will have invested in us an aggregate of \$5,286,000, comprised of the \$25,000 purchase price for the founder shares and the \$5,261,000 purchase price for the private placement warrants. Assuming a trading price of \$10.00 per share upon consummation of our initial business combination, the 2,500,000 founder shares would have an aggregate implied value of \$25.0 million. Even if the trading price of our shares was as low as approximately \$2.11 per share, the value of the founder shares would be equal to our sponsor's initial investment in us. As a result, our sponsor is likely to be able to recoup its investment in us and make a substantial profit on that investment, even if our public shares have lost significant value. Accordingly, our management team, which owns interests in our sponsor, may have an economic incentive that differs from that of the public stockholders to pursue and consummate an initial business combination rather than to liquidate and to return all of the cash in the trust to the public stockholders, even if that business combination were with a riskier or less-established target business. For the foregoing reasons, you should consider our management team's financial incentive to complete an initial business combination when evaluating whether to redeem your shares prior to or in connection with the initial business combination.

The requirement that we complete our initial business combination within the prescribed time frame may give potential target businesses leverage over us in negotiating an initial business combination and may decrease our ability to conduct due diligence on potential business combination targets as we approach our dissolution deadline, which could undermine our ability to complete our initial business combination on terms that would produce value for our stockholders.

Any potential target business with which we enter into negotiations concerning an initial business combination will be aware that we must complete our initial business combination within 12 months from the closing of this offering or during any Extension Period. Consequently, such target business may obtain leverage over us in negotiating an initial business combination, knowing that if we do not complete our initial business combination with that particular target business, we may be unable to complete our initial business combination with any target business. This risk will increase as we get closer to the timeframe described above. In addition, we may have limited time to conduct due diligence and may enter into our initial business combination on terms that we would have rejected upon a more comprehensive investigation.

We may not be able to complete our initial business combination within the prescribed time frame, in which case we would cease all operations except for the purpose of winding up and we would redeem our public shares and liquidate, in which case our public stockholders may only receive \$10.20 per share, or less than such amount in certain circumstances, and our warrants will expire worthless.

Our amended and restated certificate of incorporation will provide that we must complete our initial business combination within 12 months from the closing of this offering or during any Extension Period. We may not be able to find a suitable target business and complete our initial business combination within such time period. Our ability to complete our initial business combination may be negatively impacted by general market conditions, volatility in the capital and debt markets and the other risks described herein. For example, if the outbreak of COVID-19 continues to grow both in the U.S. and globally and, while the extent of the impact of the outbreak on us will depend on future developments, it could limit our ability to complete our initial business combination, including as a result of increased market volatility, decreased market liquidity and third-party financing being unavailable on terms acceptable to us or at all. Additionally, the outbreak of COVID-19 may negatively impact businesses we may seek to acquire.

We may seek stockholder approval of the amendments to our amended and restated certificate of incorporation and the trust agreement to be entered into between us and Continental Stock Transfer & Trust Company, LLC for any extension beyond 18 months at a meeting called for such purpose. Public stockholders will be offered the opportunity to vote on or redeem their shares in connection with any such extension.

If we have not completed our initial business combination within such time period, we will: (i) cease all operations except for the purpose of winding up, (ii) as promptly as reasonably possible but not more than ten business days thereafter, redeem the public shares, at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the trust account including interest earned on the funds held in the trust account and not previously released to us to pay our taxes (less up to \$100,000 of interest to pay dissolution expenses), divided by the number of then outstanding public shares, which redemption will completely extinguish public stockholders' rights as stockholders (including the right to receive further liquidating distributions, if any), subject to applicable law, and (iii) as promptly as reasonably possible following such redemption, subject to the approval of our remaining stockholders and our board of directors, dissolve and liquidate, subject in the case of clauses (ii) and (iii) above to our obligations under Delaware law to provide for claims of creditors and the requirements of other applicable law. In such case, our public stockholders may only receive \$10.20 per share, or less in certain circumstances, and our warrants will expire worthless. In certain circumstances, our public stockholders may receive less than \$10.20 per share on the redemption of their shares. See "— If third parties bring claims against us, the proceeds held in the trust account could be reduced and the per-share redemption amount received by stockholders may be less than \$10.20 per share" and other risk factors below.

If we seek stockholder approval of our initial business combination, our sponsor, directors, officers and their affiliates may elect to purchase shares or warrants from public stockholders, which may influence a vote on a proposed initial business combination and reduce the public "float" of our Class A common stock.

If we seek stockholder approval of our initial business combination and we do not conduct redemptions in connection with our initial business combination pursuant to the tender offer rules, our sponsor, directors, officers or their affiliates may purchase public shares or public warrants or a combination thereof in privately negotiated transactions or in the open market either prior to or following the completion of our initial business combination, although they are under no obligation to do so. However, they have no current commitments, plans or intentions to engage in such transactions and have not formulated any terms or conditions for any such transactions. None of the funds in the trust account will be used to purchase shares or public warrants in such transactions.

Such a purchase may include a contractual acknowledgement that such stockholder, although still the record holder of our shares is no longer the beneficial owner thereof and therefore agrees not to exercise its redemption rights. In the event that our sponsor, directors, officers or their affiliates purchase shares in privately negotiated transactions from public stockholders who have already elected to exercise their redemption rights or submitted a proxy to vote against our initial business combination, such selling stockholders would be required to revoke their prior elections to redeem their shares and any proxy to vote against our initial business combination. The price per share paid in any such transaction may be different than the amount per share a public stockholder would receive if it elected to redeem its shares in connection with our initial business combination. The purpose of such purchases could be to vote such shares in favor of the initial business combination and thereby increase the likelihood of obtaining stockholder approval of the initial business combination, or to satisfy a closing condition in an agreement with a target that requires us to have a minimum net worth or a certain amount of cash at the closing of our initial business combination, where it appears that such requirement would otherwise not be met. The purpose of any such purchases of public warrants could be to reduce the number of public warrants outstanding or to vote such warrants on any matters submitted to the warrant holders for approval in connection with our initial business combination.

Any such purchases of our securities may result in the completion of our initial business combination that may not otherwise have been possible. We expect that any such purchases will be reported pursuant to Section 13 and Section 16 of the Exchange Act to the extent such purchasers are subject to such reporting requirements.

In addition, if such purchases are made, the public "float" of our Class A common stock or public warrants and the number of beneficial holders of our securities may be reduced, possibly making it difficult to obtain or maintain the quotation, listing or trading of our securities on a national securities exchange.

If a stockholder fails to receive notice of our offer to redeem our public shares in connection with our initial business combination, or fails to comply with the procedures for tendering its shares, such shares may not be redeemed.

We will comply with the tender offer rules or proxy rules, as applicable, when conducting redemptions in connection with our initial business combination. Despite our compliance with these rules, if a stockholder fails to receive our tender offer or proxy materials, as applicable, such stockholder may not become aware of the opportunity to redeem its shares. In addition, proxy materials or tender offer documents, as applicable, that we will furnish to holders of our public shares in connection with our initial business combination will describe the various procedures that must be complied with in order to validly tender or redeem public shares. For example, we may require our public stockholders seeking to exercise their redemption rights, whether they are record holders or hold their shares in "street name," to either deliver their stock certificates to our transfer agent prior to the date set forth in the tender offer documents mailed to such holders, or up to two business days prior to the vote on the proposal to approve the initial business combination in the event we distribute proxy materials, or to deliver their shares to the transfer agent electronically. In the event that a stockholder fails to comply with these or any other procedures disclosed in the proxy or tender offer materials, as applicable, its shares may not be redeemed. See the section of this prospectus entitled "Proposed Business — Redemption Rights for Public Stockholders upon Completion of our Initial Business Combination — Tendering Stock Certificates in Connection with a Tender Offer or Redemption Rights."

Your only opportunity to affect the investment decision regarding a potential business combination will be limited to the exercise of your right to redeem your shares from us for cash, unless we seek stockholder approval of the initial business combination.

At the time of your investment in us, you will not be provided with an opportunity to evaluate the specific merits or risks of our initial business combination. Since our board of directors may complete an initial business combination without seeking stockholder approval, public stockholders may not have the right or opportunity to vote on the initial business combination, unless we seek such stockholder vote. Accordingly, if we do not seek stockholder approval, your only opportunity to affect the investment decision regarding a potential business combination may be limited to exercising your redemption rights within the period of time (which will be at least 20 business days) set forth in our tender offer documents mailed to our public stockholders in which we describe our initial business combination.

You will not have any rights or interests in funds from the trust account, except under certain limited circumstances. To liquidate your investment, therefore, you may be forced to sell your public shares or warrants, potentially at a loss.

Our public stockholders will be entitled to receive funds from the trust account only upon the earliest to occur of: (i) our completion of an initial business combination, and then only in connection with those shares of Class A common stock that such stockholder properly elected to redeem, subject to the limitations described herein, (ii) the redemption of any public shares properly submitted in connection with a stockholder vote to amend our amended and restated certificate of incorporation (A) to modify the substance or timing of our obligation to allow redemption in connection with our initial business combination or certain amendments to our amended and restated certificate of incorporation prior thereto or to redeem 100% of our public shares if we do not complete our initial business combination within 12 months from the closing of this offering or during any Extension Period or (B) with respect to any other provision relating to stockholders' rights or pre-initial business combination activity, and (iii) the redemption of our public shares if we are unable to complete an initial business combination within 12 months from the closing of this offering or during any Extension Period, subject to applicable law and as further described herein. In no other circumstances will a public stockholder have any right or interest of any kind in the trust account. Holders of warrants will not have any right to the proceeds held in the trust account with respect to the warrants. Accordingly, to liquidate your investment, you may be forced to sell your public shares or warrants, potentially at a loss.

You will not be entitled to protections normally afforded to investors of many other blank check companies.

Since the net proceeds of this offering and the sale of the placement warrants are intended to be used to complete an initial business combination with a target business that has not been identified, we may be deemed to be a "blank check" company under the United States securities laws. However, because we will have net tangible assets in excess of \$5,000,000 upon the successful completion of this offering and the sale of the placement warrants and will file a Current Report on Form 8-K, including an audited balance sheet demonstrating this fact, we are exempt from rules promulgated by the SEC to protect investors in blank check companies, such as Rule 419. Accordingly, investors will not be afforded the benefits or protections of those rules. Among other things, this means our units will be immediately tradable and we will have a longer period of time to complete our business combination than do companies subject to Rule 419. Moreover, if this offering were subject to Rule 419, that rule would prohibit the release of any interest earned on funds held in the trust account to us unless and until the funds in the trust account were released to us in connection with our completion of an initial business combination. For a more detailed comparison of our offering to offerings that comply with Rule 419, please see the section of this prospectus entitled "Proposed Business — Comparison of This Offering to Those of Blank Check Companies Subject to Rule 419."

If we seek stockholder approval of our initial business combination and we do not conduct redemptions pursuant to the tender offer rules, and if you or a "group" of stockholders are deemed to hold in excess of 15% of our Class A common stock, you will lose the ability to redeem all such shares in excess of 15% of our Class A common stock.

If we seek stockholder approval of our initial business combination and we do not conduct redemptions in connection with our initial business combination pursuant to the tender offer rules, our amended and restated certificate of incorporation will provide that a public stockholder, together with any affiliate of such stockholder or any other person with whom such stockholder is acting in concert or as a "group" (as defined under Section 13 of the Exchange Act), will be restricted from seeking redemption rights with respect to more than an aggregate of 15% of the shares sold in this offering without our prior consent, which we refer to as the "Excess Shares." However, we would not be restricting our stockholders' ability to vote all of their shares (including Excess Shares) for or against our initial business combination. Your inability to redeem the Excess Shares will reduce your influence over our ability to complete our initial business combination and you could suffer a material loss on your investment in us if you sell Excess Shares in open market transactions. Additionally, you will not receive redemption distributions with respect to the Excess Shares if we complete our initial business combination. And as a result, you will continue to hold that number of shares exceeding 15% and, in order to dispose of such shares, would be required to sell your shares in open market transactions, potentially at a loss.

Because of our limited resources and the significant competition for business combination opportunities, it may be more difficult for us to complete our initial business combination. If we are unable to complete our initial business combination, our public stockholders may receive only approximately \$10.20 per share on our redemption of our public shares, or less than such amount in certain circumstances, and our warrants will expire worthless.

We expect to encounter intense competition from other entities having a business objective similar to ours, including private investors (which may be individuals or investment partnerships), other blank check companies and other entities competing for the types of businesses we intend to acquire. Many of these individuals and entities are well-established and have extensive experience in identifying and effecting, directly or indirectly, acquisitions of companies operating in or providing services to various industries. Many of these competitors possess greater technical, human and other resources or more industry knowledge than we do, and our financial resources will be relatively limited when contrasted with those of many of these competitors. While we believe there are numerous target businesses we could potentially acquire with the net proceeds of this offering and the sale of the placement warrants, our ability to compete with respect to the acquisition of certain target businesses that are sizable will be limited by our available financial resources. This inherent competitive limitation gives others an advantage in pursuing the acquisition of certain target businesses. Furthermore, because we are obligated to pay cash for the shares of Class A common stock which our public stockholders redeem in connection with our initial business combination, target companies will be aware that this may reduce the resources available to us for our initial business combination. This may place us at a competitive disadvantage in successfully negotiating and completing an initial business combination. If we are unable to complete our initial business combination, our public stockholders may receive only approximately \$10.20 per share on the liquidation of our trust account, or less in certain circumstances, and our warrants will expire worthless. In certain circumstances, our public stockholders may receive less than \$10.20 per share upon our liquidation. See "— If third parties bring claims against us, the proceeds held i

If the net proceeds of this offering and the sale of the placement warrants not being held in the trust account are insufficient to allow us to operate for at least the 12 months and during any Extension Period following the closing of this offering, we may be unable to complete our initial business combination, in which case our public stockholders may only receive \$10.20 per share, or less than such amount in certain circumstances, and our warrants will expire worthless.

The funds available to us outside of the trust account to fund our working capital requirements may not be sufficient to allow us to operate for at least the 12 months following the closing of this offering, or up to 18 months, subject to the Extension Period, assuming that our initial business combination is not completed during that time. We believe that, upon the closing of this offering, the funds available to us outside of the trust account will be sufficient to allow us to operate for at least the 12 months following such closing, or up to 18 months, subject to the Extension Period; however, we cannot assure you that our estimate is accurate.

Of the funds available to us, we could use a portion of the funds available to us to pay fees to consultants to assist us with our search for a target business. We could also use a portion of the funds as a down payment or to fund a "no-shop" provision (a provision in letters of intent or merger agreements designed to keep target businesses from "shopping" around for transactions with other companies or investors on terms more favorable to such target businesses) with respect to a particular proposed initial business combination, although we do not have any current intention to do so. If we entered into a letter of intent or merger agreement where we paid for the right to receive exclusivity from a target business and were subsequently required to forfeit such funds (whether as a result of our breach or otherwise), we might not have sufficient funds to continue searching for, or conduct due diligence with respect to, a target business. If we are unable to complete our initial business combination, our public stockholders may receive only approximately \$10.20 per share on the liquidation of our trust account, or less in certain circumstances, and our warrants will expire worthless. In certain circumstances, our public stockholders may receive less than \$10.20 per share upon our liquidation. See "— If third parties bring claims against us, the proceeds held in the trust account could be reduced and the per-share redemption amount received by stockholders may be less than \$10.20 per share" and other risk factors herein.

If the net proceeds of this offering and the sale of the placement warrants not being held in the trust account are insufficient, it could limit the amount available to fund our search for a target business or businesses and complete our initial business combination and we will depend on loans from our sponsor or management team to fund our search for an initial business combination, to pay our taxes and to complete our initial business combination. If we are unable to obtain these loans, we may be unable to complete our initial business combination.

Of the net proceeds of this offering and the sale of the placement warrants, only approximately \$1,000,000 will be available to us initially outside the trust account to fund our working capital requirements. In the event that our offering expenses exceed our estimate of \$1,261,000, we may fund such excess with funds not to be held in the trust account. In such case, the amount of funds we intend to be held outside the trust account would decrease by a corresponding amount. The amount held in the trust account will not be impacted as a result of such increase or decrease. Conversely, in the event that the offering expenses are less than our estimate of \$1,261,000, the amount of funds we intend to be held outside the trust account would increase by a corresponding amount. If we are required to seek additional capital, we would need to borrow funds from our sponsor, management team or other third parties to operate or may be forced to liquidate. None of our sponsor, members of our management team nor any of their affiliates is under any obligation to advance funds to us in such circumstances. Any such advances would be repaid only from funds held outside the trust account or from funds released to us upon completion of our initial business combination. Up to \$1,500,000 of such loans may be convertible into warrants, at a price of \$1.00 per warrant at the option of the lender, upon consummation of our initial business combination. The warrants would be identical to the placement warrants. Prior to the completion of our initial business combination, we do not expect to seek loans from parties other than our sponsor or an affiliate of our sponsor as we do not believe third parties will be willing to loan such funds and provide a waiver against any and all rights to seek access to funds in our trust account. If we are unable to obtain these loans, we may be unable to complete our initial business combination. If we are unable to complete our initial business combination because we do not have sufficient funds available to us, we will be forced to cease operations and liquidate the trust account. Consequently, our public stockholders may only receive approximately \$10.20 per share on our redemption of our public shares, or less in certain circumstances, and our warrants will expire worthless. In certain circumstances, our public stockholders may receive less than \$10.20 per share on the redemption of their shares. See "— If third parties bring claims against us, the proceeds held in the trust account could be reduced and the per-share redemption amount received by stockholders may be less than \$10.20 per share" and other risk factors below.

Subsequent to the completion of our initial business combination, we may be required to take write-downs or write-offs, restructuring and impairment or other charges that could have a significant negative effect on our financial condition, results of operations and our share price, which could cause you to lose some or all of your investment.

Even if we conduct extensive due diligence on a target business with which we combine, we cannot assure you that this diligence will surface all material issues that may be present inside a particular target business, that it would be possible to uncover all material issues through a customary amount of due diligence, or that factors outside of the target business and outside of our control will not later arise. As a result of these factors, we may be forced to later write-down or write-off assets, restructure our operations, or incur impairment or other charges that could result in our reporting losses. Even if our due diligence successfully identifies certain risks, unexpected risks may arise and previously known risks may materialize in a manner not consistent with our preliminary risk analysis. Even though these charges may be non-cash items and not have an immediate impact on our liquidity, the fact that we report charges of this nature could contribute to negative market perceptions about us or our securities. In addition, charges of this nature may cause us to violate net worth or other covenants to which we may be subject as a result of assuming pre-existing debt held by a target business or by virtue of our obtaining debt financing to partially finance the initial business combination. Accordingly, any stockholders who choose to remain stockholders following the initial business combination could suffer a reduction in the value of their shares. Such stockholders are unlikely to have a remedy for such reduction in value unless they are able to successfully claim that the reduction was due to the breach by our officers or directors of a duty of care or other fiduciary duty owed to them, or if they are able to successfully bring a private claim under securities laws that the proxy solicitation or tender offer materials, as applicable, relating to the initial business combination constituted an actionable material misstatement or omission.

If third parties bring claims against us, the proceeds held in the trust account could be reduced and the per-share redemption amount received by stockholders may be less than \$10.20 per share.

Our placing of funds in the trust account may not protect those funds from third-party claims against us. Although we will seek to have all vendors, service providers, prospective target businesses and other entities with which we do business execute agreements with us waiving any right, title, interest or claim of any kind in or to any monies held in the trust account for the benefit of our public stockholders, such parties may not execute such agreements, or even if they execute such agreements they may not be prevented from bringing claims against the trust account, including, but not limited to, fraudulent inducement, breach of fiduciary responsibility or other similar claims, as well as claims challenging the enforceability of the waiver, in each case in order to gain advantage with respect to a claim against our assets, including the funds held in the trust account. If any third party refuses to execute an agreement waiving such claims to the monies held in the trust account, our management will perform an analysis of the alternatives available to it and will only enter into an agreement with a third party that has not executed a waiver if management believes that such third party's engagement would be significantly more beneficial to us than any alternative. Certain parties, including MaloneBailey LLP ("Malone"), our independent registered public accounting firm, and the underwriters of this offering, will not execute agreements with us waiving such claims to the monies held in the trust account.

Examples of possible instances where we may engage a third party that refuses to execute a waiver include the engagement of a third party consultant whose particular expertise or skills are believed by management to be significantly superior to those of other consultants that would agree to execute a waiver or in cases where management is unable to find a service provider willing to execute a waiver. In addition, there is no guarantee that such entities will agree to waive any claims they may have in the future as a result of, or arising out of, any negotiations, contracts or agreements with us and will not seek recourse against the trust account for any reason. Upon redemption of our public shares, if we are unable to complete our initial business combination within the prescribed timeframe, or upon the exercise of a redemption right in connection with our initial business combination, we will be required to provide for payment of claims of creditors that were not waived that may be brought against us within the 10 years following redemption. Accordingly, the per-share redemption amount received by public stockholders could be less than the \$10.20 per share initially held in the trust account, due to claims of such creditors. Pursuant to the letter agreement, the form of which is filed as Exhibit 10.1 to the registration statement of which this prospectus forms a part, our sponsor has agreed that it will be liable to us if and to the extent any claims by a third party for services rendered or products sold to us, or a prospective target business with which we have entered into a written letter of intent, confidentiality or similar agreement or business combination agreement, reduce the amount of funds in the trust account to below the lesser of (i) \$10.20 per public share and (ii) the actual amount per public share held in the trust account as of the date of the liquidation of the trust account, if less than \$10.20 per share due to reductions in the value of the trust assets, less taxes payable, provided that such liability will not apply to any claims by a third party or prospective target business who executed a waiver of any and all rights to the monies held in the trust account (whether or not such waiver is enforceable) nor will it apply to any claims under our indemnity of the underwriters of this offering against certain liabilities, including liabilities under the Securities Act. However, we have not asked our sponsor to reserve for such indemnification obligations, nor have we independently verified whether our sponsor has sufficient funds to satisfy its indemnity obligations and believe that our sponsor's only assets are securities of our company. Therefore, we cannot assure you that our sponsor would be able to satisfy those obligations. None of our officers or directors will indemnify us for claims by third parties including, without limitation, claims by vendors and prospective target businesses.

Our directors may decide not to enforce the indemnification obligations of our sponsor, resulting in a reduction in the amount of funds in the trust account available for distribution to our public stockholders.

In the event that the proceeds in the trust account are reduced below the lesser of (i) \$10.20 per share and (ii) the actual amount per share held in the trust account as of the date of the liquidation of the trust account if less than \$10.20 per share due to reductions in the value of the trust assets, in each case net of the interest which may be withdrawn to pay taxes, and our sponsor asserts that it is unable to satisfy its obligations or that it has no indemnification obligations related to a particular claim, our independent directors would determine whether to take legal action against our sponsor to enforce its indemnification obligations.

While we currently expect that our independent directors would take legal action on our behalf against our sponsor to enforce its indemnification obligations to us, it is possible that our independent directors in exercising their business judgment and subject to their fiduciary duties may choose not to do so in any particular instance if, for example, the cost of such legal action is deemed by the independent directors to be too high relative to the amount recoverable or if the independent directors determine that a favorable outcome is not likely. If our independent directors choose not to enforce these indemnification obligations, the amount of funds in the trust account available for distribution to our public stockholders may be reduced below \$10.20 per share.

We may not have sufficient funds to satisfy indemnification claims of our directors and executive officers.

We have agreed to indemnify our officers and directors to the fullest extent permitted by law. However, our officers and directors have agreed to waive any right, title, interest or claim of any kind in or to any monies in the trust account and to not seek recourse against the trust account for any reason whatsoever. Accordingly, any indemnification provided will be able to be satisfied by us only if (i) we have sufficient funds outside of the trust account or (ii) we consummate an initial business combination. Our obligation to indemnify our officers and directors may discourage stockholders from bringing a lawsuit against our officers or directors for breach of their fiduciary duty. These provisions also may have the effect of reducing the likelihood of derivative litigation against our officers and directors, even though such an action, if successful, might otherwise benefit us and our stockholders. Furthermore, a stockholder's investment may be adversely affected to the extent we pay the costs of settlement and damage awards against our officers and directors pursuant to these indemnification provisions.

If, after we distribute the proceeds in the trust account to our public stockholders, we file a bankruptcy petition or an involuntary bankruptcy petition is filed against us that is not dismissed, a bankruptcy court may seek to recover such proceeds, and we and our board may be exposed to claims of punitive damages.

If, after we distribute the proceeds in the trust account to our public stockholders, we file a bankruptcy petition or an involuntary bankruptcy petition is filed against us that is not dismissed, any distributions received by stockholders could be viewed under applicable debtor/creditor and/or bankruptcy laws as either a "preferential transfer" or a "fraudulent conveyance." As a result, a bankruptcy court could seek to recover some or all amounts received by our stockholders. In addition, our board of directors may be viewed as having breached its fiduciary duty to our creditors and/or having acted in bad faith, thereby exposing itself and us to claims of punitive damages, by paying public stockholders from the trust account prior to addressing the claims of creditors.

If, before distributing the proceeds in the trust account to our public stockholders, we file a bankruptcy petition or an involuntary bankruptcy petition is filed against us that is not dismissed, the claims of creditors in such proceeding may have priority over the claims of our stockholders and the per-share amount that would otherwise be received by our stockholders in connection with our liquidation may be reduced.

If, before distributing the proceeds in the trust account to our public stockholders, we file a bankruptcy petition or an involuntary bankruptcy petition is filed against us that is not dismissed, the proceeds held in the trust account could be subject to applicable bankruptcy law, and may be included in our bankruptcy estate and subject to the claims of third parties with priority over the claims of our stockholders. To the extent any bankruptcy claims deplete the trust account, the per-share amount that would otherwise be received by our stockholders in connection with our liquidation may be reduced.

If we are deemed to be an investment company under the Investment Company Act, we may be required to institute burdensome compliance requirements and our activities may be restricted, which may make it difficult for us to complete our initial business combination.

If we are deemed to be an investment company under the Investment Company Act, our activities may be restricted, including:

- restrictions on the nature of our investments; and
- restrictions on the issuance of securities, each of which may make it difficult for us to complete our initial business combination.

In addition, we may have imposed upon us burdensome requirements, including:

- registration as an investment company with the SEC;
- adoption of a specific form of corporate structure; and
- reporting, record keeping, voting, proxy and disclosure requirements and other rules and regulations that we are currently not subject to.

In order not to be regulated as an investment company under the Investment Company Act, unless we can qualify for an exclusion, we must ensure that we are engaged primarily in a business other than investing, reinvesting or trading in securities and that our activities do not include investing, reinvesting, owning, holding or trading "investment securities" constituting more than 40% of our total assets (exclusive of U.S. government securities and cash items) on an unconsolidated basis. Our business will be to identify and complete an initial business combination and thereafter to operate the post-transaction business or assets for the long term. We do not plan to buy businesses or assets with a view to resale or profit from their resale. We do not plan to buy unrelated businesses or assets or to be a passive investor.

We do not believe that our anticipated principal activities will subject us to the Investment Company Act. To this end, the proceeds held in the trust account may only be invested in United States "government securities" within the meaning of Section 2(a)(16) of the Investment Company Act having a maturity of 185 days or less or in money market funds meeting certain conditions under Rule 2a-7 promulgated under the Investment Company Act which invest only in direct U.S. government treasury obligations. Pursuant to the trust agreement, the trustee is not permitted to invest in other securities or assets. By restricting the investment of the proceeds to these instruments, and by having a business plan targeted at acquiring and growing businesses for the long term (rather than on buying and selling businesses in the manner of a merchant bank or private equity fund), we intend to avoid being deemed an "investment company" within the meaning of the Investment Company Act. This offering is not intended for persons who are seeking a return on investments in government securities or investment securities. The trust account is intended as a holding place for funds pending the earliest to occur of: (i) the completion of our initial business combination; (ii) the redemption of any public shares properly submitted in connection with a stockholder vote to amend our amended and restated certificate of incorporation (A) to modify the substance or timing of our obligation to allow redemption in connection with our initial business combination or certain amendments to our charter prior thereto or to redeem 100% of our public shares if we do not complete our initial business combination within 12 months from the closing of this offering or during any Extension Period or (B) with respect to any other provision relating to stockholders' rights or pre-initial business combination activity; or (iii) absent an initial business combination within 12 months from the closing of this offering or during any Extension Period, our return of the funds held in the trust account to our public stockholders as part of our redemption of the public shares. If we do not invest the proceeds as discussed above, we may be deemed to be subject to the Investment Company Act. If we were deemed to be subject to the Investment Company Act, compliance with these additional regulatory burdens would require additional expenses for which we have not allotted funds and may hinder our ability to complete an initial business combination or may result in our liquidation. If we are unable to complete our initial business combination, our public stockholders may receive only approximately \$10.20 per share on the liquidation of our trust account, or less in certain circumstances, and our warrants will expire worthless.

Our stockholders may be held liable for claims by third parties against us to the extent of distributions received by them upon redemption of their shares.

Under the DGCL, stockholders may be held liable for claims by third parties against a corporation to the extent of distributions received by them in a dissolution. The pro rata portion of our trust account distributed to our public stockholders upon the redemption of our public shares in the event we do not complete our initial business combination within 12 months from the closing of this offering or during any Extension Period, may be considered a liquidating distribution under Delaware law. If a corporation complies with certain procedures set forth in Section 280 of the DGCL intended to ensure that it makes reasonable provision for all claims against it, including a 60-day notice period during which any third-party claims can be brought against the corporation, a 90-day period during which the corporation may reject any claims brought, and an additional 150-day waiting period before any liquidating distributions are made to stockholders, any liability of stockholders with respect to a liquidating distribution is limited to the lesser of such stockholder's pro rata share of the claim or the amount distributed to the stockholder, and any liability of the stockholder would be barred after the third anniversary of the dissolution. However, it is our intention to redeem our public shares as soon as reasonably possible following the 18th month following the closing of this offering, or during any Extension Period in the event we do not complete our initial business combination and, therefore, we do not intend to comply with the foregoing procedures.

Because we will not be complying with Section 280, Section 281(b) of the DGCL requires us to adopt a plan, based on facts known to us at such time that will provide for our payment of all existing and pending claims or claims that may be potentially brought against us within the 10 years following our dissolution. However, because we are a blank check company, rather than an operating company, and our operations will be limited to searching for prospective target businesses to acquire, the only likely claims to arise would be from our vendors (such as lawyers, investment bankers, etc.) or prospective target businesses. If our plan of distribution complies with Section 281(b) of the DGCL, any liability of stockholders with respect to a liquidating distribution is limited to the lesser of such stockholder's pro rata share of the claim or the amount distributed to the stockholder, and any liability of the stockholder would likely be barred after the third anniversary of the dissolution. We cannot assure you that we will properly assess all claims that may be potentially brought against us. As such, our stockholders could potentially be liable for any claims to the extent of distributions received by them (but no more) and any liability of our stockholders may extend beyond the third anniversary of such date. Furthermore, if the pro rata portion of our trust account distributed to our public stockholders upon the redemption of our public shares in the event we do not complete our initial business combination within 12 months following the closing of this offering or during any Extension Period is not considered a liquidating distribution under Delaware law and such redemption distribution is deemed to be unlawful (potentially due to the imposition of legal proceedings that a party may bring or due to other circumstances that are currently unknown), then pursuant to Section 174 of the DGCL, the statute of limitations for claims of creditors could then be six years after the unlawful redemption distribution, instead

We may not hold an annual meeting of stockholders until after the consummation of our initial business combination, which could delay the opportunity for our stockholders to elect directors.

In accordance with Nasdaq corporate governance requirements, we are not required to hold an annual meeting until no later than one year after our first fiscal year end following our listing on Nasdaq.

Under Section 211(b) of the DGCL, we are, however, required to hold an annual meeting of stockholders for the purposes of electing directors in accordance with our bylaws unless such election is made by written consent in lieu of such a meeting. We may not hold an annual meeting of stockholders to elect new directors prior to the consummation of our initial business combination, and thus we may not be in compliance with Section 211(b) of the DGCL, which requires an annual meeting. Therefore, if our stockholders want us to hold an annual meeting prior to the consummation of our initial business combination, they may attempt to force us to hold one by submitting an application to the Delaware Court of Chancery in accordance with Section 211(c) of the DGCL.

The grant of registration rights to our sponsor and holders of our placement warrants may make it more difficult to complete our initial business combination, and the future exercise of such rights may adversely affect the market price of our Class A common stock.

Pursuant to an agreement to be entered into concurrently with the issuance and sale of the securities in this offering, our sponsor, holders of our placement warrants, and their permitted transferees can demand that we register the placement warrants, the Class A common stock issuable upon exercise of the placement warrants, and the Class A common stock issuable upon conversion of the founder shares, and holders of warrants that may be issued upon conversion of working capital loans may demand that we register such warrants or the Class A common stock issuable upon exercise of such warrants. We will bear the cost of registering these securities. The registration and availability of such a significant number of securities for trading in the public market may have an adverse effect on the market price of our Class A common stock. In addition, the existence of the registration rights may make our initial business combination more costly or difficult to conclude. This is because the stockholders of the target business may increase the equity stake they seek in the combined entity or ask for more cash consideration to offset the negative impact on the market price of our Class A common shares that is expected when the common stock owned by our sponsor, holders of our placement warrants or holders of our working capital loans or their respective permitted transferees are registered.

Because we are neither limited to evaluating a target business in a particular industry sector nor have we selected any specific target businesses with which to pursue our initial business combination, you will be unable to ascertain the merits or risks of any particular target business's operations.

We will seek to complete an initial business combination with companies in the pharmaceutical and medical devices sectors but may also pursue other business combination opportunities, except that we will not, under our amended and restated certificate of incorporation, be permitted to effectuate our initial business combination with another blank check company or similar company with nominal operations. Because we have not yet selected or approached any specific target business with respect to a business combination, there is no basis to evaluate the possible merits or risks of any particular target business's operations, results of operations, cash flows, liquidity, financial condition or prospects. To the extent we complete our initial business combination, we may be affected by numerous risks inherent in the business operations with which we combine. For example, if we combine with a financially unstable business or an entity lacking an established record of sales or earnings, we may be affected by the risks inherent in the business and operations of a financially unstable or a development stage entity. Although our officers and directors will endeavor to evaluate the risks inherent in a particular target business, we cannot assure you that we will properly ascertain or assess all of the significant risk factors or that we will have adequate time to complete due diligence. Furthermore, some of these risks may be outside of our control and leave us with no ability to control or reduce the chances that those risks will adversely impact a target business. We also cannot assure you that an investment in our units will ultimately prove to be more favorable to investors than a direct investment, if such opportunity were available, in a business combination target. Accordingly, any stockholders who choose to remain stockholders following our initial business combination could suffer a reduction in the value of their securities. Such stockholders are unlikely to have a remedy for such reduction in value unless they are able to successfully claim that the reduction was due to the breach by our officers or directors of a duty of care or other fiduciary duty owed to them, or if they are able to successfully bring a private claim under securities laws that the proxy solicitation or tender offer materials, as applicable, relating to the business combination contained an actionable material misstatement or material omission.

We may seek business combination opportunities in industries or sectors which may or may not be outside of our management's area of expertise.

Although we intend to focus on identifying pharmaceutical and medical devices sectors companies, we will consider an initial business combination outside of our management's area of expertise if an initial business combination candidate is presented to us and we determine that such candidate offers an attractive business combination opportunity for our company or we are unable to identify a suitable candidate in the pharmaceutical and medical devices sectors after having expanded a reasonable amount of time and effort in an attempt to do so. Although our management will endeavor to evaluate the risks inherent in any particular business combination candidate, we cannot assure you that we will adequately ascertain or assess all of the significant risk factors. We also cannot assure you that an investment in our units will not ultimately prove to be less favorable to investors in this offering than a direct investment, if an opportunity were available, in an initial business combination candidate. In the event we elect to pursue a business combination outside of the areas of our management's expertise, our management's expertise may not be directly applicable to its evaluation or operation, and the information contained in this prospectus regarding the areas of our management's expertise would not be relevant to an understanding of the business that we elect to acquire. As a result, our management may not be able to adequately ascertain or assess all of the significant risk factors. Accordingly, any stockholders who choose to remain stockholders following our initial business combination could suffer a reduction in the value of their shares. Such stockholders are unlikely to have a remedy for such reduction in value.

Although we have identified general criteria and guidelines that we believe are important in evaluating prospective target businesses, we may enter into our initial business combination with a target that does not meet such criteria and guidelines, and as a result, the target business with which we enter into our initial business combination may not have attributes entirely consistent with our general criteria and guidelines.

Although we have identified general criteria and guidelines for evaluating prospective target businesses, it is possible that a target business with which we enter into our initial business combination will not have all of these positive attributes. If we complete our initial business combination with a target that does not meet some or all of these guidelines, such combination may not be as successful as a combination with a business that does meet all of our general criteria and guidelines. In addition, if we announce a prospective business combination with a target that does not meet our general criteria and guidelines, a greater number of stockholders may exercise their redemption rights, which may make it difficult for us to meet any closing condition with a target business that requires us to have a minimum net worth or a certain amount of cash. In addition, if stockholder approval of the transaction is required by applicable law or stock exchange requirements, or we decide to obtain stockholder approval for business or other legal reasons, it may be more difficult for us to attain stockholder approval of our initial business combination if the target business does not meet our general criteria and guidelines. If we are unable to complete our initial business combination, our public stockholders may receive only approximately \$10.20 per share on the liquidation of our trust account, or less in certain circumstances, and our warrants will expire worthless. In certain circumstances, our public stockholders may receive less than \$10.20 per share on the redemption of their shares. See "— If third parties bring claims against us, the proceeds held in the trust account could be reduced and the per-share redemption amount received by stockholders may be less than \$10.20 per share" and other risk factors herein.

We may seek business combination opportunities with a financially unstable business or an entity lacking an established record of revenue, cash flow or earnings, which could subject us to volatile revenues, cash flows or earnings or difficulty in retaining key personnel.

To the extent we complete our initial business combination with a financially unstable business or an entity lacking an established record of revenues or earnings, we may be affected by numerous risks inherent in the operations of the business with which we combine. These risks include volatile revenues or earnings and difficulties in obtaining and retaining key personnel. Although our officers and directors will endeavor to evaluate the risks inherent in a particular target business, we may not be able to properly ascertain or assess all of the significant risk factors and we may not have adequate time to complete due diligence. Furthermore, some of these risks may be outside of our control and leave us with no ability to control or reduce the chances that those risks will adversely impact a target business.

We are not required to obtain a fairness opinion and consequently, you may have no assurance from an independent source that the price we are paying for the business is fair to our company from a financial point of view.

Unless we complete our initial business combination with an affiliated entity or our board cannot independently determine the fair market value of the target business or businesses, we are not required to obtain an opinion from an independent investment banking firm or another independent entity that commonly renders valuation opinions that the price we are paying is fair to our company from a financial point of view. If no opinion is obtained, our stockholders will be relying on the judgment of our board of directors, who will determine fair market value based on standards generally accepted by the financial community. Such standards used will be disclosed in our proxy materials or tender offer documents, as applicable, related to our initial business combination.

Resources could be wasted in researching business combinations that are not completed, which could materially adversely affect subsequent attempts to locate and acquire or merge with another business. If we are unable to complete our initial business combination, our public stockholders may receive only approximately \$10.20 per share, or less than such amount in certain circumstances, on the liquidation of our trust account and our warrants will expire worthless.

We anticipate that the investigation of each specific target business and the negotiation, drafting and execution of relevant agreements, disclosure documents and other instruments will require substantial management time and attention and substantial costs for accountants, attorneys, consultants and others. If we decide not to complete a specific initial business combination, the costs incurred up to that point for the proposed transaction likely would not be recoverable. Furthermore, if we reach an agreement relating to a specific target business, we may fail to complete our initial business combination for any number of reasons including those beyond our control. Any such event will result in a loss to us of the related costs incurred which could materially adversely affect subsequent attempts to locate and acquire or merge with another business.

If we are unable to complete our initial business combination, our public stockholders may receive only approximately \$10.20 per share on the liquidation of our trust account, or less in certain circumstances, and our warrants will expire worthless. In certain circumstances, our public stockholders may receive less than \$10.20 per share on the redemption of their shares. See "— If third parties bring claims against us, the proceeds held in the trust account could be reduced and the per-share redemption amount received by stockholders may be less than \$10.20 per share" and other risk factors herein.

We may have a limited ability to assess the management of a prospective target business and, as a result, may effect our initial business combination with a target business whose management may not have the skills, qualifications or abilities to manage a public company, which could, in turn, negatively impact the value of our stockholders' investment in us.

When evaluating the desirability of effecting our initial business combination with a prospective target business, our ability to assess the target business's management may be limited due to a lack of time, resources or information. Our assessment of the capabilities of the target's management, therefore, may prove to be incorrect and such management may lack the skills, qualifications or abilities we suspected. Should the target's management not possess the skills, qualifications or abilities necessary to manage a public company, the operations and profitability of the post-combination business may be negatively impacted. Accordingly, any stockholders who choose to remain stockholders following the initial business combination could suffer a reduction in the value of their shares. Such stockholders are unlikely to have a remedy for such reduction in value.

We may issue notes or other debt securities, or otherwise incur substantial debt, to complete an initial business combination, which may adversely affect our leverage and financial condition and thus negatively impact the value of our stockholders' investment in us.

Although we have no commitments as of the date of this prospectus to issue any notes or other debt securities, or to otherwise incur outstanding debt following this offering, we may choose to incur substantial debt to complete our initial business combination. We have agreed that we will not incur any indebtedness unless we have obtained from the lender a waiver of any right, title, interest or claim of any kind in or to the monies held in the trust account. As such, no issuance of debt will affect the per-share amount available for redemption from the trust account. Nevertheless, the incurrence of debt could have a variety of negative effects, including:

- default and foreclosure on our assets if our operating revenues after an initial business combination are insufficient to repay our debt obligations;
- acceleration of our obligations to repay the indebtedness even if we make all principal and interest payments when due if we breach certain covenants that require the maintenance of certain financial ratios or reserves without a waiver or renegotiation of that covenant;
- our immediate payment of all principal and accrued interest, if any, if the debt security is payable on demand;
- our inability to obtain necessary additional financing if the debt security contains covenants restricting our ability to obtain such financing while the debt security is outstanding;
- our inability to pay dividends on our common stock;
- using a substantial portion of our cash flow to pay principal and interest on our debt, which will reduce the funds available for dividends on our common stock if declared, our ability to pay expenses, make capital expenditures and acquisitions, and fund other general corporate purposes;
- limitations on our flexibility in planning for and reacting to changes in our business and in the industry in which we operate;
- increased vulnerability to adverse changes in general economic, industry and competitive conditions and adverse changes in government regulation;
- limitations on our ability to borrow additional amounts for expenses, capital expenditures, acquisitions, debt service requirements, and execution of our strategy; and
- other disadvantages compared to our competitors who have less debt.

We may issue our shares to investors in connection with our initial business combination at a price that is less than the prevailing market price of our shares at that time.

In connection with our initial business combination, we may issue shares to investors in private placement transactions (so-called PIPE transactions) at a price of \$10.20 per share or which approximates the per-share amounts in our trust account at such time. The purpose of such issuances will be to enable us to provide sufficient liquidity to the post-business combination entity. The price of the shares we issue may therefore be less, and potentially significantly less, than the market price for our shares at such time.

We may only be able to complete one business combination with the proceeds of this offering, the sale of the placement warrants, which will cause us to be solely dependent on a single business which may have a limited number of services and limited operating activities. This lack of diversification may negatively impact our operating results and profitability.

Of the net proceeds from this offering and the sale of the placement warrants, \$103,000,000 (or \$118,300,000 if the underwriters' over-allotment option is exercised in full) will be available to complete our initial business combination and pay related fees and expenses (which includes up to \$3,000,000, or up to \$3,450,000 if the over-allotment option is exercised in full, for the payment of deferred underwriting commissions).

We may effectuate our initial business combination with a single target business or multiple target businesses simultaneously or within a short period of time. However, we may not be able to effectuate our initial business combination with more than one target business because of various factors, including the existence of complex accounting issues and the requirement that we prepare and file pro forma financial statements with the SEC that present operating results and the financial condition of several target businesses as if they had been operated on a combined basis. By completing our initial business combination with only a single entity, our lack of diversification may subject us to numerous economic, competitive and regulatory developments. Further, we would not be able to diversify our operations or benefit from the possible spreading of risks or offsetting of losses, unlike other entities which may have the resources to complete several business combinations in different industries or different areas of a single industry. In addition, we intend to focus our search for an initial business combination in a single industry. Accordingly, the prospects for our success may be:

- solely dependent upon the performance of a single business, property or asset, or
- dependent upon the development or market acceptance of a single or limited number of products, processes or services.

This lack of diversification may subject us to numerous economic, competitive and regulatory risks, any or all of which may have a substantial adverse impact upon the particular industry in which we may operate subsequent to our initial business combination.

We may attempt to simultaneously complete business combinations with multiple prospective targets, which may hinder our ability to complete our initial business combination and give rise to increased costs and risks that could negatively impact our operations and profitability.

If we determine to simultaneously acquire several businesses that are owned by different sellers, we will need for each of such sellers to agree that our purchase of its business is contingent on the simultaneous closings of the other business combinations, which may make it more difficult for us, and delay our ability, to complete our initial business combination. We do not, however, intend to purchase multiple businesses in unrelated industries in conjunction with our initial business combination. With multiple business combinations, we could also face additional risks, including additional burdens and costs with respect to possible multiple negotiations and due diligence investigations (if there are multiple sellers) and the additional risks associated with the subsequent assimilation of the operations and services or products of the acquired companies in a single operating business. If we are unable to adequately address these risks, it could negatively impact our profitability and results of operations.

We may attempt to complete our initial business combination with a private company about which little information is available, which may result in an initial business combination with a company that is not as profitable as we suspected, if at all.

In pursuing our initial business combination strategy, we may seek to effectuate our initial business combination with a privately held company. Very little public information generally exists about private companies, and we could be required to make our decision on whether to pursue a potential initial business combination on the basis of limited information, which may result in an initial business combination with a company that is not as profitable as we suspected, if at all.

Our management may not be able to maintain control of a target business after our initial business combination.

We may structure an initial business combination so that the post-transaction company in which our public stockholders own shares will own less than 100% of the equity interests or assets of a target business, but we will only complete such business combination if the post-transaction company owns or acquires 50% or more of the outstanding voting securities of the target or otherwise acquires a controlling interest in the target sufficient for us not to be required to register as an investment company under the Investment Company Act. We will not consider any transaction that does not meet such criteria. Even if the post-transaction company owns 50% or more of the voting securities of the target, our stockholders prior to the initial business combination may collectively own a minority interest in the post business combination company, depending on valuations ascribed to the target and us in the initial business combination. For example, we could pursue a transaction in which we issue a substantial number of new shares of Class A common stock in exchange for all of the outstanding capital stock of a target. In this case, we would acquire a 100% interest in the target. However, as a result of the issuance of a substantial number of new shares of common stock, our stockholders immediately prior to such transaction could own less than a majority of our outstanding shares of common stock subsequent to such transaction. In addition, other minority stockholders may subsequently combine their holdings resulting in a single person or group obtaining a larger share of the company's stock than we initially acquired. Accordingly, this may make it more likely that our management will not be able to maintain our control of the target business. We cannot provide assurance that, upon loss of control of a target business, new management will possess the skills, qualifications or abilities necessary to profitably operate such business.

We do not have a specified maximum redemption threshold. The absence of such a redemption threshold may make it possible for us to complete an initial business combination with which a substantial majority of our stockholders do not agree.

Our amended and restated certificate of incorporation will not provide a specified maximum redemption threshold, except that in no event will we redeem our public shares unless our net tangible assets are at least \$5,000,001 either immediately prior to or upon consummation of our initial business combination and after payment of underwriters' fees and commissions (such that we are not subject to the SEC's "penny stock" rules) or any greater net tangible asset or cash requirement which may be contained in the agreement relating to our initial business combination. As a result, we may be able to complete our initial business combination even though a substantial majority of our public stockholders do not agree with the transaction and have redeemed their shares or, if we seek stockholder approval of our initial business combination and do not conduct redemptions in connection with our initial business combination pursuant to the tender offer rules, have entered into privately negotiated agreements to sell their shares to our sponsor, officers, directors or their affiliates. In the event the aggregate cash consideration we would be required to pay for all shares of Class A common stock that are validly submitted for redemption plus any amount required to satisfy cash conditions pursuant to the terms of the proposed initial business combination exceed the aggregate amount of cash available to us, we will not complete the initial business combination or redeem any shares, all shares of Class A common stock submitted for redemption will be returned to the holders thereof, and we instead may search for an alternate business combination.

In order to effectuate an initial business combination, blank check companies have, in the recent past, amended various provisions of their charters and other governing instruments, including their warrant agreements. We cannot assure you that we will not seek to amend our amended and restated certificate of incorporation or governing instruments in a manner that will make it easier for us to complete our initial business combination that our stockholders may not support.

In order to effectuate an initial business combination, blank check companies have, in the recent past, amended various provisions of their charters and modified governing instruments, including their warrant agreements. For example, blank check companies have amended the definition of business combination, increased redemption thresholds and extended the time to consummate an initial business combination and, with respect to their warrants, amended their warrant agreements to require the warrants to be exchanged for cash and/or other securities. Amending our amended and restated certificate of incorporation with respect to any provisions related to pre-initial business combination activity (including the requirement to deposit proceeds of this offering and the private placement of units into the trust account and not release such amounts except in specified circumstances, and to provide redemption rights to public stockholders as described herein) will require the approval of holders of 65% of our common stock entitled to vote thereon, amending other provisions of our amended and restated certificate of incorporation will require the approval of holders of a majority of our common stock and amending our warrant agreement will require a vote of holders of at least a majority of the public warrants, as the case may be (which may include public warrants acquired by our sponsor or its affiliates in this offering or thereafter in the open market). In addition, our amended and restated certificate of incorporation requires us to provide our public stockholders with the opportunity to redeem their public shares for cash if we propose an amendment to our amended and restated certificate of incorporation (A) to modify the substance or timing of our obligation to allow redemption in connection with our initial business combination or certain amendments to our charter prior thereto or to redeem 100% of our public shares if we do not complete our initial business combination within 12 months from the closing of this offer

To the extent any such amendments would be deemed to fundamentally change the nature of any securities offered through this registration statement, we would register, or seek an exemption from registration for, the affected securities. We cannot assure you that we will not seek to amend our charter or governing instruments or extend the time to consummate an initial business combination in order to effectuate our initial business combination.

The provisions of our amended and restated certificate of incorporation that relate to our pre-business combination activity (and corresponding provisions of the agreement governing the release of funds from our trust account), including an amendment to permit us to withdraw funds from the trust account such that the per share amount investors will receive upon any redemption or liquidation is substantially reduced or eliminated, may be amended with the approval of holders of at least 65% of our common stock, which is a lower amendment threshold than that of some other blank check companies. It may be easier for us, therefore, to amend our amended and restated certificate of incorporation and the trust agreement to facilitate the completion of an initial business combination that some of our stockholders may not support.

Our amended and restated certificate of incorporation will provide that any of its provisions related to pre-initial business combination activity (including the requirement to deposit proceeds of this offering and the sale of the placement warrants into the trust account and not release such amounts except in specified circumstances, and to provide redemption rights to public stockholders as described herein and including to permit us to withdraw funds from the trust account such that the per share amount investors will receive upon any redemption or liquidation is substantially reduced or eliminated) may be amended if approved by holders of at least 65% of our common stock entitled to vote thereon, and corresponding provisions of the trust agreement governing the release of funds from our trust account may be amended if approved by holders of at least 65% of our common stock entitled to vote thereon. In all other instances, our amended and restated certificate of incorporation may be amended by holders of a majority of our outstanding common stock entitled to vote thereon, subject to applicable provisions of the DGCL or applicable stock exchange rules. We may not issue additional securities that can vote on amendments to our amended and restated certificate of incorporation. Our initial stockholders, who will collectively beneficially own 20% of our common stock upon the closing of this offering (assuming they do not purchase any units in this offering and excluding the representative's shares), will participate in any vote to amend our amended and restated certificate of incorporation and/or trust agreement and will have the discretion to vote in any manner they choose. As a result, we may be able to amend the provisions of our amended and restated certificate of incorporation which govern our pre-initial business combination behavior more easily than some other blank check companies, and this may increase our ability to complete an initial business combination with which you do not agree. Our stockholders m

Our sponsor, officers and directors have agreed, pursuant to a written agreement with us, that they will not propose any amendment to our amended and restated certificate of incorporation (i) to modify the substance or timing of our obligation to allow redemption in connection with our initial business combination or certain amendments to our charter prior thereto or to redeem 100% of our public shares if we do not complete our initial business combination within 12 months from the closing of this offering or during any Extension Period or (ii) with respect to any other provision relating to stockholders' rights or pre-initial business combination activity, unless we provide our public stockholders with the opportunity to redeem their shares of Class A common stock upon approval of any such amendment at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the trust account, divided by the number of then outstanding public shares. These agreements are contained in a letter agreement that we have entered into with our sponsor, officers and directors and our director nominees. Our stockholders are not parties to, or third-party beneficiaries of, these agreements and, as a result, will not have the ability to pursue remedies against our sponsor, officers or directors for any breach of these agreements. As a result, in the event of a breach, our stockholders would need to pursue a stockholder derivative action, subject to applicable law.

Changes in the market for directors and officers liability insurance could make it more difficult and more expensive for us to negotiate and complete an initial business combination.

In recent months, the market for directors and officers liability insurance for special purpose acquisition companies has changed in ways adverse to us and our management team. Fewer insurance companies are offering quotes for directors and officers liability coverage, the premiums charged for such policies have generally increased and the terms of such policies have generally become less favorable. These trends may continue into the future.

The increased cost and decreased availability of directors and officers liability insurance could make it more difficult and more expensive for us to negotiate and complete an initial business combination. In order to obtain directors and officers liability insurance or modify its coverage as a result of becoming a public company, the post-business combination entity might need to incur greater expense and/or accept less favorable terms. Furthermore, any failure to obtain adequate directors and officers liability insurance could have an adverse impact on the post-business combination's ability to attract and retain qualified officers and directors.

In addition, after completion of any initial business combination, our directors and officers could be subject to potential liability from claims arising from conduct alleged to have occurred prior to such initial business combination. As a result, in order to protect our directors and officers, the post-business combination entity may need to purchase additional insurance with respect to any such claims ("run-off insurance"). The need for run-off insurance would be an added expense for the post-business combination entity and could interfere with or frustrate our ability to consummate an initial business combination on terms favorable to our investors.

We may be unable to obtain additional financing to complete our initial business combination or to fund the operations and growth of a target business, which could compel us to restructure or abandon a particular business combination.

We have not selected any specific business combination target, but intend to target businesses larger than we could acquire with the net proceeds of this offering and the sale of the placement warrants. As a result, we may be required to seek additional financing to complete such proposed initial business combination. We cannot assure you that such financing will be available on acceptable terms, if at all. To the extent that additional financing proves to be unavailable when needed to complete our initial business combination, we would be compelled to either restructure the transaction or abandon that particular business combination and seek an alternative target business candidate. Further, the amount of additional financing we may be required to obtain could increase as a result of future growth capital needs for any particular transaction, the depletion of the available net proceeds in search of a target business, the obligation to repurchase for cash a significant number of shares from stockholders who elect redemption in connection with our initial business combination and/or the terms of negotiated transactions to purchase shares in connection with our initial business combination. If we are unable to complete our initial business combination, our public stockholders may receive only approximately \$10.20 per share plus any pro rata interest earned on the funds held in the trust account and not previously released to us to pay our taxes on the liquidation of our trust account and our warrants will expire worthless. In addition, even if we do not need additional financing to complete our initial business combination, we may require such financing to fund the operations or growth of the target business. The failure to secure additional financing could have a material adverse effect on the continued development or growth of the target business. None of our officers, directors or stockholders is required to provide any financing to us in connection with or after our initial business combination. If we are unable to complete our initial business combination, our public stockholders may only receive approximately \$10.20 per share on the liquidation of our trust account, or less in certain circumstances, and our warrants will expire worthless. Furthermore, as described in the risk factor entitled "If third parties bring claims against us, the proceeds held in the trust account could be reduced and the per-share redemption amount received by stockholders may be less than \$10.20 per share," under certain circumstances our public stockholders may receive less than \$10.20 per share upon the liquidation of the trust account.

Our initial stockholders may exert a substantial influence on actions requiring a stockholder vote, potentially in a manner that you do not support.

Upon the closing of this offering, our initial stockholders will own shares representing 20% of our issued and outstanding shares of common stock (assuming they do not purchase any units in this offering and excluding the representative's shares). Accordingly, they may exert a substantial influence on actions requiring a stockholder vote, potentially in a manner that you do not support, including amendments to our amended and restated certificate of incorporation and approval of major corporate transactions. If our initial stockholders purchase any units in this offering or if our initial stockholders purchase any additional shares of common stock in the aftermarket or in privately negotiated transactions, this would increase their control. Factors that would be considered in making such additional purchases would include consideration of the current trading price of our Class A common stock. In addition, our board of directors, were elected by our initial stockholders. We may not hold an annual meeting of stockholders to elect new directors prior to the completion of our initial business combination, in which case all of the current directors will continue in office until at least the completion of the initial business combination. Accordingly, our initial stockholders will exert control of the board of directors for the near term, including potentially until the completion of our initial business combination.

Because we must furnish our stockholders with target business financial statements, we may lose the ability to complete an otherwise advantageous initial business combination with some prospective target businesses.

The federal proxy rules require that a proxy statement with respect to a vote on an initial business combination meeting certain financial significance tests include historical and/or pro forma financial statement disclosure in periodic reports. We will include the same financial statement disclosure in connection with our tender offer documents, whether or not they are required under the tender offer rules. These financial statements may be required to be prepared in accordance with, or be reconciled to, accounting principles generally accepted in the United States of America, or GAAP, or international financial reporting standards as issued by the International Accounting Standards Board, or IFRS, depending on the circumstances and the historical financial statements may be required to be audited in accordance with the standards of the Public Company Accounting Oversight Board (United States), or PCAOB. These financial statement requirements may limit the pool of potential target businesses we may acquire because some targets may be unable to provide such financial statements in time for us to disclose such statements in accordance with federal proxy rules and complete our initial business combination within the prescribed time frame.

Compliance obligations under the Sarbanes-Oxley Act may make it more difficult for us to effectuate our initial business combination, require substantial financial and management resources, and increase the time and costs of completing an initial business combination.

Section 404 of the Sarbanes-Oxley Act requires that we evaluate and report on our system of internal controls beginning with our Annual Report on Form 10-K for the year ending December 31, 2022. Only in the event we are deemed to be a large accelerated filer or an accelerated filer, and no longer qualify as an emerging growth company, will we be required to comply with the independent registered public accounting firm attestation requirement on our internal control over financial reporting. Further, for as long as we remain an emerging growth company, we will not be required to comply with the independent registered public accounting firm attestation requirement on our internal control over financial reporting. The fact that we are a blank check company makes compliance with the requirements of the Sarbanes-Oxley Act particularly burdensome on us as compared to other public companies because a target company with which we seek to complete our initial business combination may not be in compliance with the provisions of the Sarbanes-Oxley Act regarding adequacy of its internal controls. The development of the internal control of any such entity to achieve compliance with the Sarbanes-Oxley Act may increase the time and costs necessary to complete any such business combination.

If we effect our initial business combination with a company with operations or opportunities outside of the United States, we would be subject to a variety of additional risks that may negatively impact our operations.

If we effect our initial business combination with a company with operations or opportunities outside of the United States, we would be subject to any special considerations or risks associated with companies operating in an international setting, including any of the following:

- higher costs and difficulties inherent in managing cross-border business operations and complying with different commercial and legal requirements of overseas markets;
- rules and regulations regarding currency redemption;
- complex corporate withholding taxes on individuals;
- laws governing the manner in which future business combinations may be effected;
- tariffs and trade barriers;
- regulations related to customs and import/export matters;
- longer payment cycles and challenges in collecting accounts receivable;
- tax issues, including but not limited to tax law changes and variations in tax laws as compared to the United States;
- currency fluctuations and exchange controls;
- rates of inflation;
- cultural and language differences;
- employment regulations;
- crime, strikes, riots, civil disturbances, terrorist attacks, natural disasters and wars;

- deterioration of political relations with the United States; and
- government appropriations of assets.

We may not be able to adequately address these additional risks. If we were unable to do so, our operations might suffer, which may adversely impact our results of operations and financial condition.

If our management team following our initial business combination is unfamiliar with United States securities laws, they may have to expend time and resources becoming familiar with such laws, which could lead to various regulatory issues.

Following our initial business combination, our founding team may resign from their positions as officers or directors of the company and the management of the business combination partner will may assume the roles of executive officers and directors of our company. Such officers and directors may not be familiar with United States securities laws. If our new management following our initial business combination is unfamiliar with United States securities laws, they may have to expend time and resources becoming familiar with such laws. This could be expensive and time-consuming and could lead to various regulatory issues which may adversely affect our operations.

After our initial business combination, substantially all of our assets may be located in a foreign country and substantially all of our revenue may be derived from our operations in such country. Accordingly, our results of operations and prospects will be subject, to a significant extent, to the economic, political and social conditions and government policies, developments and conditions in the country in which we operate.

As we may acquire a business located outside of the United States as part of our initial business combination, the economic, political and social conditions, as well as government policies, of the country in which our operations would be located following our initial business combination could affect our business. Economic growth could be uneven, both geographically and among various sectors of the economy and such growth may not be sustained in the future. If in the future such country's economy experiences a downturn or grows at a slower rate than expected, there may be less demand for spending in certain industries. A decrease in demand for spending in certain industries could materially and adversely affect our ability to find an attractive target business with which to consummate our initial business combination and if we effect our initial business combination, the ability of that target business to become profitable.

Exchange rate fluctuations and currency policies may cause our target business's ability to succeed in the international markets to be diminished.

In the event we acquire a non-U.S. business as part of our initial business combination, all revenues and income would likely be received in a foreign currency, and the dollar equivalent of our net assets and distributions, if any, could be adversely affected by reductions in the value of the local currency. The value of the currencies in our target regions fluctuate and are affected by, among other things, changes in political and economic conditions. Any change in the relative value of such currency against our reporting currency may affect the attractiveness of any target business or, following consummation of our initial business combination, our financial condition and results of operations. Additionally, if a currency appreciates in value against the dollar prior to the consummation of our initial business combination, the cost of a target business as measured in dollars will increase, which may make it less likely that we are able to consummate such transaction.

We may reincorporate in another jurisdiction in connection with our initial business combination, and the laws of such jurisdiction may govern some or all of our future material agreements and we may not be able to enforce our legal rights.

In connection with our initial business combination, we may relocate the home jurisdiction of our business from the U.S. to another jurisdiction. If we determine to do this, the laws of such jurisdiction may govern some or all of our future material agreements. The system of laws and the enforcement of existing laws in such jurisdiction may not be as certain in implementation and interpretation as in the United States. The inability to enforce or obtain a remedy under any of our future agreements could result in a significant loss of business, business opportunities or capital.

There may be tax consequences to our business combinations that may adversely affect us.

While we expect to undertake any merger or acquisition so as to minimize taxes both to the acquired business and/or asset and us, such business combination might not meet the statutory requirements of a tax-free reorganization, or the parties might not obtain the intended tax-free treatment upon a transfer of shares or assets. A reorganization that does not qualify as tax-free could result in the imposition of substantial taxes on holders of our securities.

Risks Relating to our Sponsor and Management Team

Our ability to successfully effect our initial business combination and to be successful thereafter will be totally dependent upon the efforts of our key personnel, some of whom may join us following our initial business combination. The loss of key personnel could negatively impact the operations and profitability of our post-combination business.

Our ability to successfully effect our initial business combination is dependent upon the efforts of our key personnel. The role of our key personnel in the target business, however, cannot presently be ascertained. Although some of our key personnel may remain with the target business in senior management or advisory positions following our initial business combination, it is likely that some or all of the management of the target business will remain in place. While we intend to closely scrutinize any individuals we employ after our initial business combination, we cannot assure you that our assessment of these individuals will prove to be correct. These individuals may be unfamiliar with the requirements of operating a company regulated by the SEC, which could cause us to have to expend time and resources helping them become familiar with such requirements. In addition, the officers and directors of an initial business combination candidate may resign upon completion of our initial business combination. The departure of an initial business combination target's key personnel could negatively impact the operations and profitability of our post-combination business. The role of an initial business combination candidate's key personnel upon the completion of our initial business combination cannot be ascertained at this time. Although we contemplate that certain members of an initial business combination candidate following our initial business combination, it is possible that members of the management of an initial business combination candidate will not wish to remain in place. The loss of key personnel could negatively impact the operations and profitability of our post-combination business.

We are dependent upon our executive officers and directors and their departure could adversely affect our ability to operate.

Our operations are dependent upon a relatively small group of individuals and, in particular, our executive officers and directors. We believe that our success depends on the continued service of our executive officers and directors, at least until we have completed our initial business combination. We do not have an employment agreement with, or key-man insurance on the life of, any of our directors or executive officers. The unexpected loss of the services of one or more of our directors or executive officers could have a detrimental effect on us.

Our key personnel may negotiate employment or consulting agreements with a target business in connection with a particular business combination. These agreements may provide for them to receive compensation following our initial business combination and as a result, may cause them to have conflicts of interest in determining whether a particular business combination is the most advantageous.

Our key personnel may be able to remain with the company after the completion of our initial business combination only if they are able to negotiate employment or consulting agreements in connection with the initial business combination. Such negotiations would take place simultaneously with the negotiation of the initial business combination and could provide for such individuals to receive compensation in the form of cash payments and/or our securities for services they would render to us after the completion of the initial business combination. The personal and financial interests of such individuals may influence their motivation in identifying and selecting a target business. However, we believe the ability of such individuals to remain with us after the completion of our initial business combination will not be the determining factor in our decision as to whether or not we will proceed with any potential business combination. There is no certainty, however, that any of our key personnel will remain with us after the completion of our initial business combination. We cannot assure you that any of our key personnel will remain in senior management or advisory positions with us. The determination as to whether any of our key personnel will remain with us will be made at the time of the consummation of our initial business combination.

Our officers and directors will allocate their time to other businesses thereby causing conflicts of interest in their determination as to how much time to devote to our affairs. This conflict of interest could have a negative impact on our ability to complete our initial business combination.

Our officers and directors are not required to, and will not, commit their full time to our affairs, which may result in a conflict of interest in allocating their time between our operations and our search for an initial business combination and their other businesses. We do not intend to have any full-time employees prior to the completion of our initial business combination. Each of our officers is engaged in other business endeavors for which he may be entitled to substantial compensation and our officers are not obligated to contribute any specific number of hours per week to our affairs. Our independent directors may also serve as officers or board members for other entities. If our officers' and directors' other business affairs require them to devote substantial amounts of time to such affairs in excess of their current commitment levels, it could limit their ability to devote time to our affairs which may have a negative impact on our ability to complete our initial business combination. For a complete discussion of our officers' and directors' other business affairs, please see the section of this prospectus entitled "Management — Directors and Officers."

Certain of our officers and directors are now, and all of them may in the future become, affiliated with entities engaged in business activities similar to those intended to be conducted by us and, accordingly, may have conflicts of interest in allocating their time and determining to which entity a particular business opportunity should be presented.

Following the completion of this offering and until we consummate our initial business combination, we intend to engage in the business of identifying and combining with one or more businesses. Our sponsor and officers and directors are, and may in the future become, affiliated with entities (such as operating companies or investment vehicles) that are engaged in a similar business and our officers and directors may become officers or directors of another special purpose acquisition company with a class of securities intended to be registered under the Exchange Act, even prior to us entering into a definitive agreement for our initial business combination. Our officers and directors also may become aware of business opportunities which may be appropriate for presentation to us and the other entities to which they owe certain fiduciary or contractual duties.

Accordingly, they may have conflicts of interest in determining to which entity a particular business opportunity should be presented. These conflicts may not be resolved in our favor and a potential target business may be presented to another entity prior to its presentation to us. Our amended and restated certificate of incorporation will provide that we renounce our interest in any corporate opportunity offered to any director or officer unless such opportunity is expressly offered to such person solely in his or her capacity as a director or officer of our company and such opportunity is one we are legally and contractually permitted to undertake and would otherwise be reasonable for us to pursue, and to the extent the director or officer is permitted to refer that opportunity to us without violating another legal obligation.

For a complete discussion of our officers' and directors' business affiliations and the potential conflicts of interest that you should be aware of, please see the sections of this prospectus entitled "Management — Directors and Officers," "Management — Conflicts of Interest" and "Certain Relationships and Related Party Transactions."

Our officers, directors, security holders and their respective affiliates may have competitive pecuniary interests that conflict with our interests.

We have not adopted a policy that expressly prohibits our directors, officers, security holders or affiliates from having a direct or indirect pecuniary or financial interest in any investment to be acquired or disposed of by us or in any transaction to which we are a party or have an interest. In fact, we may enter into an initial business combination with a target business that is affiliated with our sponsor, our directors or officers, although we do not intend to do so. We do not have a policy that expressly prohibits any such persons from engaging for their own account in business activities of the types conducted by us. Accordingly, such persons or entities may have a conflict between their interests and ours.

The personal and financial interests of our executive officers and directors may influence their motivation in timely identifying and selecting a target business and completing a business combination. Consequently, our executive officers' and directors' discretion in identifying and selecting a suitable target business may result in a conflict of interest when determining whether the terms, conditions and timing of a particular business combination are appropriate and in our stockholders' best interest. If this were the case, it would be a breach of their fiduciary duties to us as a matter of Delaware law and we or our stockholders might have a claim against such individuals for infringing on our stockholders' rights. See the section titled "Description of Securities—Certain Differences in Corporate Law—Stockholders' Suits" for further information on the ability to bring such claims. However, we might not ultimately be successful in any claim we may make against them for such reason

The representative may have a conflict of interest if they render services to us in connection with our initial business combination.

We may elect to engage EF Hutton, division of Benchmark Investments, LLC formerly known as Kingswood Capital Markets, division of Benchmark Investments, LLC (who is the representative of the underwriters of this offering) to assist us in connection with our initial business combination. The representative shares held by the representative and/or its designees will also be worthless if we do not consummate an initial business combination. Therefore, if the representative provides services to us in connection with our initial business combination, these financial interests may result in the representative having a conflict of interest when providing such services to us.

We may engage in an initial business combination with one or more target businesses that have relationships with entities that may be affiliated with our sponsor, officers, directors or existing holders which may raise potential conflicts of interest.

In light of the involvement of our sponsor, officers and directors with other entities, we may decide to acquire one or more businesses affiliated with our sponsor, officers or directors. Our directors and officers also serve as officers and board members for other entities, including, without limitation, those described under the section of this prospectus entitled "Management — Conflicts of Interest." Such entities may compete with us for business combination opportunities. Our sponsor, officers and directors are not currently aware of any specific opportunities for us to complete our initial business combination with any entities with which they are affiliated, and there have been no preliminary discussions concerning an initial business combination with any such entity or entities. Although we will not be specifically focusing on, or targeting, any transaction with any affiliated entities, we would pursue such a transaction if we determined that such affiliated entity met our criteria for an initial business combination as set forth in the section of this prospectus entitled "Proposed Business — Selection of a Target Business and Structuring of our Initial Business Combination" and such transaction was approved by a majority of our disinterested directors. Despite our agreement to obtain an opinion from an independent investment banking firm or another independent entity that commonly renders valuation opinions, regarding the fairness to our stockholders from a financial point of view of an initial business combination with one or more businesses affiliated with our sponsor, officers, directors or existing holders, potential conflicts of interest still may exist and, as a result, the terms of the initial business combination may not be as advantageous to our public stockholders as they would be absent any conflicts of interest.

Since our sponsor, officers and directors will lose their entire investment in us if our initial business combination is not completed, a conflict of interest may arise in determining whether a particular business combination target is appropriate for our initial business combination.

On June 30, 2021, our sponsor purchased 2,875,000 founder shares for an aggregate purchase price of \$25,000, or approximately \$0.009 per share. The number of founder shares issued was determined based on the expectation that such founder shares would represent 20% of the outstanding shares after this offering (excluding the placement warrants and underlying securities, and the representative's shares). The founder shares will be worthless if we do not complete an initial business combination. Our sponsor has agreed to purchase an aggregate of 5,261,000 placement warrants at a price of \$1.00 per warrant for an aggregate purchase price of \$5,261,000 (provided that if the over-allotment option is exercised in full, the amount of placement warrants sold will be 5,711,000 for an aggregate purchase price of \$5,711,000). Each placement warrant is exercisable to purchase one whole share of common stock at \$11.50 per share. The placement warrants will be worthless if we do not complete an initial business combination. Holders of founder shares have agreed (A) to vote any shares owned by them in favor of any proposed initial business combination and (B) not to redeem any founder shares held by them in connection with a stockholder vote to approve a proposed initial business combination. In addition, we may obtain loans from our sponsor, affiliates of our sponsor or an officer or director. The personal and financial interests of our officers and directors may influence their motivation in identifying and selecting a target business combination, completing an initial business combination and influencing the operation of the business following the initial business combination.

Since our sponsor, executive officers and directors will not be eligible to be reimbursed for their out-of-pocket expenses if our business combination is not completed, a conflict of interest may arise in determining whether a particular business combination target is appropriate for our initial business combination.

Our sponsor, executive officers and directors, or any of their respective affiliates, will be reimbursed for any out-of-pocket expenses incurred in connection with activities on our behalf such as identifying potential target businesses and performing due diligence on suitable business combinations. There is no cap or ceiling on the reimbursement of out-of-pocket expenses incurred in connection with activities on our behalf; provided, however, that to the extent such expenses exceed the available proceeds not deposited in the trust account, such expenses would not be reimbursed by us unless we consummate an initial business combination. These financial interests of our sponsor, executive officers and directors may influence their motivation in identifying and selecting a target business combination and completing an initial business combination.

Risks Relating to our Securities

The securities in which we invest the funds held in the trust account could bear a negative rate of interest, which could reduce the value of the assets held in trust such that the per-share redemption amount received by public stockholders may be less than \$10.20 per share.

The proceeds held in the trust account will be invested only in U.S. government treasury obligations with a maturity of 185 days or less or in money market funds meeting certain conditions under Rule 2a-7 under the Investment Company Act, which invest only in direct U.S. government treasury obligations. While short-term U.S. government treasury obligations currently yield a positive rate of interest, they have briefly yielded negative interest rates in recent years. Central banks in Europe and Japan pursued interest rates below zero in recent years, and the Open Market Committee of the Federal Reserve has not ruled out the possibility that it may in the future adopt similar policies in the United States. In the event that we are unable to complete our initial business combination or make certain amendments to our amended and restated certificate of incorporation, our public stockholders are entitled to receive their pro-rata share of the proceeds held in the trust account, plus any interest income, net of taxes paid or payable (less, in the case we are unable to complete our initial business combination, \$100,000 of interest). Negative interest rates could reduce the value of the assets held in trust such that the pershare redemption amount received by public stockholders may be less than \$10.20 per share.

Nasdaq may delist our securities from trading on its exchange, which could limit investors' ability to make transactions in our securities and subject us to additional trading restrictions.

We have applied to have our units listed on Nasdaq. We expect that our units will be listed on Nasdaq on or promptly after the date of this prospectus. Following the date the shares of our Class A common stock and warrants are eligible to trade separately, we anticipate that the shares of our Class A common stock and warrants will be separately listed on Nasdaq. We cannot guarantee that our securities will be approved for listing on Nasdaq. Although after giving effect to this offering we expect to meet, on a pro forma basis, the minimum initial listing standards set forth in the Nasdaq listing standards, we cannot assure you that our securities will be, or will continue to be, listed on Nasdaq in the future or prior to our initial business combination. In order to continue listing our securities on Nasdaq prior to our initial business combination, we must maintain certain financial, distribution and stock price levels. Generally, we must maintain a \$50 million market value of listed securities, 1.1 million publicly available shares, a \$15 million market value of publicly held shares and a minimum number of holders of our securities (generally 400 public holders). Additionally, in connection with our initial business combination, we will be required to demonstrate compliance with Nasdaq's initial listing requirements, which are more rigorous than Nasdaq's continued listing requirements, in order to continue to maintain the listing of our securities on Nasdaq. For instance, our stock price would generally be required to be at least \$4.00 per share, the market value of listed securities would be required to be at least \$75 million, we would need to have 1.1 million publicly available shares and \$20 million of market value of unrestricted publicly held shares, and we would be required to have a minimum of 400 round lot holders (with at least 50% of such round lot holders holding securities with a market value of at least \$2,500) of our securities. We cannot assure you that we will be able to meet those initial lis

If Nasdaq delists our securities from trading on its exchange and we are not able to list our securities on another national securities exchange, we expect our securities could be quoted on an over-the-counter market. If this were to occur, we could face significant material adverse consequences, including:

- a limited availability of market quotations for our securities;
- reduced liquidity for our securities;
- a determination that our securities are a "penny stock" which will require brokers trading in our securities to adhere to more stringent rules and possibly result in a reduced level of trading activity in the secondary trading market for our securities;
- a limited amount of news and analyst coverage; and
- a decreased ability to issue additional securities or obtain additional financing in the future.

The National Securities Markets Improvement Act of 1996, which is a federal statute, prevents or preempts the states from regulating the sale of certain securities, which are referred to as "covered securities." Because we expect that our units and eventually our Class A common stock and warrants will be listed on Nasdaq, our units, Class A common stock, and warrants will be covered securities. Although the states are preempted from regulating the sale of our securities, the federal statute does allow the states to investigate companies if there is a suspicion of fraud, and, if there is a finding of fraudulent activity, then the states can regulate or bar the sale of covered securities in a particular case. While we are not aware of a state having used these powers to prohibit or restrict the sale of securities issued by blank check companies, other than the State of Idaho, certain state securities regulators view blank check companies unfavorably and might use these powers, or threaten to use these powers, to hinder the sale of securities of blank check companies in their states. Further, if we were no longer listed on Nasdaq, our securities would not be covered securities and we would be subject to regulation in each state in which we offer our securities, including in connection with our initial business combination.

If we seek stockholder approval of our initial business combination and we do not conduct redemptions pursuant to the tender offer rules, and if you or a "group" of stockholders are deemed to hold in excess of 15% of our Class A common stock, you will lose the ability to redeem all such shares in excess of 15% of our Class A common stock.

If we seek stockholder approval of our initial business combination and we do not conduct redemptions in connection with our initial business combination pursuant to the tender offer rules, our amended and restated certificate of incorporation will provide that a public stockholder, together with any affiliate of such stockholder or any other person with whom such stockholder is acting in concert or as a "group" (as defined under Section 13 of the Exchange Act), will be restricted from seeking redemption rights with respect to more than an aggregate of 15% of the shares sold in this offering without our prior consent, which we refer to as the "Excess Shares." However, we would not be restricting our stockholders' ability to vote all of their shares (including Excess Shares) for or against our initial business combination. Your inability to redeem the Excess Shares will reduce your influence over our ability to complete our initial business combination and you could suffer a material loss on your investment in us if you sell Excess Shares in open market transactions. Additionally, you will not receive redemption distributions with respect to the Excess Shares if we complete our initial business combination. And as a result, you will continue to hold that number of shares exceeding 15% and, in order to dispose of such shares, would be required to sell your shares in open market transactions, potentially at a loss.

We are not registering the shares of Class A common stock issuable upon exercise of the warrants under the Securities Act or any state securities laws at this time, and such registration may not be in place when an investor desires to exercise warrants, thus precluding such investor from being able to exercise its warrants except on a cashless basis. If the issuance of the shares upon exercise of warrants is not registered, qualified or exempt from registration or qualification, the holder of such warrant will not be entitled to exercise such warrant and such warrant may have no value and expire worthless.

We are not registering the shares of Class A common stock issuable upon exercise of the warrants under the Securities Act or any state securities laws at this time. However, under the terms of the warrant agreement, we have agreed that as soon as practicable after the closing of our initial business combination, we will use our best efforts to file with the SEC a registration statement for the registration under the Securities Act of the issuance of the shares of Class A common stock issuable upon exercise of the warrants and thereafter will use our best efforts to cause the same to become effective within 90 days following our initial business combination and to maintain a current prospectus relating to the Class A common stock issuable upon exercise of the warrants, until the expiration of the warrants in accordance with the provisions of the warrant agreement. We cannot assure you that we will be able to do so if, for example, any facts or events arise which represent a fundamental change in the information set forth in the registration statement or prospectus, the financial statements contained or incorporated by reference therein are not current, complete or correct or the SEC issues a stop order. If the shares of Class A common stock issuable upon exercise of the warrants are not registered under the Securities Act, we will be required to permit holders to exercise their warrants on a cashless basis. However, no warrant will be exercisable for cash or on a cashless basis, and we will not be obligated to issue any shares to holders seeking to exercise their warrants, unless the issuance of the shares upon such exercise is registered or qualified under the securities laws of the state of the exercising holder, or an exemption from registration is available. Notwithstanding the foregoing, if a registration statement covering the issuance of the Class A common stock issuable upon exercise of the warrants is not effective within a specified period following the consummation of our initial business combination, warrant holders may, until such time as there is an effective registration statement and during any period when we shall have failed to maintain an effective registration statement, exercise warrants on a cashless basis pursuant to the exemption provided by Section 3(a)(9) of the Securities Act, provided that such exemption is available. If that exemption, or another exemption, is not available, holders will not be able to exercise their warrants on a cashless basis. We will use our best efforts to register or qualify the shares under applicable blue sky laws to the extent an exemption is not available. In no event will we be required to net cash settle any warrant, or issue securities or other compensation in exchange for the warrants in the event that we are unable to register or qualify the shares underlying the warrants under applicable state securities laws and there is no exemption available. If the issuance of the shares upon exercise of the warrants is not so registered or qualified or exempt from registration or qualification, the holder of such warrant will not be entitled to exercise such warrant and such warrant may have no value and expire worthless. In such event, holders who acquired their warrants as part of a purchase of units will have paid the full unit purchase price solely for the shares of Class A common stock included in the units. If and when the warrants become redeemable by us, we may not exercise our redemption right if the issuance of shares of common stock upon exercise of the warrants is not exempt from registration or qualification under applicable state blue sky laws or we are unable to effect such registration or qualification. We will use our best efforts to register or qualify such shares of common stock under the blue sky laws of the state of residence in those states in which the warrants were offered by us in this offering. However, there may be instances in which holders of our public warrants may be unable to exercise such public warrants but holders of our placement warrants may be able to exercise such placement warrants.

If you exercise your public warrants on a "cashless basis," you will receive fewer shares of Class A common stock from such exercise than if you were to exercise such warrants for cash.

There are circumstances in which the exercise of the public warrants may be required or permitted to be made on a cashless basis. First, if a registration statement covering the issuance of the shares of Class A common stock issuable upon exercise of the warrants is not effective by the 90th day after the closing of our initial business combination, warrant holders may, until such time as there is an effective registration statement, exercise warrants on a cashless basis in accordance with Section 3(a)(9) of the Securities Act or another exemption. Second, if a registration statement covering the Class A common stock issuable upon exercise of the warrants is not effective within a specified period following the consummation of our initial business combination, warrant holders may, until such time as there is an effective registration statement and during any period when we shall have failed to maintain an effective registration statement, exercise warrants on a cashless basis pursuant to the exemption provided by Section 3(a)(9) of the Securities Act, provided that such exemption is available; if that exemption, or another exemption, is not available, holders will not be able to exercise their warrants on a cashless basis. Third, if we call the public warrants for redemption, our management will have the option to require all holders that wish to exercise warrants to do so on a cashless basis. In the event of an exercise on a cashless basis, a holder would pay the warrant exercise price by surrendering the warrants for that number of shares of Class A common stock equal to the quotient obtained by dividing (x) the product of the number of shares of Class A common stock underlying the warrants, multiplied by the excess of the "fair market value" (as defined in the next sentence) over the exercise price of the warrants by (y) the fair market value. The "fair market value" for this purpose shall mean the average reported last sale price of the Class A common stock for the 10 trading days ending on the third trading day prior to the date on which the notice of exercise is received by the warrant agent or on which the notice of redemption is sent to the holders of warrants, as applicable. As a result, you would receive fewer shares of Class A common stock from such exercise than if you were to exercise such warrants for cash.

We may issue additional common stock or preferred stock to complete our initial business combination or under an employee incentive plan after completion of our initial business combination. We may also issue shares of Class A common stock upon the conversion of the Class B common stock at a ratio greater than one-to-one at the time of the consummation of our initial business combination as a result of the anti-dilution provisions contained in our amended and restated certificate of incorporation. Any such issuances would dilute the interest of our stockholders and likely present other risks.

Our amended and restated certificate of incorporation will authorize the issuance of up to 125,000,000 shares of Class A common stock, par value \$0.0001 per share, 12,500,000 shares of Class B common stock, par value \$0.0001 per share, and 1,250,000 shares of preferred stock, par value \$0.0001 per share. Immediately after this offering and the sale of the placement warrants and the issuance of 100,000 shares to the representative of the underwriters, there will be 114,900,000 and 10,000,000 (assuming, in each case, that the underwriters have not exercised their over-allotment option) authorized but unissued shares of Class A common stock and Class B common stock, respectively, available for issuance, which amount does not take into account the shares of Class A common stock reserved for issuance upon exercise of outstanding warrants or the shares of Class A common stock issuable upon conversion of Class B common stock. Immediately after the consummation of this offering, there will be no shares of preferred stock issued and outstanding. Shares of Class B common stock are convertible into shares of our Class A common stock initially at a one-for-one ratio but subject to adjustment as set forth herein, including in certain circumstances in which we issue Class A common stock or equity-linked securities related to our initial business combination.

We may issue a substantial number of additional shares of common or preferred stock to complete our initial business combination or under an employee incentive plan after completion of our initial business combination (although our amended and restated certificate of incorporation will provide that we may not issue securities that can vote with common stockholders on matters related to our pre-initial business combination activity). We may also issue shares of Class A common stock upon conversion of the Class B common stock at a ratio greater than one-to-one at the time of the consummation of our initial business combination as a result of the anti-dilution provisions contained in our amended and restated certificate of incorporation. However, our amended and restated certificate of incorporation will provide, among other things, that prior to our initial business combination, we may not issue additional shares of capital stock that would entitle the holders thereof to (i) receive funds from the trust account or (ii) vote on any initial business combination. These provisions of our amended and restated certificate of incorporation, like all provisions of our amended and restated certificate of incorporation, may be amended with the approval of our stockholders. However, our executive officers, directors and director nominees have agreed, pursuant to a written agreement with us, that they will not propose any amendment to our amended and restated certificate of incorporation (A) to modify the substance or timing of our obligation to allow redemption in connection with our initial business combination or certain amendments to our charter prior thereto or to redeem 100% of our public shares if we do not complete our initial business combination within 12 months from the closing of this offering or during any Extension Period or (B) with respect to any other provision relating to stockholders' rights or pre-initial business combination activity, unless we provide our public stockholders with the opportunity to redeem their shares of common stock upon approval of any such amendment at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the trust account, including interest (which interest shall be net of taxes payable), divided by the number of then outstanding public shares.

The issuance of additional shares of common or preferred stock:

- may significantly dilute the equity interest of investors in this offering, which dilution would increase if the anti-dilution provisions in the Class B common stock resulted in the issuance of Class A shares on a greater than one-to-one basis upon conversion of the Class B common stock;
- may subordinate the rights of holders of our common stock if preferred stock is issued with rights senior to those afforded our common stock;
- could cause a change of control if a substantial number of shares of our common stock are issued, which may affect, among other things, our
 ability to use our net operating loss carry forwards, if any, and could result in the resignation or removal of our present officers and directors;
- may have the effect of delaying or preventing a change of control of us by diluting the stock ownership or voting rights of a person seeking to obtain control of us; and
- may adversely affect prevailing market prices for our units, Class A common stock and/or warrants.

We may issue our shares to investors in connection with our initial business combination at a price that is less than the prevailing market price of our shares at that time.

In connection with our initial business combination, we may issue shares to investors in private placement transactions (so-called PIPE transactions) at a price of \$10.00 per share or which approximates the per-share amounts in our trust account at such time. The purpose of such issuances will be to enable us to provide sufficient liquidity to the post-business combination entity. The price of the shares we issue may therefore be less, and potentially significantly less, than the market price for our shares at such time.

Our initial stockholders paid a nominal price for the founders' shares and, accordingly, you will experience immediate and substantial dilution from the purchase of our shares of common stock.

The difference between the public offering price per share (allocating all of the unit purchase price to the Class A common stock and none to the warrants included in the unit) and the pro forma net tangible book value per share of our Class A common stock after this offering constitutes the dilution to you and the other investors in this offering. Our sponsor acquired the founder shares at a nominal price, significantly contributing to this dilution. Upon the closing of this offering, and assuming no value is ascribed to the warrants included in the units, you and the other public stockholders will incur an immediate and substantial dilution of approximately \$10.76 per share (or \$10.81 per share, assuming the full exercise of the underwriters' over-allotment option), the difference between the pro forma net tangible book value per share and the initial offering price of \$10.00 per unit. The economic dilution to public stockholders, however, is \$0.76 per share (or \$0.81 per share, assuming the full exercise of the underwriters' over-allotment option) after taking into account redeemable temporary equity of \$10.00 per share. In addition, because of the anti-dilution rights of the founder shares, any equity or equity-linked securities issued or deemed issued in connection with our initial business combination would be disproportionately dilutive to our Class A common stock.

Unlike many other similarly structured special purpose acquisition companies, our initial stockholders will receive additional shares of Class A common stock if we issue shares to consummate an initial business combination.

The founder shares will automatically convert into Class A common stock at the time of the consummation of our initial business combination, on a one-for-one basis, subject to adjustment as provided herein. In the case that additional shares of Class A common stock, or equity-linked securities convertible or exercisable for Class A common stock, are issued or deemed issued in excess of the amounts offered in this prospectus and related to the closing of the initial business combination, the ratio at which founder shares shall convert into Class A common stock will be adjusted so that the number of Class A common stock issuable upon conversion of all founder shares will equal, in the aggregate, on an as-converted basis, 20% of the total number of all outstanding shares of common stock upon completion of the initial business combination, excluding the placement warrants and underlying securities, and the representative's shares and any shares or equity-linked securities issued, or to be issued, to any seller in the business combination and any private placement-equivalent units and their underlying securities issued to our sponsor or its affiliates upon conversion of loans made to us. This is different from most other similarly structured blank check companies in which the initial stockholder will only be issued an aggregate of 20% of the total number of shares to be outstanding prior to the initial business combination. Additionally, the aforementioned adjustment will not take into account any shares of Class A common stock redeemed in connection with the business combination. Accordingly, the holders of the founder shares could receive additional shares of Class A common stock, are issued or deemed issued solely to replace those shares that were redeemed in connection with the business combination. The foregoing may make it more difficult and expensive for us to consummate an initial business combination.

We may amend the terms of the warrants in a manner that may be adverse to holders of public warrants with the approval by the holders of at least a majority of the then outstanding public warrants. As a result, the exercise price of your warrants could be increased, the exercise period could be shortened and the number of shares of our Class A common stock purchasable upon exercise of a warrant could be decreased, all without your approval.

Our warrants will be issued in registered form under a warrant agreement between Continental Stock Transfer & Trust Company, LLC, as warrant agent, and us. The warrant agreement provides that the terms of the warrants may be amended without the consent of any holder to cure any ambiguity or correct any mistake, including to conform the provisions of the warrant agreement to the description of the terms of the warrants and the warrant agreement set forth in this prospectus, or defective provision, but requires the approval by the holders of at least a majority of the then outstanding public warrants to make any change that adversely affects the interests of the registered holders of public warrants (which may include public warrants acquired by our sponsor or its affiliates in this offering or thereafter in the open market). Accordingly, we may amend the terms of the public warrants in a manner adverse to a holder if holders of at least a majority of the then outstanding public warrants approve of such amendment. Although our ability to amend the terms of the public warrants with the consent of at least a majority of the then outstanding public warrants is unlimited, examples of such amendments could be amendments to, among other things, increase the exercise price of the warrants, convert the warrants into cash or stock, shorten the exercise period or decrease the number of shares of our Class A common stock purchasable upon exercise of a warrant.

Our warrant agreement will designate the courts of the State of New York or the United States District Court for the Southern District of New York as the sole and exclusive forum for certain types of actions and proceedings that may be initiated by holders of our warrants, which could limit the ability of warrant holders to obtain a favorable judicial forum for disputes with our company.

Our warrant agreement will provide that, subject to applicable law, (i) any action, proceeding or claim against us arising out of or relating in any way to the warrant agreement, including under the Securities Act, will be brought and enforced in the courts of the State of New York or the United States District Court for the Southern District of New York, and (ii) that we irrevocably submit to such jurisdiction, which jurisdiction shall be the exclusive forum for any such action, proceeding or claim. We will waive any objection to such exclusive jurisdiction and that such courts represent an inconvenient forum.

Notwithstanding the foregoing, these provisions of the warrant agreement will not apply to suits brought to enforce any liability or duty created by the Exchange Act or any other claim for which the federal district courts of the United States of America are the sole and exclusive forum. Any person or entity purchasing or otherwise acquiring any interest in any of our warrants shall be deemed to have notice of and to have consented to the forum provisions in our warrant agreement. If any action, the subject matter of which is within the scope the forum provisions of the warrant agreement, is filed in a court other than a court of the State of New York or the United States District Court for the Southern District of New York (a "foreign action") in the name of any holder of our warrants, such holder shall be deemed to have consented to: (x) the personal jurisdiction of the state and federal courts located in the State of New York in connection with any action brought in any such court to enforce the forum provisions (an "enforcement action"), and (y) having service of process made upon such warrant holder in any such enforcement action by service upon such warrant holder's counsel in the foreign action as agent for such warrant holder.

This choice-of-forum provision may limit a warrant holder's ability to bring a claim in a judicial forum that it finds favorable for disputes with our company, which may discourage such lawsuits. Alternatively, if a court were to find this provision of our warrant agreement inapplicable or unenforceable with respect to one or more of the specified types of actions or proceedings, we may incur additional costs associated with resolving such matters in other jurisdictions, which could materially and adversely affect our business, financial condition and results of operations and result in a diversion of the time and resources of our management and board of directors.

We may redeem your unexpired warrants prior to their exercise at a time that is disadvantageous to you, thereby making your warrants worthless.

We have the ability to redeem outstanding warrants at any time after they become exercisable and prior to their expiration, at a price of \$0.01 per warrant, provided that the reported last sale price of our Class A common stock equals or exceeds \$18.00 per share (as adjusted for stock splits, stock dividends, rights issuances, reorganizations, recapitalizations and the like) for any 20 trading days within a 30 trading-day period commencing once the warrants become exercisable and ending on the third trading day prior to the date on which we give proper notice of such redemption and provided certain other conditions are met. If and when the warrants become redeemable by us, we may not exercise our redemption right if the issuance of shares of common stock upon exercise of the warrants is not exempt from registration or qualification under applicable state blue sky laws or we are unable to effect such registration or qualification. We will use our best efforts to register or qualify such shares of common stock under the blue sky laws of the state of residence in those states in which the warrants were offered by us in this offering. Redemption of the outstanding warrants could force you (i) to exercise your warrants and pay the exercise price therefor at a time when it may be disadvantageous for you to do so, (ii) to sell your warrants at the then-current market price when you might otherwise wish to hold your warrants or (iii) to accept the nominal redemption price which, at the time the outstanding warrants are called for redemption, is likely to be substantially less than the market value of your warrants.

Our warrants and founder shares may have an adverse effect on the market price of our Class A common stock and make it more difficult to effectuate our initial business combination.

We will be issuing warrants to purchase 5,000,000 shares of our Class A common stock (or up to 5,750,000 shares of Class A common stock if the underwriters' over-allotment option is exercised in full) as part of the units offered by this prospectus and, simultaneously with the closing of this offering, we will be issuing placement warrants, in a private placement, consisting of an aggregate of 5,261,000 placement warrants (or 5,711,000 if the over-allotment option is exercised in full) exercisable for 5,261,000 shares of our Class A common stock (or 5,711,000 shares of our common stock if the over-allotment option is exercised in full). Our sponsor currently owns an aggregate of 2,875,000 founder shares. The founder shares are convertible into shares of Class A common stock on a one-for-one basis, subject to adjustment as set forth herein. In addition, if our sponsor makes any working capital loans, up to \$1,500,000 of such loans may be converted into warrants, at a price of \$1.00 per warrant at the option of the lender, upon consummation of our initial business combination. The warrants would be identical to the placement warrants. To the extent we issue shares of Class A common stock to effectuate an initial business combination, the potential for the issuance of a substantial number of additional shares of Class A common stock upon exercise of these warrants and loan conversion rights could make us a less attractive business combination vehicle to a target business. Any such issuance will increase the number of issued and outstanding shares of our Class A common stock and reduce the value of the shares of Class A common stock issued to complete the initial business combination. Therefore, our warrants and founder shares may make it more difficult to effectuate an initial business combination or increase the cost of acquiring the target business.

The private placement warrants will be identical to the warrants sold in this offering except that the private placement warrants, so long as they are held by our sponsor, or its permitted transferees, (i) may not (including the common stock shares issuable upon exercise of these warrants), subject to certain limited exceptions, be transferred, assigned or sold by the holders until 30 days after the completion of our initial business combination, and (ii) will be entitled to registration rights. The private placement warrants (including the common stock issuable upon exercise thereof) may not, subject to certain limited exceptions, be transferred, assigned or sold by the holder.

A provision of our warrant agreement may make it more difficult for us to consummate an initial business combination.

Unlike most blank check companies, if

(i) we issue additional shares of Class A common stock or equity-linked securities for capital raising purposes in connection with the closing of our initial business combination at a Newly Issued Price of less than \$9.20 per share;

- (ii) the aggregate gross proceeds from such issuances represent more than 60% of the total equity proceeds, and interest thereon, available for the funding of our initial business combination on the date of the consummation of our initial business combination (net of redemptions); and
- (iii) the Market Value is below \$9.20 per share;

then the exercise price of the warrants will be adjusted to be equal to 115% of the greater of the Market Value and the Newly Issued Price, and the \$18.00 per share redemption trigger price will be adjusted (to the nearest cent) to be equal to 180% of the greater of the Market Value and the Newly Issued Price. This may make it more difficult for us to consummate an initial business combination with a target business.

The determination of the offering price of our units, the size of this offering and the terms of the units is more arbitrary than the pricing of securities and size of an offering of an operating company in a particular industry. You may have less assurance, therefore, that the offering price of our units properly reflects the value of such units than you would have in a typical offering of an operating company.

Prior to this offering there has been no public market for any of our securities. The public offering price of the units and the terms of the warrants were negotiated between us and the underwriters. In determining the size of this offering, management held customary organizational meetings with the representative of the underwriters, both prior to our inception and thereafter, with respect to the state of capital markets, generally, and the amount the underwriters believed they reasonably could raise on our behalf. Factors considered in determining the size of this offering, prices and terms of the units, including the Class A common stock and warrants underlying the units, include:

- the history and prospects of companies whose principal business is the acquisition of other companies;
- prior offerings of those companies;
- our prospects for acquiring an operating business;
- a review of debt to equity ratios in leveraged transactions;
- our capital structure;
- an assessment of our management and their experience in identifying operating companies;
- general conditions of the securities markets at the time of this offering; and
- other factors as were deemed relevant.

Although these factors were considered, the determination of our offering price, size and terms of the units is more arbitrary than the pricing of securities of an operating company in a particular industry since we have no historical operations or financial results.

An investment in this offering may involve adverse U.S. federal income tax consequences.

An investment in this offering may involve adverse U.S. federal income tax consequences. For instance:

• Because there are no authorities that directly address instruments similar to the units we are issuing in this offering, the allocation an investor makes with respect to the purchase price of the unit among the share of Class A common stock and one half of one warrant included in the units could be challenged by the Internal Revenue Service of the United States ("IRS") or the courts.

- If we make distributions on our Class A common stock, such distributions generally will be treated as dividends for U.S. federal income tax purposes to the extent of our current or accumulated earnings and profits. The ability of a holder to seek redemption of their shares may be viewed as a position with respect to substantially similar or related property which diminishes your risk of loss and thereby affects your ability to satisfy the holding period requirements for the dividends received deduction or the preferential tax rate on qualified dividend income with respect to the time period prior to the approval of an initial business combination.
- Any capital gain or loss you realize on a sale or other disposition of our Class A common stock will generally be long-term capital gain or loss
 if your holding period for the Class A common stock is more than one year. However, the redemption feature of the Class A common stock
 could affect your ability to satisfy the holding period requirements for the long-term capital gain tax rate with respect to the time period prior to
 the approval of an initial business combination.

See the section titled "Material U.S. Federal Tax Considerations" for a summary of the material United States Federal income tax consequences of an investment in our securities. Accordingly, each prospective investor is urged to consult a tax advisor with respect to the specific tax consequences of the acquisition, ownership and disposition of our securities, including the applicability and effect of state, local, or foreign tax laws, as well as U.S. federal tax laws.

There is currently no market for our securities and a market for our securities may not develop, which would adversely affect the liquidity and price of our securities.

There is currently no market for our securities. Stockholders therefore have no access to information about prior market history on which to base their investment decision. Following this offering, the price of our securities may vary significantly due to one or more potential business combinations and general market or economic conditions. Furthermore, an active trading market for our securities may never develop or, if developed, it may not be sustained. You may be unable to sell your securities unless a market can be established and sustained.

Provisions in our amended and restated certificate of incorporation and Delaware law may inhibit a takeover of us, which could limit the price investors might be willing to pay in the future for our Class A common stock and could entrench management.

Our amended and restated certificate of incorporation will contain provisions that may discourage unsolicited takeover proposals that stockholders may consider to be in their best interests. These provisions include a staggered board of directors and the ability of the board of directors to designate the terms of and issue new series of preferred shares, which may make the removal of management more difficult and may discourage transactions that otherwise could involve payment of a premium over prevailing market prices for our securities.

We are also subject to anti-takeover provisions under Delaware law, which could delay or prevent a change of control. Together these provisions may make the removal of management more difficult and may discourage transactions that otherwise could involve payment of a premium over prevailing market prices for our securities.

Our amended and restated certificate of incorporation will require, to the fullest extent permitted by law, that derivative actions brought in our name, actions against our directors, officers, other employees or stockholders for breach of fiduciary duty and certain other actions may be brought only in the Court of Chancery in the State of Delaware and, if brought outside of Delaware, the stockholder bringing the suit will, subject to certain exceptions, be deemed to have consented to service of process on such stockholder's counsel, which may have the effect of discouraging lawsuits against our directors, officers, other employees or stockholders.

Our amended and restated certificate of incorporation will require, to the fullest extent permitted by law, that derivative actions brought in our name, actions against our directors, officers, other employees or stockholders for breach of fiduciary duty and certain other actions may be brought only in the Court of Chancery in the State of Delaware and, if brought outside of Delaware, the stockholder bringing the suit will be deemed to have consented to service of process on such stockholder's counsel except any action (A) as to which the Court of Chancery in the State of Delaware determines that there is an indispensable party not subject to the jurisdiction of the Court of Chancery (and the indispensable party does not consent to the personal jurisdiction of the Court of Chancery within ten days following such determination), (B) which is vested in the exclusive jurisdiction of a court or forum other than the Court of Chancery or (C) for which the Court of Chancery does not have subject matter jurisdiction. Any person or entity purchasing or otherwise acquiring any interest in shares of our capital stock shall be deemed to have notice of and consented to the forum provisions in our amended and restated certificate of incorporation. This choice of forum provision may limit or make more costly a stockholder's ability to bring a claim in a judicial forum that it finds favorable for disputes with us or any of our directors, officers, other employees or stockholders, which may discourage lawsuits with respect to such claims. Alternatively, if a court were to find the choice of forum provision contained in our amended and restated certificate of incorporation to be inapplicable or unenforceable in an action, we may incur additional costs associated with resolving such action in other jurisdictions, which could harm our business, operating results and financial condition.

Our amended and restated certificate of incorporation will provide that the exclusive forum provision will be applicable to the fullest extent permitted by applicable law, subject to certain exceptions. Section 27 of the Exchange Act creates exclusive federal jurisdiction over all suits brought to enforce any duty or liability created by the Exchange Act or the rules and regulations thereunder. As a result, the exclusive forum provision will not apply to suits brought to enforce any duty or liability created by the Exchange Act or any other claim for which the federal courts have exclusive jurisdiction. In addition, our amended and restated certificate of incorporation provides that, unless we consent in writing to the selection of an alternative forum, the federal district courts of the United States of America shall, to the fullest extent permitted by law, be the exclusive forum for the resolution of any complaint asserting a cause of action arising under the Securities Act, or the rules and regulations promulgated thereunder. We note, however, that there is uncertainty as to whether a court would enforce this provision and that investors cannot waive compliance with the federal securities laws and the rules and regulations thereunder. Section 22 of the Securities Act creates concurrent jurisdiction for state and federal courts over all suits brought to enforce any duty or liability created by the Securities Act or the rules and regulations thereunder.

General Risk Factors

We are a newly formed company with no operating history and no revenues, and you have no basis on which to evaluate our ability to achieve our business objective.

We are a newly formed company with no operating results, and we will not commence operations until obtaining funding through this offering. Because we lack an operating history, you have no basis upon which to evaluate our ability to achieve our business objective of completing our initial business combination with one or more target businesses. We have no plans, arrangements or understandings with any prospective target business concerning an initial business combination and may be unable to complete our initial business combination. If we fail to complete our initial business combination, we will never generate any operating revenues.

Our independent registered public accounting firm's report contains an explanatory paragraph that expresses substantial doubt about our ability to continue as a "going concern."

On July 12, 2021, we had cash of \$25,000 and a working capital deficit of \$39,798. Further, we have incurred and expect to continue to incur significant costs in pursuit of our financing and acquisition plans. Management's plans to address this need for capital through this offering are discussed in the section of this prospectus titled "Management's Discussion and Analysis of Financial Condition and Results of Operations." We cannot assure you that our plans to raise capital or to consummate an initial business combination will be successful. These factors, among others, raise substantial doubt about our ability to continue as a going concern. The financial statements contained elsewhere in this prospectus do not include any adjustments that might result from our inability to consummate this offering or our inability to continue as a going concern.

Changes in laws or regulations, or a failure to comply with any laws and regulations, may adversely affect our business, including our ability to negotiate and complete our initial business combination and results of operations.

We are subject to laws and regulations enacted by national, regional and local governments. In particular, we will be required to comply with certain SEC and other legal requirements. Compliance with, and monitoring of, applicable laws and regulations may be difficult, time consuming and costly.

Those laws and regulations and their interpretation and application may also change from time to time and those changes could have a material adverse effect on our business, investments and results of operations. In addition, a failure to comply with applicable laws or regulations, as interpreted and applied, could have a material adverse effect on our business, including our ability to negotiate and complete our initial business combination and results of operations.

Cyber incidents or attacks directed at us could result in information theft, data corruption, operational disruption and/or financial loss.

We depend on digital technologies, including information systems, infrastructure and cloud applications and services, including those of third parties with which we may deal. Sophisticated and deliberate attacks on, or security breaches in, our systems or infrastructure, or the systems or infrastructure of third parties or the cloud, could lead to corruption or misappropriation of our assets, proprietary information and sensitive or confidential data. As an early stage company without significant investments in data security protection, we may not be sufficiently protected against such occurrences. We may not have sufficient resources to adequately protect against, or to investigate and remediate any vulnerability to, cyber incidents. It is possible that any of these occurrences, or a combination of them, could have adverse consequences on our ability to consummate a business combination and lead to financial loss.

We may face risks related to companies in the pharmaceutical sector.

Business combinations with companies in the pharmaceutical sector (where we plan to search for our initial business combination target) entail special considerations and risks. If we are successful in completing a business combination with such a target business, we may be subject to, and possibly adversely affected by, the following risks:

- an inability to compete effectively in a highly competitive environment with many incumbents having substantially greater resources;
- an inability to manage rapid change, and growth;
- an inability to build strong brand identity and customer satisfaction and loyalty;
- an inability to comply with existing or newly promulgated government regulations;
- an inability to deal with customers' privacy concerns;
- an inability to attract and retain customers;
- an inability to license or enforce intellectual property rights on which our business may depend;
- any significant disruption in our computer systems or those of third parties that we would utilize in our operations;

- an inability by us, or a refusal by regulatory authorities, to provide approvals for the sale of certain products;
- potential liability for negligence, copyright or trademark infringement or other claims based on the nature and content of products that we may distribute:
- increased efforts to reduce healthcare costs;
- litigation, including product liability and intellectual property infringement litigation;
- disruption or failure of our networks, systems or technology as a result of computer viruses, "cyber-attacks," misappropriation of data or other malfeasance, as well as outages, natural disasters, terrorist attacks, accidental releases of information or similar events;
- the creation of new and improved drugs or products; and
- reliance on third-party manufactures of products.

Any of the foregoing could have an adverse impact on our business, financial condition and results of operations following a business combination.

Past performance by our management team may not be indicative of future performance of an investment in us.

Past performance by our management team is not a guarantee either (i) of success with respect to any business combination we may consummate or (ii) that we will be able to locate a suitable candidate for our initial business combination. You should not rely on the historical record of our management team's performance as indicative of our future performance of an investment in the company or the returns the company will, or is likely to, generate going forward. None of our directors has experience with blank check companies or special purpose acquisition companies. Additionally, in the course of their respective careers, members of our management team have been involved in businesses and deals that were unsuccessful.

We may be subject to an increased rate of tax on our income if we are treated as a personal holding company.

Depending on the date and size of our initial business combination, it is possible that we could be treated as a "personal holding company" for U.S. federal income tax purposes. A U.S. corporation generally will be classified as a personal holding company for U.S. federal income tax purposes in a given taxable year if more than 50% of its ownership (by value) is concentrated, within a certain period of time, in five or fewer individuals (without regard to their citizenship or residency and including as individuals for this purpose certain entities such as certain tax-exempt organizations, pension funds, and charitable trusts), and at least 60% of its income is comprised of certain passive items. See the section titled "<u>United States Federal Tax Considerations</u> — Personal Holding Company Status" for more detailed information.

We are an emerging growth company and a smaller reporting company within the meaning of the rules adopted by the Securities and Exchange Commission, and if we take advantage of certain exemptions from disclosure requirements available to emerging growth companies and smaller reporting companies, this could make our securities less attractive to investors and may make it more difficult to compare our performance with other public companies.

We are an "emerging growth company" within the meaning of the rules adopted by the Securities and Exchange Commission, as modified by the JOBS Act, and we may take advantage of certain exemptions from various reporting requirements that are applicable to other public companies that are not emerging growth companies including, but not limited to, not being required to comply with the auditor internal controls attestation requirements of Section 404 of the Sarbanes-Oxley Act, reduced disclosure obligations regarding executive compensation in our periodic reports and proxy statements, and exemptions from the requirements of holding a nonbinding advisory vote on executive compensation and stockholder approval of any golden parachute payments not previously approved. As a result, our stockholders may not have access to certain information they may deem important. We could be an emerging growth company for up to five years, although circumstances could cause us to lose that status earlier, including if the market value of our Class A common stock held by non-affiliates exceeds \$700 million as of any June 30th before that time, in which case we would no longer be an emerging growth company as of the following December 31st. We cannot predict whether investors will find our securities less attractive because we will rely on these exemptions. If some investors find our securities less attractive as a result of our reliance on these exemptions, the trading prices of our securities may be lower than they otherwise would be, there may be a less active trading market for our securities and the trading prices of our securities may be more volatile.

Further, Section 102(b)(1) of the JOBS Act exempts emerging growth companies from being required to comply with new or revised financial accounting standards until private companies (that is, those that have not had a Securities Act registration statement declared effective or do not have a class of securities registered under the Exchange Act) are required to comply with the new or revised financial accounting standards. The JOBS Act provides that a company can elect to opt out of the extended transition period and comply with the requirements that apply to non-emerging growth companies but any such an election to opt out is irrevocable. We have elected not to opt out of such extended transition period, which means that when a standard is issued or revised and it has different application dates for public or private companies, we, as an emerging growth company, can adopt the new or revised standard at the time private companies adopt the new or revised standard. This may make comparison of our financial statements with another public company which is neither an emerging growth company nor an emerging growth company which has opted out of using the extended transition period difficult or impossible because of the potential differences in accounting standards used.

Additionally, we are a "smaller reporting company" as defined in Rule 10(f)(1) of Regulation S-K. Smaller reporting companies may take advantage of certain reduced disclosure obligations, including, among other things, providing only two years of audited financial statements. We will remain a smaller reporting company until the last day of the fiscal year in which (1) the market value of our common stock held by non-affiliates equals or exceeds \$250 million as of the end of the prior June 30th, or (2) our annual revenues equaled or exceeded \$100 million during such completed fiscal year and the market value of our common stock held by non-affiliates exceeds \$700 million as of the prior June 30th. To the extent we take advantage of such reduced disclosure obligations, it may also make comparison of our financial statements with other public companies difficult or impossible.

CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS

Certain statements in this prospectus may constitute "forward-looking statements" for purposes of the federal securities laws. Our forward-looking statements include, but are not limited to, statements regarding our or our management team's expectations, hopes, beliefs, intentions or strategies regarding the future. In addition, any statements that refer to projections, forecasts or other characterizations of future events or circumstances, including any underlying assumptions, are forward-looking statements. The words "anticipate," "believe," "continue," "could," "estimate," "expect," "intend," "may," "might," "plan," "possible," "potential," "predict," "project," "should," "would" and similar expressions may identify forward-looking statements, but the absence of these words does not mean that a statement is not forward-looking. Forward-looking statements in this prospectus may include, for example, statements about:

- our ability to select an appropriate target business or businesses in our target industry or otherwise;
- our ability to complete our initial business combination in our target industry or otherwise;
- our expectations around the performance of the prospective target business or businesses in the pharmaceutical and medical devices sectors;
- our success in retaining or recruiting, or changes required in, our officers, key employees or directors following our initial business combination;

- our officers and directors allocating their time to other businesses and potentially having conflicts of interest with our business or in approving our initial business combination, as a result of which they may then receive expense reimbursements;
- our potential ability to obtain additional financing to complete our initial business combination;
- our pool of prospective target businesses in the pharmaceutical industry;
- our ability to consummate an initial business combination due to the uncertainty resulting from the ongoing COVID-19 pandemic;
- the ability of our officers and directors to generate a number of potential business combination opportunities;
- our public securities' potential liquidity and trading;
- the lack of a market for our securities;
- the use of proceeds not held in the trust account or available to us from interest income on the trust account balance;
- the trust account not being subject to claims of third parties; or
- our financial performance following this offering.

The forward-looking statements contained in this prospectus are based on our current expectations and beliefs concerning future developments and their potential effects on us. There can be no assurance that future developments affecting us will be those that we have anticipated. These forward-looking statements involve a number of risks, uncertainties (some of which are beyond our control) or other assumptions that may cause actual results or performance to be materially different from those expressed or implied by these forward-looking statements. These risks and uncertainties include, but are not limited to, those factors described under the section of this prospectus entitled "Risk Factors." Should one or more of these risks or uncertainties materialize, or should any of our assumptions prove incorrect, actual results may vary in material respects from those projected in these forward-looking statements. We undertake no obligation to update or revise any forward-looking statements, whether as a result of new information, future events or otherwise, except as may be required under applicable securities laws.

USE OF PROCEEDS

We are offering 10,000,000 units at an offering price of \$10.00 per unit. We estimate that the net proceeds of this offering together with the funds we will receive from the sale of the placement warrants will be used as set forth in the following table.

	Without Over-Allotment Option	Over-Allotment Option Fully Exercised		
Gross proceeds				
Gross proceeds from units offered to public ⁽¹⁾	\$ 100,000,000	\$	115,000,000	
Gross proceeds from placement warrants offered in the private placement to the sponsor	 5,261,000		5,711,000	
Total gross proceeds	\$ 105,261,000	\$	120,711,000	
Estimated Offering expenses ⁽²⁾				
Underwriting commissions (1% of gross proceeds from units offered to public, excluding deferred portion and 100,000 shares agreed to be issued to the representative of the				
underwriters) ⁽³⁾	\$ 1,000,000	\$	1,150,000	
Legal fees and expenses	250,000		250,000	
Accounting fees and expenses	62,000		62,000	
SEC/FINRA Expenses	31,094		31,094	
Reimbursement to underwriters for expenses	100,000		100,000	
Nasdaq listing and filing fees	75,000		75,000	
Director and Officer liability insurance premiums	650,000		650,000	
Printing, engraving and road show expenses	50,000		50,000	
Miscellaneous expenses	42,906		42,906	
Total offering expenses (excluding underwriting commissions)	\$ 1,261,000	\$	1,261,000	
Proceeds after estimated offering expenses	\$ 103,000,000	\$	118,300,000	
Held in trust account ⁽³⁾	\$ 102,000,000	\$	117,300,000	
% of public offering size	102.0%		102.0%	
Not held in trust account	\$ 1,000,000	\$	1,000,000	

The following table shows the use of the approximately \$1,000,000 of net proceeds not held in the trust account. (4)

	Amount	% of Total*
Legal, accounting, due diligence, travel, and other expenses in connection with any business		
combination ⁽⁵⁾	\$ 500,000	50.0%
Legal and accounting fees related to regulatory reporting obligations	100,000	10.0%
Nasdaq Fees	56,500	5.7%
Payment for office space, utilities and secretarial and administrative support (\$10,000 per month		
for up to 18 months)	180,000	18.0%
Working capital to cover miscellaneous expenses	163,500	16.3%
Total	\$ 1,000,000	100.0%

^{*} May not total 100% due to rounding.

- (1) Includes amounts payable to public stockholders who properly redeem their shares in connection with our successful completion of our initial business combination.
- (2) A portion of the offering expenses will be paid from the proceeds of loans from our sponsor of up to \$300,000 as described in this prospectus. As of July 12, 2021, the Company had borrowed \$64,798 under the promissory note. These amounts will be repaid upon completion of this offering out of the \$1,261,000 of offering proceeds that has been allocated for the payment of offering expenses (other than underwriting commissions). In the event that offering expenses are more than as set forth in this table, they will be repaid using a portion of the \$1,000,000 of offering proceeds not held in the trust account and set aside for post-closing working capital expenses. In the event that offering expenses are less than set forth in this table, any such amounts will be used for post-closing working capital expenses.

- (3) The underwriters have agreed to defer underwriting commissions equal to 3.0% of the gross proceeds of this offering. Upon completion of our initial business combination, \$3,000,000, which constitutes the underwriters' deferred commissions (or \$3,450,000 if the underwriters' over-allotment option is exercised in full) will be paid to the underwriters from the funds held in the trust account, and the remaining funds, less amounts released to the trustee to pay redeeming stockholders, will be released to us and can be used to pay all or a portion of the purchase price of the business or businesses with which our initial business combination occurs or for general corporate purposes, including payment of principal or interest on indebtedness incurred in connection with our initial business combination, to fund the purchases of other companies or for working capital. The underwriters will not be entitled to any interest accrued on the deferred underwriting discounts and commissions.
- (4) These expenses are estimates only. Our actual expenditures for some or all of these items may differ from the estimates set forth herein. For example, we may incur greater legal and accounting expenses than our current estimates in connection with negotiating and structuring our initial business combination based upon the level of complexity of such business combination. In the event we identify an initial business combination target in a specific industry subject to specific regulations, we may incur additional expenses associated with legal due diligence and the engagement of special legal counsel. In addition, our staffing needs may vary and as a result, we may engage a number of consultants to assist with legal and financial due diligence. We do not anticipate any change in our intended use of proceeds, other than fluctuations among the current categories of allocated expenses, which fluctuations, to the extent they exceed current estimates for any specific category of expenses, would not be available for our expenses.
- (5) Includes estimated amounts that may also be used in connection with our initial business combination to fund a "no shop" provision and commitment fees for financing.

Of the net proceeds of this offering and the sale of the placement warrants, \$102,000,000 (or \$117,300,000 if the underwriters' over-allotment option is exercised in full), including \$3,000,000 (or \$3,450,000 if the underwriters' over-allotment option is exercised in full) of deferred underwriting commissions, will be placed in a trust account in the United States with Continental Stock Transfer & Trust Company, LLC acting as trustee and JP Morgan Chase Bank N.A. acting as investment manager, and will be invested only in U.S. government treasury bills with a maturity of 185 days or less or in money market funds meeting certain conditions under Rule 2a-7 under the Investment Company Act which invest only in direct U.S. government treasury obligations. We estimate that the pre-tax interest earned on the trust account will be approximately \$91,800 per year, assuming an interest rate of 0.09% per year and no exercise of the underwriter's over-allotment option; however, we can provide no assurance regarding this amount. Except with respect to interest earned on the funds held in the trust account that may be released to us to pay our tax obligations and up to \$100,000 of interest that may be used for our dissolution expenses, the proceeds from this offering and the sale of the placement warrants will not be released from the trust account until the earliest to occur of: (a) the completion of our initial business combination, (b) the redemption of any public shares properly submitted in connection with a stockholder vote to amend our amended and restated certificate of incorporation (A) to modify the substance or timing of our obligation to allow redemption in connection with our initial business combination or certain amendments to our charter prior thereto or to redeem 100% of our public shares if we do not complete our initial business combination within 12 months from the closing of this offering or during any Extension Period or (B) with respect to any other provision relating to stockholders' rights or pre-initial business combination activity, and (c) the redemption of our public shares if we are unable to complete our initial business combination within 18 months from the closing of this offering or during any Extension Period, subject to applicable law.

The net proceeds held in the trust account may be used as consideration to pay the sellers of a target business with which we ultimately complete our initial business combination. If our initial business combination is paid for using equity or debt securities, or not all of the funds released from the trust account are used for payment of the consideration in connection with our initial business combination, we may apply the balance of the cash released from the trust account for general corporate purposes, including for maintenance or expansion of operations of the post-transaction company, the payment of principal or interest due on indebtedness incurred in completing our initial business combination, to fund the purchase of other companies or for working capital. There is no limitation on our ability to raise funds privately or through loans in connection with our initial business combination.

We believe that amounts not held in trust will be sufficient to pay the costs and expenses to which such proceeds are allocated. This belief is based on the fact that while we may begin preliminary due diligence of a target business in connection with an indication of interest, we intend to undertake in-depth due diligence, depending on the circumstances of the relevant prospective business combination, only after we have negotiated and signed a letter of intent or other preliminary agreement that addresses the terms of an initial business combination. However, if our estimate of the costs of undertaking in-depth due diligence and negotiating an initial business combination is less than the actual amount necessary to do so, we may be required to raise additional capital, the amount, availability and cost of which is currently unascertainable. If we are required to seek additional capital, we could seek such additional capital through loans or additional investments from our sponsor, members of our management team or their affiliates, but such persons are not under any obligation to advance funds to, or invest in, us.

Commencing on the date of this prospectus, we have agreed to pay Aesther Healthcare Sponsor, LLC, our sponsor, a total of \$10,000 per month for office space, utilities and secretarial and administrative support. Upon completion of our initial business combination or our liquidation, we will cease paying these monthly fees.

Prior to the closing of this offering, our sponsor has agreed to loan us up to \$300,000 to be used for a portion of the expenses of this offering. As of July 12, 2021, we had borrowed \$64,798 (of up to \$300,000 available to us) under the promissory note with our sponsor. These loans are non-interest bearing, unsecured and are due at the earlier of June 30, 2022 or the closing of this offering. The loan will be repaid upon the closing of this offering out of the offering proceeds not held in the trust account.

In addition, in order to finance transaction costs in connection with an intended initial business combination, our sponsor or an affiliate of our sponsor or certain of our officers and directors may, but are not obligated to, loan us funds on a non-interest bearing basis as may be required. If we complete our initial business combination, we would repay such loaned amounts out of the proceeds of the trust account released to us. Otherwise, such loans would be repaid only out of funds held outside the trust account. In the event that our initial business combination does not close, we may use a portion of the working capital held outside the trust account to repay such loaned amounts but no proceeds from our trust account would be used to repay such loaned amounts. Up to \$1,500,000 of such loans may be convertible into warrants, at a price of \$1.00 per warrant at the option of the lender, upon consummation of our initial business combination. The warrants would be identical to the placement warrants. Other than as described above, the terms of such loans by our officers and directors, if any, have not been determined and no written agreements exist with respect to such loans. We do not expect to seek loans from parties other than our sponsor or an affiliate of our sponsor as we do not believe third parties will be willing to loan such funds and provide a waiver against any and all rights to seek access to funds in our trust account.

If we seek stockholder approval of our initial business combination and we do not conduct redemptions in connection with our initial business combination pursuant to the tender offer rules, our sponsor, initial stockholders, directors, officers, advisors or their affiliates may purchase public shares or public warrants in privately negotiated transactions or in the open market either prior to or following the completion of our initial business combination. There is no limit on the number of shares our initial stockholders, directors, officers or their affiliates may purchase in such transactions, subject to compliance with applicable law and Nasdaq rules. However, they have no current commitments, plans or intentions to engage in such transactions and have not formulated any terms or conditions for any such transactions. If they engage in such transactions, they will not make any such purchases when they are in possession of any material nonpublic information not disclosed to the seller or if such purchases are prohibited by Regulation M under the Exchange Act. We do not currently anticipate that such purchases, if any, would constitute a tender offer subject to the tender offer rules under the Exchange Act or a going-private transaction subject to the going-private rules under the Exchange Act; however, if the purchasers determine at the time of any such purchases that the purchases are subject to such rules, the purchasers will comply with such rules. Any such purchases will be reported pursuant to Section 13 and Section 16 of the Exchange Act to the extent such purchasers are subject to such reporting requirements. None of the funds held in the trust account will be used to purchase shares or public warrants in such transactions prior to completion of our initial business combination. See "Proposed Business — Permitted purchases of our securities" for a description of how our sponsor, initial stockholders, directors, officers or any of their affiliates will select which stockholders to purchase securities fr

The purpose of any such purchases of shares could be to vote such shares in favor of the initial business combination and thereby increase the likelihood of obtaining stockholder approval of the initial business combination or to satisfy a closing condition in an agreement with a target that requires us to have a minimum net worth or a certain amount of cash at the closing of our initial business combination, where it appears that such requirement would otherwise not be met. The purpose of any such purchases of public warrants could be to reduce the number of public warrants outstanding or to vote such warrants on any matters submitted to the warrant holders for approval in connection with our initial business combination. Any such purchases of our securities may result in the completion of our initial business combination that may not otherwise have been possible. In addition, if such purchases are made, the public "float" of our shares of Class A common stock and warrants may be reduced and the number of beneficial holders of our securities may be reduced, which may make it difficult to maintain or obtain the quotation, listing or trading of our securities on a national securities exchange.

In no event will we redeem our public shares unless our net tangible assets are at least \$5,000,001 either immediately prior to or upon consummation of our initial business combination and after payment of underwriters' fees and commissions (so that we are not subject to the SEC's "penny stock" rules) and the agreement for our initial business combination may require as a closing condition that we have a minimum net worth or a certain amount of cash. If too many public stockholders exercise their redemption rights so that we cannot satisfy the net tangible asset requirement or any net worth or cash requirements, we would not proceed with the redemption of our public shares or the initial business combination, and instead may search for an alternate business combination.

A public stockholder will be entitled to receive funds from the trust account only upon the earliest to occur of: (i) our completion of an initial business combination, (ii) the redemption of any public shares properly submitted in connection with a stockholder vote to amend our amended and restated certificate of incorporation (A) to modify the substance or timing of our obligation to allow redemption in connection with our initial business combination or certain amendments to our charter prior thereto or to redeem 100% of our public shares if we do not complete our initial business combination within 12 months from the closing of this offering or during any Extension Period or (B) with respect to any other provision relating to stockholders' rights or pre-initial business combination activity, and (iii) the redemption of our public shares if we are unable to complete our initial business combination within 12 months following the closing of this offering or during any Extension Period. In no other circumstances will a public stockholder have any right or interest of any kind to or in the trust account.

Our sponsor, officers and directors have entered into a letter agreement with us, and we anticipate that our director nominees will agree to enter into with us, a letter agreement, pursuant to which they have agreed to waive their redemption rights with respect to any founder shares and any public shares held by them in connection with the completion of our initial business combination. In addition, our initial stockholders have agreed to waive their rights to liquidating distributions from the trust account with respect to any founder shares held by them if we fail to complete our initial business combination within the prescribed time frame. However, if our sponsor or any of our officers, directors or affiliates acquires public shares in or after this offering, they will be entitled to liquidating distributions from the trust account with respect to such public shares if we fail to complete our initial business combination within the prescribed time frame.

DIVIDEND POLICY

We have not paid any cash dividends on our common stock to date and do not intend to pay cash dividends prior to the completion of our initial business combination. The payment of cash dividends in the future will be dependent upon our revenues and earnings, if any, capital requirements and general financial condition subsequent to completion of our initial business combination. If we increase or decrease the size of the offering we will effect a stock dividend or a share contribution back to capital or other appropriate mechanism, as applicable, with respect to our Class B common stock immediately prior to the consummation of the offering in such amount as to maintain the ownership of our initial stockholders at 20.0% of the issued and outstanding shares of our common stock (excluding the placement warrants and underlying securities, and the representative's shares and assuming the initial stockholders do not purchase units in this offering) upon the consummation of this offering. Further, if we incur any indebtedness in connection with our initial business combination, our ability to declare dividends may be limited by restrictive covenants we may agree to in connection therewith.

DILUTION

The difference between the public offering price per share of Class A common stock, assuming no value is attributed to the warrants included in the units we are offering pursuant to this prospectus or the placement warrant, and the pro forma net tangible book value per share of our Class A common stock after this offering constitutes the dilution to investors in this offering. Such calculation does not reflect any dilution associated with the sale and exercise of warrants, including the placement warrants, which would cause the actual dilution to the public stockholders to be higher, particularly where a cashless exercise is utilized. Net tangible book value per share is determined by dividing our net tangible book value, which is our total tangible assets less total liabilities (including the value of Class A common stock which may be redeemed for cash), by the number of outstanding shares of our Class A common stock.

At July 12, 2021, our net tangible book deficit was \$(39,798), or approximately \$(0.01) per share of common stock. For purposes of the dilution calculation, in order to present the maximum estimated dilution as a result of this offering, we have assumed the number of shares included in the units offered hereby will be deemed to be 10,000,000 shares of Class A common stock, and the price per share in this offering will be deemed to be \$10.00. After giving effect to the sale of 10,000,000 shares of Class A common stock included in the units we are offering by this prospectus (or 11,500,000 shares of Class A common stock if the underwriters' over-allotment option is exercised in full), the sale of the placement warrants and the deduction of underwriting commissions and estimated expenses of this offering, our pro forma net tangible book value at July 12, 2021 would have been \$(1,975,082), or approximately \$(0.76) per share (or (\$2,425,082), or \$(0.81) per share if the underwriters' over-allotment option is exercised in full), representing an immediate decrease in net tangible book value of \$(0.75) per share (or \$(0.80) per share if the underwriters' over-allotment option is exercised in full) to our initial stockholders as of the date of this prospectus and an immediate dilution of \$(10.76) per share or 107.6% (or \$(10.81) per share or 108.1% if the underwriters' over-allotment option is exercised in full).

The following table illustrates the dilution to the public stockholders on a per-share basis, no value is attributed to the warrants included in the units:

	No exe over-all opt		Exerci over-allo option i	otment	
Public offering price		\$ 10.00		\$	10.00
Net tangible book deficit before this offering	\$ (0.01)		\$ (0.01)		
Decrease attributable to public stockholders	(0.75)		(0.80)		
Pro forma net tangible book value after this offering		(0.76)			(0.81)
Dilution to public stockholders		\$ (10.76)		\$	(10.81)
Percentage of dilution to public stockholders		107.6%			108.1%
	74	 			

For purposes of presentation, we have reduced our pro forma net tangible book value after this offering by \$102,000,000 (or \$117,300,000, assuming the full exercise of the underwriters' over-allotment option) because holders of up to 100% of our public shares may redeem their shares for a pro rata share of the aggregate amount then on deposit in the trust account at a per-share redemption price equal to the amount in the trust account as set forth in our tender offer or proxy materials (initially anticipated to be the aggregate amount held in trust two days prior to the commencement of our tender offer or general meeting, including interest earned on the funds held in the trust account and not previously released to us to pay our taxes, divided by the number of common stock shares sold in this offering).

The following table sets forth information with respect to our initial stockholders and the public stockholders:

	Shares Pu	rurchased Total Co		ideration	Average Price		
	Number	Percentage	Amount	Percentage		Per Share	
Initial Stockholders ⁽¹⁾	2,500,000	19.8%	\$ 25,000	0.02%	\$	0.01	
Public Stockholders	10,000,000	79.4%	100,000,000	99.98%	\$	10.00	
Representative shares	100,000	0.8%		%	\$	0.00	
	12,600,000	100.0%	\$ 100,025,000	100.00%			

(1) Assumes no exercise of the underwriters' over-allotment option and the corresponding forfeiture of an aggregate of 375,000 shares of Class B common stock held by our sponsor.

The pro forma net tangible book value per share after the offering is calculated as follows:

	Without Over-allotment			With Over-allotment		
Numerator:						
Net tangible book value before this offering	\$	(39,798)	\$	(39,798)		
Net proceeds from this offering and sale of the placement warrants, net of expenses ⁽¹⁾		103,000,000		118,300,000		
Plus: Offering costs paid in advance, excluded from tangible book value		64,716		64,716		
Less: Deferred underwriting commissions		(3,000,000)		(3,450,000)		
Less: Proceeds held in trust subject to redemption ⁽²⁾		(102,000,000)		(117,300,000)		
	\$	(1,975,082)	\$	(2,425,082)		

	Without Over-allotment	With Over-allotment
Denominator:		
Shares of Class B common stock outstanding prior to this offering	2,875,000	2,875,000
Shares of Class B common stock forfeited if over-allotment is not exercised	(375,000)	_
Shares of Class A common stock included in the units offered	10,000,000	11,500,000
Shares of common stock issued to underwriters	100,000	100,000
Less: Shares subject to redemption	(10,000,000)	(11,500,000)
	2,600,000	2,975,000

- (1) Expenses applied against gross proceeds include offering expenses of \$1,261,000 and underwriting commissions of \$1,000,000 (or \$1,150,000 if the over-allotment option is exercised in full) (excluding deferred underwriting fees). See "Use of Proceeds."
- (2) If we seek stockholder approval of our initial business combination and we do not conduct redemptions in connection with our initial business combination pursuant to the tender offer rules, our sponsor, initial stockholders, directors, executive officers, advisors or their affiliates may purchase public shares or public warrants in privately negotiated transactions or in the open market either prior to or following the completion of our initial business combination. In the event of any such purchases of our shares prior to the completion of our initial business combination, the number of shares of Class A common stock subject to redemption will be reduced by the amount of any such purchases, increasing the pro forma net tangible book value per share. See "Proposed Business Effecting Our Initial Business Combination Permitted Purchases of Our Securities."

CAPITALIZATION

The following table sets forth our capitalization at July 12, 2021, and as adjusted to give effect to the filing of our amended and restated certificate of incorporation, the sale of our units in this offering and the sale of the placement warrants and the application of the estimated net proceeds derived from the sale of such securities, assuming no exercise by the underwriters of their over-allotment option:

	July 12, 2021			21
		Actual		As Adjusted
Notes payable to related party $^{(1)}$	\$	(64,798)	\$	_
Redeemable Class A Common Stock ⁽²⁾		_		100,000,000
Deferred underwriting commissions		<u> </u>		3,000,000
Class A common stock, \$0.0001 par value, 125,000,000 shares authorized; -0- and 10,000,000 shares are subject to possible redemption, respectively		_		_
Preferred stock, \$0.0001 par value, 1,250,000 shares authorized; none issued and outstanding, actual and as adjusted		_		_
Class A common stock, \$0.0001 par value, 125,000,000 shares authorized; -0- and 100,000 shares issued and outstanding, actual and as adjusted, respectively		_		10
Class B common stock, \$0.0001 par value, 12,500,000 shares authorized, 2,875,000 and 2,500,000				
shares issued and outstanding, actual and as adjusted, respectively $^{(3)}$		288		250
Additional paid-in capital ⁽⁴⁾		24,712		_
Accumulated deficit		(82)		(1,974,822)
Total stockholders' equity (deficit)	\$	24,918	\$	(1,975,082)
Total capitalization	\$	(39,880)	\$	101,024,918

- (1) Our sponsor has agreed to loan us up to \$300,000 to be used for a portion of the expenses of this offering. The "as adjusted" information gives effect to the repayment of any loans made under this note out of the proceeds from this offering and the sale of the placement warrants. As of July 12, 2021, we had drawn down \$64,798 under the promissory note with our sponsor to be used for a portion of the expenses of this offering.
- (2) Upon the completion of our initial business combination, we will provide our stockholders with the opportunity to redeem their public shares for cash equal to their pro rata share of the aggregate amount then on deposit in the trust account as of two business days prior to the consummation of the initial business combination including interest earned on the funds held in the trust account and not previously released to us to pay our taxes. While redemptions cannot cause our net tangible assets to fall below \$5,000,001, all common stock shares are redeemable and classified as such on the balance sheet until such date that a redemption event takes place.
- (3) Actual share amount is prior to any forfeiture of founder shares by our sponsor and as adjusted amount assumes no exercise of the underwriters' overallotment option and the forfeiture of 375,000 Class B common stock shares.
- (4) The "as adjusted" additional paid-in capital calculation is adjusted to zero, with the off-setting balance recorded to accumulated deficit since additional paid-in capital cannot be less than zero.

MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Overview

We are a newly-organized blank check company incorporated as a Delaware corporation and formed for the purpose of effecting a merger, capital stock exchange, asset acquisition, stock purchase, reorganization or similar business combination with one or more businesses. While our efforts to identify a target business may span many industries and regions worldwide, we intend to focus our search on industries that complement our management team's background and to capitalize on the ability of our management team to identify and acquire a business, focusing on the pharmaceutical sector where our management team has extensive experience. We have not selected any specific business combination target and we have not, nor has anyone on our behalf, initiated any substantive discussions, directly or indirectly, with any business combination target. We intend to effectuate our initial business combination using cash from the proceeds of this offering and the sale of the placement warrants, the proceeds of the sale of our shares in connection with our initial business combination (including pursuant to backstop agreements we may enter into following the consummation of this offering or otherwise), shares issued to the owners of the target, debt issued to bank or other lenders or the owners of the target, or a combination of the foregoing.

The issuance of additional shares in connection with an initial business combination to the owners of the target or other investors:

- may significantly dilute the equity interest of investors in this offering, which dilution would increase if the anti-dilution provisions in the Class B common stock resulted in the issuance of Class A shares on a greater than one-to-one basis upon conversion of the Class B common stock;
- may subordinate the rights of holders of our common stock if preferred stock is issued with rights senior to those afforded our common stock;
- could cause a change in control if a substantial number of shares of our common stock is issued, which may affect, among other things, our ability to use our net operating loss carry forwards, if any, and could result in the resignation or removal of our present officers and directors;
- may have the effect of delaying or preventing a change of control of us by diluting the stock ownership or voting rights of a person seeking to obtain control of us; and
- may adversely affect prevailing market prices for our Class A common stock and/or warrants.

Similarly, if we issue debt securities or otherwise incur significant debt to bank or other lenders or the owners of a target, it could result in:

- default and foreclosure on our assets if our operating revenues after an initial business combination are insufficient to repay our debt obligations;
- acceleration of our obligations to repay the indebtedness even if we make all principal and interest payments when due if we breach certain covenants that require the maintenance of certain financial ratios or reserves without a waiver or renegotiation of that covenant;
- our immediate payment of all principal and accrued interest, if any, if the debt is payable on demand;
- our inability to obtain necessary additional financing if the debt contains covenants restricting our ability to obtain such financing while the debt is outstanding;
- our inability to pay dividends on our common stock;

- using a substantial portion of our cash flow to pay principal and interest on our debt, which will reduce the funds available for dividends on our
 common stock if declared, our ability to pay expenses, make capital expenditures and acquisitions, and fund other general corporate purposes;
- limitations on our flexibility in planning for and reacting to changes in our business and in the industry in which we operate;
- increased vulnerability to adverse changes in general economic, industry and competitive conditions and adverse changes in government regulation;
- limitations on our ability to borrow additional amounts for expenses, capital expenditures, acquisitions, debt service requirements, and execution of our strategy; and
- other purposes and other disadvantages compared to our competitors who have less debt.

As indicated in the accompanying financial statements, at July 12, 2021, we had \$25,000 cash and deferred offering costs of \$64,716. Further, we expect to continue to incur significant costs in the pursuit of our initial business combination plans. We cannot assure you that our plans to raise capital or to complete our initial business combination will be successful.

Results of Operations and Known Trends or Future Events

We have neither engaged in any operations nor generated any revenues to date. Our only activities since inception have been organizational activities and those necessary to prepare for this offering. Following this offering, we will not generate any operating revenues until after completion of our initial business combination. We will generate non-operating income in the form of interest income on cash and cash equivalents after this offering. There has been no significant change in our financial or trading position and no material adverse change has occurred since the date of our audited financial statements. After this offering, we expect to incur increased expenses as a result of being a public company (for legal, financial reporting, accounting and auditing compliance), as well as expenses as we conduct due diligence on prospective business combination candidates. We expect our expenses to increase substantially after the closing of this offering.

Liquidity and Capital Resources

As indicated in the accompanying financial statements, at July 12, 2021, we had \$25,000 cash and a working capital deficit of \$39,798. Further, we have incurred and expect to continue to incur significant costs in pursuit of our financing and acquisition plans. We cannot assure you that our plans to raise capital or to consummate an initial business combination will be successful. These factors, among others, raise substantial doubt about our ability to continue as a going concern.

Our liquidity needs have been satisfied prior to the completion of this offering through a capital contribution from our sponsor of \$25,000 for the founder shares and up to \$300,000 in loans available from our sponsor under an unsecured promissory note (of which \$64,798 had been borrowed as of July 12, 2021). We estimate that the net proceeds from (i) the sale of the units in this offering, after deducting offering expenses of approximately \$1,261,000, underwriting commissions of \$1,000,000 (\$1,150,000 if the underwriters' over-allotment option is exercised in full) (excluding deferred underwriting commissions of \$3,000,000 (or \$3,450,000 if the underwriters' over-allotment option is exercised in full), will be \$103,000,000 (or \$118,300,000 if the underwriters' over-allotment option is exercised in full), will be \$103,000,000 (or \$118,300,000 if the underwriters' over-allotment option is exercised in full) will be held in the trust account, which includes \$3,000,000 (or \$3,450,000 if the underwriters' over-allotment option is exercised in full) of deferred underwriting commissions. The proceeds held in the trust account will be invested only in U.S. government treasury obligations with a maturity of 185 days or less or in money market funds meeting certain conditions under Rule 2a-7 under the Investment Company Act which invest only in direct U.S. government treasury obligations. The remaining approximately \$1,000,000 will not be held in the trust account. In the event that our offering expenses exceed our estimate of \$1,261,000, we may fund such excess with funds not to be held in the trust account. In such case, the amount of funds we intend to be held outside the trust account would decrease by a corresponding amount. Conversely, in the event that the offering expenses are less than our estimate of \$1,261,000, the amount of funds we intend to be held outside the trust account would increase by a corresponding amount.

We intend to use substantially all of the funds held in the trust account, including any amounts representing interest earned on the trust account (less deferred underwriting commissions), to complete our initial business combination. We may withdraw interest to pay taxes. We estimate our annual franchise tax obligations, based on the number of shares of our common stock authorized and outstanding after the completion of this offering, to be \$200,000, which is the maximum amount of annual franchise taxes payable by us as a Delaware corporation per annum, which we may pay from funds from this offering held outside of the trust account or from interest earned on the funds held in our trust account and released to us for this purpose. Our annual income tax obligations will depend on the amount of interest and other income earned on the amounts held in the trust account. We expect the interest earned on the amount in the trust account will be sufficient to pay our income taxes. To the extent that our capital stock or debt is used, in whole or in part, as consideration to complete our initial business combination, the remaining proceeds held in the trust account will be used as working capital to finance the operations of the target business or businesses, make other acquisitions and pursue our growth strategies.

Prior to the completion of our initial business combination, we will have available to us the approximately \$1,000,000 of proceeds held outside the trust account. We will use these funds to identify and evaluate target businesses, perform business due diligence on prospective target businesses, travel to and from the offices, plants or similar locations of prospective target businesses or their representatives or owners, review corporate documents and material agreements of prospective target businesses, and structure, negotiate and complete an initial business combination.

In order to fund working capital deficiencies or finance transaction costs in connection with an intended initial business combination, our sponsor or an affiliate of our sponsor or certain of our officers and directors may, but are not obligated to, loan us funds on a non-interest bearing basis as may be required. If we complete our initial business combination, we would repay such loaned amounts. In the event that our initial business combination does not close, we may use a portion of the working capital held outside the trust account to repay such loaned amounts but no proceeds from our trust account would be used for such repayment. Up to \$1,500,000 of such loans may be convertible into warrants, at a price of \$1.00 per warrant at the option of the lender, upon consummation of our initial business combination. The warrants would be identical to the placement warrants. Other than as described above, the terms of such loans by our officers and directors, if any, have not been determined and no written agreements exist with respect to such loans. We do not expect to seek loans from parties other than our sponsor or an affiliate of our sponsor as we do not believe third parties will be willing to loan such funds and provide a waiver against any and all rights to seek access to funds in our trust account.

We expect our primary liquidity requirements during that period to include approximately \$500,000 for legal, accounting, due diligence, travel and other expenses associated with structuring, negotiating and documenting successful business combinations; \$100,000 for legal and accounting fees related to regulatory reporting requirements; \$56,500 for Nasdaq Fees: \$180,000 for office space, utilities and secretarial and administrative support; and approximately \$163,500 for working capital that will be used for miscellaneous expenses and reserves.

These amounts are estimates and may differ materially from our actual expenses. In addition, we could use a portion of the funds not being placed in trust to pay commitment fees for financing, fees to consultants to assist us with our search for a target business or as a down payment or to fund a "no-shop" provision (a provision designed to keep target businesses from "shopping" around for transactions with other companies or investors on terms more favorable to such target businesses) with respect to a particular proposed initial business combination, although we do not have any current intention to do so. If we entered into an agreement where we paid for the right to receive exclusivity from a target business, the amount that would be used as a down payment or to fund a "no-shop" provision would be determined based on the terms of the specific business combination and the amount of our available funds at the time. Our forfeiture of such funds (whether as a result of our breach or otherwise) could result in our not having sufficient funds to continue searching for, or conducting due diligence with respect to, prospective target businesses.

We do not believe we will need to raise additional funds following this offering in order to meet the expenditures required for operating our business. However, if our estimates of the costs of identifying a target business, undertaking in-depth due diligence and negotiating an initial business combination are less than the actual amount necessary to do so, we may have insufficient funds available to operate our business prior to our initial business combination. Moreover, we may need to obtain additional financing either to complete our initial business combination or because we become obligated to redeem a significant number of our public shares upon completion of our initial business combination, in which case we may issue additional securities or incur debt in connection with such business combination. In addition, we intend to target businesses larger than we could acquire with the net proceeds of this offering and the sale of the placement warrants, and may as a result be required to seek additional financing to complete such proposed initial business combination. Subject to compliance with applicable securities laws, we would only complete such financing simultaneously with the completion of our initial business combination. If we are unable to complete our initial business combination because we do not have sufficient funds available to us, we will be forced to cease operations and liquidate the trust account. In addition, following our initial business combination, if cash on hand is insufficient, we may need to obtain additional financing in order to meet our obligations.

Controls and Procedures

We are not currently required to maintain an effective system of internal controls as defined by Section 404 of the Sarbanes-Oxley Act. We will be required to comply with the internal control requirements of the Sarbanes-Oxley Act for the fiscal year ending December 31, 2022. Only in the event that we are deemed to be a large accelerated filer or an accelerated filer would we be required to comply with the independent registered public accounting firm attestation requirement. Further, for as long as we remain an emerging growth company as defined in the JOBS Act, we intend to take advantage of certain exemptions from various reporting requirements that are applicable to other public companies that are not emerging growth companies including, but not limited to, not being required to comply with the independent registered public accounting firm attestation requirement.

Prior to the closing of this offering, we have not completed an assessment, nor has our independent registered public accounting firm tested our systems, of internal controls. We expect to assess the internal controls of our target business or businesses prior to the completion of our initial business combination and, if necessary, to implement and test additional controls as we may determine are necessary in order to state that we maintain an effective system of internal controls. A target business may not be in compliance with the provisions of the Sarbanes-Oxley Act regarding the adequacy of internal controls. Many small and mid-sized target businesses we may consider for our initial business combination may have internal controls that need improvement in areas such as:

- staffing for financial, accounting and external reporting areas, including segregation of duties;
- reconciliation of accounts;
- proper recording of expenses and liabilities in the period to which they relate;
- evidence of internal review and approval of accounting transactions;
- documentation of processes, assumptions and conclusions underlying significant estimates; and
- documentation of accounting policies and procedures.

Because it will take time, management involvement and perhaps outside resources to determine what internal control improvements are necessary for us to meet regulatory requirements and market expectations for our operation of a target business, we may incur significant expense in meeting our public reporting responsibilities, particularly in the areas of designing, enhancing, or remediating internal and disclosure controls. Doing so effectively may also take longer than we expect, thus increasing our exposure to financial fraud or erroneous financing reporting.

Once our management's report on internal controls is complete, we will retain our independent registered public accounting firm to audit and render an opinion on such report when required by Section 404 of the Sarbanes-Oxley Act. The independent registered public accounting firm may identify additional issues concerning a target business's internal controls while performing their audit of internal control over financial reporting.

Quantitative and Qualitative Disclosures about Market Risk

The net proceeds of this offering and the sale of the placement warrants held in the trust account will be invested in U.S. government treasury bills with a maturity of 185 days or less or in money market funds meeting certain conditions under Rule 2a-7 under the Investment Company Act which invest only in direct U.S. government treasury obligations. Due to the short-term nature of these investments, we believe there will be no associated material exposure to interest rate risk.

Related Party Transactions

On June 30, 2021, our sponsor purchased 2,875,000 founder shares for an aggregate purchase price of \$25,000, or approximately \$0.009 per share. The number of founder shares issued was determined based on the expectation that such founder shares would represent 20% of the outstanding shares upon completion of this offering (excluding the placement warrants and underlying securities, and the representative's shares). The per share purchase price of the founder shares was determined by dividing the amount of cash contributed to the company by the aggregate number of founder shares issued. If we increase or decrease the size of the offering we will effect a stock dividend or a share contribution back to capital or other appropriate mechanism, as applicable, with respect to our Class B common stock immediately prior to the consummation of the offering in such amount as to maintain the ownership of our initial stockholders at 20% of the issued and outstanding shares of our common stock (excluding the placement warrants and underlying securities, and the representative's shares and assuming the initial stockholders do not purchase units in this offering) upon the consummation of this offering. Up to 375,000 founder shares held by our sponsor are subject to forfeiture by our sponsor depending on the extent to which the underwriters' over-allotment option is exercised. The founder shares (including the Class A common stock issuable upon exercise thereof) may not, subject to certain limited exceptions, be transferred, assigned or sold by the holder.

Commencing on the date of this prospectus, we have agreed to pay Aesther Healthcare Sponsor, LLC, our sponsor, a total of \$10,000 per month for office space, utilities and secretarial and administrative support. Upon completion of our initial business combination or our liquidation, we will cease paying these monthly fees.

Our sponsor, officers and directors, or any of their respective affiliates, will be reimbursed for any out-of-pocket expenses incurred in connection with activities on our behalf such as identifying potential target businesses and performing due diligence on suitable business combinations. Our audit committee will review on a quarterly basis all payments that were made to our sponsor, officers or directors or our or their affiliates and will determine which expenses and the amount of expenses that will be reimbursed. There is no cap or ceiling on the reimbursement of out-of-pocket expenses incurred by such persons in connection with activities on our behalf.

Prior to the consummation of this offering, our sponsor has agreed to loan us up to \$300,000 to be used for a portion of the expenses of this offering. These loans are non-interest bearing, unsecured and are due at the earlier of June 30, 2022 or the closing of this offering. As of July 12, 2021, we had borrowed \$64,798 from our sponsor. The loan will be repaid upon the closing of this offering out of the \$1,261,000 of offering proceeds that has been allocated to the payment of offering expenses (other than underwriting commissions).

In addition, in order to finance transaction costs in connection with an intended initial business combination, our sponsor or an affiliate of our sponsor or certain of our officers and directors may, but are not obligated to, loan us funds as may be required. If we complete our initial business combination, we would repay such loaned amounts. In the event that our initial business combination does not close, we may use a portion of the working capital held outside the trust account to repay such loaned amounts but no proceeds from our trust account would be used for such repayment. Up to \$1,500,000 of such loans may be convertible into warrants, at a price of \$1.00 per warrant at the option of the lender, upon consummation of our initial business combination. The warrants would be identical to the placement warrants. The terms of such loans by our officers and directors, if any, have not been determined and no written agreements exist with respect to such loans. We do not expect to seek loans from parties other than our sponsor or an affiliate of our sponsor as we do not believe third parties will be willing to loan such funds and provide a waiver against any and all rights to seek access to funds in our trust account.

Our sponsor has agreed to purchase an aggregate of 5,261,000 placement warrants at a price of \$1.00 per warrant for an aggregate purchase price of \$5,261,000. If the over-allotment option is exercised in full, the amount of placement warrants sold will be 5,711,000 for an aggregate purchase price of \$5,711,000. Each placement warrant is exercisable to purchase one whole share of common stock at \$11.50 per share. There will be no redemption rights or liquidating distributions from the trust account with respect to the founder shares or the placement warrants, which will expire worthless if we do not consummate a business combination within 12 months from the closing of this offering or during any Extension Period. The placement warrants are identical to the warrants sold as part of the units sold in this offering except that the placement warrants (a) will not be transferable, assignable or saleable until 30 days after the consummation of our initial business combination except to permitted transferees, and (b) will be entitled to registration rights.

Our initial stockholders have agreed to waive their redemption rights with respect to their founder shares (i) in connection with the consummation of a business combination, (ii) in connection with a stockholder vote to amend our amended and restated certificate of incorporation to modify the substance or timing of our obligation to allow redemption in connection with our initial business combination or certain amendments to our charter prior thereto or to redeem 100% of our public shares if we do not complete our initial business combination within 18 months from the closing of this offering or during any Extension Period and (iii) if we fail to consummate a business combination within 12 months from the closing of this offering or during any Extension Period or if we liquidate prior to the expiration of the 12-month period (or during any Extension Period). However, our initial stockholders will be entitled to redemption rights with respect to any public shares held by them if we fail to consummate a business combination or liquidate within the 12-month period (or during any Extension Period).

Pursuant to a registration rights agreement we will enter into with our initial stockholders on or prior to the closing of this offering, we may be required to register certain securities for sale under the Securities Act. These holders, and holders of units issued upon conversion of working capital loans, if any, are entitled under the registration rights agreement to make up to three demands that we register certain of our securities held by them for sale under the Securities Act and to have the securities covered thereby registered for resale pursuant to Rule 415 under the Securities Act. In addition, these holders have the right to include their securities in other registration statements filed by us. We will bear the costs and expenses of filing any such registration statements. See the section of this prospectus entitled "Certain Relationships and Related Party Transactions."

Off-Balance Sheet Arrangements; Commitments and Contractual Obligations; Quarterly Results

As of July 12, 2021, we did not have any off-balance sheet arrangements as defined in Item 303(a)(4)(ii) of Regulation S-K and did not have any commitments or contractual obligations. No unaudited quarterly operating data is included in this prospectus, as we have conducted no operations to date.

JOBS Act

On April 5, 2012, the JOBS Act was signed into law. The JOBS Act contains provisions that, among other things, relax certain reporting requirements for qualifying public companies. We will qualify as an "emerging growth company" and under the JOBS Act will be allowed to comply with new or revised accounting pronouncements based on the effective date for private (not publicly traded) companies. We are electing to delay the adoption of new or revised accounting standards, and as a result, we may not comply with new or revised accounting standards on the relevant dates on which adoption of such standards is required for non-emerging growth companies. As a result, our financial statements may not be comparable to companies that comply with new or revised accounting pronouncements as of public company effective dates.

Additionally, we are in the process of evaluating the benefits of relying on the other reduced reporting requirements provided by the JOBS Act. Subject to certain conditions set forth in the JOBS Act, if, as an "emerging growth company", we choose to rely on such exemptions we may not be required to, among other things, (i) provide an independent registered public accounting firm's attestation report on our system of internal controls over financial reporting pursuant to Section 404, (ii) provide all of the compensation disclosure that may be required of non-emerging growth public companies under the Dodd-Frank Wall Street Reform and Consumer Protection Act, (iii) comply with any requirement that may be adopted by the PCAOB regarding mandatory audit firm rotation or a supplement to the report of independent registered public accounting firm providing additional information about the audit and the financial statements (auditor discussion and analysis), and (iv) disclose certain executive compensation related items such as the correlation between executive compensation and performance and comparisons of the Chief Executive Officers' compensation to median employee compensation. These exemptions will apply for a period of five years following the completion of this offering or until we are no longer an "emerging growth company," whichever is earlier.

PROPOSED BUSINESS

General

We are a newly-organized blank check company incorporated in June 2021 as a Delaware corporation whose business purpose is to effect a merger, capital stock exchange, asset acquisition, stock purchase, reorganization or similar business combination with one or more businesses, which we refer to as our initial business combination. To date, our efforts have been limited to organizational activities as well as activities related to this offering. We have not selected any specific business combination target and we have not, nor has anyone on our behalf, engaged in any substantive discussions, directly or indirectly, with any business combination target with respect to an initial business combination with us.

While we may pursue an initial business combination opportunity in any business, industry, sector or geographical location, we intend to focus on industries that complement our management team's background and to capitalize on the ability of our management team to identify and acquire a business focusing on the pharmaceutical sector where our management team has extensive experience.

Our Management Team

Our management team is led by Suren Ajjarapu, our Chairman of the Board and Chief Executive Officer, and Howard A. Doss, our Chief Financial Officer.

Mr. Ajjarapu has served as our Chairman and Chief Executive Officer since our inception in June 2021. He has served as the Chairman of the Board, Chief Executive Officer and Secretary of TRxADE Health, Inc., formerly Trxade Group, Inc. (NASDAQ:MEDS)("TRxADE") since its acquisition of Trxade Group, Inc., a Nevada corporation ("Trxade Nevada") (our predecessor company) on January 8, 2014, and as the Chairman of the Board, Chief Executive Officer and Secretary of Trxade Nevada since its inception. Since March 2021, Mr. Ajjarapu has served on the Board of OceanTech Acquisitions I Corp, a Special Purpose Acquisition Company (SPAC)(NASDAQ:OTECU). Mr. Ajjarapu has served on the Board of Kano Energy, Inc., a private company, which is involved in developing renewable natural gas sites in USA since 2018. Mr. Ajjarapu was a Founder, CEO and Chairman of Sansur Renewable Energy, Inc., a company involved in developing wind power sites in the Midwest, United States, from 2009 to 2012. Mr. Ajjarapu was a Founder, President and Director of Aemetis, Inc., a biofuels company (AMTX.OB) and a Founder, Chairman and Chief Executive Officer of International Biofuels, a subsidiary of Aemetis, Inc., from 2006 to 2009. Mr. Ajjarapu was Co-Founder, COO, and Director Global Information Technology, Inc., an IT outsourcing and systems design company, headquartered in Tampa, Florida with major operations in India from 1995 to 2006. Mr. Ajjarapu holds an MS in Environmental engineering from South Dakota State University, Brookings, South Dakota, and an MBA from the University of South Florida, specializing in International Finance and Management. Mr. Ajjarapu is also a graduate of the Venture Capital and Private Equity program at Harvard University.

Mr. Doss has served as our Chief Financial Officer since our inception in June 2021. Since January 2014, he has served as the Chief Financial Officer of TRxADE. Since January 2014. Mr. Doss has served in a variety of capacities with accounting and investment firms. He joined the staff of Seidman & Seidman (BDO Seidman, Dallas) in 1977 and, in 1980, he joined the investment firm Van Kampen Investments, opening the firm's southeast office in Tampa, Florida in 1982. He remained with the firm until 1996 when he joined Franklin Templeton to develop corporate retirement plan distribution. After working for the Principal Financial Group office in Tampa, Florida, Mr. Doss was City Executive for U.S. Trust in Sarasota, Florida, responsible for high net worth individuals. He retired from that position in 2009. He served as CFO and Director for Sansur Renewable Energy, an alternative energy development company, from 2010 to 2012. Mr. Doss has also served as President of STARadio Corp. since 2005. Mr. Doss is a member of the America Institute of CPA's. He is a graduate of Illinois Wesleyan University.

We have also assembled a group of independent directors who will provide public company governance, executive leadership, operational oversight, private equity investment management and capital markets experience. Our board members have extensive experience, having served as directors or officers for numerous publicly-listed and privately-owned companies. Our directors have experience with acquisitions, divestitures and corporate strategy and implementation, which we believe will significantly benefit us as we evaluate potential acquisition or merger candidates as well as following the completion of our initial business combination. Below is biographical information for our independent directors:

Mr. Michael L. Peterson, one of our directors, was previously the CEO of PEDEVCO Corp. (NYSE American:PED), a public company engaged primarily in the acquisition, exploration, development and production of oil and natural gas shale plays in the US, formerly served as Interim President and CEO and as director of Pacific Energy Development, as a director of Aemetis, Inc. (formerly AE Biofuels Inc.), a Cupertino, California-based global advanced biofuels and renewable commodity chemicals company (AMTX.OB), and as Chairman and Chief Executive Officer of Nevo Energy, Inc. (NEVE) (formerly Solargen Energy, Inc.), a Cupertino, California-based developer of utility-scale solar farms which he helped form. Mr. Peterson served as a managing partner of American Institutional Partners, a venture investment fund based in Salt Lake City. He served as a First Vice President at Merrill Lynch, where he helped establish a new private client services division to work exclusively with high net worth investors. Mr. Peterson was employed by Goldman Sachs & Co. in a variety of positions and roles, including as a Vice President. His skills in managing businesses in public corporations, financial planning and strategic management will be a great asset for the target company.

Mr. Venkatesh Srinivasan has a tremendous amount of experience in the pharmaceutical industry and currently serves as President of Micro Labs USA and previously served as President of Rising Pharma, USA and as President and CEO of Ascend Laboratories, USA where he built the business from \$30 million (8 products, 12 people) to \$302 million (50+ products, 29 people) in six years, including building a new team and strengthening processes and systems. In addition, Mr. Srinivasan served as a Director at Pfizer India. His skillset will be extremely useful as we conduct due diligence works on target companies.

Mr. Donald G. Fell, brings along a wealth of experience in the field of economics and business to the Company, which is instrumental in executing the Company's business strategy. For over 7 years, Mr. Fell has served as an independent director of TRxADE HEALTH, INC. He is presently Professor and Institute Director for the Davis, California-based Foundation for Teaching Economics and adjunct professor of economics for the University of Colorado, Colorado Springs. Mr. Fell held positions with the University of South Florida as a member of the Executive MBA faculty, Director of Executive and Professional Education and Senior Fellow of the Public Policy Institute from 1995 to 2012. Mr. Fell was also a visiting professor at the University of LaRochelle, France, and an adjunct professor of economics at both Illinois State University and The Ohio State University. Mr. Fell holds undergraduate and graduate degrees in economics from Indiana State University and is all but dissertation (ABD) in economics from Illinois State University. Through his work with the Foundation for Teaching Economics and the University of Colorado, Colorado Springs he has overseen graduate institutes on economic policy and environmental economics in 44 states, throughout Canada, the Islands and Eastern Europe.

Mr. Siva Saravanan has more than 20 years of experience steering digital strategies and technology solutions for businesses. Mr. Saravanan is the Chief Digital Officer at Wavestone US, helping Fortune 1000 business and technology leaders accelerate digital transformation. Prior to joining Wavestone US, Siva was Chief Information Officer and SVP of Business Operations at Reviver, an exciting IoT start-up that creates connected digital license plates to enable true autonomous driving. He designed customer digital experiences, unified commerce, supply chain, field service operations and the digital agenda for Reviver. Siva was also VP for IT Digital Transformation and Program Delivery at Aristocrat Technologies. While at Aristocrat Technologies, he led the transformation of business systems for a \$3 billion leading high-tech gaming manufacturer. As a management consultant at Fitbit, he helped with strategies to scale up their technology infrastructure. Siva spent many years at Verifone as a Senior Director supporting technology operations in 40+ countries and also taking on delivery responsibilities. At VeriFone, he built a world-class global integrated supply chain network for agility and efficiency. Mr. Saravanan holds a M.S. in Systems Engineering from Tennessee State University and B.S. in Mechanical Engineering from Annamalai University in Chidambaram, India. He is on the Advisory Board of NishTech Inc., a digital commerce company and the Advisory Council of George Washington University School of Business Digital Program. Siva is a member of Forbes Technology Council contributing regularly.

We believe our management team is well positioned to take advantage of the growing set of acquisition opportunities focused on the pharmaceutical and medical devices sectors and that our contacts and relationships, ranging from owners and management teams of private and public companies, private equity funds, investment bankers, attorneys, to accountants and business brokers will allow us to generate an attractive transaction for our stockholders.

In addition, we may engage the services of financial advisory services in connection with this offering, which may become a member of our sponsor and/or invest in the private placement.

The past performance of the members of our management team, our financial advisor or their affiliates is not a guarantee that we will be able to identify a suitable candidate for our initial business combination or of success with respect to any business combination we may consummate. You should not rely on the historical record of the performance of our management team or any of its affiliates' performance as indicative of our future performance.

Business Strategy and Target Industries

While we may pursue an acquisition in any business industry or sector, we intend to focus our efforts identifying businesses primarily in the pharmaceutical and medical devices sectors. We will seek to enter into a business combination with a business that is well positioned to leverage the industry expertise, insights and relationships of the sponsor and our board of directors to create significant opportunities for value creation and improved financial performance.

The pharmaceutical and healthcare sectors have emerged into the public limelight and focus during the COVID-19 pandemic. It is, broadly speaking, a space defined by intense capital requirements, regulatory challenges and a delayed entrance to the market. While institutional investors may see the above features as a barrier to entry and success on returns, our team is confident that they are instead strong arguments for investment into the industry. Highly notable exits in the space have featured many successful companies developing novel treatments in rare disease, orphan drug populations and gene therapy. We will be focusing our efforts on the pharmaceutical sector (both innovative and generic drugs) and the MedTech sector.

3. Pharmaceutical Sector:

The global pharmaceutical market value of \$1.3 trillion in 2018 is expected to cross \$1.5 trillion by 2023 with a 4-5% annual growth globally, according to a January 2019 report by the IQVIA Institute for Human Data Science. Despite the pandemic's impact on global business, we believe the industry is very resilient and robust. The spectrum for drug consumption and pursuant development remains wide - aging populations globally are expected to drive high volume consumption generic drugs, while it is estimated that only 5% of the over 7,000 recognized rare diseases have U.S. Food and Drug Administration (FDA) approved treatments (according to the National Organization for Rare Diseases and the FDA.

There is still some uncertainty about how further regulation may play out as the U.S. government grapples with other high priority issues following the 2020 election. So far, significant drug pricing reform has been excluded from recent bills advancing through the House and Senate. The uncertainty around policy presents challenges for this sector as it seeks to optimize deployment of capital and other resources. The second half of 2021 is expected to continue the return to strong levels of merger and acquisition (M&A) activity seen in recent years. While headwinds and tailwinds exist, we believe the need to optimize portfolios and innovate will lead sector participants to deploy capital through M&A.

4. Medtech Sector:

Medtech is a \$450+ billion market with projected growth of $\sim 5.5\%$ per year (according to an October 2020 report by Statista), split across diverse segments:

- Medtech is split into many sub-segments, including: equipment, devices, consumables, in vitro diagnostics (IVD) and life sciences
- Key segments are typically driven by demographics i.e., the ageing population across the world
- Minimally invasive surgery is an important trend for equipment and device segments and prevention is driving the IVD segment

We believe some of the most important segments are - IVD, imaging, cardiology, orthopedics, dental, endoscopy, urology, drug delivery, and radiotherapy.

The North American market is by far largest market with an estimated 39% of the total market share as of 2020 (according to a May 2021 report published by Fortune Business Insights); however, we believe that it is a highly fragmented market.

According to PwC, regulatory uncertainty has recently presented concerns for the pharmaceuticals and medical devices sector in the form of more involved antitrust reviews and in March 2021, the Federal Trade Commission (FTC) announced that it and its counterpart competition enforcement agencies in the U.S. and abroad are launching a working group to update their approach to analysing the effects of pharmaceutical mergers. While we believe this could potentially slow down any potential transaction, or increase the costs of such a transaction, we do not expect recent regulatory matters to prevent us from completing a potential transaction in the Medtech space.

3. Healthcare IT

Healthcare information technology (IT) is a field of IT that involves the design, creation, development, use, and maintenance of information systems for the healthcare industry. Automatic and interoperable healthcare information systems offer a host of advantages such as low costs, minimal errors, improved medical care and public health, enhanced efficiency, and better patient satisfaction.

The U.S. Health Care Information Technology (HCIT) market was evaluated at \$61,018 million in 2017, and is estimated to reach \$149,178 million by 2025, at a CAGR of 11.7% during the forecast period.

We anticipate that the increasing digitization in the healthcare industry, demand for value-based care and rising costs of healthcare services, coupled with the increasing number of healthcare facilities will accelerate the growth of the healthcare IT market in the near term.

The U.S. healthcare IT market is driven by growing demand for improved patient safety and patient care as well as increases in government initiatives to promote HCIT. However, the U.S. HCIT is a well-established market and thus, there is a large supply and demand gap with many local as well as international key players operating in this segment, which is expected to hamper market growth. Moreover, numerous key players have collaborated with local players for developing efficient products to create opportunities in the market.

* * * * *

The following factors summarize our selection of this space and the advantages we believe we have over other similar companies:

- Broad Universe of Potential Targets. We intend to focus our investment effort broadly across the pharmaceutical industry. We believe that our investment and operating expertise across multiple industry verticals will give us a large, addressable universe of potential targets. The diversity of the target universe and the number of largely uncorrelated subsectors maximizes that likelihood that our management team will be able to identify and execute an attractive transaction.
- **Strong Access to Transformative Technology.** Our team possesses the analytical and technological capabilities to sift through a multitude of emerging private companies across disease state and drug delivery.
- Leverage our Extensive Network of Relationships to Create a Unique Pipeline of Acquisition Opportunities. We believe the combination of Mr. Ajjarapu's and our Board members' broad investment and operating experience in addition to our ability to access a deep network of public and private enterprises, experienced operators, restructuring advisors, attorneys, accountants, family offices, hedge funds, and private equity firms will enable us to identify and evaluate compelling target businesses.
- Employ a Rigorous Systematic Process of Identifying Target Companies and Acquiring a Business that will be Well-Received by the Public Markets. We believe that our directors' and officers' strong merger and acquisition (M&A) and investment track record in both private and public markets, combined with extensive public market trading experience, will provide a distinct advantage for identifying, valuing and completing a business combination that will meet our investors' expectations.

- **Providing an Alternative Path to Becoming Public.** We believe our structure will make us an attractive business combination partner to prospective target businesses that desire to become a publicly listed company. A merger with us will offer a target business an alternative process to a public listing rather than the traditional initial public offering process. We believe that target businesses may favor this alternative, which we believe is less expensive, while offering greater certainty of execution than the traditional initial public offering. Furthermore, once a proposed business combination is approved by our stockholders and the transaction is consummated, the target business will have effectively become public, whereas an initial public offering is always subject to the underwriters' ability to complete the offering, as well as general market conditions that could prevent the offering from occurring. Once public, we believe the target business would have greater access to capital and additional means of creating management incentives that are better aligned with stockholders' interests than it would as a private company. A public company can offer further benefits by augmenting a company's profile among potential new customers and vendors and aid in attracting talented management. With public company corporate governance standards, a target business may become attractive to public investors.
- Strong and Stable Financial Position with Flexibility. With funds in the trust account of \$102.0 million (or approximately \$117.3 million if the over-allotment option is exercised in full, including the deferred underwriting fees) available to use for a business combination assuming the successful completion of this offering, we believe that we offer a target business a variety of options such as providing the owners of a target business with shares in a public company and a public means to sell such shares, providing capital for the potential growth and expansion of its operations or strengthening its balance sheet by reducing its debt ratio. Because we will be able to consummate our initial business combination using our cash, debt or equity securities, or a combination of the foregoing, we will have the flexibility to use the most efficient combination that will allow us to tailor the consideration to be paid to the target business to fit its needs and desires. However, since we have no specific business combination under consideration, we have not taken any steps to secure third party financing and there can be no assurance that it will be available to us.

While we may pursue an initial business combination with any business in any industry, commercial sector or location, our initial focus will be on identifying acquisition opportunities in the pharmaceutical and medical devices sectors in the US. Trends supporting the secular growth in the healthcare industry include population aging, rise in chronic diseases and a growing consumer role in care options.

• Life sciences including pharma and biopharma:

Trends that fuel growth in these sectors include a rise in chronic diseases, regulatory changes, shift towards preventative care and demand for medicine in emerging economies. Further transforming these sectors are artificial intelligence (AI) and machine learning integration, accelerating the pace of drug discovery and development in a cost-effective manner.

Diagnostic and therapeutic devices, and related medical technologies:

The need for accurate diagnostic testing and monitoring continues to drive growth within innovative medical devices. Additionally, consumers are demanding a superior experience, as smart devices and internet connected devices find new means to solve patient problems. New healthcare technologies are reshaping the diagnosis, care, and management of illness and chronic disease.

It is important to note that the COVID-19 pandemic has had an impact on each of our sectors of interest, leading to an acceleration of several existing trends as well as a creation of new opportunities.

Acquisition Criteria

Consistent with our strategy, we have identified the following general criteria and guidelines which we believe are important in evaluating prospective target businesses. We will use these criteria and guidelines in evaluating acquisition opportunities and will initially target businesses with enterprise values of approximately \$250.0 million to \$1.0 billion, but we may decide to enter into our initial business combination with a target business that does not meet these criteria and guidelines. We currently intend to acquire one or more businesses that we believe has the following characteristics:

- Benefits from a Public Currency and Access to Public Equity Markets. Access to the public equity markets could allow the target company to utilize additional forms of capital, enhancing its ability to pursue accretive acquisitions, high-return capital projects, and/or strengthen its balance sheet and recruit and retain key employees through the use of publicly-traded equity compensation.
- *Has a Strong Competitive Position and Growing Platform.* We will seek to invest in companies that we believe possess not only established business models and sustainable competitive advantages, but also a growing platform for equity investors.
- Operated by a Talented and Incentivized Management Team. We will focus on companies with strong and experienced management teams that desire a significant equity stake in the post-business combination company. We will seek to partner with a management team and/or seller who is well-incentivized and aligned in an effort to create stockholder value.
- Benefits from Our Ability to Uniquely Structure Transaction to Unlock and Maximize Value. We will look for situations where our extensive experience and creativity can architect a win-win solution for both sides of the transaction.

These criteria are not intended to be exhaustive. Any evaluation relating to the merits of a particular initial business combination may be based, to the extent relevant, on these general guidelines as well as other considerations, factors and criteria that from time to time our management may deem relevant. In the event that we decide to enter into our initial business combination with a target business that does not meet the above criteria and guidelines, we will disclose that the target business does not meet the above criteria in our stockholder communications related to our initial business combination, which, as discussed in this prospectus, would be in the form of tender offer documents or proxy solicitation materials that we would file with the SEC

Initial Business Combination

Nasdaq rules require that we must complete one or more business combinations having an aggregate fair market value of at least 80% of the value of the assets held in the trust account (excluding the deferred underwriting commissions and taxes payable on the interest earned on the trust account) at the time of our signing a definitive agreement in connection with our initial business combination. Our board of directors will make the determination as to the fair market value of our initial business combination. If our board of directors is not able to independently determine the fair market value of our initial business combination, we will obtain an opinion from an independent investment banking firm or another independent entity that commonly renders valuation opinions with respect to the satisfaction of such criteria. While we consider it unlikely that our board of directors will not be able to make an independent determination of the fair market value of our initial business combination, it may be unable to do so if it is less familiar or experienced with the business of a particular target or if there is a significant amount of uncertainty as to the value of a target's assets or prospects. Additionally, pursuant to Nasdaq rules, any initial business combination must be approved by a majority of our independent directors.

We anticipate structuring our initial business combination so that the post-transaction company in which our public stockholders own shares will own or acquire 100% of the equity interests or assets of the target business or businesses. We may, however, structure our initial business combination such that the post-transaction company owns or acquires less than 100% of such interests or assets of the target business in order to meet certain objectives of the prior owners of the target business, the target management team or stockholders or for other reasons, but we will only complete such business combination if the post-transaction company owns or acquires 50% or more of the outstanding voting securities of the target or otherwise acquires a controlling interest in the target sufficient for it not to be required to register as an investment company under the Investment Company Act. Even if the post-transaction company owns or acquires 50% or more of the voting securities of the target, our stockholders prior to the business combination may collectively own a minority interest in the post-transaction company, depending on valuations ascribed to the target and us in the business combination transaction. For example, we could pursue a transaction in which we issue a substantial number of new shares in exchange for all of the outstanding capital stock, shares or other equity interests of a target. In this case, we would acquire a 100% controlling interest in the target. However, as a result of the issuance of a substantial number of new shares, our stockholders immediately prior to our initial business combination could own less than a majority of our issued and outstanding shares subsequent to our initial business combination. If less than 100% of the equity interests or assets of a target business or businesses are owned or acquired by the post-transaction company, the portion of such business or businesses that is owned or acquired is what will be valued for purposes of the 80% of net assets test. If the business combination involves more than one target business, the 80% of net assets test will be based on the aggregate value of all of the target businesses and we will treat the target businesses together as our initial business combination for purposes of a tender offer or for seeking stockholder approval, as applicable.

To the extent we effect our initial business combination with a company or business that may be financially unstable or in its early stages of development or growth, we may be affected by numerous risks inherent in such company or business. Although our management will endeavor to evaluate the risks inherent in a particular target business, we cannot assure you that we will properly ascertain or assess all significant risk factors.

In evaluating a prospective target business, we expect to conduct a thorough due diligence review which will encompass, among other things, meetings with incumbent management and employees, document reviews, inspection of facilities, as well as a review of financial, operational, legal and other information which will be made available to us.

The time required to select and evaluate a target business and to structure and complete our initial business combination, and the costs associated with this process, are not currently ascertainable with any degree of certainty. Any costs incurred with respect to the identification and evaluation of a prospective target business with which our initial business combination is not ultimately completed will result in our incurring losses and will reduce the funds we can use to complete another business combination.

Sourcing of Potential Initial Business Combination Targets

Certain members of our management team have spent significant portions of their careers working with businesses in the pharmaceutical and medical devices sectors and have developed a wide network of professional services contacts and business relationships in those sectors. The members of our board of directors also have significant executive management and public company experience with pharmaceutical related companies and bring additional relationships that further broaden our industry network.

This network has provided our management team with a flow of potential referrals. We believe that the network of contacts and relationships of our management team will provide us with an important source of acquisition opportunities. In addition, we anticipate that target business candidates will be brought to our attention from various unaffiliated sources, including investment market participants, private equity groups, investment banks, consultants, accounting firms, law firms and large business enterprises.

Members of our management team and our independent directors will directly or indirectly own founder shares and/or placement warrants following this offering and, accordingly, may have a conflict of interest in determining whether a particular target business is an appropriate business with which to effectuate our initial business combination. In particular, Additionally, Mr. Suren Ajjarapu, our Chief Executive Officer and Chairman currently a director of Oceantech Acquisition I Corp., traded on Nasdaq under the symbol "OTECU", a special purpose acquisition company with a class of securities registered under the Exchange Act that focuses its search for a business combination in the leisure marine, yachting and superyachting industries. As of the date of this prospectus, Oceantech Acquisition I Corp. has not entered into a business combination. Further, each of our officers and directors may have a conflict of interest with respect to evaluating a particular business combination if the retention or resignation of any such officers and directors was included by a target business as a condition to any agreement with respect to our initial business combination.

In addition, each of our officers and directors presently has, and any of them in the future may have additional, fiduciary or contractual obligations to other entities pursuant to which such officer or director is or will be required to present a business combination opportunity to such entity. We do not believe, however, that the fiduciary duties or contractual obligations of our officers or directors will materially affect our ability to complete our initial business combination.

In addition, our sponsor and our officers and directors may sponsor or form other special purpose acquisition companies similar to ours or may pursue other business or investment ventures during the period in which we are seeking an initial business combination, and some of such persons are already associated with current special purpose acquisition companies looking for acquisition targets. In the event we engage a financial advisor in the future, such financial advisor may be a sponsor of other special purposes acquisition companies. Any such companies, businesses or investments may present additional conflicts of interest in pursuing an initial business combination. However, we do not believe that any such potential conflicts would materially affect our ability to complete our initial business combination.

Status as a Public Company

We believe our structure will make us an attractive business combination partner to target businesses. As a public company, we offer a target business an alternative to the traditional initial public offering through a merger or other business combination with us. Following an initial business combination, we believe the target business would have greater access to capital and additional means of creating management incentives that are better aligned with stockholders' interests than it would as a private company. A target business can further benefit by augmenting its profile among potential new customers and vendors and aid in attracting talented employees. In a business combination transaction with us, the owners of the target business may, for example, exchange their shares of stock in the target business for our shares of Class A common stock (or shares of a new holding company) or for a combination of our shares of Class A common stock and cash, allowing us to tailor the consideration to the specific needs of the sellers.

Although there are various costs and obligations associated with being a public company, we believe target businesses will find this method a more expeditious and cost-effective method to becoming a public company than the typical initial public offering. The typical initial public offering process takes a significantly longer period of time than the typical business combination transaction process, and there are significant expenses in the initial public offering process, including underwriting discounts and commissions, marketing and road show efforts that may not be present to the same extent in connection with an initial business combination with us.

Furthermore, once a proposed initial business combination is completed, the target business will have effectively become public, whereas an initial public offering is always subject to the underwriters' ability to complete the offering, as well as general market conditions, which could delay or prevent the offering from occurring or could have negative valuation consequences. Following an initial business combination, we believe the target business would then have greater access to capital and an additional means of providing management incentives consistent with stockholders' interests and the ability to use its shares as currency for acquisitions. Being a public company can offer further benefits by augmenting a company's profile among potential new customers and vendors and aid in attracting talented employees.

While we believe that our structure and our management team's backgrounds will make us an attractive business partner, some potential target businesses may view our status as a blank check company, such as our lack of an operating history and our ability to seek stockholder approval of any proposed initial business combination, negatively.

We are an "emerging growth company," as defined in Section 2(a) of the Securities Act, as modified by the JOBS Act. As such, we are eligible to take advantage of certain exemptions from various reporting requirements that are applicable to other public companies that are not "emerging growth companies" including, but not limited to, not being required to comply with the independent registered public accounting firm attestation requirements of Section 404 of the Sarbanes-Oxley Act, reduced disclosure obligations regarding executive compensation in our periodic reports and proxy statements, and exemptions from the requirements of holding a non-binding advisory vote on executive compensation and stockholder approval of any golden parachute payments not previously approved. If some investors find our securities less attractive as a result, there may be a less active trading market for our securities and the prices of our securities may be more volatile.

In addition, Section 107 of the JOBS Act also provides that an "emerging growth company" can take advantage of the extended transition period provided in Section 7(a)(2)(B) of the Securities Act for complying with new or revised accounting standards. In other words, an "emerging growth company" can delay the adoption of certain accounting standards until those standards would otherwise apply to private companies. We intend to take advantage of the benefits of this extended transition period.

We will remain an emerging growth company until the earlier of (1) the last day of the fiscal year (a) following the fifth anniversary of the completion of this offering, (b) in which we have total annual gross revenue of at least \$1.07 billion, or (c) in which we are deemed to be a large accelerated filer, which means the market value of our Class A common stock that is held by non-affiliates exceeds \$700 million as of the prior June 30th, and (2) the date on which we have issued more than \$1.0 billion in non-convertible debt securities during the prior three-year period.

Additionally, we are a "smaller reporting company" as defined in Rule 10(f)(1) of Regulation S-K. Smaller reporting companies may take advantage of certain reduced disclosure obligations, including, among other things, providing only two years of audited financial statements. We will remain a smaller reporting company until the last day of the fiscal year in which (1) the market value of our common stock held by non-affiliates equals or exceeds \$250 million as of the end of the prior June 30th, or (2) our annual revenues equaled or exceeded \$100 million during such completed fiscal year and the market value of our common stock held by non-affiliates exceeds \$700 million as of the prior June 30th.

Financial Position

With funds available for an initial business combination initially in the amount of \$99,000,000, after payment of \$3,000,000 of deferred underwriting fees (or \$113,850,000 after payment of \$3,450,000 of deferred underwriting fees if the underwriters' over-allotment option is exercised in full), in each case before fees and expenses associated with our initial business combination (other than deferred underwriting fees), we offer a target business a variety of options such as creating a liquidity event for its owners, providing capital for the potential growth and expansion of its operations or strengthening its balance sheet by reducing its debt or leverage ratio. Because we are able to complete our initial business combination using our cash, debt or equity securities, or a combination of the foregoing, we have the flexibility to use the most efficient combination that will allow us to tailor the consideration to be paid to the target business to fit its needs and desires. However, we have not taken any steps to secure third party financing and there can be no assurance it will be available to us.

Effecting Our Initial Business Combination

We are not presently engaged in, and we will not engage in, any operations for an indefinite period of time following this offering. We intend to effectuate our initial business combination using cash from the proceeds of this offering and the sale of the placement warrants, the proceeds of the sale of our shares in connection with our initial business combination (pursuant to backstop agreements we may enter into following the consummation of this offering or otherwise), shares issued to the owners of the target, debt issued to bank or other lenders or the owners of the target, or a combination of the foregoing. We may seek to complete our initial business combination with a company or business that may be financially unstable or in its early stages of development or growth, which would subject us to the numerous risks inherent in such companies and businesses.

If our initial business combination is paid for using equity or debt securities, or not all of the funds released from the trust account are used for payment of the consideration in connection with our initial business combination or used for redemptions of our Class A common stock, we may apply the balance of the cash released to us from the trust account for general corporate purposes, including for maintenance or expansion of operations of the post-transaction company, the payment of principal or interest due on indebtedness incurred in completing our initial business combination, to fund the purchase of other companies or for working capital.

We may seek to raise additional funds through a private offering of debt or equity securities in connection with the completion of our initial business combination, and we may effectuate our initial business combination using the proceeds of such offering rather than using the amounts held in the trust account. In addition, we intend to target businesses larger than we could acquire with the net proceeds of this offering and the sale of the placement warrants, and may as a result be required to seek additional financing to complete such proposed initial business combination. Subject to compliance with applicable securities laws, we would expect to complete such financing only simultaneously with the completion of our initial business combination. In the case of an initial business combination funded with assets other than the trust account assets, our proxy materials or tender offer documents disclosing the initial business combination would disclose the terms of the financing and, only if required by applicable law or stock exchange requirements, we would seek stockholder approval of such financing. There are no prohibitions on our ability to raise funds privately, or through loans in connection with our initial business combination. At this time, we are not a party to any arrangement or understanding with any third party with respect to raising any additional funds through the sale of securities or otherwise.

We have not, nor has anyone on our behalf, initiated any substantive discussions, directly or indirectly, with any business combination target. From the period commencing with our formation through the date of this prospectus, there have been no communications or discussions between any of our officers, directors or our sponsor and any of their potential contacts or relationships regarding a potential initial business combination. Additionally, we have not engaged or retained any agent or other representative to identify or locate any suitable acquisition candidate, to conduct any research or take any measures, directly or indirectly, to locate or contact a target business. However, we may contact such targets subsequent to the closing of this offering if we become aware that such targets are interested in a potential initial business combination with us and such transaction would be attractive to our stockholders. Accordingly, there is no current basis for investors in this offering to evaluate the possible merits or risks of the target business with which we may ultimately complete our initial business combination. Although our management will assess the risks inherent in a particular target business with which we may combine, we cannot assure you that this assessment will result in our identifying all risks that a target business may encounter. Furthermore, some of those risks may be outside of our control, meaning that we can do nothing to control or reduce the chances that those risks will adversely impact a target business.

Sources of Target Businesses

We anticipate that target business candidates will be brought to our attention from various unaffiliated sources, including investment bankers and investment professionals, as a result of being solicited by us by calls or mailings. These sources may also introduce us to target businesses in which they think we may be interested on an unsolicited basis, since many of these sources will have read this prospectus and know what types of businesses we are targeting. Our officers and directors, as well as our sponsor and their affiliates, may also bring to our attention target business candidates that they become aware of through their business contacts as a result of formal or informal inquiries or discussions they may have, as well as attending trade shows or conventions. In addition, we expect to receive a number of deal flow opportunities that would not otherwise necessarily be available to us as a result of the business relationships of our officers and directors and our sponsor and their affiliates. While we do not presently anticipate engaging the services of professional firms or other individuals that specialize in business combinations on any formal basis, we may engage these firms or other individuals in the future, in which event we may pay a finder's fee, consulting fee, advisory fee or other compensation to be determined in an arm's length negotiation based on the terms of the transaction. We will engage a finder only to the extent our management determines that the use of a finder may bring opportunities to us that may not otherwise be available to us or if finders approach us on an unsolicited basis with a potential transaction that our management determines is in our best interest to pursue. Payment of finder's fees is customarily tied to completion of a transaction, in which case any such fee will be paid out of the funds held in the trust account. In no event, however, will our sponsor or any of our existing officers or directors be paid any finder's fee, reimbursement, consulting fee, monies in respect of any payment of a loan or other compensation by the company prior to, or in connection with any services rendered for any services they render in order to effectuate, the completion of our initial business combination (regardless of the type of transaction that it is). None of our sponsor, executive officers or directors, or any of their respective affiliates, will be allowed to receive any compensation, finder's fees or consulting fees from a prospective business combination target in connection with a contemplated initial business combination except as set forth herein. We have agreed to pay Aesther Healthcare Sponsor LLC, our sponsor, a total of \$10,000 per month for office space, utilities and secretarial and administrative support and to reimburse our sponsor for any out-of-pocket expenses related to identifying, investigating, and completing an initial business combination. Some of our officers and directors may enter into employment or consulting agreements with the post-transaction company following our initial business combination. The presence or absence of any such fees or arrangements will not be used as a criterion in our selection process of an initial business combination candidate.

We are not prohibited from pursuing an initial business combination with an initial business combination target that is affiliated with our sponsor, officers or directors or making the initial business combination through a joint venture or other form of shared ownership with our sponsor, officers or directors. In the event we seek to complete our initial business combination with an initial business combination target that is affiliated with our sponsor, officers or directors, we, or a committee of independent directors, would obtain an opinion from an independent investment banking firm or another independent entity that commonly renders valuation opinions that such an initial business combination is fair to our company from a financial point of view. We are not required to obtain such an opinion in any other context.

As more fully discussed in the section of this prospectus entitled "Management — Conflicts of Interest," if any of our officers or directors becomes aware of an initial business combination opportunity that falls within the line of business of any entity to which he or she has pre-existing fiduciary or contractual obligations, he or she may be required to present such business combination opportunity to such entity prior to presenting such business combination opportunity to us (including OceanTech Acquisitions I Corp.). Our officers and directors currently have certain relevant fiduciary duties or contractual obligations that may take priority over their duties to us.

Selection of a Target Business and Structuring of our Initial Business Combination

Nasdaq rules require that we must complete one or more business combinations having an aggregate fair market value of at least 80% of the value of the assets held in the trust account (excluding the deferred underwriting commissions and taxes payable on the interest earned on the trust account) at the time of our signing a definitive agreement in connection with our initial business combination. The fair market value of our initial business combination will be determined by our board of directors based upon one or more standards generally accepted by the financial community, such as discounted cash flow valuation, a valuation based on trading multiples of comparable public businesses or a valuation based on the financial metrics of M&A transactions of comparable businesses. If our board of directors is not able to independently determine the fair market value of our initial business combination, we will obtain an opinion from an independent investment banking firm or another independent entity that commonly renders valuation opinions with respect to the satisfaction of such criteria. While we consider it unlikely that our board of directors will not be able to make an independent determination of the fair market value of our initial business combination, it may be unable to do so if it is less familiar or experienced with the business of a particular target or if there is a significant amount of uncertainty as to the value of a target's assets or prospects. We do not intend to purchase multiple businesses in unrelated industries in conjunction with our initial business combination. Subject to this requirement, our management will virtually have unrestricted flexibility in identifying and selecting one or more prospective target businesses, although we will not be permitted to effectuate our initial business combination with another blank check company or a similar company with nominal operations.

In any case, we will only complete an initial business combination in which we own or acquire 50% or more of the outstanding voting securities of the target or otherwise acquire a controlling interest in the target sufficient for it not to be required to register as an investment company under the Investment Company Act. If we own or acquire less than 100% of the equity interests or assets of a target business or businesses, the portion of such business or businesses that are owned or acquired by the post-transaction company is what will be taken into account for purposes of Nasdaq's 80% fair market value test. There is no basis for investors in this offering to evaluate the possible merits or risks of any target business with which we may ultimately complete our initial business combination.

To the extent we effect our initial business combination with a company or business that may be financially unstable or in its early stages of development or growth we may be affected by numerous risks inherent in such company or business. Although our management will endeavor to evaluate the risks inherent in a particular target business, we cannot assure you that we will properly ascertain or assess all significant risk factors.

In evaluating a prospective business target, we expect to conduct a thorough due diligence review, which may encompass, among other things, meetings with incumbent management and employees, document reviews, interviews of customers and suppliers, inspection of facilities, as well as a review of financial and other information that will be made available to us.

The time required to select and evaluate a target business and to structure and complete our initial business combination, and the costs associated with this process, are not currently ascertainable with any degree of certainty. Any costs incurred with respect to the identification and evaluation of a prospective target business with which our initial business combination is not ultimately completed will result in our incurring losses and will reduce the funds we can use to complete another business combination.

Lack of Business Diversification

For an indefinite period of time after the completion of our initial business combination, the prospects for our success may depend entirely on the future performance of a single business. Unlike other entities that have the resources to complete business combinations with multiple entities in one or several industries, it is probable that we will not have the resources to diversify our operations and mitigate the risks of being in a single line of business. In addition, we intend to focus our search for an initial business combination in a single industry. By completing our initial business combination with only a single entity, our lack of diversification may:

- subject us to negative economic, competitive and regulatory developments, any or all of which may have a substantial adverse impact on the
 particular industry in which we operate after our initial business combination, and
- cause us to depend on the marketing and sale of a single product or limited number of products or services.

Limited Ability to Evaluate the Target's Management Team

Although we intend to closely scrutinize the management of a prospective target business when evaluating the desirability of effecting our initial business combination with that business, our assessment of the target business's management may not prove to be correct. In addition, the future management may not have the necessary skills, qualifications or abilities to manage a public company. Furthermore, the future role of members of our management team, if any, in the target business cannot presently be stated with any certainty. The determination as to whether any of the members of our management team will remain with the combined company will be made at the time of our initial business combination. While it is possible that one or more of our directors will remain associated in some capacity with us following our initial business combination, it is unlikely that any of them will devote their full efforts to our affairs subsequent to our initial business combination. Moreover, we cannot assure you that members of our management team will have significant experience or knowledge relating to the operations of the particular target business.

We cannot assure you that any of our key personnel will remain in senior management or advisory positions with the combined company. The determination as to whether any of our key personnel will remain with the combined company will be made at the time of our initial business combination.

Following an initial business combination, we may seek to recruit additional managers to supplement the incumbent management of the target business. We cannot assure you that we will have the ability to recruit additional managers, or that additional managers will have the requisite skills, knowledge or experience necessary to enhance the incumbent management.

Stockholders May Not Have the Ability to Approve Our Initial Business Combination

We may conduct redemptions without a stockholder vote pursuant to the tender offer rules of the SEC. However, we will seek stockholder approval if it is required by applicable law or applicable stock exchange listing requirements, or we may decide to seek stockholder approval for business or other legal reasons. Presented in the table below is a graphic explanation of the types of initial business combinations we may consider and whether stockholder approval is currently required under Delaware law for each such transaction.

	Whether
	Stockholder
	Approval is
Type of Transaction	Required
Purchase of assets	No
Purchase of stock of target not involving a merger with the company	No
Merger of target into a subsidiary of the company	No
Merger of the company with a target	Yes

Under Nasdaq's listing rules, stockholder approval would be required for our initial business combination if, for example:

- we issue shares of Class A common stock that will be equal to or in excess of 20% of the number of shares of our Class A common stock then outstanding (other than in a public offering);
- any of our directors, officers or substantial stockholders (as defined by Nasdaq rules) has a 5% or greater interest (or such persons collectively have a 10% or greater interest), directly or indirectly, in the target business or assets to be acquired or otherwise and the present or potential issuance of common stock could result in an increase in outstanding common stock or voting power of 5% or more; or
- the issuance or potential issuance of common stock will result in our undergoing a change of control.

The decision as to whether we will seek stockholder approval of a proposed business combination in those instances in which stockholder approval is not required by law will be made by us, solely in our discretion, and will be based on business and reasons, which include a variety of factors, including, but not limited to:

- the timing of the transaction, including in the event we determine stockholder approval would require additional time and there is either not enough time to seek stockholder approval or doing so would place the company at a disadvantage in the transaction or result in other additional burdens on the company;
- the expected cost of holding a stockholder vote;
- the risk that the stockholders would fail to approve the proposed business combination;
- other time and budget constraints of the company; and
- additional legal complexities of a proposed business combination that would be time-consuming and burdensome to present to stockholders.

Ability to Extend Time to Complete Business Combination

We will have until 12 months from the closing of this offering to consummate our initial business combination. However, if we anticipate that we may not be able to consummate our initial business combination within 12 months, we may, by resolution of our board if requested by our sponsor, extend the period of time to consummate a business combination up to two times, each by an additional three months (for a total of up to 18 months to complete a business combination), subject to the sponsor depositing additional funds into the trust account as set out below. Pursuant to the terms of the trust agreement to be entered into between us and Continental Stock Transfer & Trust Company, LLC on the date of this prospectus, in order to extend the time available for us to consummate our initial business combination, our initial shareholders or their affiliates or designees, upon five days advance notice prior to the applicable deadline, must deposit into the trust account for each three-month extension, \$1,000,000, or \$1,150,000 if the underwriters' over-allotment option is exercised in full (\$0.10 per share in either case) on or prior to the date of the applicable deadline, up to an aggregate of \$2,000,000 (or \$2,300,000 if the underwriters' over-allotment option is exercised in full), or approximately \$0.20 per share. Any such payments would be made in the form of a loan. Any such loans will be non-interest bearing and payable upon the consummation of our initial business combination. If we complete our initial business combination, we would repay such loaned amounts out of the proceeds of the trust account released to us. If we do not complete a business combination, we will not repay such loans. Furthermore, the letter agreement with our initial stockholders contains a provision pursuant to which our sponsor has agreed to waive its right to be repaid for such loans out of the funds held in the trust account in the event that we do not complete a business combination. In the event that we receive notice from our sponsor five days prior to the applicable deadline of its wish for us to effect an extension, we intend to issue a press release announcing such intention at least three days prior to the applicable deadline. In addition, we intend to issue a press release the day after the applicable deadline announcing whether or not the funds had been timely deposited. Our sponsor and its affiliates or designees are not obligated to fund the trust account to extend the time for us to complete our initial business combination.

Permitted Purchases of our Securities

If we seek stockholder approval of our initial business combination and we do not conduct redemptions in connection with our initial business combination pursuant to the tender offer rules, our sponsor, initial stockholders, directors, officers, advisors or their affiliates may purchase public shares or public warrants in privately negotiated transactions or in the open market either prior to or following the completion of our initial business combination. There is no limit on the number of shares our initial stockholders, directors, officers or their affiliates may purchase in such transactions, subject to compliance with applicable law and Nasdaq rules. However, they have no current commitments, plans or intentions to engage in such transactions and have not formulated any terms or conditions for any such transactions. If they engage in such transactions, they will not make any such purchases when they are in possession of any material nonpublic information not disclosed to the seller or if such purchases are prohibited by Regulation M under the Exchange Act. We do not currently anticipate that such purchases, if any, would constitute a tender offer subject to the tender offer rules under the Exchange Act or a going-private transaction subject to the going-private rules under the Exchange Act; however, if the purchasers determine at the time of any such purchases that the purchases are subject to such rules, the purchasers will comply with such rules. Any such purchases will be reported pursuant to Section 13 and Section 16 of the Exchange Act to the extent such purchasers are subject to such reporting requirements. None of the funds held in the trust account will be used to purchase shares or public warrants in such transactions prior to completion of our initial business combination.

The purpose of any such purchases of shares could be to vote such shares in favor of the initial business combination and thereby increase the likelihood of obtaining stockholder approval of the initial business combination or to satisfy a closing condition in an agreement with a target that requires us to have a minimum net worth or a certain amount of cash at the closing of our initial business combination, where it appears that such requirement would otherwise not be met. The purpose of any such purchases of public warrants could be to reduce the number of public warrants outstanding or to vote such warrants on any matters submitted to the warrant holders for approval in connection with our initial business combination. Any such purchases of our securities may result in the completion of our initial business combination that may not otherwise have been possible. In addition, if such purchases are made, the public "float" of our shares of Class A common stock or warrants may be reduced and the number of beneficial holders of our securities may be reduced, which may make it difficult to maintain or obtain the quotation, listing or trading of our securities on a national securities exchange.

Our sponsor, officers, directors and/or their affiliates anticipate that they may identify the stockholders with whom our sponsor, officers, directors or their affiliates may pursue privately negotiated purchases by either the stockholders contacting us directly or by our receipt of redemption requests submitted by stockholders following our mailing of proxy materials in connection with our initial business combination. To the extent that our sponsor, officers, directors or their affiliates enter into a private purchase, they would identify and contact only potential selling stockholders who have expressed their election to redeem their shares for a pro rata share of the trust account or vote against our initial business combination, whether or not such stockholder has already submitted a proxy with respect to our initial business combination. Our sponsor, officers, directors or their affiliates will only purchase public shares if such purchases comply with Regulation M under the Exchange Act and the other federal securities laws.

Any purchases by our sponsor, officers, directors and/or their affiliates who are affiliated purchasers under Rule 10b-18 under the Exchange Act will only be made to the extent such purchases are able to be made in compliance with Rule 10b-18, which is a safe harbor from liability for manipulation under Section 9(a)(2) and Rule 10b-5 of the Exchange Act. Rule 10b-18 has certain technical requirements that must be complied with in order for the safe harbor to be available to the purchaser. Our sponsor, officers, directors and/or their affiliates will not make purchases of common stock if the purchases would violate Section 9(a)(2) or Rule 10b-5 of the Exchange Act. We expect that any such purchases will be reported pursuant to Section 13 and Section 16 of the Exchange Act to the extent such purchases are subject to such reporting requirements.

Redemption Rights for Public Stockholders upon Completion of our Initial Business Combination

We will provide our public stockholders with the opportunity to redeem all or a portion of their shares of Class A common stock upon the completion of our initial business combination at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the trust account as of two business days prior to the consummation of the initial business combination including interest earned on the funds held in the trust account and not previously released to us to pay our taxes, divided by the number of then outstanding public shares, subject to the limitations described herein. The amount in the trust account is initially anticipated to be approximately \$10.20 per public share. The per-share amount we will distribute to investors who properly redeem their shares will not be reduced by the deferred underwriting commissions we will pay to the underwriters. Our sponsor, officers and directors have entered into a letter agreement with us, and we anticipate that our director nominees will agree to enter into with us, a letter agreement, pursuant to which they have agreed to waive their redemption rights with respect to any founder shares and any public shares held by them in connection with the completion of our initial business combination.

Manner of Conducting Redemptions

We will provide our public stockholders with the opportunity to redeem all or a portion of their public shares of Class A common stock upon the completion of our initial business combination either (i) in connection with a stockholder meeting called to approve the initial business combination or (ii) by means of a tender offer. The decision as to whether we will seek stockholder approval of a proposed initial business combination or conduct a tender offer will be made by us, solely in our discretion, and will be based on a variety of factors such as the timing of the transaction and whether the terms of the transaction would require us to seek stockholder approval under the law or stock exchange listing requirement. Under Nasdaq rules, asset acquisitions and stock purchases would not typically require stockholder approval while direct mergers with our company where we do not survive and any transactions where we issue more than 20% of our outstanding common stock or seek to amend our amended and restated certificate of incorporation would require stockholder approval. If we structure an initial business combination with a target company in a manner that requires stockholder approval, we will not have discretion as to whether to seek a stockholder vote to approve the proposed initial business combination. We may conduct redemptions without a stockholder vote pursuant to the tender offer rules of the SEC unless stockholder approval is required by law or stock exchange listing requirements or we choose to seek stockholder approval for business or other legal reasons. So long as we obtain and maintain a listing for our securities on Nasdaq, we will be required to comply with such rules.

If stockholder approval of the transaction is required by law or stock exchange listing requirement, or we decide to obtain stockholder approval for business or other legal reasons, we will, pursuant to our amended and restated certificate of incorporation:

- conduct the redemptions in conjunction with a proxy solicitation pursuant to Regulation 14A of the Exchange Act, which regulates the solicitation of proxies, and not pursuant to the tender offer rules, and
- file proxy materials with the SEC.

In the event that we seek stockholder approval of our initial business combination, we will distribute proxy materials and, in connection therewith, provide our public stockholders with the redemption rights described above upon completion of the initial business combination.

If we seek stockholder approval, we will complete our initial business combination only if a majority of the outstanding shares of common stock present and entitled to vote at the meeting to approve the initial business combination when a quorum is present are voted in favor of the initial business combination. A quorum for such meeting will consist of the holders present in person or by proxy of shares of outstanding capital stock of the Company representing a majority of the voting power of all outstanding shares of capital stock of the Company entitled to vote at such meeting. Our initial stockholders will count toward this quorum and pursuant to the letter agreement, our sponsor, officers and directors have agreed to vote and our director nominees will agree to vote, any founder shares held by them and any public shares acquired during or after this offering (including in open market and privately negotiated transactions) in favor of our initial business combination. For purposes of seeking approval of the majority of our outstanding shares of common stock voted, non-votes will have no effect on the approval of our initial business combination once a quorum is obtained. As a result, in addition to our initial stockholders' founder shares, we would need 550,001 or 5.5%, of the 10,000,000 public shares sold in this offering to be voted in favor of an initial business combination (assuming only the minimum number of shares representing a quorum are voted) in order to have our initial business combination approved (assuming the over-allotment option is not exercised and that the initial stockholders do not purchase any units in this offering or units or shares in the after-market and all shares to be issued to the representative of the underwriters and/or its designees are issued and outstanding and voted in favor of the business combination). In the event that all shares of our outstanding common stock are voted, we would need 3,700,001 or 37.0%, of the 10,000,000 public shares sold in this offering to be voted in favor of an initial business combination in order to have our initial business combination approved (assuming the over-allotment option is not exercised, and that the initial stockholders do not purchase any units in this offering or units or shares in the after-market and all shares to be issued to the representative of the underwriters and/or its designees are issued and outstanding and voted in favor of the business combination). We intend to give approximately 30 days (but not less than 10 days nor more than 60 days) prior written notice of any such meeting, if required, at which a vote shall be taken to approve our initial business combination. These quorum and voting thresholds, and the voting agreements of our initial stockholders, may make it more likely that we will consummate our initial business combination. Each public stockholder may elect to redeem its public shares irrespective of whether they vote for or against the proposed transaction (or abstain from voting).

If a stockholder vote is not required and we do not decide to hold a stockholder vote for business or other legal reasons, we will, pursuant to our amended and restated certificate of incorporation:

- conduct the redemptions pursuant to Rule 13e-4 and Regulation 14E of the Exchange Act, which regulate issuer tender offers, and
- file tender offer documents with the SEC prior to completing our initial business combination which contain substantially the same financial and other information about the initial business combination and the redemption rights as is required under Regulation 14A of the Exchange Act, which regulates the solicitation of proxies.

Upon the public announcement of our initial business combination, we or our sponsor will terminate any plan established in accordance with Rule 10b5-1 to purchase shares of our Class A common stock in the open market if we elect to redeem our public shares through a tender offer, to comply with Rule 14e-5 under the Exchange Act.

In the event we conduct redemptions pursuant to the tender offer rules, our offer to redeem will remain open for at least 20 business days, in accordance with Rule 14e-1(a) under the Exchange Act, and we will not be permitted to complete our initial business combination until the expiration of the tender offer period. In addition, we will not redeem any public shares unless our net tangible assets will be at least \$5,000,001 either immediately prior to or upon consummation of our initial business combination and after payment of underwriters' fees and commissions (so that we are not subject to the SEC's "penny stock" rules) or any greater net tangible asset or cash requirement which may be contained in the agreement relating to our initial business combination. If public stockholders tender more shares than we have offered to purchase, we will withdraw the tender offer and not complete the initial business combination.

Our amended and restated certificate of incorporation will provide that we may not redeem our public shares unless our net tangible assets are at least \$5,000,001 either immediately prior to or upon consummation of our initial business combination and after payment of underwriters' fees and commissions (so that we are not subject to the SEC's "penny stock" rules) or any greater net tangible asset or cash requirement which may be contained in the agreement relating to our initial business combination. For example, the proposed initial business combination may require: (i) cash consideration to be paid to the target or its owners, (ii) cash to be transferred to the target for working capital or other general corporate purposes or (iii) the retention of cash to satisfy other conditions in accordance with the terms of the proposed initial business combination. In the event the aggregate cash consideration we would be required to pay for all shares of Class A common stock that are validly submitted for redemption plus any amount required to satisfy cash conditions pursuant to the terms of the proposed initial business combination exceed the aggregate amount of cash available to us, we will not complete the initial business combination or redeem any shares, and all shares of Class A common stock submitted for redemption will be returned to the holders thereof.

Limitation on Redemption upon Completion of our Initial Business Combination if we Seek Stockholder Approval

Notwithstanding the foregoing, if we seek stockholder approval of our initial business combination and we do not conduct redemptions in connection with our initial business combination pursuant to the tender offer rules, our amended and restated certificate of incorporation will provide that a public stockholder, together with any affiliate of such stockholder or any other person with whom such stockholder is acting in concert or as a "group" (as defined under Section 13 of the Exchange Act), will be restricted from seeking redemption rights with respect to more than an aggregate of 15% of the shares sold in this offering, which we refer to as the "Excess Shares." Such restriction shall also be applicable to our affiliates. We believe this restriction will discourage stockholders from accumulating large blocks of shares, and subsequent attempts by such holders to use their ability to exercise their redemption rights against a proposed initial business combination as a means to force us or our management to purchase their shares at a significant premium to the then-current market price or on other undesirable terms. By limiting our stockholders' ability to redeem no more than 15% of the shares sold in this offering without our prior consent, we believe we will limit the ability of a small group of stockholders to unreasonably attempt to block our ability to complete our initial business combination, particularly in connection with an initial business combination with a target that requires as a closing condition that we have a minimum net worth or a certain amount of cash. However, we would not be restricting our stockholders' ability to vote all of their shares (including Excess Shares) for or against our initial business combination.

Tendering Stock Certificates in Connection with Redemption Rights

We may require our public stockholders seeking to exercise their redemption rights, whether they are record holders or hold their shares in "street name," to either tender their certificates to our transfer agent up to two business days prior to the vote on the proposal to approve the initial business combination, or to deliver their shares to the transfer agent electronically using the Depository Trust Company's DWAC (Deposit/Withdrawal At Custodian) System, at the holder's option. The proxy materials that we will furnish to holders of our public shares in connection with our initial business combination will indicate whether we are requiring public stockholders to satisfy such delivery requirements. Accordingly, a public stockholder would have up to two days prior to the vote on the initial business combination to tender its shares if it wishes to seek to exercise its redemption rights. Given the relatively short exercise period, it is advisable for stockholders to use electronic delivery of their public shares.

There is a nominal cost associated with the above-referenced tendering process and the act of certificating the shares or delivering them through the DWAC System. The transfer agent will typically charge the tendering broker \$80.00 and it would be up to the broker whether or not to pass this cost on to the redeeming holder. However, this fee would be incurred regardless of whether or not we require holders seeking to exercise redemption rights to tender their shares. The need to deliver shares is a requirement of exercising redemption rights regardless of the timing of when such delivery must be effectuated.

The foregoing is different from the procedures used by many special purpose acquisition companies. In order to perfect redemption rights in connection with their business combinations, many blank check companies would distribute proxy materials for the stockholders' vote on an initial business combination, and a holder could simply vote against a proposed initial business combination and check a box on the proxy card indicating such holder was seeking to exercise his or her redemption rights. After the initial business combination was approved, the company would contact such stockholder to arrange for him or her to deliver his or her certificate to verify ownership. As a result, the stockholder then had an "option window" after the completion of the initial business combination during which he or she could monitor the price of the company's stock in the market. If the price rose above the redemption price, he or she could sell his or her shares in the open market before actually delivering his or her shares to the company for cancellation. As a result, the redemption rights, to which stockholders were aware they needed to commit before the stockholder meeting, would become "option" rights surviving past the completion of the initial business combination until the redeeming holder delivered its certificate. The requirement for physical or electronic delivery prior to the meeting ensures that a redeeming holder's election to redeem is irrevocable once the initial business combination is approved.

Any request to redeem such shares, once made, may be withdrawn at any time up to the date of the stockholder meeting. Furthermore, if a holder of a public share delivered its certificate in connection with an election of redemption rights and subsequently decides prior to the applicable date not to elect to exercise such rights, such holder may simply request that the transfer agent return the certificate (physically or electronically). It is anticipated that the funds to be distributed to holders of our public shares electing to redeem their shares will be distributed promptly after the completion of our initial business combination.

If our initial business combination is not approved or completed for any reason, then our public stockholders who elected to exercise their redemption rights would not be entitled to redeem their shares for the applicable pro rata share of the trust account. In such case, we will promptly return any certificates delivered by public holders who elected to redeem their shares.

If our initial proposed initial business combination is not completed, we may continue to try to complete an initial business combination with a different target until 12 months from the closing of this offering or during any Extension Period.

Redemption of Public Shares and Liquidation if no Initial Business Combination

Our amended and restated certificate of incorporation will provide that we will have only 12 months from the closing of this offering to complete our initial business combination, subject to any Extension Period. We may seek stockholder approval of the amendments to our amended and restated certificate of incorporation and the trust agreement to be entered into between us and Continental Stock Transfer & Trust Company, LLC for any extension beyond 18 months at a meeting called for such purpose. Public stockholders will be offered the opportunity to vote on or redeem their shares in connection with any such extension. If we are unable to complete our initial business combination within such 12-month period (or during any Extension Period), we will: (i) cease all operations except for the purpose of winding up, (ii) as promptly as reasonably possible but not more than ten business days thereafter, redeem the public shares, at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the trust account including interest earned on the funds held in the trust account and not previously released to us to pay our taxes (less up to \$100,000 of interest to pay dissolution expenses), divided by the number of then outstanding public shares, which redemption will completely extinguish public stockholders' rights as stockholders (including the right to receive further liquidating distributions, if any), subject to applicable law, and (iii) as promptly as reasonably possible following such redemption, subject to the approval of our remaining stockholders and our board of directors, dissolve and liquidate, subject in the case of clauses (ii) and (iii) above to our obligations under Delaware law to provide for claims of creditors and the requirements of other applicable law. There will be no redemption rights or liquidating distributions with respect to our warrants, which will expire worthless if we fail to complete our initial business combination within the 18-month period (or during any Extension

Our sponsor, officers and directors have entered into a letter agreement with us, and we anticipate that our director nominees will agree to enter into with us, a letter agreement, pursuant to which they have waived their rights to liquidating distributions from the trust account with respect to any founder shares held by them if we fail to complete our initial business combination within 12 months from the closing of this offering or during any Extension Period. However, if our sponsor, officers or directors acquire public shares in or after this offering, they will be entitled to liquidating distributions from the trust account with respect to such public shares if we fail to complete our initial business combination within 12 months from the closing of this offering or during any Extension Period.

Our sponsor, officers and directors have agreed, pursuant to a written agreement with us, that they will not propose any amendment to our amended and restated certificate of incorporation (i) to modify the substance or timing of our obligation to allow redemption in connection with our initial business combination or certain amendments to our charter prior thereto or to redeem 100% of our public shares if we do not complete our initial business combination within 12 months from the closing of this offering or during any Extension Period or (ii) with respect to any other provision relating to stockholders' rights or pre-initial business combination activity, unless we provide our public stockholders with the opportunity to redeem their shares of Class A common stock upon approval of any such amendment at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the trust account including interest earned on the funds held in the trust account and not previously released to us to pay our taxes divided by the number of then outstanding public shares. However, we may not redeem our public shares unless our net tangible assets are at least \$5,000,001 either immediately prior to or upon consummation of our initial business combination and after payment of underwriters' fees and commissions (so that we are not subject to the SEC's "penny stock" rules). If this optional redemption right is exercised with respect to an excessive number of public shares such that we cannot satisfy the net tangible asset requirement (described above), we would not proceed with the amendment or the related redemption of our public shares at such time.

We expect that all costs and expenses associated with implementing our plan of dissolution, as well as payments to any creditors, will be funded from amounts remaining out of the approximately \$1,000,000 of proceeds held outside the trust account, although we cannot assure you that there will be sufficient funds for such purpose. We will depend on sufficient interest being earned on the proceeds held in the trust account to pay any tax obligations we may owe. However, if those funds are not sufficient to cover the costs and expenses associated with implementing our plan of dissolution, to the extent that there is any interest accrued in the trust account not required to pay taxes, we may request the trustee to release to us an additional amount of up to \$100,000 of such accrued interest to pay those costs and expenses.

If we were to expend all of the net proceeds of this offering and the sale of the placement warrants, other than the proceeds deposited in the trust account, and without taking into account interest, if any, earned on the trust account, the per-share redemption amount received by stockholders upon our dissolution would be approximately \$10.20. The proceeds deposited in the trust account could, however, become subject to the claims of our creditors which would have higher priority than the claims of our public stockholders. We cannot assure you that the actual per-share redemption amount received by stockholders will not be substantially less than \$10.20. Under Section 281(b) of the DGCL, our plan of dissolution must provide for all claims against us to be paid in full or make provision for payments to be made in full, as applicable, if there are sufficient assets. These claims must be paid or provided for before we make any distribution of our remaining assets to our stockholders. While we intend to pay such amounts, if any, we cannot assure you that we will have funds sufficient to pay or provide for all creditors' claims.

Although we will seek to have all vendors, service providers, prospective target businesses or other entities with which we do business execute agreements with us waiving any right, title, interest or claim of any kind in or to any monies held in the trust account for the benefit of our public stockholders, there is no guarantee that they will execute such agreements or even if they execute such agreements that they would be prevented from bringing claims against the trust account including but not limited to fraudulent inducement, breach of fiduciary responsibility or other similar claims, as well as claims challenging the enforceability of the waiver, in each case in order to gain an advantage with respect to a claim against our assets, including the funds held in the trust account. If any third party refuses to execute an agreement waiving such claims to the monies held in the trust account, our management will perform an analysis of the alternatives available to it and will only enter into an agreement with a third party that has not executed a waiver if management believes that such third party's engagement would be significantly more beneficial to us than any alternative. Examples of possible instances where we may engage a third party that refuses to execute a waiver include the engagement of a third party consultant whose particular expertise or skills are believed by management to be significantly superior to those of other consultants that would agree to execute a waiver or in cases where management is unable to find a service provider willing to execute a waiver. Malone, our independent registered public accounting firm, and the underwriters of the offering, will not execute agreements with us waiving such claims to the monies held in the trust account.

In addition, there is no guarantee that such entities will agree to waive any claims they may have in the future as a result of, or arising out of, any negotiations, contracts or agreements with us and will not seek recourse against the trust account for any reason. Our sponsor has agreed that it will be liable to us if and to the extent any claims by a third party for services rendered or products sold to us, or a prospective target business with which we have entered into a written letter of intent, confidentiality or similar agreement or business combination agreement, reduce the amount of funds in the trust account to below the lesser of (i) \$10.20 per public share and (ii) the actual amount per public share held in the trust account as of the date of the liquidation of the trust account, if less than \$10.20 per share due to reductions in the value of the trust assets, less taxes payable, provided that such liability will not apply to any claims by a third party or prospective target business who executed a waiver of any and all rights to the monies held in the trust account (whether or not such waiver is enforceable) nor will it apply to any claims under our indemnity of the underwriters of this offering against certain liabilities, including liabilities under the Securities Act. However, we have not asked our sponsor to reserve for such indemnification obligations, nor have we independently verified whether our sponsor has sufficient funds to satisfy its indemnity obligations and believe that our sponsor's only assets are securities of our company. Therefore, we cannot assure you that our sponsor would be able to satisfy those obligations. None of our officers or directors will indemnify us for claims by third parties including, without limitation, claims by vendors and prospective target businesses.

In the event that the proceeds in the trust account are reduced below (i) \$10.20 per public share or (ii) such lesser amount per public share held in the trust account as of the date of the liquidation of the trust account, due to reductions in value of the trust assets, in each case net of the amount of interest which may be withdrawn to pay taxes, and our sponsor asserts that it is unable to satisfy its indemnification obligations or that it has no indemnification obligations related to a particular claim, our independent directors would determine whether to take legal action against our sponsor to enforce its indemnification obligations. While we currently expect that our independent directors would take legal action on our behalf against our sponsor to enforce its indemnification obligations to us, it is possible that our independent directors in exercising their business judgment may choose not to do so if, for example, the cost of such legal action is deemed by the independent directors to be too high relative to the amount recoverable or if the independent directors determine that a favorable outcome is not likely. We have not asked our sponsor to reserve for such indemnification obligations and we cannot assure you that our sponsor would be able to satisfy those obligations. Accordingly, we cannot assure you that due to claims of creditors the actual value of the per-share redemption price will not be less than \$10.20 per public share.

We will seek to reduce the possibility that our sponsor will have to indemnify the trust account due to claims of creditors by endeavoring to have all vendors, service providers, prospective target businesses or other entities with which we do business execute agreements with us waiving any right, title, interest or claim of any kind in or to monies held in the trust account. Our sponsor will also not be liable as to any claims under our indemnity of the underwriters of this offering against certain liabilities, including liabilities under the Securities Act. We will have access to up to approximately \$1,000,000 from the proceeds of this offering and the sale of the placement warrants with which to pay any such potential claims (including costs and expenses incurred in connection with our liquidation, currently estimated to be no more than approximately \$100,000). In the event that we liquidate and it is subsequently determined that the reserve for claims and liabilities is insufficient, stockholders who received funds from our trust account could be liable for claims made by creditors. In the event that our offering expenses exceed our estimate of \$1,261,000, we may fund such excess with funds from the funds not to be held in the trust account. In such case, the amount of funds we intend to be held outside the trust account would decrease by a corresponding amount. Conversely, in the event that the offering expenses are less than our estimate of \$1,261,000, the amount of funds we intend to be held outside the trust account would increase by a corresponding amount.

Under the DGCL, stockholders may be held liable for claims by third parties against a corporation to the extent of distributions received by them in a dissolution. The pro rata portion of our trust account distributed to our public stockholders upon the redemption of our public shares in the event we do not complete our initial business combination within 12 months from the closing of this offering or during any Extension Period may be considered a liquidating distribution under Delaware law. If the corporation complies with certain procedures set forth in Section 280 of the DGCL intended to ensure that it makes reasonable provision for all claims against it, including a 60-day notice period during which any third-party claims can be brought against the corporation, a 90-day period during which the corporation may reject any claims brought, and an additional 150-day waiting period before any liquidating distributions are made to stockholders, any liability of stockholders with respect to a liquidating distribution is limited to the lesser of such stockholder's pro rata share of the claim or the amount distributed to the stockholder, and any liability of the stockholder would be barred after the third anniversary of the dissolution.

Furthermore, if the pro rata portion of our trust account distributed to our public stockholders upon the redemption of our public shares in the event we do not complete our initial business combination within 12 months from the closing of this offering (or during any Extension Period), is not considered a liquidating distribution under Delaware law and such redemption distribution is deemed to be unlawful (potentially due to the imposition of legal proceedings that a party may bring or due to other circumstances that are currently unknown), then pursuant to Section 174 of the DGCL, the statute of limitations for claims of creditors could then be six years after the unlawful redemption distribution, instead of three years, as in the case of a liquidating distribution. If we are unable to complete our initial business combination within 12 months from the closing of this offering or during any Extension Period, we will: (i) cease all operations except for the purpose of winding up, (ii) as promptly as reasonably possible but not more than ten business days thereafter, redeem the public shares, at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the trust account including interest earned on the funds held in the trust account and not previously released to us to pay our taxes (less up to \$100,000 of interest to pay dissolution expenses), divided by the number of then outstanding public shares, which redemption will completely extinguish public stockholders' rights as stockholders (including the right to receive further liquidating distributions, if any), subject to applicable law, and (iii) as promptly as reasonably possible following such redemption, subject to the approval of our remaining stockholders and our board of directors, dissolve and liquidate, subject in the case of clauses (ii) and (iii) above to our obligations under Delaware law to provide for claims of creditors and the requirements of other applicable law. Accordingly, it is our intention to redeem our public shares as soon as reasonably possible following our 18th month or the end of any Extension Period, following this offering, if we have not completed a business combination prior to such date and, therefore, we do not intend to comply with those procedures. As such, our stockholders could potentially be liable for any claims to the extent of distributions received by them (but no more) and any liability of our stockholders may extend well beyond the third anniversary of such date.

Because we will not be complying with Section 280, Section 281(b) of the DGCL requires us to adopt a plan, based on facts known to us at such time that will provide for our payment of all existing and pending claims or claims that may be potentially brought against us within the subsequent 10 years. However, because we are a blank check company, rather than an operating company, and our operations will be limited to searching for prospective target businesses to acquire, the only likely claims to arise would be from our vendors (such as lawyers, investment bankers, etc.) or prospective target businesses. As described above, pursuant to the obligation contained in our underwriting agreement, we will seek to have all vendors, service providers, prospective target businesses or other entities with which we do business execute agreements with us waiving any right, title, interest or claim of any kind in or to any monies held in the trust account. As a result of this obligation, the claims that could be made against us are significantly limited and the likelihood that any claim that would result in any liability extending to the trust account is remote. Further, our sponsor may be liable only to the extent necessary to ensure that the amounts in the trust account are not reduced below (i) \$10.20 per public share or (ii) such lesser amount per public share held in the trust account as of the date of the liquidation of the trust account, due to reductions in value of the trust assets, in each case net of the amount of interest withdrawn to pay taxes and will not be liable as to any claims under our indemnity of the underwriters of this offering against certain liabilities, including liabilities under the Securities Act. In the event that an executed waiver is deemed to be unenforceable against a third party, our sponsor will not be responsible to the extent of any liability for such third-party claims.

If we file a bankruptcy petition or an involuntary bankruptcy petition is filed against us that is not dismissed, the proceeds held in the trust account could be subject to applicable bankruptcy law, and may be included in our bankruptcy estate and subject to the claims of third parties with priority over the claims of our stockholders. To the extent any bankruptcy claims deplete the trust account, we cannot assure you we will be able to return \$10.20 per share to our public stockholders. Additionally, if we file a bankruptcy petition or an involuntary bankruptcy petition is filed against us that is not dismissed, any distributions received by stockholders could be viewed under applicable debtor/creditor and/or bankruptcy laws as either a "preferential transfer" or a "fraudulent conveyance." As a result, a bankruptcy court could seek to recover some or all amounts received by our stockholders. Furthermore, our board of directors may be viewed as having breached its fiduciary duty to our creditors and/or may have acted in bad faith, thereby exposing itself and our company to claims of punitive damages, by paying public stockholders from the trust account prior to addressing the claims of creditors. We cannot assure you that claims will not be brought against us for these reasons.

Our public stockholders will be entitled to receive funds from the trust account only upon the earlier to occur of: (i) the completion of our initial business combination, (ii) the redemption of any public shares properly tendered in connection with a stockholder vote to amend any provisions of our amended and restated certificate of incorporation (A) to modify the substance or timing of our obligation to allow redemption in connection with our initial business combination or certain amendments to our charter prior thereto or to redeem 100% of our public shares if we do not complete our initial business combination within 12 months from the closing of this offering or during any Extension Period or (B) with respect to any other provision relating to stockholders' rights or pre-initial business combination activity, and (iii) the redemption of all of our public shares if we are unable to complete our business combination within 12 months from the closing of this offering or during any Extension Period, subject to applicable law. In no other circumstances will a stockholder have any right or interest of any kind to or in the trust account. In the event we seek stockholder approval in connection with our initial business combination, a stockholder's voting in connection with the initial business combination alone will not result in a stockholder's redeeming its shares to us for an applicable pro rata share of the trust account. Such stockholder must have also exercised its redemption rights as described above. These provisions of our amended and restated certificate of incorporation, have be amended with a stockholder vote.

Comparison of Redemption or Purchase Prices in Connection with Our Initial Business Combination and if We Fail to Complete Our Initial Business Combination

The following table compares the redemptions and other permitted purchases of public shares that may take place in connection with the completion of our initial business combination and if we are unable to complete our initial business combination within 12 months from the closing of this offering or during any Extension Period.

Redemptions in Connection with our Initial Business Combination

Calculation of redemption price

Redemptions at the time of our initial business combination may be made pursuant to a tender offer or in connection with a stockholder vote. The redemption price will be the same whether we conduct redemptions pursuant to a tender offer or in connection with a stockholder vote. In either case, our public stockholders may redeem their public shares for cash equal to the aggregate amount then on deposit in the trust account as of two business days prior to consummation of the initial business combination (which is initially anticipated to be \$10.20 per public share), including interest earned on the funds held in the trust account and not previously released to us to pay our taxes divided by the number of then outstanding public shares, subject to the limitation that no redemptions will take place if all of the redemptions would cause our net tangible assets to be less than \$5,000,001 as described elsewhere in this prospectus and any limitations (including but not limited to cash requirements) agreed to in connection with the negotiation of terms of a proposed initial business combination.

Other Permitted Purchases of Public Shares by us or our Affiliates

If we seek stockholder approval of our initial business combination, our sponsor, directors, officers or their affiliates may purchase shares in privately negotiated transactions or in the open market prior to or following completion of our initial business combination. There is no limit to the prices that our sponsor, directors, officers or their affiliates may pay in these transactions.

Redemptions if we fail to Complete an Initial Business Combination

If we are unable to complete our initial business combination within 12 months from the closing of this offering, or any Extension Period, as extended by the Company's stockholders in accordance with our amended and restated certificate of incorporation, we will redeem all public shares at a per-share price, payable in cash, equal to the aggregate amount, then on deposit in the trust account (which is initially anticipated to be \$10.20 per public share) including interest earned on the funds held in the trust account and not previously released to us to pay our taxes (less up to \$100,000 of interest to pay dissolution expenses), divided by the number of then outstanding public shares.

Impact to remaining stockholders

The redemptions in connection with our initial business combination will reduce the book value per share for our remaining stockholders, who will bear the burden of the deferred underwriting commissions and taxes payable.

If the permitted purchases described above are made there would be no impact to our remaining stockholders because the purchase price would not be paid by us.

The redemption of our public shares if we fail to complete our initial business combination will reduce the book value per share for the shares held by our initial stockholders, who will be our only remaining stockholders after such redemptions.

Comparison of This Offering to Those of Blank Check Companies Subject to Rule 419

The following table compares the terms of this offering to the terms of an offering by a blank check company subject to the provisions of Rule 419. This comparison assumes that the gross proceeds, underwriting commissions and underwriting expenses of our offering would be identical to those of an offering undertaken by a company subject to Rule 419, and that the underwriters will not exercise their over-allotment option. None of the provisions of Rule 419 apply to our offering.

Escrow of offering proceeds

\$102,000,000 of the net proceeds of this offering and the sale of the placement warrants will be deposited into a trust account in the United States with Continental Stock Transfer & Trust Company, LLC acting as trustee and JP Morgan Chase Bank, N.A. acting as investment manager.

Terms of Our Offering

Investment of net proceeds

\$102,000,000 of the net offering proceeds and the sale of the placement warrants held in trust will be invested only in U.S. government treasury bills with a maturity of 185 days or less or in money market funds meeting certain conditions under Rule 2a-7 under the Investment Company Act which invest only in direct U.S. government treasury obligations.

Terms Under a Rule 419 Offering

Approximately \$85,500,000 of the offering proceeds would be deposited into either an escrow account with an insured depositary institution or in a separate bank account established by a broker-dealer in which the broker-dealer acts as trustee for persons having the beneficial interests in the account.

Proceeds could be invested only in specified securities such as a money market fund meeting conditions of the Investment Company Act or in securities that are direct obligations of, or obligations guaranteed as to principal or interest by, the United States.

Terms of Our Offering

Terms Under a Rule 419 Offering

Receipt of interest on escrowed funds

Interest on proceeds from the trust account to be paid to stockholders is reduced by (i) any taxes paid or payable, and (ii) in the event of our liquidation for failure to complete our initial business combination within the allotted time, up to \$100,000 of net interest that may be released to us should we have no or insufficient working capital to fund the costs and expenses of our dissolution and liquidation.

Interest on funds in escrow account would be held for the sole benefit of investors, unless and only after the funds held in escrow were released to us in connection with our completion of a business combination.

Limitation on fair value or net assets of target business

Nasdaq rules require that we must complete one or more business combinations having an aggregate fair market value of at least 80% of the value of the assets held in the trust account (excluding the deferred underwriting commissions and taxes payable on the income earned on the trust account) at the time of the agreement to enter into the initial business combination.

The fair value or net assets of a target business must represent at least 80% of the maximum offering proceeds.

Trading of securities issued

We expect the units will begin trading on or promptly after the date of this prospectus. The Class A common stock and warrants comprising the units will begin separate trading on the 52nd day following the date of this prospectus unless the representative informs us of its decision to allow earlier separate trading, subject to our having filed the Current Report on Form 8-K described below and having issued a press release announcing when such separate trading will begin. We will file the Current Report on Form 8-K promptly after the closing of this offering, which is anticipated to take place three business days from the date of this prospectus. If the over-allotment option is exercised following the initial filing of such Current Report on Form 8-K, an additional Current Report on Form 8-K will be filed to provide updated financial information to reflect the exercise of the over-allotment option.

No trading of the units or the underlying Class A common stock and warrants would be permitted until the completion of a business combination. During this period, the securities would be held in the escrow or trust account.

Exercise of the warrants

The warrants cannot be exercised until the later of 30 days after the completion of our initial business combination and 12 months from the closing of this offering.

The warrants could be exercised prior to the completion of a business combination, but securities received and cash paid in connection with the exercise would be deposited in the escrow or trust account.

Election to remain an investor

We will provide our public stockholders with the opportunity to redeem their public shares for cash equal to their pro rata share of the aggregate amount then on deposit in the trust account as of two business days prior to the consummation of our initial business combination, including interest earned on the funds held in the trust account and not previously released to us to pay our taxes, upon the completion of our initial business combination, subject to the limitations described herein. We may not be required by applicable law or stock exchange listing requirements to hold a stockholder vote. We intend to give approximately 30 days (but not less than 10 days nor more than 60 days) prior written notice of any such meeting, if required, at which a vote shall be taken to approve our initial business combination. If we are not required by applicable law or stock exchange listing requirements and do not otherwise decide to hold a stockholder vote, we will, pursuant to our amended and restated certificate of incorporation, conduct the redemptions pursuant to the tender offer rules of the SEC and file tender offer documents with the SEC which will contain substantially the same financial and other information about the initial business combination and the redemption rights as is required under the SEC's proxy rules. In the event we conduct redemptions pursuant to the tender offer rules, our offer to redeem will remain open for at least 20 business days, in accordance with Rule 14e-1(a) under the Exchange Act, and we will not be permitted to complete our initial business combination until the expiration of the tender offer period. If, however, we hold a stockholder vote, we will, like many blank check companies, offer to redeem shares in conjunction with a proxy solicitation pursuant to the proxy rules.

If we seek stockholder approval, we will complete our initial business combination only if a majority of the outstanding shares of common stock voted are voted in favor of the initial business combination. Additionally, each public stockholder may elect to redeem their public shares irrespective of whether they vote for or against the proposed transaction (or abstain from voting). A quorum for such meeting will consist of the holders present in person or by proxy of shares of outstanding capital stock of the company representing a majority of the voting power of all outstanding shares of capital stock of the company entitled to vote at such meeting.

A prospectus containing information pertaining to the business combination required by the SEC would be sent to each investor. Each investor would be given the opportunity to notify the company in writing, within a period of no less than 20 business days and no more than 45 business days from the effective date of a post-effective amendment to the company's registration statement, to decide if it elects to remain a stockholder of the company or require the return of its investment. If the company has not received the notification by the end of the 45th business day, funds and interest or dividends, if any, held in the trust or escrow account are automatically returned to the stockholder. Unless a sufficient number of investors elect to remain investors, all funds on deposit in the escrow account must be returned to all of the investors and none of the securities are issued.

Terms of Our Offering

Terms Under a Rule 419 Offering

Business combination deadline

If we are unable to complete an initial business combination within 12 months from the closing of this offering or during any Extension Period, we will (i) cease all operations except for the purpose of winding up, (ii) as promptly as reasonably possible but not more than ten business days thereafter, redeem 100% of the public shares, at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the trust account including interest earned on the funds held in the trust account and not previously released to us to pay our taxes (less up to \$100,000 of interest to pay dissolution expenses), divided by the number of then outstanding public shares, which redemption will completely extinguish public stockholders' rights as stockholders (including the right to receive further liquidating distributions, if any), subject to applicable law, and (iii) as promptly as reasonably possible following such redemption, subject to the approval of our remaining stockholders and our board of directors, dissolve and liquidate, subject in the case of clauses (ii) and (iii) above to our obligations under Delaware law to provide for claims of creditors and the requirements of other applicable law.

If a business combination has not been completed within 12 months after the effective date of the company's registration statement, funds held in the trust or escrow account are returned to investors.

Limitation on redemption rights of stockholders holding more than 15% of the shares sold in this offering if we hold a stockholder vote If we seek stockholder approval of our initial business combination and we do not conduct redemptions in connection with our initial business combination pursuant to the tender offer rules, our amended and restated certificate of incorporation will provide that a public stockholder (including our affiliates), together with any affiliate of such stockholder or any other person with whom such stockholder is acting in concert or as a "group" (as defined under Section 13 of the Exchange Act), will be restricted from seeking redemption rights with respect to Excess Shares (more than an aggregate of 15% of the shares sold in this offering). Our public stockholders' inability to redeem Excess Shares will reduce their influence over our ability to complete our initial business combination and they could suffer a material loss on their investment in us if they sell any Excess Shares in open market transactions.

Many blank check companies provide no restrictions on the ability of stockholders to redeem shares based on the number of shares held by such stockholders in connection with an initial business combination.

Tendering stock certificates in connection with redemption rights We may require our public stockholders seeking to exercise their redemption rights, whether they are record holders or hold their shares in "street name," to either tender their certificates to our transfer agent up to two business days prior to the vote on the proposal to approve the initial business combination, or to deliver their shares to the transfer agent electronically using The Depository Trust Company's DWAC System, at the holder's option. The proxy materials that we will furnish to holders of our public shares in connection with our initial business combination will indicate whether we are requiring public stockholders to satisfy such delivery requirements. Accordingly, a public stockholder would have up to two days prior to the vote on the initial business combination to tender its shares if it wishes to seek to exercise its redemption rights.

In order to perfect redemption rights in connection with their business combinations, holders could vote against a proposed initial business combination and check a box on the proxy card indicating such holders were seeking to exercise their redemption rights. After the business combination was approved, the company would contact such stockholders to arrange for them to deliver their certificate to verify ownership.

Release of funds

Except with respect to interest earned on the funds held in the trust account that may be released to us to pay our tax obligations and up to \$100,000 of interest that may be used for our dissolution expenses, the proceeds held in the trust account will not be released until the earliest to occur of: (i) the completion of our initial business combination, (ii) the redemption of any public shares properly submitted in connection with a stockholder vote to amend our amended and restated certificate of incorporation (A) to modify the substance or timing of our obligation to allow redemption in connection with our initial business combination or certain amendments to our charter prior thereto or to redeem 100% of our public shares if we do not complete our initial business combination within 12 months from the closing of this offering or during any Extension Period, or (B) with respect to any other provision relating to stockholders' rights or pre-business combination activity and (iii) the redemption of 100% of our public shares if we are unable to complete an initial business combination within the required time frame (subject to the requirements of applicable law). On the completion of our initial business combination, all amounts held in the trust account will be released to us, less amounts released to a separate account controlled by the trustee for disbursal to redeeming stockholders. We will use these funds to pay amounts due to any public stockholders who exercise their redemption rights as described above under "Redemption rights for public stockholders upon completion of our initial business combination," to pay the underwriters their deferred underwriting commissions, to pay all or a portion of the consideration payable to the target or owners of the target of our initial business combination and to pay other expenses associated with our initial business combination.

The proceeds held in the escrow account are not released until the earlier of the completion of a business combination or the failure to effect a business combination within the allotted time.

Competition

In identifying, evaluating and selecting a target business for our initial business combination, we may encounter intense competition from other entities having a business objective similar to ours, including other blank check companies, private equity groups and leveraged buyout funds, public companies and operating businesses seeking strategic acquisitions. Many of these entities are well established and have extensive experience identifying and effecting business combinations directly or through affiliates. Moreover, many of these competitors possess greater financial, technical, human and other resources than we do. Our ability to acquire larger target businesses will be limited by our available financial resources. This inherent limitation gives others an advantage in pursuing the acquisition of a target business. Furthermore, our obligation to pay cash in connection with our public stockholders who exercise their redemption rights may reduce the resources available to us for our initial business combination and our outstanding warrants, and the future dilution they potentially represent, may not be viewed favorably by certain target businesses. Either of these factors may place us at a competitive disadvantage in successfully negotiating an initial business combination.

Facilities

Our executive offices are located at 515 Madison Avenue, Suite 8078, New York, New York 10022, and our telephone number is (646) 908-2658.

Commencing on the date of this prospectus, we have agreed to pay Aesther Healthcare Sponsor, LLC, our sponsor, a total of \$10,000 per month for office space, utilities and secretarial and administrative support. We consider our current office space adequate for our current operations.

Employees

We currently have two officers. These individuals are not obligated to devote any specific number of hours to our matters but they intend to devote as much of their time as they deem necessary, in the exercise of their respective business judgement, to our affairs until we have completed our initial business combination. The amount of time they will devote in any time period will vary based on whether a target business has been selected for our initial business combination and the stage of the initial business combination process we are in. We do not intend to have any full-time employees prior to the completion of our initial business combination. We do not have an employment agreement with any member of our management team.

Periodic Reporting and Financial Information

We will register our units, Class A common stock and warrants under the Exchange Act and have reporting obligations, including the requirement that we file annual, quarterly and current reports with the SEC. In accordance with the requirements of the Exchange Act, our annual reports will contain financial statements audited and reported on by our independent registered public accountants.

We will provide stockholders with audited financial statements of the prospective target business as part of the tender offer materials or proxy solicitation materials sent to stockholders to assist them in assessing the target business. In all likelihood, these financial statements will need to be prepared in accordance with, or reconciled to, GAAP, or IFRS, depending on the circumstances, and the historical financial statements may be required to be audited in accordance with the standards of the PCAOB. These financial statement requirements may limit the pool of potential targets we may conduct an initial business combination with because some targets may be unable to provide such statements in time for us to disclose such statements in accordance with federal proxy rules and complete our initial business combination within the prescribed time frame. We cannot assure you that any particular target business identified by us as a potential business combination candidate will have financial statements prepared in accordance with GAAP or that the potential target business will be able to prepare its financial statements in accordance with the requirements outlined above. To the extent that these requirements cannot be met, we may not be able to acquire the proposed target business. While this may limit the pool of potential business combination candidates, we do not believe that this limitation will be material.

We will be required to evaluate our internal control procedures for the fiscal year ending December 31, 2022 as required by the Sarbanes-Oxley Act. Only in the event we are deemed to be a large accelerated filer or an accelerated filer, and no longer qualify as an emerging growth company, will we be required to have our internal control procedures audited. A target company may not be in compliance with the provisions of the Sarbanes-Oxley Act regarding adequacy of their internal controls. The development of the internal controls of any such entity to achieve compliance with the Sarbanes-Oxley Act may increase the time and costs necessary to complete any such business combination. Prior to the date of this prospectus, we will file a Registration Statement on Form 8-A with the SEC to voluntarily register our securities under Section 12 of the Exchange Act. As a result, we will be subject to the rules and regulations promulgated under the Exchange Act. We have no current intention of filing a Form 15 to suspend our reporting or other obligations under the Exchange Act prior or subsequent to the consummation of our initial business combination.

We are an "emerging growth company," as defined in Section 2(a) of the Securities Act, as modified by the JOBS Act. As such, we are eligible to take advantage of certain exemptions from various reporting requirements that are applicable to other public companies that are not "emerging growth companies" including, but not limited to, not being required to comply with the auditor attestation requirements of Section 404 of the Sarbanes-Oxley Act, reduced disclosure obligations regarding executive compensation in our periodic reports and proxy statements, and exemptions from the requirements of holding a non-binding advisory vote on executive compensation and stockholder approval of any golden parachute payments not previously approved. If some investors find our securities less attractive as a result, there may be a less active trading market for our securities and the prices of our securities may be more volatile.

In addition, Section 107 of the JOBS Act also provides that an "emerging growth company" can take advantage of the extended transition period provided in Section 7(a)(2)(B) of the Securities Act for complying with new or revised accounting standards. In other words, an "emerging growth company" can delay the adoption of certain accounting standards until those standards would otherwise apply to private companies. We intend to take advantage of the benefits of this extended transition period.

We will remain an emerging growth company until the earlier of (1) the last day of the fiscal year (a) following the fifth anniversary of the completion of this offering, (b) in which we have total annual gross revenue of at least \$1.07 billion, or (c) in which we are deemed to be a large accelerated filer, which means the market value of our shares of Class A common stock that are held by non-affiliates exceeds \$700 million as of the prior June 30th, and (2) the date on which we have issued more than \$1.0 billion in non-convertible debt during the prior three-year period. References herein to "emerging growth company" will have the meaning associated with it in the JOBS Act.

Additionally, we are a "smaller reporting company" as defined in Item 10(f)(1) of Regulation S-K. Smaller reporting companies may take advantage of certain reduced disclosure obligations, including, among other things, providing only two years of audited financial statements. We will remain a smaller reporting company until the last day of the fiscal year in which (1) the market value of our common stock held by non-affiliates equals or exceeds \$250 million as of the end of that year's second fiscal quarter, or (2) our annual revenues equaled or exceeded \$100 million during such completed fiscal year and the market value of our common stock held by non-affiliates exceeds \$700 million as of the end of that year's second fiscal quarter.

Legal Proceedings

There is no material litigation, arbitration or governmental proceeding currently pending against us or any members of our management team in their capacity as such.

MANAGEMENT

Officers, Directors and Director Nominees

Our officers, directors and director nominees are as follows:

Name	Age	Position	
Suren Ajjarapu	51	Chairman, Chief Executive Officer and Director	
Howard A. Doss	67	Chief Financial Officer and Secretary	
Michael L. Peterson*	59	Director Nominee	
Venkatesh Srinivasan*	54	Director Nominee	
Donald G. Fell*	74	Director Nominee	
Siva Saravanan*	46	Director Nominee	

^{*} This individual has indicated his assent to occupy such position on the effective date of the registration statement of which this prospectus is a part.

Suren Ajjarapu has served as our Chairman and Chief Executive Officer since our inception in June 2021. He has served as the Chairman of the Board, Chief Executive Officer and Secretary of TRxADE Health, Inc., formerly Trxade Group, Inc. (NASDAQ:MEDS)("TRxADE") since its acquisition of Trxade Group, Inc., a Nevada corporation ("Trxade Nevada") (our predecessor company) on January 8, 2014, and as the Chairman of the Board, Chief Executive Officer and Secretary of Trxade Nevada since its inception. Since March 2021, Mr. Ajjarapu has served on the Board of OceanTech Acquisitions I Corp., a Special Purpose Acquisition Company (SPAC)(NASDAQ:OTECU). Mr. Ajjarapu has served on the Board of Kano Energy, Inc, a private company, which is involved in developing renewable natural gas sites in USA since 2018. Mr. Ajjarapu was a Founder, CEO and Chairman of Sansur Renewable Energy, Inc., a company involved in developing wind power sites in the Midwest, United States, from 2009 to 2012. Mr. Ajjarapu was a Founder, President and Director of Aemetis, Inc., a biofuels company (AMTX.OB) and a Founder, Chairman and Chief Executive Officer of International Biofuels, a subsidiary of Aemetis, Inc., from 2006 to 2009. Mr. Ajjarapu was Co-Founder, COO, and Director Global Information Technology, Inc., an IT outsourcing and systems design company, headquartered in Tampa, Florida with major operations in India from 1995 to 2006. Mr. Ajjarapu holds an MS in Environmental engineering from South Dakota State University, Brookings, South Dakota, and an MBA from the University of South Florida, specializing in International Finance and Management. Mr. Ajjarapu is also a graduate of the Venture Capital and Private Equity program at Harvard University. We believe that his entrepreneurial experience and expertise in screening high quality target companies will be extremely beneficial in sourcing a target with strong growth potential and further believe that we can capitalize on his previous experiences in advising and expanding startups to help

Howard A. Doss has served as our Chief Financial Officer since our inception in June 2021. Since January 2014, he has served as the Chief Financial Officer of TRxADE. Since January 2014. Mr. Doss has served in a variety of capacities with accounting and investment firms. He joined the staff of Seidman & Seidman (BDO Seidman, Dallas) in 1977 and, in 1980, he joined the investment firm Van Kampen Investments, opening the firm's southeast office in Tampa, Florida in 1982. He remained with the firm until 1996 when he joined Franklin Templeton to develop corporate retirement plan distribution. After working for the Principal Financial Group office in Tampa, Florida, Mr. Doss was City Executive for U.S. Trust in Sarasota, Florida, responsible for high net worth individuals. He retired from that position in 2009. He served as CFO and Director for Sansur Renewable Energy, an alternative energy development company, from 2010 to 2012. Mr. Doss has also served as President of STARadio Corp. since 2005. Mr. Doss is a member of the America Institute of CPA's. He is a graduate of Illinois Wesleyan University.

Michael L. Peterson will be one of our directors as of the effective date of the registration statement of which the prospectus forms a part. Mr. Peterson has served as the president of Nevo Motors, Inc. since December 2020, which is in the process of commercializing a range extender generator technology for the heavy-duty electric vehicle market. Mr. Peterson previously served as the president of the Taipei Taiwan Mission of The Church of Jesus Christ of Latter-day Saints, in Taipei, Taiwan from June 2018 to June 2021. Since February 2021, Mr. Peterson has served on the board of directors and as the Chairman of the Audit Committee of Indonesia Energy Corporation Limited (NYSE American: INDO). Mr. Peterson served as an independent member of the Board of Directors of Trxade from August 2016 to May 2021. Mr. Peterson served as the CEO of PEDEVCO Corp. (NYSE American: PED), a public company engaged primarily in the acquisition, exploration, development and production of oil and natural gas shale plays in the US from May 2016 to May 2018. Mr. Peterson served as CFO of PEDEVCO between July 2012 and May 2016, and as Executive Vice President of Pacific Energy Development (PEDEVCO's predecessor) from July 2012 to October 2014, and as PEDEVCO's President from October 2014 to May 2018. Mr. Peterson joined Pacific Energy Development as its Executive Vice President in September 2011, assumed the additional office of Chief Financial Officer in June 2012, and served as a member of its board of directors from July 2012 to September 2013. Mr. Peterson formerly served as Interim President and CEO (from June 2009 to December 2011) and as director (from May 2008 to December 2011) of Pacific Energy Development, as a director (from May 2006 to July 2012) of Aemetis, Inc. (formerly AE Biofuels Inc.), a Cupertino, California-based global advanced biofuels and renewable commodity chemicals company (AMTX.OB), and as Chairman and Chief Executive Officer of Nevo Energy, Inc. (NEVE) (formerly Solargen Energy, Inc.), a Cupertino, California-based developer of utility-scale solar farms which he helped form in December 2008 (from December 2008 to July 2012). From 2005 to 2006, Mr. Peterson served as a managing partner of American Institutional Partners, a venture investment fund based in Salt Lake City. From 2000 to 2004, he served as a First Vice President at Merrill Lynch, where he helped establish a new private client services division to work exclusively with high net worth investors. From September 1989 to January 2000, Mr. Peterson was employed by Goldman Sachs & Co. in a variety of positions and roles, including as a Vice President. Mr. Peterson received his MBA at the Marriott School of Management and a BS in statistics/computer science from Brigham Young University. His skills in managing businesses in public corporations, financial planning and strategic management will be a great asset for the target company, and as such, believe that Mr. Peterson is well qualified to serve on the board of directors of the Company.

Venkatesh Srinivasan will be one of our directors as of the effective date of the registration statement of which the prospectus forms a part. Mr. Srinivasan has served as the President of Micor Labs USA, a pharmaceuticals company since January 2021. From May 2020 to December 2020, he served as President of Rising Pharma, USA, a pharmaceuticals company. From January 2014 to March 2020, he served as President and Chief Executive Officer of Ascend Laboratories, USA, where he grew the business, including building a new team and strengthening processes and systems. In addition, Mr. Srinivasan served as a Director at Pfizer India from August 2006 to January 2014. From January 2005 to July 2006, Mr. Srinivasan served as Vice President at Zydus Cadila India. From January 2002 through January 2005, Mr. Srinivasan served as General Manager at Dr. Reddy's Laboratories India. From May 1999 through January 2002, Mr. Srinivasan served as Deputy General Manager at Sara Lee Godrej India. From May 1997 through April 1999, Mr. Srinivasan served as Business Development Manager at Cargill India. Mr. Srinivasan received a Bachelor of Chemical Engineering degree from the University of Mumbai, a Master's of Science Degree in Chemical Engineering from Washington State University, and an MBA from the Indian Institute of Management. His skillset will be extremely useful as we conduct due diligence works on target companies, and as such, believe that Mr. Srinivasan is well qualified to serve on the board of directors of the Company.

Donald G. Fell will be one of our directors as of the effective date of the registration statement of which the prospectus forms a part. Mr. Fell is presently Professor and Institute Director for the Davis, California-based Foundation for Teaching Economics where Mr. Fell has served since 1992, and adjunct professor of economics for the University of Colorado, Colorado Springs, a position he has held since 2011. Mr. Fell has also served as an independent director of Trxade Health, Inc. since January 2014. From 1985 to May 2006, Mr. Fell served at the University of South Florida as an adjunct professor. From 2006–2012, Mr. Fell held positions with the University of South Florida as a member of the Executive MBA faculty, Director of Executive and Professional Education and Senior Fellow of the Public Policy Institute. He has also served as visiting professor of economics at the University of LaRochelle, France in 2011, and as adjunct professor of economics at both Illinois State University in 1978 and The Ohio State University between 1979 and 1984. Mr. Fell served as Professor and Department Chair at Illinois Valley Community College between 1968 and 1979. Mr. Fell served as Economics Consultant at Sundstrand Corp. from 1976 to 1978. Mr. Fell holds undergraduate and graduate degrees in economics from Indiana State University and is all but dissertation (ABD) in economics from Illinois State University. Through his work with the Foundation for Teaching Economics and the University of Colorado, Colorado Springs he has conducted graduate institutes on economic policy and environmental economics in 44 states, throughout Canada, the Islands and Eastern Europe. We believe that Mr. Fell's extensive experience in the field of economics and business will provide us with valuable insight as we seek to execute our business strategy, and as such, believe that Mr. Fell is well qualified to serve on the board of directors of the Company.

Siva Saravanan will be one of our directors as of the effective date of the registration statement of which the prospectus forms a part. Mr. Saravanan has served as the Chief Digital Officer and Managing Director of Wavestone US, a management and consulting company since March 2020. From August 2018 to February 2020, Mr. Saravanan served as the Senior Vice President and Chief Information Officer for Reviver, an IoT start-up that creates connected digital license plates for autonomous driving. From June 2017 to August 2018, Mr. Saravanan served as the Chief Executive Officer and Principal of Harvis Group Inc., a human resource consulting company. He was also VP of IT Digital Transformation and Program Delivery at Aristocrat Technologies, where he led the transformation of business systems for a leading high-tech gaming manufacturer, from January 2016 to June 2017. Mr. Saravanan spent five years at Verifone as a Senior Director supporting technology operations in 40+ countries and also taking on delivery responsibilities. From December 1998 to August 2000, Mr. Saravanan served as a Principal at Computech Corp. Mr. Saravanan holds a M.S. in Systems Engineering from Tennessee State University and B.S in Mechanical Engineering from Annamalai University in Chidambaram, India. He is also on the Advisory Board of NishTech Inc., a digital commerce company and the Advisory Council of George Washington University School of Business Digital Program. Mr. Saravanan is a member of Forbes Technology Council contributing regularly. We believe that Mr. Saravanan's significant management experience, varied operational knowledge, and experience with cutting edge technology, make him well qualified to serve on the board of directors of the Company

Terms of Office of Officers and Directors

In accordance with Nasdaq corporate governance requirements, we are not required to hold an annual meeting until one year after our first fiscal year end following our listing on Nasdaq. The term of office of our directors will expire at our first annual meeting of stockholders, subject to re-nomination and reappointment to the board by our stockholders.

Our officers are appointed by the board of directors and serve at the discretion of the board of directors, rather than for specific terms of office. Our board of directors is authorized to appoint persons to the offices set forth in our bylaws as it deems appropriate. Our bylaws provide that our officers may consist of a Chairman of the Board, Chief Executive Officer, Chief Financial Officer, President, Vice Presidents, Secretary, Treasurer, Assistant Secretaries and such other offices as may be determined by the board of directors.

Director Independence

Nasdaq listing standards require that a majority of our board of directors be independent. An "independent director" is defined generally as a person other than an officer or employee of the company or its subsidiaries or any other individual having a relationship which in the opinion of the company's board of directors, would interfere with the director's exercise of independent judgment in carrying out the responsibilities of a director. We expect that our board of directors will determine that all of our directors, other than Mr. Ajjarapu are "independent directors" as defined in the Nasdaq listing standards and applicable SEC rules. Our independent directors will have regularly scheduled meetings at which only independent directors are present.

Officer and Director Compensation

None of our officers has received any cash compensation for services rendered to us. Commencing on the date of this prospectus, we have agreed to pay Aesther Healthcare Sponsor, LLC, our sponsor, a total of \$10,000 per month for office space, utilities and secretarial and administrative support. Upon completion of our initial business combination or our liquidation, we will cease paying these monthly fees. No compensation of any kind, including any finder's fee, reimbursement, consulting fee or monies in respect of any payment of a loan, will be paid by us to our sponsor, officers or directors or any affiliate of our sponsor, officers or directors, prior to, or in connection with any services rendered in order to effectuate, the consummation of our initial business combination (regardless of the type of transaction that it is). However, these individuals will be reimbursed for any out-of-pocket expenses incurred in connection with activities on our behalf such as identifying potential target businesses and performing due diligence on suitable business combinations. Our audit committee will review on a quarterly basis all payments that were made to our sponsor, officers or directors or our or their affiliates. Any such payments prior to an initial business combination will be made using funds held outside the trust account. Other than quarterly audit committee review of such payments, we do not expect to have any additional controls in place governing our reimbursement payments to our directors and executive officers for their out-of-pocket expenses incurred in connection with identifying and consummating an initial business combination.

After the completion of our initial business combination, directors or members of our management team who remain with us may be paid consulting or management fees from the combined company. All of these fees will be fully disclosed to stockholders, to the extent then known, in the tender offer materials or proxy solicitation materials furnished to our stockholders in connection with a proposed initial business combination. We have not established any limit on the amount of such fees that may be paid by the combined company to our directors or members of management. It is unlikely the amount of such compensation will be known at the time of the proposed initial business combination, because the directors of the post-combination business will be responsible for determining officer and director compensation. Any compensation to be paid to our officers will be determined, or recommended to the board of directors for determination, either by a compensation committee constituted solely by independent directors or by a majority of the independent directors on our board of directors.

We do not intend to take any action to ensure that members of our management team maintain their positions with us after the consummation of our initial business combination, although it is possible that some or all of our officers and directors may negotiate employment or consulting arrangements to remain with us after our initial business combination. The existence or terms of any such employment or consulting arrangements to retain their positions with us may influence our management's motivation in identifying or selecting a target business but we do not believe that the ability of our management to remain with us after the consummation of our initial business combination will be a determining factor in our decision to proceed with any potential business combination. We are not party to any agreements with our officers and directors that provide for benefits upon termination of employment.

Committees of the Board of Directors

Our board of directors will have three standing committees: an audit committee, a compensation committee and a nominating and corporate governance committee. Subject to phase-in rules and a limited exception, Nasdaq rules and Rule 10A-3 of the Exchange Act require that the audit committee of a listed company be comprised solely of independent directors, and Nasdaq rules require that the compensation committee of a listed company be comprised solely of independent directors.

Audit Committee

Prior to the consummation of this offering, we will establish an audit committee of the board of directors. Messrs. Peterson, Srinivasan and Fell will serve as members of our audit committee, and Mr. Peterson will chair the audit committee. Under the Nasdaq listing standards and applicable SEC rules, we are required to have at least three members of the audit committee, all of whom must be independent. Each of Messrs. Peterson, Srinivasan and Fell meet the independent director standard under Nasdaq listing standards and under Rule 10-A-3(b)(1) of the Exchange Act.

Each member of the audit committee is financially literate and our board of directors has determined that Mr. Peterson qualifies as an "audit committee financial expert" as defined in applicable SEC rules.

We will adopt an audit committee charter, which will detail the principal functions of the audit committee, including:

- the appointment, compensation, retention, replacement, and oversight of the work of the independent registered public accounting firm engaged by us;
- pre-approving all audit and permitted non-audit services to be provided by the independent registered public accounting firm engaged by us, and establishing pre-approval policies and procedures;
- setting clear hiring policies for employees or former employees of the independent registered public accounting firm, including but not limited to, as required by applicable laws and regulations;
- setting clear policies for audit partner rotation in compliance with applicable laws and regulations;
- obtaining and reviewing a report, at least annually, from the independent registered public accounting firm describing (i) the independent registered public accounting firm's internal quality-control procedures, (ii) any material issues raised by the most recent internal quality-control review, or peer review, of the audit firm, or by any inquiry or investigation by governmental or professional authorities within the preceding five years respecting one or more independent audits carried out by the firm and any steps taken to deal with such issues and (iii) all relationships between the independent registered public accounting firm and us to assess the independent registered public accounting firm's independence;
- reviewing and approving any related party transaction required to be disclosed pursuant to Item 404 of Regulation S-K promulgated by the SEC prior to us entering into such transaction; and
- reviewing with management, the independent registered public accounting firm, and our legal advisors, as appropriate, any legal, regulatory or compliance matters, including any correspondence with regulators or government agencies and any employee complaints or published reports that raise material issues regarding our financial statements or accounting policies and any significant changes in accounting standards or rules promulgated by the Financial Accounting Standards Board, the SEC or other regulatory authorities.

Compensation Committee

Prior to the consummation of this offering, we will establish a compensation committee of the board of directors. Messrs. Peterson, Srinivasan and Fell will serve as members of our compensation committee. Under the Nasdaq listing standards and applicable SEC rules, we are required to have at least two members of the compensation committee, all of whom must be independent. Each of Messrs. Peterson, Srinivasan and Fell are independent, and Mr. Fell will chair the compensation committee.

We will adopt a compensation committee charter, which will detail the principal functions of the compensation committee, including:

- reviewing and approving on an annual basis the corporate goals and objectives relevant to our Chief Executive Officer's compensation, if any is paid by us, evaluating our Chief Executive Officer's performance in light of such goals and objectives and determining and approving the remuneration (if any) of our Chief Executive Officer based on such evaluation;
- reviewing and approving on an annual basis the compensation, if any is paid by us, of all of our other officers;
- reviewing on an annual basis our executive compensation policies and plans;
- implementing and administering our incentive compensation equity-based remuneration plans;
- assisting management in complying with our proxy statement and annual report disclosure requirements;
- approving all special perquisites, special cash payments and other special compensation and benefit arrangements for our officers and employees;
- if required, producing a report on executive compensation to be included in our annual proxy statement; and
- reviewing, evaluating and recommending changes, if appropriate, to the remuneration for directors.

Notwithstanding the foregoing, as indicated above, other than the payment to Aesther Healthcare Sponsor, LLC, our sponsor, of \$10,000 per month, for up to 18 months, for the office space, utilities, and secretarial and administrative support, no compensation of any kind, including finders, consulting or other similar fees, will be paid to any of our existing stockholders, officers, directors or any of their respective affiliates, prior to, or for any services they render in order to effectuate the consummation of an initial business combination. Accordingly, it is likely that prior to the consummation of an initial business combination, the compensation committee will only be responsible for the review and recommendation of any compensation arrangements to be entered into in connection with such initial business combination.

The charter will also provide that the compensation committee may, in its sole discretion, retain or obtain the advice of a compensation consultant, legal counsel or other adviser and will be directly responsible for the appointment, compensation and oversight of the work of any such adviser. However, before engaging or receiving advice from a compensation consultant, external legal counsel or any other adviser, the compensation committee will consider the independence of each such adviser, including the factors required by Nasdaq and the SEC.

Nominating and Corporate Governance Committee

Upon the effectiveness of the registration statement of which this prospectus forms a part, we will establish a nominating and corporate governance committee. The members of our nominating and corporate governance will be Donald G. Fell, Michael L. Peterson and Venkatesh Srinivasan. Donald G. Fell will serve as chair of the nominating and corporate governance committee.

The primary purposes of our nominating and corporate governance committee will be to assist the board in:

- identifying, screening and reviewing individuals qualified to serve as directors and recommending to the board of directors candidates for nomination for election at the annual meeting of stockholders or to fill vacancies on the board of directors;
- developing, recommending to the board of directors and overseeing implementation of our corporate governance guidelines;
- coordinating and overseeing the annual self-evaluation of the board of directors, its committees, individual directors and management in the governance of the company; and
- reviewing on a regular basis our overall corporate governance and recommending improvements as and when necessary.

The nominating and corporate governance committee will be governed by a charter that complies with the rules of the Nasdaq.

Director Nominations

Our nominating and corporate governance committee will recommend to the board of directors candidates for nomination for election at the annual meeting of the stockholders. The board of directors will also consider director candidates recommended for nomination by our stockholders during such times as they are seeking proposed nominees to stand for election at the next annual meeting of stockholders (or, if applicable, a special meeting of stockholders).

We have not formally established any specific, minimum qualifications that must be met or skills that are necessary for directors to possess. In general, in identifying and evaluating nominees for director, the board of directors considers educational background, diversity of professional experience, knowledge of our business, integrity, professional reputation, independence, wisdom, and the ability to represent the best interests of our stockholders. Prior to our initial business combination, holders of our public shares will not have the right to recommend director candidates for nomination to our board of directors.

Compensation Committee Interlocks and Insider Participation

None of our executive officers currently serves, and in the past year has not served, as a member of the board of directors or compensation committee of any entity that has one or more executive officers serving on our board of directors.

Code of Ethics

We have adopted a Code of Ethics applicable to our directors, officers and employees. We have filed a copy of our Code of Ethics and our audit and compensation committee charters as exhibits to the registration statement of which this prospectus is a part. You will be able to review these documents by accessing our public filings at the SEC's web site at *www.sec.gov*. In addition, a copy of the Code of Ethics will be provided without charge upon request from us. We intend to disclose any amendments to or waivers of certain provisions of our Code of Ethics in a Current Report on Form 8-K. See the section of this prospectus entitled "Where You Can Find Additional Information."

Conflicts of Interest

Subject to pre-existing fiduciary or contractual duties as described below, our officers and directors have agreed to present any business opportunities presented to them in their capacity as a director or officer of our company to us. Certain of our officers and directors presently have fiduciary or contractual obligations to other entities pursuant to which such officer or director is or will be required to present a business combination opportunity. Accordingly, if any of our officers or directors becomes aware of a business combination opportunity which is suitable for an entity to which he or she has then-current fiduciary or contractual obligations, he or she will honor his or her fiduciary or contractual obligations to present such opportunity to such entity. We believe, however, that the fiduciary duties or contractual obligations of our officers or directors will not materially affect our ability to complete our initial business combination. Our amended and restated certificate of incorporation will provide that we renounce our interest in any corporate opportunity offered to any director or officer unless such opportunity is expressly offered to such person solely in his or her capacity as a director or officer of our company and such opportunity is one we are legally and contractually permitted to undertake and would otherwise be reasonable for us to pursue, and to the extent the director or officer is permitted to refer that opportunity to us without violating another legal obligation.

Our officers and directors may become officers or directors of another special purpose acquisition company with a class of securities intended to be registered under the Exchange Act, even prior to us entering into a definitive agreement for our initial business combination. Additionally, Mr. Ajjarapu is currently a director of Oceantech Acquisition I Corp., traded on Nasdaq under the symbol "OTECU" a special purpose acquisition company with a class of securities registered under the Exchange Act that focuses its search for a business combination in the leisure marine, yachting and superyachting industries. As of the date of this prospectus, Oceantech Acquisition I Corp. has not entered into a business combination.

Potential investors should also be aware of the following other potential conflicts of interest:

- None of our officers or directors is required to commit his or her full time to our affairs and, accordingly, may have conflicts of interest in allocating his or her time among various business activities.
- In the course of their other business activities, our officers and directors may become aware of investment and business opportunities which may be appropriate for presentation to us as well as the other entities with which they are affiliated. Our management may have conflicts of interest in determining to which entity a particular business opportunity should be presented.

- Our initial stockholders have agreed to waive their redemption rights with respect to any founder shares and any public shares held by them in connection with the consummation of our initial business combination. Additionally, our initial stockholders have agreed to waive their redemption rights with respect to any founder shares held by them if we fail to consummate our initial business combination within 12 months from the closing of this offering or during any Extension Period. If we do not complete our initial business combination within such applicable time period, the proceeds of the sale of the placement warrants held in the trust account will be used to fund the redemption of our public shares, and the placement securities will expire worthless. With certain limited exceptions, the founder shares will not be transferable, assignable by our sponsor until the earlier to occur of: (A) one year after the completion of our initial business combination and (B) subsequent to our initial business combination, (x) if the reported last sale price of our Class A common stock equals or exceeds \$12.00 per share (as adjusted for stock splits, stock dividends, right issuances, reorganizations, recapitalizations and the like) for any 20 trading days within any 30trading day period commencing at least 150 days after our initial business combination, or (y) the date on which we complete a liquidation, merger, capital stock exchange, reorganization or other similar transaction that results in all of our stockholders having the right to exchange their shares of common stock for cash, securities or other property. With certain limited exceptions, the placement warrants and the Class A common stock underlying such warrants, will not be transferable, assignable or saleable by our sponsor or its permitted transferees until 30 days after the completion of our initial business combination. Since our sponsor and officers and directors may directly or indirectly own common stock, warrants following this offering, our officers and directors may have a conflict of interest in determining whether a particular target business is an appropriate business with which to effectuate our initial business combination.
- Our officers and directors may have a conflict of interest with respect to evaluating a particular business combination if the retention or resignation of any such officers and directors was included by a target business as a condition to any agreement with respect to our initial business combination.
- Our sponsor, officers or directors may have a conflict of interest with respect to evaluating a business combination and financing arrangements as we may obtain loans from our sponsor or an affiliate of our sponsor or any of our officers or directors to finance transaction costs in connection with an intended initial business combination. Up to \$1,500,000 of such loans may be convertible into warrants, at a price of \$1.00 per warrant at the option of the lender, upon consummation of our initial business combination. The warrants would be identical to the placement warrants.

The conflicts described above may not be resolved in our favor.

In general, officers and directors of a corporation incorporated under the laws of the State of Delaware are required to present business opportunities to a corporation if:

- the corporation could financially undertake the opportunity;
- the opportunity is within the corporation's line of business; and
- it would not be fair to our company and its stockholders for the opportunity not to be brought to the attention of the corporation.

Accordingly, as a result of multiple business affiliations, our officers and directors may have similar legal obligations relating to presenting business opportunities meeting the above-listed criteria to multiple entities. Furthermore, our amended and restated certificate of incorporation will provide that we renounce our interest in any corporate opportunity offered to any director or officer unless such opportunity is expressly offered to such person solely in his or her capacity as a director or officer of our company and such opportunity is one we are legally and contractually permitted to undertake and would otherwise be reasonable for us to pursue, and to the extent the director or officer is permitted to refer that opportunity to us without violating another legal obligation.

Below is a table summarizing the entities to which our executive officers, directors and director nominees currently have fiduciary duties or contractual obligations:

Individual ⁽¹⁾	Entity ⁽²⁾	Entity's Business	Affiliation	
Suren Ajjarapu	Oceantech Acquisitions I Corp.	Special Purpose Acquisition Company	Director	
	TRxADE HEALTH, INC.	Health services IT company	Chief Executive Officer, Chairman and Secretary	
	Kano Energy Corp.	Renewable natural gas	Executive Chairman	
Howard A. Doss	TRxADE HEALTH, INC.	Health services IT company	Chief Financial Officer	
	STARadio Corp.	Radio Broadcast Company	President	
Michael L. Peterson	Nevo Motors, Inc.	Commercializing a range extender generator technology for the heavy-duty electric vehicle market	President	
	Indonesia Energy Corporation Limited	Oil and Gas Company	Director and Chairman of Audit Committee	
Venkatesh Srinivasan	Micro Labs USA	Pharmaceutical Company	President	
Donald G. Fell	Foundation for Teaching Economics	Workshop and educational resource company	Professor and Institute Director	
	University of Colorado, Colorado Springs	University	Adjunct Professor	
	TRxADE HEALTH, INC.	Health services IT company	Independent Director, Chair of compensation and nominating and corporate governance committees, member audit committee	
Siva Saravanan	Wavestone US	Management and Digital	Chief Digital officer and Managing Director	
Siva Saravanan		Consulting		
	NishTech Inc.	Content Management and Digital Marketing	Advisory Board Member	

- (1) Each person has a fiduciary duty with respect to the listed entities next to their respective names.
- (2) Each of the entities listed in this table has priority and preference relative to our company with respect to the performance by each individual listed in this table of his obligations and the presentation by each such individual of business opportunities.

Accordingly, if any of the above executive officers, directors or any director nominee becomes aware of a business combination opportunity which is suitable for any of the above entities to which he or she has current fiduciary or contractual obligations, he or she will honor his or her fiduciary or contractual obligations to present such business combination opportunity to such entity, and only present it to us if such entity rejects the opportunity.

We are not prohibited from pursuing an initial business combination with a company that is affiliated with our sponsor, officers or directors. In the event we seek to complete our initial business combination with such a company, we, or a committee of independent directors, would obtain an opinion from an independent investment banking firm or another independent entity that commonly renders valuation opinions, that such an initial business combination is fair to our company from a financial point of view.

In the event that we submit our initial business combination to our public stockholders for a vote, pursuant to the letter agreement, our sponsor, officers and directors have agreed to vote and our director nominees will agree to vote, any founder shares held by them and any public shares purchased during or after the offering (including in open market and privately negotiated transactions) in favor of our initial business combination.

Limitation on Liability and Indemnification of Officers and Directors

Our amended and restated certificate of incorporation will provide that our officers and directors will be indemnified by us to the fullest extent authorized by Delaware law, as it now exists or may in the future be amended. In addition, our amended and restated certificate of incorporation will provide that our directors will not be personally liable for monetary damages to us or our stockholders for breaches of their fiduciary duty as directors, unless they violated their duty of loyalty to us or our stockholders, acted in bad faith, knowingly or intentionally violated the law, authorized unlawful payments of dividends, unlawful stock purchases or unlawful redemptions, or derived an improper personal benefit from their actions as directors.

We will enter into agreements with our officers and directors to provide contractual indemnification in addition to the indemnification provided for in our amended and restated certificate of incorporation. Our bylaws also will permit us to secure insurance on behalf of any officer, director or employee for any liability arising out of his or her actions, regardless of whether Delaware law would permit such indemnification. We will purchase a policy of directors' and officers' liability insurance that insures our officers and directors against the cost of defense, settlement or payment of a judgment in some circumstances and insures us against our obligations to indemnify our officers and directors.

These provisions may discourage stockholders from bringing a lawsuit against our directors for breach of their fiduciary duty. These provisions also may have the effect of reducing the likelihood of derivative litigation against officers and directors, even though such an action, if successful, might otherwise benefit us and our stockholders. Furthermore, a stockholder's investment may be adversely affected to the extent we pay the costs of settlement and damage awards against officers and directors pursuant to these indemnification provisions.

We believe that these provisions, the directors' and officers' liability insurance and the indemnity agreements are necessary to attract and retain talented and experienced officers and directors.

PRINCIPAL STOCKHOLDERS

The following table sets forth information regarding the beneficial ownership of our common stock as of the date of this prospectus, and as adjusted to reflect the sale of our common stock included in the units offered by this prospectus, and assuming no purchase of units in this offering, by:

- each person known by us to be the beneficial owner of more than 5% of our outstanding shares of common stock;
- each of our executive officers, directors and director nominees that beneficially owns shares of our common stock; and
- all our executive officers, directors and director nominees as a group.

Unless otherwise indicated, we believe that all persons named in the table have sole voting and investment power with respect to all shares of common stock beneficially owned by them. The following table does not reflect record or beneficial ownership of the placement warrants as these warrants are not exercisable within 60 days of the date of this prospectus.

On June 30, 2021, our sponsor purchased 2,875,000 founder shares for an aggregate purchase price of \$25,000, or approximately \$0.009 per share. In addition, our sponsor has committed, pursuant to a written agreement, to purchase an aggregate of 5,261,000 placement warrants for a purchase price of \$1.00 per warrant in a private placement that will occur simultaneously with the closing of this offering (assuming the underwriters do not exercise their over-allotment option). The following table presents the number of shares and percentage of our common stock beneficially owned as of the filing date, before and after this offering, by each person, or group of persons, known to us who beneficially owns more than 5% of our capital stock, each named executive officer, each of our directors and all directors and executive officers as a group. The post-offering numbers and percentages presented assume that the underwriters do not exercise their over-allotment option, that our sponsor forfeits 375,000 founder shares on a pro rata basis, and that there are 12,600,000 shares of our common stock issued and outstanding after this offering, consisting of (i) 10,100,000 shares of our Class A common stock (including 100,000 shares issuable to the representative of the underwriters as underwriter's compensation) and (ii) 2,500,000 shares of our Class B common stock, issued and outstanding after this offering.

	Before Offering		After Offering	
Name and Address of Beneficial Owner ⁽¹⁾	Number of Shares Beneficially Owned ⁽²⁾	Approximate Percentage of Outstanding Common Stock	Number of Shares Beneficially Owned ⁽²⁾	Approximate Percentage of Outstanding Common Stock
Aesther Healthcare Sponsor, LLC ⁽¹⁾⁽²⁾	2,875,000	100%	2,500,000	19.8%
Suren Ajjarapu ⁽¹⁾⁽²⁾	2,875,000	100%	2,500,000	19.8%
Howard A. Doss	_	_	_	_
Michael L. Peterson ⁽³⁾	_	_	_	_
Venkatesh Srinivasan ⁽³⁾	_	_	_	_
Donald G. Fell ⁽³⁾	_	_	_	_
Siva Saravanan ⁽³⁾	_	_	_	_
All executive officers and directors as a group (6 individuals)	2,875,000	100%	2,500,000	19.8%

- (1) Aesther Healthcare, LLC, our sponsor, is the record holder of the securities reported herein. Suren Ajjarapu, our Chairman and Chief Executive Officer, is the manager and member of our sponsor. By virtue of this relationship, Mr. Ajjarapu may be deemed to share beneficial ownership of the securities held of record by our sponsor. Mr. Ajjarapu disclaims any such beneficial ownership except to the extent of his pecuniary interest. The business address of each of these entities and individuals is 515 Madison Avenue, Suite 8078, New York, New York 10022.
- (2) Interests shown consist solely of founder shares, classified as shares of Class B common stock, after this offering (assumes the underwriters' overallotment option has not been exercised, and an aggregate of 375,000 founder shares have been forfeited by our sponsor). Founder shares are convertible into shares of Class A common stock on a one-for-one basis, subject to adjustment, as described in the section of this prospectus entitled "Description of Securities."
- (3) Will be one of our directors as of the effective date of the registration statement of which the prospectus forms a part.

After giving effect to the issuance of founder shares, our initial stockholders will own 20% of the outstanding common stock following the offering (excluding the placement warrants and the underlying securities, and the representative's shares and assuming they do not purchase any units in this offering or the public market). Because of this ownership block, our initial stockholders will have significant influence over the outcome of all matters requiring approval by our stockholders, including the election of directors, amendments to our amended and restated certificate of incorporation and approval of significant corporate transactions other than approval of our initial business combination. If we increase or decrease the size of the offering, we will effect a stock dividend or a share contribution back to capital, or other appropriate mechanism, as applicable, with respect to our Class B common stock immediately prior to the consummation of the offering in such amount as to maintain the ownership of our initial stockholders at 20% of the issued and outstanding shares of our common stock (excluding the placement warrants and the underlying securities, and the representative's shares and assuming they do not purchase any units in this offering) upon the consummation of this offering. The initial stockholders have agreed (A) to vote any shares owned by them in favor of any proposed initial business combination and (B) not to redeem any shares in connection with a stockholder vote to approve a proposed initial business combination.

Our sponsor, executive officers and directors are deemed to be our "promoters" as such term is defined under the federal securities laws.

Restrictions on Transfers of Founder Shares and Placement Warrants

The founder shares, and placement warrants, and securities contained therein, are each subject to transfer restrictions pursuant to lock-up provisions in a letter agreement with us to be entered into by our sponsor, officers and directors, and director nominees. Those lock-up provisions provide that such securities are not transferable or salable (i) in the case of the founder shares (or shares of common stock issuable upon conversion thereof), until the earlier to occur of: (A) one year after the completion of our initial business combination and (B) subsequent to our initial business combination, (x) if the reported last sale price of our Class A common stock equals or exceeds \$12.00 per share (as adjusted for stock splits, stock dividends, right issuances, reorganizations, recapitalizations and the like) for any 20 trading days within any 30-trading day period commencing at least 150 days after our initial business combination, or (y) the date on which we complete a liquidation, merger, capital stock exchange, reorganization or other similar transaction that results in all of our stockholders having the right to exchange their shares of common stock for cash, securities or other property, and (ii) in the case of the placement warrants, including the shares of Class A common stock issuable upon exercise of the placement warrants, until 30 days after the completion of our initial business combination, except in each case (a) to our or our sponsor's officers, directors, or consultants, any affiliates or family members of any of our officers, directors, or consultants, any members of our sponsor, or any affiliates of our sponsor, (b) in the case of an individual, by gift to a member of one of the members of the individual's immediate family or to a trust, the beneficiary of which is a member of one of the individual's immediate family, an affiliate of such person or to a charitable organization; (c) in the case of an individual, by virtue of laws of descent and distribution upon death of any of our officers, our directors, the initial stockholders or members of our sponsor; (d) in the case of an individual, pursuant to a qualified domestic relations order; (e) by private sales or transfers made in connection with the consummation of an initial business combination at prices no greater than the price at which the securities were originally purchased; (f) in the event of our liquidation prior to the completion of our initial business combination; (g) in the event of our liquidation, merger, capital stock exchange, reorganization or other similar transaction which results in all of our stockholders having the right to exchange their shares of common stock for cash, securities or other property subsequent to our completion of our initial business combination; or (h) to the Company for no value for cancellation in connection with the consummation of a business combination, provided, however, that in the case of clauses (a) through (e) these permitted transferees must enter into a written agreement agreeing to be bound by these transfer restrictions and the other restrictions contained in the letter agreements and by the same agreements entered into by our sponsor with respect to such securities (including provisions relating to voting, the trust account and liquidating distributions described elsewhere in this prospectus).

Registration Rights

The holders of the founder shares, placement warrants, and warrants that may be issued upon conversion of working capital loans, and any shares of Class A common stock issuable upon the exercise of the placement warrants and any warrants (and underlying Class A common stock) that may be issued upon conversion of working capital loans and Class A common stock issuable upon conversion of the founder shares, will be entitled to registration rights pursuant to a registration rights agreement to be signed prior to or on the effective date of this offering, requiring us to register such securities for resale (in the case of the founder shares, only after conversion to our Class A common stock). The holders of the majority of these securities are entitled to make up to three demands, excluding short form demands, that we register such securities. In addition, the holders have certain "piggy-back" registration rights with respect to registration statements filed subsequent to our completion of our initial business combination and rights to require us to register for resale such securities pursuant to Rule 415 under the Securities Act. The registration rights agreement does not contain liquidated damages or other cash settlement provisions resulting from delays in registering our securities. We will bear the expenses incurred in connection with the filing of any such registration statements.

CERTAIN RELATIONSHIPS AND RELATED PARTY TRANSACTIONS

On June 30, 2021, our sponsor purchased 2,875,000 founder shares for an aggregate purchase price of \$25,000, or approximately \$0.009 per share. The number of founder shares issued was determined based on the expectation that such founder shares would represent 20% of the outstanding shares upon completion of this offering (excluding the placement warrants and underlying securities, and the representative's shares). If we increase or decrease the size of the offering we will effect a stock dividend or a share contribution back to capital or other appropriate mechanism, as applicable, with respect to our Class B common stock immediately prior to the consummation of the offering in such amount as to maintain the ownership of our initial stockholders at 20.0% of the issued and outstanding shares of our common stock (excluding the placement warrants and the underlying securities, and the representative's shares and assuming they do not purchase any units in this offering) upon the consummation of this offering. Up to 375,000 founder shares held by our sponsor are subject to forfeiture by our sponsor depending on the extent to which the underwriters' over-allotment option is exercised. The founder shares (including the Class A common stock issuable upon exercise thereof) may not, subject to certain limited exceptions, be transferred, assigned or sold by the holder.

Our sponsor has agreed to purchase an aggregate of 5,261,000 placement warrants at a price of \$1.00 per warrant for an aggregate purchase price of \$5,261,000. If the over-allotment option is exercised in full, the amount of placement warrants sold will be 5,711,000 for an aggregate purchase price of \$5,711,000. There will be no redemption rights or liquidating distributions from the trust account with respect to the founder shares or placement warrants, which will expire worthless if we do not consummate a business combination within 12 months from the closing of this offering or during any Extension Period. The private placement warrants will be identical to the warrants sold in this offering except that the private placement warrants, so long as they are held by our sponsor, the underwriters or their permitted transferees, (i) may not (including the shares of Class A common stock issuable upon exercise of these warrants), subject to certain limited exceptions, be transferred, assigned or sold by the holders until 30 days after the completion of our initial business combination, and (ii) will be entitled to registration rights. The private placement warrants (including the shares of Class A common stock issuable upon exercise thereof) may not, subject to certain limited exceptions, be transferred, assigned or sold by the holder.

No compensation of any kind, including any finder's fee, reimbursement, consulting fee or monies in respect of any payment of a loan, will be paid by us to our sponsor, officers or directors or any affiliate of our sponsor, officers or directors prior to, or in connection with any services rendered in order to effectuate, the consummation of an initial business combination (regardless of the type of transaction that it is). However, these individuals will be reimbursed for any out-of-pocket expenses incurred in connection with activities on our behalf such as identifying potential target businesses and performing due diligence on suitable business combinations. Our audit committee will review on a quarterly basis all payments that were made to our sponsor, officers, directors or our or their affiliates and will determine which expenses and the amount of expenses that will be reimbursed. There is no cap or ceiling on the reimbursement of out-of-pocket expenses incurred by such persons in connection with activities on our behalf.

Prior to the closing of this offering, our sponsor has agreed to loan us up to \$300,000 to be used for a portion of the expenses of this offering. These loans are non-interest bearing, unsecured and are due at the earlier of June 30, 2022 or the closing of this offering. As of July 12, 2021, the Company had borrowed \$64,798 from the sponsor. The loan will be repaid upon the closing of this offering out of the estimated \$1,261,000 of offering proceeds that has been allocated to the payment of offering expenses (other than underwriting commissions). The value of our sponsor's interest in this transaction corresponds to the principal amount outstanding under any such loan.

In addition, in order to finance transaction costs in connection with an intended initial business combination, our sponsor or an affiliate of our sponsor or certain of our officers and directors may, but are not obligated to, loan us funds on a non-interest bearing basis as may be required. If we complete an initial business combination, we would repay such loaned amounts. In the event that the initial business combination does not close, we may use a portion of the working capital held outside the trust account to repay such loaned amounts but no proceeds from our trust account would be used for such repayment. Up to \$1,500,000 of such loans may be convertible into warrants, at a price of \$1.00 per warrant at the option of the lender, upon consummation of our initial business combination. The warrants would be identical to the placement warrants. Other than as described above, the terms of such loans by our officers and directors, if any, have not been determined and no written agreements exist with respect to such loans.

We do not expect to seek loans from parties other than our sponsor or an affiliate of our sponsor as we do not believe third parties will be willing to loan such funds and provide a waiver against any and all rights to seek access to funds in our trust account.

After our initial business combination, members of our management team who remain with us may be paid consulting, management or other fees from the combined company with any and all amounts being fully disclosed to our stockholders, to the extent then known, in the tender offer or proxy solicitation materials, as applicable, furnished to our stockholders. It is unlikely the amount of such compensation will be known at the time of distribution of such tender offer materials or at the time of a stockholder meeting held to consider our initial business combination, as applicable, as it will be up to the directors of the post-combination business to determine executive and director compensation.

The holders of the founder shares, placement warrants, and warrants that may be issued upon conversion of working capital loans (and in each case holders of their component securities, as applicable) will have registration rights to require us to register a sale of any of our securities held by them pursuant to a registration rights agreement to be signed prior to or on the effective date of this offering. These holders will be entitled to make up to three demands, excluding short form registration demands, that we register such securities for sale under the Securities Act. In addition, these holders will have "piggy-back" registration rights to include their securities in other registration statements filed by us.

We will enter into agreements with our officers and directors to provide contractual indemnification in addition to the indemnification provided for in our amended and restated certificate of incorporation. Our bylaws also will permit us to secure insurance on behalf of any officer, director or employee for any liability arising out of his or her actions, regardless of whether Delaware law would permit such indemnification. We will purchase a policy of directors' and officers' liability insurance that insures our officers and directors against the cost of defense, settlement or payment of a judgment in some circumstances and insures us against our obligations to indemnify our officers and directors.

Related Party Policy

We have not yet adopted a formal policy for the review, approval or ratification of related party transactions. Accordingly, the transactions discussed above were not reviewed, approved or ratified in accordance with any such policy.

Prior to the consummation of this offering, we will adopt a code of ethics requiring us to avoid, wherever possible, all conflicts of interests, except under guidelines or resolutions approved by our board of directors (or the appropriate committee of our board) or as disclosed in our public filings with the SEC. Under our code of ethics, conflict of interest situations will include any financial transaction, arrangement or relationship (including any indebtedness or guarantee of indebtedness) involving the company. A form of the code of ethics that we plan to adopt prior to the consummation of this offering is filed as an exhibit to the registration statement of which this prospectus is a part.

In addition, our audit committee, pursuant to a written charter that we will adopt prior to the consummation of this offering, will be responsible for reviewing and approving related party transactions to the extent that we enter into such transactions. An affirmative vote of a majority of the members of the audit committee present at a meeting at which a quorum is present will be required in order to approve a related party transaction. A majority of the members of the entire audit committee will constitute a quorum. Without a meeting, the unanimous written consent of all of the members of the audit committee will be required to approve a related party transaction. A form of the audit committee charter that we plan to adopt prior to the consummation of this offering is filed as an exhibit to the registration statement of which this prospectus is a part. We also require each of our directors and executive officers to complete a directors' and officers' questionnaire that elicits information about related party transactions.

These procedures are intended to determine whether any such related party transaction impairs the independence of a director or presents a conflict of interest on the part of a director, employee or officer.

To further minimize conflicts of interest, we have agreed not to consummate an initial business combination with an entity that is affiliated with any of our sponsor, officers or directors unless we, or a committee of independent directors, have obtained an opinion from an independent investment banking firm or another independent entity that commonly renders valuation opinions that our initial business combination is fair to our company from a financial point of view. Furthermore, no finder's fees, reimbursements, consulting fee, monies in respect of any payment of a loan or other compensation will be paid by us to our sponsor, officers or directors or any affiliate of our sponsor, officers or directors prior to, for services rendered to us prior to, or in connection with any services rendered in order to effectuate, the consummation of our initial business combination (regardless of the type of transaction that it is). However, the following payments will be made to our sponsor, officers or directors, or our or their affiliates, none of which will be made from the proceeds of this offering held in the trust account prior to the completion of our initial business combination:

- Repayment of up to an aggregate of \$300,000 in loans made to us by our sponsor to cover offering-related and organizational expenses;
- Payment to Aesther Healthcare Sponsor, LLC, our sponsor, of \$10,000 per month, for up to 18 months, for office space, utilities and secretarial
 and administrative support;
- Reimbursement for any out-of-pocket expenses related to identifying, investigating and completing an initial business combination; and
- Repayment of non-interest bearing loans which may be made by our sponsor or an affiliate of our sponsor or certain of our officers and directors to finance transaction costs in connection with an intended initial business combination, the terms of which (other than as described above) have not been determined nor have any written agreements been executed with respect thereto. Up to \$1,500,000 of such loans may be convertible into warrants, at a price of \$1.00 per warrant at the option of the lender, upon consummation of our initial business combination. The warrants would be identical to the placement warrants.

Our audit committee will review on a quarterly basis all payments that were made to our sponsor, officers, directors or our or their affiliates.

DESCRIPTION OF SECURITIES

Pursuant to our amended and restated certificate of incorporation, our authorized capital stock consists of 125,000,000 shares of Class A common stock, \$0.0001 par value, 12,500,000 shares of Class B common stock, \$0.0001 par value, and 1,250,000 shares of undesignated preferred stock, \$0.0001 par value. The following description summarizes the material terms of our capital stock. Because it is only a summary, it may not contain all the information that is important to you.

Units

Each unit has an offering price of \$10.00 and consists of one share of Class A common stock and one-half of one redeemable warrant. Each whole warrant entitles the holder to purchase one share of common stock.

We expect the Class A common stock and warrants comprising the units will begin separate trading on the 52nd day following the date of this prospectus unless the representative informs us of its decision to allow earlier separate trading, subject to our having filed the Current Report on Form 8-K described below and having issued a press release announcing when such separate trading will begin. Once the shares of Class A common stock and warrants commence separate trading, holders will have the option to continue to hold units or separate their units into the component securities. Holders will need to have their brokers contact our transfer agent in order to separate the units into shares of Class A common stock and warrants.

In no event will the Class A common stock and warrants be traded separately until we have filed with the SEC a Current Report on Form 8-K which includes an audited balance sheet reflecting our receipt of the gross proceeds at the closing of this offering. We will file a Current Report on Form 8-K which includes this audited balance sheet upon the completion of this offering, which is anticipated to take place three business days after the date of this prospectus. If the underwriters' over-allotment option is exercised following the initial filing of such Current Report on Form 8-K, a second or amended Current Report on Form 8-K will be filed to provide updated financial information to reflect the exercise of the underwriters' over-allotment option.

Common Stock

Upon the closing of this offering, 12,500,000 shares of our common stock will be outstanding (assuming no exercise of the underwriters' overallotment option and the corresponding forfeiture of 375,000 founder shares by our sponsor), consisting of:

- 10,000,000 shares of our Class A common stock underlying the units being offered in this offering; and
- 2,500,000 shares of Class B common stock held by our initial stockholders.

The initial stockholders will hold an aggregate of 20% of the issued and outstanding common stock following the offering and the expiration of the underwriters' over-allotment option (assuming they do not purchase any units in this offering or the public market and excluding the representative's shares). If we increase or decrease the size of the offering we will effect a stock dividend or a share contribution back to capital or other appropriate mechanism, as applicable, with respect to our Class B common stock immediately prior to the consummation of the offering in such amount as to maintain the ownership of our initial stockholders at 20.0% of the issued and outstanding shares of our common stock (excluding the placement warrants and the underlying securities, and the representative's shares and assuming they do not purchase any units in this offering) upon the consummation of this offering.

Common stockholders of record are entitled to one vote for each share held on all matters to be voted on by stockholders. Holders of the Class A common stock and holders of the Class B common stock will vote together as a single class on all matters submitted to a vote of our stockholders, except as required by law. Unless specified in our amended and restated certificate of incorporation or bylaws, or as required by applicable provisions of the DGCL or applicable stock exchange rules, the affirmative vote of a majority of our shares of common stock that are voted is required to approve any such matter voted on by our stockholders. Our board of directors will be divided into three classes, each of which will generally serve for a term of three years with only one class of directors being elected in each year. There is no cumulative voting with respect to the election of directors, with the result that the holders of more than 50% of the shares voted for the election of directors can elect all of the directors. Our stockholders are entitled to receive ratable dividends when, as and if declared by the board of directors out of funds legally available therefor.

Because our amended and restated certificate of incorporation authorizes the issuance of up to 125,000,000 shares of Class A common stock, if we were to enter into an initial business combination, we may (depending on the terms of such an initial business combination) be required to increase the number of shares of Class A common stock which we are authorized to issue at the same time as our stockholders vote on the initial business combination to the extent we seek stockholder approval in connection with our initial business combination.

In accordance with Nasdaq corporate governance requirements, we are not required to hold an annual meeting until no later than one year after our first fiscal year end following our listing on Nasdaq. Under Section 211(b) of the DGCL, we are, however, required to hold an annual meeting of stockholders for the purposes of electing directors in accordance with our bylaws, unless such election is made by written consent in lieu of such a meeting. We may not hold an annual meeting of stockholders to elect new directors prior to the consummation of our initial business combination, and thus we may not be in compliance with Section 211(b) of the DGCL, which requires an annual meeting. Therefore, if our stockholders want us to hold an annual meeting prior to the consummation of our initial business combination, they may attempt to force us to hold one by submitting an application to the Delaware Court of Chancery in accordance with Section 211(c) of the DGCL.

We will provide our stockholders with the opportunity to redeem all or a portion of their public shares upon the completion of our initial business combination at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the trust account as of two business days prior to the consummation of our initial business combination including interest earned on the funds held in the trust account and not previously released to us to pay our taxes, divided by the number of then outstanding public shares, subject to the limitations described herein. The amount in the trust account is initially anticipated to be approximately \$10.20 per public share. The per-share amount we will distribute to investors who properly redeem their shares will not be reduced by the deferred underwriting commissions we will pay to the underwriters. Our sponsor, officers and directors, and our director nominees, will enter into a letter agreement with us, pursuant to which they will agree to waive their redemption rights with respect to any founder shares and any public shares held by them in connection with the completion of our initial business combination. Unlike many blank check companies that hold stockholder votes and conduct proxy solicitations in conjunction with their initial business combinations and provide for related redemptions of public shares for cash upon completion of such initial business combinations even when a vote is not required by applicable law or stock exchange requirements, if a stockholder vote is not required by law and we do not decide to hold a stockholder vote for business or other legal reasons, we will, pursuant to our amended and restated certificate of incorporation, conduct the redemptions pursuant to the tender offer rules of the SEC, and file tender offer documents with the SEC prior to completing our initial business combination. Our amended and restated certificate of incorporation will require these tender offer documents to contain substantially the same financial and other information about the initial business combination and the redemption rights as is required under the SEC's proxy rules. If, however, a stockholder approval of the transaction is required by applicable law or stock exchange requirements, or we decide to obtain stockholder approval for business or other legal reasons, we will, like many blank check companies, offer to redeem shares in conjunction with a proxy solicitation pursuant to the proxy rules and not pursuant to the tender offer rules. If we seek stockholder approval, we will complete our initial business combination only if a majority of the outstanding shares of common stock voted are voted in favor of the initial business combination. A quorum for such meeting will consist of the holders present in person or by proxy of shares of outstanding capital stock of the company representing a majority of the voting power of all outstanding shares of capital stock of the company entitled to vote at such meeting. The underwriters will have the same redemption rights as a public stockholder with respect to any public shares it acquires. The representative has informed us that it has no current commitments, plans or intentions to acquire any public shares for its own account; however, if they do acquire public shares, it will do so in the ordinary course of business or in the types of transaction described in the first paragraph under "Proposed Business — Effecting our Initial Business Combination — Permitted purchases of our securities." The underwriters will not make any such purchases when in possession of any material nonpublic information not disclosed to the seller, during a restricted period under Regulation M under the Exchange Act, in transactions that would violate Section 9(a)(2) or Rule 10(b)-5 under the Exchange Act, or if prohibited by applicable state securities laws or broker-dealer regulations. To the extent our initial stockholders or purchasers of placement warrants transfer any of these securities to certain permitted transferees, such permitted transferees will agree, as a condition to such transfer, to waive these same redemption rights. If we submit our initial business combination to our public stockholders for a vote, our sponsor, the other initial stockholders, our officers and our directors have agreed to vote and our director nominees will agree to vote, their respective founder shares and any public shares held by them in favor of our initial business combination.

The participation of our sponsor, officers, directors or their affiliates in privately-negotiated transactions (as described in this prospectus), if any, could result in the approval of our initial business combination even if a majority of our public stockholders vote, or indicate their intention to vote, against such business combination. For purposes of seeking approval of the majority of our outstanding shares of common stock voted, non-votes will have no effect on the approval of our initial business combination once a quorum is obtained. We intend to give approximately 30 days (but not less than 10 days nor more than 60 days) prior written notice of any such meeting, if required, at which a vote shall be taken to approve our initial business combination. These quorum and voting thresholds, and the voting agreements of our initial stockholders, may make it more likely that we will consummate our initial business combination.

If we seek stockholder approval of our initial business combination and we do not conduct redemptions in connection with our initial business combination pursuant to the tender offer rules, our amended and restated certificate of incorporation will provide that a public stockholder, together with any affiliate of such stockholder or any other person with whom such stockholder is acting in concert or as a "group" (as defined under Section 13 of the Exchange Act), will be restricted from redeeming its shares with respect to more than an aggregate of 15% of the shares of common stock sold in this offering, which we refer to as the Excess Shares. However, we would not be restricting our stockholders' ability to vote all of their shares (including Excess Shares) for or against our initial business combination. Our stockholders' inability to redeem the Excess Shares will reduce their influence over our ability to complete our initial business combination, and such stockholders could suffer a material loss in their investment if they sell such Excess Shares on the open market. Additionally, such stockholders will not receive redemption distributions with respect to the Excess Shares if we complete the initial business combination. And, as a result, such stockholders will continue to hold that number of shares exceeding 15% and, in order to dispose such shares would be required to sell their stock in open market transactions, potentially at a loss.

If we seek stockholder approval in connection with our initial business combination, pursuant to the letter agreement our sponsor, officers and directors have agreed to vote and our director nominees will agree to vote, any founder shares held by them and any public shares they may acquire during or after this offering (including in open market and privately negotiated transactions) in favor of our initial business combination. As a result, in addition to our initial stockholders' founder shares, we would need 550,001 or 5.5%, of the 10,000,000 public shares sold in this offering to be voted in favor of an initial business combination (assuming only the minimum number of shares representing a quorum are voted) in order to have our initial business combination approved (assuming the over-allotment option is not exercised, that the initial stockholders do not purchase any units in this offering or units or shares in the after-market). In the event that all shares of our outstanding common stock are voted, we would need 3,700,001 or 37.0%, of the 10,000,000 public shares sold in this offering to be voted in favor of an initial business combination in order to have our initial business combination approved (assuming the over-allotment option is not exercised, and that the initial stockholders do not purchase any units in this offering or units or shares in the after-market). Additionally, each public stockholder may elect to redeem its public shares irrespective of whether they vote for or against the proposed transaction (or abstain from voting) (subject to the limitation described in the preceding paragraph).

Pursuant to our amended and restated certificate of incorporation, if we are unable to complete our initial business combination within 12 months from the closing of this offering or during any Extension Period, we will (i) cease all operations except for the purpose of winding up, (ii) as promptly as reasonably possible but no more than ten business days thereafter subject to lawfully available funds therefor, redeem the public shares, at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the trust account including interest earned on the funds held in the trust account and not previously released to us to pay our taxes (less up to \$100,000 of interest to pay dissolution expenses), divided by the number of then outstanding public shares, which redemption will completely extinguish public stockholders' rights as stockholders (including the right to receive further liquidating distributions, if any), subject to applicable law, and (iii) as promptly as reasonably possible following such redemption, subject to the approval of our remaining stockholders and our board of directors, dissolve and liquidate, subject in the case of clauses (ii) and (iii) above to our obligations under Delaware law to provide for claims of creditors and the requirements of other applicable law. Our sponsor, officers and directors, and our director nominees, will enter into a letter agreement with us, pursuant to which they will agree to waive their rights to liquidating distributions from the trust account with respect to any founder shares held by them if we fail to complete our initial business combination within 12 months from the closing of this offering or during any Extension Period. However, if our initial stockholders acquire public shares in or after this offering, they will be entitled to liquidating distributions from the trust account with respect to such public shares if we fail to complete our initial business combination within the prescribed time period.

In the event of a liquidation, dissolution or winding up of the company after an initial business combination, our stockholders are entitled to share ratably in all assets remaining available for distribution to them after payment of liabilities and after provision is made for each class of stock, if any, having preference over the common stock. Our stockholders have no preemptive or other subscription rights. There are no sinking fund provisions applicable to the common stock, except that we will provide our stockholders with the opportunity to redeem their public shares for cash equal to their pro rata share of the aggregate amount then on deposit in the trust account, upon the completion of our initial business combination, subject to the limitations described herein.

Founder Shares

The founder shares are identical to the shares of Class A common stock included in the units being sold in this offering, and holders of founder shares have the same stockholder rights as public stockholders, except that (i) the founder shares are subject to certain transfer restrictions, as described in more detail below, (ii) our sponsor, officers and directors have entered into a letter agreement with us, and we anticipate that our director nominees will agree to enter into with us, a letter agreement, pursuant to which they have agreed (A) to waive their redemption rights with respect to any founder shares and any public shares held by them in connection with the completion of our initial business combination, (B) to waive their redemption rights with respect to their founder shares and any public shares in connection with a stockholder vote to approve an amendment to our amended and restated certificate of incorporation (x) to modify the substance or timing of our obligation to allow redemption in connection with our initial business combination or certain amendments to our charter prior thereto or to redeem 100% of our public shares if we do not complete our initial business combination within 12 months from the closing of this offering or during any Extension Period or (y) with respect to any other provision relating to stockholders' rights or pre-initial business combination activity and (C) to waive their rights to liquidating distributions from the trust account with respect to any founder shares held by them if we fail to complete our initial business combination within 12 months from the closing of this offering or during any Extension Period, although they will be entitled to liquidating distributions from the trust account with respect to any public shares they hold if we fail to complete our initial business combination within such time period, (iii) the founder shares are shares of our Class B common stock that will automatically convert into shares of our Class A common stock at the time of the consummation of our initial business combination, on a one-for-one basis, subject to adjustment as described herein, and (iv) are entitled to registration rights. If we submit our initial business combination to our public stockholders for a vote, our sponsor, officers and directors, and our director nominees, have agreed pursuant to the letter agreement to vote any founder shares held by them and any public shares purchased during or after this offering (including in open market and privately negotiated transactions) in favor of our initial business combination.

The shares of Class B common stock will automatically convert into shares of Class A common stock at the time of the consummation of our initial business combination on a one-for-one basis (subject to adjustment for stock splits, stock dividends, reorganizations, recapitalizations and the like), and subject to further adjustment as provided herein. In the case that additional shares of Class A common stock, or equity-linked securities, are issued or deemed issued in excess of the amounts offered in this prospectus and related to the closing of the initial business combination, the ratio at which shares of Class B common stock shall convert into shares of Class A common stock will be adjusted (unless the holders of a majority of the outstanding shares of Class B common stock agree to waive such adjustment with respect to any such issuance or deemed issuance) so that the number of shares of Class A common stock issuable upon conversion of all shares of Class B common stock will equal, in the aggregate, on an as-converted basis, 20% of the sum of the total number of all shares of common stock outstanding upon completion of this offering (excluding the placement warrants and underlying securities, and the representative's shares) plus all shares of Class A common stock and equity-linked securities issued or deemed issued in connection with the initial business combination (excluding any shares or equity-linked securities issued, or to be issued, to any seller in the initial business combination, any private placement-equivalent units and their underlying securities issued to our sponsor or its affiliates upon conversion of loans made to us). We cannot determine at this time whether a majority of the holders of our Class B common stock at the time of any future issuance would agree to waive such adjustment to the conversion ratio. They may waive such adjustment due to (but not limited to) the following: (i) closing conditions which are part of the agreement for our initial business combination; (ii) negotiation with Class A stockholders on structuring an initial business combination; or (iii) negotiation with parties providing financing which would trigger the anti-dilution provisions of the Class B common stock. If such adjustment is not waived, the issuance would not reduce the percentage ownership of holders of our Class B common stock, but would reduce the percentage ownership of holders of our Class A common stock. If such adjustment is waived, the issuance would reduce the percentage ownership of holders of both classes of our common stock. The term "equity-linked securities" refers to any debt or equity securities that are convertible, exercisable or exchangeable for shares of Class A common stock issues in a financing transaction in connection with our initial business combination, including but not limited to a private placement of equity or debt. Securities could be "deemed issued" for purposes of the conversion rate adjustment if such shares are issuable upon the conversion or exercise of convertible securities, warrants or similar securities.

With certain limited exceptions, the founder shares are not transferable, assignable or saleable (except to our officers, directors, consultants and other persons or entities affiliated with our sponsor, each of whom will be subject to the same transfer restrictions) until the earlier to occur of: (A) one year after the completion of our initial business combination and (B) subsequent to our initial business combination, (x) if the reported last sale price of our Class A common stock equals or exceeds \$12.00 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like) for any 20 trading days within any 30-trading day period commencing at least 150 days after our initial business combination, or (y) the date on which we complete a liquidation, merger, capital stock exchange, reorganization or other similar transaction that results in all of our stockholders having the right to exchange their shares of common stock for cash, securities or other property.

Preferred Stock

Our amended and restated certificate of incorporation will provide that shares of preferred stock may be issued from time to time in one or more series. Our board of directors will be authorized to fix the voting rights, if any, designations, powers, preferences, the relative, participating, optional or other special rights and any qualifications, limitations and restrictions thereof, applicable to the shares of each series. Our board of directors will be able to, without stockholder approval, issue preferred stock with voting and other rights that could adversely affect the voting power and other rights of the holders of the common stock and could have anti-takeover effects. The ability of our board of directors to issue preferred stock without stockholder approval could have the effect of delaying, deferring or preventing a change of control of us or the removal of existing management. We have no preferred stock outstanding at the date hereof. Although we do not currently intend to issue any shares of preferred stock, we cannot assure you that we will not do so in the future. No shares of preferred stock are being issued or registered in this offering.

Redeemable Warrants

Public Stockholders' Warrants

Each whole warrant entitles the registered holder to purchase one share of our Class A common stock at a price of \$11.50 per share, subject to adjustment as discussed below, at any time commencing on the later of 12 months from the closing of this offering and 30 days after the completion of our initial business combination. Pursuant to the warrant agreement, a warrant holder may exercise its warrants only for a whole number of shares of Class A common stock. This means that only a whole warrant may be exercised at any given time by a warrant holder. No fractional warrants will be issued upon separation of the units and only whole warrants will trade. Accordingly, unless you purchase at least two units, you will not be able to receive or trade a whole warrant.

The warrants will expire five years after the completion of our initial business combination, at 5:00 p.m., New York City time, or earlier upon redemption or liquidation.

We will not be obligated to deliver any shares of Class A common stock pursuant to the exercise of a warrant and will have no obligation to settle such warrant exercise unless a registration statement under the Securities Act with respect to the shares of Class A common stock underlying the warrants is then effective and a prospectus relating thereto is current, subject to our satisfying our obligations described below with respect to registration. No warrant will be exercisable and we will not be obligated to issue shares of Class A common stock upon exercise of a warrant unless Class A common stock issuable upon such warrant exercise has been registered, qualified or deemed to be exempt under the securities laws of the state of residence of the registered holder of the warrants. In the event that the conditions in the two immediately preceding sentences are not satisfied with respect to a warrant, the holder of such warrant will not be entitled to exercise such warrant and such warrant may have no value and expire worthless. In no event will we be required to net cash settle any warrant. In the event that a registration statement is not effective for the exercised warrants, the purchaser of a unit containing such warrant will have paid the full purchase price for the unit solely for the share of Class A common stock underlying such unit.

We are not registering the shares of Class A common stock issuable upon exercise of the warrants at this time. However, we have agreed that as soon as practicable after the closing of our initial business combination, we will use our best efforts to file with the SEC a registration statement covering the shares of Class A common stock issuable upon exercise of the warrants, to cause such registration statement to become effective and to maintain a current prospectus relating to those shares of Class A common stock until the warrants expire or are redeemed, as specified in the warrant agreement. If a registration statement covering the shares of Class A common stock issuable upon exercise of the warrants is not effective by the 90th day after the closing of our initial business combination, warrant holders may, until such time as there is an effective registration statement and during any period when we will have failed to maintain an effective registration following the consummation of our initial business combination, warrant holders may, until such time as there is an effective within a specified period following the consummation of our initial business combination, warrant holders may, until such time as there is an effective registration statement and during any period when we shall have failed to maintain an effective registration statement, exercise warrants on a cashless basis pursuant to the exemption provided by Section 3(a)(9) of the Securities Act, provided that such exemption is available. If that exemption, or another exemption, is not available, holders will not be able to exercise their warrants on a cashless basis.

Once the warrants become exercisable, we may call the warrants for redemption:

- in whole and not in part;
- at a price of \$0.01 per warrant;
- upon not less than 30 days' prior written notice of redemption given after the warrants become exercisable (the "30-day redemption period") to each warrant holder; and
- if, and only if, the reported last sale price of the Class A common stock equals or exceeds \$18.00 per share (as adjusted for stock splits, stock dividends, right issuances, reorganizations, recapitalizations and the like and for certain capital raising transactions as discussed below) for any 20 trading days within a 30-trading day period commencing once the warrants become exercisable and ending three business days before we send the notice of redemption to the warrant holders.

If and when the warrants become redeemable by us, we may not exercise our redemption right if the issuance of shares of common stock upon exercise of the warrants is not exempt from registration or qualification under applicable state blue sky laws or we are unable to effect such registration or qualification. We will use our best efforts to register or qualify such shares of common stock under the blue sky laws of the state of residence in those states in which the warrants were offered by us in this offering.

We have established the last of the redemption criterion discussed above to prevent a redemption call unless there is at the time of the call a significant premium to the warrant exercise price. If the foregoing conditions are satisfied and we issue a notice of redemption of the warrants, each warrant holder will be entitled to exercise its warrant prior to the scheduled redemption date. However, the price of the Class A common stock may fall below the \$18.00 redemption trigger price (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like) as well as the \$11.50 warrant exercise price after the redemption notice is issued.

If we call the warrants for redemption as described above, our management will have the option to require any holder that wishes to exercise its warrant to do so on a "cashless basis." In determining whether to require all holders to exercise their warrants on a "cashless basis," our management will consider, among other factors, our cash position, the number of warrants that are outstanding and the dilutive effect on our stockholders of issuing the maximum number of shares of Class A common stock issuable upon the exercise of our warrants. If our management takes advantage of this option, all holders of warrants would pay the exercise price by surrendering their warrants for that number of shares of Class A common stock equal to the quotient obtained by dividing (x) the product of the number of shares of Class A common stock underlying the warrants, multiplied by the excess of the "fair market value" (defined below) over the exercise price of the warrants by (y) the fair market value. The "fair market value" for this purpose shall mean the average reported last sale price of the Class A common stock for the 10 trading days ending on the third trading day prior to the date on which the notice of redemption is sent to the holders of warrants. If our management takes advantage of this option, the notice of redemption will contain the information necessary to calculate the number of shares of Class A common stock to be received upon exercise of the warrants, including the "fair market value" in such case. Requiring a cashless exercise in this manner will reduce the number of shares to be issued and thereby lessen the dilutive effect of a warrant redemption. We believe this feature is an attractive option to us if we do not need the cash from the exercise of the warrants after our initial business combination.

A holder of a warrant may notify us in writing in the event it elects to be subject to a requirement that such holder will not have the right to exercise such warrant, to the extent that after giving effect to such exercise, such person (together with such person's affiliates), to the warrant agent's actual knowledge, would beneficially own in excess of 4.9% or 9.8% (or such other amount as a holder may specify) of the shares of Class A common stock outstanding immediately after giving effect to such exercise.

If the number of outstanding shares of Class A common stock is increased by a stock dividend payable in shares of Class A common stock, or by a split-up of shares of Class A common stock or other similar event, then, on the effective date of such stock dividend, split-up or similar event, the number of shares of Class A common stock issuable on exercise of each whole warrant will be increased in proportion to such increase in the outstanding shares of Class A common stock. A rights offering to holders of Class A common stock entitling holders to purchase shares of Class A common stock at a price less than the fair market value will be deemed a stock dividend of a number of shares of Class A common stock equal to the product of (i) the number of shares of Class A common stock actually sold in such rights offering (or issuable under any other equity securities sold in such rights offering that are convertible into or exercisable for Class A common stock) and (ii) one (1) minus the quotient of (x) the price per share of Class A common stock paid in such rights offering divided by (y) the fair market value. For these purposes (i) if the rights offering is for securities convertible into or exercisable for Class A common stock, in determining the price payable for Class A common stock, there will be taken into account any consideration received for such rights, as well as any additional amount payable upon exercise or conversion and (ii) fair market value means the volume weighted average price of Class A common stock as reported during the ten (10) trading day period ending on the trading day prior to the first date on which the shares of Class A common stock trade on the applicable exchange or in the applicable market, regular way, without the right to receive such rights.

In addition, if we, at any time while the warrants are outstanding and unexpired, pay a dividend or make a distribution in cash, securities or other assets to the holders of Class A common stock on account of such shares of Class A common stock (or other shares of our capital stock into which the warrants are convertible), other than (a) as described above, (b) certain ordinary cash dividends, (c) to satisfy the redemption rights of the holders of Class A common stock in connection with a proposed initial business combination, (d) to satisfy the redemption rights of the holders of Class A common stock in connection with a stockholder vote to amend our amended and restated certificate of incorporation (i) to modify the substance or timing of our obligation to allow redemption in connection with our initial business combination or certain amendments to our charter prior thereto or to redeem 100% of our Class A common stock if we do not complete our initial business combination within 12 months from the closing of this offering or during any Extension Period or (ii) with respect to any other provision relating to stockholders' rights or pre-initial business combination activity, (e) in connection with the redemption of our public shares upon our failure to complete our initial business combination, or (f) in connection with any payment in connection with the Company's liquidation and the distribution of its assets upon its failure to consummate a initial business combination, then the warrant exercise price will be decreased, effective immediately after the effective date of such event, by the amount of cash and/or the fair market value of any securities or other assets paid on each share of Class A common stock in respect of such event.

If the number of outstanding shares of our Class A common stock is decreased by a consolidation, combination, reverse stock split or reclassification of shares of Class A common stock or other similar event, then, on the effective date of such consolidation, combination, reverse stock split, reclassification or similar event, the number of shares of Class A common stock issuable on exercise of each whole warrant will be decreased in proportion to such decrease in outstanding shares of Class A common stock.

Whenever the number of shares of Class A common stock purchasable upon the exercise of the warrants is adjusted, as described above, the warrant exercise price will be adjusted by multiplying the warrant exercise price immediately prior to such adjustment by a fraction (x) the numerator of which will be the number of shares of Class A common stock purchasable upon the exercise of the warrants immediately prior to such adjustment, and (y) the denominator of which will be the number of shares of Class A common stock so purchasable immediately thereafter.

In case of any reclassification or reorganization of the outstanding shares of Class A common stock (other than those described above or that solely affects the par value of such shares of Class A common stock), or in the case of any merger or consolidation of us with or into another corporation (other than a consolidation or merger in which we are the continuing corporation and that does not result in any reclassification or reorganization of our outstanding shares of Class A common stock), or in the case of any sale or conveyance to another corporation or entity of the assets or other property of us as an entirety or substantially as an entirety in connection with which we are dissolved, the holders of the warrants will thereafter have the right to purchase and receive, upon the basis and upon the terms and conditions specified in the warrants and in lieu of the shares of our Class A common stock immediately theretofore purchasable and receivable upon the exercise of the rights represented thereby, the kind and amount of shares of stock or other securities or property (including cash) receivable upon such reclassification, reorganization, merger or consolidation, or upon a dissolution following any such sale or transfer, that the holder of the warrants would have received if such holder had exercised their warrants immediately prior to such event.

The warrants will be issued in registered form under a warrant agreement between Continental Stock Transfer & Trust Company, LLC, as warrant agent, and us. You should review a copy of the warrant agreement, which will be filed as an exhibit to the registration statement of which this prospectus is a part, for a complete description of the terms and conditions applicable to the warrants. The warrant agreement provides that the terms of the warrants may be amended without the consent of any holder to cure any ambiguity or correct any mistake, including to conform the provisions of the warrant agreement to the description of the terms of the warrants and the warrant agreement set forth in this prospectus, or defective provision, but requires the approval by the holders of at least a majority of the then outstanding public warrants to make any change that adversely affects the interests of the registered holders of public warrants.

In addition, if (x) we issue additional shares of Class A common stock or equity-linked securities for capital raising purposes in connection with the closing of our initial business combination at a Newly Issued Price of less than \$9.20 per share of Class A common stock (with such issue price or effective issue price to be determined in good faith by our board of directors and, in the case of any such issuance to our sponsor or its affiliates, without taking into account any founder shares held by our sponsor or such affiliates, as applicable, prior to such issuance), (y) the aggregate gross proceeds from such issuances represent more than 60% of the total equity proceeds, and interest thereon, available for the funding of our initial business combination on the date of the consummation of our initial business combination (net of redemptions), and (z) the Market Value is below \$9.20 per share, then the exercise price of the warrants will be adjusted (to the nearest cent) to be equal to 115% of the greater of the Market Value and the Newly Issued Price, and the \$18.00 per share redemption trigger price described above will be adjusted (to the nearest cent) to be equal to 180% of the greater of the Market Value and the Newly Issued Price.

The warrants may be exercised upon surrender of the warrant certificate on or prior to the expiration date at the offices of the warrant agent, with the exercise form on the reverse side of the warrant certificate completed and executed as indicated, accompanied by full payment of the exercise price (or on a cashless basis, if applicable), by certified or official bank check payable to us, for the number of warrants being exercised. The warrant holders do not have the rights or privileges of holders of Class A common stock and any voting rights until they exercise their warrants and receive shares of Class A common stock. After the issuance of shares of Class A common stock upon exercise of the warrants, each holder will be entitled to one (1) vote for each share held of record on all matters to be voted on by stockholders.

No fractional shares will be issued upon exercise of the warrants. If, upon exercise of the warrants, a holder would be entitled to receive a fractional interest in a share, we will, upon exercise, round up to the nearest whole number of shares of Class A common stock to be issued to the warrant holder.

We have agreed that, subject to applicable law, any action, proceeding or claim against us arising out of or relating in any way to the warrant agreement will be brought and enforced in the courts of the State of New York or the United States District Court for the Southern District of New York, and we irrevocably submit to such jurisdiction, which jurisdiction will be the exclusive forum for any such action, proceeding or claim. See "Risk Factors — Our warrant agreement will designate the courts of the State of New York or the United States District Court for the Southern District of New York as the sole and exclusive forum for certain types of actions and proceedings that may be initiated by holders of our warrants, which could limit the ability of warrant holders to obtain a favorable judicial forum for disputes with our company." This provision applies to claims under the Securities Act but does not apply to claims under the Exchange Act or any claim for which the federal district courts of the United States of America are the sole and exclusive forum.

Placement warrants

Except as described below, the placement warrants have terms and provisions that are identical to those of the warrants being sold as part of the units in this offering, including as to exercise price, exercisability, redemption, and exercise period. The placement warrants (including the Class A common stock issuable upon exercise of the placement warrants) will not be transferable, assignable or salable until 30 days after the completion of our initial business combination (except, among other limited exceptions as described under the section of this prospectus entitled "Principal Stockholders — Restrictions on Transfers of Founder Shares and Placement warrants," including to our officers, directors, consultants and other persons or entities affiliated with our sponsor).

The private placement warrants will be identical to the warrants sold in this offering except that the private placement warrants, so long as they are held by our sponsor, or its permitted transferees, (i) may not (including the common stock shares issuable upon exercise of these warrants), subject to certain limited exceptions, be transferred, assigned or sold by the holders until 30 days after the completion of our initial business combination, and (ii) will be entitled to registration rights. The private placement warrants (including the common stock issuable upon exercise thereof) may not, subject to certain limited exceptions, be transferred, assigned or sold by the holder. The purchase price of the private placement warrants will be added to the proceeds from this offering to pay for the expenses of this offering and to be held in the trust account. If we do not complete our initial business combination within 12 months (or during any Extension Period) from the closing of this offering, the proceeds of the sale of the private placement warrants will be used to fund the redemption of our public shares (subject to the requirements of applicable law) and the private placement warrants will be worthless.

In order to finance transaction costs in connection with an intended initial business combination, our sponsor or an affiliate of our sponsor or certain of our officers and directors may, but are not obligated to, loan us funds as may be required. Up to \$1,500,000 of such loans may be convertible into warrants, at a price of \$1.00 per warrant at the option of the lender, upon consummation of our initial business combination. The warrants would be identical to the placement warrants. However, as the warrants would not be issued until consummation of our initial business combination, such warrants would not be able to be voted on an amendment to the warrant agreement in connection with such business combination.

Our sponsor has agreed not to transfer, assign or sell any of the placement warrants (including the Class A common stock issuable upon exercise of any of these warrants) until the date that is 30 days after the date we complete our initial business combination, except that, among other limited exceptions as described under the section of this prospectus entitled "Principal Stockholders — Restrictions on Transfers of Founder Shares and Placement Warrants", to our officers, directors, consultants and other persons or entities affiliated with our sponsor.

Dividends

We have not paid any cash dividends on our common stock to date and do not intend to pay cash dividends prior to the completion of an initial business combination. The payment of cash dividends in the future will be dependent upon our revenues and earnings, if any, capital requirements and general financial conditions subsequent to completion of an initial business combination. The payment of any cash dividends subsequent to an initial business combination will be within the discretion of our board of directors at such time. If we increase or decrease the size of the offering we will effect a stock dividend or a share contribution back to capital or other appropriate mechanism, as applicable, with respect to our Class B common stock immediately prior to the consummation of the offering in such amount as to maintain the ownership of our initial stockholders at 20.0% of the issued and outstanding shares of our common stock (excluding the placement warrants and the underlying securities, and the representative's shares and assuming they do not purchase any units in this offering) upon the consummation of this offering. Further, if we incur any indebtedness, our ability to declare dividends may be limited by restrictive covenants we may agree to in connection therewith.

Our Transfer Agent and Warrant Agent

The transfer agent for our common stock and warrant agent for our warrants is Continental Stock Transfer & Trust Company, LLC We have agreed to indemnify Continental Stock Transfer & Trust Company, LLC in its roles as transfer agent and warrant agent, its agents and each of its stockholders, directors, officers and employees against all claims and losses that may arise out of acts performed or omitted for its activities in that capacity, except for any liability due to any gross negligence, willful misconduct or bad faith of the indemnified person or entity.

Our Amended and Restated Certificate of Incorporation

Our amended and restated certificate of incorporation will contain certain requirements and restrictions relating to this offering that will apply to us until the completion of our initial business combination. These provisions cannot be amended without the approval of the holders of at least 65% of our common stock. Our initial stockholders, who will collectively beneficially own 20% of our common stock upon the closing of this offering (assuming they do not purchase any units in this offering and excluding the representative's shares), will participate in any vote to amend our amended and restated certificate of incorporation and will have the discretion to vote in any manner they choose. Specifically, our amended and restated certificate of incorporation provides, among other things, that:

- If we are unable to complete our initial business combination within 12 months from the closing of this offering or during any Extension Period, we will (i) cease all operations except for the purpose of winding up, (ii) as promptly as reasonably possible but not more than ten business days thereafter subject to lawfully available funds therefor, redeem 100% of the public shares, at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the trust account including interest earned on the funds held in the trust account and not previously released to us to pay our taxes (less up to \$100,000 of interest to pay dissolution expenses), divided by the number of then outstanding public shares, which redemption will completely extinguish public stockholders' rights as stockholders (including the right to receive further liquidating distributions, if any), subject to applicable law, and (iii) as promptly as reasonably possible following such redemption, subject to the approval of our remaining stockholders and our board of directors, dissolve and liquidate, subject in the case of clauses (ii) and (iii) above to our obligations under Delaware law to provide for claims of creditors and the requirements of other applicable law:
- Prior to our initial business combination, we may not issue additional shares of capital stock that would entitle the holders thereof to (i) receive funds from the trust account or (ii) vote on any initial business combination;
- Although we do not intend to enter into an initial business combination with a target business that is affiliated with our sponsor, our directors or our officers, we are not prohibited from doing so. In the event we enter into such a transaction, we, or a committee of independent directors, will obtain an opinion from an independent investment banking firm or another independent entity that commonly renders valuation opinions that such an initial business combination is fair to our company from a financial point of view;
- If a stockholder vote on our initial business combination is not required by law and we do not decide to hold a stockholder vote for business or other legal reasons, we will offer to redeem our public shares pursuant to Rule 13e-4 and Regulation 14E of the Exchange Act, and will file tender offer documents with the SEC prior to completing our initial business combination which contain substantially the same financial and other information about our initial business combination and the redemption rights as is required under Regulation 14A of the Exchange Act; whether or not we maintain our registration under the Exchange Act or our listing on Nasdaq, we will provide our public stockholders with the opportunity to redeem their public shares by one of the two methods listed above;
- So long as we obtain and maintain a listing for our securities on Nasdaq, Nasdaq rules require that we must complete one or more business combinations having an aggregate fair market value of at least 80% of the value of the assets held in the trust account (excluding the deferred underwriting commissions and taxes payable on the interest earned on the trust account) at the time of our signing a definitive agreement in connection with our initial business combination;
- If our stockholders approve an amendment to our amended and restated certificate of incorporation (i) to modify the substance or timing of our obligation to allow redemption in connection with our initial business combination or certain amendments to our charter prior thereto or to redeem 100% of our public shares if we do not complete our initial business combination within 12 months from the closing of this offering or during any Extension Period or (ii) with respect to any other provision relating to stockholders' rights or pre-business combination activity, we will provide our public stockholders with the opportunity to redeem all or a portion of their shares of Class A common stock upon such approval at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the trust account, including interest earned on the funds held in the trust account and not previously released to us to pay our taxes, divided by the number of then outstanding public shares; and
- We will not effectuate our initial business combination with another blank check company or a similar company with nominal operations.

In addition, our amended and restated certificate of incorporation will provide that under no circumstances will we redeem our public shares unless our net tangible assets are at least \$5,000,001 either immediately prior to or upon consummation of our initial business combination and after payment of underwriters' fees and commissions.

Certain Anti-Takeover Provisions of Delaware Law and our Amended and Restated Certificate of Incorporation and Bylaws

We will be subject to the provisions of Section 203 of the DGCL regulating corporate takeovers upon completion of this offering. This statute prevents certain Delaware corporations, under certain circumstances, from engaging in a "business combination" with:

- a stockholder who owns 15% or more of our outstanding voting stock (otherwise known as an "interested stockholder");
- an affiliate of an interested stockholder; or
- an associate of an interested stockholder, for three years following the date that the stockholder became an interested stockholder.

A "business combination" includes a merger or sale of more than 10% of our assets. However, the above provisions of Section 203 do not apply if:

- our board of directors approves the transaction that made the stockholder an "interested stockholder," prior to the date of the transaction;
- after the completion of the transaction that resulted in the stockholder becoming an interested stockholder, that stockholder owned at least 85% of our voting stock outstanding at the time the transaction commenced, other than statutorily excluded shares of common stock; or
- on or subsequent to the date of the transaction, the initial business combination is approved by our board of directors and authorized at a meeting of our stockholders, and not by written consent, by an affirmative vote of at least two-thirds of the outstanding voting stock not owned by the interested stockholder.

Our amended and restated certificate of incorporation will provide that our board of directors will be classified into three classes of directors. As a result, in most circumstances, a person can gain control of our board only by successfully engaging in a proxy contest at two or more annual meetings.

Our authorized but unissued common stock and preferred stock are available for future issuances without stockholder approval and could be utilized for a variety of corporate purposes, including future offerings to raise additional capital, acquisitions and employee benefit plans. The existence of authorized but unissued and unreserved common stock and preferred stock could render more difficult or discourage an attempt to obtain control of us by means of a proxy contest, tender offer, merger or otherwise.

Exclusive forum for certain lawsuits

Our amended and restated certificate of incorporation will require, to the fullest extent permitted by law, that derivative actions brought in our name, actions against directors, officers and employees for breach of fiduciary duty and certain other actions may be brought only in the Court of Chancery in the State of Delaware, except any action (A) as to which the Court of Chancery in the State of Delaware determines that there is an indispensable party not subject to the jurisdiction of the Court of Chancery (and the indispensable party does not consent to the personal jurisdiction of the Court of Chancery within ten days following such determination), (B) which is vested in the exclusive jurisdiction of a court or forum other than the Court of Chancery or (C) for which the Court of Chancery does not have subject matter jurisdiction. If an action is brought outside of Delaware, the stockholder bringing the suit will be deemed to have consented to service of process on such stockholder's counsel. Although we believe this provision benefits us by providing increased consistency in the application of law in the types of lawsuits to which it applies, a court may determine that this provision is unenforceable, and to the extent it is enforceable, the provision may have the effect of discouraging lawsuits against our directors and officers.

Our amended and restated certificate of incorporation will provide that the exclusive forum provision will be applicable to the fullest extent permitted by applicable law, subject to certain exceptions. Section 27 of the Exchange Act creates exclusive federal jurisdiction over all suits brought to enforce any duty or liability created by the Exchange Act or the rules and regulations thereunder. As a result, the exclusive forum provision will not apply to suits brought to enforce any duty or liability created by the Exchange Act or any other claim for which the federal courts have exclusive jurisdiction. In addition, our amended and restated certificate of incorporation provides that, unless we consent in writing to the selection of an alternative forum, the federal district courts of the United States of America shall, to the fullest extent permitted by law, be the exclusive forum for the resolution of any complaint asserting a cause of action arising under the Securities Act, or the rules and regulations promulgated thereunder. We note, however, that there is uncertainty as to whether a court would enforce this provision and that investors cannot waive compliance with the federal securities laws and the rules and regulations thereunder. Section 22 of the Securities Act creates concurrent jurisdiction for state and federal courts over all suits brought to enforce any duty or liability created by the Securities Act or the rules and regulations thereunder.

Special meeting of stockholders

Our bylaws provide that special meetings of our stockholders may be called only by a majority vote of our board of directors, by our Chief Executive Officer or by our Chairman.

Advance notice requirements for stockholder proposals and director nominations

Our bylaws provide that stockholders seeking to bring business before our annual meeting of stockholders, or to nominate candidates for election as directors at our annual meeting of stockholders, must provide timely notice of their intent in writing. To be timely, a stockholder's notice will need to be received by the company secretary at our principal executive offices not later than the close of business on the 90th day nor earlier than the opening of business on the 120th day prior to the anniversary date of the immediately preceding annual meeting of stockholders. Pursuant to Rule 14a-8 of the Exchange Act, proposals seeking inclusion in our annual proxy statement must comply with the notice periods contained therein. Our bylaws also specify certain requirements as to the form and content of a stockholders' meeting. These provisions may preclude our stockholders from bringing matters before our annual meeting of stockholders or from making nominations for directors at our annual meeting of stockholders.

Action by written consent

Subsequent to the consummation of the offering, any action required or permitted to be taken by our common stockholders must be effected by a duly called annual or special meeting of such stockholders and may not be effected by written consent of the stockholders other than with respect to our Class B common stock.

Class B Common Stock Consent Right

For so long as any shares of Class B common stock remain outstanding, we may not, without the prior vote or written consent of the holders of a majority of the shares of Class B common stock then outstanding, voting separately as a single class, amend, alter or repeal any provision our certificate of incorporation, whether by merger, consolidation or otherwise, if such amendment, alteration or repeal would alter or change the powers, preferences or relative, participating, optional or other or special rights of the Class B common stock. Any action required or permitted to be taken at any meeting of the holders of Class B common stock may be taken without a meeting, without prior notice and without a vote, if a consent or consents in writing, setting forth the action so taken, shall be signed by the holders of the outstanding Class B common stock having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares of Class B common stock were present and voted.

Securities Eligible for Future Sale

Immediately after the consummation of this offering (assuming no exercise of the underwriters' over-allotment option) we will have 12,600,000 (or 14,475,000 if the underwriters' over-allotment option is exercised in full) shares of common stock outstanding. Of these shares, the 10,000,000 shares (or 11,500,000 if the underwriters' over-allotment option is exercised in full) sold in this offering will be freely tradable without restriction or further registration under the Securities Act, except for any shares purchased by one of our affiliates within the meaning of Rule 144 under the Securities Act. All of the remaining 2,500,000 (or 2,875,000 if the underwriters' over-allotment option is exercised in full) founder shares, all 100,000 shares which we have agreed to issue to the representative of the underwriters, and all 5,261,000 placement warrants (or 5,711,000 if the underwriter's over-allotment option is exercised in full) (and shares issuable upon exercise thereof), are restricted securities under Rule 144, in that they were issued in private transactions not involving a public offering, and the shares of Class B common stock and placement warrants are subject to transfer restrictions as set forth elsewhere in this prospectus. These restricted securities will be entitled to registration rights as more fully described below under "— Registration Rights."

Rule 144

Pursuant to Rule 144, a person who has beneficially owned restricted shares of our common stock or warrants for at least six months would be entitled to sell their securities provided that (i) such person is not deemed to have been one of our affiliates at the time of, or at any time during the three months preceding, a sale and (ii) we are subject to the Exchange Act periodic reporting requirements for at least three months before the sale and have filed all required reports under Section 13 or 15(d) of the Exchange Act during the 12 months (or such shorter period as we were required to file reports) preceding the sale.

Persons who have beneficially owned restricted shares of our common stock or warrants for at least six months but who are our affiliates at the time of, or at any time during the three months preceding, a sale, would be subject to additional restrictions, by which such person would be entitled to sell within any three-month period only a number of securities that does not exceed the greater of:

- 1% of the total number of shares of Class A common stock then outstanding, which will equal 125,000 shares immediately after this offering (or 143,750) if the underwriters exercise their over-allotment option in full); or
- the average weekly reported trading volume of the common stock during the four calendar weeks preceding the filing of a notice on Form 144 with respect to the sale.

Sales by our affiliates under Rule 144 are also limited by manner of sale provisions and notice requirements and to the availability of current public information about us.

Restrictions on the Use of Rule 144 by Shell Companies or Former Shell Companies

Rule 144 is not available for the resale of securities initially issued by shell companies (other than business combination related shell companies) or issuers that have been at any time previously a shell company. However, Rule 144 also includes an important exception to this prohibition if the following conditions are met:

the issuer of the securities that was formerly a shell company has ceased to be a shell company;

- the issuer of the securities is subject to the reporting requirements of Section 13 or 15(d) of the Exchange Act;
- the issuer of the securities has filed all Exchange Act reports and materials required to be filed, as applicable, during the preceding 12 months (or such shorter period that the issuer was required to file such reports and materials), other than Current Reports on Form 8-K; and
- at least one year has elapsed from the time that the issuer filed current Form 10 type information with the SEC reflecting its status as an entity that is not a shell company.

As a result, our initial stockholders will be able to sell their founder shares and placement warrants (and shares issuable upon exercise thereof), as applicable, pursuant to Rule 144 without registration one year after we have completed our initial business combination.

Registration Rights

The holders of the founder shares, placement warrants, and warrants that may be issued upon conversion of working capital loans, any shares of Class A common stock issuable upon the exercise of the placement warrants, and any shares of Class A common stock issuable upon the exercise of the warrants issued as part of the working capital loans and Class A common stock issuable upon conversion of the founder shares, will be entitled to registration rights pursuant to a registration rights agreement to be signed prior to or on the effective date of this offering, requiring us to register such securities for resale (in the case of the founder shares, only after conversion to our Class A common stock). The holders of the majority of these securities are entitled to make up to three demands, excluding short form demands, that we register such securities. In addition, the holders have certain "piggy-back" registration rights with respect to registration statements filed subsequent to our completion of our initial business combination and rights to require us to register for resale such securities pursuant to Rule 415 under the Securities Act. The registration rights agreement does not contain liquidated damages or other cash settlement provisions resulting from delays in registering our securities. We will bear the expenses incurred in connection with the filing of any such registration statements.

Listing of Securities

We have applied to list our units, Class A common stock and warrants on Nasdaq under the symbols "AEHAU," "AEHA" and "AEHAW," respectively. We expect that our units will be listed on Nasdaq on or promptly after the effective date of the registration statement. Following the date the shares of our Class A common stock and warrants are eligible to trade separately, we anticipate that the shares of our Class A common stock and warrants will be listed separately and as a unit on Nasdaq. We cannot guarantee that our securities will be approved for listing on Nasdaq.

UNITED STATES FEDERAL INCOME TAX CONSIDERATIONS

The following is a discussion of certain material U.S. federal income tax considerations of the acquisition, ownership and disposition of our units, shares of Class A common stock and warrants, which we refer to collectively as our securities. Because the components of a unit are separable at the option of the holder, the holder of a unit generally should be treated, for U.S. federal income tax purposes, as the owner of the underlying Class A common stock and one-half of one redeemable warrant components of the unit, as the case may be. As a result, the discussion below with respect to actual holders of Class A common stock and warrants should also apply to holders of units (as the deemed owners of the underlying Class A common stock and warrants that comprise the units). This discussion applies only to securities that are held as capital assets for U.S. federal income tax purposes and is applicable only to holders who purchased units in this offering.

This discussion assumes that the Class A common stock and warrants will trade separately and that any distributions made (or deemed made) by us on our Class A common stock and any consideration received (or deemed received) by a holder in consideration for the sale or other disposition of our securities will be in U.S. dollars.

This summary is based upon U.S. federal income tax laws as of the date of this prospectus, which is subject to change or differing interpretations, possibly with retroactive effect. This discussion is a summary only and does not describe all of the tax consequences that may be relevant to you in light of your particular circumstances, including but not limited to the alternative minimum tax, the Medicare tax on certain net investment income and the different consequences that may apply if you are subject to special rules that apply to certain types of investors, including but not limited to:

- financial institutions or financial services entities;
- broker-dealers;
- governments or agencies or instrumentalities thereof;
- regulated investment companies;
- real estate investment trusts;
- expatriates or former long-term residents of the United States;
- persons that actually or constructively own five percent or more (by vote or value) of our shares;
- persons that acquired our securities pursuant to an exercise of employee share options, in connection with employee share incentive plans or otherwise as compensation;
- insurance companies;
- dealers or traders subject to a mark-to-market method of accounting with respect to the securities;
- persons holding the securities as part of a "straddle," constructive sale, hedge, conversion or other integrated transaction or similar transaction;
- U.S. holders (as defined below) whose functional currency is not the U.S. dollar;
- partnerships (or entities or arrangements classified as partnerships or other pass-through entities for U.S. federal income tax purposes) and any beneficial owners of such entities;
- tax-exempt entities;
- · controlled foreign corporations; and
- passive foreign investment companies.

If you are a partnership (including an entity or arrangement treated as a partnership or other pass-through entity for U.S. federal income tax purposes), the U.S. federal income tax treatment of your partners, members or other beneficial owners will generally depend on the status of your partners, members or other beneficial owners and your activities, in addition to certain determinations made at the partner, member or other beneficial owner level. If you are a partner, member or other beneficial owner of a partnership (including an entity or arrangement treated as a partnership or other pass-through entity for U.S. federal income tax purposes) holding our securities, you are urged to consult your tax advisor regarding the tax consequences of the acquisition, ownership and disposition of our securities.

This discussion is based on the Internal Revenue Code of 1986, as amended (the "Code"), and administrative pronouncements, judicial decisions and final, temporary and proposed Treasury regulations as of the date hereof, which are subject to change, possibly on a retroactive basis, and changes to any of which subsequent to the date of this prospectus may affect the tax consequences described herein. This discussion does not address any aspect of state, local or non-U.S. taxation, or any U.S. federal taxes other than income taxes (such as gift and estate taxes).

We have not sought, and will not seek, a ruling from the U.S. Internal Revenue Service (the "IRS") as to any U.S. federal income tax consequence described herein. The IRS may disagree with the discussion herein, and its determination may be upheld by a court. Moreover, there can be no assurance that future legislation, regulations, administrative rulings or court decisions will not adversely affect the accuracy of the statements in this discussion. You are urged to consult your tax advisor with respect to the application of U.S. federal tax laws to your particular situation, as well as any tax consequences arising under the laws of any state, local or foreign jurisdiction.

THIS DISCUSSION IS ONLY A SUMMARY OF CERTAIN U.S. FEDERAL INCOME TAX CONSIDERATIONS ASSOCIATED WITH THE ACQUISITION, OWNERSHIP AND DISPOSITION OF OUR SECURITIES. EACH PROSPECTIVE INVESTOR IN OUR SECURITIES IS URGED TO CONSULT ITS OWN TAX ADVISOR WITH RESPECT TO THE PARTICULAR TAX CONSEQUENCES TO SUCH INVESTOR OF THE ACQUISITION, OWNERSHIP AND DISPOSITION OF OUR SECURITIES, INCLUDING THE APPLICABILITY AND EFFECT OF ANY U.S. FEDERAL NON-INCOME, STATE, LOCAL, AND NON-U.S. TAX LAWS.

Personal Holding Company Status

We could be subject to a second level of U.S. federal income tax on a portion of our income if we are determined to be a personal holding company, or PHC, for U.S. federal income tax purposes. A U.S. corporation generally will be classified as a PHC for U.S. federal income tax purposes in a given taxable year if (i) at any time during the last half of such taxable year, five or fewer individuals (without regard to their citizenship or residency and including as individuals for this purpose certain entities such as certain tax-exempt organizations, pension funds and charitable trusts) own or are deemed to own (pursuant to certain constructive ownership rules) more than 50% of the stock of the corporation by value and (ii) at least 60% of the corporation's adjusted ordinary gross income, as determined for U.S. federal income tax purposes, for such taxable year consists of PHC income (which includes, among other things, dividends, interest, certain royalties, annuities and, under certain circumstances, rents).

At least 60% of our adjusted ordinary gross income may consist of PHC income, depending on the date and size of our initial business combination. In addition, depending on the concentration of our stock in the hands of individuals, including the members of our sponsor and certain tax-exempt organizations, pension funds and charitable trusts, it is possible that more than 50% of our stock may be owned or deemed owned (pursuant to the constructive ownership rules) by such persons during the last half of a taxable year. Thus, no assurance can be given that we will not be a PHC following this offering or in the future. If we are or were to become a PHC in a given taxable year, we would be subject to an additional PHC tax, currently imposed at a rate of 20%, on our undistributed PHC income, which generally includes our taxable income, subject to certain adjustments.

Allocation of Purchase Price and Characterization of a Unit

No statutory, administrative or judicial authority directly addresses the treatment of a unit or any instrument similar to a unit for U.S. federal income tax purposes and, therefore, that treatment is not entirely clear. The acquisition of a unit should be treated for U.S. federal income tax purposes as the acquisition of one share of our Class A common stock and one-half of one warrant to acquire one our Class A common stock. For U.S. federal income tax purposes, each holder of a unit must allocate the purchase price paid by such holder for such unit between the one share of Class A common stock and the one-half of one warrant based on the relative fair market value of each at the time of issuance. Under U.S. federal income tax law, each investor must make his or her own determination of such value based on all the relevant facts and circumstances. Therefore, we strongly urge each investor to consult his or her tax adviser regarding the determination of value for these purposes. The price allocated to each share of Class A common stock and the one-half of each warrant should be the stockholder's tax basis in such share or one-half of one warrant, as the case may be. Any disposition of a unit should be treated for U.S. federal income tax purposes as a disposition of the share of Class A common stock and the one-half of one warrant comprising the unit, and the amount realized on the disposition should be allocated between the share of Class A common stock and the one-half of one warrant based on their respective relative fair market values (as determined by each such unit holder based on all the relevant facts and circumstances) at the time of disposition. The separation of shares of Class A common stock and warrants comprising the units and the combination of two halves of warrants into a single warrant should not be a taxable event for U.S. federal income tax purposes.

The foregoing treatment of the units, shares of Class A common stock and warrants and a holder's purchase price allocation are not binding on the IRS or the courts. Because there are no authorities that directly address instruments that are similar to the units, no assurance can be given that the IRS or the courts will agree with the characterization described above or the discussion below. Accordingly, each prospective investor is urged to consult its own tax advisors regarding the tax consequences of an investment in a unit (including alternative characterizations of a unit). The balance of this discussion assumes that the characterization of the units described above will be respected for U.S. federal income tax purposes.

U.S. Holders

This section applies to you if you are a "U.S. holder." A U.S. holder is a beneficial owner of our units, shares of Class A common stock or warrants who or that is, for U.S. federal income tax purposes:

- an individual who is a citizen or resident of the United States;
- a corporation (or other entity taxable as a corporation) organized in or under the laws of the United States, any state thereof or the District of Columbia;
- an estate the income of which is includible in gross income for U.S. federal income tax purposes regardless of its source; or
- a trust, if (i) a court within the United States is able to exercise primary supervision over the administration of the trust and one or more United States persons (as defined in the Code) have authority to control all substantial decisions of the trust or (ii) it has a valid election in effect under Treasury Regulations to be treated as a United States person.

Taxation of Distributions. If we pay distributions in cash or other property (other than certain distributions of our stock or rights to acquire our stock) to U.S. holders of shares of our Class A common stock, such distributions generally will constitute dividends for U.S. federal income tax purposes to the extent paid from our current or accumulated earnings and profits, as determined under U.S. federal income tax principles. Distributions in excess of current and accumulated earnings and profits will constitute a return of capital that will be applied against and reduce (but not below zero) the U.S. holder's adjusted tax basis in our Class A common stock. Any remaining excess will be treated as gain realized on the sale or other disposition of the Class A common stock and will be treated as described under "U.S. Holders — Gain or Loss on Sale, Taxable Exchange or Other Taxable Disposition of Class A Common Stock and Warrants" below.

Dividends we pay to a U.S. holder that is a taxable corporation generally will qualify for the dividends received deduction if the requisite holding period is satisfied. With certain exceptions (including, but not limited to, dividends treated as investment income for purposes of investment interest deduction limitations), and provided certain holding period requirements are met, dividends we pay to a non-corporate U.S. holder may constitute "qualified dividend income" that will be subject to tax at preferential long-term capital gains rates. It is unclear whether the redemption rights with respect to the Class A common stock described in this prospectus may prevent a U.S. holder from satisfying the applicable holding period requirements with respect to the dividends received deduction or the preferential tax rate on qualified dividend income, as the case may be. If the holding period requirements are not satisfied, then a corporation may not be able to qualify for the dividends received deduction and would have taxable income equal to the entire dividend amount, and non-corporate U.S. holders may be subject to tax on such dividend at regular ordinary income tax rates instead of the preferential rate that applies to qualified dividend income.

Gain or Loss on Sale, Taxable Exchange or Other Taxable Disposition of Class A Common Stock and Warrants. Upon a sale or other taxable disposition of our Class A common stock or warrants (which, in general, would include a redemption of Class A common stock or warrants that is treated as a sale of such securities as described below, including as a result of a dissolution and liquidation in the event we do not consummate an initial business combination within the required time period), a U.S. holder generally will recognize capital gain or loss in an amount equal to the difference between the amount realized and the U.S. holder's adjusted tax basis in the Class A common stock or warrants. Any such capital gain or loss generally will be long-term capital gain or loss if the U.S. holder's holding period for the Class A common stock or warrants so disposed of exceeds one year. It is unclear, however, whether the redemption rights with respect to the Class A common stock described in this prospectus may suspend the running of the applicable holding period for this purpose. If the running of the holding period for the Class A common stock is suspended, then non-corporate U.S. holders may not be able to satisfy the one-year holding period requirement for long-term capital gain treatment, in which case any gain on a sale or taxable disposition of the shares would be subject to short-term capital gain treatment and would be taxed at regular ordinary income tax rates. Long-term capital gains recognized by non-corporate U.S. holders may be eligible to be taxed at reduced rates. The deductibility of capital losses is subject to limitations.

Generally, the amount of gain or loss recognized by a U.S. holder is an amount equal to the difference between (i) the sum of the amount of cash and the fair market value of any property received in such disposition (or, if the Class A common stock or warrants are held as part of units at the time of the disposition, the portion of the amount realized on such disposition that is allocated to the Class A common stock or the warrants based upon the then relative fair market values of the Class A common stock or the warrants included in the units) and (ii) the U.S. holder's adjusted tax basis in its Class A common stock or warrants generally will equal the U.S. holder's acquisition cost (that is, as discussed above, the portion of the purchase price of a unit allocated to a share of Class A common stock or one-half of one warrant or, as discussed below, the U.S. holder's initial basis for Class A common stock received upon exercise of warrants) less, in the case of a share of Class A common stock, any prior distributions treated as a return of capital.

Redemption of Class A Common Stock. In the event that a U.S. holder's Class A common stock is redeemed pursuant to the redemption provisions described in this prospectus under the section of this prospectus entitled "Description of Securities — Common Stock" or if we purchase a U.S. holder's Class A common stock in an open market transaction (each of which we refer to as a "redemption"), the treatment of the transaction for U.S. federal income tax purposes will depend on whether the redemption qualifies as a sale of the Class A common stock under Section 302 of the Code. If the redemption qualifies as a sale of Class A common stock, the U.S. holder will be treated as described under "U.S. Holders — Gain or Loss on Sale, Taxable Exchange or Other Taxable Disposition of Class A Common Stock and Warrants" above. If the redemption does not qualify as a sale of Class A common stock, the U.S. holder will be treated as receiving a corporate distribution with the tax consequences described above under "U.S. Holders — Taxation of Distributions." Whether a redemption qualifies for sale treatment will depend largely on the total number of shares of our stock treated as held by the U.S. holder (including any stock constructively owned by the U.S. holder as a result of owning warrants) relative to all of our shares outstanding both before and after the redemption. The redemption of Class A common stock generally will be treated as a sale of the Class A common stock (rather than as a corporate distribution) if the redemption (i) is "substantially disproportionate" with respect to the U.S. holder. These tests are explained more fully below.

In determining whether any of the foregoing tests are satisfied, a U.S. holder takes into account not only stock actually owned by the U.S. holder, but also shares of our stock that are constructively owned by it. A U.S. holder may constructively own, in addition to stock owned directly, stock owned by certain related individuals and entities in which the U.S. holder has an interest or that have an interest in such U.S. holder, as well as any stock the U.S. holder has a right to acquire by exercise of an option, which would generally include Class A common stock which could be acquired pursuant to the exercise of the warrants. In order to meet the substantially disproportionate test, the percentage of our outstanding voting stock actually and constructively owned by the U.S. holder immediately following the redemption of Class A common stock must, among other requirements, be less than 80% of the percentage of our outstanding voting stock actually and constructively owned by the U.S. holder immediately before the redemption. There will be a complete termination of a U.S. holder's interest if either (i) all of the shares of our stock actually and constructively owned by the U.S. holder are redeemed or (ii) all of the shares of our stock actually owned by the U.S. holder are redeemed and the U.S. holder is eligible to waive, and effectively waives in accordance with specific rules, the attribution of stock owned by certain family members and the U.S. holder does not constructively own any other shares of our stock (including any stock constructively owned by the U.S. holder as a result of owning warrants). The redemption of the Class A common stock will not be essentially equivalent to a dividend if a U.S. holder's redemption results in a "meaningful reduction" of the U.S. holder's proportionate interest in us. Whether the redemption will result in a meaningful reduction in a U.S. holder's proportionate interest in us will depend on the particular facts and circumstances. However, the IRS has indicated in a published ruling that even a small reduction in the proportionate interest of a small minority stockholder in a publicly held corporation who exercises no control over corporate affairs may constitute such a "meaningful reduction." A U.S. holder should consult with its own tax advisors as to the tax consequences of a redemption.

If none of the foregoing tests is satisfied, then the redemption of any Class A common stock will be treated as a corporate distribution and the tax effects will be as described under "U.S. Holders — Taxation of Distributions," above. After the application of those rules, any remaining tax basis of the U.S. holder in the redeemed Class A common stock will be added to the U.S. holder's adjusted tax basis in its remaining stock, or, if it has none, to the U.S. holder's adjusted tax basis in its warrants or possibly in other stock constructively owned by it.

Exercise, Lapse or Redemption of a Warrant. A U.S. holder generally will not recognize taxable gain or loss on the acquisition of our Class A common stock upon exercise of a warrant for cash. The U.S. holder's tax basis in the share of our Class A common stock received upon exercise of the warrant generally will be an amount equal to the sum of the U.S. holder's initial investment in the warrant (i.e., the portion of the U.S. holder's purchase price for a unit that is allocated to the warrant, as described above under "Allocation of Purchase Price and Characterization of a Unit") and the exercise price. It is unclear whether the U.S. holder's holding period for the Class A common stock received upon exercise of the warrants will begin on the date following the date of exercise or on the date of exercise of the warrants; in either case, the holding period will not include the period during which the U.S. holder held the warrants. If a warrant is allowed to lapse unexercised, a U.S. holder generally will recognize a capital loss equal to such holder's tax basis in the warrant.

The tax consequences of a cashless exercise of a warrant are not clear under current tax law. A cashless exercise may not be taxable, either because the exercise is not a realization event or because the exercise is treated as a recapitalization for U.S. federal income tax purposes. In either situation, a U.S. holder's basis in the Class A common stock received would equal the holder's basis in the warrants exercised therefor. If the cashless exercise were treated as not being a realization event, it is unclear whether a U.S. holder's holding period in the Class A common stock would be treated as commencing on the date following the date of exercise or on the date of exercise of the warrant; in either case, the holding period would not include the period during which the U.S. holder held the warrants. If the cashless exercise were treated as a recapitalization, the holding period of the Class A common stock would include the holding period of the warrants exercised therefor.

It is also possible that a cashless exercise could be treated in part as a taxable exchange in which gain or loss would be recognized. In such event, a U.S. holder could be deemed to have surrendered a number of warrants equal to the number of shares of Class A common stock having an aggregate fair market value equal to the exercise price for the total number of warrants to be exercised. The U.S. holder would recognize capital gain or loss in an amount equal to the difference between the fair market value of the Class A common stock received in respect of the warrants deemed surrendered and the U.S. holder's tax basis in such warrants. Such gain or loss would be long-term or short-term, depending on the U.S. holder's holding period in the warrants deemed surrendered. In this case, a U.S. holder's aggregate tax basis in the Class A common stock received would equal the sum of the U.S. holder's initial investment in the warrants deemed exercised (i.e., the portion of the U.S. holder's purchase price for the units that is allocated to the warrants, as described above under "Allocation of Purchase Price and Characterization of a Unit") and the exercise price of such warrants. It is unclear whether a U.S. holder's holding period for the Class A common stock would commence on the date following the date of exercise or on the date of exercise of the warrant; in either case, the holding period would not include the period during which the U.S. holder held the warrant.

Due to the absence of authority on the U.S. federal income tax treatment of a cashless exercise, including when a U.S. holder's holding period would commence with respect to the Class A common stock received, there can be no assurance regarding which, if any, of the alternative tax consequences and holding periods described above would be adopted by the IRS or a court of law. Accordingly, U.S. holders should consult their tax advisors regarding the tax consequences of a cashless exercise.

If we redeem warrants for cash pursuant to the redemption provisions described in the section of this prospectus entitled "Description of Securities — Redeemable Warrants — Public Stockholders' Warrants" or if we purchase warrants in an open market transaction, such redemption or purchase generally will be treated as a taxable disposition to the U.S. holder, taxed as described above under "U.S. Holders — Gain or Loss on Sale, Taxable Exchange or Other Taxable Disposition of Class A Common Stock and Warrants."

Possible Constructive Distributions. The terms of each warrant provide for an adjustment to the number of shares of Class A common stock for which the warrant may be exercised or to the exercise price of the warrant in certain events, as discussed in the section of this prospectus entitled "Description of Securities — Redeemable Warrants — Public Stockholders' Warrants." An adjustment which has the effect of preventing dilution generally is not taxable. The U.S. holders of the warrants would, however, be treated as receiving a constructive distribution from us if, for example, the adjustment to the number of such shares or to such exercise price increases the warrant holders' proportionate interest in our assets or earnings and profits (e.g., through an increase in the number of shares of Class A common stock that would be obtained upon exercise or through a decrease in the exercise price of the warrant), which adjustment may be made as a result of a distribution of cash or other property, such as other securities, to the holders of shares of our Class A common stock, or as a result of the issuance of a stock dividend to holders of shares of our Class A common stock, in each case, which is taxable to the U.S. holders of such shares as a distribution. Such constructive distribution would be subject to tax as described under "U.S. Holders — Taxation of Distributions" in the same manner as if the U.S. holders of the warrants received a cash distribution from us equal to the fair market value of such increased interest.

Information Reporting and Backup Withholding. In general, information reporting requirements may apply to dividends paid to a U.S. holder and to the proceeds of the sale or other disposition of our units, shares of Class A common stock and warrants, unless the U.S. holder is an exempt recipient. Backup withholding may apply to such payments if the U.S. holder fails to provide a taxpayer identification number, a certification of exempt status or has been notified by the IRS that it is subject to backup withholding (and such notification has not been withdrawn).

Backup withholding is not an additional tax. Any amounts withheld under the backup withholding rules will be allowed as a credit against a U.S. holder's U.S. federal income tax liability and may entitle such holder to a refund, provided the required information is timely furnished to the IRS.

Non-U.S. Holders

This section applies to you if you are a "Non-U.S. holder." As used herein, the term "Non-U.S. holder" means a beneficial owner of our units, Class A common stock or warrants who or that is for U.S. federal income tax purposes:

- a non-resident alien individual (other than certain former citizens and residents of the United States subject to U.S. tax as expatriates);
- a foreign corporation; or
- an estate or trust that is not a U.S. holder;

but generally, does not include an individual who is present in the United States for 183 days or more in the taxable year of the disposition of our units, Class A common stock or warrants. If you are such an individual, you should consult your tax advisor regarding the U.S. federal income tax consequences of the acquisition, ownership or sale or other disposition of our securities.

Taxation of Distributions. In general, any distributions (including constructive distributions) we make to a Non-U.S. holder of shares of our Class A common stock, to the extent paid out of our current or accumulated earnings and profits (as determined under U.S. federal income tax principles), will constitute dividends for U.S. federal income tax purposes and, provided such dividends are not effectively connected with the Non-U.S. holder's conduct of a trade or business within the United States, we will be required to withhold tax from the gross amount of the dividend at a rate of 30%, unless such Non-U.S. holder is eligible for a reduced rate of withholding tax under an applicable income tax treaty and provides proper certification of its eligibility for such reduced rate (usually on an IRS Form W-8BEN or W-8BEN-E). In the case of any constructive dividend, it is possible that this tax would be withheld from any amount owed to a Non-U.S. holder by us or the applicable withholding agent, including cash distributions on other property or sale proceeds from warrants or other property subsequently paid or credited to such holder. Any distribution not constituting a dividend will be treated first as reducing (but not below zero) the Non-U.S. holder's adjusted tax basis in its shares of our Class A common stock and, to the extent such distribution exceeds the Non-U.S. holder's adjusted tax basis, as gain realized from the sale or other disposition of the Class A common stock, which will be treated as described under "Non-U.S. Holders — Gain on Sale, Taxable Exchange or Other Taxable Disposition of Class A Common Stock and Warrants" below. In addition, if we determine that we are likely to be classified as a "United States real property holding corporation" (see "Non-U.S. Holders — Gain on Sale, Taxable Exchange or Other Taxable Disposition of Class A Common Stock and Warrants" below), we generally will withhold 15% of any distribution that exceeds our current and accumulated earnings and profits.

The withholding tax generally does not apply to dividends paid to a Non-U.S. holder who provides a Form W-8ECI, certifying that the dividends are effectively connected with the Non-U.S. holder's conduct of a trade or business within the United States. Instead, the effectively connected dividends will be subject to regular U.S. federal income tax as if the Non-U.S. holder were a U.S. resident, subject to an applicable income tax treaty providing otherwise. A corporate Non-U.S. holder receiving effectively connected dividends may also be subject to an additional "branch profits tax" imposed at a rate of 30% (or a lower applicable treaty rate).

Exercise, Lapse or Redemption of a Warrant. The U.S. federal income tax treatment of a Non-U.S. holder's exercise of a warrant, or the lapse of a warrant held by a Non-U.S. holder, generally will correspond to the U.S. federal income tax treatment of the exercise or lapse of a warrant by a U.S. holder, as described under "U.S. Holders — Exercise, Lapse or Redemption of a Warrant" above, although to the extent a cashless exercise results in a taxable exchange, the consequences would be similar to those described below in "Non-U.S. Holders — Gain on Sale, Taxable Exchange or Other Taxable Disposition of Class A Common Stock and Warrants."

The characterization for U.S. federal income tax purposes of the redemption of the Non-U.S. holder's warrants generally will correspond to the U.S. federal income tax treatment of such a redemption of a U.S. holder's warrants, as described under "U.S. Holders—Exercise, Lapse or Redemption of a Warrant" above, and the consequences of the redemption to the Non-U.S. holder will be as described below under the heading "Non-U.S. Holders — Gain on Sale, Taxable Exchange or Other Taxable Disposition of Class A Common Stock and Warrants" depending on such characterization.

Gain on Sale, Taxable Exchange or Other Taxable Disposition of Class A Common Stock and Warrants. A Non-U.S. holder generally will not be subject to U.S. federal income or withholding tax in respect of gain recognized on a sale, taxable exchange or other taxable disposition of our Class A common stock, which would include a dissolution and liquidation in the event we do not complete an initial business combination within the required time period, or warrants (including an expiration or redemption of our warrants), in each case without regard to whether those securities were held as part of a unit, unless:

- the gain is effectively connected with the conduct by the Non-U.S. holder of a trade or business within the United States (and, under certain income tax treaties, is attributable to a United States permanent establishment or fixed base maintained by the Non-U.S. holder); or
- we are or have been a "United States real property holding corporation" for U.S. federal income tax purposes at any time during the shorter of the five-year period ending on the date of disposition or the period that the Non-U.S. holder held our Class A common stock, and, in the case where shares of our Class A common stock are regularly traded on an established securities market, the Non-U.S. holder has owned, directly or constructively, more than 5% of our Class A common stock at any time within the shorter of the five-year period preceding the disposition or such Non-U.S. holder's holding period for the shares of our Class A common stock. There can be no assurance that our Class A common stock will be treated as regularly traded on an established securities market for this purpose.

Unless an applicable treaty provides otherwise, gain described in the first bullet point above will be subject to tax at generally applicable U.S. federal income tax rates as if the Non-U.S. holder were a U.S. resident. Any gains described in the first bullet point above of a Non-U.S. holder that is a foreign corporation may also be subject to an additional "branch profits tax" imposed at a 30% rate (or lower treaty rate).

If the second bullet point above applies to a Non-U.S. holder, gain recognized by such holder on the sale, exchange or other disposition of our Class A common stock or warrants will be subject to tax at generally applicable U.S. federal income tax rates. In addition, a buyer of our Class A common stock or warrants from such holder may be required to withhold U.S. federal income tax at a rate of 15% of the amount realized upon such disposition. We cannot determine whether we will be a "United States real property holding corporation" in the future until we complete an initial business combination. We will be classified as a United States real property holding corporation if the fair market value of our "United States real property interests" equals or exceeds 50% of the sum of the fair market value of our worldwide real property interests plus our other assets used or held for use in a trade or business, as determined for U.S. federal income tax purposes.

Redemption of Class A Common Stock. The characterization for U.S. federal income tax purposes of the redemption of a Non-U.S. holder's Class A common stock pursuant to the redemption provisions described in the section of this prospectus entitled "Description of Securities — Common Stock" generally will correspond to the U.S. federal income tax characterization of such a redemption of a U.S. holder's Class A common stock, as described under "U.S. Holders — Redemption of Class A Common Stock" above, and the consequences of the redemption to the Non-U.S. holder will be as described above under "Non-U.S. Holders — Taxation of Distributions" and "Non-U.S. Holders — Gain on Sale, Taxable Exchange or Other Taxable Disposition of Class A Common Stock and Warrants," as applicable.

Because it may not be certain at the time a Non-U.S. holder is redeemed whether such Non-U.S. holder's redemption will be treated as a sale of shares or a distribution constituting a dividend, and because such determination will depend in part on a Non-U.S. holder's particular circumstances, we or the applicable withholding agent may not be able to determine whether (or to what extent) a Non-U.S. holder is treated as receiving a dividend for U.S. federal income tax purposes. Therefore, we or the applicable withholding agent may withhold tax at a rate of 30% on the gross amount of any consideration paid to a Non-U.S. holder in redemption of such Non-U.S. holder's Class A common stock unless special procedures are available to Non-U.S. holders to certify that they are entitled to exemptions from, or reductions in, such withholding tax. However, there can be no assurance that such special certification procedures will be available. A Non-U.S. holder generally may obtain a refund of any such excess amounts withheld by timely filing an appropriate claim for refund with the IRS. Non-U.S. holders should consult their own tax advisors regarding the application of the foregoing rules in light of their particular facts and circumstances.

Possible Constructive Distributions. The terms of each warrant provide for an adjustment to the number of shares of Class A common stock for which the warrant may be exercised or to the exercise price of the warrant in certain events, as discussed in the section of this prospectus entitled "Description of Securities — Redeemable Warrants — Public Stockholders' Warrants." An adjustment which has the effect of preventing dilution generally is not a taxable event. Nevertheless, a Non-U.S. holder of warrants would be treated as receiving a constructive distribution from us if, for example, the adjustment increases the holder's proportionate interest in our assets or earnings and profits (e.g., through an increase in the number of shares of Class A common stock that would be obtained upon exercise or through a decrease in the exercise price of the warrants), which adjustment may be made as a result of a distribution of cash or other property, such as other securities, to the holders of shares of our Class A common stock, or as a result of the issuance of a stock dividend to holders of shares of our Class A common stock, in each case, which is taxable to such holders as a distribution. Any constructive distribution received by a Non-U.S. holder would be subject to U.S. federal income tax (including any applicable withholding) in the same manner as if such Non-U.S. holder received a cash distribution from us equal to the fair market value of such increased interest without any corresponding receipt of cash. It is possible that any withholding tax on such a constructive distribution might be satisfied by us or the applicable withholding agent through a sale of a portion of the Non-U.S. holder's shares of Class A common stock, warrants or other property held or controlled by us or the applicable withholding agent on behalf of the Non-U.S. holder.

Information Reporting and Backup Withholding. Information returns will be filed with the IRS in connection with payments of dividends and the proceeds from a sale or other disposition of our units, shares of Class A common stock and warrants. A Non-U.S. holder may have to comply with certification procedures to establish that it is not a United States person in order to avoid information reporting and backup withholding requirements. The certification procedures required to claim a reduced rate of withholding under a treaty generally will satisfy the certification requirements necessary to avoid the backup withholding as well. Backup withholding is not an additional tax. The amount of any backup withholding from a payment to a Non-U.S. holder will be allowed as a credit against such holder's U.S. federal income tax liability and may entitle such holder to a refund, provided that the required information is timely furnished to the IRS.

FATCA Withholding Taxes. Provisions commonly referred to as "FATCA" impose withholding of 30% on payments of dividends (including constructive dividends) on our Class A common stock to "foreign financial institutions" (which is broadly defined for this purpose and in general includes investment vehicles) and certain other non-U.S. entities unless various U.S. information reporting and due diligence requirements (generally relating to ownership by United States persons of interests in or accounts with those entities) have been satisfied by, or an exemption applies to, the payee (typically certified as to by the delivery of a properly completed IRS Form W-8BEN-E). Foreign financial institutions located in jurisdictions that have an intergovernmental agreement with the United States governing FATCA may be subject to different rules. Under certain circumstances, a Non-U.S. holder might be eligible for refunds or credits of such withholding taxes, and a Non-U.S. holder might be required to file a U.S. federal income tax return to claim such refunds or credits. Thirty percent withholding under FATCA was scheduled to apply to payments of gross proceeds from the sale or other disposition of property that produces U.S.-source interest or dividends beginning on January 1, 2019, but on December 13, 2018, the IRS released proposed regulations that, if finalized in their proposed form, would eliminate the obligation to withhold on gross proceeds. Such proposed regulations also delayed withholding on certain other payments received from other foreign financial institutions that are allocable, as provided for under final Treasury Regulations, to payments of U.S.-source dividends, and other fixed or determinable annual or periodic income. Although these proposed Treasury Regulations are not final, taxpayers generally may rely on them until final Treasury Regulations are issued. Prospective investors should consult their tax advisors regarding the effects of FATCA on their investment in our securities.

UNDERWRITING

EF Hutton, a division of Benchmark Investments, LLC, the "representative" is acting as the sole book-running manager of the offering and as representative of the underwriters named below. Subject to the terms and conditions of the underwriting agreement dated the date of this prospectus, the underwriters named below, through the representative, have severally agreed to purchase, and we have agreed to sell to the underwriters, the following respective number of units set forth opposite the underwriter's name.

Underwriters	Number of Units
EF Hutton, division of Benchmark Investments, LLC	
Total	10,000,000

The underwriting agreement provides that the obligations of the underwriters to purchase the units included in this offering are subject to approval of legal matters by counsel and to other conditions. The underwriters are obligated to purchase all of the units (other than those covered by the over-allotment option described below) if they purchase any of the units.

Units sold by the underwriters to the public will initially be offered at the initial public offering price set forth on the cover of this prospectus. Any units sold by the underwriters to securities dealers may be sold at a discount from the initial public offering price not to exceed \$_____ per unit. If all of the units are not sold at the initial offering price, the underwriters may change the offering price and the other selling terms. The underwriters have advised us that they do not intend to make sales to discretionary accounts.

If the underwriters sell more units than the total number set forth in the table above, we have granted to the underwriters an option, exercisable for 45 days from the date of this prospectus, to purchase up to 1,500,000 additional units at the public offering price less the underwriting discount. The underwriters may exercise this option solely for the purpose of covering over-allotments, if any, in connection with this offering. Any units issued or sold under the option will be issued and sold on the same terms and conditions as the other units that are the subject of this offering.

We, our sponsor and our officers and directors have agreed that, for a period of 180 days from the date of this prospectus, we and they will not, without the prior written consent of the representative, sell, offer to sell, contract or agree to sell, hypothecate, pledge, grant any option to purchase or otherwise dispose of or agree to dispose of, directly or indirectly, or establish or increase a put equivalent position or liquidate or decrease a call equivalent position within the meaning of Section 16 of the Exchange Act with respect to any units, shares of Class A common stock, warrants, or any securities convertible into, or exercisable, or exchangeable for, shares of Class A common stock, (ii) enter into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any units, shares of common stock, warrants, or any securities convertible into, or exercisable, or exchangeable for, shares of Class A common stock, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (iii) publicly announce any intention to effect any transaction specified in clause (i) or (ii), subject to certain exceptions. The representative in its sole discretion may release any of the securities subject to the lock-up agreement at any time without notice, other than in the case of the officers and directors, which shall be with notice. Our sponsor, officers and directors are also subject to separate transfer restrictions on their founder shares and placement warrants pursuant to the letter agreement as described herein.

Our initial stockholders have agreed not to transfer, assign or sell any of their founder shares (or shares of common stock issuable upon conversion thereof) until the earlier to occur of: (A) one year after the completion of our initial business combination and (B) subsequent to our initial business combination, (x) if the reported last sale price of our Class A common stock equals or exceeds \$12.00 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like) for any 20 trading days within any 30-trading day period commencing at least 150 days after our initial business combination, or (y) the date on which we complete a liquidation, merger, capital stock exchange, right issuances, reorganization or other similar transaction that results in all of our stockholders having the right to exchange their shares of common stock for cash, securities or other property (except with respect to permitted transferees as described herein under the section of this prospectus entitled "Principal Stockholders — Restrictions on Transfers of Founder Shares and Placement Warrants"). The placement warrants (including the Class A common stock issuable upon exercise of the placement warrants) will not be transferable, assignable or salable until 30 days after the completion of our initial business combination (except with respect to permitted transferees as described herein under the section of this prospectus entitled "Principal Stockholders — Restrictions on Transfers of Founder Shares and Placement Warrants").

Prior to this offering, there has been no public market for our securities. Consequently, the initial public offering price for the units was determined by negotiations between us and the underwriters. Among the factors considered in determining the initial public offering price were the history and prospects of companies whose principal business is the acquisition of other companies, prior offerings of those companies, our management, our capital structure, and currently prevailing general conditions in the equity securities markets, including current market valuations of publicly traded companies considered comparable to our company. We cannot assure you, however, that the price at which the units, Class A common stock or warrants will sell in the public market after this offering will not be lower than the initial public offering price or that an active trading market in our units, Class A common stock or warrants will develop and continue after this offering.

We have applied to list our units on Nasdaq under the symbol "AEHAU." We cannot guarantee that our securities will be approved for listing on Nasdaq. We expect that our units will be listed on Nasdaq on or promptly after the date of this prospectus. We expect that our Class A common stock and warrants will be listed under the symbols "AEHA" and "AEHAW," respectively, once the Class A common stock and warrants begin separate trading.

The following table shows the underwriting discounts and commissions that we are to pay to the underwriters in connection with this offering. These amounts are shown assuming both no exercise and full exercise of the underwriters' over-allotment option. The upfront portion of the underwriting discounts and commissions will be \$0.10 per unit.

Pavable by Aesther

			Healthcare quisition Corp.		
	_	No Exercise		Full Exercise	
er Unit ⁽¹⁾	\$	0.40	\$	0.40	
Total ⁽¹⁾	\$	4,000,000	\$	4,600,000	

(1) Includes \$0.30 per unit, or \$3,000,000 (or \$3,450,000 if the underwriters' over-allotment option is exercised in full) in the aggregate payable to the underwriters for deferred underwriting commissions to be placed in a trust account located in the United States as described herein. The deferred commissions will be released to the representative only on completion of an initial business combination as described in this prospectus.

If we do not complete our initial business combination and subsequently liquidate, the underwriters have agreed that (i) they will forfeit any rights or claims to their deferred underwriting discounts and commissions, including any accrued interest thereon, then in the trust account upon liquidation, and (ii) that the deferred underwriters' discounts and commissions will be distributed on a pro rata basis, including interest earned on the funds held in the trust account and not previously released to us to pay our taxes, to the public stockholders.

Stabilization

In connection with the offering, the underwriters may purchase and sell units in the open market. Purchases and sales in the open market may include short sales, purchases to cover short positions, which may include purchases pursuant to the over-allotment option and stabilizing purchases, in accordance with Regulation M under the Exchange Act.

- Short sales involve secondary market sales by the underwriters of a greater number of units than it is required to purchase in the offering.
- "Covered" short sales are sales of units in an amount up to the number of units represented by the underwriters' over-allotment option.

- "Naked" short sales are sales of units in an amount in excess of the number of units represented by the underwriters' over-allotment option.
- Covering transactions involve purchases of units either pursuant to the over-allotment option or in the open market after the distribution has been completed in order to cover short positions.
- To close a naked short position, the underwriters must purchase units in the open market after the distribution has been completed. A naked short position is more likely to be created if the underwriters are concerned that there may be downward pressure on the price of the units in the open market after pricing that could adversely affect investors who purchase in the offering.
- To close a covered short position, the underwriters must purchase units in the open market after the distribution has been completed or must exercise the over-allotment option. In determining the source of units to close the covered short position, the underwriters will consider, among other things, the price of units available for purchase in the open market as compared to the price at which they may purchase units through the over-allotment option.
- Stabilizing transactions involve bids to purchase units so long as the stabilizing bids do not exceed a specified maximum.

Purchases to cover short positions and stabilizing purchases, as well as other purchases by the underwriters for their own account, may have the effect of preventing or retarding a decline in the market price of the units. They may also cause the price of the units to be higher than the price that would otherwise exist in the open market in the absence of these transactions. The underwriters may conduct these transactions in the over-the-counter market or otherwise. If the underwriters commence any of these transactions, it may discontinue them at any time.

We estimate that our portion of the total expenses of this offering payable by us will be \$1,261,000, excluding underwriting discounts and commissions. We have agreed to reimburse the underwriters for certain expenses, including legal fees which will not exceed \$100,000 in the aggregate as well as fees related to the review by FINRA.

We have agreed to indemnify the underwriters against certain liabilities, including liabilities under the Securities Act, or to contribute to payments the underwriters may be required to make because of any of those liabilities.

We are not under any contractual obligation to engage the underwriters to provide any services for us after this offering, and have no present intent to do so. However, the underwriters may introduce us to potential target businesses or assist us in raising additional capital in the future. If the underwriters provide services to us after this offering, we may pay the underwriters fair and reasonable fees that would be determined at that time in an arm's length negotiation; provided that no agreement will be entered into with the underwriters and no fees for such services will be paid to the underwriters prior to the date that is 60 days from the date of this prospectus, unless FINRA determines that such payment would not be deemed underwriters' compensation in connection with this offering and we may pay the underwriters of this offering or any entity with which it is affiliated a finder's fee or other compensation for services rendered to us in connection with the completion of an initial business combination.

The underwriters and their affiliates may in the future engage in, investment banking and other commercial dealings in the ordinary course of business with us or our affiliates, for which it may in the future receive, customary fees and commissions for any such transactions.

The underwriters and their respective affiliates are financial institutions engaged in various activities, which may include, among other things, securities trading, commercial and investment banking, financial advisory, investment management, financing and brokerage activities. Such underwriters and their affiliates have in the past, in the ordinary course of business, provided certain of these services to affiliates of our sponsor, and have arrangements in place whereby they may currently or in the future provide such services to affiliates of our sponsor, for which they have received and may receive customary fees, interest, commissions and other compensation.

In addition, in the ordinary course of their business activities, the underwriters and their affiliates may make or hold a broad array of investments and actively trade debt and equity securities (or related derivative securities) and financial instruments (including bank loans) for their own account and for the accounts of their customers. Such investments and securities activities may involve securities and/or instruments of ours or our affiliates. The underwriters and their affiliates may also make investment recommendations and/or publish or express independent research views in respect of such securities or financial instruments and may hold, or recommend to clients that they acquire, long and/or short positions in such securities and instruments.

A prospectus in electronic format may be made available by e-mail or on the web sites or through online services maintained by one or more of the underwriters or their affiliates. In those cases, prospective investors may view offering terms online and may be allowed to place orders online. The underwriters may agree with us to allocate a specific number of units for sale to online brokerage account holders. Any such allocation for online distributions will be made by the underwriters on the same basis as other allocations. Other than the prospectus in electronic format, the information on the underwriters' web sites and any information contained in any other web site maintained by any of the underwriters is not part of this prospectus, has not been approved and/or endorsed by us or the underwriters and should not be relied upon by investors.

Representative Shares

We have agreed to issue to the representative and/or its designees, 100,000 shares of common stock upon the consummation of this offering. The representative has agreed not to transfer, assign or sell any such shares of common stock until the completion of our initial business combination. In addition, the representative has agreed (i) to waive its redemption rights with respect to such shares of common stock in connection with the completion of our initial business combination and (ii) to waive its rights to liquidating distributions from the trust account with respect to such shares of common stock if we fail to complete our initial business combination within 12 months (as such period may be extended up to 18 months at the election of the Company subject to the satisfaction of certain conditions) from the closing of this offering.

The shares of common stock have been deemed compensation by FINRA and are therefore subject to a lock-up for a period of 180 days immediately following the commencement of sales in this offering pursuant to FINRA Rule 5110(e)(1). Pursuant to FINRA Rule 5110(e)(1), these securities may not be sold, transferred, assigned, pledged or hypothecated or the subject of any hedging, short sale, derivative, put or call transaction that would result in the economic disposition of the securities by any person for a period of 180 days immediately following the commencement of sales in this offering, except to any underwriter and selected dealer participating in the offering and their officers or partners, registered persons or affiliates or as otherwise permitted under Rule 5110(e)(2).

Selling Restrictions

Notice to Prospective Investors in the European Economic Area and the United Kingdom

In relation to each member state of the European Economic Area and the United Kingdom (each, a "relevant state"), no units have been offered or will be offered pursuant to the offering to the public in that relevant state prior to the publication of a prospectus in relation to the units that has been approved by the competent authority in that relevant state or, where appropriate, approved in another relevant state and notified to the competent authority in that relevant state, all in accordance with the Prospectus Regulation, except that offers of our units may be made to the public in that relevant state at any time under the following exemptions under the Prospectus Regulation:

- to any legal entity which is a qualified investor as defined under the Prospectus Regulation;
- to fewer than 150 natural or legal persons (other than qualified investors as defined under the Prospectus Regulation), subject to obtaining the prior consent of the representative for any such offer; or
- in any other circumstances falling within Article 1(4) of the Prospectus Regulation.

provided that no such offer of units shall require the issuer or the representative to publish a prospectus pursuant to Article 3 of the Prospectus Regulation or supplement a prospectus pursuant to Article 23 of the Prospectus Regulation.

Each person in a relevant state who initially acquires any units or to whom any offer is made will be deemed to have represented, acknowledged and agreed to and with the Company and the representative that it is a qualified investor within the meaning of the Prospectus Regulation.

In the case of any units being offered to a financial intermediary as that term is used in Article 5(1) of the Prospectus Regulation, each such financial intermediary will be deemed to have represented, acknowledged and agreed that the units acquired by it in the offer have not been acquired on a non-discretionary basis on behalf of, nor have they been acquired with a view to their offer or resale to, persons in circumstances which may give rise to an offer to the public other than their offer or resale in a relevant state to qualified investors, in circumstances in which the prior consent of the representative has been obtained to each such proposed offer or resale.

We, the representative and each of our and the representative's respective affiliates will rely upon the truth and accuracy of the foregoing representations, acknowledgements and agreements.

For the purposes of this provision, the expression an "offer to the public" in relation to any units in any relevant state means the communication in any form and by any means of sufficient information on the terms of the offer and any units to be offered so as to enable an investor to decide to purchase or subscribe for any units, and the expression "Prospectus Regulation" means Regulation (EU) 2017/1129.

References to the Prospectus Regulation include, in relation to the United Kingdom, the Prospectus Regulation as it forms part of United Kingdom domestic law by virtue of the European Union (Withdrawal) Act 2018.

The above selling restriction is in addition to any other selling restrictions set out below.

In connection with the offering, the representative is not acting for anyone other than the issuer and will not be responsible to anyone other than the issuer for providing the protections afforded to its clients nor for providing advice in relation to the offering.

Notice to Prospective Investors in the United Kingdom

This prospectus is for distribution only to persons who (i) have professional experience in matters relating to investments and who qualify as investment professionals within the meaning of Article 19(5) of the Financial Services and Markets Act 2000 (Financial Promotion) Order 2005 (as amended, the "Financial Promotion Order"), (ii) are persons falling within Article 49(2)(a) to (d) ("high net worth companies, unincorporated associations etc.") of the Financial Promotion Order, (iii) are outside the United Kingdom, or (iv) are persons to whom an invitation or inducement to engage in investment activity (within the meaning of Section 21 of the Financial Services and Markets Act 2000, as amended ("FSMA")) in connection with the issue or sale of any securities may otherwise lawfully be communicated or caused to be communicated (all such persons together being referred to as "relevant persons"). This document is directed only at relevant persons and must not be acted on or relied on by persons who are not relevant persons. Any investment or investment activity to which this document relates is available only to relevant persons and will be engaged in only with relevant persons.

Notice to Prospective Investors in France

Neither this prospectus nor any other offering material relating to the units described in this prospectus has been submitted to the clearance procedures of the Autorité des Marchés Financiers or of the competent authority of another member state of the European Economic Area and notified to the Autorité des Marchés Financiers. The units have not been offered or sold and will not be offered or sold, directly or indirectly, to the public in France. Neither this prospectus nor any other offering material relating to the units has been or will be:

- released, issued, distributed or caused to be released, issued or distributed to the public in France; or
- used in connection with any offer for subscription or sale of the units to the public in France. Such offers, sales and distributions will be made in France only:
- To qualified investors (investisseurs qualifiés) and/or to a restricted circle of investors (cercle restreint d'investisseurs), in each case investing for their own account, all as defined in, and in accordance with, articles L.411-2, D.411-1, D.411-2, D.734-1, D.744-1, D.754-1 and D.764-1 of the French Code monétaire et financier;
- to investment services providers authorized to engage in portfolio management on behalf of third parties; or
- in a transaction that, in accordance with article L.411-2-II-1° -or-2° -or 3° of the French Code monétaire et financier and article 211-2 of the General Regulations (Règlement Général) of the Autorité des Marchés Financiers, does not constitute a public offer (appel public à l'épargne).

The units may be resold directly or indirectly, only in compliance with articles L.411-1, L.411-2, L.412-1 and L.621-8 through L.621-8-3 of the French Code monétaire et financier.

Notice to Prospective Investors in Hong Kong

The units have not been offered or sold and will not be offered or sold in Hong Kong, by means of any document, other than (a) to "professional investors" as defined in the Securities and Futures Ordinance (Cap. 571) of Hong Kong and any rules made under that Ordinance; or (b) in other circumstances which do not result in the document being a "prospectus" as defined in the Companies Ordinance (Cap. 32) of Hong Kong or which do not constitute an offer to the public within the meaning of that Ordinance. No advertisement, invitation or document relating to the units has been or may be issued or has been or may be in the possession of any person for the purposes of issue, whether in Hong Kong or elsewhere, which is directed at, or the contents of which are likely to be accessed or read by, the public of Hong Kong (except if permitted to do so under the securities laws of Hong Kong) other than with respect to units which are or are intended to be disposed of only to persons outside Hong Kong or only to "professional investors" as defined in the Securities and Futures Ordinance and any rules made under that Ordinance.

Notice to Prospective Investors in Japan

The units have not been and will not be registered under the Financial Instruments and Exchange Law of Japan (Law No. 25 of 1948, as amended) and, accordingly, will not be offered or sold, directly or indirectly, in Japan, or for the benefit of any Japanese Person or to others for re-offering or resale, directly or indirectly, in Japan or to any Japanese Person, except in compliance with all applicable laws, regulations and ministerial guidelines promulgated by relevant Japanese governmental or regulatory authorities in effect at the relevant time. For the purposes of this paragraph, "Japanese Person" shall mean any person resident in Japan, including any corporation or other entity organized under the laws of Japan.

Notice to Prospective Investors in Singapore

This prospectus has not been registered as a prospectus with the Monetary Authority of Singapore. Accordingly, the units were not offered or sold or caused to be made the subject of an invitation for subscription or purchase and will not be offered or sold or caused to be made the subject of an invitation for subscription or purchase, and this prospectus or any other document or material in connection with the offer or sale, or invitation for subscription or purchase, of the units, has not been circulated or distributed, nor will it be circulated or distributed, whether directly or indirectly, to any person in Singapore other than (i) to an institutional investor (as defined in Section 4A of the Securities and Futures Act (Chapter 289) of Singapore, as modified or amended from time to time (the "SFA")) pursuant to Section 274 of the SFA, (ii) to a relevant person (as defined in Section 275(2) of the SFA) pursuant to Section 275(1) of the SFA, or any person pursuant to Section 275(1A) of the SFA, and in accordance with the conditions specified in Section 275 of the SFA, or (iii) otherwise pursuant to, and in accordance with the conditions of, any other applicable provision of the SFA.

Where the units are subscribed or purchased under Section 275 of the SFA by a relevant person which is:

- (a) a corporation (which is not an accredited investor (as defined in Section 4A of the SFA)) the sole business of which is to hold investments and the entire share capital of which is owned by one or more individuals, each of whom is an accredited investor; or
- (b) a trust (where the trustee is not an accredited investor) whose sole purpose is to hold investments and each beneficiary of the trust is an individual who is an accredited investor.

securities or securities-based derivatives contracts (each term as defined in Section 2(1) of the SFA) of that corporation or the beneficiaries' rights and interest (howsoever described) in that trust shall not be transferred within six months after that corporation or that trust has acquired the units pursuant to an offer made under Section 275 of the SFA except:

- (a) to an institutional investor or to a relevant person, or to any person arising from an offer referred to in Section 275(1A) or Section 276(4)(i)(B) of the SFA;
 - (b) where no consideration is or will be given for the transfer;
 - (c) where the transfer is by operation of law; or
 - (d) as specified in Section 276(7) of the SFA.

Notice to Prospective Investors in Canada

The units may be sold in Canada only to purchasers purchasing, or deemed to be purchasing, as principal that are accredited investors, as defined in National Instrument 45-106 *Prospectus Exemptions* or subsection 73.3(1) of the *Securities Act* (Ontario), and are permitted clients, as defined in National Instrument 31-103 *Registration Requirements, Exemptions and Ongoing Registrant Obligations*. Any resale of the units must be made in accordance with an exemption from, or in a transaction not subject to, the prospectus requirements of applicable securities laws.

Securities legislation in certain provinces or territories of Canada may provide a purchaser with remedies for rescission or damages if this prospectus (including any amendment thereto) contains a misrepresentation, provided that the remedies for rescission or damages are exercised by the purchaser within the time limit prescribed by the securities legislation of the purchaser's province or territory. The purchaser should refer to any applicable provisions of the securities legislation of the purchaser's province or territory for particulars of these rights or consult with a legal advisor.

Pursuant to section 3A.3 (or, in the case of securities issued or guaranteed by the government of a non-Canadian jurisdiction, section 3A.4) of National Instrument 33-105 *Underwriting Conflicts* (NI 33-105), the underwriters are not required to comply with the disclosure requirements of NI 33-105 regarding underwriter conflicts of interest in connection with this offering.

Notice to Prospective Investors in the Dubai International Financial Centre

This prospectus relates to an Exempt Offer in accordance with the Offered Securities Rules of the Dubai Financial Services Authority ("DFSA"). This prospectus is intended for distribution only to persons of a type specified in the Offered Securities Rules of the DFSA. It must not be delivered to, or relied on by, any other person. The DFSA has no responsibility for reviewing or verifying any documents in connection with Exempt Offers. The DFSA has not approved this prospectus nor taken steps to verify the information set forth herein and has no responsibility for the prospectus. The securities to which this prospectus relates may be illiquid and/or subject to restrictions on their resale.

Prospective purchasers of the securities offered should conduct their own due diligence on the securities. If you do not understand the contents of this prospectus you should consult an authorized financial advisor.

Notice to Prospective Investors in Australia

No placement document, prospectus, product disclosure statement or other disclosure document has been lodged with the Australian Securities and Investments Commission ("ASIC"), in relation to the offering. This prospectus does not constitute a prospectus, product disclosure statement or other disclosure document under the Corporations Act"), and does not purport to include the information required for a prospectus, product disclosure statement or other disclosure document under the Corporations Act. Any offer in Australia of the securities may only be made to persons (the "Exempt Investors") who are "sophisticated investors" (within the meaning of section 708(8) of the Corporations Act), "professional investors" (within the meaning of section 708(11) of the Corporations Act) or otherwise pursuant to one or more exemptions contained in section 708 of the Corporations Act so that it is lawful to offer the securities without disclosure to investors under Chapter 6D of the Corporations Act.

The securities applied for by Exempt Investors in Australia must not be offered for sale in Australia in the period of 18 months after the date of allotment under the offering, except in circumstances where disclosure to investors under Chapter 6D of the Corporations Act would not be required pursuant to an exemption under section 708 of the Corporations Act or otherwise or where the offer is pursuant to a disclosure document which complies with Chapter 6D of the Corporations Act. Any person acquiring securities must observe such Australian on-sale restrictions. This prospectus contains general information only and does not take account of the investment objectives, financial situation or particular needs of any particular person. It does not contain any securities recommendations or financial product advice. Before making an investment decision, investors need to consider whether the information in this prospectus is appropriate to their needs, objectives and circumstances, and, if necessary, seek expert advice on those matters.

Notice to Prospective Investors in Switzerland

The securities may not be publicly offered in Switzerland and will not be listed on the SIX Swiss Exchange ("SIX") or on any other stock exchange or regulated trading facility in Switzerland. This document has been prepared without regard to the disclosure standards for issuance prospectuses under art. 652a or art. 1156 of the Swiss Code of Obligations or the disclosure standards for listing prospectuses under art. 27 ff. of the SIX Listing Rules or the listing rules of any other stock exchange or regulated trading facility in Switzerland. Neither this document nor any other offering or marketing material relating to the securities or the offering may be publicly distributed or otherwise made publicly available in Switzerland.

Neither this document nor any other offering or marketing material relating to the offering, the company, the shares have been or will be filed with or approved by any Swiss regulatory authority. In particular, this document will not be filed with, and the offer of securities will not be supervised by, the Swiss Financial Market Supervisory Authority FINMA (FINMA), and the offer of securities has not been and will not be authorized under the Swiss Federal Act on Collective Investment Schemes ("CISA"). The investor protection afforded to acquirers of interests in collective investment schemes under the CISA does not extend to acquirers of securities.

Notice to Prospective Investors in Israel

In the State of Israel, this prospectus shall not be regarded as an offer to the public to purchase securities under the Israeli Securities Law, 5728 – 1968, which requires a prospectus to be published and authorized by the Israel Securities Authority, if it complies with certain provisions of Section 15 of the Israeli Securities Law, 5728 – 1968, including, inter alia, if: (i) the offer is made, distributed or directed to not more than 35 investors, subject to certain conditions (the "Addressed Investors"); or (ii) the offer is made, distributed or directed to certain qualified investors defined in the First Addendum of the Israeli Securities Law, 5728 –1968, subject to certain conditions (the "Qualified Investors"). The Qualified Investors shall not be taken into account in the count of the Addressed Investors and may be offered to purchase securities in addition to the 35 Addressed Investors. The Company has not and will not take any action that would require it to publish a prospectus in accordance with and subject to the Israeli Securities Law, 5728 – 1968. We have not and will not distribute this prospectus or make, distribute or direct an offer to subscribe for our securities to any person within the State of Israel, other than to Qualified Investors and up to 35 Addressed Investors.

Qualified Investors may have to submit written evidence that they meet the definitions set out in of the First Addendum to the Israeli Securities Law, 5728 – 1968. In particular, we may request, as a condition to be offered securities, that Qualified Investors will each represent, warrant and certify to us and/or to anyone acting on our behalf: (i) that it is an investor falling within one of the categories listed in the First Addendum to the Israeli Securities Law, 5728 – 1968 regarding Qualified Investors is applicable to it; (iii) that it will abide by all provisions set forth in the Israeli Securities Law, 5728 – 1968 and the regulations promulgated thereunder in connection with the offer to be issued securities; (iv) that the securities that it will be issued are, subject to exemptions available under the Israeli Securities Law, 5728 – 1968: (a) for its own account; (b) for investment purposes only; and (c) not issued with a view to resale within the State of Israel, other than in accordance with the provisions of the Israeli Securities Law, 5728 – 1968; and (v) that it is willing to provide further evidence of its Qualified Investor status. Addressed Investors may have to submit written evidence in respect of their identity and may have to sign and submit a declaration containing, inter alia, the Addressed Investor's name, address and passport number or Israeli identification number.

We have not authorized and do not authorize the making of any offer of securities through any financial intermediary on our behalf, other than offers made by the underwriters and their respective affiliates, with a view to the final placement of the securities as contemplated in this document. Accordingly, no purchaser of the shares, other than the underwriters, is authorized to make any further offer of shares on our behalf or on behalf of the underwriters.

LEGAL MATTERS

The Loev Law Firm, PC, Bellaire, Texas, is acting as counsel in connection with the registration of our securities under the Securities Act, and as such, will pass upon the validity of the securities offered in this prospectus.

Certain legal matters will be passed upon on behalf of the underwriters by Schiff Hardin LLP, Washington, DC.

EXPERTS

The financial statements of Aesther Healthcare Acquisition Corp. as of July 12, 2021, and for the period from June 17, 2021 (inception) through July 12, 2021 appearing in this prospectus have been audited by MaloneBailey LLP, independent registered public accounting firm, as stated in their report herein (which contains an explanatory paragraph relating to substantial doubt about the ability of Aesther Healthcare Acquisition Corp. to continue as a going concern as described in Note 1 to the financial statements), appearing elsewhere in this prospectus, and are included in reliance upon such report given on the authority of such firm as experts in accounting and auditing.

WHERE YOU CAN FIND ADDITIONAL INFORMATION

We have filed with the SEC a registration statement on Form S-1 under the Securities Act with respect to the securities we are offering by this prospectus. This prospectus does not contain all of the information included in the registration statement. For further information about us and our securities, you should refer to the registration statement and the exhibits and schedules filed with the registration statement. Whenever we make reference in this prospectus to any of our contracts, agreements or other documents, the references are materially complete but may not include a description of all aspects of such contracts, agreements or other documents, and you should refer to the exhibits attached to the registration statement for copies of the actual contract, agreement or other document.

Upon completion of this offering, we will be subject to the information requirements of the Exchange Act and will file annual, quarterly and current event reports, proxy statements and other information with the SEC. You can read our SEC filings, including the registration statement, over the Internet at the SEC's website at *www.sec.gov*.

AESTHER HEALTHCARE ACQUISITION CORP.

INDEX TO FINANCIAL STATEMENTS

	Page(s)
Report of Independent Registered Public Accounting Firm	F-1
Financial Statements:	
Balance Sheet as of July 12, 2021	F-2
Statement of Operations for the period from June 17, 2021 (inception) through July 12, 2021	F-3
Statement of Changes in Stockholders' Equity for the period from June 17, 2021 (inception) through July 12, 2021	F-4
Statement of Cash Flows for the period from June 17, 2021 (inception) through July 12, 2021	F-5
Notes to the Financial Statements	F-6 - F-13
160	

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Shareholders and Board of Directors of Aesther Healthcare Acquisition Corp.

Opinion on the Financial Statements

We have audited the accompanying balance sheet of Aesther Healthcare Acquisition Corp. (the "Company") as of July 12, 2021, and the related statements of operations, stockholders' equity, and cash flows for the period from June 17, 2021 (inception) through July 12, 2021, and the related notes (collectively referred to as the "financial statements"). In our opinion, the financial statements present fairly, in all material respects, the financial position of the Company as of July 12, 2021, and the results of its operations and its cash flows for the period from June 17, 2021 (inception) through July 12, 2021, in conformity with accounting principles generally accepted in the United States of America.

Going Concern Matter

The accompanying financial statements have been prepared assuming that the Company will continue as a going concern. As discussed in Note 1 to the financial statements, the Company has no revenue, its business plan is dependent on the completion of a financing transaction and the Company's cash and working capital are not sufficient to complete its planned activities for the upcoming year. These conditions raise substantial doubt about the Company's ability to continue as a going concern. Management's plans in regard to these matters are also described in Note 1. The financial statements do not include any adjustments that might result from the outcome of this uncertainty.

Basis for Opinion

These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on the Company's financial statements based on our audit. We are a public accounting firm registered with the Public Company Accounting Oversight Board (United States) ("PCAOB") and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audit in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement, whether due to error or fraud. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. As part of our audit we are required to obtain an understanding of internal control over financial reporting but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion.

Our audits included performing procedures to assess the risks of material misstatement of the financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements. Our audit also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that our audit provides a reasonable basis for our opinion.

/s/ MaloneBailey, LLP www.malonebailey.com We have served as the Company's auditor since 2021. Houston, Texas

July 16, 2021, except for footnotes 1 and 4, as to which the date is September 2, 2021.

AESTHER HEALTHCARE ACQUISITION CORP. BALANCE SHEET JULY 12, 2021

Assets:	
Cash	\$ 25,000
Deferred offering costs	 64,716
Total Assets	\$ 89,716
Liabilities and Stockholders' Equity	
Promissory note - related party	 64,798
Total current liabilities	64,798
Commitments and Contingencies (Note 6)	
Stockholders' Equity	
Preferred stock, \$0.0001 par value; 1,250,000 shares authorized; none issued and outstanding	-
Class A common stock, \$0.0001 par value; 125,000,000 shares authorized; none issued and outstanding	-
Class B common stock, \$0.0001 par value; 12,500,000 shares authorized; 2,875,000 shares issued and outstanding ⁽¹⁾	288
Additional paid-in capital	24,712
Accumulated deficit	 (82)
Total Stockholders' equity	24,918
Total Liabilities and Stockholders' Equity	\$ 89,716

(1) Includes up to 375,000 shares of Class B common stock subject to forfeiture if the over-allotment option is not exercised in full or in part by the underwriters (see Note 6).

The accompanying notes are an integral part of these financial statements.

AESTHER HEALTHCARE ACQUISITION CORP. STATEMENT OF OPERATIONS FOR THE PERIOD FROM JUNE 17, 2021 (INCEPTION) THROUGH JULY 12, 2021

Formation and operating costs	\$ 82
Net loss	\$ (82)
Basic and diluted weighted average shares outstanding ⁽¹⁾	2,500,000
Basic and diluted net loss per common share	\$ (0.00)

(1) Excludes up to 375,000 shares of Class B common stock subject to forfeiture if the over-allotment option is not exercised in full or in part by the underwriters (see Note 6).

The accompanying notes are an integral part of these financial statements.

AESTHER HEALTHCARE ACQUISITION CORP. STATEMENT OF CHANGES IN STOCKHOLDERS' EQUITY FOR THE PERIOD FROM JUNE 17, 2021 (INCEPTION) THROUGH JULY 12, 2021

	Class B Common Stock		Additional Paid-In		Accumulated		Stockholders'		
	Shares ⁽¹⁾	Amount		(Capital	D	eficit]	Equity
Balance as of June 17, 2021 (inception)		\$		\$	_	\$	_	\$	_
Class B common stock issued to Sponsor	2,875,000		288		24,712		_		25,000
Net loss	_		_		_		(82)		(82)
Balance as of July 12, 2021	2,875,000	\$	288	\$	24,712	\$	(82)	\$	24,918

(1) Includes up to 375,000 shares of Class B common stock subject to forfeiture if the over-allotment option is not exercised in full or in part by the underwriters (see Note 6).

The accompanying notes are an integral part of these financial statements.

AESTHER HEALTHCARE ACQUISITION CORP. STATEMENT OF CASH FLOWS FOR THE PERIOD FROM JUNE 17, 2021 (INCEPTION) THROUGH JULY 12, 2021

Cash flows from operating activities:	
Net loss	\$ (82)
Adjustments to reconcile net loss to net cash used in operating activities:	
Formation and operating costs paid	82
Net cash used in operating activities	_
Cash flows from financing activities:	
Proceeds from issuance of founder shares	25,000
Net cash provided by financing activities	25,000
Net change in cash	25,000
Cash, beginning of the period	
Cash, end of the period	\$ 25,000
Supplemental disclosure of cash flow information:	
Deferred offering costs paid by Sponsor under the promissory note	\$ 64,716
The accompanying notes are an integral part of these financial statements.	
F-5	

AESTHER HEALTHCARE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENTS

Note 1— Organization, Business Operations and Going Concern

Aesther Healthcare Acquisition Corp. (the "Company") is a newly organized blank check company formed for the purpose of effecting a merger, capital stock exchange, asset acquisition, stock purchase, reorganization or similar business combination with one or more businesses (the "Business Combination"). The Company has not selected any potential Business Combination target and the Company has not, nor has anyone on its behalf, initiated any substantive discussions, directly or indirectly, with any potential Business Combination target.

As of July 12, 2021, the Company had not commenced any operations. All activity for the period from June 17, 2021 (inception) through July 12, 2021 relates to the Company's formation and the Proposed Public Offering (as defined below). The Company will not generate any operating revenues until after the completion of its initial Business Combination, at the earliest. The Company will generate non-operating income in the form of interest income on cash and cash equivalents from the proceeds derived from the Proposed Public Offering (as defined below). The Company has selected December 31 as its fiscal year end.

The Company's sponsor is Aesther Healthcare Sponsor, LLC, a Delaware limited liability company (the "Sponsor"). The Company's ability to commence operations is contingent upon obtaining adequate financial resources through a proposed public offering of 10,000,000 units (the "Units" and, with respect to the Class A common stock included in the Units being offered, the "public shares") at \$10.00 per Unit (or 11,500,000 Units if the underwriters' overallotment option is exercised in full), which is discussed in Note 3 (the "Proposed Public Offering") and the sale of 5,261,000 warrants (or 5,711,000 warrants if the underwriters' over-allotment option is exercised on full) (the "Private Placement Warrants") at a price of \$1.00 per Private Placement Warrant in a private placement to the Sponsor that will close simultaneously with the Proposed Public Offering. The Company's management has broad discretion with respect to the specific application of the net proceeds of the Proposed Public Offering and sale of the Private Placement Warrants, although substantially all of the net proceeds are intended to be applied generally toward consummating a Business Combination.

The Company's Business Combination must be with one or more target businesses that together have a fair market value equal to at least 80% of the net balance in the Trust Account (as defined below) (excluding the amount of deferred underwriting discounts held and taxes payable on the income earned on the Trust Account) at the time of the signing an agreement to enter into a Business Combination. However, the Company will only complete a Business Combination if the post-Business Combination company owns or acquires 50% or more of the outstanding voting securities of the target or otherwise acquires a controlling interest in the target sufficient for it not to be required to register as an investment company under the Investment Company Act of 1940, as amended (the "Investment Company Act"). There is no assurance that the Company will be able to successfully effect a Business Combination.

Upon the closing of the Proposed Public Offering, management has agreed that \$10.20 per Unit sold in the Proposed Public Offering, including the proceeds of the sale of the Private Placement Warrants, will be held in a Trust Account ("Trust Account") and will be invested only in U.S. government securities with a maturity of 185 days or less or in money market funds meeting certain conditions under Rule 2a-7 under the Investment Company Act which invest only in direct U.S. government treasury obligations. Except with respect to interest earned on the funds held in the Trust Account that may be released to the Company to pay its franchise and income tax obligations (less up to \$100,000 of interest to pay dissolution expenses), the proceeds from the Proposed Public Offering and the sale of the Private Placement Warrants will not be released from the Trust Account until the earliest of (a) the completion of the initial Business Combination, (b) the redemption of any public shares properly submitted in connection with a stockholder vote to amend the Company's amended and restated certificate of incorporation (i) to modify the substance or timing of the Company's obligation to redeem 100% of the public shares if the Company does not complete the initial Business Combination within 12 months from the closing of the Proposed Public Offering (or up to 18 months, if the time to complete a business combination is extended as provided for in the amended and restated certificate of incorporation of the Company) or (ii) with respect to any other provision relating to stockholders' rights or pre-Business Combination activity, and (c) the redemption of the public shares if the Company is unable to complete the initial Business Combination within 18 months from the closing of the Proposed Public Offering, subject to applicable law. The proceeds deposited in the Trust Account could become subject to the claims of the Company's creditors, if any, which could have priority over the claims of the public stockholders.

The Company will provide its public stockholders with the opportunity to redeem all or a portion of their public shares upon the completion of the initial Business Combination either: (1) in connection with a stockholder meeting called to approve the Business Combination; or (2) by means of a tender offer. The decision as to whether the Company will seek stockholder approval of a proposed Business Combination or conduct a tender offer will be made by the Company, solely in its discretion, and will be based on a variety of factors such as the timing of the transaction and whether the terms of the transaction would require the Company to seek stockholder approval under applicable law or stock exchange listing requirement. The public stockholders will be entitled to redeem all or a portion of their public shares upon the completion of the initial Business Combination at a per share price, payable in cash, equal to the aggregate amount then on deposit in the Trust Account as of two business days prior to the consummation of the initial Business Combination, including interest earned on the funds held in the Trust Account and not previously released to the Company to pay its franchise and income taxes, divided by the number of then outstanding public shares, subject to the limitations described herein. The amount in the Trust Account is initially anticipated to be \$10.20 per public share.

The shares of common stock subject to redemption will be recorded at a redemption value and classified as temporary equity upon the completion of the Proposed Public Offering, in accordance with Financial Accounting Standards Board's ("FASB") Accounting Standards Codification ("ASC") Topic 480 "Distinguishing Liabilities from Equity." In such case, the Company will proceed with a Business Combination if the Company has net tangible assets of at least \$5,000,001 upon such consummation of a Business Combination and, if the Company seeks stockholder approval, a majority of the issued and outstanding shares voted are voted in favor of the Business Combination.

The Company will have only 12 months from the closing of the Proposed Public Offering (or up to 18 months, if the time to complete a business combination is extended as provided for in the amended and restated certificate of incorporation of the Company) (the "Combination Period") to complete the initial Business Combination. If the Company is unable to complete the initial Business Combination within the Combination Period, the Company will: (1) cease all operations except for the purpose of winding up, (ii) as promptly as reasonably possible but not more than ten business days thereafter, redeem the public shares, at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the Trust Account including interest earned on the funds held in the Trust Account and not previously released to the Company to pay its franchise and income taxes (less up to \$100,000 of interest to pay dissolution expenses), divided by the number of then outstanding public shares, which redemption will completely extinguish public stockholders' rights as stockholders (including the right to receive further liquidating distributions, if any), subject to applicable law, and (iii) as promptly as reasonably possible following such redemption, subject to the approval of the Company's remaining stockholders and the board of directors, dissolve and liquidate, subject in each case to the Company's obligations under Delaware law to provide for claims of creditors and the requirements of other applicable law.

The Sponsor, officers and directors have entered into a letter agreement with the Company, pursuant to which they have agree: (1) waive their redemption rights with respect to their founder shares and public shares in connection with the completion of the initial Business Combination, (ii) waive their redemption rights with respect to their founder shares and public shares in connection with a stockholder vote to approve an amendment to the Company's amended and restated certificate of incorporation (A) to modify the substance or timing of the Company's obligation to redeem 100% of the public shares if the Company does not complete the initial Business Combination within the Combination Period or (B) with respect to any other provision relating to stockholders' rights or pre-initial Business Combination activity and (iii) waive their rights to liquidating distributions from the Trust Account with respect to their founder shares if the Company fails to complete the initial Business Combination within the Combination Period, although they will be entitled to liquidating distributions from the Trust Account with respect to any public shares they hold if we fail to complete the initial Business Combination within the Combination Period.

The Sponsor has agreed that it will be liable to the Company if and to the extent any claims by a third party for services rendered or products sold to the Company, or a prospective target business with which the Company has entered into a written letter of intent, confidentiality or similar agreement or Business Combination agreement, reduce the amount of funds in the Trust Account to below the lesser of (i) \$10.20 per public share and (ii) the actual amount per public share held in the Trust Account as of the date of the liquidation of the Trust Account, if less than \$10.20 per share due to reductions in the value of the trust assets, less taxes payable, provided that such liability will not apply to any claims by a third party or prospective target business who executed a waiver of any and all rights to the monies held in the Trust Account (whether or not such waiver is enforceable) nor will it apply to any claims under the Company's indemnity of the underwriters of the Proposed Public Offering against certain liabilities, including liabilities under the Securities Act. However, the Company has not asked the Sponsor to reserve for such indemnification obligations, nor have we independently verified whether the Sponsor has sufficient funds to satisfy its indemnity obligations and believe that the Sponsor's only assets are securities of the Company. Therefore, the Company cannot assure you that the Sponsor would be able to satisfy those obligations.

Going Concern Consideration

As of July 12, 2021, the Company had \$25,000 in cash and a working capital deficit of \$39,798 (excluding deferred offering costs). The Company has incurred and expects to continue to incur significant costs in pursuit of its financing and acquisition plans. These conditions raise substantial doubt about the Company's ability to continue as a going concern. Management plans to address this uncertainty through a Proposed Public Offering as discussed in Note 3. There is no assurance that the Company's plans to raise capital or to consummate a Business Combination will be successful within the Combination Period. The financial statements do not include any adjustments that might result from the outcome of this uncertainty.

Risks and Uncertainties

Management is currently evaluating the impact of the COVID-19 pandemic and has concluded that while it is reasonably possible that the virus could have a negative effect on the Company's financial position, results of its operations and/or search for a target company, the specific impact is not readily determinable as of the date of these financial statements. The financial statements do not include any adjustments that might result from the outcome of this uncertainty.

Note 2— Significant Accounting Policies

Basis of Presentation

The accompanying financial statements are presented in conformity with accounting principles generally accepted in the United States of America ("US GAAP") and pursuant to the rules and regulations of the U.S. Securities and Exchange Commission (the "SEC").

Emerging Growth Company Status

The Company is an "emerging growth company," as defined in Section 2(a) of the Securities Act, as modified by the Jumpstart Our Business Startups Act of 2012, (the "JOBS Act"), and it may take advantage of certain exemptions from various reporting requirements that are applicable to other public companies that are not emerging growth companies including, but not limited to, not being required to comply with the auditor attestation requirements of Section 404 of the Sarbanes-Oxley Act, reduced disclosure obligations regarding executive compensation in its periodic reports and proxy statements, and exemptions from the requirements of holding a nonbinding advisory vote on executive compensation and stockholder approval of any golden parachute payments not previously approved.

Further, Section 102(b)(1) of the JOBS Act exempts emerging growth companies from being required to comply with new or revised financial accounting standards until private companies (that is, those that have not had a Securities Act registration statement declared effective or do not have a class of securities registered under the Exchange Act) are required to comply with the new or revised financial accounting standards. The JOBS Act provides that a company can elect to opt out of the extended transition period and comply with the requirements that apply to non-emerging growth companies but any such election to opt out is irrevocable. The Company has elected not to opt out of such extended transition period which means that when a standard is issued or revised and it has different application dates for public or private companies, the Company, as an emerging growth company, can adopt the new or revised standard at the time private companies adopt the new or revised standard. This may make comparison of the Company's financial statements with another public company which is neither an emerging growth company nor an emerging growth company which has opted out of using the extended transition period difficult or impossible because of the potential differences in accounting standards used.

Use of Estimates

The preparation of financial statements in conformity with US GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of expenses during the reporting period. Actual results could differ from those estimates.

Cash and Cash Equivalents

The Company considers all short-term investments with an original maturity of three months or less when purchased to be cash equivalents. The Company did not have cash equivalents as of July 12, 2021.

Deferred Offering Costs

Deferred offering costs consist of accounting and legal expenses incurred through the balance sheet date that are directly related to the Proposed Public Offering and that will be charged to stockholders' equity upon the completion of the Proposed Public Offering. Should the Proposed Public Offering prove to be unsuccessful, these deferred costs, as well as additional expenses to be incurred, will be charged to operations.

Fair Value of Financial Instruments

The fair value of the Company's assets and liabilities, which qualify as financial instruments under the FASB ASC 820, "Fair Value Measurements and Disclosures," approximates the carrying amounts represented in the balance sheet, primarily due to its short-term nature.

Derivative Financial Instruments

The Company evaluates its financial instruments to determine if such instruments are derivatives or contain features that qualify as embedded derivatives in accordance with ASC Topic 815, "Derivatives and Hedging". For derivative financial instruments that are accounted for as liabilities, the derivative instrument is initially recorded at its fair value on the grant date and is then re-valued at each reporting date, with changes in the fair value reported in the statements of operations. The classification of derivative instruments, including whether such instruments should be recorded as liabilities or as equity, is evaluated at the end of each reporting period. Derivative liabilities are classified in the balance sheet as current or non-current based on whether or not net-cash settlement or conversion of the instrument could be required within 12 months of the balance sheet date.

Net Loss Per Common Share

The Company complies with the accounting and disclosure requirements of FASB ASC Topic 260, "Earnings Per Share." Net loss per common share is computed by dividing net loss by the weighted average number of shares of common stock outstanding during the period, excluding common stock subject to forfeiture. Weighted average shares were reduced for the effect of an aggregate of 375,000 shares of common stock that are subject to forfeiture if the over-allotment option is not exercised by the underwriters (see Note 6). At July 12, 2021, the Company did not have any dilutive securities and other contracts that could, potentially, be exercised or converted into common stock and then share in the earnings of the Company. As a result, diluted loss per common share is the same as basic loss per common share for the period presented.

Income Taxes

The Company accounts for income taxes under FASB ASC 740, "Income Taxes" ("ASC 740"). ASC 740 requires the recognition of deferred tax assets and liabilities for both the expected impact of differences between the financial statement and tax basis of assets and liabilities and for the expected future tax benefit to be derived from tax loss and tax credit carry forwards. ASC 740 additionally requires a valuation allowance to be established when it is more likely than not that all or a portion of deferred tax assets will not be realized.

ASC 740 also clarifies the accounting for uncertainty in income taxes recognized in an enterprise's financial statements and prescribes a recognition threshold and measurement process for financial statement recognition and measurement of a tax position taken or expected to be taken in a tax return. For those benefits to be recognized, a tax position must be more-likely-than-not to be sustained upon examination by taxing authorities. ASC 740 also provides guidance on derecognition, classification, interest and penalties, accounting in interim period, disclosure and transition.

The Company recognizes accrued interest and penalties related to unrecognized tax benefits as income tax expense. There were no unrecognized tax benefits and no amounts accrued for interest and penalties as of July 12, 2021. The Company is currently not aware of any issues under review that could result in significant payments, accruals or material deviation from its position.

The Company has identified the United States as its only "major" tax jurisdiction.

The Company is subject to income tax examinations by major taxing authorities since inception. These examinations may include questioning the timing and amount of deductions, the nexus of income among various tax jurisdictions and compliance with federal and state tax laws. The Company's management does not expect that the total amount of unrecognized tax benefits will materially change over the next twelve months.

The provision for income taxes was deemed to be immaterial for the period from June 17, 2021(inception) through July 12, 2021.

Recent Accounting Standards

Management does not believe that any recently issued, but not effective, accounting standards, if currently adopted, would have a material effect on the Company's financial statements.

Note 3— Proposed Public Offering

Pursuant to the Proposed Public Offering, the Company will offer for sale 10,000,000 Units (or 11,500,000 Units if the underwriters' overallotment option is exercised in full) at a purchase price of \$10.00 per Unit. Each Unit will consist of one share of Class A common stock and one—half of one warrant ("public warrant"). Each whole public warrant will entitle the holder to purchase one share of Class A common stock at an exercise price of \$11.50 per whole share, subject to adjustment (see Note 6).

Note 4 — Related Party Transactions

Founder Shares

On June 30, 2021, the Sponsor paid \$25,000 in cash consideration for 2,875,000 Class B shares (the "founder shares"). The number of founder shares outstanding was determined based on the expectation that the total size of the Proposed Public Offering would be a maximum of 11,500,000 units if the underwriters' over-allotment option is exercised in full, and therefore that such founder shares would represent 20% of the outstanding shares after the Proposed Public Offering, excluding the representative's shares. Up to 375,000 of the founder shares will be forfeited depending on the extent to which the underwriters' over-allotment option is exercised.

The Company' initial stockholders have agreed not to transfer, assign or sell any of their founder shares until the earlier to occur of: (i) one year after the date of the consummation of the initial Business Combination or (ii) the date on which the Company consummates a liquidation, merger, stock exchange or other similar transaction which results in all of the stockholders having the right to exchange their shares of Class A common stock for cash, securities or other property. Any permitted transferees will be subject to the same restrictions and other agreements of the initial stockholders with respect to any founder shares. Notwithstanding the foregoing, if the closing price of the shares of Class A common stock equals or exceeds \$12.00 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like) for any 20 trading days within any 30-trading day period commencing 150 days after the initial Business Combination, the founder shares will no longer be subject to such transfer restrictions.

The Company will have until 12 months from the closing of the Proposed Public Offering to consummate its initial Business Combination. However, if the Company anticipates that it may not be able to consummate its initial Business Combination within 12 months, it may, by resolution of its board if requested by the Sponsor, extend the period of time to consummate a Business Combination up to two times, each by an additional three months (for a total of up to 18 months to complete a Business Combination), subject to the Sponsor depositing additional funds into the Trust Account. In order to extend the time available for the Company to consummate its initial Business Combination, the Sponsor or its affiliates or designees, upon five days advance notice prior to the applicable deadline, must deposit into the Trust Account for each three-month extension, \$1,000,000, or \$1,150,000 if the underwriters' overallotment option is exercised in full (\$0.10 per share in either case) on or prior to the date of the applicable deadline, up to an aggregate of \$2,000,000 (or \$2,300,000 if the underwriters' over-allotment option is exercised in full), or approximately \$0.20 per share. Any such payments would be made in the form of a loan. Any such loans will be non-interest bearing and payable upon the consummation of the Company's initial Business Combination. If the Company completes an initial Business Combination, it would repay such loaned amounts out of the proceeds of the Trust Account released to it. If the Company does not complete a Business Combination, we will not repay such loans. Furthermore, the letter agreement with the Company's initial stockholders contains a provision pursuant to which the Sponsor has agreed to waive its right to be repaid for such loans out of the funds held in the Trust Account in the event that the Company does not complete a Business Combination. In the event that the Company receives notice from the Sponsor five days prior to the applicable deadline of its wish for the Company to effect an extension, the Company intends to issue a press release announcing such intention at least three days prior to the applicable deadline. In addition, the Company intends to issue a press release the day after the applicable deadline announcing whether or not the funds had been timely deposited. The Sponsor and its affiliates or designees are not obligated to fund the Trust Account to extend the time for the Company to complete its initial Business Combination.

Promissory Note — Related Party

On June 30, 2021, the Sponsor agreed to loan the Company up to \$300,000 to be used for a portion of the expenses of the Proposed Public Offering. These loans are non-interest bearing, unsecured and due at the earlier of June 30, 2022 or the closing of the Proposed Public Offering. These loans will be repaid upon the closing of the Proposed Public Offering out of the \$1,261,000 of offering proceeds that has been allocated to the payment of offering expenses. As of July 12, 2021, the Company had borrowed \$64,798 under the promissory note.

Related Party Loans

In order to finance transaction costs in connection with an intended initial Business Combination, the Sponsor, an affiliate of the Sponsor or certain of the Company's officers and directors may, but is not obligated to, loan the Company funds as may be required (the "Working Capital Loans"). If the Company completes an initial Business Combination, the Company would repay such loaned amounts out of the proceeds of the Trust Account released to the Company. Otherwise, such loans would be repaid only out of funds held outside the Trust Account. In the event that the initial Business Combination does not close, the Company may use a portion of the working capital held outside the Trust Account to repay such loaned amounts but no proceeds from the Trust Account would be used to repay such loaned amounts. Up to \$1,500,000 of such loans may be convertible into Private Placement Warrants of the post Business Combination entity, at a price of \$1.00 per warrant at the option of the lender. The warrants would be identical to the Private Placement Warrants issued to the Sponsor. At July 12, 2021, no such Working Capital Loans were outstanding.

Administrative Support Agreement

The Company has agreed to pay Aesther Healthcare Sponsor, LLC, our sponsor a total of \$10,000 per month for office space, utilities and secretarial and administrative support. The administrative support agreement will begin on that date of this prospectus and continue monthly until (i) the completion of the Company's initial Business Combination or (ii) liquidation of the Company.

Note 5— Commitments and Contingencies

Registration Rights

The holders of the founder shares, Private Placement Warrants and warrants that may be issued upon conversion of Working Capital Loans (and any shares of common stock issuable upon the exercise of the Private Placement Warrants or warrants issued upon conversion of the working capital loans and upon conversion of the founder shares) will be entitled to registration rights pursuant to a registration rights agreement to be signed prior to or on the effective date of the Proposed Public Offering requiring the Company to register such securities for resale (in the case of the founder shares, only after conversion to shares of Class A common stock). The holders of these securities will be entitled to make up to three demands, excluding short form registration demands, that the Company registers such securities. In addition, the holders have certain "piggy-back" registration rights with respect to registration statements filed subsequent to the Company's completion of the initial Business Combination and rights to require the Company to register for resale such securities pursuant to Rule 415 under the Securities Act. The Company will bear the expenses incurred in connection with the filing of any such registration statements.

Underwriters Agreement

The Company will grant the underwriters a 45-day option to purchase up to 1,500,000 additional Units to cover any over-allotments, if any, at the Proposed Public Offering price less the underwriting discounts and commissions.

The underwriters will be entitled to a cash underwriting discount of one percent (1%) of the gross proceeds of the Proposed Public Offering, or \$1,000,000 (or up to \$1,150,000 if the underwriters' over-allotment option is exercised in full). Additionally, the underwriters will be entitled to 100,000 shares of common stock and a deferred underwriting discount of 3.0% of the gross proceeds of the Proposed Public Offering, or \$3,000,000 (or up to \$3,450,000 if the underwriters' over-allotment option is exercised in full) held in the Trust Account upon the completion of the Company's initial Business Combination subject to the terms of the underwriting agreement.

Representative's Class A Common Stock

The 100,000 shares issuable to the representative have been deemed compensation by FINRA and are therefore subject to a lock-up for a period of 180 days immediately following the commencement of sales of this offering pursuant to Rule 5110(e)(1) of FINRA's Rules. Pursuant to FINRA Rule 5110(e) (1), these securities will not be the subject of any hedging, short sale, derivative, put or call transaction that would result in the economic disposition of the securities by any person for a period of 180 days immediately following the effective date of the registration statement of which this prospectus forms a part, nor may they be sold, transferred, assigned, pledged or hypothecated for a period of 180 days immediately following the commencement of sales of this offering except as permitted by FINRA Rule 5110(e)(2).

Note 6— Stockholders' Equity

Preferred Stock

The Company is authorized to issue 1,250,000 shares of preferred stock with a par value of \$0.0001 per share. At July 12, 2021, there were no shares of preferred stock issued or outstanding.

Class A Common Stock

The Company is authorized to issue 125,000,000 shares of Class A common stock with a par value of \$0.0001 per share. Holders of Class A common stock are entitled to one vote for each share. At July 12, 2021, there were no shares of Class A common stock issued or outstanding.

Class B Common Stock

The Company is authorized to issue 12,500,000 shares of Class B common stock with a par value of \$0.0001 per share. Holders of the Class B common stock are entitled to one vote for each common stock. At July 12, 2021, there were 2,875,000 shares of Class B common stock issued and outstanding, of which 375,000 shares are subject to forfeiture to the extent that the underwriters' over-allotment option is not exercised in full so that the founder shares will represent, on an as-converted basis, 20% of the Company's issued and outstanding shares after the Proposed Public Offering (assuming the Sponsor does not purchase any public shares in the Proposed Public Offering and excluding the representative's shares).

The Company' initial stockholders have agreed not to transfer, assign or sell any of their founder shares until the earlier to occur of (i) one year after the date of the consummation of the initial Business Combination or (ii) the date on which the Company consummates a liquidation, merger, stock exchange or other similar transaction which results in all of the stockholders having the right to exchange their shares of Class A common stock for cash, securities or other property (except as described herein under the section of this prospectus entitled "Principal Stockholders — Restrictions on Transfers of Founder Shares and Private Placement Warrants"). Any permitted transferees will be subject to the same restrictions and other agreements of the initial stockholders with respect to any founder shares. Notwithstanding the foregoing, if the closing price of the shares of Class A common stock equals or exceeds \$12.00 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like) for any 20 trading days within any 30-trading day period commencing 150 days after the initial Business Combination, the founder shares will no longer be subject to the Lock-up.

The shares of Class B common stock will automatically convert into shares of Class A common stock at the time of the initial Business Combination on a one-for-one basis, subject to adjustment for stock splits, stock dividends, reorganizations, recapitalizations and the like, and subject to further adjustment as provided herein. In the case that additional shares of Class A common stock, or equity-linked securities, are issued or deemed issued in excess of the amounts offered in this prospectus and related to the closing of the initial Business Combination, the ratio at which shares of Class B common stock shall convert into shares of Class A common stock will be adjusted (unless the holders of a majority of the outstanding shares of Class B common stock agree to waive such adjustment with respect to any such issuance or deemed issuance) so that the number of shares of Class A common stock issuable upon conversion of all shares of Class B common stock will equal, in the aggregate, on an as-converted basis, 20% of the sum of the total number of all shares of common stock outstanding upon the completion of the Proposed Public Offering (not including the shares of Class A common stock issuable to EF Hutton) plus all shares of Class A common stock and equity-linked securities issued or deemed issued in connection with the initial Business Combination (excluding any shares or equity-linked securities issued, or to be issued, to any seller in the initial Business Combination or any private placement-equivalent units issued to the Sponsor, its affiliates or certain of the Company's officers and directors upon conversion of Working Capital Loans made to the Company, and excluding the representative's shares).

Holders of the Class A common stock and holders of the Class B common stock will vote together as a single class on all matters submitted to a vote of the Company's stockholders, with each share of common stock entitling the holder to one vote.

Warrants

Warrants may only be exercised for a whole number of shares. No fractional shares will be issued upon exercise of the Warrants. The Warrants will become exercisable on the later of (a) 30 days after the consummation of a Business Combination or (b) 12 months from the closing of the Business Combination. No Warrants will be exercisable for cash unless the Company has an effective and current registration statement covering the common stock issuable upon exercise of the Warrants and a current prospectus relating to such common stock. Notwithstanding the foregoing, if a registration statement covering the common stock issuable upon the exercise of the Warrants is not effective within 90 days from the consummation of a Business Combination, the holders may, until such time as there is an effective registration statement and during any period when the Company shall have failed to maintain an effective registration statement, exercise the Warrants on a cashless basis pursuant to an available exemption from registration under the Securities Act. If an exemption from registration is not available, holders will not be able to exercise their Warrants on a cashless basis. The Warrants will expire five years from the consummation of a Business Combination or earlier upon redemption or liquidation.

The Company may call the Warrants for redemption, in whole and not in part, at a price of \$0.01 per warrant:

- at any time while the Warrants are exercisable,
- upon not less than 30 days' prior written notice of redemption to each Warrant holder,
- if, and only if, the reported last sale price of the common stock equals or exceeds \$18 per share, for any 20 trading days within a 30-trading day period ending on the third trading day prior to the notice of redemption to Warrant holders, and
- if, and only if, there is a current registration statement in effect with respect to the common stock underlying such warrants at the time of redemption and for the entire 30-day trading period referred to above and continuing each day thereafter until the date of redemption.

The Placement Warrants will be identical to the Public Warrants underlying the Units being sold in the Proposed Offering, except that the Placement Warrants and the common stock issuable upon the exercise of the Placement Warrants will not be transferable, assignable or salable until 30 days after the completion of a Business Combination, subject to certain limited exceptions.

If the Company calls the Warrants for redemption, management will have the option to require all holders that wish to exercise the Warrants to do so on a "cashless basis," as described in the warrant agreement. The exercise price and number of common stock issuable upon exercise of the warrants may be adjusted in certain circumstances including in the event of a stock dividend, extraordinary dividend or recapitalization, reorganization, merger, or consolidation. However, the warrants will not be adjusted for issuances of common stock at a price below its exercise price. Additionally, in no event will the Company be required to net cash settle the warrants. If the Company is unable to complete a Business Combination within the Combination Period and the Company liquidates the funds held in the Trust Account, holders of warrants will not receive any of such funds with respect to their warrants, nor will they receive any distribution from the Company's assets held outside of the Trust Account with respect to such warrants. Accordingly, the warrants may expire worthless.

The exercise price is \$11.50 per share, subject to adjustment as described herein. In addition, if (x) if the Company issues additional shares of Class A common stock or equity-linked securities for capital raising purposes in connection with the closing of our initial business combination at an issue price or effective issue price of less than \$9.20 per share of Class A common stock (with such issue price or effective issue price to be determined in good faith by our board of directors and, in the case of any such issuance to our sponsor or its affiliates, without taking into account any founder shares held by our sponsor or such affiliates, as applicable, prior to such issuance) (the "Newly Issued Price"), (y) the aggregate gross proceeds from such issuances represent more than 60% of the total equity proceeds, and interest thereon, available for the funding of our initial business combination on the date of the consummation of our initial business combination (net of redemptions), and (z) the volume weighted average trading price of our Class A common stock during the 20 trading day period starting on the trading day prior to the day on which we consummate our initial business combination (such price, the "Market Value") is below \$9.20 per share, then the exercise price of the warrants will be adjusted (to the nearest cent) to be equal to 115% of the greater of the Market Value and the Newly Issued Price, and the \$18.00 per share redemption trigger price described below under "Redemption of warrants" will be adjusted (to the nearest cent) to be equal to 180% of the greater of the Market Value and the Newly Issued Price.

Note 7 — Subsequent Events

The Company evaluated subsequent events and transactions that occurred after the balance sheet date up to through July 16, 2021 the date that the financial statements were issued. Other than as described below, the Company did not identify any subsequent events that would have required adjustment or disclosure in the financial statements.



Aesther Healthcare Acquisition Corp.

PRELIMINARY PROSPECTUS

, 2021

EF HUTTON

division of Benchmark Investments, LLC

Until , 2021 (25 days after the date of this prospectus), all dealers that buy, sell or trade our Class A common stock, whether or not participating in this offering, may be required to deliver a prospectus. This is in addition to the dealers' obligation to deliver a prospectus when acting as underwriters and with respect to its unsold allotments or subscriptions.

You should rely only on the information contained in this prospectus. We have not, and the underwriters have not, authorized anyone to provide you with different information. If anyone provides you with different or inconsistent information, you should not rely on it. We are not, and the underwriters are not, making an offer to sell securities in any jurisdiction where the offer or sale is not permitted. You should not assume that the information contained in this prospectus is accurate as of any date other than the date on the front of this prospectus.

No dealer, salesperson or any other person is authorized to give any information or make any representations in connection with this offering other than those contained in this prospectus and, if given or made, the information or representations must not be relied upon as having been authorized by us. This prospectus does not constitute an offer to sell or a solicitation of an offer to buy any security other than the securities offered by this prospectus, or an offer to sell or a solicitation of an offer to buy any securities by anyone in any jurisdiction in which the offer or solicitation is not authorized or is unlawful.

PART II

INFORMATION NOT REQUIRED IN PROSPECTUS

Item 13. Other Expenses of Issuance and Distribution.

The estimated approximate expenses payable by us in connection with the offering described in this registration statement (other than the underwriting discount and commissions) will be as follows:

SEC/FINRA Expenses	31,204
Accounting fees and expenses	62,000
Printing, road show and engraving expenses	50,000
Reimbursement to underwriters for expenses	100,000
Directors & officers liability insurance premiums ⁽¹⁾	650,000
Legal fees and expenses	250,000
Nasdaq listing and filing fees	75,000
Miscellaneous	43,906
Total	\$ 1,261,000

(1) This amount represents the approximate amount of annual director and officer liability insurance premiums the registrant anticipates paying following the completion of its initial public offering and until it completes an initial business combination.

Item 14. Indemnification of Directors and officers.

Our amended and restated certificate of incorporation will provide that all of our directors, officers, employees and agents shall be entitled to be indemnified by us to the fullest extent permitted by Section 145 of the Delaware General Corporation Law ("DGCL"). Section 145 of the Delaware General Corporation Law concerning indemnification of officers, directors, employees and agents is set forth below.

Section 145. Indemnification of officers, directors, employees and agents; insurance.

- (a) A corporation shall have power to indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (other than an action by or in the right of the corporation) by reason of the fact that the person is or was a director, officer, employee or agent of the corporation, or is or was serving at the request of the corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise, against expenses (including attorneys' fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by the person in connection with such action, suit or proceeding if the person acted in good faith and in a manner the person reasonably believed to be in or not opposed to the best interests of the corporation, and, with respect to any criminal action or proceeding, had no reasonable cause to believe the person reasonably believed to be in or not opposed to the best interests of the corporation, suit or proceeding by judgment, order, settlement, conviction, or upon a plea of nolo contendere or its equivalent, shall not, of itself, create a presumption that the person did not act in good faith and in a manner which the person reasonably believed to be in or not opposed to the best interests of the corporation, and, with respect to any criminal action or proceeding, had reasonable cause to believe that the person's conduct was unlawful.
- (b) A corporation shall have power to indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action or suit by or in the right of the corporation to procure a judgment in its favor by reason of the fact that the person is or was a director, officer, employee or agent of the corporation, or is or was serving at the request of the corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise against expenses (including attorneys' fees) actually and reasonably incurred by the person in connection with the defense or settlement of such action or suit if the person acted in good faith and in a manner the person reasonably believed to be in or not opposed to the best interests of the corporation and except that no indemnification shall be made in respect of any claim, issue or matter as to which such person shall have been adjudged to be liable to the corporation unless and only to the extent that the Court of Chancery or the court in which such action or suit was brought shall determine upon application that, despite the adjudication of liability but in view of all the circumstances of the case, such person is fairly and reasonably entitled to indemnity for such expenses which the Court of Chancery or such other court shall deem proper.

- (c) To the extent that a present or former director or officer of a corporation has been successful on the merits or otherwise in defense of any action, suit or proceeding referred to in subsections (a) and (b) of this section, or in defense of any claim, issue or matter therein, such person shall be indemnified against expenses (including attorneys' fees) actually and reasonably incurred by such person in connection therewith.
- (d) Any indemnification under subsections (a) and (b) of this section (unless ordered by a court) shall be made by the corporation only as authorized in the specific case upon a determination that indemnification of the present or former director, officer, employee or agent is proper in the circumstances because the person has met the applicable standard of conduct set forth in subsections (a) and (b) of this section. Such determination shall be made, with respect to a person who is a director or officer at the time of such determination, (1) by a majority vote of the directors who are not parties to such action, suit or proceeding, even though less than a quorum, or (2) by a committee of such directors designated by majority vote of such directors, even though less than a quorum, or (3) if there are no such directors, or if such directors so direct, by independent legal counsel in a written opinion, or (4) by the stockholders.
- (e) Expenses (including attorneys' fees) incurred by an officer or director in defending any civil, criminal, administrative or investigative action, suit or proceeding may be paid by the corporation in advance of the final disposition of such action, suit or proceeding upon receipt of an undertaking by or on behalf of such director or officer to repay such amount if it shall ultimately be determined that such person is not entitled to be indemnified by the corporation as authorized in this section. Such expenses (including attorneys' fees) incurred by former officers and directors or other employees and agents may be so paid upon such terms and conditions, if any, as the corporation deems appropriate.
- (f) The indemnification and advancement of expenses provided by, or granted pursuant to, the other subsections of this section shall not be deemed exclusive of any other rights to which those seeking indemnification or advancement of expenses may be entitled under any bylaw, agreement, vote of stockholders or disinterested directors or otherwise, both as to action in such person's official capacity and as to action in another capacity while holding such office. A right to indemnification or to advancement of expenses arising under a provision of the certificate of incorporation or a bylaw shall not be eliminated or impaired by an amendment to such provision after the occurrence of the act or omission that is the subject of the civil, criminal, administrative or investigative action, suit or proceeding for which indemnification or advancement of expenses is sought, unless the provision in effect at the time of such act or omission explicitly authorizes such elimination or impairment after such action or omission has occurred.
- (g) A corporation shall have power to purchase and maintain insurance on behalf of any person who is or was a director, officer, employee or agent of the corporation, or is or was serving at the request of the corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise against any liability asserted against such person and incurred by such person in any such capacity, or arising out of such person's status as such, whether or not the corporation would have the power to indemnify such person against such liability under this section.
- (h) For purposes of this section, references to "the corporation" shall include, in addition to the resulting corporation, any constituent corporation (including any constituent of a constituent) absorbed in a consolidation or merger which, if its separate existence had continued, would have had power and authority to indemnify its directors, officers, and employees or agents, so that any person who is or was a director, officer, employee or agent of such constituent corporation, or is or was serving at the request of such constituent corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise, shall stand in the same position under this section with respect to the resulting or surviving corporation as such person would have with respect to such constituent corporation if its separate existence had continued.

- (i) For purposes of this section, references to "other enterprises" shall include employee benefit plans; references to "fines" shall include any excise taxes assessed on a person with respect to any employee benefit plan; and references to "serving at the request of the corporation" shall include any service as a director, officer, employee or agent of the corporation which imposes duties on, or involves services by, such director, officer, employee or agent with respect to an employee benefit plan, its participants or beneficiaries; and a person who acted in good faith and in a manner such person reasonably believed to be in the interest of the participants and beneficiaries of an employee benefit plan shall be deemed to have acted in a manner "not opposed to the best interests of the corporation" as referred to in this section.
- (j) The indemnification and advancement of expenses provided by, or granted pursuant to, this section shall, unless otherwise provided when authorized or ratified, continue as to a person who has ceased to be a director, officer, employee or agent and shall inure to the benefit of the heirs, executors and administrators of such a person.
- (k) The Court of Chancery is hereby vested with exclusive jurisdiction to hear and determine all actions for advancement of expenses or indemnification brought under this section or under any by law, agreement, vote of stockholders or disinterested directors, or otherwise. The Court of Chancery may summarily determine a corporation's obligation to advance expenses (including attorneys' fees).

Insofar as indemnification for liabilities arising under the Securities Act may be permitted to our directors, officers, and controlling persons pursuant to the foregoing provisions, or otherwise, we have been advised that, in the opinion of the SEC, such indemnification is against public policy as expressed in the Securities Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment of expenses incurred or paid by a director, officer or controlling person in a successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, we will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to the court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Securities Act and will be governed by the final adjudication of such issue.

In accordance with Section 102(b)(7) of the DGCL, our amended and restated certificate of incorporation, will provide that no director shall be personally liable to us or any of our stockholders for monetary damages resulting from breaches of their fiduciary duty as directors, except to the extent such limitation on or exemption from liability is not permitted under the DGCL. The effect of this provision of our amended and restated certificate of incorporation is to eliminate our rights and those of our stockholders (through stockholders' derivative suits on our behalf) to recover monetary damages against a director for breach of the fiduciary duty of care as a director, including breaches resulting from negligent or grossly negligent behavior, except, as restricted by Section 102(b)(7) of the DGCL. However, this provision does not limit or eliminate our rights or the rights of any stockholder to seek non-monetary relief, such as an injunction or rescission, in the event of a breach of a director's duty of care.

If the DGCL is amended to authorize corporate action further eliminating or limiting the liability of directors, then, in accordance with our amended and restated certificate of incorporation, the liability of our directors to us or our stockholders will be eliminated or limited to the fullest extent authorized by the DGCL, as so amended. Any repeal or amendment of provisions of our amended and restated certificate of incorporation limiting or eliminating the liability of directors, whether by our stockholders or by changes in law, or the adoption of any other provisions inconsistent therewith, will (unless otherwise required by law) be prospective only, except to the extent such amendment or change in law permits us to further limit or eliminate the liability of directors on a retroactive basis.

Our amended and restated certificate of incorporation will also provide that we will, to the fullest extent authorized or permitted by applicable law, indemnify our current and former officers and directors, as well as those persons who, while directors or officers of our corporation, are or were serving as directors, officers, employees or agents of another entity, trust or other enterprise, including service with respect to an employee benefit plan, in connection with any threatened, pending or completed proceeding, whether civil, criminal, administrative or investigative, against all expense, liability and loss (including, without limitation, attorney's fees, judgments, fines, ERISA excise taxes and penalties and amounts paid in settlement) reasonably incurred or suffered by any such person in connection with any such proceeding.

Notwithstanding the foregoing, a person eligible for indemnification pursuant to our amended and restated certificate of incorporation will be indemnified by us in connection with a proceeding initiated by such person only if such proceeding was authorized by our board of directors, except for proceedings to enforce rights to indemnification.

The right to indemnification which will be conferred by our amended and restated certificate of incorporation is a contract right that includes the right to be paid by us the expenses incurred in defending or otherwise participating in any proceeding referenced above in advance of its final disposition, provided, however, that if the DGCL requires, an advancement of expenses incurred by our officer or director (solely in the capacity as an officer or director of our corporation) will be made only upon delivery to us of an undertaking, by or on behalf of such officer or director, to repay all amounts so advanced if it is ultimately determined that such person is not entitled to be indemnified for such expenses under our amended and restated certificate of incorporation or otherwise.

The rights to indemnification and advancement of expenses will not be deemed exclusive of any other rights which any person covered by our amended and restated certificate of incorporation may have or hereafter acquire under law, our amended and restated certificate of incorporation, our bylaws, an agreement, vote of stockholders or disinterested directors, or otherwise.

Any repeal or amendment of provisions of our amended and restated certificate of incorporation affecting indemnification rights, whether by our stockholders or by changes in law, or the adoption of any other provisions inconsistent therewith, will (unless otherwise required by law) be prospective only, except to the extent such amendment or change in law permits us to provide broader indemnification rights on a retroactive basis, and will not in any way diminish or adversely affect any right or protection existing at the time of such repeal or amendment or adoption of such inconsistent provision with respect to any act or omission occurring prior to such repeal or amendment or adoption of such inconsistent provision. Our amended and restated certificate of incorporation will also permit us, to the extent and in the manner authorized or permitted by law, to indemnify and to advance expenses to persons other that those specifically covered by our amended and restated certificate of incorporation.

Our bylaws include the provisions relating to advancement of expenses and indemnification rights consistent with those which will be set forth in our amended and restated certificate of incorporation. In addition, our bylaws provide for a right of indemnity to bring a suit in the event a claim for indemnification or advancement of expenses is not paid in full by us within a specified period of time. Our bylaws also permit us to purchase and maintain insurance, at our expense, to protect us and/or any director, officer, employee or agent of our corporation or another entity, trust or other enterprise against any expense, liability or loss, whether or not we would have the power to indemnify such person against such expense, liability or loss under the DGCL.

Any repeal or amendment of provisions of our bylaws affecting indemnification rights, whether by our board of directors, stockholders or by changes in applicable law, or the adoption of any other provisions inconsistent therewith, will (unless otherwise required by law) be prospective only, except to the extent such amendment or change in law permits us to provide broader indemnification rights on a retroactive basis, and will not in any way diminish or adversely affect any right or protection existing thereunder with respect to any act or omission occurring prior to such repeal or amendment or adoption of such inconsistent provision.

We will enter into indemnification agreements with each of our officers and directors a form of which is to be filed as an exhibit to this Registration Statement. These agreements will require us to indemnify these individuals to the fullest extent permitted under Delaware law against liabilities that may arise by reason of their service to us, and to advance expenses incurred as a result of any proceeding against them as to which they could be indemnified.

Pursuant to the Underwriting Agreement to be filed as Exhibit 1.1 to this Registration Statement, we have agreed to indemnify the underwriters and the underwriters have agreed to indemnify us against certain civil liabilities that may be incurred in connection with this offering, including certain liabilities under the Securities Act.

Item 15. Recent Sales of Unregistered Securities.

On June 30, 2021, Aesther Healthcare Sponsor, LLC, our sponsor, purchased an aggregate of 2,875,000 founder shares, for an aggregate offering price of \$25,000 at an average purchase price of approximately \$0.009 per share. The number of founder shares issued was determined based on the expectation that the founder shares would represent 20% of the outstanding shares of common stock upon completion of this offering (without taking into account the , representative's shares). All such securities were issued in connection with our organization pursuant to the exemption from registration contained in Section 4(a)(2) of the Securities Act. Our sponsor is an accredited investor for purposes of Rule 501 of Regulation D.

On or before the date of the prospectus accompanying this registration statement, our sponsor has agreed to purchase an aggregate of 5,261,000 placement warrants (or 5,711,000 placement warrants if the over-allotment option is exercised in full), at a price of \$1.00 per placement warrant, for an aggregate purchase price of \$5,261,000 (\$5,711,000 if the over-allotment option is exercised in full). These placement warrants will be issued pursuant to the exemption from registration contained in Section 4(2) of the Securities Act as they will be sold to "accredited investors" as defined in Rule 501(a) of the Securities Act. No underwriting discounts or commissions will be paid with respect to such sales. Subscription agreements will be entered into with our sponsor in connection with these placement warrants and copies of such agreements will be attached as exhibits to this registration statement.

Item 16. Exhibits and Financial Statement Schedules.

- (a) *Exhibits*. The list of exhibits following the signature page to this registration statement is incorporated herein by reference.
- (b) Financial Statements. See page F-1 for an index to the financial statements and schedules included in the registration statement.

Item 17. Undertakings.

- (a) The undersigned registrant hereby undertakes to provide to the underwriters at the closing specified in the underwriting agreements, certificates in such denominations and registered in such names as required by the underwriters to permit prompt delivery to each purchaser.
- (b) Insofar as indemnification for liabilities arising under the Securities Act of 1933 may be permitted to directors, officers and controlling persons of the registrant pursuant to the foregoing provisions, or otherwise, the registrant has been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the registrant of expenses incurred or paid by a director, officer or controlling person of the registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Act and will be governed by the final adjudication of such issue.

- (c) The undersigned registrant hereby undertakes that:
 - (1) For purposes of determining any liability under the Securities Act of 1933, the information omitted from the form of prospectus filed as part of this registration statement in reliance upon Rule 430A and contained in a form of prospectus filed by the registrant pursuant to Rule 424(b) (1) or (4) or 497(h) under the Securities Act shall be deemed to be part of this registration statement as of the time it was declared effective.
 - (2) For the purpose of determining any liability under the Securities Act of 1933, each post-effective amendment that contains a form of prospectus shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
 - (3) For the purpose of determining liability under the Securities Act of 1933 to any purchaser, if the registrant is subject to Rule 430C, each prospectus filed pursuant to Rule 424(b) as part of a registration statement relating to an offering, other than registration statements relying on Rule 430B or other than prospectuses filed in reliance on Rule 430A, shall be deemed to be part of and included in the registration statement as of the date it is first used after effectiveness. Provided, however, that no statement made in a registration statement or prospectus that is part of the registration statement or made in a document incorporated or deemed incorporated by reference into the registration statement or prospectus that is part of the registration statement will, as to a purchaser with a time of contract of sale prior to such first use, supersede or modify any statement that was made in the registration statement or prospectus that was part of the registration statement or made in any such document immediately prior to such date of first use.
 - (4) For the purpose of determining liability of a registrant under the Securities Act of 1933 to any purchaser in the initial distribution of the securities, the undersigned registrant undertakes that in a primary offering of securities of an undersigned registrant pursuant to this registration statement, regardless of the underwriting method used to sell the securities to the purchaser, if the securities are offered or sold to such purchaser by means of any of the following communications, the undersigned registrant will be a seller to the purchaser and will be considered to offer or sell such securities to such purchaser:
 - (i) Any preliminary prospectus or prospectus of the undersigned registrant relating to the offering required to be filed pursuant to Rule 424;
 - (ii) Any free writing prospectus relating to the offering prepared by or on behalf of the undersigned registrant or used or referred to by an undersigned registrant;
 - (iii) The portion of any other free writing prospectus relating to the offering containing material information about the undersigned registrant or its securities provided by or on behalf of the undersigned registrant; and
 - (iv) Any other communication that is an offer in the offering made by the undersigned registrant to the purchaser.

SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, as amended, the registrant has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of New York, State of New York, on the 2nd day of September, 2021.

Aesther Healthcare Acquisition Corp.

By: /s/ Suren Ajjarapu

Name: Suren Ajjarapu Title: Chief Executive Officer

KNOW ALL MEN BY THESE PRESENTS, that each person whose signature appears below constitutes and appoints each of Suren Ajjarapu and Howard A. Doss or any one of them, with full power of substitution and re-substitution for him or her and in his or her name, place and stead, to act in the absence of the other, in any and all capacities, to sign any and all amendments (including post-effective amendments) to this Registration Statement, and to file the same, with all exhibits thereto, and other documents in connection therewith, with the Securities and Exchange Commission, granting unto said attorneys-in-fact and agents, and each of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises, as fully to all intents and purposes as she or he might or could do in person, hereby ratifying and confirming all that each of said attorney-in-fact or his or her substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Act of 1933, as amended, this Registration Statement has been signed below by the following persons in the capacities and on the dates indicated.

NAME	POSITION	DATE
/s/ Suren Ajjarapu Suren Ajjarapu	Chief Executive Officer and Chairman (principal executive officer)	September 2, 2021
/s/ Howard A. Doss Howard A. Doss	Chief Financial Officer (principal financial and accounting officer)	September 2, 2021
	II-7	

EXHIBIT INDEX

Exhibit	Description
1.1	Form of Underwriting Agreement*
3.1	Certificate of Incorporation*
3.2	Form of Amended and Restated Certificate of Incorporation*
3.3	By Laws*
4.1	Specimen Unit Certificate*
4.2	Specimen Class A Common Stock Certificate*
4.3	Specimen Warrant Certificate (included in Exhibit 4.4)*
4.4	Form of Warrant Agreement between Continental Stock Transfer & Trust Company, LLC and the Registrant*
5.1	Opinion of The Loev Law Firm, PC*
10.1	Form of Letter Agreement among the Registrant and our officers, directors and Aesther Healthcare Sponsor, LLC*
10.2	Promissory Note, dated June 30, 2021, issued to Aesther Healthcare Sponsor, LLC*
10.3	Form of Investment Management Trust Agreement between Continental Stock Transfer & Trust Company, LLC and the Registrant*
10.4	Form of Registration Rights Agreement between the Registrant and certain security holders*
10.5	Securities Subscription Agreement, dated June 30, 2021, between the Registrant and Aesther Healthcare Sponsor, LLC*
10.6	Form of Placement Warrant Purchase Agreement between the Registrant and Aesther Healthcare Sponsor, LLC*
10.7	Form of Indemnity Agreement*
10.8	Form of Administrative Support Agreement by and between the Registrant and Aesther Healthcare Sponsor, LLC*
14.1	Code of Ethics*
23.1	Consent of MaloneBailey LLP*
23.2	Consent of The Loev Law Firm, PC (included in Exhibit 5.1)*
24.1	Power of Attorney (included on the signature page to the initial filing of this Registration Statement)**
99.1	Audit Committee Charter*
99.2	Compensation Committee Charter*
99.3	Nominating and Corporate Governance Committee Charter*
99.4	Consent of Michael L. Peterson**
99.5	Consent of Venkatesh Srinivasan**
99.6	Consent of Donald G. Fell**
99.7	Consent of Siva Saravanan**

^{*} Filed herewith.

^{**} Previously filed.

10,000,000 Units Aesther Healthcare Acquisition Corp. UNDERWRITING AGREEMENT

[], 2021

EF Hutton, division of Benchmark Investments, LLC 590 Madison Avenue, 39th Floor New York, NY 10022

As Representative of the Underwriters named on <u>Schedule A</u> hereto

Ladies and Gentlemen:

The undersigned, Aesther Healthcare Acquisition Corp., a Delaware corporation ("**Company**"), hereby confirms its agreement with EF Hutton, division of Benchmark Investments, LLC (hereinafter referred to as "**you**", "**EF Hutton**", or as the "**Representative**") and with the other underwriters named on *Schedule A* hereto for which you are acting as representative (the Representative and the other Underwriters being collectively referred to herein as the "**Underwriters**" or, individually, an "**Underwriter**"), as follows:

1. Purchase and Sale of Securities.

1.1. Firm Securities.

1.1.1. Purchase of Firm Units. On the basis of the representations and warranties herein contained, but subject to the terms and conditions herein set forth, the Company agrees to issue and sell, severally and not jointly, to the several Underwriters, an aggregate of 10,000,000 units (the "Firm Units") of the Company at a purchase price (net of discounts and commissions, including the Deferred Underwriting Commission described in Section 1.3 below) of \$9.60 per Firm Unit, subject to certain adjustments set forth in Section 1.3 below. The Underwriters, severally and not jointly, agree to purchase from the Company the number of Firm Units set forth opposite their respective names on *Schedule A* attached hereto and made a part hereof at a purchase price (net of discounts and commissions, including the Deferred Underwriting Commission described in Section 1.3 below) of \$9.60 per Firm Unit. The Firm Units (and the Option Units (as hereinafter defined), if any) are to be offered initially to the public (the "Offering") at the offering price of \$10.00 per Firm Unit. Each Firm Unit consists of (i) one share of Class A common stock of the Company, par value \$.0001 (the "Class A Common Stock"), and (ii) one-half of one redeemable warrant ("Warrant(s)") with each Warrant entitling the holder thereof to purchase one share of Class A Common Stock.. The Class A Common Stock and Warrants included in the Firm Units will not be separately transferable until the 52^{nd} day after the date that the Registration Statement (as defined below) is declared effective (the "Effective Date") or the announcement by the Company of the Representative's decision to allow earlier trading, subject, however, to the Company filing a Current Report on Form 8-K ("Form 8-K") with the Commission (as defined below) containing an audited balanced sheet reflecting the Company's receipt of the gross proceeds of the Offering and issuing a press release announcing when such separate trading will begin. In no event will the Company allow separate trading until (i) the preparation of an audited balance sheet of the Company reflecting receipt by the Company of the proceeds of the Offering and the filing of such audited balance sheet with the Commission (as herein defined) on a Form 8-K or similar form by the Company which includes such balance sheet and (ii) the issuance of a press release announcing when such separate trading shall begin. Each whole Warrant entitles the holder thereof to purchase one share of Class A Common Stock at a price of \$11.50 per full share during the period commencing on the later of (a) 30 days after the completion of an initial Business Combination (as defined below), and (b) 12 months from the Effective Date, and terminating on the five year anniversary of the closing of a Business Combination. As used herein, the term "Business Combination" shall mean any share exchange, share reconstruction and amalgamation, share purchase, purchasing all or substantially all of the assets of, entering into contractual arrangements with, or engaging in any other similar business combination with, one or more operating businesses by the Company. As used herein, the term "Business Day" shall mean any day other than a Saturday, Sunday or any day on which national banks in New York, New York are not open for business.

1.1.2. Payment and Delivery. Delivery and payment for the Firm Units shall be made at 10:00 A.M., New York time, on the second (2nd) Business Day following the Effective Date of the Registration Statement (or the third Business Day following the Effective Date, if the Registration Statement is declared effective at or after 4:00 p.m.) or at such earlier time as shall be agreed upon by the Representative and the Company at the offices of the Representative or at such other place as shall be agreed upon by the Representative and the Company. The closing of the public offering contemplated by this Agreement is referred to herein as the "Closing" and the hour and date of delivery and payment for the Firm Units is referred to herein as the "Closing Date." Payment for the Firm Units shall be made on the Closing Date at the Representative's election by wire transfer in Federal (same day) funds or by certified or bank cashier's check(s) in New York Clearing House funds. \$102,000,000 (\$117,300,000 if the Over-allotment Option (as defined in Section 1.2) is exercised in full), or approximately \$10.20 per unit, of the proceeds received by the Company for the Firm Units and from the Private Placement (as defined in Section 1.6) shall be deposited in the trust account established by the Company for the benefit of the public shareholders as described in the Registration Statement (the "Trust Account") pursuant to the terms of an Investment Management Trust Agreement (the "Trust Agreement"). Such amount includes an aggregate of \$3,000,000 (\$3,450,000 if the Over-allotment Option is exercised in full), or \$0.30 per unit, payable to the Representative as a Deferred Underwriting Commission in accordance with Section 1.3 hereof, to be placed by the Underwriters in the Trust Account. The proceeds (less commissions, expense allowance and actual expense payments or other fees payable pursuant to this Agreement) shall be paid to the order of the Company upon delivery to the Representative of certificates (in form and substance satisfactory to the Underwriters) representing the Firm Units (or through the facilities of the Depository Trust Company ("DTC")) for the account of the Underwriters. The Firm Units shall be registered in such name or names and in such authorized denominations as the Representative may request in writing at least one (1) Business Day prior to the Closing Date. The Company will permit the Representative to examine and package the Firm Units for delivery at least one (1) full Business Day prior to the Closing Date. The Company shall not be obligated to sell or deliver the Firm Units except upon tender of payment by the Representative for all the Firm Units.

1.2. Over-Allotment Option.

1.2.1. Option Units. For the purpose of covering any over-allotments in connection with the distribution and sale of the Firm Units, the Underwriters are hereby granted, severally and not jointly, an option to purchase up to an additional 1,500,000 units from the Company (the "Overallotment Option"). Such additional 1,500,000 units shall be identical in all respects to the Firm Units and are hereinafter referred to as "Option Units." The Firm Units and the Option Units are hereinafter collectively referred to as the "Units," and the Units, the Class A Common Stock and the Warrants included in the Units are hereinafter referred to collectively as the "Public Securities." The purchase price to be paid for the Option Units (net of discounts and commissions), including the Deferred Underwriting Commission described in Section 1.3 below will be \$9.60 per Option Unit. The Option Units are to be offered initially to the public at the offering price of \$10.00 per Option Unit.

1.2.2. Exercise of Option. The Over-allotment Option granted pursuant to Section 1.2.1 hereof may be exercised by the Representative as to all (at any time) or any part (from time to time) of the Option Units within 45 days after the Effective Date. The Underwriters will not be under any obligation to purchase any Option Units prior to the exercise of the Over-allotment Option. The Over-allotment Option granted hereby may be exercised by the giving of oral notice to the Company from the Representative, which must be confirmed in writing by overnight mail or facsimile or e-mail transmission setting forth the number of Option Units to be purchased and the date and time for delivery of and payment for the Option Units, which will not be later than five Business Days after the date of the notice or such other time as shall be agreed upon by the Company and the Representative, at the offices of the Representative or at such other place or in such other manner as shall be agreed upon by the Company and the Representative. If such delivery and payment for the Option Units does not occur on the Closing Date, the date and time of the closing for such Option Units will be as set forth in the notice (hereinafter the "Option Closing Date"). Upon exercise of the Over-allotment Option, the Company will become obligated to convey to the Underwriters, and, subject to the terms and conditions set forth herein, the Underwriters will become obligated to purchase, the number of Option Units specified in such notice.

- 1.2.3. <u>Payment and Delivery.</u> Delivery and payment for the Option Units shall be made at 10:00 AM, New York time, on the Option Closing Date or at such earlier time as shall be agreed upon by the Representative and the Company at the offices of the Representative or at such other place as shall be agreed upon by the Representative and the Company. Payment for the Option Units shall be made on the Option Closing Date (or at such earlier time as shall be agreed upon by the Representative and the Company) at the Representative's election by wire transfer in Federal (same day) funds or by certified or bank cashier's check(s) in New York Clearing House funds, by deposit of the sum of \$9.60 per Option Unit in the Trust Account pursuant to the Trust Agreement upon delivery to the Representative of certificates (in form and substance satisfactory to the Underwriters) representing the Option Units (or through the facilities of DTC) for the account of the Underwriters. The Underwriters shall also place an aggregate of \$0.30 per Option Unit (up to \$450,000), payable to the Representative, as Deferred Underwriting Commission, in accordance with Section 1.3 hereof, in the Trust Account. The certificates representing the Option Units to be delivered will be in such denominations and registered in such names as the Representative requests not less than one (1) Business Day prior to the Closing Date or the Option Closing Date, as the case may be, and will be made available to the Representative for inspection, checking and packaging at the aforesaid office of the Company's transfer agent or correspondent not less than one full Business Day prior to such Closing Date or Option Closing Date.
- 1.3. <u>Deferred Underwriting Commission</u>. The Underwriters agree that 3.0% of the gross proceeds from the sale of the Firm Units (\$3,000,000) and 3.0% of the gross proceeds from the sale of the Option Units (up to \$450,000) (the "**Deferred Underwriting Commission**") will be deposited in and held in the Trust Account and payable directly from the Trust Account, without accrued interest, to the Representative for its own account upon consummation of the Business Combination. Each of the Underwriters acknowledges that only the Representative shall receive the Deferred Underwriting Commission. In the event that the Company is unable to consummate a Business Combination and Continental Stock Transfer & Trust Company ("CST"), as the trustee of the Trust Account (in this context, the "**Trustee**"), commences liquidation of the Trust Account as provided in the Trust Agreement, the Underwriters agree that: (i) the Underwriters hereby forfeit any rights or claims to the Deferred Underwriting Commission; and (ii) the Deferred Underwriting Commission, together with all other amounts on deposit in the Trust Account, shall be distributed on a pro-rata basis among the public shareholders.
- 1.4. Representative's Shares. The Company hereby agrees to issue to the Representative (or its designees) 100,000 shares of Class A Common Stock (the "Representative Shares") on the Closing Date. The Representative has agreed not to transfer, assign or sell any such Representative Shares until the completion of the initial Business Combination. In addition, the Representative has agreed (i) to waive its redemption rights with respect to such shares in connection with the completion of the initial Business Combination and (ii) to waive its rights to liquidating distributions from the trust account with respect to such shares if the Company fails to complete the initial Business Combination within 12 months from the closing of the Offering (or up to 18 months from the closing of this Offering if the Company extends the period of time to consummate an initial Business Combination by the full amount of time). The Representative will not sell, transfer, assign, pledge or hypothecate the Representative Shares, or cause the Representative Shares to be the subject of any hedging, short sale, derivative, put, or call transaction that would result in the effective economic disposition of the Representative Shares by any person, for a period of 180 days (pursuant to FINRA Rule 5110(e)(1)) following the Effective Date to anyone other than (i) the Representative or an underwriter or selected dealer in connection with the Offering, or (ii) any officers, partners, registered persons or affiliates of Representative or of any such underwriter or selected dealer. On and after the 181st day following the Effective Date, transfers to others may be made subject to compliance with or exemptions from applicable securities laws.
- 1.5. <u>Delivery and Payment</u>. Delivery of the Representative Shares shall be made on the Closing Date. The Company shall deliver to Representative, and its designees, book-entry confirmations for the Representative Shares in the name or names and in such authorized denominations as Representative may request.

1.6 Private Placement.

1.6.1 <u>Placement Warrants</u>. Simultaneously with the consummation of the Offering, Aesther Healthcare Sponsor, LLC (the "**Sponsor**") shall purchase from the Company pursuant to the Subscription Agreement (as defined in Section 2.25.2 hereof) an aggregate of 5,261,000 Warrants (the "**Placement Warrants**") at a purchase price of \$1.00 per Placement Warrant in a private placement (the "**Private Placement**"). The Placement Warrants, and the Class A Common Stock issuable upon exercise of the Placement Warrants are hereinafter referred to collectively as the "**Placement Securities**." Except as disclosed in the Registration Statement, there will be no placement agent in the Private Placement and no party shall be entitled to a placement fee or expense allowance from the sale of the Placement Securities. The Subscription Agreement shall state that the Sponsor has agreed (i) to waive its redemption rights with respect to the Placement Securities in connection with the completion of the initial Business Combination and (ii) to waive its rights to liquidating distributions from the trust account with respect to the Placement Securities if the Company fails to complete the initial Business Combination within 12 months from the closing of the Offering (or up to 18 months from the closing of this Offering if the Company extends the period of time to consummate an initial Business Combination by the full amount of time).

1.6.2 Additional Placement Warrants. Immediately prior to the Option Closing, the Sponsor shall purchase from the Company pursuant to the Subscription Agreement an additional number of Placement Warrants (up to a maximum of 450,000 Placement Warrants) *pro rata* with the amount of the Over-allotment Option exercised by the Representative so that at least \$10.20 per Firm Unit and Option Unit sold to the public in the Offering is held in trust regardless of whether the Over-allotment Option is exercised in full or part (the "Additional Placement Warrants"), at a purchase price of \$1.00 per Additional Placement Warrant in a private placement (the "Additional Private Placement"). The Additional Placement Warrants, and the Class A Common Stock issuable upon exercise of the Additional Placement Warrants are hereinafter referred to collectively as the "Additional Placement Securities." Except as disclosed in the Registration Statement, there will be no placement agent in the Additional Private Placement and no party shall be entitled to a placement fee or expense allowance from the sale of the Placement Securities. The Subscription Agreement shall state that the Sponsor has agreed to waive its redemption rights with respect to the Additional Placement Securities in connection with the completion of the initial Business Combination and (ii) to waive its rights to liquidating distributions from the trust account with respect to the Additional Placement Securities if the Company fails to complete the initial Business Combination within 12 months from the closing of the Offering (or up to 18 months from the closing of this Offering if the Company extends the period of time to consummate an initial Business Combination by the full amount of time).

2. Representations and Warranties of the Company. The Company represents and warrants to the Underwriters as follows:

2.1. Filing of Registration Statement.

2.1.1. Pursuant to the Act. The Company has filed with the Securities and Exchange Commission (the "Commission") a registration statement and an amendment or amendments thereto, on Form S-1 (File No. 333-258012), including any related preliminary prospectus (the "Preliminary **Prospectus**", including any prospectus that is included in the Registration Statement immediately prior to the effectiveness of the Registration Statement), for the registration of the Public Securities and Representative Shares under the Securities Act of 1933, as amended (the "Act"), which registration statement and amendment or amendments have been prepared by the Company in conformity in all material respects with the requirements of the Act, and the rules and regulations (the "Regulations") of the Commission under the Act. The conditions for use of Form S-1 to register the Offering under the Act, as set forth in the General Instructions to such Form, have been satisfied in all material respects. Except as the context may otherwise require, such registration statement, as amended, on file with the Commission at the time the registration statement becomes effective (including the prospectus, financial statements, schedules, exhibits and all other documents filed as a part thereof or incorporated therein and all information deemed to be a part thereof as of such time pursuant to Rule 430A of the Regulations), is hereinafter called the "Registration Statement," and the form of the final prospectus dated the Effective Date included in the Registration Statement (or, if applicable, the form of final prospectus containing information permitted to be omitted at the time of effectiveness by Rule 430A of the Regulations filed with the Commission pursuant to Rule 424 of the Regulations), is hereinafter called the "Prospectus." For purposes of this Agreement, "Time of Sale", as used in the Act, means 5:00 p.m., New York City time, on the date of this Agreement. If the Company has filed, or is required pursuant to the terms hereof to file, a registration statement pursuant to Rule 462(b) under the Act registering the Securities (a "Rule 462(b) Registration Statement"), then, unless otherwise specified, any reference herein to the term "Registration Statement" shall be deemed to include such Rule 462(b) Registration Statement. Other than a Rule 462(b) Registration Statement, which, if filed, becomes effective upon filing, no other document with respect to the Registration Statement has heretofore been filed with the Commission. All of the Public Securities have been registered under the Act pursuant to the Registration Statement or, if any Rule 462(b) Registration Statement is filed, will be duly registered under the Act with the filing of such Rule 462(b) Registration Statement. The Registration Statement has been declared effective by the Commission on the date hereof. If, subsequent to the date of this Agreement, the Company or the Representative has determined that at the Time of Sale the Prospectus included an untrue statement of a material fact or omitted a statement of material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, and have agreed to provide an opportunity to purchasers of the Firm Units to terminate their old purchase contracts and enter into new purchase contracts, the Prospectus will be deemed to include any additional information available to purchasers at the time of entry into the first such new purchase contract.

- 2.1.2. <u>Pursuant to the Exchange Act</u>. The Company has filed with the Commission a Form 8-A (File Number 001-[]) providing for the registration under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), of the Units, the Class A Common Stock, and the Warrants. The registration of the Units, Class A Common Stock and Warrants under the Exchange Act will be declared effective by the Commission on or prior to the Effective Date.
- 2.2. <u>No Stop Orders, Etc.</u> Neither the Commission nor, to the best of the Company's knowledge, any state regulatory authority has issued any order or threatened to issue any order preventing or suspending the use of any Preliminary Prospectus or Prospectus or any part thereof, or has instituted or, to the best of the Company's knowledge, threatened to institute any proceedings with respect to such an order.

2.3. <u>Disclosures in Registration Statement</u>.

2.3.1. 10b-5 Representation. At the time the Registration Statement became effective, upon the filing or first use (within the meaning of the Regulations) of the Prospectus and at the Closing Date and the Option Closing Date, if any, the Registration Statement and the Prospectus contained or will contain all material statements that are required to be stated therein in accordance with the Act and the Regulations, and did or will in all material respects conform to the requirements of the Act and the Regulations. Neither the Registration Statement nor any Preliminary Prospectus or the Prospectus, nor any amendment or supplement thereto, on their respective dates, did or will contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein (in the case of the Preliminary Prospectus and the Prospectus, in light of the circumstances under which they were made), not misleading. When any Preliminary Prospectus was first filed with the Commission (whether filed as part of the Registration Statement for the registration of the Securities or any amendment thereto or pursuant to Rule 424(a) of the Regulations) or first used (within the meaning of the Regulations) and when any amendment thereof or supplement thereto was first filed with the Commission or first used (within the meaning of the Regulations), such Preliminary Prospectus and any amendments thereof and supplements thereto complied or will have been corrected in the Prospectus to comply in all material respects with the applicable provisions of the Act and the Regulations and did not and will not contain an untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. The representation and warranty made in this Section 2.3.1 does not apply to statements made or statements omitted in reliance upon and in conformity with written information furnished to the Company with respect to the Underwriters by the Representative expressly for use in the Registration Statement or Prospectus or any amendment thereof or supplement thereto. It is understood the following identified statements set forth in the Prospectus under the heading "Underwriting" constitute, for the purposes of this Agreement, information furnished by the Representative with respect to the Underwriters: (i) the names of the Underwriters; (ii) the table of underwriters in the first paragraph of the section captioned "Underwriting", and (iii) the subsections titled "Price Stabilization, Short Positions," and "Electronic Distribution" included in the section captioned "Underwriting."

- 2.3.2. Disclosure of Agreements. The agreements and documents described in the Registration Statement, the Preliminary Prospectus and the Prospectus conform to the descriptions thereof contained therein and there are no agreements or other documents required to be described in the Registration Statement, the Preliminary Prospectus or the Prospectus or to be filed with the Commission as exhibits to the Registration Statement, that have not been so described or filed. Each agreement or other instrument (however characterized or described) to which the Company is a party or by which its property or business is or may be bound or affected and (i) that is referred to in the Registration Statement, Preliminary Prospectus or the Prospectus or attached as an exhibit thereto, or (ii) is material to the Company's business, has been duly and validly executed by the Company, is in full force and effect in all material respects and is enforceable against the Company and, to the Company's knowledge, the other parties thereto, in accordance with its terms, except (x) as such enforceability may be limited by bankruptcy, insolvency, reorganization or similar laws affecting creditors' rights generally, (y) as enforceability of any indemnification or contribution provision may be limited under the federal and state securities laws, and (z) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to the equitable defenses and to the discretion of the court before which any proceeding therefor may be brought, and none of such agreements or instruments has been assigned by the Company, and neither the Company nor, to the Company's knowledge, any other party is in breach or default thereunder and, to the Company's knowledge, no event has occurred that, with the lapse of time or the giving of notice, or both, would constitute a breach or default thereunder. To the Company's knowledge, performance by the Company of the material provisions of such agreements or instruments will not result in a material violation of any existing applicable law, rule, regulation, judgment, order or decree of any governmental agency or court, domestic or foreign, having jurisdiction over the Company or any of its assets or businesses, including, without limitation, those relating to environmental laws and regulations.
- 2.3.3. <u>Prior Securities Transactions</u>. No securities of the Company have been sold by the Company or by or on behalf of, or for the benefit of, any person or persons controlling, controlled by, or under common control with the Company since the date of the Company's formation, except as disclosed in the Registration Statement.
- 2.3.4. <u>Regulations</u>. The disclosures in the Registration Statement, the Preliminary Prospectus and the Prospectus concerning the effects of federal, state and local regulation on the Company's business as currently contemplated fairly summarize, to the best of the Company's knowledge, such effects and do not omit to state a material fact necessary to make the statements therein, in light of the circumstances in which they were made, not misleading.

2.4. Changes After Dates in Registration Statement.

- 2.4.1. No Material Adverse Change. Except as contemplated or disclosed in the Prospectus, since the respective dates as of which information is given in the Registration Statement, any Preliminary Prospectus and/or the Prospectus (i) there has been no material adverse change in the condition, financial or otherwise, or business prospects of the Company; (ii) there have been no material transactions entered into by the Company, other than as contemplated pursuant to this Agreement; (iii) no member of the Company's board of directors or management has resigned from any position with the Company and (iv) no event or occurrence has taken place which materially impairs, or would likely materially impair, with the passage of time, the ability of the members of the Company's board of directors or management to act in their capacities with the Company as described in the Registration Statement and the Prospectus.
- 2.4.2. <u>Recent Securities Transactions, Etc.</u> Except as contemplated in the Prospectus, subsequent to the respective dates as of which information is given in the Registration Statement and the Prospectus, and except as may otherwise be indicated or contemplated herein or therein, the Company has not: (i) issued any securities or incurred any liability or obligation, direct or contingent, for borrowed money; or (ii) declared or paid any dividend or made any other distribution on or in respect to its shares.
- 2.5. <u>Independent Accountants</u>. To the best of the Company's knowledge, MaloneBailey LLP ("**MaloneBailey**"), whose report is filed with the Commission as part of the Registration Statement and included in the Registration Statement, the Preliminary Prospectus and the Prospectus, are independent accountants as required by the Act and the Regulations and the Public Company Accounting Oversight Board (including the rules and regulations promulgated by such entity, the "**PCAOB**"). To the best of the Company's knowledge, MaloneBailey is duly registered and in good standing with the PCAOB. MaloneBailey has not, during the periods covered by the financial statements included in the Registration Statement and the Prospectus, provided to the Company any non-audit services, as such term is used in Section 10A(g) of the Exchange Act.

2.6. Financial Statements; Statistical Data.

2.6.1. <u>Financial Statements</u>. The financial statements, including the notes thereto and supporting schedules, included in the Registration Statement, the Preliminary Prospectus and the Prospectus fairly present in all material respects the financial position and the results of operations of the Company at the dates and for the periods to which they apply; and such financial statements have been prepared in conformity with generally accepted accounting principles, consistently applied throughout the periods involved; and the supporting schedules included in the Registration Statement present fairly in all material respects the information required to be stated therein. To the best of the Company's knowledge, no other financial statements or supporting schedules are required to be included or incorporated by reference in the Registration Statement, the Preliminary Prospectus or the Prospectus. The Registration Statement, the Preliminary Prospectus and the Prospectus disclose all material off-balance sheet transactions, arrangements, obligations (including contingent obligations), and other relationships of the Company with unconsolidated entities or other persons that may have a material current or future effect on the Company's financial condition, changes in financial condition, results of operations, liquidity, capital expenditures, capital resources, or significant components of revenues or expenses. To the best of the Company's knowledge, there are no pro forma or as adjusted financial statements which are required to be included in the Registration Statement and the Prospectus in accordance with Regulation S-X which have not been included as so required.

2.6.2. <u>Statistical Data</u>. The statistical, industry-related and market-related data included in the Registration Statement, the Preliminary Prospectus and the Prospectus are based on or derived from sources which the Company reasonably and in good faith believes are reliable and accurate, and such data agree with the sources from which they are derived.

2.7. <u>Authorized Capital</u>; <u>Options</u>, <u>Etc.</u> The Company had at the date or dates indicated in the Registration Statement, the Preliminary Prospectus and the Prospectus, as the case may be, duly authorized, issued and outstanding capitalization as set forth in the Registration Statement, the Preliminary Prospectus and the Prospectus. Based on the assumptions stated in the Registration Statement, the Preliminary Prospectus and the Prospectus, the Company will have on the Closing Date the adjusted share capitalization set forth therein. Except as set forth in, or contemplated by, the Registration Statement, the Preliminary Prospectus and the Prospectus, on the Effective Date and on the Closing Date and the Option Closing Date, if any, there will be no options, warrants, or other rights to purchase or otherwise acquire any authorized, but unissued Class A Common Stock or any security convertible into Class A Common Stock, or any contracts or commitments to issue or sell Class A Common Stock or any such options, warrants, rights or convertible securities.

2.8. Valid Issuance of Securities, Etc.

2.8.1. <u>Outstanding Securities</u>. All issued and outstanding securities of the Company (including, without limitation, the Placement Securities and the Additional Placement Securities) have been duly authorized and validly issued and are fully paid and non-assessable; the holders thereof have no rights of rescission with respect thereto, and are not subject to personal liability by reason of being such holders; and none of such securities were issued in violation of the preemptive rights of any holders of any security of the Company or similar contractual rights granted by the Company. The Public Securities conform in all material respects to all statements relating thereto contained in the Registration Statement, the Preliminary Prospectus and the Prospectus with respect to the Placement Securities and the Additional Placement Securities, the offers and sales of the outstanding Class A Common Stock were at all relevant times either registered under the Act and the applicable state securities or Blue Sky laws or, based in part on the representations and warranties of the purchasers of such Class A Common Stock, exempt from such registration requirements.

- 2.8.2. Securities Sold. The Securities have been duly authorized and reserved for issuance and when issued and paid for, will be validly issued, fully paid and non-assessable; the holders thereof are not and will not be subject to personal liability by reason of being such holders; the Securities are not and will not be subject to the preemptive rights of any holders of any security of the Company or similar contractual rights granted by the Company; and all corporate actions required to be taken for the authorization, issuance and sale of the Securities have been duly and validly taken. The Securities conform in all material respects to all statements with respect thereto contained in the Registration Statement, the Preliminary Prospectus and the Prospectus, as the case may be. When issued, the Warrants will constitute valid and binding obligations of the Company to issue and sell, upon exercise thereof and payment of the respective exercise prices therefor, the number and type of securities of the Company called for thereby in accordance with the terms thereof and the Warrants, and the Warrants are enforceable against the Company in accordance with their respective terms, except: (i) as such enforceability may be limited by bankruptcy, insolvency, reorganization or similar laws affecting creditors' rights generally; (ii) as enforceability of any indemnification or contribution provision may be limited under federal and state securities laws; and (iii) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to the equitable defenses and to the discretion of the court before which any proceeding therefor may be brought. The Representative Shares, when issued in accordance with the terms hereof, will be duly and validly authorized, validly issued, fully paid and non-assessable; the holders thereof are not and will not be subject to personal liability by reason of being such holders.
- 2.8.3. <u>Placement Securities</u>. The Placement Securities and the Additional Placement Securities have been duly authorized and reserved for issuance and when issued and paid for, will be validly issued, fully paid and non-assessable; the Placement Securities and the Additional Placement Securities are not and will not be subject to the preemptive rights of any holders of any security of the Company or similar contractual rights granted by the Company; and all corporate actions required to be taken for the authorization, issuance and sale of the Placement Securities and the Additional Placement Securities have been duly and validly taken. When issued, the Placement Warrants and the Additional Placement Warrants will constitute valid and binding obligations of the Company to issue and sell, upon purchase or exercise thereof and payment of the purchase price therefor, the number and type of securities of the Company called for thereby in accordance with the terms thereof, and such Placement Warrants and Additional Placement Warrants are enforceable against the Company in accordance with their respective terms, except: (i) as such enforceability may be limited by bankruptcy, insolvency, reorganization or similar laws affecting creditors' rights generally; (ii) as enforceability of any indemnification or contribution provision may be limited under federal and state securities laws; and (iii) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to the equitable defenses and to the discretion of the court before which any proceeding therefor may be brought. The Class A Common Stock underlying the Placement Warrants and the Additional Placement Warrants and Additional Placement Warrants and upon the exercise of such Placement Warrants and Additional Placement Warrants and upon the exercise of such Placement Warrants and Additional Placement Warrants and validly authorized, validly issued, fully paid and non-assessable, and the holders thereof are not and will not be s
- 2.8.4. No Integration. Subject to the disclosure contained in the Registration Statement, the Preliminary Prospectus and/or the Prospectus with respect to the Placement Securities, neither the Company nor any of its affiliates has, prior to the date hereof, made any offer or sale of any securities which are required to be "integrated" pursuant to the Act or the Regulations with the offer and sale of the Public Securities pursuant to the Registration Statement.
- 2.9. <u>Registration Rights of Third Parties</u>. Except as set forth in the Registration Statement, the Preliminary Prospectus or the Prospectus, no holders of any securities of the Company or any rights exercisable for or convertible or exchangeable into securities of the Company have the right to require the Company to register any such securities of the Company under the Act or to include any such securities in a registration statement to be filed by the Company.
- 2.10. <u>Validity and Binding Effect of Agreements</u>. This Agreement, the Trust Agreement (as defined in <u>Section 2.26</u> hereof), the Registration Rights Agreement (as defined in <u>Section 2.25.3</u> hereof), the Subscription Agreement (as defined in <u>Section 2.25.2</u> hereof), and the Warrant Agreement (as defined in <u>Section 2.24</u> hereof) have been duly and validly authorized by the Company and when executed and delivered by the Company and the other parties thereto, will constitute valid and binding agreements of the Company, enforceable against the Company in accordance with their respective terms, except: (i) as such enforceability may be limited by bankruptcy, insolvency, reorganization or similar laws affecting creditors' rights generally; (ii) as enforceability of any indemnification or contribution provision may be limited under the federal and state securities laws; and (iii) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to the equitable defenses and to the discretion of the court before which any proceeding therefor may be brought.

- 2.11. No Conflicts, Etc. The execution, delivery, and performance by the Company of this Agreement, the Trust Agreement, the Warrant Agreement, the Registration Rights Agreement, and the Subscription Agreement, the consummation by the Company of the transactions herein and therein contemplated and the compliance by the Company with the terms hereof and thereof do not and will not, with or without the giving of notice or the lapse of time or both: (i) result in a material breach of, or conflict with any of the terms and provisions of, or constitute a material default under, or result in the creation, modification, termination or imposition of any lien, charge or encumbrance upon any property or assets of the Company pursuant to the terms of any agreement or instrument to which the Company is a party; (ii) result in any violation of the provisions of the certificate of incorporation or bylaws of the Company (the "Charter Documents"); or (iii) violate any existing applicable law, rule, regulation, judgment, order or decree of any governmental agency or court, domestic or foreign, having jurisdiction over the Company or any of its properties or business.
- 2.12. No Defaults; Violations. No material default exists in the due performance and observance of any term, covenant or condition of any material license, contract, indenture, mortgage, deed of trust, note, loan or credit agreement, or any other agreement or instrument evidencing an obligation for borrowed money, or any other material agreement or instrument to which the Company is a party or by which the Company may be bound or to which any of the properties or assets of the Company is subject. The Company is not in violation of any material agreement, license, permit, applicable law, rule, regulation, judgment or decree of any governmental agency or court, domestic or foreign, having jurisdiction over the Company or any of its properties or businesses, except for such violations which would not reasonably be expected to have a material adverse effect on the Company. The Company is not in violation of any term or provision of its Certificate of Incorporation.

2.13. Corporate Power; Licenses; Consents.

- 2.13.1. <u>Conduct of Business</u>. The Company has all requisite corporate power and authority, and has all necessary authorizations, approvals, orders, licenses, certificates and permits of and from all governmental regulatory officials and bodies that it needs as of the date hereof to conduct its business for the purposes described in the Registration Statement, the Preliminary Prospectus and the Prospectus. To the Company's knowledge, the disclosures in the Registration Statement and the Prospectus concerning the effects of federal, state and local regulation on the Offering and the Company's business purpose as currently contemplated are correct in all material respects and do not omit to state a material fact required to be stated therein or necessary in order to make the statements therein (with respect to the Prospectus, in light of the circumstances under which they were made), not misleading. Since its formation, the Company has conducted no business and has incurred no liabilities other than in connection with and in furtherance of the Offering.
- 2.13.2. <u>Transactions Contemplated Herein</u>. The Company has all corporate power and authority to enter into this Agreement and to carry out the provisions and conditions hereof, and all consents, authorizations, approvals and orders required in connection therewith have been obtained. No consent, authorization or order of, and no filing with, any court, government agency or other body is required for the valid issuance, sale and delivery, of the Securities and the consummation of the transactions and agreements contemplated by this Agreement, the Trust Agreement, the Warrant Agreement, the Subscription Agreement and the Registration Rights Agreement and as contemplated by the Prospectus, except with respect to applicable federal and state securities laws and the rules and regulations promulgated by the Financial Industry Regulatory Authority ("FINRA").
- 2.14. <u>D&O Questionnaires</u>. To the knowledge of the Company, all information contained in the questionnaires (the "**Questionnaires**") completed by each of the Company's shareholders immediately prior to the Offering (the "**Initial Shareholders**") and each of the Company's officers and directors and included by the Company in the Registration Statement is true and correct in all material respects and the Company has not become aware of any information which would cause the information disclosed in the Questionnaires completed by each Initial Shareholder, officer or director, to become inaccurate and incorrect in any material respect.
- 2.15. <u>Litigation</u>; <u>Governmental Proceedings</u>. There is no action, suit, proceeding, inquiry, arbitration, investigation, litigation or governmental proceeding pending or, to the best of the Company's knowledge, threatened against, or involving the Company or, to the best of the Company's knowledge, any of its officers, directors or Initial Shareholders which is required to be disclosed and has not been disclosed in the Registration Statement, the Questionnaires, the Preliminary Prospectus and the Prospectus.

- 2.16. <u>Good Standing</u>. The Company has been duly incorporated, is validly existing and is in good standing under the laws of its country of incorporation and is duly qualified to do business and is in good standing as a foreign company in each jurisdiction in which its ownership or lease of property or the conduct of business requires such qualification, except where the failure to qualify would not have a material adverse effect on the Company.
- 2.17. <u>No Contemplation of a Business Combination</u>. Prior to the date hereof, neither the Company, its officers and directors nor the Initial Shareholders had, and as of the Closing, the Company and such officers and directors and Initial Shareholders will not have had: (i) any specific Business Combination under consideration or contemplation; or (ii) any substantive interactions or discussions with any target business regarding a possible Business Combination.

2.18. Transactions Affecting Disclosure to FINRA.

- 2.18.1. Except as described in the Preliminary Prospectus and/or the Prospectus, there are no claims, payments, arrangements, agreements or understandings relating to the payment of a finder's, consulting or origination fee by the Company or its officers or directors or any Initial Shareholder with respect to the sale of the Securities hereunder or any other arrangements, agreements or understandings of the Company or, to the Company's knowledge, any of its officers, directors or Initial Shareholders that may affect the Underwriters' compensation, as determined by FINRA.
- 2.18.2. The Company has not made any direct or indirect payments (in cash, securities or otherwise) to: (i) any person, as a finder's fee, consulting fee or otherwise, in consideration of such person raising capital for the Company or introducing to the Company persons who raised or provided capital to the Company; (ii) to any FINRA member; or (iii) to any person or entity that has any direct or indirect affiliation or association with any FINRA member, within the twelve (12) months prior to the Effective Date, other than payments to the Representative.
- 2.18.3. No officer, director, or beneficial owner of any class of the Company's securities (whether debt or equity, registered or unregistered, regardless of the time acquired or the source from which derived) (any such individual or entity, a "Company Affiliate") is a member, a person associated, or affiliated with a member of FINRA.
- 2.18.4. No Company Affiliate is an owner of stock or other securities of any member of FINRA (other than securities purchased on the open market).
 - 2.18.5. No Company Affiliate has made a subordinated loan to any member of FINRA.
- 2.18.6. No proceeds from the sale of the Public Securities (excluding underwriting compensation) or the Placement Securities or Additional Placement Securities will be paid to any FINRA member, or any persons associated or affiliated with a member of FINRA, except as specifically authorized herein and in the Subscription Agreement.
- 2.18.7. The Company has not issued any warrants or other securities, or granted any options, directly or indirectly to anyone who is a potential underwriter in the Offering or a related person (as defined by FINRA rules) of such an underwriter within the 180-day period prior to the initial filing date of the Registration Statement.
- 2.18.8. No person to whom securities of the Company have been privately issued within the 180-day period prior to the initial filing date of the Registration Statement has any relationship or affiliation or association with any member of FINRA.

- 2.18.9. To the Company's best knowledge, no FINRA member intending to participate in the Offering has a conflict of interest with the Company. For this purpose, a "conflict of interest" exists when a member of FINRA and its associated persons, parent or affiliates in the aggregate beneficially own 10% or more of the Company's outstanding subordinated debt or equity, or 10% or more of the Company's preferred equity. "Members participating in the Offering" include managing agents, syndicate group members and all dealers which are members of FINRA.
- 2.18.10. Except with respect to the Representative in connection with the Offering, the Company has not entered into any agreement or arrangement (including, without limitation, any consulting agreement or any other type of agreement) during the 180-day period prior to the initial filing date of the Registration Statement, which arrangement or agreement provides for the receipt of any item of value and/or the transfer of any warrants, options, or other securities from the Company to a FINRA member, any person associated with a member (as defined by FINRA rules), any potential underwriters in the Offering and any related persons.

2.19. Taxes.

- 2.19.1. There are no transfer taxes or other similar fees or charges under Delaware law, U.S. federal law or the laws of any U.S. state or any political subdivision thereof, required to be paid in connection with the execution and delivery of this Agreement or the issuance or sale by the Company of the Public Securities.
- 2.19.2. The Company has filed all non-U.S. and U.S. federal, state and local tax returns that are required to be a filed or has requested extensions thereof, except in any case in which the failure to so file would not have a Material Adverse Effect, and has paid all taxes required to be paid by it and any other assessment, fine or penalty levied against it, to the extent that any of the foregoing in due and payable, except for any such assessment, fine or penalty that is currently being contested in good faith or as would not have a Material Adverse Effect.
- 2.20. Foreign Corrupt Practices Act. Neither the Company nor, to the Company's knowledge, any of its officers, directors or Initial Shareholders or any other person acting on behalf of the Company has, directly or indirectly, given or agreed to give any money, gift or similar benefit (other than legal price concessions to customers in the ordinary course of business) to any customer, supplier, employee or agent of a customer or supplier, or official or employee of any governmental agency or instrumentality of any government (domestic or foreign) or any political party or candidate for office (domestic or foreign) or other person who was, is, or may be in a position to help or hinder the business of the Company (or assist it in connection with any actual or proposed transaction) that: (i) might subject the Company to any damage or penalty in any civil, criminal or governmental litigation or proceeding; (ii) if not given in the past, might have had a material adverse effect on the assets, business or operations of the Company as reflected in any of the financial statements contained in the Registration Statement, the Preliminary Prospectus and/or the Prospectus; or (iii) if not continued in the future, might adversely affect the assets, business, operations or prospects of the Company. The Company's internal accounting controls and procedures are sufficient to cause the Company to comply with the Foreign Corrupt Practices Act of 1977, as amended.
- 2.21. <u>Currency and Foreign Transactions Reporting Act</u>. The operations of the Company are and have been conducted at all times in material compliance with applicable financial recordkeeping and reporting requirements of the Currency and Foreign Transaction Reporting Act of 1970, as amended, the money laundering statutes of all applicable jurisdictions, the rules and regulations thereunder and any related or similar rules, regulations or guidelines, issued, administered or enforced by any governmental agency (collectively, the "**Money Laundering Laws**") and no action, suit or proceeding by or before any court or governmental agency, authority or body or any arbitrator involving the Company with respect to the Money Laundering Laws is pending or, to the best knowledge of the Company, threatened.
- 2.22. <u>Bank Secrecy Act; Patriot Act</u>. Neither the Company nor, to the Company's knowledge, any officer, director or Initial Shareholder has violated: (i) the Bank Secrecy Act of 1970, as amended; (ii) the Money Laundering Laws; or (iii) the Uniting and Strengthening of America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism (USA PATRIOT ACT) Act of 2001, and/or the rules and regulations promulgated under any such law, or any successor law.

- 2.23. Officers' Certificate. Any certificate signed by any duly authorized officer of the Company and delivered to the Representative or to the Representative's counsel shall be deemed a representation and warranty by the Company to the Underwriters as to the matters covered thereby.
- 2.24. <u>Warrant Agreement</u>. The Company has entered into a warrant agreement with respect to the Warrants (including the Placement Warrants) with CST substantially in the form filed as an exhibit to the Registration Statement (the "Warrant Agreement").

2.25. Agreements With Officers, Directors and Initial Shareholders.

- 2.25.1. <u>Insider Letters</u>. The Company has caused to be duly executed legally binding and enforceable agreements (except: (i) as such enforceability may be limited by bankruptcy, insolvency, reorganization or similar laws affecting creditors' rights generally; (ii) as enforceability of any indemnification, contribution or non-compete provision may be limited under the federal and state securities laws; and (iii) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to the equitable defenses and to the discretion of the court before which any proceeding therefor may be brought) annexed as exhibits to the Registration Statement (the "**Insider Letter**"), pursuant to which each of the officers, directors and Initial Shareholders of the Company agree to certain matters, including but not limited to, certain matters described as being agreed to by them under the "Proposed Business" Section of the Prospectus.
- 2.25.2. <u>Subscription Agreement</u>. The Sponsor has executed and delivered an agreement, annexed as an exhibit to the Registration Statement (the "**Subscription Agreement**"), pursuant to which the Sponsor, among other things, has purchased an aggregate of up to 5,261,000 Placement Warrants in the Private Placement, and an aggregate of up to 450,000 Additional Placement Warrants should the Representative exercise the Over-allotment Option. Pursuant to the Subscription Agreement, all of the proceeds from the sale of the Placement Warrants and Additional Placement Warrants will be deposited by the Company in the Trust Account in accordance with the terms of the Trust Agreement prior to the Closing.
- 2.25.3. <u>Registration Rights Agreement</u>. The Company, the Sponsor, the Initial Shareholders, and the Representative have entered into a registration rights agreement (the "**Registration Rights Agreement**") substantially in the form annexed as an exhibit to the Registration Statement, whereby the parties will be entitled to certain registration rights with respect to their securities, as set forth in such Registration Rights Agreement and described more fully in the Registration Statement.
- 2.26. <u>Investment Management Trust Agreement</u>. The Company has entered into the Trust Agreement with respect to certain proceeds of the Offering and the Private Placement substantially in the form filed as an exhibit to the Registration Statement.

2.27. Reserved.

- 2.28. <u>Covenants Not to Compete</u>. To the Company's knowledge, no officer, director or Initial Shareholder of the Company is subject to any noncompetition agreement or non-solicitation agreement with any employer or prior employer which could materially affect his or her ability to be an Initial Shareholder, employee, officer or director of the Company.
- 2.29. <u>Investments</u>. No more than 45% of the "value" (as defined in Section 2(a)(41) of the Investment Company Act of 1940, as amended (the "Investment Company Act")) of the Company's total assets consist of, and no more than 45% of the Company's net income after taxes is derived from, securities other than "Government Securities" (as defined in Section 2(a)(16) of the Investment Company Act).
- 2.30. <u>Subsidiaries</u>. The Company does not own an interest in any company, corporation, partnership, limited liability company, joint venture, trust or other business entity.

- 2.31. Related Party Transactions. No relationship, direct or indirect, exists between or among any of the Company or any Company Affiliate, on the one hand, and any director, officer, shareholder, customer or supplier of the Company or any Company Affiliate, on the other hand, which is required by the Act, the Exchange Act or the Regulations to be described in the Registration Statement, the Preliminary Prospectus and/or the Prospectus which is not so described and described as required. There are no outstanding loans, advances (except normal advances for business expenses in the ordinary course of business) or guarantees of indebtedness by the Company to or for the benefit of any of the officers or directors of the Company or any of their respective family members, except as disclosed in the Registration Statement, the Preliminary Prospectus and/or the Prospectus. The Company has not extended or maintained credit, arranged for the extension of credit, or renewed an extension of credit, in the form of a personal loan to or for any director or officer of the Company.
- 2.32. <u>No Influence</u>. The Company has not offered, or caused the Underwriters to offer, the Firm Units to any person or entity with the intention of unlawfully influencing: (i) a customer or supplier of the Company or any Company Affiliate to alter the customer's or supplier's level or type of business with the Company or such affiliate; or (ii) a journalist or publication to write or publish favorable information about the Company or any such affiliate.
- 2.33. <u>Trading of the Public Securities on the Nasdaq Global Market.</u> As of the Effective Date and the Closing Date, the Public Securities will have been authorized for listing on the Nasdaq Global Market and no proceedings have been instituted or threatened which would effect, and no event or circumstance has occurred as of the Effective Date which is reasonably likely to effect, the listing of the Public Securities on the Nasdaq Global Market.
- 2.34. Free-Writing Prospectus and Testing-the-Waters. The Company has not made any offer relating to the Public Securities that would constitute an issuer free writing prospectus, as defined in Rule 433 under the Act, or that would otherwise constitute a "free writing prospectus" as defined in Rule 405. The Company (a) has not engaged in any Testing-the-Waters Communication other than Testing-the-Waters Communications with the consent of the Representative with entities that are qualified institutional buyers within the meaning of Rule 144A under the Act or institutions that are accredited investors within the meaning of Rule 501 under the Act and (b) has not authorized anyone to engage in Testing-the-Waters Communications other than its officers and the Representative and individuals engaged by the Representative. The Company has not distributed any Written Testing-the-Waters Communications other than those listed on Schedule B hereto. "Testing-the-Waters Communication" means any oral or written communication with potential investors undertaken in reliance on Section 5(d) of the Act.
- 2.35. <u>Disclosure Controls and Procedures</u>. The Company maintains effective "disclosure controls and procedures" (as defined under Rule 13a-15(e) under the Exchange Act), to the extent required by such rule.
- 2.36. <u>Definition of "Knowledge"</u>. As used in herein, the term "**knowledge of the Company**" (or similar language) shall mean the knowledge of the officers and directors of the Company who are named in the Prospectus, with the assumption that such officers and directors shall have made reasonable and diligent inquiry of the matters presented.
- 3. <u>Covenants of the Company</u>. The Company covenants and agrees as follows:
- 3.1. <u>Amendments to Registration Statement</u>. The Company will deliver to the Representative, prior to filing, any amendment or supplement to the Registration Statement or Prospectus proposed to be filed after the Effective Date and will not file any such amendment or supplement to which the Representative shall reasonably object in writing.

3.2. Federal Securities Laws.

3.2.1. Compliance. During the time when a Prospectus is required to be delivered under the Act, the Company will use all reasonable efforts to comply with all requirements imposed upon it by the Act, the Regulations and the Exchange Act and by the regulations under the Exchange Act, as from time to time in force, so far as necessary to permit the continuance of sales of or dealings in the Public Securities in accordance with the provisions hereof and the Prospectus. If at any time when a Prospectus relating to the Public Securities is required to be delivered under the Act, any event shall have occurred as a result of which, in the opinion of counsel for the Company or counsel for the Underwriters, the Prospectus, as then amended or supplemented, includes an untrue statement of a material fact or omits to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, or if it is necessary during such period to amend the Registration Statement or amend or supplement the Prospectus to comply with the Act, the Company will notify the Representative promptly and prepare and file with the Commission, subject to Section 3.1 hereof, an appropriate amendment to the Registration Statement or amendment or supplement to the Prospectus (at the expense of the Company) so as to correct such statement or omission or effect such compliance.

- 3.2.2. <u>Filing of Final Prospectus</u>. The Company will file the Prospectus (in form and substance satisfactory to the Representative) with the Commission pursuant to the requirements of Rule 424 of the Regulations.
- 3.2.3. Exchange Act Registration. For a period of five (5) years from the Effective Date (except in connection with a going private transaction), or until such earlier time upon which the Company is required to be liquidated, the Company will use its best efforts to maintain the registration of the Units, Class A Common Stock, and Warrants under the provisions of the Exchange Act. The Company will not deregister the Units, Class A Common Stock, and Warrants under the Exchange Act without the prior written consent of the Representative.
- 3.2.4. <u>Free Writing Prospectuses</u>. The Company will not make any offer relating to the Public Securities that would constitute an issuer free writing prospectus, as defined in Rule 433 under the Act.
- 3.2.5. <u>Sarbanes-Oxley Compliance</u>. As soon as it is legally required to do so, the Company shall take all actions necessary to obtain and thereafter maintain material compliance with each applicable provision of the Sarbanes-Oxley Act of 2002, as amended, and the rules and regulations promulgated thereunder and related or similar rules and regulations promulgated by any other governmental or self-regulatory entity or agency with jurisdiction over the Company.
- 3.3. <u>Blue Sky Filing</u>. Unless the Public Securities are listed on the Nasdaq Global Market or another national securities exchange, the Company at its expense will endeavor in good faith, in cooperation with the Representative, at or prior to the time the Registration Statement becomes effective, to qualify the Public Securities for offering and sale under the securities laws of such jurisdictions as the Representative may reasonably designate, provided that no such qualification shall be required in any jurisdiction where, as a result thereof, the Company would be subject to service of general process or to taxation as a foreign company doing business in such jurisdiction. In each jurisdiction where such qualification shall be effected, the Company will, unless the Representative agrees that such action is not at the time necessary or advisable, use all reasonable efforts to file and make such statements or reports at such times as are or may be required by the laws of such jurisdiction.
- 3.4. <u>Delivery to Underwriters of Prospectuses</u>. The Company will deliver to each of the several Underwriters, without charge, from time to time during the period when the Prospectus is required to be delivered under the Act or the Exchange Act such number of copies of each Preliminary Prospectus and Prospectus and all amendments and supplements to such documents as such Underwriters may reasonably request and, as soon as the Registration Statement or any amendment or supplement thereto becomes effective, deliver to the Representative two original executed Registration Statements, including exhibits, and all post-effective amendments thereto and copies of all exhibits filed therewith or incorporated therein by reference and all original executed consents of certified experts.
- 3.5. Effectiveness and Events Requiring Notice to the Representative. The Company will use its best efforts to cause the Registration Statement to remain effective and will notify the Representative immediately and confirm the notice in writing: (i) of the effectiveness of the Registration Statement and any amendment thereto; (ii) of the issuance by the Commission of any stop order suspending the effectiveness of the Registration Statement, or any post-effective amendment thereto or preventing or suspending the use of any Preliminary Prospectus or the Prospectus or of the initiation, or the threatening, of any proceedings for that purpose; (iii) of the issuance by any state securities commission of any proceedings for the suspension of the qualification of the Public Securities for offering or sale in any jurisdiction or of the initiation, or the threatening, of any proceeding for that purpose; (iv) of the mailing and delivery to the Commission for filing of any amendment or supplement to the Registration Statement or Prospectus; (v) of the receipt of any comments or request for any additional information from the Commission; and (vi) of the happening of any event during the period described in Section 3.5 hereof that, in the judgment of the Company, makes any statement of a material fact made in the Registration Statement, the Preliminary Prospectus and/or the Prospectus untrue or that requires the making of any changes in the Registration Statement, the Preliminary Prospectus and/or the Prospectus in order to make the statements therein (with respect to the Preliminary Prospectus and the Prospectus in light of the circumstances under which they were made), not misleading in any material respect. If the Commission or any state securities commission shall enter a stop order or suspend such qualification at any time, the Company will make every reasonable effort to obtain promptly the lifting of such order.

3.6. Review of Financial Statements. Until the earlier of five (5) years from the Effective Date, or until such earlier date upon which the Company is required to be liquidated, the Company, at its expense, shall cause its regularly engaged independent certified public accountants to review (but not audit) the Company's financial statements for each of the first three fiscal quarters prior to the announcement or filing of quarterly financial information, if any.

3.7. Affiliated Transactions.

- 3.7.1. <u>Business Combinations</u>. The Company will not consummate a Business Combination with any entity which is affiliated with any of its officers, directors or Initial Shareholders unless the Company obtains an opinion from an independent investment banking firm or another independent entity that commonly renders valuation opinions stating the Business Combination is fair to the Company's shareholders from a financial perspective.
- 3.7.2. <u>Compensation</u>. Except as set forth in this <u>Section 3.7.2</u>, the Company shall not pay any of its officers, directors or Initial Shareholders or any of their affiliates any fees or compensation from the Company, for services rendered to the Company prior to, or in connection with, this Offering or the consummation of a Business Combination; *provided* that the officers, directors and the Initial Shareholder shall be entitled to reimbursement from the Company for their out-of-pocket expenses incurred on the Company's behalf, which includes any loans and advances made to the Company prior to the Closing.
- 3.7.3 <u>Administrative Services</u>. The Company has entered into an administrative services agreement (the "Services Agreement") with the Sponsor pursuant to which the Sponsor will make available to the Company general and administrative services including office space, utilities, receptionist and secretarial support for the Company's use for \$10,000 per month for up to eighteen months.
- 3.8. Secondary Market Trading. In the event the Public Securities are not listed on the Nasdaq Global Market or such other national securities exchange, the Company will apply to be included in Mergent, Inc. Manual for a period of five (5) years from the consummation of a Business Combination. Promptly after the consummation of the Offering, the Company shall take such commercially reasonable steps as may be necessary to obtain a secondary market trading exemption for the Company's securities in all applicable jurisdictions. The Company shall also take such other action as may be reasonably requested by the Representative to obtain a secondary market trading exemption in such other states as may be requested by the Representative; provided that no qualification shall be required in any jurisdiction where, as a result thereof, the Company would be subject to service of general process or to taxation as a foreign entity doing business in such jurisdiction.
- 3.9. <u>Financial Public Relations Firm</u>. Promptly after the execution of a definitive agreement for a Business Combination, the Company shall retain a financial public relations firm reasonably acceptable to the Representative for a term to be agreed upon by the Company and the Representative.

3.10. Reports to the Representative.

- 3.10.1. Periodic Reports, Etc. For a period of five (5) years from the Effective Date or until such earlier time upon which the Company is dissolved, the Company will furnish to the Representative and its counsel copies of such financial statements and other periodic and special reports as the Company from time to time furnishes generally to holders of any class of its securities, and promptly furnish to the Representative: (i) a copy of each periodic report the Company shall be required to file with the Commission; (ii) a copy of every press release and every news item and article with respect to the Company or its affairs which was released by the Company; (iii) a copy of each Form 8-K or Schedules 13D, 13G, 14D-1 or 13E-4 received or prepared by the Company; (iv) five (5) copies of each Registration Statement; and (v) such additional documents and information with respect to the Company and the affairs of any future subsidiaries of the Company as the Representative may from time to time reasonably request; provided that the Representative shall sign, if requested by the Company, a Regulation FD compliant confidentiality agreement which is reasonably acceptable to the Representative and its counsel in connection with the Representative's receipt of such information. Documents filed with the Commission pursuant to its Electronic Data Gathering, Analysis and Retrieval System ("EDGAR") shall be deemed to have been delivered to the Representative pursuant to this section.
- 3.10.2. <u>Transfer Sheets</u>. For a period of five (5) years following the Effective Date or until such earlier time upon which the Company is dissolved, the Company shall retain a transfer and warrant agent acceptable to the Representative (the "**Transfer Agent**"). In the event the Public Securities are not listed on the Nasdaq Global Market or such other national securities exchange, the Company will furnish to the Underwriters at the Company's sole cost and expense such transfer sheets of the Company's securities as the Representative may request, including the daily and monthly consolidated transfer sheets of the Transfer Agent and DTC. CST is an acceptable Transfer Agent to the Representative.
- 3.10.3. <u>Trading Reports</u>. If the Public Securities are quoted on the OTC Bulletin Board (or any successor trading market) or a market operated by the OTC Market Group Inc. (or similar publisher of quotations), then during such time the Company shall provide to the Representative, at its expense, such reports published by the OTC Bulletin Board or the OTC Market Group Inc. relating to price trading of the Public Securities, as the Representative shall reasonably request. In addition to the requirements of the preceding sentence, if the Public Securities are not listed on a national securities exchange, for a period of two (2) years from the Closing Date, the Company, at its expense, shall provide the Representative a subscription to the Company's weekly Depository Transfer Company Security Position Reports.
- 3.11. <u>Disqualification of Form S-1</u>. For a period of seven (7) years from the date hereof, the Company will not take any action or actions which may prevent or disqualify the Company's use of Form S-1 (or other appropriate form) for the registration of the Warrants under the Act.

3.12. Payment of Expenses.

3.12.1. General Expenses Related to the Offering. The Company hereby agrees to pay on each of the Closing Date and the Option Closing Date, if any, to the extent not paid at Closing Date, all expenses incident to the performance of the obligations of the Company under this Agreement, including, but not limited to: (i) the preparation, printing, filing and mailing (including the payment of postage with respect to such mailing) of the Registration Statement, the Preliminary Prospectus and/or the final Prospectus and the printing and mailing of this Agreement and related documents, including the cost of all copies thereof and any amendments thereof or supplements thereto supplied to the Underwriters in quantities as may be required by the Underwriters; (ii) the printing, engraving, issuance and delivery of the Units, and the Class A Common Stock and Warrants included in the Units, including any transfer or other taxes payable thereon; (iii) if the public securities are not listed on a national securities exchange, the qualification of the Public Securities under state or foreign securities or Blue Sky laws, including the costs of printing and mailing the "Preliminary Blue Sky Memorandum," and all amendments and supplements thereto, fees and disbursements for counsel of the Representative's choice retained for such purpose; (iv) filing fees incurred in registering the Offering with FINRA (including all Public Offering System filing fees); (v) fees and disbursements of the transfer and warrant agent; (vi) the Company's expenses associated with "road show" marketing "due diligence" meetings arranged by the Representative (none of which will be received or paid on behalf of an underwriter and related person); (vii) the preparation of leather bound volumes and Lucite cube or similar commemorative items in a style as reasonably requested by the Representative; (viii) background checks on the Company's directors, director nominees and executive officers as requested by the Representative; (ix) transfer taxes, all fees and any expenses and fees incurred by the Representative's counsel, transfer and warrant agent and registrar fees; and (x) all other reasonable costs and expenses incident to the performance of its obligations hereunder which are not otherwise specifically provided for in this Section 3.12.1. The Representative may deduct from the net proceeds of the Offering payable to the Company on the Closing Date, or the Option Closing Date, if any, the expenses set forth above to be paid by the Company to the Representative and others, as agreed to by the Company in writing; provided, however, that such fees and expenses deducted from the net proceeds of the Offering payable to the Company shall not exceed \$100,000 in the aggregate (less any amounts previously paid).

- 3.12.2 *Fee on Termination of Offering*. Notwithstanding anything contained herein to the contrary, upon termination of the Offering other than as a result of a breach of this Agreement by the Underwriters, the Company shall: (A) reimburse the Representative for, or otherwise pay and bear, the expenses and fees to be paid and borne by the Company as provided for in Section 3.12.1 above, as applicable, and (B) reimburse the Representative for the full amount of its accountable out-of-pocket expenses actually incurred to such date (which shall include, but shall not be limited to, all fees and disbursements of the Representative's counsel, travel, lodging and other "road show" expenses, mailing, printing and reproduction expenses, and any expenses incurred by the Representative in conducting its due diligence, including background checks of the Company's officers and directors), up to an aggregate amount of \$50,000, less the amounts previously paid and any amounts previously paid to the Representative in reimbursement for such expenses. If applicable, and solely in the event of a termination of this Offering, the Representative shall refund to the Company any portion of advances previously received by the Representative which are in excess of the accountable out-of-pocket expenses actually incurred to such date by the Representative.
- 3.13. <u>Application of Net Proceeds</u>. The Company will apply the net proceeds from the Offering received by it in a manner consistent with the application described under the caption "Use of Proceeds" in the Prospectus.
- 3.14. <u>Delivery of Earnings Statements to Security Holders</u>. The Company will make generally available to its security holders as soon as practicable, but not later than the first day of the fifteenth (15th) full calendar month following the Effective Date, an earnings statement (which need not be certified by independent public or independent certified public accountants unless required by the Act or the Regulations, but which shall satisfy the provisions of Rule 158(a) under Section 11(a) of the Act) covering a period of at least twelve (12) consecutive months beginning after the Effective Date.

3.15. Notice to FINRA.

- 3.15.1. <u>Business Combination</u>. In the event any person or entity (regardless of any FINRA affiliation or association) is engaged to assist the Company in its search for a merger candidate or to provide any other merger and acquisition services, the Company will provide the following to the FINRA and the Representative prior to the consummation of the Business Combination: (i) complete details of all services and copies of agreements governing such services; and (ii) justification as to why the person or entity providing the merger and acquisition services should not be considered an "underwriter and related person" (as such term is defined in Rule 5110 of FINRA's Rules) with respect to the Offering. The Company also agrees that proper disclosure of such arrangement or potential arrangement will be made in any proxy or tender offer statement which the Company files in connection with the Business Combination.
- 3.15.2. <u>Broker/Dealer</u>. In the event the Company intends to register as a broker/dealer, merge with or acquire a registered broker/dealer, or otherwise become a member of FINRA, it shall promptly notify FINRA.
- 3.16. <u>Stabilization</u>. Neither the Company, nor, to its knowledge, any of its employees, directors or shareholders (without the consent of the Representative) has taken or will take, directly or indirectly, any action designed to or that has constituted or that might reasonably be expected to cause or result in, under the Exchange Act, or otherwise, stabilization or manipulation of the price of any security of the Company to facilitate the sale or resale of the Units.
- 3.17. <u>Internal Controls</u>. The Company will maintain a system of internal accounting controls sufficient to provide reasonable assurances that: (i) transactions are executed in accordance with management's general or specific authorization; (ii) transactions are recorded as necessary in order to permit preparation of financial statements in accordance with generally accepted accounting principles and to maintain accountability for assets; (iii) access to assets is permitted only in accordance with management's general or specific authorization; and (iv) the recorded accountability for assets is compared with existing assets at reasonable intervals and appropriate action is taken with respect to any differences.

- 3.18. <u>Accountants</u>. For a period of five years from the Effective Date or until such earlier time upon which the Company is required to be liquidated, the Company shall retain MaloneBailey or other independent public accountants reasonably acceptable to the Representative.
- 3.19. Form 8-K. The Company shall, on the date hereof, retain its independent public accountants to audit the financial statements of the Company as of the Closing Date (the "Audited Financial Statements") reflecting the receipt by the Company of the proceeds of the Offering and the Private Placement, as well as the proceeds from the exercise of the Over-Allotment if such exercise has occurred on the date of the Prospectus. Within four Business Days of the Closing Date, the Company will file a Current Report on Form 8-K with the Commission, which Report shall contain the Audited Financial Statements.
- 3.20. <u>FINRA</u>. The Company shall advise FINRA if it is aware that any 5% or greater shareholder of the Company becomes an affiliate or associated person of a FINRA member participating in the distribution of the Company's Public Securities.
- 3.21. <u>Corporate Proceedings</u>. All corporate proceedings and other legal matters necessary to carry out the provisions of this Agreement and the transactions contemplated hereby shall have been done to the reasonable satisfaction of counsel for the Underwriters.
- 3.22. <u>Investment Company</u>. The Company shall cause the proceeds of the Offering to be held in the Trust Account to be invested only in "government securities" with specific maturity dates or in money market funds meeting certain conditions under Rule 2a-7 promulgated under the Investment Company Act as set forth in the Trust Agreement and disclosed in the Prospectus. The Company will otherwise conduct its business in a manner so that it will not become subject to the Investment Company Act. Furthermore, once the Company consummates a Business Combination, it will be engaged in a business other than that of investing, reinvesting, owning, holding or trading securities.
- 3.23. <u>Business Combination Announcement</u>. Within four (4) Business Days following the consummation by the Company of a Business Combination, the Company shall cause an announcement ("**Business Combination Announcement**") to be issued by a press release service announcing the consummation of the Business Combination and indicating that the Representative was one of the co-managing underwriters in the Offering and also indicating the name and location of any other financial advisors engaged by the Company as its merger and acquisitions advisor. The Company shall supply the Representative with a draft of the Business Combination Announcement and provide the Representative with a reasonable advance opportunity to comment thereon. The Company will not issue the Business Combination Announcement without the final approval of the Representative, which approval shall not be unreasonably withheld, conditioned or delayed.
- 3.24. <u>Press Releases</u>. The Company agrees that it will not issue press releases or engage in any other publicity, without the Representative's prior written consent (not to be unreasonably withheld), for a period of forty five (45) days from the Effective Date; provided that in no event shall the Company be prohibited from issuing any press release or engaging in any other publicity required by law.
- 3.25. Electronic Prospectus. The Company shall cause to be prepared and delivered to the Representative, at its expense, within one (1) Business Day from the Effective Date, an Electronic Prospectus to be used by the Underwriters in connection with the Offering. As used herein, the term "Electronic Prospectus" means a form of prospectus, and any amendment or supplement thereto, that meets each of the following conditions: (i) it shall be encoded in an electronic format, satisfactory to the Representative, that may be transmitted electronically by the other Underwriters to offerees and purchasers of the Units for at least the period during which a Prospectus relating to the Units is required to be delivered under the Act; (ii) it shall disclose the same information as the paper prospectus and prospectus filed pursuant to EDGAR, except to the extent that graphic and image material cannot be disseminated electronically, in which case such graphic and image material shall be replaced in the electronic prospectus with a fair and accurate narrative description or tabular representation of such material, as appropriate; and (iii) it shall be in or convertible into a paper format or an electronic format, satisfactory to the Representative, that will allow recipients thereof to store and have continuously ready access to the prospectus at any future time, without charge to such recipients (other than any fee charged for subscription to the Internet as a whole and for on-line time). The Company hereby confirms that it has included or will include in the Prospectus filed pursuant to EDGAR or otherwise with the Commission and in the Registration Statement at the time it was declared effective an undertaking that, upon receipt of a request by an investor or his or her representative within the period when a prospectus relating to the Units is required to be delivered under the Act, the Company shall transmit or cause to be transmitted promptly, without charge, a paper copy of the Prospectus.

- 3.26. <u>Reservation of Shares</u>. The Company will reserve and keep available that maximum number of its authorized but unissued securities which are issuable upon exercise of the Warrants, the Placement Warrants, and the Additional Placement Warrants outstanding from time to time.
- 3.27. <u>Private Placement Proceeds</u>. Immediately upon establishment of the Trust Account and prior to the Closing, the Company shall deposit all of the proceeds from the Private Placement in the Trust Account and shall provide the Representative with evidence of the same.

3.28. No Amendment to Certificate of Incorporation.

- (i) Prior to the closing of a Business Combination, the Company covenants and agrees it will not seek to amend or modify its Certificate of Incorporation, as amended, without the prior approval of its Board of Directors and the affirmative vote of a majority of the voting power of the Class A Common Stock.
- (ii) The Company acknowledges that the purchasers of the Units in this Offering shall be deemed to be third party beneficiaries of this Section 3.28.
- (iii) The Representative and the Company specifically agree that this <u>Section 3.28</u> shall not be modified or amended in any way without the approval of at least 65% of the voting power of the Class A Common Stock that were issued in the Offering.
- 3.29. <u>Financial Printer</u>. The Company shall retain a financial printer, reasonably acceptable to the Representative, for the purpose of facilitating the Company's EDGAR filings and the printing of the Preliminary Prospectus and Prospectus.
- 3.30. <u>Listing on the Nasdaq Global Market</u>. The Company will use commercially reasonable efforts to maintain the listing of the Public Securities on the Nasdaq Global Market or such other national securities exchange until the earlier of five (5) years from the Effective Date or until the Public Securities are no longer registered under the Exchange Act.
- 3.31. <u>Payment of Deferred Underwriting Commission on Business Combination</u>. Upon the consummation of a Business Combination, the Company agrees that it will cause the Trustee to pay the Deferred Underwriting Commission directly from the Trust Account to the Representative, in accordance with <u>Section 1.3</u>.
- 4. <u>Conditions of Underwriters' Obligations</u>. The obligations of the several Underwriters to purchase and pay for the Units, as provided herein, shall be subject to the continuing accuracy of the representations and warranties of the Company as of the date hereof and as of each of the Closing Date and the Option Closing Date, if any, to the accuracy of the statements of officers of the Company made pursuant to the provisions hereof and to the performance by the Company of its obligations hereunder and to the following conditions:

4.1 Regulatory Matters.

4.1.1. <u>Effectiveness of Registration Statement</u>. The Registration Statement shall have become effective not later than 5:00 P.M., New York time, on the date of this Agreement or such later date and time as shall be consented to in writing by the Representative, and, at each of the Closing Date and the Option Closing Date, no stop order suspending the effectiveness of the Registration Statement shall have been issued and no proceedings for the purpose shall have been instituted or shall be pending or contemplated by the Commission and any request on the part of the Commission for additional information shall have been complied with to the reasonable satisfaction of Schiff Hardin LLP ("Schiff").

- 4.1.2. <u>FINRA Clearance</u>. By the Effective Date, the Representative shall have received a notice of no objection from FINRA as to the terms and arrangements for acting as, and the amount of compensation allowable or payable to, the Underwriters as described in the Registration Statement.
- 4.1.3. <u>No Commission Stop Order</u>. At each of the Closing Date and the Option Closing Date, the Commission has not issued any order or threatened to issue any order preventing or suspending the use of any Preliminary Prospectus or the Prospectus or any part thereof, and has not instituted or threatened to institute any proceedings with respect to such an order.
- 4.1.4. No Blue Sky Stop Orders. No order suspending the sale of the Units in any jurisdiction designated by the Representative pursuant to Section 3.3 hereof, if any, shall have been issued on either the Closing Date or the Option Closing Date, and no proceedings for that purpose shall have been instituted or shall be contemplated.
- 4.1.5. <u>The Nasdaq Global Market</u>. By the Effective Date, the Securities shall have been approved for trading on the Nasdaq Global Market.

4.2. Company Counsel Matters.

4.2.1. <u>Closing Date Opinion of Counsel</u>. On the Closing Date, the Representative shall have received the favorable opinion of The Loev Law Firm, PC ("**Loev**"), counsel to the Company, dated the Closing Date, addressed to the Representative and the other Underwriters and in form and substance reasonably satisfactory to the Representative.

The opinion of Loev shall further include a statement to the effect that such counsel has participated in conferences with officers and other representatives of the Company, representatives of the independent public accountants for the Company and representatives of the Underwriters at which the contents of the Registration Statement, final Preliminary Prospectus, the Prospectus and related matters were discussed and although such counsel is not passing upon and does not assume any responsibility for the accuracy, completeness or fairness of the statements contained in the Registration Statement, final Preliminary Prospectus and the Prospectus (except as otherwise set forth in such opinion), no facts have come to the attention of such counsel which lead them to believe that either the Registration Statement, final Preliminary Prospectus or the Prospectus or any amendment or supplement thereto, as of the date of such opinion contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading (it being understood that such counsel need express no opinion with respect to the financial statements and related notes and schedules and other financial and statistical data included in the Registration Statement, final Preliminary Prospectus or matters relating to the sale of securities in any jurisdiction outside the U.S.).

- 4.2.2. <u>Option Closing Date Opinion of Counsel</u>. On each Option Closing Date, if any, the Representative shall have received the favorable opinion of Loev, dated each Option Closing Date, addressed to the Representative and in form and substance reasonably satisfactory to counsel to the Representative, confirming as of each Option Closing Date, the statements made by Love in its opinion delivered on the Closing Date.
- 4.2.3. <u>Reliance</u>. In rendering such opinions, such counsels may rely: (i) as to matters involving the application of laws other than the laws of the United States and jurisdictions in which they are admitted, to the extent such counsel deems proper and to the extent specified in such opinion, if at all, upon an opinion or opinions (in form and substance reasonably satisfactory to the Representative) of other counsel reasonably acceptable to the Representative, familiar with the applicable laws; and (ii) as to matters of fact, to the extent they deem proper, on certificates or other written statements of officers of the Company and officers of departments of various jurisdictions having custody of documents respecting the corporate existence or good standing of the Company, provided that copies of any such statements or certificates shall be delivered to the Underwriters' counsel if requested. The opinion of Loev and any opinion relied upon by such counsel for the Company shall include a statement to the effect that it may be relied upon by counsel for the Underwriters in its opinion delivered to the Underwriters.

- 4.3. <u>Cold Comfort Letter</u>. At the time this Agreement is executed, and at each of the Closing Date and the Option Closing Date, if any, the Representative shall have received a letter, addressed to the Representative and in form and substance satisfactory in all respects (including the nature of the changes or decreases, if any, referred to in clause (iii) below) to the Representative from MaloneBailey dated, respectively, as of the date of this Agreement and as of the Closing Date and the Option Closing Date, if any:
- (i) Confirming that they are an independent registered public accounting firm with respect to the Company within the meaning of the Act and the applicable Regulations;
- (ii) Stating that in their opinion the financial statements of the Company included in the Registration Statement, the Preliminary Prospectus and the Prospectus comply as to form in all material respects with the applicable accounting requirements of the Act and the Regulations thereunder;
- (iii) Stating that, on the basis of limited procedures which included a reading of the latest available minutes of the shareholders and board of directors and the various committees of the board of directors, consultations with officers and other employees of the Company responsible for financial and accounting matters and other specified procedures and inquiries, nothing has come to their attention which would lead them to believe that: (a) at a date not later than five (5) days prior to the Effective Date, Closing Date or Option Closing Date, as the case may be, there was any change in the share capital or long-term debt of the Company, other than as set forth in or contemplated by the Registration Statement, the Preliminary Prospectus and the Prospectus, or, if there was any decrease, setting forth the amount of such decrease; and (c) during the period from [], 2021 (balance sheet date) to a specified date not later than five (5) days prior to the Effective Date, Closing Date or Option Closing Date, as the case may be, there was any decrease in net earnings or net earnings per shares Class A Common Stock, in each case as compared with Statement of Operations for the period from [], 2021 (inception) to [], 2021, other than as set forth in or contemplated by the Registration Statement, the Preliminary Prospectus and the Prospectus, or, if there was any such decrease, setting forth the amount of such decrease;
- (iv) Stating they have compared specific dollar amounts, numbers of shares, percentages of earnings, statements and other financial information pertaining to the Company set forth in the Registration Statement, the Preliminary Prospectus and the Prospectus in each case to the extent that such amounts, numbers, percentages, statements and information may be derived from the general accounting records, including work sheets, of the Company and excluding any questions requiring an interpretation by legal counsel, with the results obtained from the application of specified readings, inquiries and other appropriate procedures (which procedures do not constitute an examination in accordance with the standards of the PCAOB) set forth in the letter and found them to be in agreement; and
 - (v) Statements as to such other matters incident to the transaction contemplated hereby as the Representative may reasonably request.

4.4. Officers' Certificates.

4.4.1. Officers' Certificate. At each of the Closing Date and the Option Closing Date, if any, the Representative shall have received a certificate of the Company signed by the Chief Executive Officer or the President and the Secretary or Assistant Secretary of the Company, dated the Closing Date or the Option Closing Date, as the case may be, to the effect that the Company has performed all covenants and complied with all conditions required by this Agreement to be performed or complied with by the Company prior to and as of the Closing Date, or the Option Closing Date, as the case may be, and that the conditions set forth in Section 4.5 hereof have been satisfied as of such date and that, as of Closing Date and the Option Closing Date, as the case may be, the representations and warranties of the Company set forth in Section 2 hereof are true and correct. In addition, the Representative will have received such other and further certificates of officers of the Company as the Representative may reasonably request.

- 4.4.2. <u>Secretary's Certificate</u>. At each of the Closing Date and the Option Closing Date, if any, the Representative shall have received a certificate of the Company signed by the Secretary or Assistant Secretary of the Company, dated the Closing Date or the Option Closing Date, as the case may be, respectively, certifying: (i) that the amended and restated Certificate of Incorporation and Bylaws of the Company filed as exhibits to the Registration Statement are true and complete, has not been modified and is in full force and effect; (ii) that the board resolutions relating to the Offering are in full force and effect and have not been modified; (iii) all correspondence between the Company or its counsel and the Commission; (iv) all correspondence between the Company or its counsel and the Nasdaq Stock Market; and (v) as to the incumbency of the officers of the Company. The documents referred to in such certificate and a good standing certificate for the Company shall be attached to such certificate.
- 4.5. No Material Changes. Prior to and on each of the Closing Date and the Option Closing Date, if any: (i) there shall have been no material adverse change or development that is likely to result in a material adverse change in the condition or prospects or the business activities, financial or otherwise, of the Company from the latest dates as of which such condition is set forth in the Registration Statement and Prospectus; (ii) no action suit or proceeding, at law or in equity, shall have been pending or threatened against the Company or any Initial Shareholder before or by any court or federal or state commission, board or other administrative agency wherein an unfavorable decision, ruling or finding may materially adversely affect the business, operations, prospects or financial condition or income of the Company, except as set forth in the Registration Statement, the Preliminary Prospectus and Prospectus; (iii) no stop order shall have been issued under the Act and no proceedings therefor shall have been initiated or threatened by the Commission; and (iv) the Registration Statement, the Preliminary Prospectus and any amendments or supplements thereto shall contain all material statements which are required to be stated therein in accordance with the Act and the Regulations and shall conform in all material respects to the requirements of the Act and the Regulations, and neither the Registration Statement, the Preliminary Prospectus nor the Prospectus nor any amendment or supplement thereto shall contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein (in the case of the Prospectus, in light of the circumstances under which they were made), not misleading.

4.6. Delivery of Agreements.

- 4.6.1. <u>Effective Date Deliveries</u>. On the Effective Date, the Company shall have delivered to the Representative executed copies of the Trust Agreement, the Warrant Agreement, the Registration Rights Agreement, all of the Insider Letters and the Subscription Agreement.
- 4.6.2. <u>Closing Date Deliveries</u>. On the Closing Date, the Company shall deliver to the Representative, or its designees, book-entry confirmation representing the Representative Shares.

5. Indemnification.

5.1. Indemnification of Underwriters.

5.1.1. General. Subject to the conditions set forth below, the Company agrees to indemnify and hold harmless each of the Underwriters and each dealer selected by the Representative that participates in the offer and sale of the Units (each a "Selected Dealer") and each of their respective directors, officers and employees and each person, if any, who controls any such Underwriter ("Controlling Person") within the meaning of Section 15 of the Act or Section 20(a) of the Exchange Act, against any and all loss, liability, claim, damage and expense whatsoever (including but not limited to any and all legal or other expenses reasonably incurred in investigating, preparing or defending against any litigation, commenced or threatened, or any claim whatsoever, whether arising out of any action between any of the Underwriters and the Company or between any of the Underwriters and any third party or otherwise) to which they or any of them may become subject under the Act, the Exchange Act or any other federal, state or local statute, law, rule, regulation or ordinance or at common law or otherwise or under the laws, rules and regulation of foreign countries, arising out of or based upon any untrue statement or alleged untrue statement of a material fact contained in: (i) any Preliminary Prospectus, the Registration Statement, or the Prospectus (as from time to time each may be amended and supplemented); (ii) in any post-effective amendment or amendments or any new registration statement and prospectus in which is included securities of the Company; or (iii) any application or other document or written communication (in this Section 5, collectively called "Application") executed by the Company or based upon written information furnished by the Company in any jurisdiction in order to qualify the Units under the securities laws thereof or filed with the Commission, any state securities commission or agency, the Nasdaq Stock Market or any securities exchange; or the omission or alleged omission therefrom of a material fact required to be stated therein or necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading, unless such statement or omission was made in reliance upon and in conformity with written information furnished to the Company with respect to an Underwriter by or on behalf of such Underwriter expressly for use in any Preliminary Prospectus, the Registration Statement, or the Prospectus, or any amendment or supplement thereof. With respect to any untrue statement or omission or alleged untrue statement or omission made in the Preliminary Prospectus, the indemnity agreement contained in this paragraph shall not inure to the benefit of any Underwriter to the extent that any loss, liability, claim, damage or expense of such Underwriter results from the fact that a copy of the Prospectus was not given or sent to the person asserting any such loss, liability, claim or damage at or prior to the written confirmation of sale of the Securities to such person as required by the Act and the Regulations, and if the untrue statement or omission has been corrected in the Prospectus, unless such failure to deliver the Prospectus was a result of non-compliance by the Company with its obligations under Section 3.4 hereof. The Company agrees to promptly notify the Representative of the commencement of any litigation or proceedings against the Company or any of its officers, directors or Controlling Persons in connection with the issue and sale of the Securities or in connection with the Preliminary Prospectus, the Registration Statement, or the Prospectus. For purposes of this Section 5, the term Underwriter or Underwriters shall refer to each Underwriter whether acting as an underwriter or as a qualified independent underwriter. The Company agrees to advance, or at the option of the Underwriters or any Controlling Person reimburse, each Underwriter or Controlling Person for all expenses as they are incurred in connection with such person's enforcement of his, her or its rights under this Agreement.

5.1.2. Procedure. If any action is brought against an Underwriter or Controlling Person in respect of which indemnity may be sought against the Company pursuant to Section 5.1.1, such Underwriter shall promptly notify the Company in writing of the institution of such action and the Company shall assume the defense of such action, including the employment and fees of counsel (subject to the reasonable approval of such Underwriter) and payment of actual expenses. Such Underwriter or Controlling Person shall have the right to employ its or their own counsel in any such case, but the fees and expenses of such counsel shall be at the expense of such Underwriter or such Controlling Person unless: (i) the employment of such counsel at the expense of the Company shall have been authorized in writing by the Company in connection with the defense of such action within reasonable time under the circumstances; (ii) the Company shall not have employed counsel to have charge of the defense of such action; or (iii) such indemnified party or parties shall have reasonably concluded that there may be defenses available to it or them which are different from or additional to those available to the Company (in which case the Company shall not have the right to direct the defense of such action on behalf of the indemnified party or parties), in any of which events the reasonable fees and expenses of not more than one additional firm of attorneys (together with local counsel) selected by the Underwriter and/or Controlling Person shall be borne by the Company. Notwithstanding anything to the contrary contained herein, if the Underwriter or Controlling Person shall assume the defense of such action as provided above, the Company shall have the right to approve the terms of any settlement of such action which approval shall not be unreasonably withheld. The advancement and reimbursement obligations of the Company required hereby regarding expenses shall be made by periodic payments of the amount thereof during the course of the investi

5.2. <u>Indemnification of the Company.</u> Each Underwriter, severally and not jointly, agrees to indemnify and hold harmless the Company, its directors, officers and employees and agents who control the Company within the meaning of Section 15 of the Act or Section 20 of the Exchange Act against any and all loss, liability, claim, damage and expense described in the foregoing indemnity from the Company to the several Underwriters, as incurred, but only with respect to untrue statements or omissions, or alleged untrue statements or omissions made in any Preliminary Prospectus, the Registration Statement, or the Prospectus, or any amendment or supplement thereto, or in any Application, in reliance upon, and in strict conformity with, written information furnished to the Company with respect to such Underwriter by or on behalf of the Underwriter expressly for use in such Preliminary Prospectus, the Registration Statement, or the Prospectus, or any amendment or supplement thereto or in any such Application, which furnished written information, it is expressly agreed, consists solely of the information described in the last sentence of <u>Section 2.3.1</u>. In case any action shall be brought against the Company or any other person so indemnified based on any Preliminary Prospectus, the Registration Statement, or the Prospectus, or any amendment or supplement thereto or any Application, and in respect of which indemnity may be sought against any Underwriter, such Underwriter shall have the rights and duties given to the Section 5.1.2.

5.3. Contribution.

5.3.1. Contribution Rights. In order to provide for just and equitable contribution under the Act in any case in which: (i) any person entitled to indemnification under this Section 5 makes a claim for indemnification pursuant hereto but it is judicially determined (by the entry of a final judgment or decree by a court of competent jurisdiction and the expiration of time to appeal or the denial of the last right of appeal) that such indemnification may not be enforced in such case notwithstanding the fact that this Section 5 provides for indemnification in such case; or (ii) contribution under the Act, the Exchange Act or otherwise may be required on the part of any such person in circumstances for which indemnification is provided under this Section 5, then, and in each such case, the Company and the Underwriters shall contribute to the aggregate losses, liabilities, claims, damages and expenses of the nature contemplated by said indemnity agreement incurred by the Company and the Underwriters, as incurred, in such proportions that the Underwriters are responsible for that portion represented by the percentage that the underwriting discount appearing on the cover page of the Prospectus bears to the initial offering price appearing thereon and the Company is responsible for the balance; provided, that, no person guilty of a fraudulent misrepresentation (within the meaning of Section 11(f) of the Act) shall be entitled to contribution from any person who was not guilty of such fraudulent misrepresentation. Notwithstanding the provisions of this Section 5.3.1, no Underwriter shall be required to contribute any amount in excess of the amount by which the total price at which the Public Securities underwritten by it and distributed to the public were offered to the public exceeds the amount of any damages that such Underwriter has otherwise been required to pay in respect of such losses, liabilities, claims, damages and expenses. For purposes of this Section, each director, officer and employee of an Underwriter or the Company, as applicable, and each person, if any, who controls an Underwriter or the Company, as applicable, within the meaning of Section 15 of the Act shall have the same rights to contribution as the Underwriters or the Company, as applicable.

5.3.2. Contribution Procedure. Within fifteen (15) days after receipt by any party to this Agreement (or its representative) of notice of the commencement of any action, suit or proceeding, such party will, if a claim for contribution in respect thereof is to be made against another party ("Contributing Party"), notify the Contributing Party of the commencement thereof, but the omission to so notify the Contributing Party will not relieve it from any liability which it may have to any other party other than for contribution hereunder. In case any such action, suit or proceeding is brought against any party, and such party notifies a Contributing Party or its representative of the commencement thereof within the aforesaid fifteen (15) days, the Contributing Party will be entitled to participate therein with the notifying party and any other Contributing Party similarly notified. Any such Contributing Party shall not be liable to any party seeking contribution on account of any settlement of any claim, action or proceeding effected by such party seeking contribution without the written consent of such Contributing Party. The contribution provisions contained in this Section are intended to supersede, to the extent permitted by law, any right to contribution under the Act, the Exchange Act or otherwise available. The Underwriters' obligations to contribute pursuant to this Section 5.3 are several and not joint.

6. Default by an Underwriter.

6.1. <u>Default Not Exceeding 10% of Firm Units or Option Units</u>. If any Underwriter or Underwriters shall default in its or their obligations to purchase the Firm Units or the Option Units, if the Over-allotment Option is exercised, hereunder, and if the number of the Firm Units or Option Units with respect to which such default relates does not exceed in the aggregate 10% of the number of Firm Units or Option Units that all Underwriters have agreed to purchase hereunder, then such Firm Units or Option Units to which the default relates shall be purchased by the non-defaulting Underwriters in proportion to their respective commitments hereunder.

- 6.2. <u>Default Exceeding 10% of Firm Units or Option Units</u>. In the event that the default addressed in <u>Section 6.1</u> above relates to more than 10% of the Firm Units or Option Units, the Representative may in its discretion arrange for itself or for another party or parties to purchase such Firm Units or Option Units to which such default relates on the terms contained herein. If, within one (1) Business Day after such default relating to more than 10% of the Firm Units or Option Units, the Representative does not arrange for the purchase of such Firm Units or Option Units, then the Company shall be entitled to a further period of one (1) Business Day within which to procure another party or parties satisfactory to the Company and the Representative to purchase said Firm Units or Option Units on such terms. In the event the Representative does not arrange for the purchase of the Firm Units or Option Units to which a default relates as provided in this Section 6, this Agreement may be terminated by the Company without liability on the part of the Company (except as provided in <u>Sections 3.12</u> and <u>5</u> hereof) or the several Underwriters (except as provided in <u>Section 5</u> hereof); *provided*, *however*, that if such default occurs with respect to the Option Units, this Agreement will not terminate as to the Firm Units; and *provided further* that nothing herein shall relieve a defaulting Underwriter of its liability, if any, to the other several Underwriters and to the Company for damages occasioned by its default hereunder.
- 6.3. <u>Postponement of Closing Date</u>. In the event the Firm Units or Option Units to which the default relates are to be purchased by the non-defaulting Underwriters, or are to be purchased by another party or parties as aforesaid, the Representative or the Company shall have the right to postpone the Closing Date or Option Closing Date for a reasonable period, but not in any event exceeding five (5) Business Days, in order to effect whatever changes may thereby be made necessary in the Registration Statement, the Preliminary Prospectus and/or the Prospectus, as the case may be, or in any other documents and arrangements, and the Company agrees to file promptly any amendment to, or to supplement, the Registration Statement, the Preliminary Prospectus and/or the Prospectus, as the case may be, that in the opinion of counsel for the Underwriters may thereby be made necessary. The term "Underwriter" as used in this Agreement shall include any party substituted under this Section 6 with like effect as if it had originally been a party to this Agreement with respect to such Securities.

7. Additional Covenants.

- 7.1. <u>Additional Shares or Options</u>. The Company hereby agrees that until the Company consummates a Business Combination, it shall not issue any Class A Common Stock or any options or other securities convertible into Class A Common Stock, or any class of shares which participate in any manner in the Trust Account or which vote as a class with the Class A Common Stock on a Business Combination.
- 7.2. Trust Account Waiver Acknowledgments. The Company hereby agrees that it will not commence its due diligence investigation of any operating business or businesses which the Company seeks to acquire (each, a "Target Business") unless and until such Target Business acknowledges in writing, whether through a letter of intent, memorandum of understanding or other similar document (and subsequently acknowledges the same in any definitive document replacing any of the foregoing), that: (i) it has read the Prospectus and understands that the Company has established the Trust Account, initially in an amount of \$102,000,000 (or \$117,300,000 if the Over-allotment Option is exercised in full) for the benefit of the public shareholders, and that (ii) for and in consideration of the Company agreeing to evaluate such Target Business for purposes of consummating a Business Combination with it, such Target Business agrees that it does not have any right, title, interest or claim of any kind in or to any monies of the Trust Account ("Claim") and waives any Claim it may have in the future as a result of, or arising out of, any negotiations, contracts or agreements with the Company and will not seek recourse against the Trust Account for any reason whatsoever. The Company further agrees that it will use its best efforts, prior to obtaining the services of any vendor, to obtain a written acknowledgment from such vendor, whether through a letter of intent, memorandum of understanding or other similar document (and subsequently acknowledges the same in any definitive document replacing any of the foregoing), that: (i) such vendor has read the Prospectus and understands that the Company has established the Trust Account, initially in an amount of \$102,000,000 (or \$117,300,000 if the Overallotment Option is exercised in full) for the benefit of the public shareholders, and that (ii) for and in consideration of the Company agreeing to engage the services of the vendor, such vendor agrees that it does not have any Claim and waives any Claim it may have in the future as a result of, or arising out of, any negotiations, contracts or agreements with the Company and will not seek recourse against the Trust Account for any reason whatsoever. The foregoing letters shall substantially be in the form attached hereto as Exhibit A and B, respectively.

- 7.3. <u>Insider Letters</u>. The Company shall not take any action or omit to take any action which would cause a breach of any of the Insider Letters executed between each of the Company's officers, directors and Initial Shareholders or the Subscription Agreement and will not allow any amendments to, or waivers of, such Insider Letters or the Subscription Agreement without the prior written consent of the Representative.
- 7.4. <u>Certificate of Incorporation</u>. The Company shall not take any action or omit to take any action that would cause the Company to be in breach or violation of its Certificate of Incorporation. Except as provided in <u>Section 3.28</u>, prior to the consummation of a Business Combination, the Company will not amend its Certificate of Incorporation, without the prior written consent of the Representative.
- 7.5. <u>Tender Offer Documents, Proxy Materials and Other Information</u>. The Company shall provide counsel to the Representative with copies of all tender offer documents or proxy information and all related material filed with the Commission in connection with a Business Combination concurrently with such filing with the Commission. In addition, the Company shall furnish any other state in which the Offering was registered, such information as may be requested by such state.
- 7.6. <u>Acquisition/Liquidation Procedure</u>. The Company agrees that it will comply with Article IX of its Certificate of Incorporation in connection with the consummation of a Business Combination or the failure to consummate a Business Combination within 12 months from the Effective Date (or up to 18 months from the closing of this Offering if the Company extends the period of time to consummate an initial Business Combination by the full amount of time).
- 7.7. <u>Rule 419</u>. The Company agrees that it will use its best efforts to prevent the Company from becoming subject to Rule 419 under the Act prior to the consummation of any Business Combination, including, but not limited to, using its best efforts to prevent any of the Company's outstanding securities from being deemed to be a "penny stock" as defined in Rule 3a-51-1 under the Exchange Act during such period.
- 7.8. <u>Presentation of Potential Target Businesses</u>. The Company shall cause each of the Initial Shareholders to agree that, in order to minimize potential conflicts of interest which may arise from multiple affiliations, the Initial Shareholders will present to the Company for its consideration, prior to presentation to any other person or company, any suitable opportunity to acquire an operating business, until the earlier of the consummation by the Company of a Business Combination or the liquidation of the Company, subject to any pre-existing fiduciary obligations the Initial Shareholders might have.
- 8. <u>Representations and Agreements to Survive Delivery.</u> Except as the context otherwise requires, all representations, warranties and agreements contained in this Agreement shall be deemed to be representations, warranties and agreements at the Closing Date or the Option Closing Date, if any, and such representations, warranties and agreements of the Underwriters and the Company, including the indemnity agreements contained in Section 5 hereof, shall remain operative and in full force and effect regardless of any investigation made by or on behalf of any Underwriter, the Company or any Controlling Person, and shall survive termination of this Agreement or the issuance and delivery of the Securities to the several Underwriters until the earlier of the expiration of any applicable statute of limitations and the seventh anniversary of the later of the Closing Date or the Option Closing Date, if any, at which time the representations, warranties and agreements shall terminate and be of no further force and effect.
- 9. Effective Date of This Agreement and Termination Thereof.
- 9.1. Effective Date. This Agreement shall become effective on the Effective Date at the time the Registration Statement is declared effective by the Commission.

- 9.2. Termination. The Representative shall have the right to terminate this Agreement at any time prior to any Closing Date: (i) if any domestic or international event or act or occurrence has materially disrupted, or in the Representative's opinion will in the immediate future materially disrupt, general securities markets in the United States; or (ii) if trading on the New York Stock Exchange, the NYSE American or the Nasdaq Stock Market shall have been suspended, or minimum or maximum prices for trading shall have been fixed, or maximum ranges for prices for securities shall have been required on the over the counter markets or by order of the Commission or any other government authority having jurisdiction; or (iii) if the United States shall have become involved in a war or an initiation or increase in major hostilities, or (iv) if a banking moratorium has been declared by a New York State or federal authority, or (v) if a moratorium on foreign exchange trading has been declared which materially adversely impacts the United States securities markets; or (vi) if the Company shall have sustained a material loss by fire, flood, accident, hurricane, earthquake, theft, sabotage or other calamity or malicious act which, whether or not such loss shall have been insured, will, in the Representative's opinion, make it inadvisable to proceed with the delivery of the Units; or (vii) if any of the Company's representations, warranties or covenants hereunder are breached in any material respect; or (viii) if the Representative shall have become aware after the date hereof of such a material adverse change in the conditions or prospects of the Company, or such material adverse change in general market conditions, including, without limitation, as a result of terrorist activities after the date hereof, as in the Representative's judgment would make it impracticable to proceed with the offering, sale and/or delivery of the Units or to enforce contracts made by the Underwriters for the sale of the Units.
- 9.3. Expenses. In the event this Agreement shall not be carried out for any reason whatsoever, except as a result of the Representative's or any Underwriters' breach or default with respect to any of its material obligations pursuant to this Agreement, within the time specified herein or any extensions thereof pursuant to the terms herein, the obligations of the Company to pay the out-of-pocket expenses actually incurred by the Representative related to the transactions contemplated herein shall be governed by Section 3.12 hereof.
- 9.4. <u>Indemnification</u>. Notwithstanding any contrary provision contained in this Agreement, any election hereunder or any termination of this Agreement, and whether or not this Agreement is otherwise carried out, the provisions of Section 5 shall not be in any way affected by such election or termination or failure to carry out the terms of this Agreement or any part hereof.

10. Miscellaneous.

10.1. <u>Notices</u>. All communications hereunder, except as herein otherwise specifically provided, shall be in writing and shall be mailed, delivered by hand or reputable overnight courier or delivered by facsimile transmission (with printed confirmation of receipt) and confirmed, or by electronic transmission via PDF, and shall be deemed given when so mailed, delivered or faxed or transmitted (or if mailed, three (3) days after such mailing):

If to the Representative:

EF Hutton, division of Benchmark Investments, LLC 590 Madison Avenue, 39th Floor
New York, NY 10022

Attn: Mr. Joseph T. Rallo, Chief Executive Officer

Fax: (212) 404-7002

Email: jrallo@efhuttongroup.com

Copy to (which copy shall not be deemed to constitute notice to the Representative):

Schiff Hardin LLP 901 K Street NW Suite 700 Washington, DC 20001 Attn: Cavas Pavri, Esq.

Email: cpavri@schiffhardin.com

Fax: (202) 778-6460

If to the Company:

Aesther Healthcare Acquisition Corp. 515 Madison Avenue, Suite 8078 New York, New York 10022

Attn: Suren Ajjarapu

Email: Suren@aestherhealthcarespac.com

Copy to (which copy shall not be deemed to constitute notice to the Company):

The Loev Law Firm, PC 6300 West Loop South Suite 280 Bellaire, Texas 77401 Attn: David M. Loev, Esq.

Facsimile: (713) 524-4122 Email: dloev@loevlaw.com

- 10.2. <u>Headings</u>. The headings contained herein are for the sole purpose of convenience of reference, and shall not in any way limit or affect the meaning or interpretation of any of the terms or provisions of this Agreement.
 - 10.3. Amendment. This Agreement may only be amended by a written instrument executed by each of the parties hereto.
- 10.4. Entire Agreement. This Agreement (together with the other agreements and documents being delivered pursuant to or in connection with this Agreement) constitutes the entire agreement of the parties hereto with respect to the subject matter hereof (and thereof), and supersedes all prior agreements and understandings of the parties, oral and written, with respect to the subject matter hereof.
- 10.5. <u>Binding Effect</u>. This Agreement shall inure solely to the benefit of and shall be binding upon the Representative, the Underwriters, the Company and the Controlling Persons, directors and officers referred to in Section 5 hereof, and their respective successors, legal representatives and assigns, and no other person shall have or be construed to have any legal or equitable right, remedy or claim under or in respect of or by virtue of this Agreement or any provisions herein contained.

10.6. Governing Law, Venue, etc.

10.6.1. This Agreement shall be governed by and construed and enforced in accordance with the internal laws of the State of New York, without giving effect to the conflict of laws principles thereof. Each of the Representative and the Company (and any individual signatory hereto): (i) agrees that any legal suit, action or proceeding arising out of or relating to this Agreement and/or the transactions contemplated hereby shall be instituted exclusively in New York Supreme Court, County of New York, or in the United States District Court for the Southern District of New York; (ii) waives any objection which such party may have or hereafter have to the venue of any such suit, action or proceeding; and (iii) irrevocably and exclusively consents to the jurisdiction of the New York Supreme Court, County of New York, and the United States District Court for the Southern District of New York in any such suit, action or proceeding.

10.6.2. Each of the Representative and the Company (and any individual signatory hereto) further agrees to accept and acknowledge service of any and all process which may be served in any such suit, action or proceeding in the New York Supreme Court, County of New York, or in the United States District Court for the Southern District of New York and agrees that service of process upon the Company or any such individual mailed by certified mail to the Company's address shall be deemed in every respect effective service of process upon the Representative's addresses shall be deemed in every respect effective service process upon the Representative, in any such suit, action or proceeding.

10.6.3. THE COMPANY (ON BEHALF OF ITSELF AND, TO THE FULLEST EXTENT PERMITTED BY LAW, ON BEHALF OF ITS EQUITY HOLDERS AND CREDITORS) HEREBY WAIVES ANY RIGHT TO A TRIAL BY JURY IN RESPECT OF ANY CLAIM BASED UPON, ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT AND THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT, THE REGISTRATION STATEMENT AND THE PROSPECTUS.

10.6.4. Each of the Company and Representative agrees that the prevailing party(ies) in any such action shall be entitled to recover from the other party(ies) all of its reasonable attorneys' fees and expenses relating to such action or proceeding and/or incurred in connection with the preparation therefor.

- 10.7. Execution in Counterparts. This Agreement may be executed in one or more counterparts, and by the different parties hereto in separate counterparts, each of which shall be deemed to be an original, but all of which taken together shall constitute one and the same agreement, and shall become effective when one or more counterparts has been signed by each of the parties hereto and delivered to each of the other parties hereto. Delivery of a signed counterpart of this Agreement by fax or email/.pdf transmission shall constitute valid and sufficient delivery thereof.
- 10.8. Waiver, Etc. The failure of any of the parties hereto to at any time enforce any of the provisions of this Agreement shall not be deemed or construed to be a waiver of any such provision, nor to in any way effect the validity of this Agreement or any provision hereof or the right of any of the parties hereto to thereafter enforce each and every provision of this Agreement. No waiver of any breach, non-compliance or non-fulfillment of any of the provisions of this Agreement shall be effective unless set forth in a written instrument executed by the party or parties against whom or which enforcement of such waiver is sought; and no waiver of any such breach, non-compliance or non-fulfillment shall be construed or deemed to be a waiver of any other or subsequent breach, non-compliance or non-fulfillment.
- 10.9. No Fiduciary Relationship. The Company hereby acknowledges that the Underwriters are acting solely as underwriters in connection with the Offering. The Company further acknowledges that the Underwriters are acting pursuant to a contractual relationship created solely by this Agreement entered into on an arm's length basis and in no event do the parties intend that the Underwriters act or be responsible as a fiduciary to the Company, its management, shareholders, creditors or any other person in connection with any activity that the Underwriters may undertake or have undertaken in furtherance of the Offering, either before or after the date hereof. The Underwriters hereby expressly disclaim any fiduciary or similar obligations to the Company, either in connection with the transactions contemplated by this Agreement or any matters leading up to such transactions, and the Company hereby confirms its understanding and agreement to that effect. The Company and the Underwriters agree that they are each responsible for making their own independent judgments with respect to any such transactions, and that any opinions or views expressed by the Underwriters to the Company regarding such transactions, including but not limited to any opinions or views with respect to the price or market for the Company's securities, do not constitute advice or recommendations to the Company. The Company hereby waives and releases, to the fullest extent permitted by law, any claims that the Company may have against the Underwriters with respect to any breach or alleged breach of any fiduciary or similar duty to the Company in connection with the transactions contemplated by this Agreement or any matters leading up to such transactions.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

If the foregoing correctly sets forth the understanding be provided below for that purpose, whereupon this letter shall constitute a bind	between the Underwriters and the Company, please so indicate in the space ling agreement between us.					
	Very truly yours,					
	AESTHER HEALTHCARE ACQUISITION CORP.					
	Ву:					
	Name:					
	Title:					
Agreed to and accepted on the date first above written.						
EF Hutton, division of Benchmark Investments, LLC, as Representative of the several Underwriters						
By:						
Name:	_					

Title:

SCHEDULE A

AESTHER HEALTHCARE ACQUISITION CORP.

10,000,000 Units

	Number of Firm
	Units
Underwriter	to be Purchased
EF Hutton, division of Benchmark Investments, LLC	_
TOTAL	10,000,000

SCHEDULE B

AESTHER HEALTHCARE ACQUISITION CORP.

Written Communications

EXHIBIT A

Form of Target Business Letter

Aesther Healthcare Acquisition Corp. 515 Madison Avenue, Suite 8078 New York, New York 10022

Ladies and Gentlemen:

Reference is made to the Final Prospectus of Aesther Healthcare Acquisition Capitalized terms used and not otherwise defined herein shall have the meanings assign	1 \ 1 \ 1 \ 7			
We have read the Prospectus and understand that the Company has established \$117,300,000 if the Over-allotment Option is exercised in full) for the benefit of the public offering (the " Underwriters ") and that, except for interest earned on the trust incurs, proceeds in the trust account will not be released until (a) the consummation of Company if it is unable to consummate a Business Combination within the allotted time	"public shareholders" and the underwriters of the Company's initial taccount that may be released to the Company to pay any taxes is a Business Combination, or (b) the dissolution and liquidation of the			
For and in consideration of the Company agreeing to evaluate the undersigned for purposes of consummating a business combination or other form of acquisition with it, the undersigned hereby agrees that it does not have any right, title, interest or claim of any kind in or to any monies in the trust account (the "Claim") and hereby waives any Claim it may have in the future as a result of, or arising out of, any negotiations, contracts or agreement with the Company and will not seek recourse against the trust account for any reason whatsoever.				
Print Nar	ne of Target Business			
Authoriz	ed Signature of Target Business			

EXHIBIT B

Form of Vendor Letter

Aesther Healthcare Acquisition Corp. 515 Madison Avenue, Suite 8078 New York, New York 10022 Ladies and Gentlemen:

Zaares and Gendement	
Reference is made to the Final Prospectus of Aesther Healthce Capitalized terms used and not otherwise defined herein shall have the	rare Acquisition Corp. (the " Company "), dated, 2021 (the " Prospectus ") meanings assigned to them in the Prospectus.
\$117,300,000 if the Over-allotment Option is exercised in full) for the public offering (the " Underwriters ") and that, except for interest ear	y has established a "trust account", initially in an amount of at least \$102,000,000 (or be benefit of the "public shareholders" and the underwriters of the Company's initial med on the trust account that may be released to the Company to pay any taxes it consummation of a Business Combination, or (b) the dissolution and liquidation of the athe allotted time.
not have any right, title, interest or claim of any kind in or to any monie	e products or services of the undersigned, the undersigned hereby agrees that it does es in the trust account (the "Claim") and hereby waives any Claim it may have in the greements with the Company and will not seek recourse against the trust account for
	Print Name of Vendor
	Authorized Signature of Vendor

State of Delaware Secretary of State Division of Corporations Delivered 05:38 PM 06/17/2021 FILED 05:38 PM 06/17/2021 SR 2012484845 - File Number 6012147

CERTIFICATE OF INCORPORATION OF AESTHER HEALTHCARE ACQUISITION CORP. (a Delaware corporation)

June 17, 2021

The undersigned, for the purposes of forming a corporation pursuant to the General Corporation Law of the State of Delaware ("<u>DGCL</u>"), does make, file, and record this Certificate of Incorporation (the "<u>Certificate</u>"), and does hereby certify as follows:

ARTICLE I NAME

The name of the corporation is Aesther Healthcare Acquisition Corp. (the "Corporation").

ARTICLE II PURPOSE

The purpose of the Corporation is to engage in any lawful act or activity for which corporations may be organized under DGCL.

ARTICLE III REGISTERED AGENT

The Registered Office of the corporation in the State of Delaware is located at 8 The Green STE A, in the City of Dover, County of Kent County, Zip Code 19901. The name of the Registered Agent at such address upon whom process against this corporation may be served is A Registered Agent, Inc.

ARTICLE IV CAPITALIZATION

Section 4.1 <u>Authorized Capital Stock</u>. The total number of shares of all classes of capital stock, each with a par value of \$0.0001 per share, which the Corporation is authorized to issue is 138,750,000 shares, consisting of (a) 137,500,000 shares of common stock (the "<u>Common Stock</u>"), including (i) 125,000,000 shares of Class A Common Stock (the "<u>Class A Common Stock</u>"), and (ii) 12,500,000 shares of Class B Common Stock (the "<u>Class B Common Stock</u>"), and (b) 1,250,000 shares of preferred stock (the "<u>Preferred Stock</u>").

Section 4.2 <u>Preferred Stock</u>. The board of directors of the Corporation (the "<u>Board</u>") is hereby expressly authorized to provide, out of the unissued shares of the Preferred Stock, one or more series of Preferred Stock, and to establish from time to time the number of shares to be included in each such series and to fix the voting rights, if any, designations, powers, preferences and relative, participating, optional, special and other rights, if any, of each such series and any qualifications, limitations and restrictions thereof, as shall be stated in the resolution or resolutions adopted by the Board providing for the issuance of such series and included in a certificate of designation (a "<u>Preferred Stock Designation</u>") filed pursuant to the DGCL, and the Board is hereby expressly vested with the authority to the full extent provided by law, now or hereafter, to adopt any such resolution or resolutions.

Section 4.3 Common Stock.

- (a) Voting.
- (i) Except as otherwise required by law or this Certificate (including any Preferred Stock Designation), the holders of the Common Stock shall exclusively possess all voting power with respect to the Corporation.
- (ii) Except as otherwise required by law or this Certificate (including any Preferred Stock Designation), the holders of shares of Common Stock shall be entitled to one vote for each such share on each matter properly submitted to the stockholders on which the holders of the Common Stock are entitled to vote.
- (iii) Except as otherwise required by law or this Certificate (including any Preferred Stock Designation), at any annual or special meeting of the stockholders of the Corporation, holders of the Class A Common Stock and holders of the Class B Common Stock, voting together as a single class, shall have the exclusive right to vote for the election of directors and on all other matters properly submitted to a vote of the stockholders. Notwithstanding the foregoing, except as otherwise required by law or this Certificate (including any Preferred Stock Designation), holders of shares of any series of Common Stock shall not be entitled to vote on any amendment to this Certificate (including any amendment to any Preferred Stock Designation) that relates solely to the terms of one or more outstanding series of Preferred Stock or other series of Common Stock if the holders of such affected series of Preferred Stock or Common Stock, as applicable, are entitled, either separately or together with the holders of one or more other such series, to vote thereon pursuant to this Certificate (including any Preferred Stock Designation) or the DGCL.
- (b) Class B Common Stock.
- (i) Shares of Class B Common Stock shall be convertible into shares of Class A Common Stock on a one-for-one basis (the "**Initial Conversion Ratio**") automatically upon the consummation of the Business Combination (as defined below).
- (ii) Notwithstanding the Initial Conversion Ratio, in the case that additional shares of Class A Common Stock, or equity-linked securities, are issued or deemed issued in excess of the amounts sold in the Corporation's initial public offering of securities and related to the closing of the initial Business Combination, all issued and outstanding shares of Class B Common Stock shall automatically convert into shares of Class A Common Stock at the time of the closing of the Corporation's initial merger, capital stock exchange, asset acquisition, stock purchase, reorganization or similar business combination with one or more businesses (the "Business Combination") at a ratio for which:

Certificate of Incorporation of Aesther Healthcare Acquisition Corp. Page 2 of 8

- the numerator shall be equal to the sum of (A) 20% of all shares of Class A Common Stock issued or issuable (upon the conversion or exercise of any equity-linked securities or otherwise) by the Corporation, related to or in connection with the consummation of the initial Business Combination (excluding any securities issued or issuable to any seller in the initial Business Combination) plus (B) the number of shares of Class B Common Stock issued and outstanding prior to the closing of the initial Business Combination; and
- the denominator shall be the number of shares of Class B Common Stock issued and outstanding prior to the closing of the initial Business.

Notwithstanding anything to the contrary contained herein, (i) the foregoing adjustment to the Initial Conversion Ratio may be waived as to any particular issuance or deemed issuance of additional shares of Class A Common Stock or equity-linked securities by the written consent or agreement of holders of a majority of the shares of Class B Common Stock then outstanding consenting or agreeing separately as a single class in the manner provided in Section 4.3(b)(iii), and (ii) in no event shall the Class B Common Stock convert into Class A Common Stock at a ratio that is less than one-for-one.

The foregoing conversion ratio shall also be adjusted to account for any subdivision (by stock split, subdivision, exchange, stock dividend, reclassification, recapitalization or otherwise) or combination (by reverse stock split, exchange, reclassification, recapitalization or otherwise) or similar reclassification or recapitalization of the outstanding shares of Class A Common Stock into a greater or lesser number of shares occurring after the original filing of this Certificate without a proportionate and corresponding subdivision, combination or similar reclassification or recapitalization of the outstanding shares of Class B Common Stock.

Each share of Class B Common Stock shall convert into its pro rata number of shares of Class A Common Stock pursuant to this Section 4.3(b). The pro rata share for each holder of Class B Common Stock will be determined as follows: Each share of Class B Common Stock shall convert into such number of shares of Class A Common Stock as is equal to the product of one (1) multiplied by a fraction, the numerator of which shall be the total number of shares of Class A Common Stock into which all of the issued and outstanding shares of Class B Common Stock shall be converted pursuant to this Section 4.3(b) and the denominator of which shall be the total number of issued and outstanding shares of Class B Common Stock at the time of conversion.

Certificate of Incorporation of Aesther Healthcare Acquisition Corp. Page 3 of 8

(iii) Voting. Except as otherwise required by law or this Certificate (including any Preferred Stock Designation), for so long as any shares of Class B Common Stock shall remain outstanding, the Corporation shall not, without the prior vote or written consent of the holders of a majority of the shares of Class B Common Stock then outstanding, voting separately as a single class, amend, alter or repeal any provision of this Certificate, whether by merger, consolidation or otherwise, if such amendment, alteration or repeal would alter or change the powers, preferences or relative, participating, optional or other or special rights of the Class B Common Stock. Any action required or permitted to be taken at any meeting of the holders of Class B Common Stock may be taken without a meeting, without prior notice and without a vote, if a consent or consents in writing, setting forth the action so taken, shall be signed by the holders of the outstanding Class B Common Stock having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares of Class B Common Stock were present and voted and shall be delivered to the Corporation by delivery to its registered office in the State of Delaware, its principal place of business, or an officer or agent of the Corporation having custody of the book in which minutes of proceedings of stockholders are recorded. Delivery made to the Corporation's registered office shall be by hand or by certified or registered mail, return receipt requested. Prompt written notice of the taking of corporate action without a meeting by less than unanimous written consent of the holders of Class B Common Stock shall, to the extent required by law, be given to those holders of Class B Common Stock who have not consented in writing and who, if the action had been taken at a meeting, would have been entitled to notice of the meeting if the record date for notice of such meeting had been the date that written consents signed by a sufficient number of holders of Class B Common Stock to take the action were delivered to the Corporation.

(c) *Dividends*. Subject to applicable law, the rights, if any, of the holders of any outstanding series of the Preferred Stock, the holders of shares of Common Stock shall be entitled to receive such dividends and other distributions (payable in cash, property or capital stock of the Corporation) when, as and if declared thereon by the Board from time to time out of any assets or funds of the Corporation legally available therefor and shall share equally on a per share basis in such dividends and distributions.

(d) *Liquidation, Dissolution or Winding Up of the Corporation.* Subject to applicable law, the rights, if any, of the holders of any outstanding series of the Preferred Stock, in the event of any voluntary or involuntary liquidation, dissolution or winding up of the Corporation, after payment or provision for payment of the debts and other liabilities of the Corporation, the holders of shares of Common Stock shall be entitled to receive all the remaining assets of the Corporation available for distribution to its stockholders, ratably in proportion to the number of shares of Class A Common Stock (on an as converted basis with respect to the Class B Common Stock) held by them.

Section 4.4 <u>Rights and Options</u>. The Corporation has the authority to create and issue rights, warrants and options entitling the holders thereof to acquire from the Corporation any shares of its capital stock of any class or classes, with such rights, warrants and options to be evidenced by or in instrument(s) approved by the Board. The Board is empowered to set the exercise price, duration, times for exercise and other terms and conditions of such rights, warrants or options; provided, however, that the consideration to be received for any shares of capital stock issuable upon exercise thereof may not be less than the par value thereof.

Certificate of Incorporation of Aesther Healthcare Acquisition Corp. Page 4 of 8

ARTICLE V INCORPORATOR

The name and mailing address of the sole incorporator of the Corporation are as follows:

Name

John S. Gillies

c/o The Loev Law Firm, PC
6300 West Loop South, Suite 280
Bellaire, Texas 77401

ARTICLE VI DIRECTORS

- A. <u>Power of the Board of Directors</u>. The business and affairs of the Company shall be managed by or under the direction of the Board of Directors. In furtherance, and not in limitation, of the powers conferred by the laws of the State of Delaware, the Board of Directors shall be expressly authorized to:
 - determine the rights, powers, duties, rules and procedures that affect the power of the Board of Directors to manage and direct the business and affairs of the Company;
 - 2. establish one or more committees of the Board of Directors, by the affirmative vote of a majority of the Board of Directors, to which may be delegated any or all of the powers and duties of the Board of Directors to the fullest extent permitted by law; and
 - 3. exercise all such powers and do all such acts as may be exercised by the Company, subject to the provisions of the laws of the State of Delaware, this Certificate, and the Bylaws of the Company (as the same may be amended and/or restated from time to time, the "Bylaws").
- B. <u>Number of Directors</u>. The number of directors constituting the Board of Directors shall be fixed from time to time exclusively by a vote of a majority of the Board of Directors in the manner provided in the Bylaws.
- C. <u>Vacancies</u>. Except as otherwise required by law and subject to the rights of the holders of any class or series of Preferred Stock to elect directors, any vacancies on the Board of Directors for any reason, including from the death, resignation, disqualification or removal of any director, and any newly created directorships resulting by reason of any increase in the number of directors may be filled exclusively by the Board of Directors, acting by the affirmative vote of a majority of the remaining directors then in office, even if less than a quorum, or by a sole remaining director, and shall not be required to be filled by stockholders. Any directors elected to fill a vacancy shall hold office until the next annual meeting of stockholders or until their successors are duly elected and qualified.

Certificate of Incorporation of Aesther Healthcare Acquisition Corp. Page 5 of 8

- D. <u>Removal of Directors</u>. Except as otherwise required by law and subject to the rights of the holders of any class or series of Preferred Stock, any director, or the entire Board of Directors, may be removed from office at any time, with or without cause only by the affirmative vote of the holders of a majority of the voting power of all of the shares of capital stock of the Company then entitled to vote generally in the election of directors, voting as a single class.
- E. <u>Appointment of Directors</u>. Elections of directors need not be by written ballot unless the Bylaws of the Company shall so provide. There shall be no cumulative voting in the election of directors.

ARTICLE VII BYLAWS

- A. <u>Amendment by the Board of Directors</u>. In furtherance, and not in limitation, of the powers conferred upon it by law, the Board of Directors is expressly authorized and empowered to amend, alter, change, adopt or repeal the Bylaws of the Company; <u>provided</u>, <u>however</u>, that no Bylaws hereafter adopted shall invalidate any prior act of the directors that would have been valid if such Bylaws had not been adopted.
- B. <u>Amendment by Stockholders</u>. In addition to any requirements of DGCL (and notwithstanding the fact that a lesser percentage may be specified by DGCL), unless otherwise specified in the Bylaws, the affirmative vote of the holders of a majority of the voting power of all the shares of capital stock of the Company then entitled to vote generally in the election of directors, voting together as a single class, shall be required for the stockholders of the Company to amend, alter, change or repeal or to adopt any provision of the Bylaws of the Company.

ARTICLE VIII LIMITED LIABILITY; INDEMNIFICATION

Section 8.1 <u>Limitation of Director Liability</u>. A director of the Corporation shall not be personally liable to the Corporation or its stockholders for monetary damages for any breach of fiduciary duty by such director as a director, except for liability (i) for any breach of the director's duty of loyalty to the Corporation or its stockholders, (ii) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) under <u>Section 174</u> of the DGCL, or (iv) for any transaction from which the director derived an improper personal benefit. If the DGCL is amended to authorize corporate action further eliminating or limiting the personal liability of directors, then the liability of a director of the Corporation shall be eliminated or limited to the fullest extent permitted by the DGCL, as so amended. Any repeal or modification of this <u>Section 8.1</u> by the stockholders of the Corporation shall not adversely affect any right or protection of a director of the Corporation with respect to events occurring prior to the time of such repeal or modification.

Certificate of Incorporation of Aesther Healthcare Acquisition Corp. Page 6 of 8 Section 8.2 <u>Indemnification</u>. The Corporation, to the full extent permitted by <u>Section 145</u> of the DGCL, as amended from time to time, shall indemnify all persons whom it may indemnify pursuant thereto. Expenses (including attorneys' fees) incurred by an officer or director in defending any civil, criminal, administrative, or investigative action, suit or proceeding for which such officer or director may be entitled to indemnification hereunder shall be paid by the Corporation in advance of the final disposition of such action, suit or proceeding upon receipt of an undertaking by or on behalf of such director or officer to repay such amount if it shall ultimately be determined that he is not entitled to be indemnified by the Corporation as authorized hereby.

ARTICLE IX INSOLVENCY; SALE, LEASE OR EXCHANGE OF ASSETS

Whenever a compromise or arrangement is proposed between this Corporation and its creditors or any class of them and/or between this Corporation and its stockholders or any class of them, any court of equitable jurisdiction within the State of Delaware may, on the application in a summary way of this Corporation or of any creditor or stockholder thereof or on the application of any receiver or receivers appointed for this Corporation under Section 291 of Title 8 of the DGCL or on the application of trustees in dissolution or of any receiver or receivers appointed for this Corporation under Section 279 of Title 8 of the DGCL order a meeting of the creditors or class of creditors, and/or of the stockholders or class of stockholders or this Corporation, as the case may be, to be summoned in such manner as the said court directs. If a majority in number representing three fourths in value of the creditors or class of creditors, and/or of the stockholders or class of stockholders of this Corporation, as the case may be, agree to any compromise or arrangement and to any reorganization of this Corporation as a consequence of such compromise or arrangement, the said compromise or arrangement and the said reorganization shall, if sanctioned by the court to which the said application has been made, be binding on all the creditors or class of creditors, and/or on all the stockholders or class of stockholders, of this Corporation, as the case may be, and also on this Corporation.

ARTICLE X AMENDMENT OF CERTIFICATE OF INCORPORATION

The Corporation reserves the right to amend, alter, change, add or repeal any provision contained in this Certificate (including any Preferred Stock Designation), in the manner now or hereafter prescribed by this Certificate and the DGCL; and except as set forth in <u>ARTICLE VIII</u>, all rights, preferences and privileges herein conferred upon stockholders, directors or any other persons by and pursuant to this Certificate in its present form or as hereafter amended are granted subject to the right reserved in this Article.

ARTICLE X POWERS OF INCORPORATOR; INITIAL DIRECTORS

A. <u>Powers of Incorporator; Initial Directors of the Company</u>. The powers of the incorporator are to terminate upon the filing of this Certificate with the Secretary of State of the State of Delaware. The name and mailing address of each person who is to serve as an initial director of the Company until the first annual meeting of stockholders or until their successors are duly elected and qualified, is:

Suren Ajjarapu 515 Madison Avenue Floor 8 New York, New York 10022

> Certificate of Incorporation of Aesther Healthcare Acquisition Corp. Page 7 of 8

IN WITNESS WHEREOF, the undersigned has duly executed this Certificate of Incorporation as of the date first written above.

By: /s/ John Gillies
Name: John Gillies

Sole Incorporator

Certificate of Incorporation of Aesther Healthcare Acquisition Corp. Page 8 of 8

AMENDED AND RESTATED CERTIFICATE OF INCORPORATION OF AESTHER HEALTHCARE ACQUISITION CORP. [________], 2021

Aesther Healthcare Acquisition Corp., a corporation organized and existing under the laws of the State of Delaware (the "Corporation"), DOES HEREBY CERTIFY AS FOLLOWS:

- 1. The name of the Corporation is "<u>Aesther Healthcare Acquisition Corp.</u>" The original certificate of incorporation of the Corporation was filed with the Secretary of State of the State of Delaware on June 17, 2021 (the "<u>Certificate</u>").
- 2. This Amended and Restated Certificate of Incorporation (this "<u>Amended and Restated Certificate</u>"), which both restates and amends the provisions of the Certificate, was duly adopted in accordance with Sections 228, 242 and 245 of the General Corporation Law of the State of Delaware, as amended from time to time (the "<u>DGCL</u>").
 - 3. This Amended and Restated Certificate shall become effective on the date of filing with Secretary of State of Delaware.
 - 4. Certain capitalized terms used in this Amended and Restated Certificate are defined where appropriate herein.
 - 5. The text of the Certificate, as amended to the date hereof, is hereby restated and amended in its entirety to read as follows:

ARTICLE I NAME

The name of the corporation is Aesther Healthcare Acquisition Corp. (the "Corporation").

ARTICLE II PURPOSE

The purpose of the Corporation is to engage in any lawful act or activity for which corporations may be organized under the DGCL. In addition to the powers and privileges conferred upon the Corporation by law and those incidental thereto, the Corporation shall possess and may exercise all the powers and privileges that are necessary or convenient to the conduct, promotion or attainment of the business or purposes of the Corporation, including, but not limited to, effecting a merger, capital stock exchange, asset acquisition, stock purchase, reorganization or similar business combination, involving the Corporation and one or more businesses (a "Business Combination").

ARTICLE III REGISTERED AGENT

The address of the Corporation's registered office in the State of Delaware is 8 The Green STE A, City of Dover, County of Kent County, 19901. The name of its registered agent at that address is A Registered Agent, Inc.

Page 1 of 15

ARTICLE IV CAPITALIZATION

Section 4.1 <u>Authorized Capital Stock</u>. The total number of shares of all classes of capital stock, each with a par value of \$0.0001 per share, which the Corporation is authorized to issue is 138,750,000 shares, consisting of (a) 137,500,000 shares of common stock (the "<u>Common Stock</u>"), including (i) 125,000,000 shares of Class A Common Stock (the "<u>Class A Common Stock</u>"), and (ii) 12,500,000 shares of Class B Common Stock (the "<u>Class B Common Stock</u>"), and (b) 1,250,000 shares of preferred stock (the "<u>Preferred Stock</u>").

Section 4.2 <u>Preferred Stock</u>. Subject to Article IX of this Amended and Restated Certificate, the Board of Directors of the Corporation (the "<u>Board</u>") is hereby expressly authorized to provide out of the unissued shares of the Preferred Stock for one or more series of Preferred Stock and to establish from time to time the number of shares to be included in each such series and to fix the voting rights, if any, designations, powers, preferences and relative, participating, optional, special and other rights, if any, of each such series and any qualifications, limitations and restrictions thereof, as shall be stated in the resolution or resolutions adopted by the Board providing for the issuance of such series and included in a certificate of designation (a "<u>Preferred Stock Designation</u>") filed pursuant to the DGCL, and the Board is hereby expressly vested with the authority to the full extent provided by law, now or hereafter, to adopt any such resolution or resolutions.

Section 4.3 Common Stock.

- (a) Voting.
- (i) Except as otherwise required by law or this Amended and Restated Certificate (including any Preferred Stock Designation and Section 9.9 hereof), the holders of the Common Stock shall exclusively possess all voting power with respect to the Corporation.
- (ii) Except as otherwise required by law or this Amended and Restated Certificate (including any Preferred Stock Designation and Section 9.9 hereof), the holders of shares of Common Stock shall be entitled to one vote for each such share on each matter properly submitted to the stockholders on which the holders of the Common Stock are entitled to vote.
- (iii) Except as otherwise required by law or this Amended and Restated Certificate (including any Preferred Stock Designation and Section 9.9 hereof), at any annual or special meeting of the stockholders of the Corporation, holders of the Class A Common Stock and holders of the Class B Common Stock, voting together as a single class, shall have the exclusive right to vote for the election of directors and on all other matters properly submitted to a vote of the stockholders. Notwithstanding the foregoing, except as otherwise required by law or this Amended and Restated Certificate (including any Preferred Stock Designation), holders of shares of any series of Common Stock shall not be entitled to vote on any amendment to this Amended and Restated Certificate (including any amendment to any Preferred Stock Designation) that relates solely to the terms of one or more outstanding series of Preferred Stock or other series of Common Stock if the holders of such affected series of Preferred Stock or Common Stock, as applicable, are entitled exclusively, either separately or together with the holders of one or more other such series, to vote thereon pursuant to this Amended and Restated Certificate (including any Preferred Stock Designation) or the DGCL.
 - (b) Class B Common Stock.
- (i) Shares of Class B Common Stock shall be convertible into shares of Class A Common Stock on a one-for-one basis (the "**Initial Conversion Ratio**") automatically on the closing of the Business Combination.
- (ii) Notwithstanding the Initial Conversion Ratio, in the case that additional shares of Class A Common Stock, or Equity-linked Securities (as defined below), are issued or deemed issued in excess of the amounts sold in the Corporation's initial public offering of securities (the "Offering") and related to the closing of the initial Business Combination, all issued and outstanding shares of Class B Common Stock shall automatically convert into shares of Class A Common Stock at the time of the closing of the initial Business Combination at a ratio for which:

- the numerator shall be equal to the sum of (A) 25% of all shares of Class A Common Stock issued and outstanding or issuable (upon the conversion or exercise of any Equity-linked Securities or otherwise) by the Corporation, related to or in connection with the consummation of the initial Business Combination (excluding any securities issued or issuable to any seller in the initial Business Combination, any private placement warrants (or underlying securities) issued to Aesther Healthcare Sponsor, LLC (the "Sponsor") or its affiliates upon conversion of loans to the Corporation and any securities issued as compensation to the underwriters of the Corporation's initial public offering or their representative(s), or their affiliates) plus (B) the number of shares of Class B Common Stock issued and outstanding prior to the closing of the initial Business Combination; and
- the denominator shall be the number of shares of Class B Common Stock issued and outstanding prior to the closing of the initial Business Combination.

As used herein, the term "<u>Equity-linked Securities</u>" means any debt or equity securities that are convertible, exercisable or exchangeable for shares of Class A common stock issued in a financing transaction in connection with a Business Combination, including but not limited to a private placement of equity or debt.

Notwithstanding anything to the contrary contained herein, (i) the foregoing adjustment to the Initial Conversion Ratio may be waived as to any particular issuance or deemed issuance of additional shares of Class A Common Stock or Equity-linked Securities by the written consent or agreement of holders of a majority of the shares of Class B Common Stock then outstanding consenting or agreeing separately as a single class in the manner provided in Section 4.3(b)(iii), and (ii) in no event shall the Class B Common Stock convert into Class A Common Stock at a ratio that is less than one-for-one.

The foregoing conversion ratio shall also be adjusted to account for any subdivision (by stock split, subdivision, exchange, stock dividend, reclassification, recapitalization or otherwise) or combination (by reverse stock split, exchange, reclassification, recapitalization or otherwise) or similar reclassification or recapitalization of the outstanding shares of Class A Common Stock into a greater or lesser number of shares occurring after the original filing of this Amended and Restated Certificate without a proportionate and corresponding subdivision, combination or similar reclassification or recapitalization of the outstanding shares of Class B Common Stock.

Each share of Class B Common Stock shall convert into its pro rata number of shares of Class A Common Stock pursuant to this Section 4.3(b). The pro rata share for each holder of Class B Common Stock will be determined as follows: Each share of Class B Common Stock shall convert into such number of shares of Class A Common Stock as is equal to the product of one (1) multiplied by a fraction, the numerator of which shall be the total number of shares of Class A Common Stock into which all of the issued and outstanding shares of Class B Common Stock shall be converted pursuant to this Section 4.3(b) and the denominator of which shall be the total number of issued and outstanding shares of Class B Common Stock at the time of conversion.

(iii) *Voting.* Except as otherwise required by law or this Amended and Restated Certificate (including any Preferred Stock Designation), for so long as any shares of Class B Common Stock shall remain outstanding, the Corporation shall not, without the prior vote or written consent of the holders of a majority of the shares of Class B Common Stock then outstanding, voting separately as a single class, amend, alter or repeal any provision of this Amended and Restated Certificate, whether by merger, consolidation or otherwise, if such amendment, alteration or repeal would alter or change the powers, preferences or relative, participating, optional or other or special rights of the Class B Common Stock. Any action required or permitted to be taken at any meeting of the holders of Class B Common Stock may be taken without a meeting, without prior notice and without a vote, if a consent or consents in writing, setting forth the action so taken, shall be signed by the holders of the outstanding Class B Common Stock having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares of Class B Common Stock were present and voted and shall be delivered to the Corporation by delivery to its registered office in the State of Delaware, its principal place of business, or an officer or agent of the Corporation having custody of the book in which minutes of proceedings of stockholders are recorded. Delivery made to the Corporation's registered office shall be by hand, by certified or registered mail or by electronic mail, return receipt requested. Prompt written notice of the taking of corporate action without a meeting by less than unanimous written consent of the holders of Class B Common Stock shall, to the extent required by law, be given to those holders of Class B Common Stock who have not consented in writing and who, if the action had been taken at a meeting, would have been entitled to notice of the meeting if the record date for notice of such

- (c) *Dividends*. Subject to applicable law, the rights, if any, of the holders of any outstanding series of the Preferred Stock and the provisions of Article IX hereof, the holders of shares of Common Stock shall be entitled to receive such dividends and other distributions (payable in cash, property or capital stock of the Corporation) when, as and if declared thereon by the Board from time to time out of any assets or funds of the Corporation legally available therefor and shall share equally on a per share basis in such dividends and distributions.
- (d) *Liquidation, Dissolution or Winding Up of the Corporation*. Subject to applicable law, the rights, if any, of the holders of any outstanding series of the Preferred Stock and the provisions of Article IX hereof, in the event of any voluntary or involuntary liquidation, dissolution or winding up of the Corporation, after payment or provision for payment of the debts and other liabilities of the Corporation, the holders of shares of Common Stock shall be entitled to receive all the remaining assets of the Corporation available for distribution to its stockholders, ratably in proportion to the number of shares of Common Stock held by them.

Section 4.4 <u>Rights and Options</u>. The Corporation has the authority to create and issue rights, warrants and options entitling the holders thereof to acquire from the Corporation any shares of its capital stock of any class or classes, with such rights, warrants and options to be evidenced by or in instrument(s) approved by the Board. The Board is empowered to set the exercise price, duration, times for exercise and other terms and conditions of such rights, warrants or options; provided, however, that the consideration to be received for any shares of capital stock issuable upon exercise thereof may not be less than the par value thereof.

ARTICLE V BOARD OF DIRECTORS

Section 5.1 <u>Board Powers</u>. The business and affairs of the Corporation shall be managed by, or under the direction of, the Board. In addition to the powers and authority expressly conferred upon the Board by statute, this Amended and Restated Certificate or the Bylaws of the Corporation ("<u>Bylaws</u>"), the Board is hereby empowered to exercise all such powers and do all such acts and things as may be exercised or done by the Corporation, subject, nevertheless, to the provisions of the DGCL, this Amended and Restated Certificate and any Bylaws adopted by the stockholders of the Corporation; provided, however, that no Bylaws hereafter adopted by the stockholders of the Corporation shall invalidate any prior act of the Board that would have been valid if such Bylaws had not been adopted.

Section 5.2 Number, Election and Term.

(a) The number of directors of the Corporation, other than those who may be elected by the holders of one or more series of the Preferred Stock voting separately by class or series, shall be fixed from time to time exclusively by the Board pursuant to a resolution adopted by a majority of the Board.

Page 4 of 15

(b) Subject to Section 5.5 hereof, the Board shall be divided into three classes, as nearly equal in number as possible and designated Class I, Class II and Class III. The Board is authorized to assign members of the Board already in office to Class I, Class II or Class III. The authorized number of directors may be changed only by resolution of the board of directors. The term of the initial Class I Directors shall expire at the first annual meeting of the stockholders of the Corporation following the effectiveness of this Amended and Restated Certificate, the term of the initial Class II Directors shall expire at the second annual meeting of the stockholders of the Corporation following the effectiveness of this Amended and Restated Certificate, and the term of the initial Class III Directors shall expire at the third annual meeting of the stockholders of the Corporation following the effectiveness of this Amended and Restated Certificate. At each succeeding annual meeting of the stockholders of the Corporation, beginning with the first annual meeting of the stockholders of the Corporation following the effectiveness of this Amended and Restated Certificate, each of the successors elected to replace the class of directors whose term expires at that annual meeting shall be elected for a three-year term or until the election and qualification of their respective successors in office, subject to their earlier death, resignation or removal. Subject to Section 5.5 hereof, if the number of directors that constitutes the Board is changed, any increase or decrease shall be apportioned by the Board among the classes so as to maintain the number of directors in each class as nearly equal as possible, but in no case shall a decrease in the number of directors constituting the Board shorten the term of any incumbent director. Subject to the rights of the holders of one or more series of Preferred Stock, voting separately by class or series, to elect directors pursuant to the terms of one or more series of Preferred Stock, the election of directors shall be determined by a plurality of the votes cast by the stockholders present in person or represented by proxy at the meeting and entitled to vote thereon. The Board is hereby expressly authorized, by resolution or resolutions thereof, to assign members of the Board already in office to the aforesaid classes at the time this Amended and Restated Certificate (and therefore such classification) becomes effective in accordance with the DGCL.

(c) Subject to Section 5.5 hereof, a director shall hold office until the annual meeting for the year in which his or her term expires and until his or her successor has been elected and qualified, subject, however, to such director's earlier death, resignation, retirement, disqualification or removal.

(d) Unless and except to the extent that the Bylaws shall so require, the election of directors need not be by written ballot. The holders of shares of Common Stock shall not have cumulative voting rights.

Section 5.3 <u>Newly Created Directorships and Vacancies</u>. Subject to Sections 5.5 and 9.9 hereof, newly created directorships resulting from an increase in the number of directors and any vacancies on the Board resulting from death, resignation, retirement, disqualification, removal or other cause may be filled solely and exclusively by a majority vote of the remaining directors then in office, even if less than a quorum, or by a sole remaining director (and not by stockholders), and any director so chosen shall hold office for the remainder of the full term of the class of directors to which the new directorship was added or in which the vacancy occurred and until his or her successor has been elected and qualified, subject, however, to such director's earlier death, resignation, retirement, disqualification or removal.

Section 5.4 <u>Removal</u>. Subject to Sections 5.5 hereof and except as otherwise required by this Amended and Restated Certificate (including Section 9.9 hereof), any or all of the directors may be removed from office at any time, but only for cause and only by the affirmative vote of holders of a majority of the voting power of all then outstanding shares of capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class.

Section 5.5 <u>Preferred Stock - Directors</u>. Notwithstanding any other provision of this Article V, and except as otherwise required by law, whenever the holders of one or more series of the Preferred Stock shall have the right, voting separately by class or series, to elect one or more directors, the term of office, the filling of vacancies, the removal from office and other features of such directorships shall be governed by the terms of such series of the Preferred Stock as set forth in this Amended and Restated Certificate (including any Preferred Stock Designation) and such directors shall not be included in any of the classes created pursuant to this Article V unless expressly provided by such terms.

Section 5.6 Quorum. A quorum for the transaction of business by the directors shall be set forth in the Bylaws.

ARTICLE VI BYLAWS

In furtherance and not in limitation of the powers conferred upon it by law, the Board shall have the power and is expressly authorized to adopt, amend, alter or repeal the Bylaws. The affirmative vote of a majority of the Board shall be required to adopt, amend, alter or repeal the Bylaws. The Bylaws also may be adopted, amended, altered or repealed by the stockholders of the Corporation; provided, however, that in addition to any vote of the holders of any class or series of capital stock of the Corporation required by law or by this Amended and Restated Certificate (including any Preferred Stock Designation), the affirmative vote of the holders of at least a majority of the voting power of all then outstanding shares of capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class, shall be required for the stockholders to adopt, amend, alter or repeal the Bylaws; and provided further, however, that no Bylaws hereafter adopted by the stockholders shall invalidate any prior act of the Board that would have been valid if such Bylaws had not been adopted.

ARTICLE VII MEETINGS OF STOCKHOLDERS; ACTION BY WRITTEN CONSENT

Section 7.1 Meetings. Subject to the rights, if any, of the holders of any outstanding series of the Preferred Stock, and to the requirements of applicable law, special meetings of stockholders of the Corporation may be called only by the Chairman of the Board, Chief Executive Officer of the Corporation, or the Board pursuant to a resolution adopted by a majority of the Board, and the ability of the stockholders of the Corporation to call a special meeting is hereby specifically denied. Except as provided in the foregoing sentence, special meetings of stockholders of the Corporation may not be called by another person or persons.

Section 7.2 <u>Advance Notice</u>. Advance notice of stockholder nominations for the election of directors and of business to be brought by stockholders before any meeting of the stockholders of the Corporation shall be given in the manner provided in the Bylaws.

Section 7.3 <u>Action by Written Consent</u>. Except as may be otherwise provided for or fixed pursuant to this Amended and Restated Certificate (including any Preferred Stock Designation) relating to the rights of the holders of any outstanding series of Preferred Stock, subsequent to the consummation of the Offering, any action required or permitted to be taken by the stockholders of the Corporation must be effected by a duly called annual or special meeting of such stockholders and may not be effected by written consent of the stockholders other than with respect to the Class B Common Stock with respect to which action may be taken by written consent.

ARTICLE VIII LIMITED LIABILITY; INDEMNIFICATION

Section 8.1 <u>Limitation of Director Liability</u>. A director of the Corporation shall not be personally liable to the Corporation or its stockholders for monetary damages for breach of fiduciary duty as a director, except to the extent such exemption from liability or limitation thereof is not permitted under the DGCL as the same exists or may hereafter be amended unless they violated their duty of loyalty to the Corporation or its stockholders, acted in bad faith, knowingly or intentionally violated the law, authorized unlawful payments of dividends, unlawful stock purchases or unlawful redemptions, or derived improper personal benefit from their actions as directors. Any amendment, modification or repeal of the foregoing sentence shall not adversely affect any right or protection of a director of the Corporation hereunder in respect of any act or omission occurring prior to the time of such amendment, modification or repeal.

Section 8.2 Indemnification and Advancement of Expenses.

Page 6 of 15

- (a) To the fullest extent permitted by applicable law, as the same exists or may hereafter be amended, the Corporation shall indemnify and hold harmless each person who is or was made a party or is threatened to be made a party to or is otherwise involved in any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (a "proceeding") by reason of the fact that he or she is or was a director or officer of the Corporation or, while a director or officer of the Corporation, is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust, other enterprise or nonprofit entity, including service with respect to an employee benefit plan (an "indemnitee"), whether the basis of such proceeding is alleged action in an official capacity as a director, officer, employee or agent, or in any other capacity while serving as a director, officer, employee or agent, against all liability and loss suffered and expenses (including, without limitation, attorneys' fees, judgments, fines, ERISA excise taxes and penalties and amounts paid in settlement) reasonably incurred by such indemnitee in connection with such proceeding. The Corporation shall to the fullest extent not prohibited by applicable law pay the expenses (including attorneys' fees) incurred by an indemnitee in defending or otherwise participating in any proceeding in advance of its final disposition; provided, however, that, to the extent required by applicable law, such payment of expenses in advance of the final disposition of the proceeding shall be made only upon receipt of an undertaking, by or on behalf of the indemnitee, to repay all amounts so advanced if it shall ultimately be determined that the indemnitee is not entitled to be indemnified under this Section 8.2 or otherwise. The rights to indemnification and advancement of expenses conferred by this Section 8.2 shall be contract rights and such rights shall continue as to an indemnitee who has ceased to be a director, officer, employee or agent and shall inure to the benefit of his or her heirs, executors and administrators. Notwithstanding the foregoing provisions of this Section 8.2(a), except for proceedings to enforce rights to indemnification and advancement of expenses, the Corporation shall indemnify and advance expenses to an indemnitee in connection with a proceeding (or part thereof) initiated by such indemnitee only if such proceeding (or part thereof) was authorized by the Board.
- (b) The rights to indemnification and advancement of expenses conferred on any indemnitee by this Section 8.2 shall not be exclusive of any other rights that any indemnitee may have or hereafter acquire under law, this Amended and Restated Certificate, the Bylaws, an agreement, vote of stockholders or disinterested directors, or otherwise.
- (c) Any repeal or amendment of this Section 8.2 by the stockholders of the Corporation or by changes in law, or the adoption of any other provision of this Amended and Restated Certificate inconsistent with this Section 8.2, shall, unless otherwise required by law, be prospective only (except to the extent such amendment or change in law permits the Corporation to provide broader indemnification rights on a retroactive basis than permitted prior thereto), and shall not in any way diminish or adversely affect any right or protection existing at the time of such repeal or amendment or adoption of such inconsistent provision in respect of any proceeding (regardless of when such proceeding is first threatened, commenced or completed) arising out of, or related to, any act or omission occurring prior to such repeal or amendment or adoption of such inconsistent provision.
- (d) This Section 8.2 shall not limit the right of the Corporation, to the extent and in the manner authorized or permitted by law, to indemnify and to advance expenses to persons other than indemnitees.

ARTICLE IX BUSINESS COMBINATION REQUIREMENTS; EXISTENCE

Section 9.1 General.

(a) The provisions of this Article IX shall apply during the period commencing upon the effectiveness of this Amended and Restated Certificate and terminating upon the consummation of the Corporation's initial Business Combination and, subject to Section 9.9 hereof, no amendment to this Article IX shall be effective prior to the consummation of the initial Business Combination unless approved by the affirmative vote of the holders of at least sixty-five percent (65%) of all then outstanding shares of the Common Stock.

(b) Immediately after the Offering, a certain amount of the net offering proceeds received by the Corporation in the Offering (including the proceeds of any exercise of the underwriters' over-allotment option, if any) and certain other amounts specified in the Corporation's registration statement on Form S-1, as initially filed with the U.S. Securities and Exchange Commission (the "SEC") on July 19, 2021, as amended (the "Registration Statement"), shall be deposited in a trust account (the "Trust Account"), established for the benefit of the Public Stockholders (as defined below) pursuant to a trust agreement described in the Registration Statement (the "Trust Agreement"). Except for the withdrawal of interest to pay taxes (less up to \$100,000 interest to pay dissolution expenses), none of the funds held in the Trust Account (including the interest earned on the funds held in the Trust Account) will be released from the Trust Account until the earliest to occur of (i) the completion of the initial Business Combination, (ii) the redemption of 100% of the Offering Shares (as defined below) if the Corporation is unable to complete its initial Business Combination by the Termination date (or, if the Office of the Delaware Division of Corporations shall not be open for business (including filing of corporate documents) on such date the next date upon which the Office of the Delaware Division of Corporations shall be open) (the "Deadline Date") and (iii) the redemption of shares in connection with a stockholder vote to amend any provisions of this Amended and Restated Certificate (a) to modify the substance or timing of the Corporation's obligation to provide for the redemption of the Offering Shares in connection with an initial Business Combination or to redeem 100% of such shares if the Corporation has not consummated an initial Business Combination by the Deadline Date or (b) with respect to any other provision relating to stockholders' rights or pre-initial Business Combination activity (as described in Section 9.7). Holders of shares of Common Stock included as part of the units sold in the Offering (the "Offering Shares") (whether such Offering Shares were purchased in the Offering or in the secondary market following the Offering and whether or not such holders are the Sponsor or officers or directors of the Corporation, or affiliates of any of the foregoing) are referred to herein as "Public Stockholders."

Section 9.2 Redemption Rights.

(a) Prior to the consummation of the initial Business Combination, the Corporation shall provide all holders of Offering Shares with the opportunity to have their Offering Shares redeemed upon the consummation of the initial Business Combination pursuant to, and subject to the limitations of, Sections 9.2(b) and 9.2(c) (such rights of such holders to have their Offering Shares redeemed pursuant to such Sections, the "Redemption Rights") hereof for cash equal to the applicable redemption price per share determined in accordance with Section 9.2(b) hereof (the "Redemption Price"); provided, however, that the Corporation will only redeem Offering Shares so long as (after such redemption), the Corporation's net tangible assets (as determined in accordance with Rule 3a51-1(g)(1) of the Securities Exchange Act of 1934, as amended (the "Exchange Act") (or any successor rule)), or of any entity that succeeds the Corporation as a public company, will be at least \$5,000,001 or any greater net tangible asset or cash requirement which may be contained in the agreement relating to the initial Business Combination either immediately prior to or upon consummation of the initial Business Combination and after payment of underwriters' fees and commissions (such limitation hereinafter called the "Redemption Limitation"). Notwithstanding anything to the contrary contained in this Amended and Restated Certificate, there shall be no Redemption Rights or liquidating distributions with respect to any warrant issued pursuant to the Offering. This redemption obligation exists only with respect to the proposed business combination with respect to which it relates.

- (b) If the Corporation offers to redeem the Offering Shares other than in conjunction with a stockholder vote on an initial Business Combination with a proxy solicitation pursuant to Regulation 14A of the Exchange Act (or any successor rules or regulations) and filing proxy materials with the SEC, the Corporation shall offer to redeem the Offering Shares, with payment to be made upon the consummation of the initial Business Combination, subject to lawfully available funds therefor, in accordance with the provisions of Section 9.2(a) hereof pursuant to a tender offer in accordance with Rule 13e-4 and Regulation 14E of the Exchange Act (or any successor rule or regulation) (such rules and regulations hereinafter called the "Tender Offer Rules") which it shall commence prior to the consummation of the initial Business Combination and shall file tender offer documents with the SEC prior to the consummation of the initial Business Combination that contain substantially the same financial and other information about the initial Business Combination and the Redemption Rights as is required under Regulation 14A of the Exchange Act (or any successor rule or regulation) (such rules and regulations hereinafter called the "Proxy Solicitation Rules"), even if such information is not required under the Tender Offer Rules; provided, however, that if a stockholder vote is required by law to approve the proposed initial Business Combination, or the Corporation decides to submit the proposed initial Business Combination to the stockholders for their approval for business or other legal reasons, the Corporation shall offer to redeem the Offering Shares, subject to lawfully available funds therefor, in accordance with the provisions of Section 9.2(a) hereof in conjunction with a proxy solicitation pursuant to the Proxy Solicitation Rules (and not the Tender Offer Rules) at a price per share equal to the Redemption Price calculated in accordance with the following provisions of this Section 9.2(b). In the event that the Corporation offers to redeem the Offering Shares pursuant to a tender offer in accordance with the Tender Offer Rules, the Redemption Price per share of the Common Stock payable to holders of the Offering Shares tendering their Offering Shares pursuant to such tender offer shall be equal to the quotient obtained by dividing: (i) the aggregate amount on deposit in the Trust Account as of two business days prior to the consummation of the initial Business Combination, including interest earned on the funds held in the Trust Account (which interest shall be net of taxes payable), by (ii) the total number of then outstanding Offering Shares. If the Corporation offers to redeem the Offering Shares in conjunction with a stockholder vote on the proposed initial Business Combination pursuant to a proxy solicitation, the Redemption Price per share of the Common Stock payable to holders of the Offering Shares exercising their Redemption Rights (irrespective of whether they voted in favor or against the Business Combination) shall be equal to the quotient obtained by dividing: (x) the aggregate amount on deposit in the Trust Account as of two business days prior to the consummation of the initial Business Combination, including interest earned on the funds held in the Trust Account (which interest shall be net of taxes payable), by (y) the total number of then outstanding Offering Shares.
- (c) If the Corporation offers to redeem the Offering Shares in conjunction with a stockholder vote on an initial Business Combination pursuant to a proxy solicitation, a Public Stockholder, together with any affiliate of such stockholder or any other person with whom such stockholder is acting in concert or as a "**group**" (as defined under Section 13(d)(3) of the Exchange Act), shall be restricted from seeking Redemption Rights with respect to more than an aggregate of 15% of the Offering Shares without the prior consent of the Corporation.
- (d) In the event that the Corporation has not consummated a Business Combination within 12 months from the consummation of the Offering, the Board of Directors may extend the period of time to consummate a Business Combination up to two times (the latest such date being referred to as the "Termination Date"), each by an additional three months, for an aggregate of six additional months, providing that (i) for each such extension the Sponsor (or its designees) must deposit into the Trust Account \$1,000,000 (or up to \$1,150,000 if the underwriters' over-allotment option is exercised in full) per extension in exchange for a non-interest bearing, unsecured promissory note, for maximum aggregate proceeds to the Corporation of \$2,000,000 (or up to \$2,300,000 if the underwriters' over-allotment option is exercised in full) if two extensions occur and (ii) the procedures relating to any such extension, as set forth in the Trust Agreement, shall have been complied with. The gross proceeds from the issuance of such promissory notes will be added to the proceeds from the Offering to be held in the Trust Account and shall be used to fund the redemption of the Offering Shares in accordance with this Article IX. In the event that the Corporation has not consummated an initial Business Combination by the Deadline Date, the Corporation shall (i) cease all operations except for the purpose of winding up, (ii) as promptly as reasonably possible but not more than ten business days thereafter subject to lawfully available funds therefor, redeem 100% of the Offering Shares in consideration of a per-share price, payable in cash, equal to the quotient obtained by dividing (A) the aggregate amount then on deposit in the Trust Account, including interest earned on the funds held in the Trust Account (which interest shall be net of taxes payable and up to \$100,000 of interest to pay dissolution expenses), by (B) the total number of then outstanding Offering Shares, which redemption will completely extinguish rights of the Public Stockholders (including the right to receive further liquidating distributions, if any), subject to applicable law, and (iii) as promptly as reasonably possible following such redemption, subject to the approval of the remaining stockholders and the Board in accordance with applicable law, dissolve and liquidate, subject in each case to the Corporation's obligations under the DGCL to provide for claims of creditors and other requirements of applicable law.
- (e) If the Corporation offers to redeem the Offering Shares in conjunction with a stockholder vote on an initial Business Combination, the Corporation shall consummate the proposed initial Business Combination only if (i) such initial Business Combination is approved by the affirmative vote of the holders of a majority of the shares of the Common Stock that are voted at a stockholder meeting held to consider such initial Business Combination and (ii) the Redemption Limitation is not exceeded.
- (f) If the Corporation conducts a tender offer pursuant to Section 9.2(b), the Corporation shall consummate the proposed initial Business Combination only if the Redemption Limitation is not exceeded.

Section 9.3 Distributions from the Trust Account.

(a) A Public Stockholder shall be entitled to receive funds from the Trust Account only as provided in Sections 9.2(a), 9.2(b), 9.2(d) or 9.7 hereof. In no other circumstances shall a Public Stockholder have any right or interest of any kind in or to distributions from the Trust Account, and no stockholder other than a Public Stockholder shall have any interest in or to the Trust Account.

- (b) Each Public Stockholder that does not exercise its Redemption Rights shall retain its interest in the Corporation and shall be deemed to have given its consent to the release of the remaining funds in the Trust Account to the Corporation, and following payment to any Public Stockholders exercising their Redemption Rights, the remaining funds in the Trust Account shall be released to the Corporation.
- (c) The exercise by a Public Stockholder of the Redemption Rights shall be conditioned on such Public Stockholder following the specific procedures for redemptions set forth by the Corporation in any applicable tender offer or proxy materials sent to the Public Stockholders relating to the proposed initial Business Combination. Payment of the amounts necessary to satisfy the Redemption Rights properly exercised shall be made as promptly as practical after the consummation of the initial Business Combination.
- Section 9.4 <u>Share Issuances</u>. Prior to the consummation of the Corporation's initial Business Combination, the Corporation shall not issue any additional shares of capital stock of the Corporation (other than by way of a stock split or share dividend to the holders of the Class A Common Stock) that would entitle the holders thereof to receive funds from the Trust Account or vote as a class with the Class A Common Stock on any initial Business Combination, on any pre-Business Combination activity or on any amendment to this Article IX.
- Section 9.5 <u>Transactions with Affiliates</u>. In the event the Corporation enters into an initial Business Combination with a target business that is affiliated with the Sponsor, or the directors or officers of the Corporation, the Corporation, or a committee of the independent directors of the Corporation, shall obtain an opinion from an independent investment banking firm or another independent firm that commonly renders valuation opinions that such Business Combination is fair to the Corporation from a financial point of view.
- Section 9.6 No Transactions with Other Blank Check Companies. The Corporation shall not enter into an initial Business Combination with another blank check company or a similar company with nominal operations.
- Section 9.7 <u>Additional Redemption Rights</u>. If, in accordance with Section 9.1(a), any amendment is made to this Amended and Restated Certificate (a) to modify (i) the substance or timing of the Corporation's obligation to allow redemption in connection with the Corporation's initial Business Combination or to redeem 100% of the Offering Shares if the Corporation has not consummated an initial Business Combination by the Deadline Date, or (ii) with respect to any other provision relating to stockholders' rights or pre-initial Business Combination activity, the Public Stockholders shall be provided with the opportunity to redeem their Offering Shares upon the approval of any such amendment, at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the Trust Account, including interest (net of taxes payable), divided by the number of then outstanding Offering Shares; provided, however, that any such amendment will be voided, and this Article IX will remain unchanged, if any stockholders who wish to redeem are unable to redeem due to the Redemption Limitation.

Section 9.8 Minimum Value of Target. So long as the Corporation maintains its listing on Nasdaq, the Corporation's initial Business Combination must occur with one or more target businesses that together have a fair market value of at least 80% of the assets held in the Trust Account (excluding the deferred underwriting discount held in, and taxes payable on the income earned on, the Trust Account) at the time of the agreement to enter into the initial Business Combination in accordance with Nasdaq rules.

Section 9.9 <u>Appointment and Removal of Directors</u>. Notwithstanding any other provision in this Amended and Restated Certificate, prior to the closing of the initial Business Combination, the holders of Class B Common Stock shall have the exclusive right to elect and remove, with or without cause, any director, and the holders of Class A Common Stock shall have no right to vote on the election or removal of any director. This Section 9.9 may be amended only by a resolution passed by holders of at least 90% of the outstanding Common Stock voting thereon.

ARTICLE X BUSINESS COMBINATIONS

Section 10.1 Section 203 of the DGCL. The Corporation will be subject to Section 203 of the DGCL.

Section 10.2 <u>Limitations on Business Combinations</u>. Notwithstanding Section 10.1, the Corporation shall not engage in any business combination (as defined below), at any point in time at which the Corporation's common stock is registered under Section 12(b) or 12(g) of the Exchange Act, with any interested stockholder (as defined below) for a period of three years following the time that such stockholder of the Corporation became an interested stockholder, unless:

- (a) prior to such time, the Board approved either the business combination or the transaction which resulted in the stockholder becoming an interested stockholder, or
- (b) upon consummation of the transaction which resulted in the stockholder becoming an interested stockholder, the interested stockholder owned at least 85% of the voting stock (as defined below) of the Corporation outstanding at the time the transaction commenced, excluding for purposes of determining the voting stock outstanding (but not the outstanding voting stock owned by the interested stockholder) those shares owned by (i) persons who are directors and also officers of the Corporation or (ii) employee stock plans in which employee participants do not have the right to determine confidentially whether shares held subject to the plan will be tendered in a tender or exchange offer, or
- (c) at or subsequent to such time, the business combination is approved by the Board and authorized at an annual or special meeting of stockholders, and not by written consent, by the affirmative vote of at least 66 2/3% of the outstanding voting stock of the Corporation which is not owned by the interested stockholder.

Section 10.3 *Definitions*. For the purposes of this Article X:

- (a) "affiliate" means a person that directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, another person.
- (b) "associate," when used to indicate a relationship with any person, means: (i) any corporation, partnership, unincorporated association or other entity of which such person is a director, officer or partner or is, directly or indirectly, the owner of 20% or more of any class of voting stock; (ii) any trust or other estate in which such person has at least a 20% beneficial interest or as to which such person serves as trustee or in a similar fiduciary capacity; and (iii) any relative or spouse of such person, or any relative of such spouse, who has the same residence as such person.
 - (c) "business combination," when used in reference to the Corporation and any interested stockholder, means:
- (i) any merger or consolidation of the Corporation or any direct or indirect majority-owned subsidiary of the Corporation (A) with the interested stockholder, or (B) with any other corporation, partnership, unincorporated association or other entity if the merger or consolidation is caused by the interested stockholder and as a result of such merger or consolidation Article XI is not applicable to the surviving entity;
- (ii) any sale, lease, exchange, mortgage, pledge, transfer or other disposition (in one transaction or a series of transactions), except proportionately as a stockholder of the Corporation, to or with the interested stockholder, whether as part of a dissolution or otherwise, of assets of the Corporation or of any direct or indirect majority-owned subsidiary of the Corporation which assets have an aggregate market value equal to 10% or more of either the aggregate market value of all the outstanding stock of the Corporation;

Page 11 of 15

(iii) any transaction which results in the issuance or transfer by the Corporation or by any direct or indirect majority-owned subsidiary of the Corporation of any stock of the Corporation or of such subsidiary to the interested stockholder, except: (A) pursuant to the exercise, exchange or conversion of securities exercisable for, exchangeable for or convertible into stock of the Corporation or any such subsidiary which securities were outstanding prior to the time that the interested stockholder became such; (B) pursuant to a merger under Section 251(g) of the DGCL; (C) pursuant to a dividend or distribution paid or made, or the exercise, exchange or conversion of securities exercisable for, exchangeable for or convertible into stock of the Corporation or any such subsidiary which security is distributed, pro rata to all holders of a class or series of stock of the Corporation subsequent to the time the interested stockholder became such; (D) pursuant to an exchange offer by the Corporation to purchase stock made on the same terms to all holders of said stock; or (E) any issuance or transfer of stock by the Corporation; provided, however, that in no case under items (C)-(E) of this subsection (iii) shall there be an increase in the interested stockholder's proportionate share of the stock of any class or series of the Corporation or of the voting stock of the Corporation (except as a result of immaterial changes due to fractional share adjustments);

(iv) any transaction involving the Corporation or any direct or indirect majority-owned subsidiary of the Corporation which has the effect, directly or indirectly, of increasing the proportionate share of the stock of any class or series, or securities convertible into the stock of any class or series, of the Corporation or of any such subsidiary which is owned by the interested stockholder, except as a result of immaterial changes due to fractional share adjustments or as a result of any purchase or redemption of any shares of stock not caused, directly or indirectly, by the interested stockholder; or

(v) any receipt by the interested stockholder of the benefit, directly or indirectly (except proportionately as a stockholder of the Corporation), of any loans, advances, guarantees or pledges (other than those expressly permitted in subsections (i)-(iv) above) provided by or through the Corporation or any direct or indirect majority-owned subsidiary.

(d) "control," including the terms "controlling," "controlled by" and "under common control with," means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a person, whether through the ownership of voting stock, by contract, or otherwise. A person who is the owner of 20% or more of the outstanding voting stock of the Corporation, partnership, unincorporated association or other entity shall be presumed to have control of such entity, in the absence of proof by a preponderance of the evidence to the contrary. Notwithstanding the foregoing, a presumption of control shall not apply where such person holds voting stock, in good faith and not for the purpose of circumventing this Article X, as an agent, bank, broker, nominee, custodian or trustee for one or more owners who do not individually or as a group have control of such entity.

(e) "interested stockholder" means any person (other than the Corporation or any direct or indirect majority-owned subsidiary of the Corporation) that (i) is the owner of 15% or more of the outstanding voting stock of the Corporation, or (ii) is an affiliate or associate of the Corporation and was the owner of 15% or more of the outstanding voting stock of the Corporation at any time within the three year period immediately prior to the date on which it is sought to be determined whether such person is an interested stockholder, and the affiliates and associates of such person; provided, however, that the term "interested stockholder" shall not include (A) the Principal Stockholder or Principal Stockholder Transferees or any "group" (within the meaning of Rule 13d-5 of the Exchange Act) that includes any Principal Stockholder or Principal Stockholder Transferee or (B) any person whose ownership of shares in excess of the 15% limitation set forth herein is the result of any action taken solely by the Corporation; provided that such person specified in this clause (B) shall be an interested stockholder if thereafter such person acquires additional shares of voting stock of the Corporation, except as a result of further corporate action not caused, directly or indirectly, by such person. For the purpose of determining whether a person is an interested stockholder, the voting stock of the Corporation deemed to be outstanding shall include stock deemed to be owned by the person through application of the definition of "owner" below but shall not include any other unissued stock of the Corporation which may be issuable pursuant to any agreement, arrangement or understanding, or upon exercise of conversion rights, warrants or options, or otherwise.

- (f) "owner," including the terms "own" and "owned," when used with respect to any stock, means a person that individually or with or through any of its affiliates or associates:
 - (i) beneficially owns such stock, directly or indirectly; or
- (ii) has (A) the right to acquire such stock (whether such right is exercisable immediately or only after the passage of time) pursuant to any agreement, arrangement or understanding, or upon the exercise of conversion rights, exchange rights, warrants or options, or otherwise; provided, however, that a person shall not be deemed the owner of stock tendered pursuant to a tender or exchange offer made by such person or any of such person's affiliates or associates until such tendered stock is accepted for purchase or exchange; or (B) the right to vote such stock pursuant to any agreement, arrangement or understanding; provided, however, that a person shall not be deemed the owner of any stock because of such person's right to vote such stock if the agreement, arrangement or understanding to vote such stock arises solely from a revocable proxy or consent given in response to a proxy or consent solicitation made to ten or more persons; or
- (iii) has any agreement, arrangement or understanding for the purpose of acquiring, holding, voting (except voting pursuant to a revocable proxy or consent as described in item (B) of subsection (ii) above), or disposing of such stock with any other person that beneficially owns, or whose affiliates or associates beneficially own, directly or indirectly, such stock.
 - (g) "person" means any individual, corporation, partnership, unincorporated association or other entity.
 - (h) "stock" means, with respect to any corporation, capital stock and, with respect to any other entity, any equity interest.
- (i) "Principal Stockholder" means, collectively, (i) the Sponsor and (ii) any affiliate or successor of a person referenced in clauses (i) and (ii) of this definition.
- (j) "Principal Stockholder Transferee" means any Person who acquires voting stock of the Corporation from the Principal Stockholder (other than in connection with a public offering) and who is designated in writing by the Principal Stockholder as a "Principal Stockholder Transferee."
- (k) "voting stock" means stock of any class or series entitled to vote generally in matters submitted for stockholders' approval other than the election of directors.

ARTICLE XI CORPORATE OPPORTUNITY

To the extent allowed by law, the doctrine of corporate opportunity, or any other analogous doctrine, shall not apply with respect to the Corporation or any of its officers or directors, or any of their respective affiliates, in circumstances where the application of any such doctrine would conflict with any fiduciary duties or contractual obligations they may have as of the date of this Amended and Restated Certificate or in the future, and the Corporation renounces any expectancy that any of the directors or officers of the Corporation will offer any such corporate opportunity of which he or she may become aware to the Corporation, except, the doctrine of corporate opportunity shall apply with respect to any of the directors or officers of the Corporation with respect to a corporate opportunity that was offered to such person solely in his or her capacity as a director or officer of the Corporation and (i) such opportunity is one the Corporation is legally and contractually permitted to undertake and would otherwise be reasonable for the Corporation to pursue and (ii) the director or officer is permitted to refer that opportunity to the Corporation without violating any legal obligation.

ARTICLE XII AMENDMENT OF AMENDED AND RESTATED CERTIFICATE OF INCORPORATION

The Corporation reserves the right at any time and from time to time to amend, alter, change or repeal any provision contained in this Amended and Restated Certificate (including any Preferred Stock Designation), and other provisions authorized by the laws of the State of Delaware at the time in force that may be added or inserted, in the manner now or hereafter prescribed by this Amended and Restated Certificate and the DGCL; and, except as set forth in Article VIII, all rights, preferences and privileges of whatever nature herein conferred upon stockholders, directors or any other persons by and pursuant to this Amended and Restated Certificate in its present form or as hereafter amended are granted subject to the right reserved in this Article XI; provided, however, that Article IX of this Amended and Restated Certificate may be amended only as provided therein.

ARTICLE XIII EXCLUSIVE FORUM FOR CERTAIN LAWSUITS; CONSENT TO JURISDICTION

Section 13.1 Forum. Subject to the last sentence in this Section 13.1, and unless the Corporation consents in writing to the selection of an alternative forum, to the fullest extent permitted by the applicable law, the Court of Chancery of the State of Delaware shall be the sole and exclusive forum for any stockholder (including a beneficial owner) to bring (i) any derivative action or proceeding brought on behalf of the Corporation, (ii) any action asserting a claim of breach of a fiduciary duty owed by any director, officer or other employee of the Corporation to the Corporation or the Corporation's stockholders, (iii) any action asserting a claim against the Corporation, its directors, officers or employees arising pursuant to any provision of the DGCL or this Amended and Restated Certificate or the Bylaws, or (iv) any action asserting a claim against the Corporation, its directors, officers or employees governed by the internal affairs doctrine and, if brought outside of Delaware, the stockholder bringing the suit will be deemed to have consented to service of process on such stockholder's counsel except any action (A) as to which the Court of Chancery in the State of Delaware determines that there is an indispensable party not subject to the jurisdiction of the Court of Chancery (and the indispensable party does not consent to the personal jurisdiction of the Court of Chancery within ten days following such determination), (B) which is vested in the exclusive jurisdiction of a court or forum other than the Court of Chancery, or (C) for which the Court of Chancery does not have subject matter jurisdiction. Notwithstanding the foregoing, (i) the provisions of this Section 13.1 will not apply to suits brought to enforce any liability or duty created by the Exchange Act or any other claim for which the federal courts have exclusive jurisdiction and (ii) unless the Corporation consents in writing to the selection of an alternative forum, the federal district courts of the United States of America shall, to the

Section 13.2 <u>Consent to Jurisdiction</u>. If any action the subject matter of which is within the scope of Section 13.1 immediately above is filed in a court other than a court located within the State of Delaware (a "<u>Foreign Action</u>") in the name of any stockholder, such stockholder shall be deemed to have consented to (i) the personal jurisdiction of the state and federal courts located within the State of Delaware in connection with any action brought in any such court to enforce Section 13.1 immediately above (an "<u>FSC Enforcement Action</u>") and (ii) having service of process made upon such stockholder in any such FSC Enforcement Action by service upon such stockholder's counsel in the Foreign Action as agent for such stockholder.

Section 13.3 <u>Severability</u>. If any provision or provisions of this Article XIII shall be held to be invalid, illegal or unenforceable as applied to any person or entity or circumstance for any reason whatsoever, then, to the fullest extent permitted by law, the validity, legality and enforceability of such provisions in any other circumstance and of the remaining provisions of this Article XIII (including, without limitation, each portion of any sentence of this Article XIII containing any such provision held to be invalid, illegal or unenforceable that is not itself held to be invalid, illegal or unenforceable) and the application of such provision to other persons or entities and circumstances shall not in any way be affected or impaired thereby. Any person or entity purchasing or otherwise acquiring any interest in shares of capital stock of the Corporation shall be deemed to have notice of and consented to the provisions of this Article XIII.

IN WITNESS WHEREOF, Aesther Healthcare	Acquisition Co	rp. has ca	used this	Amended and	Restated	Certificate	to be duly	executed and
acknowledged in its name and on its behalf by an authoriz	ed officer as of t	he date fir	st set fort	h above.				

AESTHER HEALTHCARE ACQUISITION CORP.

Ву:	
Name: Suren Ajjarapu	
Title: Chief Executive Officer	

Page 15 of 15

BY LAWS OF AESTHER HEALTHCARE ACQUISITION CORP. (THE "CORPORATION")

ARTICLE I

OFFICES

Section 1.1. Registered Office. The registered office of the Corporation within the State of Delaware shall be located at either (a) the principal place of business of the Corporation in the State of Delaware or (b) the office of the corporation or individual acting as the Corporation's registered agent in Delaware.

Section 1.2. Additional Offices. The Corporation may, in addition to its registered office in the State of Delaware, have such other offices and places of business, both within and outside the State of Delaware, as the Board of Directors of the Corporation (the "<u>Board</u>") may from time to time determine or as the business and affairs of the Corporation may require.

ARTICLE II

STOCKHOLDERS MEETINGS

Section 2.1. Annual Meetings. The annual meeting of stockholders shall be held at such place, either within or without the State of Delaware, and time and on such date as shall be determined by the Board and stated in the notice of the meeting, provided that the Board may in its sole discretion determine that the meeting shall not be held at any place, but may instead be held solely by means of remote communication pursuant to <u>Section 9.5(a)</u>. At each annual meeting, the stockholders entitled to vote on such matters shall elect those directors of the Corporation to fill any term of a directorship that expires on the date of such annual meeting and may transact any other business as may properly be brought before the meeting.

Section 2.2. Special Meetings. Subject to the rights of the holders of any outstanding series of the preferred stock of the Corporation ("<u>Preferred Stock</u>"), and to the requirements of applicable law, special meetings of stockholders, for any purpose or purposes, may be called only by the Chairman of the Board, Chief Executive Officer, or the Board pursuant to a resolution adopted by a majority of the Board, and may not be called by any other person. Special meetings of stockholders shall be held at such place, either within or without the State of Delaware, and at such time and on such date as shall be determined by the Board and stated in the Corporation's notice of the meeting, provided that the Board may in its sole discretion determine that the meeting shall not be held at any place, but may instead be held solely by means of remote communication pursuant to Section 9.5(a).

Section 2.3. Notices. Written notice of each stockholders meeting stating the place, if any, date, and time of the meeting, and the means of remote communication, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such meeting and the record date for determining the stockholders entitled to vote at the meeting, if such date is different from the record date for determining stockholders entitled to notice of the meeting, shall be given in the manner permitted by <u>Section 9.3</u> to each stockholder entitled to vote thereat as of the record date for determining the stockholders entitled to notice of the meeting, by the Corporation not less than 10 nor more than 60 days before the date of the meeting unless otherwise required by the General Corporation Law of the State of Delaware (the "<u>DGCL</u>"). If said notice is for a stockholders meeting other than an annual meeting, it shall in addition state the purpose or purposes for which the meeting is called, and the business transacted at such meeting shall be limited to the matters so stated in the Corporation's notice of meeting (or any supplement thereto). Any meeting of stockholders as to which notice has been given may be postponed, and any meeting of stockholders as to which notice has been given may be cancelled, by the Board upon public announcement (as defined in Section 2.7(c)) given before the date previously scheduled for such meeting.

Section 2.4. Quorum. Except as otherwise provided by applicable law, the Corporation's Certificate of Incorporation, as the same may be amended or restated from time to time (the "Certificate of Incorporation") or these By Laws, the presence, in person or by proxy, at a stockholders meeting of the holders of shares of outstanding capital stock of the Corporation representing a majority of the voting power of all outstanding shares of capital stock of the Corporation entitled to vote at such meeting shall constitute a quorum for the transaction of business at such meeting, except that when specified business is to be voted on by a class or series of stock voting as a class, the holders of shares representing a majority of the voting power of the outstanding shares of such class or series shall constitute a quorum of such class or series for the transaction of such business. If a quorum shall not be present or represented by proxy at any meeting of the stockholders of the Corporation, the chairman of the meeting may adjourn the meeting from time to time in the manner provided in Section 2.6 until a quorum shall attend. The stockholders present at a duly convened meeting may continue to transact business until adjournment, notwithstanding the withdrawal of enough stockholders to leave less than a quorum. Shares of its own stock belonging to the Corporation or to another corporation, if a majority of the voting power of the shares entitled to vote in the election of directors of such other corporation is held, directly or indirectly, by the Corporation, shall neither be entitled to vote nor be counted for quorum purposes; provided, however, that the foregoing shall not limit the right of the Corporation or any such other corporation to vote shares held by it in a fiduciary capacity.

Section 2.5. Voting of Shares.

- (a) <u>Voting Lists</u>. The Secretary of the Corporation (the "<u>Secretary</u>") shall prepare, or shall cause the officer or agent who has charge of the stock ledger of the Corporation to prepare and make, at least 10 days before every meeting of stockholders, a complete list of the stockholders of record entitled to vote at such meeting; provided, however, that if the record date for determining the stockholders entitled to vote is less than 10 days before the meeting date, the list shall reflect the stockholders entitled to vote as of the tenth day before the meeting date, arranged in alphabetical order and showing the address and the number and class of shares registered in the name of each stockholder. Nothing contained in this Section 2.5(a) shall require the Corporation to include electronic mail addresses or other electronic contact information on such list. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting, during ordinary business hours for a period of at least 10 days prior to the meeting: (i) on a reasonably accessible electronic network, provided that the information required to gain access to such list is provided with the notice of the meeting, or (ii) during ordinary business hours, at the principal place of business of the Corporation. In the event that the Corporation determines to make the list available on an electronic network, the Corporation may take reasonable steps to ensure that such information is available only to stockholders of the Corporation. If the meeting is to be held at a place, then the list shall be produced and kept at the time and place of the meeting during the whole time thereof, and may be inspected by any stockholder who is present. If a meeting of stockholders is to be held solely by means of remote communication as permitted by Section 9.5(a), the list shall be open to the examination of any stockholder during the whole time of the meeting on a reasonably accessible electronic network, and the information required to acces
- (b) <u>Manner of Voting</u>. At any stockholders meeting, every stockholder entitled to vote may vote in person or by proxy. If authorized by the Board, the voting by stockholders or proxy holders at any meeting conducted by remote communication may be effected by a ballot submitted by electronic transmission (as defined in <u>Section 9.3</u>), provided that any such electronic transmission must either set forth or be submitted with information from which the Corporation can determine that the electronic transmission was authorized by the stockholder or proxy holder. The Board, in its discretion, or the chairman of the meeting of stockholders, in such person's discretion, may require that any votes cast at such meeting shall be cast by written ballot.
- (c) <u>Proxies</u>. Each stockholder entitled to vote at a meeting of stockholders or to express consent or dissent to corporate action in writing without a meeting may authorize another person or persons to act for such stockholder by proxy, but no such proxy shall be voted or acted upon after three years from its date, unless the proxy provides for a longer period. Proxies need not be filed with the Secretary until the meeting is called to order, but shall be filed with the Secretary before being voted. Without limiting the manner in which a stockholder may authorize another person or persons to act for such stockholder as proxy, either of the following shall constitute a valid means by which a stockholder may grant such authority. No stockholder shall have cumulative voting rights.

- (i) A stockholder may execute a writing authorizing another person or persons to act for such stockholder as proxy. Execution may be accomplished by the stockholder or such stockholder's authorized officer, director, employee or agent signing such writing or causing such person's signature to be affixed to such writing by any reasonable means, including, but not limited to, by facsimile signature.
- (ii) A stockholder may authorize another person or persons to act for such stockholder as proxy by transmitting or authorizing the transmission of an electronic transmission to the person who will be the holder of the proxy or to a proxy solicitation firm, proxy support service organization or like agent duly authorized by the person who will be the holder of the proxy to receive such transmission, provided that any such electronic transmission must either set forth or be submitted with information from which it can be determined that the electronic transmission was authorized by the stockholder. Any copy, facsimile telecommunication or other reliable reproduction of the writing or transmission authorizing another person or persons to act as proxy for a stockholder may be substituted or used in lieu of the original writing or transmission for any and all purposes for which the original writing or transmission could be used; provided that such copy, facsimile telecommunication or other reproduction shall be a complete reproduction of the entire original writing or transmission.
- (d) <u>Required Vote</u>. Subject to the rights of the holders of one or more series of Preferred Stock, voting separately by class or series, to elect directors pursuant to the terms of one or more series of Preferred Stock, at all meetings of stockholders at which a quorum is present, the election of directors shall be determined by a plurality of the votes cast by the stockholders present in person or represented by proxy at the meeting and entitled to vote thereon. All other matters presented to the stockholders at a meeting at which a quorum is present shall be determined by the vote of a majority of the votes cast by the stockholders present in person or represented by proxy at the meeting and entitled to vote thereon, unless the matter is one upon which, by applicable law, the Certificate of Incorporation, these By Laws or applicable stock exchange rules, a different vote is required, in which case such provision shall govern and control the decision of such matter.
- (e) <u>Inspectors of Election</u>. The Board may, and shall if required by law, in advance of any meeting of stockholders, appoint one or more persons as inspectors of election, who may be employees of the Corporation or otherwise serve the Corporation in other capacities, to act at such meeting of stockholders or any adjournment thereof and to make a written report thereof. The Board may appoint one or more persons as alternate inspectors to replace any inspector who fails to act. If no inspectors of election or alternates are appointed by the Board, the chairman of the meeting shall appoint one or more inspectors to act at the meeting. Each inspector, before discharging his or her duties, shall take and sign an oath faithfully to execute the duties of inspector with strict impartiality and according to the best of his or her ability. The inspectors shall ascertain and report the number of outstanding shares and the voting power of each; determine the number of shares present in person or represented by proxy at the meeting and the validity of proxies and ballots; count all votes and ballots and report the results; determine and retain for a reasonable period a record of the disposition of any challenges made to any determination by the inspectors; and certify their determination of the number of shares represented at the meeting and their count of all votes and ballots. No person who is a candidate for an office at an election may serve as an inspector at such election. Each report of an inspector shall be in writing and signed by the inspector or by a majority of them if there is more than one inspector acting at such meeting. If there is more than one inspector, the report of a majority shall be the report of the inspectors.

Section 2.6. Adjournments. Any meeting of stockholders, annual or special, may be adjourned by the chairman of the meeting, from time to time, whether or not there is a quorum, to reconvene at the same or some other place. Notice need not be given of any such adjourned meeting if the date, time, and place, if any, thereof, and the means of remote communication, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such adjourned meeting are announced at the meeting at which the adjournment is taken. At the adjourned meeting the stockholders, or the holders of any class or series of stock entitled to vote separately as a class, as the case may be, may transact any business that might have been transacted at the original meeting. If the adjournment is for more than 30 days, notice of the adjourned meeting shall be given to each stockholder of record entitled to vote at the meeting. If after the adjourned meeting in accordance with Section 9.2, and shall give notice of the adjourned meeting to each stockholder of record entitled to vote at such adjourned meeting as of the record date for notice of such adjourned meeting to each stockholder of record entitled to vote at such adjourned meeting as of the record date for notice of such adjourned meeting.

Section 2.7. Advance Notice for Business.

- (a) <u>Annual Meetings of Stockholders</u>. No business may be transacted at an annual meeting of stockholders, other than business that is either (i) specified in the Corporation's notice of meeting (or any supplement thereto) given by or at the direction of the Board, (ii) otherwise properly brought before the annual meeting by or at the direction of the Board or (iii) otherwise properly brought before the annual meeting by any stockholder of the Corporation (x) who is a stockholder of record entitled to vote at such annual meeting on the date of the giving of the notice provided for in this <u>Section 2.7(a)</u> and on the record date for the determination of stockholders entitled to vote at such annual meeting and (y) who complies with the notice procedures set forth in this <u>Section 2.7(a)</u>. Notwithstanding anything in this <u>Section 2.7(a)</u> to the contrary, only persons nominated for election as a director to fill any term of a directorship that expires on the date of the annual meeting pursuant to <u>Section 3.5</u> will be considered for election at such meeting.
- (i) In addition to any other applicable requirements, for business (other than nominations) to be properly brought before an annual meeting by a stockholder, such stockholder must have given timely notice thereof in proper written form to the Secretary and such business must otherwise be a proper matter for stockholder action. Subject to Section 2.7(a)(iii), a stockholder's notice to the Secretary with respect to such business, to be timely, must be received by the Secretary at the principal executive offices of the Corporation not later than the close of business on the 90th day nor earlier than the opening of business on the 120th day before the anniversary date of the immediately preceding annual meeting of stockholders; provided, however, that in the event that the annual meeting is more than 30 days before or more than 60 days after such anniversary date, notice by the stockholder to be timely must be so delivered not earlier than the close of business on the 120th day before the meeting and not later than the later of (x) the close of business on the 90th day before the meeting or (y) the close of business on the 10th day following the day on which public announcement of the date of the annual meeting is first made by the Corporation. The public announcement of an adjournment or postponement of an annual meeting shall not commence a new time period (or extend any time period) for the giving of a stockholder's notice as described in this Section 2.7(a).
- (ii) To be in proper written form, a stockholder's notice to the Secretary with respect to any business (other than nominations) must set forth as to each such matter such stockholder proposes to bring before the annual meeting (A) a brief description of the business desired to be brought before the annual meeting, the text of the proposal or business (including the text of any resolutions proposed for consideration and in the event such business includes a proposal to amend these By Laws, the language of the proposed amendment) and the reasons for conducting such business at the annual meeting, (B) the name and record address of such stockholder and the name and address of the beneficial owner, if any, on whose behalf the proposal is made, (C) the class or series and number of shares of capital stock of the Corporation that are owned beneficially and of record by such stockholder and by the beneficial owner, if any, on whose behalf the proposal is made, (D) a description of all arrangements or understandings between such stockholder and the beneficial owner, if any, on whose behalf the proposal is made and any other person or persons (including their names) in connection with the proposal of such business by such stockholder, (E) any material interest of such stockholder and the beneficial owner, if any, on whose behalf the proposal is made in such business and (F) a representation that such stockholder (or a qualified representative of such stockholder) intends to appear in person or by proxy at the annual meeting to bring such business before the meeting.

- (iii) The foregoing notice requirements of this <u>Section 2.7(a)</u> shall be deemed satisfied by a stockholder as to any proposal (other than nominations) if the stockholder has notified the Corporation of such stockholder's intention to present such proposal at an annual meeting in compliance with Rule 14a-8 (or any successor thereof) of the Securities Exchange Act of 1934, as amended (the "<u>Exchange Act</u>"), and such stockholder has complied with the requirements of such Rule for inclusion of such proposal in a proxy statement prepared by the Corporation to solicit proxies for such annual meeting. No business shall be conducted at the annual meeting of stockholders except business brought before the annual meeting in accordance with the procedures set forth in this <u>Section 2.7(a)</u>, provided, however, that once business has been properly brought before the annual meeting in accordance with such procedures, nothing in this <u>Section 2.7(a)</u> shall be deemed to preclude discussion by any stockholder of any such business. If the Board or the chairman of the annual meeting determines that any stockholder proposal was not made in accordance with the provisions of this <u>Section 2.7(a)</u> or that the information provided in a stockholder's notice does not satisfy the information requirements of this <u>Section 2.7(a)</u>, such proposal shall not be presented for action at the annual meeting. Notwithstanding the foregoing provisions of this <u>Section 2.7(a)</u>, if the stockholder (or a qualified representative of the stockholder) does not appear at the annual meeting of stockholders of the Corporation to present the proposed business, such proposed business shall not be transacted, notwithstanding that proxies in respect of such matter may have been received by the Corporation.
- (iv) In addition to the provisions of this <u>Section 2.7(a)</u>, a stockholder shall also comply with all applicable requirements of the Exchange Act and the rules and regulations thereunder with respect to the matters set forth herein. Nothing in this <u>Section 2.7(a)</u> shall be deemed to affect any rights of stockholders to request inclusion of proposals in the Corporation's proxy statement pursuant to Rule 14a-8 under the Exchange Act.
- (b) <u>Special Meetings of Stockholders</u>. Only such business shall be conducted at a special meeting of stockholders as shall have been brought before the meeting pursuant to the Corporation's notice of meeting. Nominations of persons for election to the Board may be made at a special meeting of stockholders at which directors are to be elected pursuant to the Corporation's notice of meeting only pursuant to <u>Section 3.5</u>.
- (c) <u>Public Announcement</u>. For purposes of these By Laws, "<u>public announcement</u>" shall mean disclosure in a press release reported by the Dow Jones News Service, Associated Press or comparable national news service or in a document publicly filed by the Corporation with the Securities and Exchange Commission pursuant to Sections 13, 14 or 15(d) of the Exchange Act (or any successor thereto).

Section 2.8. Conduct of Meetings. The chairman of each annual and special meeting of stockholders shall be the Chairman of the Board or, in the absence (or inability or refusal to act) of the Chairman of the Board, any Chief Executive Officer (if he or she shall be a director) or, in the absence (or inability or refusal to act of a Chief Executive Officer or if a Chief Executive Officer is not a director, the President (if he or she shall be a director) or, in the absence (or inability or refusal to act) of the President or if the President is not a director, such other person as shall be appointed by the Board. The date and time of the opening and the closing of the polls for each matter upon which the stockholders will vote at a meeting shall be announced at the meeting by the chairman of the meeting. The Board may adopt such rules and regulations for the conduct of the meeting of stockholders as it shall deem appropriate. Except to the extent inconsistent with these By Laws or such rules and regulations as adopted by the Board, the chairman of any meeting of stockholders shall have the right and authority to convene and to adjourn the meeting, to prescribe such rules, regulations and procedures and to do all such acts as, in the judgment of such chairman, are appropriate for the proper conduct of the meeting. Such rules, regulations or procedures, whether adopted by the Board or prescribed by the chairman of the meeting, may include, without limitation, the following: (a) the establishment of an agenda or order of business for the meeting; (b) rules and procedures for maintaining order at the meeting and the safety of those present; (c) limitations on attendance at or participation in the meeting to stockholders of record of the Corporation, their duly authorized and constituted proxies or such other persons as the chairman of the meeting shall determine; (d) restrictions on entry to the meeting after the time fixed for the commencement thereof; and (e) limitations on the time allotted to questions or comments by participants. Unless and to the extent determined by the Board or the chairman of the meeting, meetings of stockholders shall not be required to be held in accordance with the rules of parliamentary procedure. The secretary of each annual and special meeting of stockholders shall be the Secretary or, in the absence (or inability or refusal to act) of the Secretary, an Assistant Secretary so appointed to act by the chairman of the meeting. In the absence (or inability or refusal to act) of the Secretary and all Assistant Secretaries, the chairman of the meeting may appoint any person to act as secretary of the meeting.

Section 2.9. Consents in Lieu of Meeting. Unless otherwise provided by the Certificate of Incorporation, until the Corporation consummates an initial public offering ("Offering"), any action required to be taken at any annual or special meeting of stockholders, or any action which may be taken at any annual or special meeting of such stockholders, may be taken without a meeting, without prior notice and without a vote, if a consent in writing, setting forth the action so taken, shall be signed by the holders of outstanding stock entitled to vote thereon having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted, and shall be delivered to the Corporation by delivery to its registered office in the State of Delaware, its principal place of business, or an officer or agent of the Corporation having custody of the book in which proceedings of meetings of stockholders are recorded. Delivery made to the Corporation's registered office shall be by hand or by certified or registered mail, return receipt requested.

Every written consent shall bear the date of signature of each stockholder who signs the consent, and no written consent shall be effective to take the corporate action referred to therein unless, within 60 days of the earliest dated consent delivered in the manner required by this section and the DGCL to the Corporation, written consents signed by a sufficient number of holders entitled to vote to take action are delivered to the Corporation by delivery to its registered office in Delaware, its principal place of business or an officer or agent of the Corporation having custody of the book in which proceedings of meetings of stockholders are recorded. Delivery made to the Corporation's registered office shall be by hand or by certified or registered mail, return receipt requested.

ARTICLE III

DIRECTORS

Section 3.1. Powers; Number. The business and affairs of the Corporation shall be managed by or under the direction of the Board, which may exercise all such powers of the Corporation and do all such lawful acts and things as are not by statute or by the Certificate of Incorporation or by these By Laws required to be exercised or done by the stockholders. Directors need not be stockholders or residents of the State of Delaware. Subject to the Certificate of Incorporation, the number of directors shall be fixed exclusively by resolution of the Board. Each director shall hold office until a successor is duly elected and qualified or until the director's earlier death, resignation, disqualification, or removal.

Section 3.2. Newly Created Directorships and Vacancies. Any newly created directorships resulting from an increase in the authorized number of directors and any vacancies occurring in the Board of Directors, may be filled by the affirmative votes of a majority of the remaining members of the Board of Directors, although less than a quorum, or by a sole remaining director. A director so elected shall be elected to hold office until the earlier of the expiration of the term of office of the director whom he or she has replaced, a successor is duly elected and qualified, or the earlier of such director's death, resignation, or removal.

Section 3.3. Resignations. Any director may resign at any time by notice given in writing or by electronic transmission to the Corporation. Such resignation shall take effect at the date of receipt of such notice by the Corporation or at such later effective date or upon the happening of an event or events as is therein specified.

Section 3.4. Removals. Except as prohibited by applicable law or the Certificate of Incorporation, the stockholders holding a majority of the shares then entitled to vote at an election of directors may remove any director from office with or without cause.

Section 3.5. Advance Notice for Nomination of Directors.

(a) Only persons who are nominated in accordance with the following procedures shall be eligible for election as directors of the Corporation, except as may be otherwise provided by the terms of one or more series of Preferred Stock with respect to the rights of holders of one or more series of Preferred Stock to elect directors. Nominations of persons for election to the Board at any annual meeting of stockholders, or at any special meeting of stockholders called for the purpose of electing directors as set forth in the Corporation's notice of such special meeting, may be made (i) by or at the direction of the Board or (ii) by any stockholder of the Corporation (x) who is a stockholder of record entitled to vote in the election of directors on the date of the giving of the notice provided for in this Section 3.5 and on the record date for the determination of stockholders entitled to vote at such meeting and (y) who complies with the notice procedures set forth in this Section 3.5.

- (b) In addition to any other applicable requirements, for a nomination to be made by a stockholder, such stockholder must have given timely notice thereof in proper written form to the Secretary. To be timely, a stockholder's notice to the Secretary must be received by the Secretary at the principal executive offices of the Corporation (i) in the case of an annual meeting, not later than the close of business on the 90th day nor earlier than the opening of business on the 120th day before the anniversary date of the immediately preceding annual meeting of stockholders; provided, however, that in the event that the annual meeting is more than 30 days before or more than 60 days after such anniversary date, notice by the stockholder to be timely must be so received not earlier than the opening of business on the 120th day before the meeting and not later than the later of (x) the close of business on the 90th day before the meeting or (y) the close of business on the 10th day following the day on which public announcement of the date of the annual meeting was first made by the Corporation; and (ii) in the case of a special meeting of stockholders called for the purpose of electing directors, not later than the close of business on the 10th day following the day on which public announcement of the date of the special meeting is first made by the Corporation. In no event shall the public announcement of an adjournment or postponement of an annual meeting or special meeting commence a new time period (or extend any time period) for the giving of a stockholder's notice as described in this Section 3.5.
- (c) Notwithstanding anything in paragraph (b) to the contrary, in the event that the number of directors to be elected to the Board at an annual meeting is greater than the number of directors whose terms expire on the date of the annual meeting and there is no public announcement by the Corporation naming all of the nominees for the additional directors to be elected or specifying the size of the increased Board before the close of business on the 90th day prior to the anniversary date of the immediately preceding annual meeting of stockholders, a stockholder's notice required by this Section 3.5 shall also be considered timely, but only with respect to nominees for the additional directorships created by such increase that are to be filled by election at such annual meeting, if it shall be received by the Secretary at the principal executive offices of the Corporation not later than the close of business on the 10th day following the date on which such public announcement was first made by the Corporation.
- (d) To be in proper written form, a stockholder's notice to the Secretary must set forth (i) as to each person whom the stockholder proposes to nominate for election as a director (A) the name, age, business address and residence address of the person, (B) the principal occupation or employment of the person, (C) the class or series and number of shares of capital stock of the Corporation that are owned beneficially or of record by the person and (D) any other information relating to the person that would be required to be disclosed in a proxy statement or other filings required to be made in connection with solicitations of proxies for election of directors pursuant to Section 14 of the Exchange Act and the rules and regulations promulgated thereunder; and (ii) as to the stockholder giving the notice (A) the name and record address of such stockholder as they appear on the Corporation's books and the name and address of the beneficial owner, if any, on whose behalf the nomination is made, (B) the class or series and number of shares of capital stock of the Corporation that are owned beneficially and of record by such stockholder and the beneficial owner, if any, on whose behalf the nomination is made, (C) a description of all arrangements or understandings relating to the nomination to be made by such stockholder among such stockholder, the beneficial owner, if any, on whose behalf the nomination is made, each proposed nominee and any other person or persons (including their names), (D) a representation that such stockholder (or a qualified representative of such stockholder) intends to appear in person or by proxy at the meeting to nominate the persons named in its notice and (E) any other information relating to such stockholder and the beneficial owner, if any, on whose behalf the nomination is made that would be required to be disclosed in a proxy statement or other filings required to be made in connection with solicitations of proxies for election of directors pursuant to Section 14 of the Exchange Act a

- (e) If the Board or the chairman of the meeting of stockholders determines that any nomination was not made in accordance with the provisions of this <u>Section 3.5</u>, or that the information provided in a stockholder's notice does not satisfy the information requirements of this <u>Section 3.5</u>, then such nomination shall not be considered at the meeting in question. Notwithstanding the foregoing provisions of this <u>Section 3.5</u>, if the stockholder (or a qualified representative of the stockholder) does not appear at the meeting of stockholders of the Corporation to present the nomination shall be disregarded, notwithstanding that proxies in respect of such nomination may have been received by the Corporation.
- (f) In addition to the provisions of this <u>Section 3.5</u>, a stockholder shall also comply with all of the applicable requirements of the Exchange Act and the rules and regulations thereunder with respect to the matters set forth herein. Nothing in this <u>Section 3.5</u> shall be deemed to affect any rights of the holders of Preferred Stock to elect directors pursuant to the Certificate of Incorporation.

Section 3.6. Compensation. Unless otherwise restricted by the Certificate of Incorporation or these By Laws, the Board shall have the authority to fix the compensation of directors, including for service on a committee of the Board, and may be paid either a fixed sum for attendance at each meeting of the Board or other compensation as director. The directors may be reimbursed their expenses, if any, of attendance at each meeting of the Board. No such payment shall preclude any director from serving the Corporation in any other capacity and receiving compensation therefor. Members of committees of the Board may be allowed like compensation and reimbursement of expenses for service on the committee.

ARTICLE IV

BOARD MEETINGS

Section 4.1. Annual Meetings. The Board shall meet as soon as practicable after the adjournment of each annual stockholders meeting at the place of the annual stockholders meeting unless the Board shall fix another time and place and give notice thereof in the manner required herein for special meetings of the Board. No notice to the directors shall be necessary to legally convene this meeting, except as provided in this <u>Section 4.1</u>.

Section 4.2. Regular Meetings. Regularly scheduled, periodic meetings of the Board may be held without notice at such times, dates and places (within or without the State of Delaware) as shall from time to time be determined by the Board.

Section 4.3. Special Meetings. Special meetings of the Board (a) may be called by the Chairman of the Board or President and (b) shall be called by the Chairman of the Board, President or Secretary on the written request of at least a majority of directors then in office, or the sole director, as the case may be, and shall be held at such time, date and place (within or without the State of Delaware) as may be determined by the person calling the meeting or, if called upon the request of directors or the sole director, as specified in such written request. Notice of each special meeting of the Board shall be given, as provided in Section 9.3, to each director (i) at least 24 hours before the meeting if such notice is oral notice given personally or by telephone or written notice given by hand delivery or by means of a form of electronic transmission and delivery; (ii) at least two days before the meeting if such notice is sent by a nationally recognized overnight delivery service; and (iii) at least five days before the meeting if such notice is sent through the United States mail. If the Secretary shall fail or refuse to give such notice, then the notice may be given by the officer who called the meeting or the directors who requested the meeting. Any and all business that may be transacted at a regular meeting of the Board may be transacted at a special meeting. Except as may be otherwise expressly provided by applicable law, the Certificate of Incorporation, or these By Laws, neither the business to be transacted at, nor the purpose of, any special meeting need be specified in the notice or waiver of notice of such meeting. A special meeting may be held at any time without notice if all the directors are present or if those not present waive notice of the meeting in accordance with Section 9.4.

Section 4.4. Quorum; Required Vote. A majority of the Board shall constitute a quorum for the transaction of business at any meeting of the Board, and the act of a majority of the directors present at any meeting at which there is a quorum shall be the act of the Board, except as may be otherwise specifically provided by applicable law, the Certificate of Incorporation or these By Laws. If a quorum shall not be present at any meeting, a majority of the directors present may adjourn the meeting from time to time, without notice other than announcement at the meeting, until a quorum is present.

Section 4.5. Consent In Lieu of Meeting. Unless otherwise restricted by the Certificate of Incorporation or these By Laws, any action required or permitted to be taken at any meeting of the Board or any committee thereof may be taken without a meeting if all members of the Board or committee, as the case may be, consent thereto in writing or by electronic transmission, and the writing or writings or electronic transmission or transmissions (or paper reproductions thereof) are filed with the minutes of proceedings of the Board or committee. Such filing shall be in paper form if the minutes are maintained in paper form and shall be in electronic form if the minutes are maintained in electronic form.

Section 4.6. Organization. The chairman of each meeting of the Board shall be the Chairman of the Board or, in the absence (or inability or refusal to act) of the Chairman of the Board, any Chief Executive Officer (if he or she shall be a director) or, in the absence (or inability or refusal to act) of a Chief Executive Officer is not a director, the President (if he or she shall be a director) or in the absence (or inability or refusal to act) of the President or if the President is not a director, a chairman elected from the directors present. The Secretary shall act as secretary of all meetings of the Board. In the absence (or inability or refusal to act) of the Secretary, an Assistant Secretary shall perform the duties of the Secretary at such meeting. In the absence (or inability or refusal to act) of the Secretary and all Assistant Secretaries, the chairman of the meeting may appoint any person to act as secretary of the meeting.

ARTICLE V

COMMITTEES OF DIRECTORS

Section 5.1. Establishment. The Board may by resolution of the Board designate one or more committees, each committee to consist of one or more of the directors of the Corporation. Each committee shall keep regular minutes of its meetings and report the same to the Board when required by the resolution designating such committee. The Board shall have the power at any time to fill vacancies in, to change the membership of, or to dissolve any such committee.

Section 5.2. Available Powers. Any committee established pursuant to <u>Section 5.1</u> hereof, to the extent permitted by applicable law and by resolution of the Board, shall have and may exercise all of the powers and authority of the Board in the management of the business and affairs of the Corporation, and may authorize the seal of the Corporation to be affixed to all papers that may require it.

Section 5.3. Alternate Members. The Board may designate one or more directors as alternate members of any committee, who may replace any absent or disqualified member at any meeting of such committee. In the absence or disqualification of a member of the committee, the member or members thereof present at any meeting and not disqualified from voting, whether or not he, she or they constitute a quorum, may unanimously appoint another member of the Board to act at the meeting in place of any such absent or disqualified member.

Section 5.4. Procedures. Unless the Board otherwise provides, the time, date, place, if any, and notice of meetings of a committee shall be determined by such committee. At meetings of a committee, a majority of the number of members of the committee (but not including any alternate member, unless such alternate member has replaced any absent or disqualified member at the time of, or in connection with, such meeting) shall constitute a quorum for the transaction of business. The act of a majority of the members present at any meeting at which a quorum is present shall be the act of the committee, except as otherwise specifically provided by applicable law, the Certificate of Incorporation, these By Laws or the Board. If a quorum is not present at a meeting of a committee, the members present may adjourn the meeting from time to time, without notice other than an announcement at the meeting, until a quorum is present. Unless the Board otherwise provides and except as provided in these By Laws, each committee designated by the Board may make, alter, amend and repeal rules for the conduct of its business. In the absence of such rules each committee shall conduct its business in the same manner as the Board is authorized to conduct its business pursuant to <u>Article III</u> and <u>Article IV</u> of these By Laws.

ARTICLE VI

OFFICERS

- **Section 6.1. Officers.** The officers of the Corporation elected by the Board shall be one or more Chief Executive Officers, a Chief Financial Officer, a Secretary and such other officers (including without limitation, a Chairman of the Board, Presidents, Vice Presidents, Assistant Secretaries and a Treasurer) as the Board from time to time may determine. Officers elected by the Board shall each have such powers and duties as generally pertain to their respective offices, subject to the specific provisions of this <u>Article VI</u>. Such officers shall also have such powers and duties as from time to time may be conferred by the Board. Any Chief Executive Officer or President may also appoint such other officers (including without limitation one or more Vice Presidents and Controllers) as may be necessary or desirable for the conduct of the business of the Corporation. Such other officers shall have such powers and duties and shall hold their offices for such terms as may be provided in these By Laws or as may be prescribed by the Board or, if such officer has been appointed by any Chief Executive Officer or President, as may be prescribed by the appointing officer.
- (a) <u>Chairman of the Board</u>. The Chairman of the Board shall preside when present at all meetings of the stockholders and the Board. The Chairman of the Board shall have general supervision and control of the acquisition activities of the Corporation subject to the ultimate authority of the Board, and shall be responsible for the execution of the policies of the Board with respect to such matters. In the absence (or inability or refusal to act) of the Chairman of the Board, any Chief Executive Officer (if he or she shall be a director) shall preside when present at all meetings of the stockholders and the Board. The powers and duties of the Chairman of the Board shall not include supervision or control of the preparation of the financial statements of the Corporation (other than through participation as a member of the Board). The position of Chairman of the Board and Chief Executive Officer may be held by the same person and may be held by more than one person.
- (b) <u>Chief Executive Officer</u>. One or more Chief Executive Officers shall be the chief executive officer(s) of the Corporation, shall have general supervision of the affairs of the Corporation and general control of all of its business subject to the ultimate authority of the Board, and shall be responsible for the execution of the policies of the Board with respect to such matters, except to the extent any such powers and duties have been prescribed to the Chairman of the Board pursuant to <u>Section 6.1(a)</u> above. In the absence (or inability or refusal to act) of the Chairman of the Board, any Chief Executive Officer (if he or she shall be a director) shall preside when present at all meetings of the stockholders and the Board. The position of Chief Executive Officer and President may be held by the same person and may be held by more than one person.
- (c) <u>President</u>. The President shall make recommendations to any Chief Executive Officer on all operational matters that would normally be reserved for the final executive responsibility of any Chief Executive Officer. In the absence (or inability or refusal to act) of the Chairman of the Board and a Chief Executive Officer, the President (if he or she shall be a director) shall preside when present at all meetings of the stockholders and the Board. The President shall also perform such duties and have such powers as shall be designated by the Board. The position of President and Chief Executive Officer may be held by the same person.
- (d) <u>Vice Presidents</u>. In the absence (or inability or refusal to act) of the President, the Vice President (or in the event there be more than one Vice President, the Vice Presidents in the order designated by the Board) shall perform the duties and have the powers of the President. Any one or more of the Vice Presidents may be given an additional designation of rank or function.

(e) Secretary.

- (i) The Secretary shall attend all meetings of the stockholders, the Board and (as required) committees of the Board and shall record the proceedings of such meetings in books to be kept for that purpose. The Secretary shall give, or cause to be given, notice of all meetings of the stockholders and special meetings of the Board and shall perform such other duties as may be prescribed by the Board, the Chairman of the Board, any Chief Executive Officer or President. The Secretary shall have custody of the corporate seal of the Corporation and the Secretary, or any Assistant Secretary, shall have authority to affix the same to any instrument requiring it, and when so affixed, it may be attested by his or her signature or by the signature of such Assistant Secretary. The Board may give general authority to any other officer to affix the seal of the Corporation and to attest the affixing thereof by his or her signature.
- (ii) The Secretary shall keep, or cause to be kept, at the principal executive office of the Corporation or at the office of the Corporation's transfer agent or registrar, if one has been appointed, a stock ledger, or duplicate stock ledger, showing the names of the stockholders and their addresses, the number and classes of shares held by each and, with respect to certificated shares, the number and date of certificates issued for the same and the number and date of certificates cancelled.
- (f) <u>Assistant Secretaries</u>. The Assistant Secretary or, if there be more than one, the Assistant Secretaries in the order determined by the Board shall, in the absence (or inability or refusal to act) of the Secretary, perform the duties and have the powers of the Secretary.
- (g) <u>Chief Financial Officer</u>. The Chief Financial Officer shall perform all duties commonly incident to that office (including, without limitation, the care and custody of the funds and securities of the Corporation, which from time to time may come into the Chief Financial Officer's hands and the deposit of the funds of the Corporation in such banks or trust companies as the Board, any Chief Executive Officer or the President may authorize).
- (h) <u>Treasurer</u>. The Treasurer shall, in the absence (or inability or refusal to act) of the Chief Financial Officer, perform the duties and exercise the powers of the Chief Financial Officer.
- **Section 6.2. Term of Office; Removal; Vacancies**. The elected officers of the Corporation shall be appointed by the Board and shall hold office until their successors are duly elected and qualified by the Board or until their earlier death, resignation, retirement, disqualification, or removal from office. Any officer may be removed, with or without cause, at any time by the Board. Any officer appointed by any Chief Executive Officer or President may also be removed, with or without cause, by any Chief Executive Officer or President, as the case may be, unless the Board otherwise provides. Any vacancy occurring in any elected office of the Corporation may be filled by the Board. Any vacancy occurring in any office appointed by any Chief Executive Officer or President may be filled by any Chief Executive Officer, or President, as the case may be, unless the Board then determines that such office shall thereupon be elected by the Board, in which case the Board shall elect such officer.
- **Section 6.3. Other Officers**. The Board may delegate the power to appoint such other officers and agents, and may also remove such officers and agents or delegate the power to remove same, as it shall from time to time deem necessary or desirable.
- **Section 6.4. Multiple Officeholders; Stockholder and Director Officers.** Any number of offices may be held by the same person unless the Certificate of Incorporation or these By Laws otherwise provide. Officers need not be stockholders or residents of the State of Delaware.

ARTICLE VII

SHARES

Section 7.1. Certificated and Uncertificated Shares. The shares of the Corporation may be certificated or uncertificated, subject to the sole discretion of the Board and the requirements of the DGCL.

Section 7.2. Multiple Classes of Stock. If the Corporation shall be authorized to issue more than one class of stock or more than one series of any class, the Corporation shall (a) cause the powers, designations, preferences and relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights to be set forth in full or summarized on the face or back of any certificate that the Corporation issues to represent shares of such class or series of stock or (b) in the case of uncertificated shares, within a reasonable time after the issuance or transfer of such shares, send to the registered owner thereof a written notice containing the information required to be set forth on certificates as specified in clause (a) above; provided, however, that, except as otherwise provided by applicable law, in lieu of the foregoing requirements, there may be set forth on the face or back of such certificate or, in the case of uncertificated shares, on such written notice a statement that the Corporation will furnish without charge to each stockholder who so requests the powers, designations, preferences and relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences or rights.

Section 7.3. Signatures. Each certificate representing capital stock of the Corporation shall be signed by or in the name of the Corporation by (a) the Chairman of the Board, any Chief Executive Officer, the President or a Vice President and (b) the Treasurer, an Assistant Treasurer, the Secretary or an Assistant Secretary of the Corporation. Any or all the signatures on the certificate may be a facsimile. In case any officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such officer, transfer agent or registrar before such certificate is issued, such certificate may be issued by the Corporation with the same effect as if such person were such officer, transfer agent or registrar on the date of issue.

Section 7.4. Consideration and Payment for Shares.

- (a) Subject to applicable law and the Certificate of Incorporation, shares of stock may be issued for such consideration, having in the case of shares with par value a value not less than the par value thereof, and to such persons, as determined from time to time by the Board. The consideration may consist of any tangible or intangible property or any benefit to the Corporation including cash, promissory notes, services performed, contracts for services to be performed or other securities, or any combination thereof.
- (b) Subject to applicable law and the Certificate of Incorporation, shares may not be issued until the full amount of the consideration has been paid, unless upon the face or back of each certificate issued to represent any partly paid shares of capital stock or upon the books and records of the Corporation in the case of partly paid uncertificated shares, there shall have been set forth the total amount of the consideration to be paid therefor and the amount paid thereon up to and including the time said certificate representing certificated shares or said uncertificated shares are issued.

Section 7.5. Lost, Destroyed or Wrongfully Taken Certificates.

- (a) If an owner of a certificate representing shares claims that such certificate has been lost, destroyed or wrongfully taken, the Corporation shall issue a new certificate representing such shares or such shares in uncertificated form if the owner: (i) requests such a new certificate before the Corporation has notice that the certificate representing such shares has been acquired by a protected purchaser; (ii) if requested by the Corporation, delivers to the Corporation a bond sufficient to indemnify the Corporation against any claim that may be made against the Corporation on account of the alleged loss, wrongful taking or destruction of such certificate or the issuance of such new certificate or uncertificated shares; and (iii) satisfies other reasonable requirements imposed by the Corporation.
- (b) If a certificate representing shares has been lost, apparently destroyed or wrongfully taken, and the owner fails to notify the Corporation of that fact within a reasonable time after the owner has notice of such loss, apparent destruction or wrongful taking and the Corporation registers a transfer of such shares before receiving notification, the owner shall be precluded from asserting against the Corporation any claim for registering such transfer or a claim to a new certificate representing such shares or such shares in uncertificated form.

Section 7.6. Transfer of Stock.

- (a) If a certificate representing shares of the Corporation is presented to the Corporation with an endorsement requesting the registration of transfer of such shares or an instruction is presented to the Corporation requesting the registration of transfer of uncertificated shares, the Corporation shall register the transfer as requested if:
 - (i) in the case of certificated shares, the certificate representing such shares has been surrendered;
- (ii) (A) with respect to certificated shares, the endorsement is made by the person specified by the certificate as entitled to such shares; (B) with respect to uncertificated shares, an instruction is made by the registered owner of such uncertificated shares; or (C) with respect to certificated shares or uncertificated shares, the endorsement or instruction is made by any other appropriate person or by an agent who has actual authority to act on behalf of the appropriate person;
- (iii) the Corporation has received a guarantee of signature of the person signing such endorsement or instruction or such other reasonable assurance that the endorsement or instruction is genuine and authorized as the Corporation may request;
- (iv) the transfer does not violate any restriction on transfer imposed by the Corporation that is enforceable in accordance with $\underline{\text{Section 7.8(a)}}$; and
 - (v) such other conditions for such transfer as shall be provided for under applicable law have been satisfied.
- (b) Whenever any transfer of shares shall be made for collateral security and not absolutely, the Corporation shall so record such fact in the entry of transfer if, when the certificate for such shares is presented to the Corporation for transfer or, if such shares are uncertificated, when the instruction for registration of transfer thereof is presented to the Corporation, both the transferor and transfere request the Corporation to do so.
- **Section 7.7. Registered Stockholders**. Before due presentment for registration of transfer of a certificate representing shares of the Corporation or of an instruction requesting registration of transfer of uncertificated shares, the Corporation may treat the registered owner as the person exclusively entitled to inspect for any proper purpose the stock ledger and the other books and records of the Corporation, vote such shares, receive dividends or notifications with respect to such shares and otherwise exercise all the rights and powers of the owner of such shares, except that a person who is the beneficial owner of such shares (if held in a voting trust or by a nominee on behalf of such person) may, upon providing documentary evidence of beneficial ownership of such shares and satisfying such other conditions as are provided under applicable law, may also so inspect the books and records of the Corporation.

Section 7.8. Effect of the Corporation's Restriction on Transfer.

- (a) A written restriction on the transfer or registration of transfer of shares of the Corporation or on the amount of shares of the Corporation that may be owned by any person or group of persons, if permitted by the DGCL and noted conspicuously on the certificate representing such shares or, in the case of uncertificated shares, contained in a notice, offering circular or prospectus sent by the Corporation to the registered owner of such shares within a reasonable time prior to or after the issuance or transfer of such shares, may be enforced against the holder of such shares or any successor or transferee of the holder including an executor, administrator, trustee, guardian or other fiduciary entrusted with like responsibility for the person or estate of the holder.
- (b) A restriction imposed by the Corporation on the transfer or the registration of shares of the Corporation or on the amount of shares of the Corporation that may be owned by any person or group of persons, even if otherwise lawful, is ineffective against a person without actual knowledge of such restriction unless: (i) the shares are certificated and such restriction is noted conspicuously on the certificate; or (ii) the shares are uncertificated and such restriction was contained in a notice, offering circular or prospectus sent by the Corporation to the registered owner of such shares within a reasonable time prior to or after the issuance or transfer of such shares.
- **Section 7.9. Regulations**. The Board shall have power and authority to make such additional rules and regulations, subject to any applicable requirement of law, as the Board may deem necessary and appropriate with respect to the issue, transfer or registration of transfer of shares of stock or certificates representing shares. The Board may appoint one or more transfer agents or registrars and may require for the validity thereof that certificates representing shares bear the signature of any transfer agent or registrar so appointed.

ARTICLE VIII

INDEMNIFICATION

Section 8.1. Right to Indemnification. To the fullest extent permitted by applicable law, as the same exists or may hereafter be amended, the Corporation shall indemnify and hold harmless each person who was or is made a party or is threatened to be made a party to or is otherwise involved in any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (hereinafter a "**proceeding**"), by reason of the fact that he or she is or was a director or officer of the Corporation or, while a director or officer of the Corporation, is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust, other enterprise or nonprofit entity, including service with respect to an employee benefit plan (hereinafter an "**Indemnitee**"), whether the basis of such proceeding is alleged action in an official capacity as a director, officer, employee or agent, or in any other capacity while serving as a director, officer, employee or agent, against all liability and loss suffered and expenses (including, without limitation, attorneys' fees, judgments, fines, ERISA excise taxes and penalties and amounts paid in settlement) reasonably incurred by such Indemnitee in connection with such proceeding; provided, however, that, except as provided in Section 8.3 with respect to proceedings to enforce rights to indemnification, the Corporation shall indemnify an Indemnitee in connection with a proceeding (or part thereof) initiated by such Indemnitee only if such proceeding (or part thereof) was authorized by the Board.

Section 8.2. Right to Advancement of Expenses. In addition to the right to indemnification conferred in Section 8.1, an Indemnitee shall also have the right to be paid by the Corporation to the fullest extent not prohibited by applicable law the expenses (including, without limitation, attorneys' fees) incurred in defending or otherwise participating in any such proceeding in advance of its final disposition (hereinafter an "advancement of expenses"); provided, however, that, if the DGCL requires, an advancement of expenses incurred by an Indemnitee in his or her capacity as a director or officer of the Corporation (and not in any other capacity in which service was or is rendered by such Indemnitee, including, without limitation, service to an employee benefit plan) shall be made only upon the Corporation's receipt of an undertaking (hereinafter an "undertaking"), by or on behalf of such Indemnitee, to repay all amounts so advanced if it shall ultimately be determined that such Indemnitee is not entitled to be indemnified under this Article VIII or otherwise.

Section 8.3. Right of Indemnitee to Bring Suit. If a claim under Section 8.1 or Section 8.2 is not paid in full by the Corporation within 60 days after a written claim therefor has been received by the Corporation, except in the case of a claim for an advancement of expenses, in which case the applicable period shall be 20 days, the Indemnitee may at any time thereafter bring suit against the Corporation to recover the unpaid amount of the claim. If successful in whole or in part in any such suit, or in a suit brought by the Corporation to recover an advancement of expenses pursuant to the terms of an undertaking, the Indemnitee shall also be entitled to be paid the expense of prosecuting or defending such suit. In (a) any suit brought by the Indemnitee to enforce a right to indemnification hereunder (but not in a suit brought by an Indemnitee to enforce a right to an advancement of expenses) it shall be a defense that, and (b) in any suit brought by the Corporation to recover an advancement of expenses pursuant to the terms of an undertaking, the Corporation shall be entitled to recover such expenses upon a final judicial decision from which there is no further right to appeal (hereinafter a "final adjudication") that, the Indemnitee has not met any applicable standard for indemnification set forth in the DGCL. Neither the failure of the Corporation (including its directors who are not parties to such action, a committee of such directors, independent legal counsel, or its stockholders) to have made a determination prior to the commencement of such suit that indemnification of the Indemnitee is proper in the circumstances because the Indemnitee has met the applicable standard of conduct set forth in the DGCL, nor an actual determination by the Corporation (including a determination by its directors who are not parties to such action, a committee of such directors, independent legal counsel, or its stockholders) that the Indemnitee has not met such applicable standard of conduct, shall create a presumption that the Indemnitee has not met the applicable standard of conduct or, in the case of such a suit brought by the Indemnitee, shall be a defense to such suit. In any suit brought by the Indemnitee to enforce a right to indemnification or to an advancement of expenses hereunder, or by the Corporation to recover an advancement of expenses pursuant to the terms of an undertaking, the burden of proving that the Indemnitee is not entitled to be indemnified, or to such advancement of expenses, under this Article VIII or otherwise shall be on the Corporation.

Section 8.4. Non-Exclusivity of Rights. The rights provided to any Indemnitee pursuant to this <u>Article VIII</u> shall not be exclusive of any other right, which such Indemnitee may have or hereafter acquire under applicable law, the Certificate of Incorporation, these By Laws, an agreement, a vote of stockholders or disinterested directors, or otherwise.

Section 8.5. Insurance. The Corporation may maintain insurance, at its expense, to protect itself and/or any director, officer, employee or agent of the Corporation or another corporation, partnership, joint venture, trust or other enterprise against any expense, liability or loss, whether or not the Corporation would have the power to indemnify such person against such expense, liability or loss under the DGCL.

Section 8.6. Indemnification of Other Persons. This <u>Article VIII</u> shall not limit the right of the Corporation to the extent and in the manner authorized or permitted by law to indemnify and to advance expenses to persons other than Indemnitees. Without limiting the foregoing, the Corporation may, to the extent authorized from time to time by the Board, grant rights to indemnification and to the advancement of expenses to any employee or agent of the Corporation and to any other person who is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust or other enterprise, including service with respect to an employee benefit plan, to the fullest extent of the provisions of this <u>Article VIII</u> with respect to the indemnification and advancement of expenses of Indemnitees under this <u>Article VIII</u>.

Section 8.7. Amendments. Any repeal or amendment of this <u>Article VIII</u> by the Board or the stockholders of the Corporation or by changes in applicable law, or the adoption of any other provision of these By Laws inconsistent with this <u>Article VIII</u>, will, to the extent permitted by applicable law, be prospective only (except to the extent such amendment or change in applicable law permits the Corporation to provide broader indemnification rights to Indemnitees on a retroactive basis than permitted prior thereto), and will not in any way diminish or adversely affect any right or protection existing hereunder in respect of any act or omission occurring prior to such repeal or amendment or adoption of such inconsistent provision; provided however, that amendments or repeals of this <u>Article VIII</u> shall require the affirmative vote of the stockholders holding at least 66.7% of the voting power of all outstanding shares of capital stock of the Corporation.

Section 8.8. Certain Definitions. For purposes of this Article VIII, (a) references to "other enterprise" shall include any employee benefit plan; (b) references to "fines" shall include any excise taxes assessed on a person with respect to an employee benefit plan; (c) references to "serving at the request of the Corporation" shall include any service that imposes duties on, or involves services by, a person with respect to any employee benefit plan, its participants, or beneficiaries; and (d) a person who acted in good faith and in a manner such person reasonably believed to be in the interest of the participants and beneficiaries of an employee benefit plan shall be deemed to have acted in a manner "not opposed to the best interest of the Corporation" for purposes of Section 145 of the DGCL.

Section 8.9. Contract Rights. The rights provided to Indemnitees pursuant to this <u>Article VIII</u> shall be contract rights and such rights shall continue as to an Indemnitee who has ceased to be a director, officer, agent or employee and shall inure to the benefit of the Indemnitee's heirs, executors and administrators.

Section 8.10. Severability. If any provision or provisions of this <u>Article VIII</u> shall be held to be invalid, illegal or unenforceable for any reason whatsoever: (a) the validity, legality and enforceability of the remaining provisions of this <u>Article VIII</u> shall not in any way be affected or impaired thereby; and (b) to the fullest extent possible, the provisions of this <u>Article VIII</u> (including, without limitation, each such portion of this <u>Article VIII</u> containing any such provision held to be invalid, illegal or unenforceable) shall be construed so as to give effect to the intent manifested by the provision held invalid, illegal or unenforceable.

ARTICLE IX

MISCELLANEOUS

Section 9.1. Place of Meetings. If the place of any meeting of stockholders, the Board or committee of the Board for which notice is required under these By Laws is not designated in the notice of such meeting, such meeting shall be held at the principal business office of the Corporation; provided, however, if the Board has, in its sole discretion, determined that a meeting shall not be held at any place, but instead shall be held by means of remote communication pursuant to <u>Section 9.5</u> hereof, then such meeting shall not be held at any place.

Section 9.2. Fixing Record Dates.

(a) In order that the Corporation may determine the stockholders entitled to notice of any meeting of stockholders or any adjournment thereof, the Board may fix a record date, which shall not precede the date upon which the resolution fixing the record date is adopted by the Board, and which record date shall not be more than 60 nor less than 10 days before the date of such meeting. If the Board so fixes a date, such date shall also be the record date for determining the stockholders entitled to vote at such meeting unless the Board determines, at the time it fixes such record date, that a later date on or before the date of the meeting shall be the date for making such determination. If no record date is fixed by the Board, the record date for determining stockholders entitled to notice of and to vote at a meeting of stockholders shall be at the close of business on the business day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the business day next preceding the day on which the meeting is held. A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board may fix a new record date for the adjourned meeting, and in such case shall also fix as the record date for stockholders entitled to notice of such adjourned meeting the same or an earlier date as that fixed for determination of stockholders entitled to vote in accordance with the foregoing provisions of this Section 9.2(a) at the adjourned meeting.

(b) In order that the Corporation may determine the stockholders entitled to receive payment of any dividend or other distribution or allotment of any rights or the stockholders entitled to exercise any rights in respect of any change, conversion or exchange of stock, or for the purpose of any other lawful action, the Board may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted, and which record date shall be not more than 60 days prior to such action. If no record date is fixed, the record date for determining stockholders for any such purpose shall be at the close of business on the day on which the Board adopts the resolution relating thereto.

Section 9.3. Means of Giving Notice.

(a) <u>Notice to Directors</u>. Whenever under applicable law, the Certificate of Incorporation or these By Laws notice is required to be given to any director, such notice shall be given either (i) in writing and sent by mail, or by a nationally recognized delivery service, (ii) by means of facsimile telecommunication or other form of electronic transmission, or (iii) by oral notice given personally or by telephone. A notice to a director will be deemed given as follows: (i) if given by hand delivery, orally, or by telephone, when actually received by the director, (ii) if sent through the United States mail, when deposited in the United States mail, with postage and fees thereon prepaid, addressed to the director at the director's address appearing on the records of the Corporation, (iii) if sent for next day delivery by a nationally recognized overnight delivery service, when deposited with such service, with fees thereon prepaid, addressed to the director at the director's address appearing on the records of the Corporation, (iv) if sent by facsimile telecommunication, when sent to the facsimile transmission number for such director appearing on the records of the Corporation, (v) if sent by electronic mail, when sent to the electronic mail address for such director appearing on the records of the Corporation.

- (b) Notice to Stockholders. Whenever under applicable law, the Certificate of Incorporation or these By Laws notice is required to be given to any stockholder, such notice may be given (i) in writing and sent either by hand delivery, through the United States mail, or by a nationally recognized overnight delivery service for next day delivery, or (ii) by means of a form of electronic transmission consented to by the stockholder, to the extent permitted by, and subject to the conditions set forth in Section 232 of the DGCL. A notice to a stockholder shall be deemed given as follows: (i) if given by hand delivery, when actually received by the stockholder, (ii) if sent through the United States mail, when deposited in the United States mail, with postage and fees thereon prepaid, addressed to the stockholder at the stockholder's address appearing on the stock ledger of the Corporation, (iii) if sent for next day delivery by a nationally recognized overnight delivery service, when deposited with such service, with fees thereon prepaid, addressed to the stockholder at the stockholder's address appearing on the stock ledger of the Corporation, and (iv) if given by a form of electronic transmission consented to by the stockholder to whom the notice is given and otherwise meeting the requirements set forth above, (A) if by facsimile transmission, when directed to a number at which the stockholder has consented to receive notice, (B) if by electronic mail, when directed to an electronic mail address at which the stockholder has consented to receive notice, (C) if by a posting on an electronic network together with separate notice to the stockholder of such specified posting, upon the later of (1) such posting and (2) the giving of such separate notice, and (D) if by any other form of electronic transmission, when directed to the stockholder. A stockholder may revoke such stockholder's consent to receiving notice by means of electronic communication by giving written notice of such revocation to the Corporation. Any such consent shall be deemed revoked if (1) the Corporation is unable to deliver by electronic transmission two consecutive notices given by the Corporation in accordance with such consent and (2) such inability becomes known to the Secretary or an Assistant Secretary or to the Corporation's transfer agent, or other person responsible for the giving of notice; provided, however, the inadvertent failure to treat such inability as a revocation shall not invalidate any meeting or other action.
- (c) <u>Electronic Transmission</u>. "<u>Electronic transmission</u>" means any form of communication, not directly involving the physical transmission of paper, that creates a record that may be retained, retrieved and reviewed by a recipient thereof, and that may be directly reproduced in paper form by such a recipient through an automated process, including but not limited to transmission by telex, facsimile telecommunication, electronic mail, telegram and cablegram.
- (d) <u>Notice to Stockholders Sharing Same Address</u>. Without limiting the manner by which notice otherwise may be given effectively by the Corporation to stockholders, any notice to stockholders given by the Corporation under any provision of the DGCL, the Certificate of Incorporation or these By Laws shall be effective if given by a single written notice to stockholders who share an address if consented to by the stockholders at that address to whom such notice is given. A stockholder may revoke such stockholder's consent by delivering written notice of such revocation to the Corporation. Any stockholder who fails to object in writing to the Corporation within 60 days of having been given written notice by the Corporation of its intention to send such a single written notice shall be deemed to have consented to receiving such single written notice.
- (e) Exceptions to Notice Requirements. Whenever notice is required to be given, under the DGCL, the Certificate of Incorporation or these By Laws, to any person with whom communication is unlawful, the giving of such notice to such person shall not be required and there shall be no duty to apply to any governmental authority or agency for a license or permit to give such notice to such person. Any action or meeting that shall be taken or held without notice to any such person with whom communication is unlawful shall have the same force and effect as if such notice had been duly given. In the event that the action taken by the Corporation is such as to require the filing of a certificate with the Secretary of State of Delaware, the certificate shall state, if such is the fact and if notice is required, that notice was given to all persons entitled to receive notice except such persons with whom communication is unlawful. Whenever notice is required to be given by the Corporation, under any provision of the DGCL, the Certificate of Incorporation or these By Laws, to any stockholder to whom (1) notice of two consecutive annual meetings of stockholders and all notices of stockholder meetings or of the taking of action by written consent of stockholders without a meeting to such stockholder during the period between such two consecutive annual meetings, or (2) all, and at least two payments (if sent by first-class mail) of dividends or interest on securities during a 12-month period, have been mailed addressed to such stockholder at such stockholder's address as shown on the records of the Corporation and have been returned undeliverable, the giving of such notice to such stockholder shall not be required. Any action or meeting that shall be taken or held without notice to such stockholder shall have the same force and effect as if such notice had been duly given. If any such stockholder shall deliver to the Corporation a written notice setting forth such stockholder's then current address, the requirement that notice be given to such stockholder shall be reinstated. In the event that the action taken by the Corporation is such as to require the filing of a certificate with the Secretary of State of Delaware, the certificate need not state that notice was not given to persons to whom notice was not required to be given pursuant to Section 230(b) of the DGCL. The exception in subsection (1) of the first sentence of this paragraph to the requirement that notice be given shall not be applicable to any notice returned as undeliverable if the notice was given by electronic transmission.

Section 9.4. Waiver of Notice. Whenever any notice is required to be given under applicable law, the Certificate of Incorporation, or these By Laws, a written waiver of such notice, signed by the person or persons entitled to said notice, or a waiver by electronic transmission by the person entitled to said notice, whether before or after the time stated therein, shall be deemed equivalent to such required notice. All such waivers shall be kept with the books of the Corporation. Attendance at a meeting shall constitute a waiver of notice of such meeting, except where a person attends for the express purpose of objecting to the transaction of any business on the ground that the meeting was not lawfully called or convened.

$Section\ 9.5.\ Meeting\ Attendance\ via\ Remote\ Communication\ Equipment.$

- (a) <u>Stockholder Meetings</u>. If authorized by the Board in its sole discretion, and subject to such guidelines and procedures as the Board may adopt, stockholders entitled to vote at such meeting and proxy holders not physically present at a meeting of stockholders may, by means of remote communication:
 - (i) participate in a meeting of stockholders; and
- (ii) be deemed present in person and vote at a meeting of stockholders, whether such meeting is to be held at a designated place or solely by means of remote communication, provided that (A) the Corporation shall implement reasonable measures to verify that each person deemed present and permitted to vote at the meeting by means of remote communication is a stockholder or proxy holder, (B) the Corporation shall implement reasonable measures to provide such stockholders and proxy holders a reasonable opportunity to participate in the meeting and, if entitled to vote, to vote on matters submitted to the applicable stockholders, including an opportunity to read or hear the proceedings of the meeting substantially concurrently with such proceedings, and (C) if any stockholder or proxy holder votes or takes other action at the meeting by means of remote communication, a record of such votes or other action shall be maintained by the Corporation.
- (b) <u>Board Meetings</u>. Unless otherwise restricted by applicable law, the Certificate of Incorporation or these By Laws, members of the Board or any committee thereof may participate in a meeting of the Board or any committee thereof by means of conference telephone or other communications equipment by means of which all persons participating in the meeting can hear each other. Such participation in a meeting shall constitute presence in person at the meeting, except where a person participates in the meeting for the express purpose of objecting to the transaction of any business on the ground that the meeting was not lawfully called or convened.

Section 9.6. Dividends. The Board may from time to time declare, and the Corporation may pay, dividends (payable in cash, property or shares of the Corporation's capital stock) on the Corporation's outstanding shares of capital stock, subject to applicable law and the Certificate of Incorporation.

Section 9.7. Reserves. The Board may set apart out of the funds of the Corporation available for dividends a reserve or reserves for any proper purpose and may abolish any such reserve.

Section 9.8. Contracts and Negotiable Instruments. Except as otherwise provided by applicable law, the Certificate of Incorporation or these By Laws, any contract, bond, deed, lease, mortgage or other instrument may be executed and delivered in the name and on behalf of the Corporation by such officer or officers or other employee or employees of the Corporation as the Board may from time to time authorize. Such authority may be general or confined to specific instances as the Board may determine. The Chairman of the Board, any Chief Executive Officer, the President, the Chief Financial Officer, the Treasurer or any Vice President may execute and deliver any contract, bond, deed, lease, mortgage or other instrument in the name and on behalf of the Corporation. Subject to any restrictions imposed by the Board, the Chairman of the Board, any Chief Executive Officer, President, the Chief Financial Officer, the Treasurer or any Vice President may delegate powers to execute and deliver any contract, bond, deed, lease, mortgage or other instrument in the name and on behalf of the Corporation to other officers or employees of the Corporation under such person's supervision and authority, it being understood, however, that any such delegation of power shall not relieve such officer of responsibility with respect to the exercise of such delegated power.

Section 9.9. Fiscal Year. The fiscal year of the Corporation shall be fixed by the Board.

Section 9.10. Seal. The Board may adopt a corporate seal, which shall be in such form as the Board determines. The seal may be used by causing it or a facsimile thereof to be impressed, affixed or otherwise reproduced.

Section 9.11. Books and Records. The books and records of the Corporation may be kept within or outside the State of Delaware at such place or places as may from time to time be designated by the Board.

Section 9.12. Resignation. Any director, committee member or officer may resign by giving notice thereof in writing or by electronic transmission to the Chairman of the Board, any Chief Executive Officer, the President or the Secretary. The resignation shall take effect at the time it is delivered unless the resignation specifies a later effective date or an effective date determined upon the happening of an event or events. Unless otherwise specified therein, the acceptance of such resignation shall not be necessary to make it effective.

Section 9.13. Surety Bonds. Such officers, employees and agents of the Corporation (if any) as the Chairman of the Board, any Chief Executive Officer, President or the Board may direct, from time to time, shall be bonded for the faithful performance of their duties and for the restoration to the Corporation, in case of their death, resignation, retirement, disqualification or removal from office, of all books, papers, vouchers, money and other property of whatever kind in their possession or under their control belonging to the Corporation, in such amounts and by such surety companies as the Chairman of the Board, any Chief Executive Officer, President or the Board may determine. The premiums on such bonds shall be paid by the Corporation and the bonds so furnished shall be in the custody of the Secretary.

Section 9.14. Securities of Other Corporations. Powers of attorney, proxies, waivers of notice of meeting, consents in writing and other instruments relating to securities owned by the Corporation may be executed in the name of and on behalf of the Corporation by the Chairman of the Board, any Chief Executive Officer, President, any Vice President or any officers authorized by the Board. Any such officer, may, in the name of and on behalf of the Corporation, take all such action as any such officer may deem advisable to vote in person or by proxy at any meeting of security holders of any corporation in which the Corporation may own securities, or to consent in writing, in the name of the Corporation as such holder, to any action by such corporation, and at any such meeting or with respect to any such consent shall possess and may exercise any and all rights and power incident to the ownership of such securities and which, as the owner thereof, the Corporation might have exercised and possessed. The Board may from time to time confer like powers upon any other person or persons.

Section 9.15. Amendments. The Board shall have the power to adopt, amend, alter or repeal the By Laws. The affirmative vote of a majority of the Board shall be required to adopt, amend, alter or repeal the By Laws. The By Laws also may be adopted, amended, altered or repealed by the stockholders; provided, however, that in addition to any vote of the holders of any class or series of capital stock of the Corporation required by applicable law or the Certificate of Incorporation, the affirmative vote of the holders of at least a majority of the voting power (except as otherwise provided in Section 8.7) of all outstanding shares of capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class, shall be required for the stockholders to adopt, amend, alter or repeal the By Laws.

NUMBER	UNITS
AEHAU	

SEE REVERSE FOR CERTAIN DEFINITIONS

AESTHER HEALTHCARE ACQUISITION CORP.

CUSIP 00809L205

UNITS CONSISTING OF ONE SHARE OF CLASS A COMMON STOCK AND ONE-HALF OF ONE WARRANT	UNITS CONSISTING OF	ONE SHARE OF CLASS	A COMMON STOCK AND	ONE-HALF OF ONE WARRANT
---	---------------------	--------------------	--------------------	-------------------------

THIS CERTIFIES THAT	
is the owner ofUnits.	
Each Unit ("Unit") consists of one share of Class A common stock, with a par value \$0.0001 per share ("Class A Common Stock"), of Aesther Healthd Acquisition Corp., a Delaware corporation (the "Company"), and one-half of one redeemable warrant ("Warrant"). Only whole Warrants are exercisal Each redeemable Warrant entitles the holder thereof to purchase one share of Class A Common Stock at a price of \$11.50 per full share (subject adjustment), upon the later to occur of (i) 30 days after the Company's completion of a merger, share exchange, asset acquisition, share purch recapitalization, reorganization or other similar business combination with one or more businesses or entities (a "Business Combination") or (ii) one of the closing of the Company's initial public offering. The shares of Class A Common Stock and Warrants comprising the Units represented by certificate are not transferable separately prior to the fifty-second (52nd) day after the date of the prospectus relating to the Company's initial public offering, unless EF Hutton, division of Benchmark Investments, LLC ("EF") determines that an earlier date is acceptable, but in no event will the share Class A Common Stock and Warrants be traded separately until the Company files with the Securities and Exchange Commission (the "SEC") a curreport on Form 8-K which includes an audited balance sheet reflecting the receipt by the Company of the gross proceeds from its initial public offer including the proceeds received by the Company from the exercise of the over-allotment option thereto, if the over-allotment option is exercised. If over-allotment option is exercised after the date of the prospectus, we will also include in the Form 8-K or a new Form 8-K to provide update financial information to reflect the exercise of the over-allotment option. We will also include in the Form 8-K or a new Form 8-K to provide update financial information to reflect the exercise of the over-allotment option is calculated by the Company for the balance of the prospectus.	able. It to lase, year this blic es of rent ring the ated uent
The terms of the Warrants are governed by a Warrant Agreement, dated as of , 2021, between the Company and Continental Stock Transfer Trust Company, as Warrant Agent, and is subject to the terms and provisions contained therein, all of which terms and provisions the holder of certificate consents to by acceptance hereof. A copy of the Warrant Agreement is on file at the office of Continental Stock Transfer & Trust Company of State Street, 30th Floor, New York, New York 10004, and is available to any Warrant holder on written request and without cost.	this
This certificate is not valid unless countersigned by the Transfer Agent and Registrar of the Company.	
Witness the facsimile seal of the Company and the facsimile signatures of its duly authorized officers.	
This Unit Certificate shall be governed and construed in accordance with the internal laws of the State of New York, without regard to conflicts of laprinciples thereof.	aws
[Seal]	
Ву	
Chairman Chief Financial Officer	

Aesther Healthcare Acquisition Corp.

The Company will furnish without charge to each stockholder who so requests, a statement of the powers, designations, preferences and relative, participating, optional or other special rights of each class of shares or series thereof of the Company and the qualifications, limitations, or restrictions of such preferences and/or rights.

The following abbreviations, when used in the inscription on the face of this certificate, shall be construed as though they were written out in full according to applicable laws or regulations:

TEN COM –	as tenants in common	UNIF GIFT MIN ACT Custodian
TEN ENT – JT TEN –	as tenants by the entireties as joint tenants with right of survivorship	(Cust) (Minor) under Uniform Gifts to Minors
JI IEN –	and not as tenants in common	Act
	and not as tenants in common	(State)
		(State)
Additional Abbreviations may	y also be used though not in the above list.	
For value received,	hereby sell(s), a	ussign(s) and transfer(s) unto
PLEASE INSERT SOCIAL S IDENTIFYING NUMBE		
(PLEA	SE PRINT OR TYPEWRITE NAME AND ADD	DRESS, INCLUDING ZIP CODE, OF ASSIGNEE(S))
_		
		Units represented by the within Certificate, and do hereby
irrevocably constitute and ap		Attorney to transfer the said
Units on the books of the with	nin named Company will full power of substitution	in the premises.
Dated		
		is assignment must correspond with the name as written upon the face of the particular, without alteration or enlargement or any change whatever.
Signature(s) Guaranteed:		
		A D A NIMOD ANGOMETICAN

THE SIGNATURE(S) SHOULD BE GUARANTEED BY AN ELIGIBLE GUARANTOR INSTITUTION (BANKS, STOCKBROKERS, SAVINGS AND LOAN ASSOCIATIONS AND CREDIT UNIONS WITH MEMBERSHIP IN AN APPROVED SIGNATURE GUARANTEE MEDALLION PROGRAM, PURSUANT TO S.E.C. RULE 17Ad-15).

The holder of this certificate shall be entitled to receive funds with respect to the underlying common stock from the trust fund only in the event of the Company's liquidation upon failure to consummate a business combination or if the holder seeks to convert his or her respective common stock underlying the unit upon consummation of such business combination or in connection with certain amendments to the Company's Amended and Restated Certificate of Incorporation. In no other circumstances shall the holder have any right or interest of any kind in or to the trust fund.

SHARES

SEE REVERSE FOR CERTAIN DEFINITIONS

CUSIP 00809L106

AESTHER HEALTHCARE ACQUISITION CORP. INCORPORATED UNDER THE LAWS OF THE STATE OF DELAWARE

CLASS A COMMON STOCK

This Certifies that

is the owner of

FULLY PAID AND NON-ASSESSABLE SHARES OF CLASS A COMMON STOCK, PAR VALUE OF \$0.0001 (THE "COMMON STOCK"), OF

AESTHER HEALTHCARE ACQUISITION CORP. (THE "COMPANY")

transferable on the books of the Company in person or by duly authorized attorney upon surrender of this certificate properly endorsed.

The Company will be required to redeem all of its shares of Common Stock if it is unable to complete a business combination within the time period set forth in the Company's Amended and Restated Certificate of Incorporation, as the same may be amended from time to time (the "*Charter*"), all as more fully described in the Company's final prospectus dated , 2021.

This certificate is not valid unless countersigned by the Transfer Agent and registered by the Registrar of the Company.

Witness the facsimile signatures of its duly authorized officers.

Chief Executive Officer	Chief Financial Officer

AESTHER HEALTHCARE ACQUISITION CORP.

The Company will furnish without charge to each stockholder who so requests a statement of the powers, designations, preferences and relative, participating, optional or other special rights of each class of stock or series thereof of the Company and the qualifications, limitations, or restrictions of such preferences and/or rights. This certificate and the shares represented thereby are issued and shall be held subject to all the provisions of the Charter and resolutions of the Board of Directors providing for the issue of securities (copies of which may be obtained from the secretary of the Company), to all of which the holder of this certificate by acceptance hereof assents. The following abbreviations, when used in the inscription on the face of this certificate, shall be construed as though they were written out in full according to applicable laws or regulations:

TEN COM	_	as tenants in common	UNIF GIFT MIN ACT	_	Custodian	
TEN ENT	_	as tenants by the entireties		(0	Cust)	(Minor)
JT TEN	-	 as joint tenants with right of survi- not as tenants in common 	vorship and	U	under Uniform Gifts to Minors Act	
					(State)	
Additional abbre	eviations m	ay also be used though not in the above list	t.			
For value receiv	red,	hereby sells, assigns and transfers unt	0			
	(P)	LEASE INSERT SOCIAL SECURITY OF	R OTHER IDENTIFY	ING NU	MBER(S) OF ASSIGNEE(S))	
(PLEASE PRINT OR TYPEWRITE NAME(S) AND ADDRESS(ES), INCLUDING ZIP CODE, OF ASSIGNEE(S))						
shares of the Common Stock represented by the within Certificate, and hereby irrevocably constitutes and appoints						
Attorney to trans	sfer the said	l shares of Common Stock on the books of	the within named Co	ompany w	ith full power of substitution in the pr	remises.

NOTICE: THE SIGNATURE(S) TO THIS ASSIGNMENT MUST CORRESPOND WITH THE NAME AS WRITTEN UPON THE FACE OF THE CERTIFICATE IN EVERY PARTICULAR, WITHOUT ALTERATION OR ENLARGEMENT OR ANY CHANGE WHATEVER.

Signature(s) Guaranteed:

By

Dated:

THE SIGNATURE(S) MUST BE GUARANTEED BY AN ELIGIBLE GUARANTOR INSTITUTION (BANKS, STOCKBROKERS, SAVINGS AND LOAN ASSOCIATIONS AND CREDIT UNIONS WITH MEMBERSHIP IN AN APPROVED SIGNATURE GUARANTEE MEDALLION PROGRAM, PURSUANT TO S.E.C. RULE 17Ad-15 OF THE SECURITIES EXCHANGE ACT OF 1934, AS AMENDED (OR ANY SUCCESSOR RULE)).

As more fully described in, and subject to the terms and conditions described in, the Company's final prospectus for its initial public offering dated 2021, the holder(s) of this certificate shall be entitled to receive a pro-rata portion of certain funds held in the trust account established in connection with the Company's initial public offering only in the event that (i) the Company redeems the shares of Common Stock sold in the Company's initial public offering and liquidates because it does not consummate an initial business combination by the date set forth in the Charter, as the same may be amended from time to time (such date being referred to herein as the "Last Date"), (ii) the Company redeems the shares of Common Stock sold in its initial public offering properly submitted in connection with a stockholder vote to amend the Charter to modify the substance or timing of the Company's obligation to redeem 100% of the Common Stock if it does not consummate an initial business combination by the Last Date or with respect to any other provisions relating to stockholders' rights or pre-initial business combination activity, or (iii) if the holder(s) seek(s) to redeem for cash his, her or its respective shares of Common Stock in connection with a tender offer (or proxy solicitation, solely in the event the Company seeks stockholder approval of the proposed initial business combination) setting forth the details of a proposed initial business combination. In no other circumstances shall the holder(s) have any right or interest of any kind in or to the trust account.

WARRANT AGREEMENT

This Warrant Agreement (this "<u>Agreement</u>") is made as of [●], 2021, between Aesther Healthcare Acquisition Corp., a Delaware corporation (the "<u>Company</u>"), and Continental Stock Transfer & Trust Company, LLC, a corporation, as warrant agent (in such capacity, the "<u>Warrant Agent</u>").

WHEREAS, the Company is engaged in an initial public offering (the "<u>Public Offering</u>") of units of the Company's equity securities, each such unit is comprised of one share of Class A common stock of the Company, par value \$0.0001 per share ("<u>Common Stock</u>"), and one-half of one redeemable Public Warrant (as defined below) (the "<u>Public Units</u>") and, in connection therewith, has determined to issue and deliver up to 5,000,000 warrants (or up to 5,750,000 warrants if the Over-allotment Option (as defined below) is exercised in full) to public investors in the Offering (the "<u>Public Warrants</u>"). Each whole Warrant entitles the holder thereof to purchase one half of one share of Common Stock for \$11.50 per share, subject to adjustment as described herein; and

WHEREAS, on [_____], 2021, the Company entered into that certain Placement Warrant Purchase Agreement with Aesther Healthcare Sponsor, LLC, a Delaware limited liability company (the "Sponsor"), pursuant to which the Sponsor agreed to purchase an aggregate of 5,261,000 private placement warrants (or up to 5,711,000 private placement warrants if the underwriters in the Offering exercise their Over-allotment Option in full) simultaneously with the closing of the Offering (and the closing the Over-allotment Option, if applicable) (the "Private Placement Warrants" and, together with the Public Warrants, the "Warrants") at a purchase price of \$1.00 per Private Placement Warrant bearing the legend set forth in Exhibit B attached hereto; and

WHEREAS, in order to finance the Company's transaction costs in connection with an intended initial Business Combination (as defined below), the Sponsor or an affiliate of the Sponsor or certain of the Company's executive officers and directors may, but are not obligated to, loan to the Company funds as the Company may require, of which up to \$1,500,000 of such loans may be convertible into up to an additional 1,500,000 Warrants at a price of \$1.00 per Warrant, and, in connection therewith, will issue and deliver up to an aggregate of 1,500,000 warrants (the "Working Capital Warrants"); and

WHEREAS, following consummation of the Offering, the Company may issue additional warrants ("<u>Post IPO Warrants</u>," and, together with the Private Placement Warrants, the Working Capital Warrants and the Public Warrants, the "<u>Warrants</u>") in connection with, or following the consummation by the Company of, a Business Combination (defined below); and

WHEREAS, the Company has filed with the Securities and Exchange Commission (the "Commission") registration statement on Form S-1, File No. 333-258012 (the "Registration Statement") and prospectus (the "Prospectus"), for the registration, under the Securities Act of 1933, as amended (the "Securities Act"), of the Public Units, the Public Warrants, the Common Stock included in the Units and the Common Stock to be issued to the Representative (as defined below); and

WHEREAS, following consummation of the Public Offering, the Company may issue additional warrants ("Post IPO Warrants" and together with the Public Warrants, Private Warrants, and Working Capital Warrants, the "Warrants") in connection with, or following the consummation by the Company of, a Business Combination (defined below); and

WHEREAS, the Company desires the Warrant Agent to act on behalf of the Company, and the Warrant Agent is willing to so act, in connection with the issuance, registration, transfer, exchange, redemption, and exercise of the Warrants; and

WHEREAS, the Company desires to provide for the form and provisions of the Warrants, the terms upon which they shall be issued and exercised, and the respective rights, limitation of rights, and immunities of the Company, the Warrant Agent, and the holders of the Warrants; and

WHEREAS, all acts and things have been done and performed which are necessary to make the Warrants, when executed on behalf of the Company and countersigned by or on behalf of the Warrant Agent (if a physical certificate is issued), as provided herein, the valid, binding, and legal obligations of the Company, and to authorize the execution and delivery of this Agreement.

NOW, THEREFORE, in consideration of the mutual agreements herein contained, the parties hereto agree as follows:

1. <u>Appointment of Warrant Agent</u>. The Company hereby appoints the Warrant Agent to act as agent for the Company for the Warrants, and the Warrant Agent hereby accepts such appointment and agrees to perform the same in accordance with the terms and conditions set forth in this Agreement.

2. Warrants.

- 2.1. <u>Form of Warrant</u>. Each Warrant shall be issued in registered form only, and, if a physical certificate is issued, shall be in substantially the form of Exhibit A hereto, the provisions of which are incorporated herein and shall be signed by, or bear the facsimile signature of, the Chairman of the Board of Directors or Chief Executive Officer and Treasurer, Secretary or Assistant Secretary of the Company and shall bear a facsimile of the Company's seal. In the event the person whose facsimile signature has been placed upon any Warrant shall have ceased to serve in the capacity in which such person signed the Warrant before such Warrant is issued, it may be issued with the same effect as if he or she had not ceased to be such at the date of issuance.
- 2.2. <u>Uncertificated Warrants</u>. Notwithstanding anything herein to the contrary, any Warrant, or portion thereof, may be issued as part of, and be represented by, a Unit, and any Warrant may be issued in uncertificated or book-entry form through the Warrant Agent and/or the facilities of The Depository Trust Company (the "<u>Depositary</u>") or other book-entry depositary system, in each case as determined by the Board of Directors of the Company or by an authorized committee thereof. Any Warrant so issued shall have the same terms, force and effect as a certificated Warrant that has been duly countersigned by the Warrant Agent in accordance with the terms of this Agreement. If the Depositary subsequently ceases to make its book-entry settlement system available for the Public Warrants, the Company may instruct the Warrant Agent regarding making other arrangements for book-entry settlement. In the event that the Public Warrants are not eligible for, or it is no longer necessary to have the Public Warrants available in, book-entry form, the Warrant Agent shall provide written instructions to the Depositary to deliver to the Warrant Agent for cancellation each Book-Entry Warrant Certificate, and the Company shall instruct the Warrant Agent to deliver to the Depositary definitive certificates in physical form evidencing such Warrants ("<u>Definitive Warrant Certificate</u>"). Such Definitive Warrant Certificate shall be in the form annexed hereto as Exhibit A, with appropriate insertions, modifications and omissions, as provided above.
- 2.3. <u>Effect of Countersignature</u>. Except with respect to uncertificated Warrants as described above, unless and until countersigned by the Warrant Agent pursuant to this Agreement, a Warrant shall be invalid and of no effect and may not be exercised by the holder thereof.

2.4. Registration.

- 2.4.1. <u>Warrant Register</u>. The Warrant Agent shall maintain books ("<u>Warrant Register</u>") for the registration of original issuance and the registration of transfer of the Warrants. Upon the initial issuance of the Warrants, the Warrant Agent shall issue and register the Warrants in the names of the respective holders thereof in such denominations and otherwise in accordance with instructions delivered to the Warrant Agent by the Company.
- 2.4.2. <u>Registered Holder</u>. Prior to due presentment for registration of transfer of any Warrant, the Company and the Warrant Agent may deem and treat the person in whose name such Warrant is then registered in the Warrant Register ("<u>registered holder</u>") as the absolute owner of such Warrant and of each Warrant represented thereby (notwithstanding any notation of ownership or other writing on the Warrant certificate made by anyone other than the Company or the Warrant Agent), for the purpose of any exercise thereof, and for all other purposes, and neither the Company nor the Warrant Agent shall be affected by any notice to the contrary.

- 2.5. <u>Detachability of Warrants</u>. The securities comprising the Public Units will not be separately transferable until the 52nd day following the date of the Prospectus or, if such 52nd day is not on a day, other than a Saturday, Sunday or federal holiday, on which banks in New York City are generally open for normal business (a "<u>Business Day</u>"), then on the immediately succeeding Business Day following such date, or earlier with the consent of EF Hutton (the "<u>Representative</u>"), but in no event will the Representative allow separate trading of the securities comprising the Public Units until (i) the Company has filed a Current Report on Form 8-K which includes an audited balance sheet reflecting the receipt by the Company of the gross proceeds of the Public Offering including the proceeds received by the Company from the exercise of the underwriters' over-allotment option in the Public Offering, if the over-allotment option is exercised prior to the filing of the Form 8-K, and (ii) the Company has issued a press release and has filed a Current Report on Form 8-K announcing when such separate trading shall begin (the "<u>Detachment Date</u>"). If the over-allotment option is exercised following the filing of a current report on Form 8-K pursuant to (i) above, a second or amended current report on Form 8-K will be filed to provide updated financial information to reflect the exercise of the over-allotment option.
- 2.6 <u>Fractional Warrants</u>. The Company shall not issue fractional Warrants other than as part of the Public Units, each of which is comprised of one share of Common Stock and one half of one Public Warrant. If, upon the detachment of Public Warrants from Public Units or otherwise, a holder of Warrants would be entitled to receive a fractional Warrant, the Company shall round down to the nearest whole number the number of Warrants to be issued to such holder.
 - 2.8 Private Warrants and Working Capital Warrants. The Private Warrants and Working Capital Warrants will be identical to the Public Warrants.
- 2.9 <u>Post IPO Warrants</u>. The Post IPO Warrants, when and if issued, shall have the same terms and be in the same form as the Public Warrants except as may be agreed upon by the Company.

3. Terms and Exercise of Warrants

- 3.1. Warrant Price. Each Warrant shall, when countersigned by the Warrant Agent (except with respect to uncertificated Warrants), entitle the registered holder thereof, subject to the provisions of such Warrant and of this Agreement, to purchase from the Company the number of shares of Common Stock stated therein, at the price of \$11.50 per share, subject to the adjustments provided in Section 4 hereof and in the last sentence of this Section 3.1. The term "Warrant Price" as used in this Agreement refers to the price per share at which the shares of Common Stock may be purchased at the time a Warrant is exercised. The Company in its sole discretion may lower the Warrant Price at any time prior to the Expiration Date (as defined below) for a period of not less than twenty (20) Business Days; provided, that the Company shall provide at least twenty (20) days' prior written notice of such reduction to registered holders of the Warrants and, provided further that any such reduction shall be applied consistently to all of the Warrants.
- 3.2. <u>Duration of Warrants</u>. A Warrant may be exercised only during the period commencing on the later of: (i) the date that is thirty (30) days after the first date on which the Company completes a merger, capital stock exchange, asset acquisition, stock purchase, reorganization or similar business combination, involving the Company and one or more businesses (a "<u>Business Combination</u>"), or (ii) the date that is twelve (12) months from the date of the closing of the Offering (as described more fully in the Registration Statement), and terminating at 5:00 p.m., New York City time on the earlier to occur of (i) the date that is five (5) years after the date on which the Company consummates a Business Combination, (ii) at 5:00 p.m., New York City time on the Redemption Date as provided in Section 6.2 of this Agreement and (iii) the liquidation of the Trust Account (defined below) ("<u>Expiration Date</u>"). The period of time from the date the Warrants will first become exercisable until the expiration of the Warrants shall hereafter be referred to as the "<u>Exercise Period.</u>" Except with respect to the right to receive the Redemption Price (as set forth in Section 6 hereunder), as applicable, each outstanding Warrant not exercised on or before the Expiration Date shall become void, and all rights thereunder and all rights in respect thereof under this Agreement shall cease at 5:00 p.m. New York City time on the Expiration Date. The Company in its sole discretion may extend the duration of the Warrants by delaying the Expiration Date; provided, however, that the Company will provide at least twenty (20) days' prior written notice of any such extension to registered holders and, provided further that any such extension shall be applied consistently to all of the Warrants. Notwithstanding anything to the contrary contained herein, for so long as any Private Placement Warrant is held by EF Hutton and/or their designees, such Private Placement Warrant may not be exercised after five years from the effective date of the Re

3.3. Exercise of Warrants.

- 3.3.1. <u>Payment</u>. Subject to the provisions of the Warrant and this Agreement, a Warrant, when countersigned by the Warrant Agent (if a physical certificate is issued), may be exercised by the registered holder thereof by surrendering it, at the office of the Warrant Agent, or at the office of its successor as Warrant Agent, in the Borough of Manhattan, City and State of New York, with the subscription form, as set forth in the Warrant, duly executed, and by paying in full the Warrant Price for each share of Common Stock as to which the Warrant is exercised and any and all applicable taxes due in connection with the exercise of the Warrant, as follows:
 - (a) in lawful money of the United States, by good certified check or good bank draft payable to the order of the Warrant Agent or wire transfer;
 - (b) in the event of a redemption pursuant to Section 6.1 hereof in which the Company's Board of Directors (the "Board") has elected to force all holders of Warrants to exercise such Warrants on a "cashless basis," by surrendering the Warrants for that number of shares of Common Stock equal to the quotient obtained by dividing (x) the product of the number of shares of Common Stock underlying the Warrants, multiplied by the excess of the "Fair Market Value" (defined in this Section 3.3.1(b) below), over the Warrant Price by (y) the Fair Market Value. Solely for purposes of this Section 3.3.1(b), the "Fair Market Value" shall mean the average reported closing price of the shares of Common Stock for the ten (10) trading days ending on the third trading day prior to the date on which the notice of redemption is sent to holders of the Warrants pursuant to Section 6 hereof; or
 - (c) in the event the registration statement required by Section 7.4 hereof is not effective and current within sixty (60) Business Days after the closing of a Business Combination, by surrendering such Warrants for that number of shares of Common Stock equal to the quotient obtained by dividing (x) the product of the number of shares of Common Stock underlying the Warrants, multiplied by the excess of the "Fair Market Value" (defined in this Section 3.3.1(c) below), over the Warrant Price by (y) the Fair Market Value; provided, however, that no cashless exercise shall be permitted unless the Fair Market Value is equal to or higher than the exercise price. Solely for purposes of this Section 3.3.1(c), the "Fair Market Value" shall mean the average reported last sale price of the shares of Common Stock for the ten (10) trading days ending on the trading day prior to the date of exercise.
- 3.3.2. <u>Issuance of Shares of Common Stock.</u> As soon as practicable after the exercise of any Warrant and the clearance of the funds in payment of the Warrant Price (if any), the Company shall issue to the registered holder of such Warrant a certificate or certificates, or book entry position, for the number of shares of Common Stock to which he, she or it is entitled, registered in such name or names as may be directed by him, her or it, and if such Warrant shall not have been exercised in full, a new countersigned Warrant, or book entry position, for the number of shares as to which such Warrant shall not have been exercised. Notwithstanding the foregoing, in no event will the Company be required to net cash settle the Warrant exercise. No Warrant shall be exercisable for cash and the Company shall not be obligated to issue shares of Common Stock upon exercise of a Warrant unless the Common Stock issuable upon such Warrant exercise has been registered, qualified or deemed to be exempt from registration or qualification under the securities laws of the state of residence of the registered holder of the Warrants. In the event that the condition in the immediately preceding sentences are not satisfied with respect to a Warrant, the holder of such Warrant shall not be entitled to exercise such Warrant for cash and such Warrant may have no value and expire worthless, in which case the purchaser of a Unit containing such Public Warrants shall have paid the full purchase price for the Unit solely for the shares of Common Stock underlying such Unit. Warrants may not be exercised by, or securities issued to, any registered holder in any state in which such exercise or issuance would be unlawful. The Company may require holders of Public Warrants to settle the Warrant on a "cashless basis" pursuant to Section 7.4. If, by reason of any exercise of Warrants on a "cashless basis", the holder of any Warrant would be entitled, upon the exercise of such Warrant, to receive a fractional interest in a share of Common Stock, th

- 3.3.3. <u>Valid Issuance</u>. All shares of Common Stock issued upon the proper exercise of a Warrant in conformity with this Agreement shall be validly issued, fully paid and nonassessable.
- 3.3.4. <u>Date of Issuance</u>. Each person in whose name any book entry position or certificate for shares of Common Stock are issued shall for all purposes be deemed to have become the holder of record of such shares on the date on which the Warrant, or book entry position representing such Warrant, was surrendered and payment of the Warrant Price was made, irrespective of the date of delivery of such certificate in the case of a certificated Warrant, except that, if the date of such surrender and payment is a date when the share transfer books of the Company or book entry system of the Warrant Agent are closed, such person shall be deemed to have become the holder of such shares of Common Stock at the close of business on the next succeeding date on which the share transfer books or book entry system are open.

3.3.5 Maximum Percentage. A holder of a Warrant may notify the Company in writing in the event it elects to be subject to the provisions contained in this subsection 3.3.5; however, no holder of a Warrant shall be subject to this subsection 3.3.5 unless he, she or it makes such election. If the election is made by a holder, the Warrant Agent shall not effect the exercise of the holder's Warrant, and such holder shall not have the right to exercise such Warrant, to the extent that after giving effect to such exercise, such person and any of its affiliates or any other person subject to aggregation with such person for purposes of the "beneficial ownership" test under Section 13 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or any "group" (within the meaning of Section 13 of the Exchange Act) of which such person is or may be deemed to be a part, would beneficially own (within the meaning of Section 13 of the Exchange Act) (or to the extent that for any reason the equivalent calculation under Section 16 of the Exchange Act and the rules and regulations thereunder would result in a higher ownership percentage, such higher percentage would be) in excess of 4.9% or 9.8% (as specified by the holder) (the "Maximum Percentage") of the shares of Common Stock outstanding immediately after giving effect to such exercise. For purposes of the foregoing sentence, the aggregate number of shares of Common Stock beneficially owned by such person and its affiliates or any such other person or group shall include the number of shares of Common Stock issuable upon exercise of the Warrant with respect to which the determination of such sentence is being made, but shall exclude shares of Common Stock that would be issuable upon (x) exercise of the remaining, unexercised portion of the Warrant beneficially owned by such person and its affiliates and (y) exercise or conversion of the unexercised or unconverted portion of any other securities of the Company beneficially owned by such person and its affiliates (including, without limitation, any convertible notes or convertible preferred stock or warrants) subject to a limitation on conversion or exercise analogous to the limitation contained herein. Except as set forth in the preceding sentence, for purposes of this paragraph, beneficial ownership shall be calculated in accordance with Section 13(d) of the Exchange Act. For purposes of the Warrant, in determining the number of outstanding shares of Common Stock, the holder may rely on the number of outstanding shares of Common Stock as reflected in (1) the Company's most recent annual report on Form 10-K, quarterly report on Form 10-Q, current report on Form 8-K or other public filing with the Commission as the case may be, (2) a more recent public announcement by the Company or (3) any other notice by the Company or the Warrant Agent setting forth the number of shares of Common Stock outstanding. For any reason at any time, upon the written request of the holder of the Warrant, the Company shall, within two (2) Business Days, confirm orally and in writing to such holder the number of shares of Common Stock then outstanding. In any case, the number of outstanding shares of Common Stock shall be determined after giving effect to the conversion or exercise of equity securities of the Company by the holder and its affiliates since the date as of which such number of outstanding shares of Common Stock were reported. By written notice to the Company, the holder of a Warrant may from time to time increase or decrease the Maximum Percentage applicable to such holder to any other percentage specified in such notice; provided, however, that any such increase shall not be effective until the sixty-first (61st) day after such notice is delivered to the Company.

4. Adjustments.

- 4.1. Stock Dividends; Split Ups. If after the date hereof, and subject to the provisions of Section 4.6 below, the number of outstanding shares of Common Stock are increased by a stock dividend payable in shares of Common Stock, or by a split up of shares of Common Stock, or other similar event, then, on the effective date of such stock dividend, split up or similar event, the number of shares of Common Stock issuable on exercise of each Warrant shall be increased in proportion to such increase in outstanding shares of Common Stock. A rights offering to holders of the Common Stock entitling holders to purchase shares of Common Stock at a price less than the "Fair Market Value" (as defined below) shall be deemed a stock dividend of a number of shares of Common Stock equal to the product of (i) the number of shares of Common Stock actually sold in such rights offering (or issuable under any other equity securities sold in such rights offering that are convertible into or exercisable for the Common Stock) and (ii) one (1) minus the quotient of (x) the price per share of Common Stock paid in such rights offering divided by (y) the Fair Market Value. For purposes of this subsection 4.1, (i) if the rights offering is for securities convertible into or exercisable for Common Stock, in determining the price payable for Common Stock, there shall be taken into account any consideration received for such rights, as well as any additional amount payable upon exercise or conversion and (ii) "Fair Market Value" means the volume weighted average price of the Common Stock as reported during the ten (10) trading day period ending on the trading day prior to the first date on which the shares of Common Stock trade on the applicable exchange or in the applicable market, regular way, without the right to receive such rights.
- 4.2. <u>Aggregation of Shares</u>. If after the date hereof, the number of outstanding shares of Common Stock is decreased by a consolidation, combination, reverse stock split or reclassification of shares of Common Stock or other similar event, then, on the effective date of such consolidation, combination, reverse stock split, reclassification or similar event, the number of shares of Common Stock issuable on exercise of each Warrant shall be decreased in proportion to such decrease in outstanding shares of Common Stock.
- 4.3 Extraordinary Dividends. If the Company, at any time while the Warrants are outstanding and unexpired, shall pay a dividend or make a distribution in cash, securities or other assets to the holders of the shares of Common Stock or other shares of the Company's capital stock into which the Warrants are convertible (an "Extraordinary Dividend"), then the Warrant Price shall be decreased, effective immediately after the effective date of such Extraordinary Dividend, by the amount of cash and the fair market value (as determined by the Company's Board of Directors, in good faith) of any securities or other assets paid in respect of such Extraordinary Dividend divided by all outstanding shares of the Company at such time (whether or not any stockholders waived their right to receive such dividend); provided, however, that none of the following shall be deemed an Extraordinary Dividend for purposes of this provision: (a) any adjustment described in Section 4.1 above, (b) any cash dividends or cash distributions which, when combined on a per share basis with all other cash dividends and cash distributions paid on the Common Stock during the 365-day period ending on the date of declaration of such dividend or distribution does not exceed \$0.50 per share (taking into account all of the outstanding shares of the Company at such time (whether or not any shareholders waived their right to receive such dividend) and as adjusted to appropriately reflect any of the events referred to in other subsections of this Section 4 and excluding cash dividends or cash distributions that resulted in an adjustment to the Warrant Price or to the number of shares of Common Stock issuable on exercise of each Warrant) but only with respect to the amount of the aggregate cash dividends or cash distributions equal to or less than \$0.50, (c) any payment to satisfy the redemption rights of the holders of the shares of Common Stock in connection with a proposed initial Business Combination or certain amendments to the Company's Amended and Restated Certificate of Incorporation (as described in the Registration Statement) or (d) any payment in connection with the Company's liquidation and the distribution of its assets upon its failure to consummate a Business Combination. Solely for purposes of illustration, if the Company, at a time while the Warrants are outstanding and unexpired, pays a cash dividend of \$0.35 and previously paid an aggregate of \$0.40 of cash dividends and cash distributions on the Common Stock during the 365-day period ending on the date of declaration of such \$0.35 dividend, then the Warrant Price will be decreased, effectively immediately after the effective date of such \$0.35 dividend, by \$0.25 (the absolute value of the difference between \$0.75 (the aggregate amount of all cash dividends and cash distributions paid or made in such 365-day period, including such \$0.35 dividend) and \$0.50 (the greater of (x) \$0.50 and (y) the aggregate amount of all cash dividends and cash distributions paid or made in such 365-day period prior to such \$0.35 dividend)). Furthermore, solely for the purposes of illustration, if following the closing of the Company's initial Business Combination, there were total shares outstanding of 100,000,000 and the Company paid a \$1.00 dividend to 17,500,000 of such shares (with the remaining 82,500,000 shares waiving their right to receive such dividend), then no adjustment to the Warrant Price would occur as a \$17.5 million dividend payment divided by 100,000,000 shares equals \$0.175 per share which is less than \$0.50 per share.

- 4.4 <u>Adjustments in Exercise Price</u>. Whenever the number of shares of Common Stock purchasable upon the exercise of the Warrants is adjusted, as provided in Sections 4.1 and 4.2 above, the Warrant Price shall be adjusted (to the nearest cent) by multiplying such Warrant Price immediately prior to such adjustment by a fraction (x) the numerator of which shall be the number of shares of Common Stock purchasable upon the exercise of the Warrants immediately prior to such adjustment, and (y) the denominator of which shall be the number of shares of Common Stock so purchasable immediately thereafter.
- 4.5. Replacement of Securities upon Reorganization, etc. In case of any reclassification or reorganization of the outstanding shares of Common Stock (other than a change covered by Section 4.1, 4.2 or 4.3 hereof or that solely affects the par value of the Common Stock), or in the case of any merger or consolidation of the Company with or into another corporation (other than a consolidation or merger in which the Company is the continuing corporation and that does not result in any reclassification or reorganization of the outstanding Common Stock), or in the case of any sale or conveyance to another corporation or entity of the assets or other property of the Company as an entirety or substantially as an entirety in connection with which the Company is dissolved, the Warrant holders shall thereafter have the right to purchase and receive, upon the basis and upon the terms and conditions specified in the Warrants and in lieu of the shares of Common Stock of the Company immediately theretofore purchasable and receivable upon the exercise of the rights represented thereby, the kind and amount of shares of stock or other securities or property (including cash) receivable upon such reclassification, reorganization, merger or consolidation, or upon a dissolution following any such sale or transfer, that the Warrant holder would have received if such Warrant holder had exercised his, her or its Warrant(s) immediately prior to such event. If any reclassification also results in a change in the Common Stock covered by Section 4.1, 4.2 or 4.3, then such adjustment shall be made pursuant to Sections 4.1, 4.2, 4.3, 4.4 and this Section 4.5. The provisions of this Section 4.5 shall similarly apply to successive reclassifications, reorganizations, mergers or consolidations, sales or other transfers. In no event will the Warrant Price be reduced to less than the par value per share issuable upon exercise of the Warrant. Notwithstanding anything to the contrary herein, in the event of any tender offer for shares of shares of C
- 4.6. <u>Issuance in connection with a Business Combination</u>. If, in connection with a Business Combination, the Company (a) issues additional shares of Common Stock or equity-linked securities at an issue price or effective issue price of less than \$9.20 per share (with such issue price or effective issue price as determined by the Company's Board of Directors, in good faith, and in the case of any such issuance to the Company's initial stockholders, or their affiliates, without taking into account any founders' shares held by them prior to such issuance), (b) the aggregate gross proceeds from such issuances represent more than 60% of the total equity proceeds, and interest thereon, available for the funding of the Business Combination on the date of the consummation of such Business Combination (net of redemptions), and (c) the Fair Market Value (as defined below) is below \$9.20 per share, the exercise price of the warrants will be adjusted (to the nearest cent) to be equal to 115% of the greater of (i) the Fair Market Value or (ii) the price at which the Company issues the shares of Common Stock or equity-linked securities, and the \$18.00 per share redemption trigger price will be adjusted (to the nearest cent) to be equal to 180% of the higher of the Fair Market Value and the price at which the Company issues shares of Common Stock or equity-linked securities. Solely for purposes of this Section 4.6, the "Fair Market Value" shall mean the volume weighted average reported trading price of the shares of Common Stock for the twenty (20) trading days starting on the trading day prior to the date of the consummation of the Business Combination.

- 4.7 <u>Notices of Changes in Warrant</u>. Upon every adjustment of the Warrant Price or the number of shares issuable upon exercise of a Warrant, the Company shall give written notice thereof to the Warrant Agent, which notice shall state the Warrant Price resulting from such adjustment and the increase or decrease, if any, in the number of shares purchasable at such price upon the exercise of a Warrant, setting forth in reasonable detail the method of calculation and the facts upon which such calculation is based. Upon the occurrence of any event specified in Sections 4.1, 4.2, 4.3, 4.4, 4.5, or 4.6, then, in any such event, the Company shall give written notice to each Warrant holder, at the last address set forth for such holder in the Warrant Register, of the record date or the effective date of the event. Failure to give such notice, or any defect therein, shall not affect the legality or validity of such event.
- 4.8. <u>No Fractional Warrants or Shares</u>. Notwithstanding any provision contained in this Agreement to the contrary, the Company shall not issue fractional shares upon exercise of Warrants. If, by reason of any adjustment made pursuant to this Section 4, the holder of any Warrant would be entitled, upon the exercise of such Warrant, to receive a fractional interest in a share, the Company shall, upon such exercise, round up to the nearest whole number of shares of Common Stock to be issued to the Warrant holder.
- 4.9. Form of Warrant. The form of Warrant need not be changed because of any adjustment pursuant to this Section 4, and Warrants issued after such adjustment may state the same Warrant Price and the same number of shares as is stated in the Warrants initially issued pursuant to this Agreement. However, the Company may at any time in its sole discretion make any change in the form of Warrant that the Company may deem appropriate and that does not affect the substance thereof, and any Warrant thereafter issued or countersigned, whether in exchange or substitution for an outstanding Warrant or otherwise, may be in the form as so changed.
- 4.10 Other Events. In case any event shall occur affecting the Company as to which none of the provisions of preceding subsections of this Section 4 are strictly applicable, but which would require an adjustment to the terms of the Warrants in order to (i) avoid an adverse impact on the Warrants and (ii) effectuate the intent and purpose of this Section 4, then, in each such case, the Company shall appoint a firm of independent public accountants, investment banking or other appraisal firm of recognized national standing, which shall give its opinion as to whether or not any adjustment to the rights represented by the Warrants is necessary to effectuate the intent and purpose of this Section 4 and, if they determine that an adjustment is necessary, the terms of such adjustment, provided, however, that under no circumstances shall the Warrants be adjusted pursuant to this Section 4.10, as a result of any issuance of securities in connection with the Business Combination. The Company shall adjust the terms of the Warrants in a manner that is consistent with any adjustment recommended in such opinion.
- 4.11 <u>No Adjustment.</u> For the avoidance of doubt, no adjustment shall be made to the terms of the Warrants solely as a result of an adjustment to the conversion ratio of the Class B Common Stock into shares of Common Stock or the conversion of the shares of Class B Common Stock into shares of Common Stock, in each case, pursuant to the Company's Amended and Restated Certificate of Incorporation, as further amended from time to time.

5. Transfer and Exchange of Warrants.

- 5.1. <u>Registration of Transfer</u>. The Warrant Agent shall register the transfer, from time to time, of any outstanding Warrant upon the Warrant Register, upon surrender of such Warrant for transfer, properly endorsed with signatures, in the case of certificated Warrants, properly guaranteed and accompanied by appropriate instructions for transfer. Upon any such transfer, a new Warrant representing an equal aggregate number of Warrants shall be issued and the old Warrant shall be cancelled by the Warrant Agent. In the case of certificated Warrants, the Warrants so cancelled shall be delivered by the Warrant Agent to the Company from time to time upon request.
- 5.2. <u>Procedure for Surrender of Warrants</u>. Warrants may be surrendered to the Warrant Agent, either in certificated form or in book entry position, together with a written request for exchange or transfer, and thereupon the Warrant Agent shall issue in exchange therefor one or more new Warrants, or book entry positions, as requested by the registered holder of the Warrants so surrendered, representing an equal aggregate number of Warrants; provided, however, that in the event that a Warrant surrendered for transfer bears a restrictive legend, the Warrant Agent shall not cancel such Warrant and issue new Warrants in exchange therefor until the Warrant Agent has received an opinion of counsel for the Company stating that such transfer may be made and indicating whether the new Warrants must also bear a restrictive legend.

- 5.3. <u>Fractional Warrants</u>. The Warrant Agent shall not be required to effect any registration of transfer or exchange which will result in the issuance of a warrant certificate or book-entry position for a fraction of a Warrant.
 - 5.4. Service Charges. No service charge shall be made for any exchange or registration of transfer of Warrants.
- 5.5. <u>Warrant Execution and Countersignature</u>. The Warrant Agent is hereby authorized to countersign and to deliver, in accordance with the terms of this Agreement, the Warrants required to be issued pursuant to the provisions of this Section 5, and the Company, whenever required by the Warrant Agent, will supply the Warrant Agent with Warrants duly executed on behalf of the Company for such purpose.
- 5.6. Private Warrants and Working Capital Warrants. The Warrant Agent shall not register any transfer of Private Warrants or Working Capital Warrants until 30 days after the consummation by the Company of an initial Business Combination, except for transfers (i) among the initial shareholders or to the initial shareholders' or the Company's officers, directors, consultants or their affiliates, (ii) to a holder's shareholders or members upon the holder's liquidation, in each case if the holder is an entity, (iii) by bona fide gift to a member of the holder's immediate family or to a trust, the beneficiary of which is the holder or a member of the holder's immediate family, in each case for estate planning purposes, (iv) by virtue of the laws of descent and distribution upon death, (v) pursuant to a qualified domestic relations order, (vi) to the Company for no value for cancellation in connection with the consummation of a Business Combination by private sales at prices no greater than the price at which the Private Warrants were originally purchased, (viii) in the event of the Company's liquidation prior to its consummation of an initial Business Combination or (ix) in the event that, subsequent to the consummation of an initial Business Combination, the Company completes a liquidation, merger, share exchange or other similar transaction which results in all of the Company's shareholders having the right to exchange their Common Stock for cash, securities or other property, in each case (except for clauses (vi), (viii) or (ix) or with the Company's prior written consent) on the condition that prior to such registration for transfer, the Warrant Agent shall be presented with written documentation pursuant to which each transferee (each, a "Permitted Transferee") or the trustee or legal guardian for such transferee agrees to be bound by the transfer restrictions contained in this section and any other applicable agreement the transferor is bound by.
- 5.7. <u>Transfers prior to Detachment</u>. Prior to the Detachment Date, the Public Warrants may be transferred or exchanged only together with the Unit in which such Warrant is included, and only for the purpose of effecting, or in conjunction with, a transfer or exchange of such Unit. Furthermore, each transfer of a Unit on the register relating to such Public Units shall operate also to transfer the Warrants included in such Unit. Notwithstanding the foregoing, the provisions of this Section 5.7 shall have no effect on any transfer of Warrants on or after the Detachment Date.

6. Redemption.

6.1. Redemption. Subject to Section 6.4 hereof, not less than all of the outstanding Warrants may be redeemed, at the option of the Company, at any time during the Exercise Period, at the office of the Warrant Agent, upon the notice referred to in Section 6.2, at the price of \$0.01 per Warrant ("Redemption Price"), provided that the closing price of the Common Stock equals or exceeds \$18.00 per share (subject to adjustment in accordance with Section 4 hereof), on each of twenty (20) trading days within any thirty (30) trading day period commencing after the Warrants become exercisable and ending on the third trading day prior to the date on which notice of redemption is given and provided that there is an effective registration statement covering the shares of Common Stock issuable upon exercise of the Warrants, and a current prospectus relating thereto, available throughout the 30-day redemption or the Company has elected to require the exercise of the Warrants on a "cashless basis" pursuant to subsection 3.3.1(b); provided, however, that if and when the Public Warrants become redeemable by the Company, the Company may not exercise such redemption right if the issuance of shares of Common Stock upon exercise of the Public Warrants is not exempt from registration or qualification under applicable state blue sky laws or the Company is unable to effect such registration or qualification.

- 6.2. <u>Date Fixed for, and Notice of, Redemption</u>. In the event the Company shall elect to redeem all of the Warrants that are subject to redemption, the Company shall fix a date for the redemption (the "<u>Redemption Date</u>"). Notice of redemption shall be mailed by first class mail, postage prepaid, by the Company not less than thirty (30) days prior to the Redemption Date to the registered holders of the Warrants to be redeemed at their last addresses as they shall appear on the registration books. Any notice mailed in the manner herein provided shall be conclusively presumed to have been duly given whether or not the registered holder received such notice.
- 6.3. Exercise After Notice of Redemption. The Public Warrants may be exercised, for cash (or on a "cashless basis" in accordance with Section 3 of this Agreement) at any time after notice of redemption shall have been given by the Company pursuant to Section 6.2 hereof and prior to the Redemption Date. In the event the Company determines to require all holders of Public Warrants to exercise their Warrants on a "cashless basis" pursuant to Section 3.3.1(b), the notice of redemption will contain the information necessary to calculate the number of shares of Common Stock to be received upon exercise of the Warrants, including the "Fair Market Value" in such case. On and after the Redemption Date, the record holder of the Warrants shall have no further rights except to receive, upon surrender of the Warrants, the Redemption Price.

7. Other Provisions Relating to Rights of Holders of Warrants.

- 7.1. <u>No Rights as Stockholder</u>. A Warrant does not entitle the registered holder thereof to any of the rights of a stockholder of the Company, including, without limitation, the right to receive dividends, or other distributions, exercise any preemptive rights to vote or to consent or to receive notice as stockholders in respect of the meetings of stockholders or the election of directors of the Company or any other matter.
- 7.2. <u>Lost, Stolen, Mutilated, or Destroyed Warrants</u>. If any Warrant is lost, stolen, mutilated, or destroyed, the Company and the Warrant Agent may on such terms as to indemnity or otherwise as they may in their discretion impose (which shall, in the case of a mutilated Warrant, include the surrender thereof), issue a new Warrant of like denomination, tenor, and date as the Warrant so lost, stolen, mutilated, or destroyed. Any such new Warrant shall constitute a substitute contractual obligation of the Company, whether or not the allegedly lost, stolen, mutilated, or destroyed Warrant shall be at any time enforceable by anyone.
- 7.3. <u>Reservation of Shares of Common Stock</u>. The Company shall at all times reserve and keep available a number of its authorized but unissued shares of Common Stock that will be sufficient to permit the exercise in full of all outstanding Warrants issued pursuant to this Agreement.
- 7.4. Registration of Shares of Common Stock. The Company agrees that as soon as practicable after the closing of its initial Business Combination, it shall use its best efforts to file with the Securities and Exchange Commission a registration statement for the registration, under the Act, of the Common Stock issuable upon exercise of the Warrants, and it shall use its best efforts to take such action as is necessary to register or qualify for sale, in those states in which the Warrants were initially offered by the Company and in those states where holders of Warrants then reside, the Common Stock issuable upon exercise of the Warrants, to the extent an exemption is not available. The Company will use its best efforts to cause the same to become effective and to maintain the effectiveness of such registration statement, and a current prospectus relating thereto, until the expiration of the Warrants in accordance with the provisions of this Agreement. If any such registration statement has not been declared effective by the 90th day following the closing of the Business Combination, holders of the Warrants shall have the right, during the period beginning on the 91st day after the closing of the Business Combination and ending upon such registration statement being declared effective by the Securities and Exchange Commission, and during any other period when the Company shall fail to have maintained an effective registration statement covering the shares of Common Stock issuable upon exercise of the Warrants, to exercise such Warrants on a "cashless basis" as determined in accordance with Section 3.3.1(c). The Company shall provide the Warrant Agent with an opinion of counsel for the Company (which shall be an outside law firm with securities law experience) stating that (i) the exercise of the Warrants on a cashless basis in accordance with this Section 7.4 is not required to be registered under the Act and (ii) the shares of Common Stock issued upon such exercise will be freely tradable under U.S. federal securities laws by anyone who is not an affiliate (as such term is defined in Rule 144 under the Act) of the Company and, accordingly, will not be required to bear a restrictive legend. For the avoidance of any doubt, unless and until all of the Warrants have been exercised on a cashless basis, the Company shall continue to be obligated to comply with its registration obligations under the first three sentences of this Section 7.4. The provisions of this Section 7.4 may not be modified, amended, or deleted without the prior written consent of the Representative.

8. Concerning the Warrant Agent and Other Matters.

8.1. <u>Payment of Taxes</u>. The Company will from time to time promptly pay all taxes and charges that may be imposed upon the Company or the Warrant Agent in respect of the issuance or delivery of shares of Common Stock upon the exercise of Warrants, but the Company shall not be obligated to pay any transfer taxes in respect of the Warrants or such shares of Common Stock.

8.2. Resignation, Consolidation, or Merger of Warrant Agent.

- 8.2.1. Appointment of Successor Warrant Agent. The Warrant Agent, or any successor to it hereafter appointed, may resign its duties and be discharged from all further duties and liabilities hereunder after giving sixty (60) days' notice in writing to the Company. If the office of the Warrant Agent becomes vacant by resignation or incapacity to act or otherwise, the Company shall appoint in writing a successor Warrant Agent in place of the Warrant Agent. If the Company shall fail to make such appointment within a period of thirty (30) days after it has been notified in writing of such resignation or incapacity by the Warrant Agent or by the holder of the Warrant (who shall, with such notice, submit his Warrant for inspection by the Company), then the holder of any Warrant may apply to the Supreme Court of the State of New York for the County of New York for the appointment of a successor Warrant Agent at the Company's cost. Any successor Warrant Agent, whether appointed by the Company or by such court, shall be a corporation organized and existing under the laws of the State of New York, in good standing and having its principal office in the Borough of Manhattan, City and State of New York, and authorized under such laws to exercise corporate trust powers and subject to supervision or examination by federal or state authority. After appointment, any successor Warrant Agent shall be vested with all the authority, powers, rights, immunities, duties, and obligations of its predecessor Warrant Agent with like effect as if originally named as Warrant Agent hereunder, without any further act or deed; but if for any reason it becomes necessary or appropriate, the predecessor Warrant Agent all the authority, powers, and rights of such predecessor Warrant Agent hereunder; and upon request of any successor Warrant Agent the Company shall make, execute, acknowledge, and deliver any and all instruments in writing for more fully and effectually vesting in and confirming to such successor Warrant Agent all such authority, po
- 8.2.2. <u>Notice of Successor Warrant Agent</u>. In the event a successor Warrant Agent shall be appointed, the Company shall give notice thereof to the predecessor Warrant Agent and the transfer agent for the shares of Common Stock not later than the effective date of any such appointment.
- 8.2.3. <u>Merger or Consolidation of Warrant Agent</u>. Any corporation into which the Warrant Agent may be merged or with which it may be consolidated or any corporation resulting from any merger or consolidation to which the Warrant Agent shall be a party shall be the successor Warrant Agent under this Agreement without any further act.

8.3. Fees and Expenses of Warrant Agent.

8.3.1. <u>Remuneration</u>. The Company agrees to pay the Warrant Agent reasonable remuneration for its services as such Warrant Agent hereunder and will reimburse the Warrant Agent upon demand for all expenditures that the Warrant Agent may reasonably incur in the execution of its duties hereunder.

8.3.2. <u>Further Assurances</u>. The Company agrees to perform, execute, acknowledge, and deliver or cause to be performed, executed, acknowledged, and delivered all such further and other acts, instruments, and assurances as may reasonably be required by the Warrant Agent for the carrying out or performing of the provisions of this Agreement.

8.4. Liability of Warrant Agent.

- 8.4.1. Reliance on Company Statement. Whenever in the performance of its duties under this Agreement, the Warrant Agent shall deem it necessary or desirable that any fact or matter be proved or established by the Company prior to taking or suffering any action hereunder, such fact or matter (unless other evidence in respect thereof be herein specifically prescribed) may be deemed to be conclusively proved and established by a statement signed by the Chief Executive Officer, Chief Financial Officer, Secretary or Chairman of the Board of Directors of the Company and delivered to the Warrant Agent. The Warrant Agent may rely upon such statement for any action taken or suffered in good faith by it pursuant to the provisions of this Agreement.
- 8.4.2. <u>Indemnity</u>. The Warrant Agent shall be liable hereunder only for its own fraud, gross negligence, willful misconduct or bad faith. The Company agrees to indemnify the Warrant Agent and save it harmless against any and all liabilities, including judgments, costs and reasonable counsel fees, for anything done or omitted by the Warrant Agent in the execution of this Agreement except as a result of the Warrant Agent's fraud, gross negligence, willful misconduct, or bad faith.
- 8.4.3. Exclusions. The Warrant Agent shall have no responsibility with respect to the validity of this Agreement or with respect to the validity or execution of any Warrant (except its countersignature thereof); nor shall it be responsible for any breach by the Company of any covenant or condition contained in this Agreement or in any Warrant; nor shall it be responsible to make any adjustments required under the provisions of Section 4 hereof or responsible for the manner, method, or amount of any such adjustment or the ascertaining of the existence of facts that would require any such adjustment; nor shall it by any act hereunder be deemed to make any representation or warranty as to the authorization or reservation of any shares of Common Stock to be issued pursuant to this Agreement, the Amended and Restated Certificate of Incorporation of the Company, or any Warrant or as to whether any shares of Common Stock will, when issued, be valid and fully paid and nonassessable.
- 8.5. <u>Acceptance of Agency</u>. The Warrant Agent hereby accepts the agency established by this Agreement and agrees to perform the same upon the terms and conditions herein set forth and among other things, shall account promptly to the Company with respect to Warrants exercised and concurrently account for, and pay to the Company, all monies received by the Warrant Agent for the purchase of shares of Common Stock through the exercise of Warrants.

9. Miscellaneous Provisions.

- 9.1. <u>Successors</u>. All the covenants and provisions of this Agreement by or for the benefit of the Company or the Warrant Agent shall bind and inure to the benefit of their respective successors and assigns.
- 9.2. <u>Notices</u>. Any notice, statement or demand authorized by this Agreement to be given or made by the Warrant Agent or by the holder of any Warrant to or on the Company shall be sufficiently given (i) if by email when the email is sent, (ii) if by hand or overnight delivery, when so delivered, or (iii) if sent by certified mail or private courier service within five (5) days after deposit of such notice, postage prepaid, addressed (until another address is filed in writing by the Company with the Warrant Agent), as follows:

Aesther Healthcare Acquisition Corp. 515 Madison Ave. Suite 8078 New York, New York 10022 Attn: Suren Ajjarapu, Chief Executive Officer

E-mail: Suren@aestherhealthcarespac.com

Any notice, statement or demand authorized by this Agreement to be given or made by the holder of any Warrant or by the Company to or on the Warrant Agent shall be sufficiently given (i) if by email, when the email is sent, (ii) if by hand or overnight delivery, when so delivered, or (iii) if sent by certified mail or private courier service within five days after deposit of such notice, postage prepaid, addressed (until another address is filed in writing by the Warrant Agent with the Company), as follows:

Continental Stock Transfer & Trust Company, LLC 1 State Street, 30th Floor New York, NY 10004 Attn: Compliance Department

with a copy in each case to:

The Loev Law Firm, PC Attn: David M. Loev, Esq. 6300 West Loop South, Suite 280 Bellaire, Texas 77401 Email: dloev@loevlaw.com

- 9.3. <u>Applicable Law.</u> The validity, interpretation, and performance of this Agreement and of the Warrants shall be governed in all respects by the laws of the State of New York, without giving effect to conflicts of law principles that would result in the application of the substantive laws of another jurisdiction. The Company hereby agrees that any action, proceeding or claim against it arising out of or relating in any way to this Agreement shall be brought and enforced in the courts of the State of New York or the United States District Court for the Southern District of New York. The Company hereby waives any objection that such courts represent an inconvenient forum. Any such process or summons to be served upon the Company may be served by transmitting a copy thereof by registered or certified mail, return receipt requested, postage prepaid, addressed to it at the address set forth in Section 9.2 hereof. Such mailing shall be deemed personal service and shall be legal and binding upon the Company in any action, proceeding or claim.
- 9.4. <u>Persons Having Rights under this Agreement</u>. Nothing in this Agreement expressed and nothing that may be implied from any of the provisions hereof is intended, or shall be construed, to confer upon, or give to, any person or corporation other than the parties hereto and the registered holders of the Warrants and, for the purposes of Sections 7.4, 9.4 and 9.8 hereof, the Representative, any right, remedy, or claim under or by reason of this Warrant Agreement or of any covenant, condition, stipulation, promise, or agreement hereof. The Representative shall be deemed to be a third-party beneficiary of this Agreement with respect to Sections 7.4, 9.4 and 9.8 hereof. All covenants, conditions, stipulations, promises, and agreements contained in this Warrant Agreement shall be for the sole and exclusive benefit of the parties hereto (and the Representative with respect to the Sections 7.4, 9.4 and 9.8 hereof) and their successors and assigns and of the registered holders of the Warrants.
- 9.5. Examination of the Warrant Agreement. A copy of this Agreement shall be available at all reasonable times at the office of the Warrant Agent in the Borough of Manhattan, City and State of New York, for inspection by the registered holder of any Warrant. The Warrant Agent may require any such holder to submit his Warrant for inspection by it.
- 9.6. <u>Counterparts</u>. This Agreement may be executed in any number of original or facsimile counterparts and each of such counterparts shall for all purposes be deemed to be an original, and all such counterparts shall together constitute but one and the same instrument.
- 9.7. Effect of Headings. The section headings herein are for convenience only and are not part of this Agreement and shall not affect the interpretation thereof.

- 9.8 Amendments. This Agreement may be amended by the parties hereto without the consent of any registered holder for the purpose of curing any ambiguity, or of curing, correcting or supplementing any defective provision contained herein or adding or changing any other provisions with respect to matters or questions arising under this Agreement as the parties may deem necessary or desirable and that the parties deem shall not adversely affect the interest of the registered holders. All other modifications or amendments, including any amendment to increase the Warrant Price or shorten the Exercise Period, shall require the written consent or vote of the registered holders of (i) a majority of the then outstanding Public Warrants if such modification or amendment is being undertaken prior to, or in connection with, the consummation of a Business Combination. Notwithstanding the foregoing, the Company may lower the Warrant Price or extend the duration of the Exercise Period pursuant to Sections 3.1 and 3.2, respectively, without the consent of the registered holders. The provisions of this Section 9.8 may not be modified, amended or deleted without the prior written consent of the Representative.
- 9.9 <u>Trust Account Waiver</u>. The Warrant Agent acknowledges and agrees that it shall not make any claims or proceed against the trust account established by the Company in connection with the Public Offering (as more fully described in the Registration Statement) ("<u>Trust Account</u>"), including by way of set-off, and shall not be entitled to any funds in the Trust Account under any circumstance. In the event that the Warrant Agent has a claim against the Company under this Agreement, the Warrant Agent will pursue such claim solely against the Company and not against the property held in the Trust Account.
- 9.10 <u>Severability</u>. This Agreement shall be deemed severable, and the invalidity or unenforceability of any term or provision hereof shall not affect the validity or enforceability of this Agreement or of any other term or provision hereof. Furthermore, in lieu of any such invalid or unenforceable term or provision, the parties hereto intend that there shall be added as a part of this Agreement a provision as similar in terms to such invalid or unenforceable provision as may be possible and be valid and enforceable.

[signature page follows]

IN WITNESS WHEREOF, this Agreement has been duly executed b	y the parties hereto as of the day and year first above written.
	AESTHER HEALTHCARE ACQUISITION CORP.
	Ву:
	Name: Suren Ajjarapu
	Title: Chief Executive Officer
	CONTINENTAL STOCK TRANSFER & TRUST CO.
	By:
	Name:
	Title:
[Signature Page	to Warrant Agreement

EXHIBIT A

NUMBER	
_	

(SEE REVERSE SIDE FOR LEGEND) THIS WARRANT WILL BE VOID IF NOT EXERCISED PRIOR TO THE EXPIRATION DATE (DEFINED BELOW)

WARRANTS

AESTHER HEALTHCARE ACQUISITION CORP.

CUSH	1 C	

WARRANT

THIS CERTIFIES THAT, for value received

is the registered holder of a warrant or warrants (the "Warrant(s)") of Aesther Healthcare Acquisition Corp., a Delaware corporation (the "Company"), expiring at 5:00 p.m., New York City time, on the five year anniversary of the Company's completion of an initial merger, business combination, capital share exchange, asset acquisition, stock purchase, recapitalization, reorganization or other similar business combination with one or more businesses or entities (a "Business Combination"), to purchase one fully paid and non-assessable share of common stock, par value \$0.0001 per share ("Shares"), of the Company for each whole Warrant evidenced by this Warrant Certificate. The Warrant entitles the holder thereof to purchase from the Company, commencing on the later of (a) 30 days after the Company's completion of an initial Business Combination, and (b) 12 months from the closing of the Company's initial public offering ("IPO"), such number of Shares of the Company at the Warrant Price (as defined below), upon surrender of this Warrant Certificate and payment of the Warrant Price at the office or agency of Continental Stock Transfer & Trust Company (the "Warrant Agent"), but only subject to the conditions set forth herein and in the Warrant Agreement between the Company and Continental Stock Transfer & Trust Company. In no event will the Company be required to net cash settle any warrant exercise. The term "Warrant Price" as used in this Warrant Certificate refers to the price per Share at which Shares may be purchased at the time the Warrant is exercised. The initial Warrant Price per Share is equal to \$11.50 per share. The Warrant Agreement provides that upon the occurrence of certain events the Warrant Price, the Redemption Trigger Price (as defined below), and the number of Shares purchasable hereunder, set forth on the face hereof, may, subject to certain conditions, be adjusted.

No fraction of a Share will be issued upon any exercise of a Warrant. If the holder of a Warrant would be entitled to receive a fraction of a Share upon any exercise of a Warrant, the Company shall, upon such exercise, round up to the nearest whole number the number of Shares to be issued to such holder.

Upon any exercise of the Warrant for less than the total number of full Shares provided for herein, there shall be issued to the registered holder hereof or the registered holder's assignee a new Warrant Certificate covering the number of Shares for which the Warrant has not been exercised.

Warrant Certificates, when surrendered at the office or agency of the Warrant Agent by the registered holder in person or by attorney duly authorized in writing, may be exchanged in the manner and subject to the limitations provided in the Warrant Agreement, but without payment of any service charge, for another Warrant Certificate or Warrant Certificates of like tenor and evidencing in the aggregate a like number of Warrants.

Upon due presentment for registration of transfer of the Warrant Certificate at the office or agency of the Warrant Agent, a new Warrant Certificate or Warrant Certificates of like tenor and evidencing in the aggregate a like number of Warrants shall be issued to the transferee in exchange for this Warrant Certificate, subject to the limitations provided in the Warrant Agreement, without charge except for any applicable tax or other governmental charge.

The Company and the Warrant Agent may deem and treat the registered holder as the absolute owner of this Warrant Certificate (notwithstanding any notation of ownership or other writing hereon made by anyone), for the purpose of any exercise hereof, of any distribution to the registered holder, and for all other purposes, and neither the Company nor the Warrant Agent shall be affected by any notice to the contrary.

This Warrant does not entitle the registered holder to any of the rights of a stockholder of the Company.

The Company reserves the right to call the Warrant at any time prior to its exercise with a notice of call in writing to the holders of record of the Warrant, giving at least 30 days' notice of such call, at any time while the Warrant is exercisable, if the last sale price of the Shares has been at least \$18.00 per share (the "Redemption Trigger Price") on each of 20 trading days within any 30 trading day period (the "30-day trading period") ending on the third business day prior to the date on which notice of such call is given and if, and only if, there is a current registration statement in effect with respect to the Shares underlying the Warrants for the entire 30-day trading period and continuing each day thereafter until the date of redemption. The call price of the Warrants is to be \$0.01 per Warrant. Any Warrant either not exercised or tendered back to the Company by the end of the date specified in the notice of call shall be canceled on the books of the Company and have no further value except for the \$0.01 call price.

Ву		
President	Secretary	
	16	

SUBSCRIPTION FORM

To Be Executed by the Registered Holder in Order to Exercise Warrants

The undersigned Registered Holder irrevocably elects to exercise Common Stock issuable upon the exercise of such Warrants, and requests	Warrants represented by this Warrant Certificate, and to purchase the that Certificates for such shares shall be issued in the name of
(PLEASE TYPE OR PF	RINT NAME AND ADDRESS)
(SOCIAL SECURITY OR T	'AX IDENTIFICATION NUMBER)
and be delivered to	PRINT OR TYPE NAME AND ADDRESS)
(PLEASE F	PRINT OR T TPE NAME AND ADDRESS)
and, if such number of Warrants shall not be all the Warrants evidenced by Warrants be registered in the name of, and delivered to, the Registered Ho	this Warrant Certificate, that a new Warrant Certificate for the balance of such lder at the address stated below:
Dated:	_
	(SIGNATURE)
	(ADDRESS)
	(TAX IDENTIFICATION NUMBER)
	17

ASSIGNMENT

To Be Executed by the Registered Holder in Order to Assign Warrants

For Value Received,	hereby sell, assign, and transfer unto
	(PLEASE TYPE OR PRINT NAME AND ADDRESS)
and be delivered to	(SOCIAL SECURITY OR TAX IDENTIFICATION NUMBER)
	(PLEASE PRINT OR TYPE NAME AND ADDRESS)
	of the Warrants represented by this Warrant Certificate, and hereby irrevocably constitute and appoint Attorney to transfer this Warrant Certificate on the books of the Company, with full power of substitution in the
premises.	<u> </u>
Dated:	
	(SIGNATURE)
PARTICULAR, WITHOUT ALTE	SSIGNMENT OF THE SUBSCRIPTION FORM MUST CORRESPOND TO THE NAME WRITTEN UPON THE FACE OF THIS WARRANT CERTIFICATE IN EVERY EXATION OR ENLARGEMENT OR ANY CHANGE WHATSOEVER, AND MUST BE GUARANTEED BY A COMMERCIAL BANK OR TRUST COMPANY OR A MEMBER ICAN, NASDAQ, NEW YORK STOCK EXCHANGE, PACIFIC STOCK EXCHANGE, OR CHICAGO STOCK EXCHANGE.
	19

EXHIBIT B

LEGEND FOR PRIVATE PLACEMENT WARRANTS AND WORKING CAPITAL WARRANTS

"THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY STATE SECURITIES LAWS, AND MAY NOT BE OFFERED, SOLD, TRANSFERRED OR OTHERWISE DISPOSED OF UNLESS REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, AND ANY APPLICABLE STATE SECURITIES LAWS OR AN EXEMPTION FROM REGISTRATION IS AVAILABLE. IN ADDITION, SUBJECT TO ANY ADDITIONAL LIMITATIONS ON TRANSFER DESCRIBED IN THE LETTER AGREEMENT BY AND AMONG AESTHER HEALTHCARE ACQUISITION CORP. (THE "COMPANY"), AESTHER HEALTHCARE SPONSOR, LLC, EF HUTTON AND THE OTHER PARTIES THERETO, THE SECURITIES REPRESENTED BY THIS CERTIFICATE MAY NOT BE SOLD OR TRANSFERRED PRIOR TO THE DATE THAT IS THIRTY (30) DAYS AFTER THE DATE UPON WHICH THE COMPANY COMPLETES ITS INITIAL BUSINESS COMBINATION (AS DEFINED IN SECTION 3 OF THE WARRANT AGREEMENT REFERRED TO HEREIN) EXCEPT TO A PERMITTED TRANSFEREE (AS DEFINED IN SECTION 5.6 OF THE WARRANT AGREEMENT) WHO AGREES IN WRITING WITH THE COMPANY TO BE SUBJECT TO SUCH TRANSFER PROVISIONS.

SECURITIES EVIDENCED BY THIS CERTIFICATE AND SHARES OF CLASS A COMMON STOCK OF THE COMPANY ISSUED UPON EXERCISE OF SUCH SECURITIES SHALL BE ENTITLED TO REGISTRATION RIGHTS UNDER A REGISTRATION RIGHTS AGREEMENT TO BE EXECUTED BY THE COMPANY."

The Loev Law Firm, PC

CORPORATE, SECURITIES, PUBLIC/PRIVATE OFFERINGS, CONTRACTS, MERGERS & ACQUISITIONS, LITIGATION

David M. Loev John S. Gillies Timothy J. Henderson* 6300 West Loop South, Suite 280 Bellaire, Texas 77401 Telephone (713) 524-4110 Facsimile (713) 524-4122 www.loevlaw.com

* Of Counsel, Board Certified Civil Trial Law (Texas)

September 2, 2021

Aesther Healthcare Acquisition Corp. 515 Madison Avenue, Suite 8078 New York, New York 10022

Re: Registration Statement on Form S-1

Ladies and Gentlemen:

We have acted as counsel to Aesther Healthcare Acquisition Corp., a Delaware corporation (the "Company"), in connection with the preparation and filing of a Registration Statement on Form S-1, File No. 333-258012 (as amended, the "Registration Statement"), initially filed on July 19, 2021, with the U.S. Securities and Exchange Commission (the "Commission") under the Securities Act of 1933, as amended (the "Act"). The Registration Statement relates to the proposed issuance and sale from time to time of up to 11,500,000 units of the Company, including the underwriters' over-allotment option (collectively the "Units"), with each Unit consisting of one share of Class A common stock of the Company, par value \$0.0001 per share (the "Common Stock"); and one-half of one redeemable warrant of the Company, each whole warrant entitling the holder thereof to purchase one share of Common Stock (the "Warrants"). Such Common Stock, Warrants, and Units are referred to collectively in this opinion as "Securities."

In connection with the opinion expressed herein, we have examined originals or copies, certified or otherwise identified to our satisfaction, of (i) the form of Amended and Restated Certificate of Incorporation of the Company that is filed as Exhibit 3.2 to the Registration Statement; (ii) the Bylaws of the Company that are filed as Exhibit 3.3 to the Registration Statement; (iii) the Registration Statement; (iv) the form of the underwriting agreement proposed to be entered into between the Company and the underwriters named therein (the "Underwriting Agreement") that is filed as Exhibit 1.1 to the Registration Statement; (vi) the Specimen Unit Certificate that is filed as Exhibit 4.2 to the Registration Statement; (vii) the Specimen Warrant Certificate that is filed as Exhibit 4.3 to the Registration Statement; (viii) the form of warrant agreement proposed to be entered into by and between the Company and Continental Stock Transfer & Trust Company, as warrant agent (the "Warrant Agent"), that is filed as Exhibit 4.4 to the Registration Statement; and (ix) such other certificates, statutes and other instruments and documents as we considered appropriate for purposes of the opinion hereafter expressed. We have also reviewed such questions of law as we have deemed necessary or appropriate. As to matters of fact relevant to the opinion expressed herein, and as to factual matters arising in connection with our examination of corporate documents, records and other documents and writings, we relied upon certificates and other communications of corporate officers of the Company, without further investigation as to the facts set forth therein.

Aesther Healthcare Acquisition Corp. Page 2 of 3

In connection with rendering the opinion set forth below, we have assumed that (i) all information contained in all documents reviewed by us is true and correct, (ii) all signatures on all documents examined by us are genuine, (iii) all documents submitted to us as originals are authentic and all documents submitted to us as copies conform to the originals of those documents, (iv) the Registration Statement, and any amendments thereto (including post-effective amendments), will have become effective, and (v) all Units will be issued and sold in compliance with applicable federal and state securities laws and in the manner specified in the Registration Statement and the Prospectus relating thereto. We have further assumed that each of the documents identified in clauses (i) through (viii) of the preceding paragraph will be entered into, adopted or filed as appropriate. As to matters of fact material to this opinion, we have relied, without independent verification, upon statements and representations of representatives of the Company and public officials.

This opinion is limited in all respects to the General Corporation Law of the State of Delaware (including the applicable provisions of the Delaware Constitution and the reported judicial decisions interpreting those laws currently in effect) and, as to the Units, and the Warrants, constituting legally binding obligations of the Company, solely with respect to the laws of the State of New York, and no opinion is expressed with respect to the laws of any other jurisdiction or any effect which such laws may have on the opinions expressed herein. Our opinion is based on these laws as in effect on the date hereof and as of the effective date of the Registration Statement, and we assume no obligation to revise or supplement this opinion after the effective date of the Registration Statement should the law be changed by legislative action, judicial decision or otherwise. Where our opinions expressed herein refer to events to occur at a future date, we have assumed that there will have been no changes in the relevant law or facts between the date hereof and such future date. Our opinions expressed herein are limited to the matters expressly stated herein and no opinion is implied or may be inferred beyond the matters expressly stated. Not in limitation of the foregoing, we are not rendering any opinion as to the compliance with any other federal or state law, rule or regulation relating to securities (including any blue sky laws), or to the sale or issuance thereof.

Based upon the foregoing, and subject to the assumptions, qualifications and limitations set forth herein, we are of the opinion that:

1. Units. When the Registration Statement becomes effective under the Act, and when the offering is completed as contemplated by the Registration Statement, and the Units are delivered to and paid for by the underwriters in accordance with the terms of the Underwriting Agreement, such Units will be legally binding obligations of the Company, enforceable in accordance with their terms except: (a) as such enforceability may be limited by bankruptcy, insolvency, reorganization or similar laws affecting creditors' rights generally and by general equitable principles (regardless of whether enforceability is considered in a proceeding in equity or at law); (b) as enforceability of any indemnification or contribution provision may be limited under the federal and state securities laws; (c) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to the equitable defenses and to the discretion of the court before which any proceeding therefor may be brought; (d) we express no opinion as to whether a state court outside of the State of New York or a federal court of the United States would give effect to the choice of New York law provided for in the Warrant Agreement; and (e) with respect to the Common Stock, we express no opinion to the extent that, notwithstanding its current reservation of shares of Common Stock, future issuances of securities, including the Common Stock, of the Company and/or adjustments to outstanding securities, including the Warrants underlying the Units, of the Company may cause the number of shares of Common Stock underlying the Units, including the Common Stock issuable upon exercise of the Warrants underlying the Units, to exceed the number that remain authorized but unissued.

Aesther Healthcare Acquisition Corp. Page 3 of 3

- **2.** <u>Common Stock.</u> When the Registration Statement becomes effective under the Act and when the offering is completed as contemplated by the Registration Statement, and such the shares of Common Stock is delivered to and paid for by the underwriters in accordance with the terms of the Underwriting Agreement, the shares of Common Stock will be validly issued, fully paid and non-assessable.
- **3.** Warrants. When the Registration Statement becomes effective under the Act and when the Warrants underlying the Units are issued, delivered and paid for as part of the Units, as contemplated by the Registration Statement and the Underwriting Agreement, and assuming the due authorization, execution and delivery of such Warrants by the Warrant Agent, such Warrants will be legally binding obligations of the Company enforceable in accordance with their terms except: (a) as such enforceability may be limited by bankruptcy, insolvency, reorganization or similar laws affecting creditors' rights generally and by general equitable principles (regardless of whether enforceability is considered in a proceeding in equity or at law); (b) as enforceability of any indemnification or contribution provision may be limited under the federal and state securities laws; (c) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to the equitable defenses and to the discretion of the court before which any proceeding therefor may be brought; (d) we express no opinion as to whether a state court outside of the State of New York or a federal court of the United States would give effect to the choice of New York law provided for in the Warrant Agreement; (e) with respect to the Common Stock, we express no opinion to the extent that, notwithstanding its current reservation of shares of Common Stock, future issuances of securities, including the Common Stock, of the Company and/or adjustments to outstanding securities, including the Warrants, of the Company may cause the Warrants to be exercisable for more shares of Common Stock than the number that remain authorized but unissued and (f) we have assumed the Exercise Price (as defined in the Warrant Agreement) will not be adjusted to an amount below the par value per share of the Common Stock.

We consent to the filing of this opinion as an exhibit to the Registration Statement and to the reference to us under the caption "<u>Legal Matters</u>" in the prospectus forming a part of the Registration Statement. In giving such consent, we do not thereby admit that we are in the category of persons whose consent is required under Section 7 of the Act or the rules and regulations of the Commission thereunder.

Very truly yours,

/s/ The Loev Law Firm, PC

The Loev Law Firm, PC

], 2021

Aesther Healthcare Acquisition Corp. 515 Madison Avenue, Suite 8078 New York. New York 10022

Re: Initial Public Offering

Ladies and Gentlemen:

This letter (this "Letter Agreement") is being delivered to you in accordance with the Underwriting Agreement (the "<u>Underwriting Agreement</u>") to be entered into by and between Aesther Healthcare Acquisition Corp., a Delaware corporation (the "<u>Company</u>") and EF Hutton, division of Benchmark Investments, LLC, as representative of the underwriters (each, an "<u>Underwriter</u>" and collectively, the "<u>Underwriters</u>"), relating to an underwritten initial public offering (the "<u>Public Offering</u>"), of up to 11,500,000 of the Company's units (including up to 1,500,000 units that may be purchased to cover overallotments, if any) (the "<u>Units</u>"), each comprised of one share of the Company's Class A common stock, par value \$0.0001 per share (the "<u>Common Stock</u>"), and one-half of one redeemable warrant. Each whole warrant (each, a "<u>Warrant</u>") entitles the holder thereof to purchase one share of Common Stock at a price of \$11.50 per share, subject to adjustment. The Units will be sold in the Public Offering pursuant to a registration statement on Form S-1 (File No. 333-258012) and prospectus (the "<u>Prospectus</u>") filed by the Company with the U.S. Securities and Exchange Commission (the "<u>Commission</u>") and the Company has applied to have the Units listed on the Nasdaq Global Market. Certain capitalized terms used herein are defined in paragraph 11 hereof.

In order to induce the Company and the Underwriters to enter into the Underwriting Agreement and to proceed with the Public Offering and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Aesther Healthcare Sponsor, LLC (the "Sponsor") and each of the undersigned individuals, each of whom is a member of the Company's board of directors and/or management team or an advisor of the Company (each, an "Insider" and collectively, the "Insiders"), hereby agrees with the Company as follows:

1. The Sponsor and each Insider agrees that if the Company seeks stockholder approval of a proposed Business Combination, then in connection with such proposed Business Combination, it, he or she shall (i) vote any shares of Capital Stock owned by it, him or her in favor of any proposed Business Combination and (ii) not redeem any shares of Common Stock owned by it, him or her in connection with such stockholder approval. If the Company engages in a tender offer in connection with any proposed Business Combination, the Sponsor and each Insider agrees that it, he or she will not seek to sell its, his or her shares of Common Stock to the Company in connection with such tender offer.

2. The Sponsor and each Insider hereby agrees that in the event that the Company fails to consummate a Business Combination within 12 months from the closing of the Public Offering (or up to 18 months if the Company extended such period in accordance with the Company's amended and restated certificate of incorporation (the "Charter")) or such later period approved by the Company's stockholders in accordance with the Company's Charter, the Sponsor and each Insider shall take all reasonable steps to cause the Company to (i) cease all operations except for the purpose of winding up, (ii) as promptly as reasonably possible but not more than 10 business days thereafter, subject to lawfully available funds therefor, redeem 100% of the Common Stock sold as part of the Units in the Public Offering (the "Offering Shares"), at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the Trust Account, including interest earned on the funds held in the Trust Account and not previously released to the Company to pay its taxes (less up to \$100,000 of interest to pay dissolution expenses), divided by the number of then outstanding Offering Shares, which redemption will completely extinguish all Public Stockholders' rights as stockholders of the Company (including the right to receive further liquidation distributions, if any), subject to applicable law, and (iii) as promptly as reasonably possible following such redemption, subject to the approval of the Company's remaining stockholders and the Company's board of directors, dissolve and liquidate, subject in each case to the Company's obligations under Delaware law to provide for claims of creditors and other requirements of applicable law. The Sponsor and each Insider agree not to propose any amendment to the Company's amended and restated certificate of incorporation that would modify (i) the substance or timing of the Company's obligation to redeem 100% of the Offering Shares if the Company does not complete a Business Combination within 12 months from the closing of the Public Offering (or up to 18 months if the Company extended such period in accordance with the Company's Charter) or (ii) the other provisions relating to stockholders' rights or pre-initial business combination activities, unless the Company provides its Public Stockholders with the opportunity to redeem their Offering Shares upon approval of any such amendment at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the Trust Account, including interest (which interest shall be net of amounts released for payment of taxes) divided by the number of then outstanding Offering Shares. The Sponsor and each Insider agree to waive its redemption rights with respect to shares of Capital Stock owned by it in connection with a stockholder vote to approve an amendment to the Company's amended and restated certificate of incorporation (A) to modify the substance or timing of the Company's obligation to redeem 100% of the Offering Shares if the Company does not complete a Business Combination within 12 months from the closing of the Public Offering (or up to 18 months if the Company extended such period in accordance with the Company's Charter) or (B) with respect to any other provision relating to stockholders' rights or pre-initial business combination activity.

Each of the Sponsor and each Insider acknowledges that it, he or she has no right, title, interest or claim of any kind in or to any monies held in the Trust Account or any other asset of the Company as a result of any liquidation of the Company with respect to the Founder Shares held by it, him or her. The Sponsor and each Insider hereby further waives, with respect to any shares of Common Stock held by it, him or her, if any, whether acquired now or hereafter, any redemption rights it, he or she may have in connection with the consummation of a Business Combination, including, without limitation, any such rights available in the context of a stockholder vote to approve such Business Combination or a stockholder vote to approve an amendment to the amended and restated certificate of incorporation to modify (i) (A) the substance or timing of the Company's obligation to redeem 100% of the Offering Shares if the Company has not consummated a Business Combination within the time period set forth in the amended and restated certificate of incorporation or (B) any other provisions relating to stockholders' rights or pre-initial Business Combination activity or (ii) in the context of a tender offer made by the Company to purchase shares of Common Stock (although the Sponsor, the Insiders and their respective affiliates shall be entitled to redemption and liquidation rights with respect to any Offering Shares it or they hold if the Company fails to consummate a Business Combination within 12 months from the date of the closing of the Public Offering (or up to 18 months if the Company extended such period in accordance with the Company's Charter)).

3. During the period commencing on the effective date of the Underwriting Agreement and ending 180 days after such date, the Sponsor and each Insider shall not, without the prior written consent of the Representative, (i) sell, offer to sell, contract or agree to sell, hypothecate, pledge, grant any option to purchase or otherwise dispose of or agree to dispose of, directly or indirectly, or establish or increase a put equivalent position or liquidate or decrease a call equivalent position within the meaning of Section 16 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and the rules and regulations of the Commission promulgated thereunder, with respect to any Units, shares of Capital Stock, Warrants or any securities convertible into, or exercisable, or exchangeable for, shares of Capital Stock owned by it, him or her, (ii) enter into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any Units, shares of Capital Stock, Warrants or any securities convertible into, or exercisable, or exchangeable for, shares of Capital Stock owned by it, him or her, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (iii) publicly announce any intention to effect any transaction specified in clause (i) or (ii). Each of the Insiders and the Sponsor acknowledges and agrees that, prior to the effective date of any release or waiver, of the restrictions set forth in this paragraph 3 or paragraph 7 below, the Company shall announce the impending release or waiver by press release through a major news service at least two business days before the effective date of the release or waiver. Any release or waiver granted shall only be effective two business days after the publication date of such press release. The provisions of this paragraph will not apply if the release or waiver is effected solely to permit a transfer not for consideration and the transfere has agreed in writing to be

4. In the event of the liquidation of the Trust Account upon the failure of the Company to consummate its initial Business Combination within the time period set forth in the amended and restated certificate of incorporation, as amended, the Sponsor (which for purposes of clarification shall not extend to any other stockholders, members or managers of the Sponsor) agrees to indemnify and hold harmless the Company against any and all loss, liability, claim, damage and expense whatsoever (including, but not limited to, any and all legal or other expenses reasonably incurred in investigating, preparing or defending against any litigation, whether pending or threatened, or any claim whatsoever) to which the Company may become subject as a result of any claim by (i) any third party (other than the Company's independent accountants) for services rendered or products sold to the Company or (ii) a prospective target business with which the Company has entered into a letter of intent, confidentiality or other similar agreement for a Business Combination (a "Target"); provided, however, that such indemnification of the Company by the Sponsor shall apply only to the extent necessary to ensure that such claims by a third party for services rendered (other than the Company's independent public accountants) or products sold to the Company or a Target do not reduce the amount of funds in the Trust Account to below (i) \$10.20 per share of the Offering Shares or (ii) such lesser amount per share of the Offering Shares held in the Trust Account due to reductions in the value of the trust assets as of the date of the liquidation of the Trust Account, in each case, net of the amount of interest earned on the property in the Trust Account which may be withdrawn to pay taxes, except as to any claims by a third party (including a Target) who executed a waiver of any and all rights to seek access to the Trust Account and except as to any claims under the Company's indemnity of the Underwriters against certain liabilities, including liabilities under the Securities Act. In the event that any such executed waiver is deemed to be unenforceable against such third party, the Sponsor shall not be responsible to the extent of any liability for such third-party claims. The Sponsor shall have the right to defend against any such claim with counsel of its choice reasonably satisfactory to the Company if, within 15 days following written receipt of notice of the claim to the Sponsor, the Sponsor notifies the Company in writing that it shall undertake such defense.

- 5. To the extent that the Underwriters do not exercise their over-allotment option to purchase up to an additional 1,500,000 Units in full within 45 days from the date of the Prospectus (and as further described in the Prospectus), the Sponsor agrees to forfeit, at no cost, a number of Founder Shares in the aggregate equal to 375,000 multiplied by a fraction, (i) the numerator of which is 1,500,000 minus the number of Units purchased by the Underwriters upon the exercise of their over-allotment option, and (ii) the denominator of which is 1,500,000. The Sponsor will be required to forfeit only that number of Founder Shares as is necessary so that the Initial Stockholders will own an aggregate of 20.0% of the Company's issued and outstanding shares of Capital Stock after the Public Offering.
- 6. The Sponsor and each Insider hereby agrees and acknowledges that: (i) the Underwriters and the Company would be irreparably injured in the event of a breach by such Sponsor or an Insider of its, his or her obligations under paragraphs 1, 2, 3, 4, 5, 6, 7(a), 7(b), and 9 of this Letter Agreement (ii) monetary damages may not be an adequate remedy for such breach and (iii) the non-breaching party shall be entitled to injunctive relief, in addition to any other remedy that such party may have in law or in equity, in the event of such breach.
- 7. (a) The Sponsor and each Insider agrees that it or he shall not Transfer any Founder Shares (or shares of Common Stock issuable upon conversion thereof) until the earlier of (A) one year after the date of the Company's initial Business Combination or (B) subsequent to the Company's initial Business Combination, (x) if the reported last sale price of the Common Stock equals or exceeds \$12.00 per share (as adjusted for stock splits, stock dividends, right issuances, reorganizations, recapitalizations and the like) for any 20 trading days within any 30-trading day period commencing at least 150 days after the Company's initial Business Combination, or (y) the date on which the Company completes a liquidation, merger, capital stock exchange, reorganization or other similar transaction that results in all of our stockholders having the right to exchange their shares of common stock for cash, securities or other property (the "Founder Shares Lock-up Period").
- (b) The Sponsor and each Insider agrees that it, he or she shall not Transfer any Private Placement Warrants or shares of Common Stock issued or issuable upon the exercise of the Private Placement Warrants, until 30 days after the completion of the initial Business Combination (the "Private Placement Lock-up Period"), together with the Founder Shares Lock-up Period, the "Lock-up Periods").
- (c) Notwithstanding the provisions set forth in paragraphs 7(a) and 7(b), Transfers of the Founder Shares, Private Placement Warrants and shares of Common Stock issued or issuable upon the exercise or conversion of the Private Placement Warrants or the Founder Shares that are held by the Sponsor, any Insider or any of their permitted transferees (that have complied with this paragraph 7(c)), are permitted (i) to the Company's officers or directors, any affiliate or family member of any of the Company's officers or directors or any members of the Sponsor or any affiliates of the Sponsor; (b) in the case of an individual, by gift to a member of such individual's immediate family or to a trust, the beneficiary of which is a member of such individual's immediate family, an affiliate of such individual or to a charitable organization; (c) in the case of an individual, by virtue of laws of descent and distribution upon death of such individual; (d) in the case of an individual, pursuant to a qualified domestic relations order; (e) by private sales or transfers made in connection with the consummation of an initial Business Combination at prices no greater than the price at which the securities were originally purchased; (f) in the event of the Company's liquidation prior to the completion of an initial Business Combination; (g) by virtue of the laws of the State of Delaware or the Sponsor's limited liability company agreement upon dissolution of the Sponsor; or (h) in the event of the Company's liquidation, merger, capital stock exchange, reorganization or other similar transaction which results in all of the Company's stockholders having the right to exchange their shares of Common Stock for cash, securities or other property subsequent to the Company's completion of an initial Business Combination; provided, however, that in the case of clauses (a) through (e) or (g), these permitted transferees must enter into a written agreement with the Company agreeing to be bound by the transfer restrictions herein and the oth

- 8. Each of the Sponsor and the Insiders represents and warrants that it, he or she has never been suspended or expelled from membership in any securities or commodities exchange or association or had a securities or commodities license or registration denied, suspended or revoked. Each Insider's biographical information furnished to the Company (including any such information included in the Prospectus) is true and accurate in all respects and does not omit any material information with respect to the Insider's background. The Sponsor and each Insider's questionnaire furnished to the Company is true and accurate in all respects. The Sponsor and each Insider represents and warrants that: it, he or she is not subject to or a respondent in any legal action for, any injunction, cease-and-desist order or order or stipulation to desist or refrain from any act or practice relating to the offering of securities in any jurisdiction; it, he or she has never been convicted of, or pleaded guilty to, any crime (i) involving fraud, (ii) relating to any financial transaction or handling of funds of another person, or (iii) pertaining to any dealings in any securities and it, he or she is not currently a defendant in any such criminal proceeding. The Company represents and warrants that, to its knowledge, (i) none of its advisors has been suspended or expelled from membership in any securities or commodities exchange or association or had a securities or commodities license or registration denied, suspended or revoked, (ii) each advisor's biographical information furnished to the Company (including any such information included in the Prospectus) is true and accurate in all respects and does not omit any material information with respect to such advisor's background and each advisor's questionnaire furnished to the Company is true and accurate in all respects, (iii) none of its advisors is subject to or a respondent in any legal action for, any injunction, cease-and-desist order or order or stipulation to desist or refrain from any act or practice relating to the offering of securities in any jurisdiction; and (iii) none of its advisors has been convicted of, or pleaded guilty to, any crime (x) involving fraud, (y) relating to any financial transaction or handling of funds of another person, or (z) pertaining to any dealings in any securities and none of its advisors is currently a defendant in any such criminal proceeding.
- 9. (a) Except as disclosed in the Prospectus and cash or other compensation to the Company's officers or advisors to be engaged subsequent to the consummation of the Public Offering (which will be disclosed in the Company's other filings with the Securities and Exchange Commission), neither the Sponsor nor any individual who is an officer, director or advisor of the Company as of the date hereof nor any affiliate thereof shall receive from the Company any finder's fee, reimbursement, consulting fee, monies in respect of any repayment of a loan or other compensation prior to, or in connection with any services rendered in order to effectuate the consummation of the Company's initial Business Combination (regardless of the type of transaction that it is), other than the following, none of which will be made from the proceeds held in the Trust Account prior to the completion of the initial Business Combination: repayment of a loan and advances up to an aggregate of \$300,000 made to the Company by Aesther Healthcare Sponsor, LLC; reimbursement for any out-of-pocket expenses related to identifying, investigating and consummating an initial Business Combination; payment to an affiliate of the Sponsor of \$10,000 per month, for up to 18 months, for office space, utilities and secretarial and administrative support; and repayment of non-interest bearing loans, if any, and on such terms as to be determined by the Company from time to time, made by the Sponsor or any of the Company's officers or directors to finance transaction costs in connection with an intended initial Business Combination, provided, that, if the Company does not consummate an initial Business Combination, a portion of the working capital held outside the Trust Account may be used by the Company to repay such loaned amounts so long as no proceeds from the Trust Account are used for such repayment. Up to \$1,500,000 of such loans may be convertible into warrants, at a price of \$1.00 per warrant, at the option of the lender, upon consummation of the initi
- 10. Each of the Sponsor and each Insider has full right and power, without violating any agreement to which it is bound (including, without limitation, any non-competition or non-solicitation agreement with any employer or former employer), to enter into this Letter Agreement and, as applicable, to serve as an officer and/or a director on the board of directors of the Company and hereby consents to being named in the Prospectus as an officer and/or a director of the Company.
- 11. As used herein, (i) "Business Combination" shall mean a merger, capital stock exchange, asset acquisition, stock purchase, reorganization or similar business combination, involving the Company and one or more businesses; (ii) "Capital Stock" shall mean, collectively, the Common Stock and the Founder Shares; (iii) "Founder Shares" shall mean the 2,875,000 shares of the Company's Class B common stock, par value \$0.0001 per share, initially held by the Sponsor (up to 375,000 Shares of which are subject to complete or partial forfeiture by the Sponsor if the over-allotment option is not exercised in full by the Underwriters); (iv) "Initial Stockholders" shall mean the Sponsor and any other holder of Founder Shares immediately prior to the Public Offering; (v) "Private Placement Warrants" shall mean the 5,261,000 warrants to purchase shares of Common Stock comprising the Private Placement Warrants (or up to 5,711,000 warrants to purchase shares of Common Stock if the over-allotment option is exercised in full); (vi) "Public Stockholders" shall mean the holders of securities issued in the Public Offering; (vii) "Trust Account" shall mean the trust fund into which a portion of the net proceeds of the Public Offering and the sale of the Private Placement Units shall be deposited; and (viii) "Transfer" shall mean the (a) sale or assignment of, offer to sell, contract or agreement to sell, hypothecate, pledge, grant of any option to purchase or otherwise dispose of or agreement to dispose of, directly or indirectly, or establishment or increase of a put equivalent position or liquidation with respect to or decrease of a call equivalent position within the meaning of Section 16 of the Securities Exchange Act of 1934, as amended, and the rules and regulations of the Commission promulgated thereunder with respect to, any security, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (c) public announcement of any intention to effect any transaction specified in clause (a

- 12. This Letter Agreement constitutes the entire agreement and understanding of the parties hereto in respect of the subject matter hereof and supersedes all prior understandings, agreements, or representations by or among the parties hereto, written or oral, to the extent they relate in any way to the subject matter hereof or the transactions contemplated hereby. This Letter Agreement may not be changed, amended, modified or waived (other than to correct a typographical error) as to any particular provision, except by a written instrument executed by all parties hereto.
- 13. No party hereto may assign either this Letter Agreement or any of its rights, interests, or obligations hereunder without the prior written consent of the other parties. Any purported assignment in violation of this paragraph shall be void and ineffectual and shall not operate to transfer or assign any interest or title to the purported assignee. This Letter Agreement shall be binding on the Sponsor and each Insider and their respective successors, heirs and assigns and permitted transferees.
- 14. Nothing in this Letter Agreement shall be construed to confer upon, or give to, any person or corporation other than the parties hereto any right, remedy or claim under or by reason of this Letter Agreement or of any covenant, condition, stipulation, promise or agreement hereof. All covenants, conditions, stipulations, promises and agreements contained in this Letter Agreement shall be for the sole and exclusive benefit of the parties hereto and their successors, heirs, personal representatives and assigns and permitted transferees.
- 15. This Letter Agreement may be executed in any number of original or facsimile counterparts and each of such counterparts shall for all purposes be deemed to be an original, and all such counterparts shall together constitute but one and the same instrument.
- 16. This Letter Agreement shall be deemed severable, and the invalidity or unenforceability of any term or provision hereof shall not affect the validity or enforceability of this Letter Agreement or of any other term or provision hereof. Furthermore, in lieu of any such invalid or unenforceable term or provision, the parties hereto intend that there shall be added as a part of this Letter Agreement a provision as similar in terms to such invalid or unenforceable provision as may be possible and be valid and enforceable.
- 17. This Letter Agreement shall be governed by and construed and enforced in accordance with the laws of the State of New York, without giving effect to conflicts of law principles that would result in the application of the substantive laws of another jurisdiction. The parties hereto (i) all agree that any action, proceeding, claim or dispute arising out of, or relating in any way to, this Letter Agreement shall be brought and enforced in the courts of New York City, in the State of New York, and irrevocably submit to such jurisdiction and venue, which jurisdiction and venue shall be exclusive and (ii) waive any objection to such exclusive jurisdiction and venue or that such courts represent an inconvenient forum.
- 18. Any notice, consent or request to be given in connection with any of the terms or provisions of this Letter Agreement shall be in writing and shall be sent by express mail or similar private courier service, by certified mail (return receipt requested), by hand delivery or facsimile transmission.
- 19. This Letter Agreement shall terminate on the earlier of (i) the expiration of the Lock-up Periods or (ii) the liquidation of the Company; provided, however, that this Letter Agreement shall earlier terminate in the event that the Public Offering is not consummated and closed by [______] 31, 2021; provided further that paragraph 4 of this Letter Agreement shall survive such liquidation.

·	
Aesther Healthcare Acquisition Corp.	
Ву:	
Name:Suren Ajjarapu Title: Chief Executive Officer	
Name: Suren Ajjarapu	
Name: Howard Doss	
Name: Michael L. Peterson	
Name: Venkatesh Srinivasan	
Name: Donald G. Fell	
Name: Siva Saravanan	
Aesther Healthcare Sponsor, LLC	
By:	
By: Suren Ajjarapu, Chief Executive Officer	
[Signature Page to Letter Agreement]	
7	

Sincerely,

THIS NOTE HAS NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"). THIS NOTE HAS BEEN ACQUIRED FOR INVESTMENT ONLY AND MAY NOT BE SOLD, TRANSFERRED OR ASSIGNED IN THE ABSENCE OF REGISTRATION OF THE RESALE THEREOF UNDER THE SECURITIES ACT OR AN OPINION OF COUNSEL REASONABLY SATISFACTORY IN FORM, SCOPE AND SUBSTANCE TO THE COMPANY THAT SUCH REGISTRATION IS NOT REQUIRED.

PROMISSORY NOTE

Principal Amount: Up to \$300,000.00 Dated as of June 30, 2021

Aesther Healthcare Acquisition Corp., a Delaware corporation (the "**Maker**"), promises to pay to the order of **Aesther Healthcare Sponsor, LLC** or its registered assigns or successors in interest (the "**Payee**") the principal sum of up to Three Hundred Thousand Dollars (\$300,000.00) in lawful money of the United States of America, on the terms and conditions described below. All payments on this Note shall be made by check or wire transfer of immediately available funds or as otherwise determined by the Maker to such account as the Payee may from time to time designate by written notice in accordance with the provisions of this Note.

- 1. **Principal.** The principal balance of this Promissory Note (this "**Note**") shall be payable on the earlier of: (i) June 30, 2022 or (ii) on the date on which the Maker consummates an initial public offering of its securities. The principal balance may be prepaid at any time. Under no circumstances shall any individual, including but not limited to any officer, director, employee or shareholder of the Maker, be obligated personally for any obligations or liabilities of the Maker hereunder.
- 2. **Drawdown Requests**. Maker and Payee agree that Maker may request up to Three Hundred Thousand Dollars (\$300,000.00) for costs reasonably related to Maker's initial public offering of its securities. The principal of this Note may be drawn down from time to time prior to the earlier of: (i) June 30, 2022 or (ii) the date on which Maker consummates an initial public offering of its securities, upon written request from Maker to Payee (each, a "**Drawdown Request**"). Each Drawdown Request must state the amount to be drawn down and Payee shall fund each Drawdown Request no later than five (5) business days after receipt of a Drawdown Request; provided, however, that the maximum amount of drawdowns collectively under this Note is Three Hundred Thousand Dollars (\$300,000.00). Once an amount is drawn down under this Note, it shall not be available for future Drawdown Requests even if prepaid. No fees, payments or other amounts shall be due to Payee in connection with, or as a result of, any Drawdown Request by Maker. Notwithstanding the foregoing, all payments shall be applied first to payment in full of any costs incurred in the collection of any sum due under this Note, including (without limitation) reasonable attorneys' fees, and then to the reduction of the unpaid principal balance of this Note.
- 3. **Interest.** No interest shall accrue on the unpaid principal balance of this Note.
- 4. **Application of Payments.** All payments shall be applied first to payment in full of any costs incurred in the collection of any sum due under this Note, including (without limitation) reasonable attorney's fees, then to the payment in full of any late charges and finally to the reduction of the unpaid principal balance of this Note.

- 5. **Events of Default.** The following shall constitute an event of default ("**Event of Default**"):
 - (a) **Failure to Make Required Payments.** Failure by Maker to pay the principal of this Note within five (5) business days following the date when due.
 - (b) **Voluntary Liquidation, Etc.** The commencement by Maker of a proceeding relating to its bankruptcy, insolvency, reorganization, rehabilitation or other similar action, or the consent by it to the appointment of, or taking possession by, a receiver, liquidator, assignee, trustee, custodian, sequestrator (or other similar official) for Maker or for any substantial part of its property, or the making by it of any assignment for the benefit of creditors, or the failure of Maker generally to pay its debts as such debts become due, or the taking of corporate action by Maker in furtherance of any of the foregoing.
 - (c) **Involuntary Bankruptcy, Etc.** The entry of a decree or order for relief by a court having jurisdiction in the premises in respect of maker in an involuntary case under any applicable bankruptcy, insolvency or similar law, for the appointing of a receiver, liquidator, assignee, custodian, trustee, sequestrator (or similar official) for Maker or for any substantial part of its property, or ordering the winding-up or liquidation of the affairs of Maker, and the continuance of any such decree or order unstayed and in effect for a period of 60 consecutive days.

6. Remedies.

- (a) Upon the occurrence of an Event of Default specified in Section 5(a) hereof, Payee may, by written notice to Maker, declare this Note to be due immediately and payable, whereupon the unpaid principal amount of this Note, and all other amounts payable hereunder, shall become immediately due and payable without presentment, demand, protest or other notice of any kind, all of which are hereby expressly waived, anything contained herein or in the documents evidencing the same to the contrary notwithstanding.
- (b) Upon the occurrence of an Event of Default specified in Sections 5(b) and 5(c), the unpaid principal balance of this Note, and all other sums payable with regard to this Note, shall automatically and immediately become due and payable, in all cases without any action on the part of Payee.
- Waivers. Maker and all endorsers and guarantors of, and sureties for, this Note waive presentment for payment, demand, notice of dishonor, protest, and notice of protest with regard to the Note, all errors, defects and imperfections in any proceedings instituted by Payee under the terms of this Note, and all benefits that might accrue to Maker by virtue of any present or future laws exempting any property, real or personal, or any part of the proceeds arising from any sale of any such property, from attachment, levy or sale under execution, or providing for any stay of execution, exemption from civil process, or extension of time for payment; and Maker agrees that any real estate that may be levied upon pursuant to a judgment obtained by virtue hereof, or any writ of execution issued hereon, may be sold upon any such writ in whole or in part in any order desired by Payee.
- 8. **Unconditional Liability.** Maker hereby waives all notices in connection with the delivery, acceptance, performance, default, or enforcement of the payment of this Note, and agrees that its liability shall be unconditional, without regard to the liability of any other party, and shall not be affected in any manner by any indulgence, extension of time, renewal, waiver or modification granted or consented to by Payee, and consents to any and all extensions of time, renewals, waivers, or modifications that may be granted by Payee with respect to the payment or other provisions of this Note, and agrees that additional makers, endorsers, guarantors, or sureties may become parties hereto without notice to Maker or affecting Maker's liability hereunder.

- 9. **Notices.** Any notice called for hereunder shall be deemed properly given if (i) sent by certified mail, return receipt requested, (ii) personally delivered, (iii) dispatched by any form of private or governmental express mail or delivery service providing receipted delivery or (iv) sent by facsimile or (v) to such address as either party may designate to the other party.
 - Notice shall be deemed given on the earlier of (i) actual receipt by the receiving party, (ii) the date shown on a facsimile transmission confirmation, (iii) the date reflected on a signed delivery receipt, or (iv) two (2) Business Days following tender of delivery or dispatch by express mail or delivery service.
- 10. **Construction.** THIS NOTE SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH THE LAWS OF NEW YORK, WITHOUT REGARD TO CONFLICT OF LAW PROVISIONS THEREOF.
- 11. **Jurisdiction.** The courts of New York have exclusive jurisdiction to settle any dispute arising out of or in connection with this agreement (including a dispute relating to any non-contractual obligations arising out of or in connection with this agreement) and the parties submit to the exclusive jurisdiction of the courts of New York.
- 12. **Severability.** Any provision contained in this Note which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.
- 13. **Trust Waiver.** Notwithstanding anything herein to the contrary, the Payee hereby waives any and all right, title, interest or claim of any kind ("Claim") in or to any amounts contained in the trust account in which the proceeds of the initial public offering (the "IPO") conducted by the Maker and the proceeds of the sale of securities in a private placement to occur prior to the effectiveness of the IPO, as described in greater detail in the registration statement and prospectus to be filed with the Securities and Exchange Commission in connection with the IPO, will be placed, and hereby agrees not to seek recourse, reimbursement, payment or satisfaction for any Claim from the trust account or any distribution therefrom for any reason whatsoever.
- 14. **Amendment; Waiver.** Any amendment hereto or waiver of any provision hereof may be made with, and only with, the written consent of the Maker and the Payee.
- 15. **Assignment.** No assignment or transfer of this Note or any rights or obligations hereunder may be made by any party hereto (by operation of law or otherwise) without the prior written consent of the other party hereto and any attempted assignment without the required consent shall be void.
- 16. **Further Assurance.** The Maker shall, at its own cost and expense, execute and do (or procure to be executed and done by any other necessary party) all such deeds, documents, acts and things as the Payee may from time to time require as may be necessary to give full effect to this Promissory Note.

IN WITNESS WHEREOF, Maker, intending to be legally bound hereby, has caused this Note to be duly executed on the day and year first above written.

AESTHER HEALTHCARE ACQUISITION CORP.

By: <u>/s/ Suren Ajjarapu</u>

Name: Suren Ajjarapu Title: Chief Executive Officer

[Signature Page to Promissory Note]

INVESTMENT MANAGEMENT TRUST AGREEMENT

This Investment Management Trust Agreement (this "Agreement") is made effective as of [], 2021 by and between Aesther Healthcare Acquisition Corp., a Delaware corporation (the "Company"), and Continental Stock Transfer & Trust Company, LLC, a New York corporation (the "Trustee").

WHEREAS, the Company's registration statement on Form S-1, No. 333-258012 (the "*Registration Statement*") and prospectus (the "*Prospectus*") for the initial public offering of the Company's units (the "*Units*"), each of which consists of one share of the Company's Class A common stock, par value \$0.0001 per share (the "*Common Stock*") and one-half of one redeemable warrant, with each warrant entitling the holder thereof to purchase one share of Common Stock (such initial public offering hereinafter referred to as the "*Offering*"), has been declared effective as of the date hereof by the U.S. Securities and Exchange Commission; and

WHEREAS, the Company has entered into an Underwriting Agreement (the "*Underwriting Agreement*") with EF Hutton, division of Benchmark Investments, LLC, as representative (the "*Representative*") of the several underwriters (the "*Underwriters*"); and

WHEREAS, as described in the Prospectus, \$102,000,000 of the gross proceeds of the Offering and sale of the Private Placement Units (as defined in the Underwriting Agreement) (or \$117,300,000 if the Underwriters' over-allotment option is exercised in full), plus any amounts to be deposited in connection with the Extended Terms (as defined below), will be delivered to the Trustee to be deposited and held in a segregated trust account located at all times in the United States (the "*Trust Account*") for the benefit of the Company and the holders of the Common Stock included in the Units issued in the Offering as hereinafter provided (the amount to be delivered to the Trustee (and any interest subsequently earned thereon) is referred to herein as the "*Initial Property*," the stockholders for whose benefit the Trustee shall hold the Property will be referred to as the "*Public Stockholders*," and the Public Stockholders and the Company will be referred to together as the "*Beneficiaries*"); and

WHEREAS, as described in the Prospectus, the Company will have twelve (12) months from the effective date of the Registration Statement (the "Initial Term") to complete an initial business combination (the "Business Combination"), with the option to extend the Initial Term for two (2) additional three-month terms (the first such extension, the "First Extended Term" and the second such extension, the "Second Extended Term" and, together, the "Extended Terms"), each Extended Term requiring the Company to deposit an additional amount of \$0.10 per Unit into the Trust Account such that (i) if the Company elects the First Extended Term, it will deliver to the Trustee an amount equal to \$1,000,000 (or \$1,150,000 if the Underwriters' overallotment option is exercised in full) (such additional amount, the "First Extension Property"), increasing the Initial Amount of gross proceeds held in the Trust Account to \$103,000,000 (or \$118,450,000 if the Underwriter's overallotment option is exercised in full) (such increased amount, the "Second Extension Property" and together with the Initial Property and the First Extension Property, the "Property"), increasing the gross proceeds held in the Trust Account to \$104,000,000 (or \$119,600,000 if the Underwriter's overallotment option is exercised in full) for the Second Extended Term; and

WHEREAS, the Company and the Trustee desire to enter into this Agreement to set forth the terms and conditions pursuant to which the Trustee shall hold the Property.

NOW THEREFORE, IT IS AGREED:

- 1. <u>Agreements and Covenants of Trustee</u>. The Trustee hereby agrees and covenants to:
- (a) Hold the Property in trust for the Beneficiaries in accordance with the terms of this Agreement in the Trust Account established by the Trustee in the United States at J.P. Morgan Chase Bank, N.A. (or at another U.S. chartered commercial bank with consolidated assets of \$100 billion or more) and at a brokerage institution selected by the Trustee that is reasonably satisfactory to the Company;

- (b) Manage, supervise and administer the Trust Account subject to the terms and conditions set forth herein;
- (c) In a timely manner, upon the written instruction of the Company, invest and reinvest the Property in United States government securities within the meaning of Section 2(a)(16) of the Investment Company Act of 1940, as amended, having a maturity of 185 days or less, or in money market funds meeting the conditions of paragraphs (d)(1), (d)(2), (d)(3) and (d)(4) of Rule 2a-7 promulgated under the Investment Company Act of 1940, as amended (or any successor rule), which invest only in direct U.S. government treasury obligations, as determined by the Company; the Trustee may not invest in any other securities or assets, it being understood that the Trust Account will earn no interest while account funds are uninvested awaiting the Company's instructions hereunder; while account funds are invested or uninvested, the Trustee may earn bank credits or other consideration;
- (d) Collect and receive, when due, all interest or other income arising from the Property, which shall become part of the "Property," as such term is used herein;
- (e) Promptly notify the Company and the Representative of all communications received by the Trustee with respect to any Property requiring action by the Company;
- (f) Supply any necessary information or documents as may be requested by the Company (or its authorized agents) in connection with the Company's preparation of the tax returns relating to assets held in the Trust Account;
- (g) Participate in any plan or proceeding for protecting or enforcing any right or interest arising from the Property if, as and when instructed by the Company to do so;
- (h) Render to the Company monthly written statements of the activities of, and amounts in, the Trust Account reflecting all receipts and disbursements of the Trust Account;
- (i) Commence liquidation of the Trust Account only after and promptly after (x) receipt of, and only in accordance with, the terms of a letter from the Company ("*Termination Letter*") in a form substantially similar to that attached hereto as either Exhibit A or Exhibit B, as applicable, signed on behalf of the Company by its Chief Executive Officer, Chief Financial Officer, Secretary or Chairman of the board of directors of the Company (the "Board") or other authorized officer of the Company, and, in the case of Exhibit A, acknowledged and agreed to by the Representative, and complete the liquidation of the Trust Account and distribute the Property in the Trust Account, including interest not previously released to the Company to pay its taxes, only as directed in the Termination Letter and the other documents referred to therein, or (y) upon the date which is the later of (i) expiration of the Initial Term, (ii) the expiration of the First Extended Term, (iii) the expiration of the Second Extended Term and (iv) such later date as may be approved by the Company's stockholders in accordance with the Company's amended and restated certificate of incorporation, if a Termination Letter has not been received by the Trustee prior to such date, in which case the Trust Account, including interest not previously released to the Company to pay its taxes (less up to \$100,000 of interest that may be released to the Company to pay dissolution expenses) shall be distributed to the Public Stockholders of record as of such date; provided, however, that in the event the Trustee receives a Termination Letter in a form substantially similar to Exhibit B hereto, or if the Trustee begins to liquidate the Property because it has received no such Termination Letter by the date specified in clause (y) of this Section 1(i), the Trustee shall keep the Trust Account open until twelve (12) months following the date the Property has been distributed to the Public Stockholders;
- (j) Upon written request from the Company, which may be given from time to time in a form substantially similar to that attached hereto as Exhibit C, withdraw from the Trust Account and distribute to the Company the amount of interest earned on the Property requested by the Company to cover any tax obligation owed by the Company as a result of assets of the Company or interest or other income earned on the Property, which amount shall be delivered directly to the Company by electronic funds transfer or other method of prompt payment, and the Company shall forward such payment to the relevant taxing authority; provided, however, that to the extent there is not sufficient cash in the Trust Account to pay such tax obligation, the Trustee shall liquidate such assets held in the Trust Account as shall be designated by the Company in writing to make such distribution so long as there is no reduction in the principal amount per share initially deposited in the Trust Account; provided, further, that if the tax to be paid is a franchise tax, the written request by the Company to make such distribution shall be accompanied by a copy of the franchise tax bill from the State of Delaware for the Company and a written statement from the principal financial officer of the Company setting forth the actual amount payable. The written request of the Company in the form of Exhibit C referenced above shall constitute presumptive evidence that the Company is entitled to said funds, and the Trustee shall have no responsibility to look beyond said request (it being acknowledged and agreed that any such amount in excess of interest income earned on the Property shall not be payable from the Trust Account);

- (k) Upon receipt of an extension letter ("*Extension Letter*") substantially similar to Exhibit D hereto at least five business days prior to the Applicable Deadline, signed on behalf of the Company by an executive officer, and receipt of the dollar amount specified in the Extension Letter on or prior to the applicable Extended Term, the Trustee shall follow the instructions set forth in the Extension Letter.
- (l) Upon written request from the Company, which may be given from time to time in a form substantially similar to that attached hereto as Exhibit E, the Trustee shall distribute to the Public Stockholders of record as of such date the amount requested by the Company to be used to redeem shares of Common Stock from Public Stockholders properly submitted in connection with a stockholder vote to approve an amendment to the Company's amended and restated certificate of incorporation (a) to modify the substance or timing of the ability of Public Stockholders to seek redemption in connection with an initial Business Combination or the Company's obligation to redeem 100% of the shares of Common Stock included in the Units sold in the Offering if the Company has not consummated an initial Business Combination within such time as is described in the Company's amended and restated certificate of incorporation or (b) with respect to any other provisions relating to stockholders' rights or pre-initial Business Combination activity. The written request of the Company referenced above shall constitute presumptive evidence that the Company is entitled to distribute said funds, and the Trustee shall have no responsibility to look beyond said request; and
 - (m) Not make any withdrawals or distributions from the Trust Account other than pursuant to Section 1(i), (j) or (l) above.
- 2. Agreements and Covenants of the Company. The Company hereby agrees and covenants to:
- (a) Give all instructions to the Trustee hereunder in writing, signed by the Company's Chairman of the Board, Chief Executive Officer, Chief Financial Officer or Secretary. In addition, except with respect to its duties under Sections 1(i), 1(j) and 1(l) hereof, the Trustee shall be entitled to rely on, and shall be protected in relying on, any verbal or telephonic advice or instruction which it, in good faith and with reasonable care, believes to be given by any one of the persons authorized above to give written instructions, provided that the Company shall promptly confirm such instructions in writing;
- (b) Subject to Section 4 hereof, hold the Trustee harmless and indemnify the Trustee from and against any and all expenses, including reasonable counsel fees and disbursements, or losses suffered by the Trustee in connection with any action taken by it hereunder and in connection with any action, suit or other proceeding brought against the Trustee involving any claim, or in connection with any claim or demand, which in any way arises out of or relates to this Agreement, the services of the Trustee hereunder, or the Property or any interest earned on the Property, except for expenses and losses resulting from the Trustee's gross negligence, fraud or willful misconduct. Promptly after the receipt by the Trustee of notice of demand or claim or the commencement of any action, suit or proceeding, pursuant to which the Trustee intends to seek indemnification under this Section 2(b), it shall notify the Company in writing of such claim (hereinafter referred to as the "Indemnified Claim"). The Trustee shall have the right to conduct and manage the defense against such Indemnified Claim; provided that the Trustee shall obtain the consent of the Company with respect to the selection of counsel, which consent shall not be unreasonably withheld or delayed. The Trustee may not agree to settle any Indemnified Claim without the prior written consent of the Company, which such consent shall not be unreasonably withheld or delayed. The Company may participate in such action with its own counsel;

- (c) Pay the Trustee the fees set forth on Schedule A hereto, including an initial acceptance fee, annual administration fee, and transaction processing fee which fees shall be subject to modification by the parties from time to time. It is expressly understood that the Property shall not be used to pay such fees unless and until the consummation of the Business Combination (as defined below). The Company shall pay the Trustee the initial acceptance fee and the first annual administration fee at the consummation of the Offering. The Company shall not be responsible for any other fees or charges of the Trustee except as set forth in this Section 2(c), Schedule A and as may be provided in Section 2(b) hereof;
- (d) In connection with any vote of the Company's stockholders regarding a Business Combination, provide to the Trustee an affidavit or certificate of the inspector of elections for the stockholder meeting verifying the vote of such stockholders regarding such Business Combination;
- (e) Provide the Representative with a copy of any Termination Letter(s) and/or any other correspondence that is sent to the Trustee with respect to any proposed withdrawal from the Trust Account promptly after it issues the same;
- (f) Instruct the Trustee to make only those distributions that are permitted under this Agreement, and refrain from instructing the Trustee to make any distributions that are not permitted under this Agreement;
- (g) Within four (4) business days after the Underwriters exercise the over-allotment option (or any unexercised portion thereof) or such over-allotment option expires, provide the Trustee with a notice in writing of the total amount of the Deferred Discount, which shall in no event be less than \$3,000,000;
- (h) Promptly following an applicable Extended Term, disclose whether or not the term the Company has to consummate a Business Combination has been extended; and
- (i) Unless otherwise agreed between the Company and the Representative, ensure that any Instruction Letter (as defined in Exhibit A) delivered in connection with a Termination Letter in the form of Exhibit A expressly provides that the Deferred Discount is paid directly to the account or accounts directed by the Representative on behalf of the Underwriters prior to any transfer of the funds held in the Trust Account to the Company or any other person.
- 3. <u>Limitations of Liability</u>. The Trustee shall have no responsibility or liability to:
- (a) Imply obligations, perform duties, inquire or otherwise be subject to the provisions of any agreement or document other than this Agreement and that which is expressly set forth herein;
- (b) Take any action with respect to the Property, other than as directed in Section 1 hereof, and the Trustee shall have no liability to any third party except for liability arising out of the Trustee's gross negligence, fraud or willful misconduct;
- (c) Institute any proceeding for the collection of any principal and income arising from, or institute, appear in or defend any proceeding of any kind with respect to, any of the Property unless and until it shall have received instructions from the Company given as provided herein to do so and the Company shall have advanced or guaranteed to it funds sufficient to pay any expenses incident thereto;
 - (d) Refund any depreciation in principal of any Property;
- (e) Assume that the authority of any person designated by the Company to give instructions hereunder shall not be continuing unless provided otherwise in such designation, or unless the Company shall have delivered a written revocation of such authority to the Trustee;
- (f) The other parties hereto or to anyone else for any action taken or omitted by it, or any action suffered by it to be taken or omitted, in good faith and in the Trustee's reasonable best judgment, except for the Trustee's gross negligence, fraud or willful misconduct. The Trustee may rely conclusively and shall be protected in acting upon any order, notice, demand, certificate, opinion or advice of counsel (including counsel chosen by the Trustee, which counsel may be the Company's counsel), statement, instrument, report or other paper or document (not only as to its due execution and the validity and effectiveness of its provisions, but also as to the truth and acceptability of any information therein contained) which the Trustee believes, in good faith and with reasonable care, to be genuine and to be signed or presented by the proper person or persons. The Trustee shall not be bound by any notice or demand, or any waiver, modification, termination or rescission of this Agreement or any of the terms hereof, unless evidenced by a written instrument delivered to the Trustee, signed by the proper party or parties and, if the duties or rights of the Trustee are affected, unless it shall give its prior written consent thereto;

- (g) Verify the accuracy of the information contained in the Registration Statement;
- (h) Provide any assurance that any Business Combination entered into by the Company or any other action taken by the Company is as contemplated by the Registration Statement;
- (i) File information returns with respect to the Trust Account with any local, state or federal taxing authority or provide periodic written statements to the Company documenting the taxes payable by the Company, if any, relating to any interest income earned on the Property;
- (j) Prepare, execute and file tax reports, income or other tax returns and pay any taxes with respect to any income generated by, and activities relating to, the Trust Account, regardless of whether such tax is payable by the Trust Account or the Company, including, but not limited to, franchise and income tax obligations, except pursuant to Section 1(j) hereof; or
- (k) Verify calculations, qualify or otherwise approve the Company's written requests for distributions pursuant to Sections 1(i), 1(j) and 1(l) hereof.
- 4. <u>Trust Account Waiver</u>. The Trustee has no right of set-off or any right, title, interest or claim of any kind ("*Claim*") to, or to any monies in, the Trust Account, and hereby irrevocably waives any Claim to, or to any monies in, the Trust Account that it may have now or in the future. In the event the Trustee has any Claim against the Company under this Agreement, including, without limitation, under Section 2(b) or Section 2(c) hereof, the Trustee shall pursue such Claim solely against the Company and its assets outside the Trust Account and not against the Property or any monies in the Trust Account.
- 5. <u>Termination</u>. This Agreement shall terminate as follows:
- (a) If the Trustee gives written notice to the Company that it desires to resign under this Agreement, the Company shall use its reasonable efforts to locate a successor trustee, pending which the Trustee shall continue to act in accordance with this Agreement. At such time that the Company notifies the Trustee that a successor trustee has been appointed and has agreed to become subject to the terms of this Agreement, the Trustee shall transfer the management of the Trust Account to the successor trustee, including but not limited to the transfer of copies of the reports and statements relating to the Trust Account, whereupon this Agreement shall terminate; provided, however, that in the event that the Company does not locate a successor trustee within ninety (90) days of receipt of the resignation notice from the Trustee, the Trustee may submit an application to have the Property deposited with any court in the State of New York or with the United States District Court for the Southern District of New York and upon such deposit, the Trustee shall be immune from any liability whatsoever; or
- (b) At such time that the Trustee has completed the liquidation of the Trust Account and its obligations in accordance with the provisions of Section 1(i) hereof and distributed the Property in accordance with the provisions of the Termination Letter, this Agreement shall terminate except with respect to Section 2(b).

6. Miscellaneous.

(a) The Company and the Trustee each acknowledge that the Trustee will follow the security procedures set forth below with respect to funds transferred from the Trust Account. The Company and the Trustee will each restrict access to confidential information relating to such security procedures to authorized persons. Each party must notify the other party immediately if it has reason to believe unauthorized persons may have obtained access to such confidential information, or of any change in its authorized personnel. In executing funds transfers, the Trustee shall rely upon all information supplied to it by the Company, including, account names, account numbers, and all other identifying information relating to a Beneficiary, Beneficiary's bank or intermediary bank. Except for any liability arising out of the Trustee's gross negligence, fraud or willful misconduct, the Trustee shall not be liable for any loss, liability or expense resulting from any error in the information or transmission of the funds.

- (b) This Agreement shall be governed by and construed and enforced in accordance with the laws of the State of New York, without giving effect to conflicts of law principles that would result in the application of the substantive laws of another jurisdiction. This Agreement may be executed in several original or facsimile counterparts, each one of which shall constitute an original, and together shall constitute but one instrument.
- (c) This Agreement contains the entire agreement and understanding of the parties hereto with respect to the subject matter hereof. This Agreement or any provision hereof may only be changed, amended or modified (other than to correct a typographical error) by a writing signed by each of the parties hereto.
- (d) This Agreement or any provision hereof may only be changed, amended or modified pursuant to Section 6(c) hereof with the Consent of the Stockholders, it being the specific intention of the parties hereto that each of the Company's stockholders is, and shall be, a third party beneficiary of this Section 6(d) with the same right and power to enforce this Section 6(d) as the other parties hereto. For purposes of this Section 6(d), the "Consent of the Stockholders" means receipt by the Trustee of a certificate from the inspector of elections of the stockholder meeting certifying that either (i) the Company's stockholders of record as of a record date established in accordance with Section 213(a) of the Delaware General Corporation Law, as amended ("DGCL") (or any successor rule), who hold sixty-five percent (65%) or more of all then outstanding shares of the Common Stock and Class B common stock, par value \$0.0001 per share, of the Company voting together as a single class, have voted in favor of such change, amendment or modification, or (ii) the Company's stockholders of record as of the record date who hold sixty-five percent (65%) or more of all then outstanding shares of the Common Stock and Class B common stock, par value \$0.0001 per share, of the Company voting together as a single class, have delivered to such entity a signed writing approving such change, amendment or modification. No such amendment will affect any Public Stockholder who has otherwise indicated his election to redeem his shares of Common Stock in connection with a stockholder vote sought to amend this Agreement. Except for any liability arising out of the Trustee's gross negligence, fraud or willful misconduct, the Trustee may rely conclusively on the certification from the inspector or elections referenced above and shall be relieved of all liability to any party for executing the proposed amendment in reliance thereon.
- (e) The parties hereto consent to the jurisdiction and venue of any state or federal court located in the City of New York, State of New York, for purposes of resolving any disputes hereunder. AS TO ANY CLAIM, CROSS-CLAIM OR COUNTERCLAIM IN ANY WAY RELATING TO THIS AGREEMENT, EACH PARTY WAIVES THE RIGHT TO TRIAL BY JURY.
- (f) Any notice, consent or request to be given in connection with any of the terms or provisions of this Agreement shall be in writing and shall be sent by express mail or similar private courier service, by certified mail (return receipt requested), by hand delivery or by facsimile or email transmission:

if to the Trustee, to:

Continental Stock Transfer & Trust Company 1 State Street, 30th Floor New York, NY 10004 Attn: [Francis Wolf and Celeste Gonzalez] Email: [fwolf@continentalstock.com] Email: [cgonzalez@continentalstock.com]

if to the Company, to:

Aesther Healthcare Acquisition Corp. 515 Madison Avenue, Suite 8078 New York, New York 10022 Telephone: (646) 908-2658

Attn.: Mr. Suren Ajjarapu, Chief Executive Officer

in each case, with copies to:

The Loev Law Firm, PC Attn.: David M. Loev, Esq. John S. Gillies, Esq. The Loev Law Firm, PC 6300 West Loop South, Suite 280 Bellaire, TX 77401

Telephone No.: (832) 930-6432

Email: <u>dloev@loevlaw.com</u>; and john@loevlaw.com

and

EF Hutton, division of Benchmark Investments, LLC 590 Madison Avenue, 39th Floor New York, NY 10022 Attn.: Legal Department

and

Cavas Pavri, Esq. Schiff Harden LLP 100 N. 18th Street, Suite 300 Philadelphia, PA 19103 Telephone: (202) 778-6400 Email: CPavri@schiffhardin.com

- (g) Each of the Company and the Trustee hereby represents that it has the full right and power and has been duly authorized to enter into this Agreement and to perform its respective obligations as contemplated hereunder. The Trustee acknowledges and agrees that it shall not make any claims or proceed against the Trust Account, including by way of set-off, and shall not be entitled to any funds in the Trust Account under any circumstance.
- (h) This Agreement is the joint product of the Trustee and the Company and each provision hereof has been subject to the mutual consultation, negotiation and agreement of such parties and shall not be construed for or against any party hereto.
- (i) This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original, but all such counterparts shall together constitute one and the same instrument. Delivery of a signed counterpart of this Agreement by facsimile or electronic transmission shall constitute valid and sufficient delivery thereof.
- (j) Each of the Company and the Trustee hereby acknowledges and agrees that the Representative, on behalf of the Underwriters, is a third-party beneficiary of this Agreement.
- (k) Except as specified herein, no party to this Agreement may assign its rights or delegate its obligations hereunder to any other person or entity.

[Signature Page Follows]

IN WITNESS WHEREOF, the parties have duly executed this Investment Management Trust Agreement as of the date first written above.		
Continental Stock Transfer & Trust Company, as Trustee		
	Ву:	
	Name:	
	Title:	
	Aesther Healthcare Acquisition Corp.	
	Ву:	
	Name: Suren Ajjarapu	
	Title: Chief Executive Officer	

Signature Page to Investment Management Trust Agreement

SCHEDULE A

Fee Item	Time and method of payment	Amoun	t
Initial set-up fee.	Initial closing of Offering by wire transfer.	\$	3,500.00
Trustee administration fee	Payable annually. First year fee payable at initial closing of Offering by wire transfer thereafter by wire transfer or check.	\$	10,000.00
Transaction processing fee for disbursements to Company under Sections 1(i), 1(j) and 1(l)	Billed to Company following disbursement made to Company under Section 1	\$	250.00 per item presented
Desire Agent services as required assumed to	Dilled to Common verse delicery of service recovery to		
Paying Agent services as required pursuant to Section 1(i) and 1(l)	Billed to Company upon delivery of service pursuant to Section 1(i) and 1(l)		Prevailing rates

EXHIBIT A

[Letterhead of Company]

[Insert date]

Continental Stock Transfer & Trust Company 1 State Street, 30th Floor New York, New York 10004 Attn: Francis Wolf and Celeste Gonzalez

Re: Trust Account No. Termination Letter

Dear Mr. Wolf and Ms. Gonzalez:

Pursuant to Section 1(i) of the Investment Management Trust Agreement between Aesther Healthcare Acquisition Corp. (the "Company") and Continental Stock Transfer & Trust Company (the "Trustee"), dated as of [], 2021 (the "Trust Agreement"), this is to advise you that the Company has entered into an agreement with [insert name] (the "Target Business") to consummate a business combination with Target Business (the "Business") Combination") on or about [insert date]. The Company shall notify you at least seventy-two (72) hours in advance of the actual date of the consummation of the Business Combination (or such shorter time period as you may agree) (the "Consummation Date"). Capitalized terms used but not defined herein shall have the meanings set forth in the Trust Agreement.

In accordance with the terms of the Trust Agreement, we hereby authorize you to commence to liquidate all of the assets of the Trust Account and to transfer the proceeds into a segregated account held by you on behalf of the Beneficiaries to the effect that, on the Consummation Date, all of the funds held in the Trust Account will be immediately available for transfer to the account or accounts that the Company shall direct on the Consummation Date (including as directed to it by the Representative on behalf of the Underwriters (with respect to the Deferred Discount)). It is acknowledged and agreed that while the funds are on deposit in the trust operating account at J.P. Morgan Chase Bank, N.A., awaiting distribution, the Company will not earn any interest or dividends.

On the Consummation Date (i) counsel for the Company shall deliver to you written notification that the Business Combination has been consummated, or will be consummated concurrently with your transfer of funds to the accounts as directed by the Company (the "Notification") and (ii) the Company shall deliver to you a certification by the Chief Executive Officer of the vote with an affidavit which verifies that the Business Combination has been approved by a vote of the Company's stockholders, if a vote is held and (b) a joint written instruction signed by the Company and the Representative with respect to the transfer of the funds held in the Trust Account, including payment of the Deferred Discount to the Representative from the Trust Account (the "Instruction Letter"). You are hereby directed and authorized to transfer the funds held in the Trust Account immediately upon your receipt of the Notification and the Instruction Letter, in accordance with the terms of the Instruction Letter. In the event that certain deposits held in the Trust Account may not be liquidated by the Consummation Date without penalty, you will notify the Company in writing of the same and the Company shall direct you as to whether such funds should remain in the Trust Account and be distributed after the Consummation Date to the Company. Upon the distribution of all the funds, net of any payments necessary for reasonable unreimbursed expenses related to liquidating the Trust Account, your obligations under the Trust Agreement shall be terminated.

In the event that the Business Combination is not consummated on the Consummation Date described in the notice thereof and we have not notified you on or before the original Consummation Date of a new Consummation Date, then upon receipt by the Trustee of written instructions from the Company, the funds held in the Trust Account shall be reinvested as provided in Section 1(c) of the Trust Agreement on the business day immediately following the Consummation Date as set forth in such written instruction as soon thereafter as possible.

Very truly yours, Aesther Healthcare Acquisition Corp. Bv: Name: Suren Ajjarapu Title: Chief Executive Officer EF Hutton, division of Benchmark Investments, LLC

By:

Name: Edward Tsuker Title: Head of Capital Markets

EXHIBIT B

[Letterhead of Company]

[Insert date]

Continental Stock Transfer & Trust Company 1 State Street, 30th Floor New York, New York 10004 Attn: Francis Wolf and Celeste Gonzalez

Re: Trust Account No. Termination Letter

Dear Mr. Wolf and Ms. Gonzalez:

Pursuant to Section 1(i) of the Investment Management Trust Agreement between Aesther Healthcare Acquisition Corp. (the "Company") and Continental Stock Transfer & Trust Company (the "Trustee"), dated as of [], 2021 (the "Trust Agreement"), this is to advise you that the Company has been unable to effect a business combination with a Target Business within the time frame specified in the Company's Amended and Restated Certificate of Incorporation, as described in the Company's Prospectus relating to the Offering. Capitalized terms used but not defined herein shall have the meanings set forth in the Trust Agreement.

In accordance with the terms of the Trust Agreement, we hereby authorize you to liquidate all of the assets in the Trust Account and to transfer the total proceeds into a segregated account held by you on behalf of the Beneficiaries to await distribution to the Public Stockholders. The Company has selected [](1) as the effective date for the purpose of determining when the Public Stockholders will be entitled to receive their share of the liquidation proceeds. You agree to be the Paying Agent of record and, in your separate capacity as Paying Agent, agree to distribute said funds directly to the Company's Public Stockholders in accordance with the terms of the Trust Agreement and the Amended and Restated Certificate of Incorporation of the Company. Upon the distribution of all the funds, net of any payments necessary for reasonable unreimbursed expenses related to liquidating the Trust Account, your obligations under the Trust Agreement shall be terminated, except to the extent otherwise provided in Section 1(i) of the Trust Agreement 12 months from the closing of the Offering (or up to 18 months if the Company extends the period of time as described in more detail in the Prospectus) or at a later date, if extended.

Very truly yours,

Aesther Healthcare Acquisition Corp.

By:

Name: Suren Ajjarapu

Title: Chief Executive Officer

EXHIBIT C

[Letterhead of Company]

[Insert date]

Continental Stock Transfer & Trust Company 1 State Street, 30th Floor New York, New York 10004 Attn: Francis Wolf and Celeste Gonzalez

Re: Trust Account No. Tax Payment Withdrawal Instruction

Dear Mr. Wolf and Ms. Gonzalez:

Pursuant to Section 1(j) of the Investment Management Trust Agreement between Aesther Healthcare Acquisition Corp. (the "Company") and Continental Stock Transfer & Trust Company (the "Trustee"), dated as of [], 2021 (the "Trust Agreement"), the Company hereby requests that you deliver to the Company \$ of the interest income earned on the Property as of the date hereof. Capitalized terms used but not defined herein shall have the meanings set forth in the Trust Agreement.

The Company needs such funds to pay for the tax obligations as set forth on the attached tax return or tax statement. In accordance with the terms of the Trust Agreement, you are hereby directed and authorized to transfer (via wire transfer) such funds promptly upon your receipt of this letter to the Company's operating account at:

[WIRE INSTRUCTION INFORMATION]

Very truly yours,

Aesther Healthcare Acquisition Corp.

By:

Name: Suren Ajjarapu
Title: Chief Executive Officer

EXHIBIT D

[Letterhead of Company]

[Insert date]

Continental Stock Transfer & Trust Company 1 State Street, 30th Floor New York, New York 10004 Attn: Francis Wolf and Celeste Gonzalez

Re: Trust Account No. Extension Letter	
Dear Mr. Wolf and Ms. Gonzalez:	
Stock Transfer & Trust Company, LLC, dated as of [], 2021 (the "Trust	Detween Aesther Healthcare Acquisition Corp. ("Company") and Continental Agreement"), this is to advise you that the Company is extending the time asinesses for an additional three (3) months, from to (the
This Extension Letter shall serve as the notice required with respect to the E have the meanings ascribed to them in the Trust Agreement.	stended Terms. Capitalized words used herein and not otherwise defined shall
In accordance with the terms of the Trust Agreement, we hereby authorize option was exercised in full, or in any case, \$0.10 per Unit), which will be wi	you to deposit \$1,000,000 (or \$1,150,000 if the underwriters' over-allotment red to you, into the Trust Account investments upon receipt.
This is the of up to two Extension Letters.	
	Very truly yours,
	Aesther Healthcare Acquisition Corp.
	Ву:
	Name: Suren Ajjarapu Title: Chief Executive Officer

EXHIBIT E

[Letterhead of Company]

[Insert date]

Continental Stock Transfer & Trust Company 1 State Street, 30th Floor New York, New York 10004 Attn: Francis Wolf and Celeste Gonzalez

Re: Trust Account No. Stockholder Redemption Withdrawal Instruction

Dear Mr. Wolf and Ms. Gonzalez:

Pursuant to Section 1(k) of the Investment Management Trust Agreement between Aesther Healthcare Acquisition Corp. (the "Company") and Continental Stock Transfer & Trust Company (the "Trustee"), dated as of [], 2021 (the "Trust Agreement"), the Company hereby requests that you deliver to the redeeming Public Stockholders of the Company \$ of the principal and interest income earned on the Property as of the date hereof into a segregated account held by you on behalf of the Beneficiaries. Capitalized terms used but not defined herein shall have the meanings set forth in the Trust Agreement.

The Company needs such funds to pay its Public Stockholders who have properly elected to have their shares of Common Stock redeemed by the Company in connection with a stockholder vote to approve an amendment to the Company's amended and restated certificate of incorporation that would affect the substance or timing of the Company's obligation to redeem 100% of its public shares of Common Stock if the Company has not consummated an initial Business Combination within such time as is described in the Company's amended and restated certificate of incorporation or with respect to any other provisions relating to stockholders' rights or pre-initial Business Combination activity. As such, you are hereby directed and authorized to transfer (via wire transfer) such funds promptly upon your receipt of this letter into a segregated account held by you on behalf of the Beneficiaries.

Very truly yours,

Aesther Healthcare Acquisition Corp.

By:

Name: Suren Ajjarapu
Title: Chief Executive Officer

REGISTRATION RIGHTS AGREEMENT

THIS REGISTRATION RIGHTS AGREEMENT (this "Agreement"), dated as of [], 2021, is made and entered into by and among Aesther Healthcare Acquisition Corp., a Delaware corporation (the "Company"), Aesther Healthcare Sponsor, LLC, a Delaware limited liability company (the "Sponsor"), and the undersigned parties listed on the signature page hereto (each such party, together with the Sponsor and any person or entity who hereafter becomes a party to this Agreement pursuant to Section 5.2 of this Agreement, a "Holder" and collectively the "Holders").

RECITALS

WHEREAS, the Company and the Sponsor have entered into that certain Securities Subscription Agreement (the "Founder Shares Purchase Agreement"), dated as of June 30, 2021, pursuant to which the Sponsor purchased an aggregate of 2,875,000 shares (the "Founder Shares") of the Company's Class B common stock, par value \$0.0001 per share (the "Class B Common Stock"), up to 375,000 shares of which will be forfeited to the Company for no consideration depending on the extent to which the underwriters of the Company's initial public offering exercise their over-allotment option;

WHEREAS, the Founder Shares will automatically convert into shares of the Company's Class A common stock, par value \$0.0001 per share (the "Common Stock"), at the time of the Company's initial Business Combination (as defined below) on a one-for-one basis, subject to adjustment, on the terms and conditions provided in the Company's amended and restated certificate of incorporation;

WHEREAS, on [], 2021, the Company and the Sponsor entered into that certain Private Placement Warrants Purchase Agreement, pursuant to which the Sponsor agreed to purchase 5,261,000 warrants (or 5,711,000 warrants if the over-allotment option is exercised in full) at a price of \$1.00 per warrant (the "Private Placement Warrants"), in a private placement transaction occurring simultaneously with the closing of the Company's initial public offering;

WHEREAS, in order to finance the Company's transaction costs in connection with an intended initial Business Combination (as defined below) the Sponsor or an affiliate of the Sponsor or certain of the Company's officers and directors may loan to the Company funds as the Company may require, of which up to \$1,500,000 of such loans may be convertible into warrants, each to purchase one share of Common Stock ("Working Capital Warrants") at a price of \$1.00 per warrant; and

WHEREAS, the Company and the Holders desire to enter into this Agreement, pursuant to which the Company shall grant the Holders certain registration rights with respect to certain securities of the Company, as set forth in this Agreement.

NOW, THEREFORE, in consideration of the representations, covenants and agreements contained herein, and certain other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

ARTICLE I

DEFINITIONS

- 1.1 <u>Definitions</u>. The terms defined in this Article I shall, for all purposes of this Agreement, have the respective meanings set forth below:
- "Adverse Disclosure" shall mean any public disclosure of material non-public information, which disclosure, in the good faith judgment of the Chief Executive Officer or principal financial officer of the Company, after consultation with counsel to the Company, (i) would be required to be made in any Registration Statement or Prospectus in order for the applicable Registration Statement or Prospectus not to contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements contained therein (in the case of any prospectus and any preliminary prospectus, in the light of the circumstances under which they were made) not misleading, (ii) would not be required to be made at such time if the Registration Statement were not being filed, and (iii) the Company has a bona fide business purpose for not making such information public.
 - "Agreement" shall have the meaning given in the Preamble.
 - "Board" shall mean the Board of Directors of the Company.
- "Business Combination" shall mean any merger, capital stock exchange, asset acquisition, stock purchase, reorganization or other similar business combination with one or more businesses, involving the Company.
 - "Commission" shall mean the Securities and Exchange Commission.
 - "Common Stock" shall have the meaning given in the Recitals hereto.
 - "Company" shall have the meaning given in the Preamble.
 - "Demand Registration" shall have the meaning given in subsection 2.1.1.
 - "Demanding Holder" shall have the meaning given in subsection 2.1.1.
 - "Exchange Act" shall mean the Securities Exchange Act of 1934, as it may be amended from time to time.
 - "Form S-1" shall have the meaning given in subsection 2.1.1.
 - "Form S-3" shall have the meaning given in subsection 2.3.
- "Founder Shares" shall have the meaning given in the Recitals hereto and shall be deemed to include the shares of Common Stock issuable upon conversion thereof.
- "Founder Shares Lock-up Period" shall mean, with respect to the Founder Shares, the period ending on the earlier of (A) one year after the completion of the Company's initial Business Combination or (B) subsequent to the Business Combination, (x) if the last sale price of the Common Stock equals or exceeds \$12.00 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like) for any 20 trading days within any 30-trading day period commencing at least 150 days after the Company's initial Business Combination or (y) the date on which the Company completes a liquidation, merger, capital stock exchange, reorganization or other similar transaction that results in all of the Company's stockholders having the right to exchange their shares of Common Stock for cash, securities or other property.
 - "Founder Shares Purchase Agreement" shall have the meaning given in the Recitals hereto.
 - "Holders" shall have the meaning given in the Preamble.
- "Insider Letter" shall mean that certain letter agreement, dated as of [], 2021, by and among the Company, the Sponsor and each of the Company's officers, directors, director nominees and advisors.
 - "Maximum Number of Securities" shall have the meaning given in subsection 2.1.4.

- "*Misstatement*" shall mean an untrue statement of a material fact or an omission to state a material fact required to be stated in a Registration Statement or Prospectus, or necessary to make the statements in a Registration Statement or Prospectus in the light of the circumstances under which they were made not misleading.
- "Permitted Transferees" shall mean any person or entity to whom a Holder of Registrable Securities is permitted to transfer such Registrable Securities prior to the expiration of the Founder Shares Lock-up Period, Private Placement Lock-up Period, or Working Capital Warrant Lock-up Period, as the case may be, under the Insider Letter, this Agreement, and any other applicable agreement between such Holder and the Company, and to any transferee thereafter.
 - "Piggyback Registration" shall have the meaning given in subsection 2.2.1.
- "Private Placement Lock-up Period" shall mean, with respect to Private Placement Warrants that are held by the initial purchasers of such Private Placement Warrants or their Permitted Transferees, and any of the securities underlying the Private Placement Warrants, including the Common Stock issued or issuable upon the exercise of the Private Placement Warrants, that are held by the initial purchasers of the Private Placement Warrants or their Permitted Transferees, the period ending 30 days after the completion of the Company's initial Business Combination.
 - "Private Placement Warrants" shall have the meaning given in the Recitals hereto.
 - "Pro Rata" shall have the meaning given in subsection 2.1.4.
- "*Prospectus*" shall mean the prospectus included in any Registration Statement, as supplemented by any and all prospectus supplements and as amended by any and all post-effective amendments and including all material incorporated by reference in such prospectus.
- "Registrable Security" shall mean (a) the Founder Shares and the shares of Common Stock issued or issuable upon the conversion of any Founder Shares, (b) the Private Placement Warrants and the Common Stock issued or issuable upon the exercise of the Private Placement Warrants, (c) any outstanding share of the Common Stock or any other equity security (including the shares of Common Stock issued or issuable upon the exercise of any other equity security) of the Company held by a Holder as of the date of this Agreement, (d) any Working Capital Warrants (including the Common Stock issued or issuable upon the exercise of the Working Capital Warrants), and (e) any other equity security of the Company issued or issuable with respect to any such share of the Common Stock by way of a stock dividend or stock split or in connection with a combination of shares, recapitalization, merger, consolidation or reorganization; provided, however, that, as to any particular Registrable Security, such securities shall cease to be Registrable Securities when: (A) a Registration Statement with respect to the sale of such securities shall have become effective under the Securities Act and such securities when: (B) such securities for such securities not bearing a legend restricting further transfer shall have been delivered by the Company and subsequent public distribution of such securities shall not require registration under the Securities Act; (C) such securities shall have ceased to be outstanding; (D) such securities may be sold without registration pursuant to Rule 144 promulgated under the Securities Act (or any successor rule promulgated thereafter by the Commission) (but with no volume or other restrictions or limitations); or (E) such securities have been sold to, or through, a broker, dealer or underwriter in a public distribution or other public securities transaction.
- "*Registration*" shall mean a registration effected by preparing and filing a registration statement or similar document in compliance with the requirements of the Securities Act, and the applicable rules and regulations promulgated thereunder, and any such registration statement having been declared effective by, or become effective pursuant to rules promulgated by, the Commission.
 - "Registration Expenses" shall mean the out-of-pocket expenses of a Registration, including, without limitation, the following:
- (A) all registration and filing fees (including fees with respect to filings required to be made with the Financial Industry Regulatory Authority, Inc.) and any securities exchange on which the Common Stock is then listed;

- (B) fees and expenses of compliance with securities or blue sky laws (including reasonable fees and disbursements of counsel for the Underwriters in connection with blue sky qualifications of Registrable Securities);
 - (C) printing, messenger, telephone and delivery expenses;
 - (D) reasonable fees and disbursements of counsel for the Company;
- (E) reasonable fees and disbursements of all independent registered public accountants of the Company incurred specifically in connection with such Registration; and
- (F) reasonable fees and expenses of one (1) legal counsel selected by the majority-in-interest of the Demanding Holders initiating a Demand Registration to be registered for offer and sale in the applicable Registration.
- "*Registration Statement*" shall mean any registration statement that covers the Registrable Securities pursuant to the provisions of this Agreement, including the Prospectus included in such registration statement, amendments (including post-effective amendments) and supplements to such registration statement, and all exhibits to and all material incorporated by reference in such registration statement.
 - "Requesting Holder" shall have the meaning given in subsection 2.1.1.
 - "Securities Act" shall mean the Securities Act of 1933, as amended from time to time.
 - "Sponsor" shall have the meaning given in the Recitals hereto.
- "*Underwriter*" shall mean a securities dealer who purchases any Registrable Securities as principal in an Underwritten Offering and not as part of such dealer's market-making activities.
- "*Underwritten Registration*" or "*Underwritten Offering*" shall mean a Registration in which securities of the Company are sold to an Underwriter in a firm commitment underwriting for distribution to the public.
 - "Working Capital Warrants" shall have the meaning given in the Recitals hereto.
- "Working Capital Warrant Lock-up Period" shall mean, with respect to Working Capital Warrants and the shares of Common Stock issued or issuable upon the exercise of the warrants underlying the Working Capital Warrants, that are held by the initial purchasers of such warrants or their Permitted Transferees, the period ending 30 days after the completion of the Company's initial Business Combination.

ARTICLE II

REGISTRATIONS

2.1 Demand Registration.

2.1.1 Request for Registration. Subject to the provisions of subsection 2.1.4 and Section 2.4 hereof, at any time and from time to time on or after the date the Company consummates the Business Combination, the Holders of at least a majority in interest of the then-outstanding number of Registrable Securities (the "Demanding Holders") may make a written demand for Registration under the Securities Act of all or part of their Registrable Securities, which written demand shall describe the amount and type of securities to be included in such Registration and the intended method(s) of distribution thereof (such written demand a "Demand Registration"). The Company shall, within ten (10) days of the Company's receipt of the Demand Registration, notify, in writing, all other Holders of Registrable Securities of such demand, and each Holder of Registrable Securities who thereafter wishes to include all or a portion of such Holder's Registrable Securities in a Registration pursuant to a Demand Registration (each such Holder that includes all or a portion of such Holder's Registrable Securities in such Registration, a "Requesting Holder") shall so notify the Company, in writing, within five (5) days after the receipt by the Holder of the notice from the Company. Upon receipt by the Company of any such written notification from a Requesting Holder(s) to the Company, such Requesting Holder(s) shall be entitled to have their Registrable Securities included in a Registration pursuant to a Demand Registration and the Company shall effect, as soon thereafter as practicable, but not more than forty five (45) days immediately after the Company's receipt of the Demand Registration, the Registration of all Registrable Securities requested by the Demanding Holders and Requesting Holders pursuant to such Demand Registration. Under no circumstances shall the Company be obligated to effect more than an aggregate of three (3) Registrations pursuant to a Demand Registration under this subsection 2.1.1 with respect to any or all Registrable Securities; provided, however, that a Registration shall not be counted for such purposes unless a Form S-1 or any similar long-form registration statement that may be available at such time ("Form S-1") has become effective and all of the Registrable Securities requested by the Requesting Holders to be registered on behalf of the Requesting Holders in such Form S-1 Registration have been sold, in accordance with Section 3.1 of this Agreement.

2.1.2 Effective Registration. Notwithstanding the provisions of subsection 2.1.1 above or any other part of this Agreement, a Registration pursuant to a Demand Registration shall not count as a Registration unless and until (i) the Registration Statement filed with the Commission with respect to a Registration pursuant to a Demand Registration has been declared effective by the Commission and (ii) the Company has complied with all of its obligations under this Agreement with respect thereto; provided, further, that if, after such Registration Statement has been declared effective, an offering of Registrable Securities in a Registration pursuant to a Demand Registration is subsequently interfered with by any stop order or injunction of the Commission, federal or state court or any other governmental agency the Registration Statement with respect to such Registration shall be deemed not to have been declared effective, unless and until, (i) such stop order or injunction is removed, rescinded or otherwise terminated, and (ii) a majority-in-interest of the Demanding Holders initiating such Demand Registration thereafter affirmatively elect to continue with such Registration and accordingly notify the Company in writing, but in no event later than five (5) days, of such election; and provided, further, that the Company shall not be obligated or required to file another Registration Statement until the Registration Statement that has been previously filed with respect to a Registration pursuant to a Demand Registration becomes effective or is subsequently terminated.

2.1.3 <u>Underwritten Offering</u>. Subject to the provisions of subsection 2.1.4 and Section 2.4 hereof, if a majority-in-interest of the Demanding Holders so advise the Company as part of their Demand Registration that the offering of the Registrable Securities pursuant to such Demand Registration shall be in the form of an Underwritten Offering, then the right of such Demanding Holder or Requesting Holder (if any) to include its Registrable Securities in such Registration shall be conditioned upon such Holder's participation in such Underwritten Offering and the inclusion of such Holder's Registrable Securities in such Underwritten Offering to the extent provided herein. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this subsection 2.1.3 shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the majority-in-interest of the Demanding Holders initiating the Demand Registration.

2.1.4 Reduction of Underwritten Offering. If the managing Underwriter or Underwriters in an Underwritten Registration pursuant to a Demand Registration, in good faith, advises the Company, the Demanding Holders and the Requesting Holders (if any) in writing that the dollar amount or number of Registrable Securities that the Demanding Holders and the Requesting Holders (if any) desire to sell, taken together with all other shares of Common Stock or other equity securities that the Company desires to sell and the shares of Common Stock, if any, as to which a Registration has been requested pursuant to separate written contractual piggy-back registration rights held by any other stockholders who desire to sell, exceeds the maximum dollar amount or maximum number of equity securities that can be sold in the Underwritten Offering without adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of such offering (such maximum dollar amount or maximum number of such securities, as applicable, the "Maximum Number of Securities"), then the Company shall include in such Underwritten Offering, as follows: (i) first, the Registrable Securities of the Demanding Holders and the Requesting Holders (if any) (pro rata based on the respective number of Registrable Securities that each Demanding Holder and Requesting Holder (if any) has requested be included in such Underwritten Registration and the aggregate number of Registrable Securities that the Demanding Holders and Requesting Holders have requested be included in such Underwritten Registration (such proportion is referred to herein as "Pro Rata")) that can be sold without exceeding the Maximum Number of Securities; (ii) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (i), the Registrable Securities of Holders (Pro Rata, based on the respective number of Registrable Securities that each Holder has so requested) exercising their rights to register their Registrable Securities pursuant to subsection 2.2.1 hereof, without exceeding the Maximum Number of Securities; and (iii) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii), the Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; and (iv) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i), (ii) and (iii), the Common Stock or other equity securities of other persons or entities that the Company is obligated to register in a Registration pursuant to separate written contractual arrangements with such persons and that can be sold without exceeding the Maximum Number of Securities.

2.1.5 <u>Demand Registration Withdrawal</u>. A majority-in-interest of the Demanding Holders initiating a Demand Registration or a majority-in-interest of the Requesting Holders (if any), pursuant to a Registration under subsection 2.1.1 shall have the right to withdraw from a Registration pursuant to such Demand Registration for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of their intention to withdraw from such Registration prior to the effectiveness of the Registration Statement filed with the Commission with respect to the Registration of their Registrable Securities pursuant to such Demand Registration. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with a Registration pursuant to a Demand Registration prior to its withdrawal under this subsection 2.1.5.

2.2 Piggyback Registration.

2.2.1 Piggyback Rights. If, at any time on or after the date the Company consummates an initial Business Combination, the Company proposes to file a Registration Statement under the Securities Act with respect to an offering of equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into equity securities, for its own account or for the account of stockholders of the Company (or by the Company and by the stockholders of the Company including, without limitation, pursuant to Section 2.1 hereof), other than a Registration Statement (i) filed in connection with any employee stock option or other benefit plan, (ii) for an exchange offer or offering of securities solely to the Company's existing stockholders, (iii) for an offering of debt that is convertible into equity securities of the Company or (iv) for a dividend reinvestment plan, then the Company shall give written notice of such proposed filing to all of the Holders of Registrable Securities as soon as practicable but not less than ten (10) days before the anticipated filing date of such Registration Statement, which notice shall (A) describe the amount and type of securities to be included in such offering, the intended method(s) of distribution, and the name of the proposed managing Underwriter or Underwriters, if any, in such offering, and (B) offer to all of the Holders of Registrable Securities the opportunity to register the sale of such number of Registrable Securities as such Holders may request in writing within five (5) days after receipt of such written notice (such Registration a "Piggyback Registration"). The Company shall, in good faith, cause such Registrable Securities to be included in such Piggyback Registration and shall use its best efforts to cause the managing Underwriter or Underwriters of a proposed Underwritten Offering to permit the Registrable Securities requested by the Holders pursuant to this subsection 2.2.1 to be included in a Piggyback Registration on the same terms and conditions as any similar securities of the Company included in such Registration and to permit the sale or other disposition of such Registrable Securities in accordance with the intended method(s) of distribution thereof. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this subsection 2.2.1 shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the Company.

2.2.2 <u>Reduction of Piggyback Registration</u>. If the managing Underwriter or Underwriters in an Underwritten Registration that is to be a Piggyback Registration, in good faith, advises the Company and the Holders of Registrable Securities participating in the Piggyback Registration in writing that the dollar amount or number of shares of Common Stock that the Company desires to sell, taken together with (i) the shares of Common Stock, if any, as to which Registration has been demanded pursuant to separate written contractual arrangements with persons or entities other than the Holders of Registrable Securities hereunder, (ii) the Registrable Securities as to which registration has been requested pursuant to Section 2.2 hereof, and (iii) the shares of Common Stock, if any, as to which Registration has been requested pursuant to separate written contractual piggy-back registration rights of other stockholders of the Company, exceeds the Maximum Number of Securities, then:

(a) If the Registration is undertaken for the Company's account, the Company shall include in any such Registration (A) first, the shares of Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to subsection 2.2.1 hereof, pro rata, based on the respective number of Registrable Securities that each Holder has so requested exercising its rights to register its Registrable Securities, which can be sold without exceeding the Maximum Number of Securities; and (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), the shares of Common Stock, if any, as to which Registration has been requested pursuant to written contractual piggy-back registration rights of other stockholders of the Company, which can be sold without exceeding the Maximum Number of Securities;

(b) If the Registration is pursuant to a request by persons or entities other than the Holders of Registrable Securities, then the Company shall include in any such Registration (A) first, the shares of Common Stock or other equity securities, if any, of such requesting persons or entities, other than the Holders of Registrable Securities, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to subsection 2.2.1, pro rata based on the respective number of Registrable Securities that each Holder has requested be included in such Underwritten Registration and the aggregate number of Registrable Securities that the Holders have requested to be included in such Underwritten Registration, which can be sold without exceeding the Maximum Number of Securities; (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), the shares of Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; and (D) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A), (B) and (C), the shares of Common Stock or other equity securities for the account of other persons or entities that the Company is obligated to register pursuant to separate written contractual arrangements with such persons or entities, which can be sold without exceeding the Maximum Number of Securities.

- 2.2.3 <u>Piggyback Registration Withdrawal</u>. Any Holder of Registrable Securities shall have the right to withdraw from a Piggyback Registration for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of his, her or its intention to withdraw from such Piggyback Registration prior to the effectiveness of the Registration Statement filed with the Commission with respect to such Piggyback Registration. The Company (whether on its own good faith determination or as the result of a request for withdrawal by persons pursuant to separate written contractual obligations) may withdraw a Registration Statement filed with the Commission in connection with a Piggyback Registration at any time prior to the effectiveness of such Registration Statement. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with the Piggyback Registration prior to its withdrawal under this subsection 2.2.3.
- 2.2.4 <u>Unlimited Piggyback Registration Rights</u>. For purposes of clarity, any Registration effected pursuant to Section 2.2 hereof shall not be counted as a Registration pursuant to a Demand Registration effected under Section 2.1 hereof.
- 2.3 Registrations on Form S-3. The Holders of Registrable Securities may at any time, and from time to time, request in writing that the Company, pursuant to Rule 415 under the Securities Act (or any successor rule promulgated thereafter by the Commission), register the resale of any or all of their Registrable Securities on Form S-3 or any similar short form registration statement that may be available at such time ("Form S-3"); provided, however, that the Company shall not be obligated to effect such request through an Underwritten Offering. Within five (5) days of the Company's receipt of a written request from a Holder or Holders of Registrable Securities for a Registration on Form S-3, the Company shall promptly give written notice of the proposed Registration on Form S-3 to all other Holders of Registrable Securities, and each Holder of Registrable Securities who thereafter wishes to include all or a portion of such Holder's Registrable Securities in such Registration on Form S-3 shall so notify the Company, in writing, within ten (10) days after the receipt by the Holder of the notice from the Company. As soon as practicable thereafter, but not more than twelve (12) days after the Company's initial receipt of such written request for a Registration on Form S-3, the Company shall file a Registration Statement relating to all or such portion of such Holder's Registrable Securities as are specified in such written request, together with all or such portion of Registrable Securities of any other Holder or Holders; provided, however, that the Company shall not be obligated to effect any such Registration pursuant to Section 2.3 hereof if (i) a Form S-3 is not available for such offering; or (ii) the Holders of Registrable Securities and such other equity securities (if any) at any aggregate price to the public of less than \$10,000,000.

2.4 Restrictions on Registration Rights. If (A) during the period starting with the date sixty (60) days prior to the Company's good faith estimate of the date of the filing of, and ending on a date one hundred and twenty (120) days after the effective date of, a Company initiated Registration and provided that the Company has delivered written notice to the Holders prior to receipt of a Demand Registration pursuant to subsection 2.1.1 and it continues to actively employ, in good faith, all reasonable efforts to cause the applicable Registration Statement to become effective; (B) the Holders have requested an Underwritten Registration and the Company and the Holders are unable to obtain the commitment of underwriters to firmly underwrite the offer; or (C) in the good faith judgment of the Board such Registration would be seriously detrimental to the Company and the Board concludes as a result that it is essential to defer the filing of such Registration Statement at such time, then in each case the Company shall furnish to such Holders a certificate signed by the Chairman of the Board stating that in the good faith judgment of the Board it would be seriously detrimental to the Company for such Registration Statement to be filed in the near future and that it is therefore essential to defer the filing of such Registration Statement. In such event, the Company shall have the right to defer such filing for a period of not more than thirty (30) days; provided, however, that the Company shall not defer its obligation in this manner more than once in any 12 month period.

ARTICLE III

COMPANY PROCEDURES

- 3.1 <u>General Procedures</u>. If at any time on or after the date the Company consummates an initial Business Combination the Company is required to effect the Registration of Registrable Securities, the Company shall use its best efforts to effect such Registration to permit the sale of such Registrable Securities in accordance with the intended plan of distribution thereof, and pursuant thereto the Company shall, as expeditiously as possible:
- 3.1.1 prepare and file with the Commission as soon as practicable a Registration Statement with respect to such Registrable Securities and use its reasonable best efforts to cause such Registration Statement to become effective and remain effective until all Registrable Securities covered by such Registration Statement have been sold;
- 3.1.2 prepare and file with the Commission such amendments and post-effective amendments to the Registration Statement, and such supplements to the Prospectus, as may be reasonably requested by the majority-in-interest of the Holders with Registrable Securities registered on such Registration Statement or any Underwriter of Registrable Securities or as may be required by the rules, regulations or instructions applicable to the registration form used by the Company or by the Securities Act or rules and regulations thereunder to keep the Registration Statement effective until all Registrable Securities covered by such Registration Statement are sold in accordance with the intended plan of distribution set forth in such Registration Statement or supplement to the Prospectus;
- 3.1.3 prior to filing a Registration Statement or Prospectus, or any amendment or supplement thereto, furnish without charge to the Underwriters, if any, and the Holders of Registrable Securities included in such Registration, and such Holders' legal counsel, copies of such Registration Statement as proposed to be filed, each amendment and supplement to such Registration Statement (in each case including all exhibits thereto and documents incorporated by reference therein), the Prospectus included in such Registration Statement (including each preliminary Prospectus), and such other documents as the Underwriters and the Holders of Registrable Securities included in such Registration or the legal counsel for any such Holders may request in order to facilitate the disposition of the Registrable Securities owned by such Holders; *provided*, that the Company will not have any obligation to provide any document pursuant to this clause that is available on the Commission's EDGAR system;
- 3.1.4 prior to any public offering of Registrable Securities, use its best efforts to (i) register or qualify the Registrable Securities covered by the Registration Statement under such securities or "blue sky" laws of such jurisdictions in the United States as the Holders of Registrable Securities included in such Registration Statement (in light of their intended plan of distribution) may request and (ii) take such action necessary to cause such Registrable Securities covered by the Registration Statement to be registered with or approved by such other governmental authorities as may be necessary by virtue of the business and operations of the Company and do any and all other acts and things that may be necessary or advisable to enable the Holders of Registrable Securities included in such Registration Statement to consummate the disposition of such Registrable Securities in such jurisdictions; provided, however, that the Company shall not be required to qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify or take any action to which it would be subject to general service of process or taxation in any such jurisdiction where it is not then otherwise so subject;

- 3.1.5 cause all such Registrable Securities to be listed on each securities exchange or automated quotation system on which similar securities issued by the Company are then listed;
- 3.1.6 provide a transfer agent or warrant agent, as applicable, and registrar for all such Registrable Securities no later than the effective date of such Registration Statement;
- 3.1.7 advise each seller of such Registrable Securities, promptly after it shall receive notice or obtain knowledge thereof, of the issuance of any stop order by the Commission suspending the effectiveness of such Registration Statement or the initiation or threatening of any proceeding for such purpose and promptly use its reasonable best efforts to prevent the issuance of any stop order or to obtain its withdrawal if such stop order should be issued;
- 3.1.8 at least five (5) days prior to the filing of any Registration Statement or Prospectus or any amendment or supplement to such Registration Statement or Prospectus or any document that is to be incorporated by reference into such Registration Statement or Prospectus, furnish a copy thereof to each seller of such Registrable Securities or its counsel; *provided*, that the Company will not have any obligation to provide any document pursuant to this clause that is available on the Commission's EDGAR system;
- 3.1.9 notify the Holders at any time when a Prospectus relating to such Registration Statement is required to be delivered under the Securities Act, of the happening of any event as a result of which the Prospectus included in such Registration Statement, as then in effect, includes a Misstatement, and then to correct such Misstatement as set forth in Section 3.4 hereof;
- 3.1.10 permit a representative of the Holders (such representative to be selected by a majority of the participating Holders), the Underwriters, if any, and any attorney or accountant retained by such Holders, or Underwriters to participate, at each such person's own expense, in the preparation of the Registration Statement, and cause the Company's officers, directors and employees to supply all information reasonably requested by any such representative, Underwriter, attorney or accountant in connection with the Registration; provided, however, that such representative or Underwriters enter into a confidentiality agreement, in form and substance reasonably satisfactory to the Company, prior to the release or disclosure of any such information and provided further, the Company may not include the name of any Holder or Underwriter or any information regarding any Holder or Underwriter in any Registration Statement or Prospectus, any amendment or supplement to such Registration Statement or Prospectus, any document that is to be incorporated by reference into such Registration Statement or Prospectus, or any response to any comment letter, without the prior written consent of such Holder or Underwriter and providing each such Holder or Underwriter a reasonable amount of time to review and comment on such applicable document, which comments the Company shall include unless contrary to applicable law;
- 3.1.11 obtain a "cold comfort" letter from the Company's independent registered public accountants in the event of an Underwritten Registration, which the participating Holders may rely on, in customary form and covering such matters of the type customarily covered by "cold comfort" letters as the managing Underwriter may reasonably request, and reasonably satisfactory to a majority-in-interest of the participating Holders;
- 3.1.12 on the date the Registrable Securities are delivered for sale pursuant to such Registration, obtain an opinion, dated such date, of counsel representing the Company for the purposes of such Registration, addressed to the Holders, the placement agent or sales agent, if any, and the Underwriters, if any, covering such legal matters with respect to the Registration in respect of which such opinion is being given as the Holders, placement agent, sales agent, or Underwriter may reasonably request and as are customarily included in such opinions and negative assurance letters, and reasonably satisfactory to a majority in interest of the participating Holders;

- 3.1.13 in the event of any Underwritten Offering, enter into and perform its obligations under an underwriting agreement, in usual and customary form, with the managing Underwriter of such offering;
- 3.1.14 make available to its security holders, as soon as reasonably practicable, an earnings statement covering the period of at least twelve (12) months beginning with the first day of the Company's first full calendar quarter after the effective date of the Registration Statement which satisfies the provisions of Section 11(a) of the Securities Act and Rule 158 thereunder (or any successor rule promulgated thereafter by the Commission);
- 3.1.15 if the Registration involves the Registration of Registrable Securities involving gross proceeds in excess of \$51,000,000, use its reasonable efforts to make available senior executives of the Company to participate in customary "road show" presentations that may be reasonably requested by the Underwriter in any Underwritten Offering; and
- 3.1.16 otherwise, in good faith, cooperate reasonably with, and take such customary actions as may reasonably be requested by the Holders, in connection with such Registration.
- 3.2 <u>Registration Expenses</u>. The Registration Expenses of all Registrations shall be borne by the Company. It is acknowledged by the Holders that the Holders shall bear all incremental selling expenses relating to the sale of Registrable Securities, such as Underwriters' commissions and discounts, brokerage fees, Underwriter marketing costs and, other than as set forth in the definition of "Registration Expenses," all reasonable fees and expenses of any legal counsel representing the Holders.
- 3.3 Requirements for Participation in Underwritten Offerings. No person may participate in any Underwritten Offering for equity securities of the Company pursuant to a Registration initiated by the Company hereunder unless such person (i) agrees to sell such person's securities on the basis provided in any underwriting arrangements approved by the Company and (ii) completes and executes all customary questionnaires, powers of attorney, indemnities, lock-up agreements, underwriting agreements and other customary documents as may be reasonably required under the terms of such underwriting arrangements.
- 3.4 <u>Suspension of Sales; Adverse Disclosure</u>. Upon receipt of written notice from the Company that a Registration Statement or Prospectus contains a Misstatement, each of the Holders shall forthwith discontinue disposition of Registrable Securities until he, she or it has received copies of a supplemented or amended Prospectus correcting the Misstatement (it being understood that the Company hereby covenants to prepare and file such supplement or amendment as soon as practicable after the time of such notice), or until he, she or it is advised in writing by the Company that the use of the Prospectus may be resumed. If the filing, initial effectiveness or continued use of a Registration Statement in respect of any Registration at any time would require the Company to make an Adverse Disclosure or would require the inclusion in such Registration Statement of financial statements that are unavailable to the Company for reasons beyond the Company's control, the Company may, upon giving prompt written notice of such action to the Holders, delay the filing or initial effectiveness of, or suspend use of, such Registration Statement for the shortest period of time, but in no event more than ninety (90) days in any 12-month period, determined in good faith by the Company to be necessary for such purpose. In the event the Company exercises its rights under the preceding sentence, the Holders agree to suspend, immediately upon their receipt of the notice referred to above, their use of the Prospectus relating to any Registration in connection with any sale or offer to sell Registrable Securities. The Company shall immediately notify the Holders of the expiration of any period during which it exercised its rights under this Section 3.4.
- 3.5 Reporting Obligations. As long as any Holder shall own Registrable Securities, the Company, at all times while it shall be a reporting company under the Exchange Act, covenants to file timely (or obtain extensions in respect thereof and file within the applicable grace period) all reports required to be filed by the Company after the date hereof pursuant to Sections 13(a) or 15(d) of the Exchange Act and to promptly furnish the Holders with true and complete copies of all such filings. The Company further covenants that it shall take such further action as any Holder may reasonably request, all to the extent required from time to time to enable such Holder to sell shares of the Common Stock held by such Holder without registration under the Securities Act within the limitation of the exemptions provided by Rule 144 promulgated under the Securities Act (or any successor rule promulgated thereafter by the Commission), including providing any legal opinions. Upon the request of any Holder, the Company shall deliver to such Holder a written certification of a duly authorized officer as to whether it has complied with such requirements.

ARTICLE IV

INDEMNIFICATION AND CONTRIBUTION

4.1 Indemnification.

4.1.1 The Company agrees to indemnify, to the extent permitted by law, each Holder of Registrable Securities, its officers and directors and each person who controls such Holder (within the meaning of the Securities Act) against all losses, claims, damages, liabilities and expenses (including attorneys' fees) caused by any untrue or alleged untrue statement of material fact contained in any Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, except insofar as the same are caused by or contained in any information furnished in writing to the Company by such Holder expressly for use therein. The Company shall indemnify the Underwriters, their officers and directors and each person who controls such Underwriters (within the meaning of the Securities Act) to the same extent as provided in the foregoing with respect to the indemnification of the Holder.

4.1.2 In connection with any Registration Statement in which a Holder of Registrable Securities is participating, such Holder shall furnish to the Company in writing such information and affidavits as the Company reasonably requests for use in connection with any such Registration Statement or Prospectus and, to the extent permitted by law, shall indemnify the Company, its directors and officers and agents and each person who controls the Company (within the meaning of the Securities Act) against any losses, claims, damages, liabilities and expenses (including without limitation reasonable attorneys' fees) resulting from any untrue statement of material fact contained in the Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, but only to the extent that such untrue statement or omission is contained in any information or affidavit so furnished in writing by such Holder expressly for use therein. The Holders of Registrable Securities shall indemnify the Underwriters, their officers, directors and each person who controls such Underwriters (within the meaning of the Securities Act) to the same extent as provided in the foregoing with respect to indemnification of the Company. For the avoidance of doubt, the obligation to indemnify under this Section 4.1.2 shall be several, not joint and several, among the Holders of Registrable Securities, and the total indemnification liability of a Holder under this Section 5 statement.

4.1.3 Any person entitled to indemnification herein shall (i) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification (provided that the failure to give prompt notice shall not impair any person's right to indemnification hereunder to the extent such failure has not materially prejudiced the indemnifying party) and (ii) unless in such indemnified party's reasonable judgment a conflict of interest between such indemnified and indemnifying parties may exist with respect to such claim, permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying party shall not be subject to any liability for any settlement made by the indemnified party without its consent (but such consent shall not be unreasonably withheld). An indemnifying party who is not entitled to, or elects not to, assume the defense of a claim shall not be obligated to pay the fees and expenses of more than one counsel for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment of any indemnified party a conflict of interest may exist between such indemnified party and any other of such indemnified parties with respect to such claim. No indemnifying party shall, without the consent of the indemnified party, consent to the entry of any judgment or enter into any settlement which cannot be settled in all respects by the payment of money (and such money is so paid by the indemnifying party pursuant to the terms of such settlement) or which settlement does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnified party of a release from all liability in respect to such claim or litigation.

- 4.1.4 The indemnification provided for under this Agreement shall remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party or any officer, director or controlling person of such indemnified party and shall survive the transfer of securities. The Company and each Holder of Registrable Securities participating in an offering also agrees to make such provisions as are reasonably requested by any indemnified party for contribution to such party in the event the Company's or such Holder's indemnification is unavailable for any reason.
- 4.1.5 If the indemnification provided under Section 4.1 hereof from the indemnifying party is unavailable or insufficient to hold harmless an indemnified party in respect of any losses, claims, damages, liabilities and expenses referred to herein, then the indemnifying party, in lieu of indemnifying the indemnified party, shall contribute to the amount paid or payable by the indemnified party as a result of such losses, claims, damages, liabilities and expenses in such proportion as is appropriate to reflect the relative fault of the indemnifying party and the indemnified party, as well as any other relevant equitable considerations. The relative fault of the indemnifying party and indemnified party shall be determined by reference to, among other things, whether any action in question, including any untrue or alleged untrue statement of a material fact or omission or alleged omission to state a material fact, was made by, or relates to information supplied by, such indemnifying party or indemnified party, and the indemnifying party's and indemnified party's relative intent, knowledge, access to information and opportunity to correct or prevent such action; provided, however, that the liability of any Holder under this subsection 4.1.5 shall be limited to the amount of the net proceeds received by such Holder in such offering giving rise to such liability. The amount paid or payable by a party as a result of the losses or other liabilities referred to above shall be deemed to include, subject to the limitations set forth in subsections 4.1.1, 4.1.2 and 4.1.3 above, any legal or other fees, charges or expenses reasonably incurred by such party in connection with any investigation or proceeding. The parties hereto agree that it would not be just and equitable if contribution pursuant to this subsection 4.1.5 were determined by pro rata allocation or by any other method of allocation, which does not take account of the equitable considerations referred to in this subsection 4.1.5. No person guilty of fraudulent mi

ARTICLE V

MISCELLANEOUS

5.1 Notices. Any notice or communication under this Agreement must be in writing and given by (i) deposit in the United States mail, addressed to the party to be notified, postage prepaid and registered or certified with return receipt requested, (ii) delivery in person or by courier service providing evidence of delivery, or (iii) transmission by hand delivery, electronic mail or facsimile. Each notice or communication that is mailed, delivered, or transmitted in the manner described above shall be deemed sufficiently given, served, sent, and received, in the case of mailed notices, on the third business day following the date on which it is mailed and, in the case of notices delivered by courier service, hand delivery, electronic mail or facsimile, at such time as it is delivered to the addressee (with the delivery receipt or the affidavit of messenger) or at such time as delivery is refused by the addressee upon presentation. Any notice or communication under this Agreement must be addressed, if to the Company, to: 515 Madison Avenue, Suite 8078, New York, New York 10022, and, if to any Holder, at such Holder's address or contact information as set forth in the Company's books and records. Any party may change its address for notice at any time and from time to time by written notice to the other parties hereto, and such change of address shall become effective thirty (30) days after delivery of such notice as provided in this Section 5.1.

5.2 Assignment; No Third Party Beneficiaries.

- 5.2.1 This Agreement and the rights, duties and obligations of the Company hereunder may not be assigned or delegated by the Company in whole or in part.
- 5.2.2 Prior to the expiration of the Founder Shares Lock-up Period, the Private Placement Lock-up Period, or the Working Capital Warrant Lock-up Period, as the case may be, no Holder may assign or delegate such Holder's rights, duties or obligations under this Agreement, in whole or in part, except in connection with a transfer of Registrable Securities by such Holder to a Permitted Transferee but only if such Permitted Transferee agrees to become bound by the transfer restrictions set forth in this Agreement. After the expiration of the Founder Shares Lock-up Period, the Private Placement Lock-up Period, or the Working Capital Warrant Lock-up Period, as the case may be, the Holder may assign or delegate such Holder's rights, duties or obligations under this Agreement, in whole or in part, to any transferee.

- 5.2.3 This Agreement and the provisions hereof shall be binding upon and shall inure to the benefit of each of the parties and its successors and the permitted assigns of the Holders, which shall include Permitted Transferees.
- 5.2.4 This Agreement shall not confer any rights or benefits on any persons that are not parties hereto, other than as expressly set forth in this Agreement and Section 5.2 hereof.
- 5.2.5 No assignment by any party hereto of such party's rights, duties and obligations hereunder shall be binding upon or obligate the Company unless and until the Company shall have received (i) written notice of such assignment as provided in Section 5.1 hereof and (ii) the written agreement of the assignee, in a form reasonably satisfactory to the Company, to be bound by the terms and provisions of this Agreement (which may be accomplished by an addendum or certificate of joinder to this Agreement). Any transfer or assignment made other than as provided in this Section 5.2 shall be null and void.
- 5.3 <u>Counterparts</u>. This Agreement may be executed in multiple counterparts (including facsimile or PDF counterparts), each of which shall be deemed an original, and all of which together shall constitute the same instrument, but only one of which need be produced.
- 5.4 Governing Law; Venue. NOTWITHSTANDING THE PLACE WHERE THIS AGREEMENT MAY BE EXECUTED BY ANY OF THE PARTIES HERETO, THE PARTIES EXPRESSLY AGREE THAT (I) THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED UNDER THE LAWS OF THE STATE OF NEW YORK AS APPLIED TO AGREEMENTS AMONG NEW YORK RESIDENTS ENTERED INTO AND TO BE PERFORMED ENTIRELY WITHIN NEW YORK, WITHOUT REGARD TO THE CONFLICT OF LAW PROVISIONS OF SUCH JURISDICTION AND (II) THE VENUE FOR ANY ACTION TAKEN WITH RESPECT TO THIS AGREEMENT SHALL BE ANY STATE OR FEDERAL COURT IN NEW YORK COUNTY IN THE STATE OF NEW YORK.

EACH PARTY HERETO ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND, THEREFORE, EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT TO ANY ACTION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT.

- 5.5 <u>Amendments and Modifications</u>. Upon the written consent of the Company and the Holders of at least a majority in interest of the Registrable Securities at the time in question, compliance with any of the provisions, covenants and conditions set forth in this Agreement may be waived, or any of such provisions, covenants or conditions may be amended or modified; provided, however, that notwithstanding the foregoing, any amendment hereto or waiver hereof that adversely affects one Holder, solely in his, her or its capacity as a holder of the shares of capital stock of the Company, in a manner that is materially different from the other Holders (in such capacity) shall require the consent of the Holder so affected. No course of dealing between any Holder or the Company and any other party hereto or any failure or delay on the part of a Holder or the Company in exercising any rights or remedies under this Agreement shall operate as a waiver of any rights or remedies of any Holder or the Company. No single or partial exercise of any rights or remedies under this Agreement by a party shall operate as a waiver or preclude the exercise of any other rights or remedies hereunder or thereunder by such party.
- 5.6 Other Registration Rights. The Company represents and warrants that no person, other than a Holder of Registrable Securities, has any right to require the Company to register any securities of the Company for sale or to include such securities of the Company in any Registration filed by the Company for the sale of securities for its own account or for the account of any other person. Further, the Company represents and warrants that this Agreement supersedes any other registration rights agreement or agreement with similar terms and conditions and in the event of a conflict between any such agreement or agreements and this Agreement, the terms of this Agreement shall prevail.
- 5.7 Term. This Agreement shall terminate upon the earlier of (i) the tenth anniversary of the date of this Agreement or (ii) the date as of which (A) all of the Registrable Securities have been sold pursuant to a Registration Statement (but in no event prior to the applicable period referred to in Section 4(a)(3) of the Securities Act and Rule 174 thereunder (or any successor rule promulgated thereafter by the Commission)) or (B) the Holders of all Registrable Securities are permitted to sell the Registrable Securities under Rule 144 (or any similar provision) under the Securities Act without limitation on the amount of securities sold or the manner of sale and without compliance with the current public reporting requirements set forth under Rule 144(i)(2). The provisions of Section 3.5 and Article IV shall survive any termination.

[SIGNATURE PAGES FOLLOW]

IN WITNESS WHEREOF, the undersigned have caused this Agreement to be executed as of the date first written above.

COMPANY: AESTHER HEALTHCARE ACQUISITION CORP. a Delaware corporation By: Name: Suren Ajjarapu Title: Chief Executive Officer HOLDER: AESTHER HEALTHCARE SPONSOR, LLC, a Delaware limited liability company By: Name: Suren Ajjarapu Title: Chief Executive Officer Suren Ajjarapu Howard Doss Michael L. Peterson Venkatesh Srinivasan Donald G. Fell

Siva Saravanan

[Signature Page to Registration Rights Agreement]

Aesther Healthcare Acquisition Corp. 515 Madison Ave, Suite 8078 New York, New York 10022 (646) 908-2658

June 30, 2021

Aesther Healthcare Sponsor, LLC 515 Madison Ave, Suite 8078 New York, New York 10022

RE: Securities Subscription Agreement

Ladies and Gentlemen:

This agreement (the "<u>Agreement</u>") is entered into on June 30, 2021 by and between Aesther Healthcare Sponsor, LLC, a Delaware limited liability company (the "<u>Subscriber</u>" or "<u>you</u>"), and Aesther Healthcare Acquisition Corp., a Delaware corporation (the "<u>Company</u>," "<u>we</u>" or "<u>us</u>"). Pursuant to the terms hereof, the Company hereby accepts the offer the Subscriber has made to purchase **2,875,000** shares of the Company's Class B common stock, \$0.0001 par value per share (the "<u>Shares</u>"), up to **375,000** of which are subject to forfeiture by you if the underwriters of the initial public offering ("<u>IPO</u>") of units ("<u>Units</u>") of the Company, do not fully exercise their over-allotment option (the "<u>Over-allotment Option</u>"). The Company and the Subscriber's agreements regarding such Shares are as follows:

1. Purchase of Securities.

1.1. <u>Purchase of Shares.</u> For the sum of \$25,000 (the "<u>Purchase Price</u>"), which the Company acknowledges receiving in cash, the Company hereby issues the Shares to the Subscriber, and the Subscriber hereby purchases the Shares from the Company, subject to forfeiture, on the terms and subject to the conditions set forth in this Agreement. Concurrently with the Subscriber's execution of this Agreement, the Company shall, at its option, deliver to the Subscriber a certificate registered in the Subscriber's name representing the shares (the "<u>Original Certificate</u>"), or effect such delivery in book-entry form.

2. Representations, Warranties and Agreements.

- 2.1. <u>Subscriber's Representations, Warranties and Agreements.</u> To induce the Company to issue the Shares to the Subscriber, the Subscriber hereby represents and warrants to the Company and agrees with the Company as follows:
- 2.1.1. No Conflicts. The execution, delivery and performance of this Agreement and the consummation by the Subscriber of the transactions contemplated hereby do not violate, conflict with or constitute a default under (i) the formation and governing documents of the Subscriber, (ii) any agreement, indenture or instrument to which the Subscriber is a party or (iii) any law, statute, rule or regulation to which the Subscriber is subject, or any agreement, order, judgment or decree to which the Subscriber is subject.
- 2.1.2. <u>Organization and Authority.</u> The Subscriber is a Delaware limited liability company, validly existing and in good standing under the laws of Delaware and possesses all requisite power and authority necessary to carry out the transactions contemplated by this Agreement. Upon execution and delivery by you, this Agreement is a legal, valid and binding agreement of Subscriber, enforceable against Subscriber in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, fraudulent conveyance or similar laws affecting the enforcement of creditors' rights generally and subject to general principles of equity (regardless of whether enforcement is sought in a proceeding at law or in equity).

- 2.1.3. Experience, Financial Capability and Suitability. Subscriber is: (i) sophisticated in financial matters and is able to evaluate the risks and benefits of the investment in the Shares and (ii) able to bear the economic risk of its investment in the Shares for an indefinite period of time because the Shares have not been registered under the Securities Act (as defined below) and therefore cannot be sold unless subsequently registered under the Securities Act or an exemption from such registration is available. Subscriber is capable of evaluating the merits and risks of its investment in the Company and has the capacity to protect its own interests. Subscriber must bear the economic risk of this investment until the Shares are sold pursuant to: (i) an effective registration statement under the Securities Act or (ii) an exemption from registration available with respect to such sale. Subscriber is able to bear the economic risks of an investment in the Shares and to afford a complete loss of Subscriber's investment in the Shares.
- 2.1.4. <u>Regulation D Offering.</u> Subscriber represents that it is an "accredited investor" as such term is defined in Rule 501(a) of Regulation D under the Securities Act of 1933, as amended (the "<u>Securities Act</u>") and acknowledges the sale contemplated hereby is being made in reliance on a private placement exemption to "accredited investors" within the meaning of Section 501(a) of Regulation D under the Securities Act or similar exemptions under state law.
- 2.1.5. <u>Investment Purposes.</u> The Subscriber is purchasing the Shares solely for investment purposes, for the Subscriber's own account and not for the account or benefit of any other person, and not with a view towards the distribution or dissemination thereof. The Subscriber did not decide to enter into this Agreement as a result of any general solicitation or general advertising within the meaning of Rule 502 under the Securities Act.
- 2.1.6. Restrictions on Transfer; Shell Company. Subscriber understands the Shares are being offered in a transaction not involving a public offering within the meaning of the Securities Act. Subscriber understands the Shares will be "restricted securities" within the meaning of Rule 144(a)(3) under the Securities Act, and Subscriber understands that the certificates or book-entries representing the Shares will contain a legend in respect of such restrictions. If in the future the Subscriber decides to offer, resell, pledge or otherwise transfer the Shares, such Shares may be offered, resold, pledged or otherwise transferred only pursuant to: (i) registration under the Securities Act, or (ii) an available exemption from registration. Subscriber agrees that if any transfer of its Shares or any interest therein is proposed to be made, as a condition precedent to any such transfer, Subscriber may be required to deliver to the Company an opinion of counsel satisfactory to the Company. Absent registration or an exemption, the Subscriber agrees not to resell the Shares. Subscriber further acknowledges that because the Company is a shell company, Rule 144 may not be available to the Subscriber for the resale of the Shares until one year following consummation of the initial business combination of the Company, despite technical compliance with the requirements of Rule 144 and the release or waiver of any contractual transfer restrictions.
- 2.2. <u>Company's Representations, Warranties and Agreements.</u> To induce the Subscriber to purchase the Shares, the Company hereby represents and warrants to the Subscriber and agrees with the Subscriber as follows:
- 2.2.1. <u>Organization and Corporate Power.</u> The Company is a Delaware corporation and is qualified to do business in every jurisdiction in which the failure to so qualify would reasonably be expected to have a material adverse effect on the financial condition, operating results or assets of the Company. The Company possesses all requisite corporate power and authority necessary to carry out the transactions contemplated by this Agreement.
- 2.2.2. No Conflicts. The execution, delivery and performance of this Agreement and the consummation by the Company of the transactions contemplated hereby do not violate, conflict with or constitute a default under (i) the certificate of incorporation or bylaws of the Company, (ii) any agreement, indenture or instrument to which the Company is a party or (iii) any law, statute, rule or regulation to which the Company is subject, or any agreement, order, judgment or decree to which the Company is subject.
- 2.2.3. <u>Title to Securities.</u> Upon issuance in accordance with, and payment pursuant to, the terms hereof, the Shares will be duly and validly issued, fully paid and non-assessable. Upon issuance in accordance with, and payment pursuant to, the terms hereof, the Subscriber will have or receive good title to the Shares, free and clear of all liens, claims and encumbrances of any kind, other than (a) transfer restrictions hereunder and other agreements to which the Shares may be subject which have been notified to the Subscriber in writing, (b) transfer restrictions under federal and state securities laws, and (c) liens, claims or encumbrances imposed due to the actions of the Subscriber.

3. Forfeiture of Shares.

- 3.1. <u>Partial or No Exercise of the Over-allotment Option.</u> In the event the Over-allotment Option granted to the underwriters of the IPO is not exercised in full, the Subscriber acknowledges and agrees that it (or, if applicable, it and any transferees of Shares) shall forfeit any and all rights to such number of Shares (up to an aggregate of **375,000** Shares and pro rata based upon the percentage of the Over-allotment Option exercised) such that immediately following such forfeiture, the Subscriber (and all other initial stockholders prior to the IPO, if any) will own an aggregate number of Shares, not including Shares issuable upon exercise of any warrants or any Common Stock purchased by Subscriber in the IPO or in the aftermarket, equal to 20% of the issued and outstanding Shares immediately following the IPO.
- 3.2. <u>Termination of Rights as Stockholder.</u> If any of the Shares are forfeited in accordance with this Section 3, then after such time the Subscriber (or successor in interest), shall no longer have any rights as a holder of such forfeited Shares, and the Company shall take such action as is appropriate to cancel such forfeited Shares.
- 3.3. Share Certificates. In the event an adjustment to the Original Certificates, if any, is required pursuant to this Section 3, then the Subscriber shall return such Original Certificates to the Company or its designated agent as soon as practicable upon its receipt of notice from the Company advising Subscriber of such adjustment, following which a new certificate (the "New Certificate"), if any, shall be issued in such amount representing the adjusted number of Shares held by the Subscriber. The New Certificate, if any, shall be returned to the Subscriber as soon as practicable. Any such adjustment for any uncertificated securities held by the Subscriber shall be made in book-entry form.
- 4. <u>Waiver of Liquidation Distributions</u>; <u>Redemption Rights</u>. In connection with the Shares purchased pursuant to this Agreement, the Subscriber hereby waives any and all right, title, interest or claim of any kind in or to any distributions by the Company from the trust account which will be established for the benefit of the Company's public stockholders and into which substantially all of the proceeds of the IPO will be deposited (the "<u>Trust Account</u>"), in the event of a liquidation of the Company upon the Company's failure to timely complete an initial business combination. For purposes of clarity, in the event the Subscriber purchases Shares in the IPO or in the aftermarket, any additional Shares so purchased shall be eligible to receive any liquidating distributions by the Company. However, in no event will the Subscriber have the right to redeem any Shares issued pursuant to this Agreement into funds held in the Trust Account upon the successful completion of an initial business combination.

5. Restrictions on Transfer.

- 5.1. <u>Securities Law Restrictions.</u> In addition to any restrictions to be contained in a letter agreement (the "<u>Insider Letter</u>") to be dated as of the closing of the IPO by and between Subscriber and the Company, Subscriber agrees not to sell, transfer, pledge, hypothecate or otherwise dispose of all or any part of the Shares unless, prior thereto (a) a registration statement on the appropriate form under the Securities Act and applicable state securities laws with respect to the Shares proposed to be transferred shall then be effective or (b) the Company has received an opinion from counsel reasonably satisfactory to the Company, that such registration is not required because such transaction is exempt from registration under the Securities Act and the rules promulgated by the Securities and Exchange Commission thereunder and with all applicable state securities laws.
 - 5.2. <u>Lock-up</u>. Subscriber acknowledges that the Securities will be subject to lock-up provisions (the "<u>Lock-up</u>") contained in the Insider Letter.
 - 5.3. Restrictive Legends. Any certificates representing the Shares shall have endorsed thereon legends substantially as follows:

"THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY STATE SECURITIES LAWS AND NEITHER THE SECURITIES NOR ANY INTEREST THEREIN MAY BE OFFERED, SOLD, TRANSFERRED, PLEDGED OR OTHERWISE DISPOSED OF EXCEPT PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER SUCH ACT OR SUCH LAWS OR AN EXEMPTION FROM REGISTRATION UNDER SUCH ACT AND SUCH LAWS WHICH, IN THE OPINION OF COUNSEL, IS AVAILABLE."

"THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO A LOCKUP AND MAY NOT BE OFFERED, SOLD, TRANSFERRED, PLEDGED OR OTHERWISE DISPOSED DURING THE TERM OF THE LOCKUP."

- 5.4. <u>Additional Shares or Substituted Securities.</u> In the event of the declaration of a share dividend, the declaration of an extraordinary dividend payable in a form other than Shares, a spin-off, a share split, a recapitalization or a similar transaction affecting the Company's outstanding Shares without receipt of consideration, any new, substituted or additional securities or other property which are by reason of such transaction distributed with respect to any Shares subject to this Section 5 shall immediately be subject to this Section 5 and Section 3. Appropriate adjustments to reflect the distribution of such securities or property shall be made to the number of Shares subject to this Section 5 and Section 3.
- 5.5. <u>Registration Rights.</u> Subscriber acknowledges that the Shares are being purchased pursuant to an exemption from the registration requirements of the Securities Act and will become freely tradable only after certain conditions are met or they are registered pursuant to a registration rights agreement to be entered into with the Company prior to the closing of the IPO (the "Registration Rights Agreement") which shall contain customary registration rights.

6. Other Agreements.

- 6.1. <u>Further Assurances</u>. Subscriber agrees to execute such further instruments and to take such further action as may reasonably be necessary to carry out the intent of this Agreement.
- 6.2. Notices. All notices, statements or other documents which are required or contemplated by this Agreement shall be: (i) in writing and delivered personally or sent by first class registered or certified mail, overnight courier service to the address designated in writing, or (ii) by electronic mail, to the electronic mail address most recently provided to such party or such other electronic mail address as may be designated in writing by such party. Any notice or other communication so transmitted shall be deemed to have been given (w) on the day of delivery, if delivered personally, (x) on the business day following receipt of written confirmation, if sent by electronic transmission, (y) one (1) business day after delivery to an overnight courier service or (z) five (5) days after mailing if sent by mail.
- 6.3. Entire Agreement. This Agreement, together with the Insider Letter and the Registration Rights Agreement, each substantially in the form to be filed as an exhibit to the Registration Statement on Form S-1 associated with the Company's IPO, embodies the entire agreement and understanding between the Subscriber and the Company with respect to the subject matter hereof and supersedes all prior oral or written agreements and understandings relating to the subject matter hereof. No statement, representation, warranty, covenant or agreement of any kind not expressly set forth in this Agreement shall affect, or be used to interpret, change or restrict, the express terms and provisions of this Agreement.
- 6.4. <u>Modifications and Amendments.</u> The terms and provisions of this Agreement may be modified or amended only by written agreement executed by all parties hereto.
- 6.5. Waivers and Consents. The terms and provisions of this Agreement may be waived, or consent for the departure therefrom granted, only by a written document executed by the party entitled to the benefits of such terms or provisions. No such waiver or consent shall be deemed to be or shall constitute a waiver or consent with respect to any other terms or provisions of this Agreement, whether or not similar. Each such waiver or consent shall be effective only in the specific instance and for the purpose for which it was given, and shall not constitute a continuing waiver or consent.
- 6.6. <u>Assignment.</u> The rights and obligations under this Agreement may not be assigned by either party hereto without the prior written consent of the other party.
- 6.7. <u>Benefit.</u> All statements, representations, warranties, covenants and agreements in this Agreement shall be binding on the parties hereto and shall inure to the benefit of the respective successors and permitted assigns of each party hereto. Nothing in this Agreement shall be construed to create any rights or obligations except among the parties hereto, and no person or entity shall be regarded as a third-party beneficiary of this Agreement.

- 6.8. <u>Governing Law.</u> This Agreement and the rights and obligations of the parties hereunder shall be construed in accordance with and governed by the laws of New York applicable to contracts wholly performed within the borders of such state, without giving effect to the conflict of law principles thereof.
- 6.9. Severability. In the event that any court of competent jurisdiction shall determine that any provision, or any portion thereof, contained in this Agreement shall be unreasonable or unenforceable in any respect, then such provision shall be deemed limited to the extent that such court deems it reasonable and enforceable, and as so limited shall remain in full force and effect. In the event that such court shall deem any such provision, or portion thereof, wholly unenforceable, the remaining provisions of this Agreement shall nevertheless remain in full force and effect.
- 6.10. <u>Survival of Representations and Warranties.</u> All representations and warranties made by the parties hereto in this Agreement or in any other agreement, certificate or instrument provided for or contemplated hereby, shall survive the execution and delivery hereof and any investigations made by or on behalf of the parties.
- 6.11. No Broker or Finder. Each of the parties hereto represents and warrants to the other that no broker, finder or other financial consultant has acted on its behalf in connection with this Agreement or the transactions contemplated hereby in such a way as to create any liability on the other. Each of the parties hereto agrees to indemnify and save the other harmless from any claim or demand for commission or other compensation by any broker, finder, financial consultant or similar agent claiming to have been employed by or on behalf of such party and to bear the cost of legal expenses incurred in defending against any such claim.
- 6.14. Counterparts. This Agreement may be executed in one or more counterparts, all of which when taken together shall be considered one and the same agreement and shall become effective when counterparts have been signed by each party and delivered to the other party, it being understood that both parties need not sign the same counterpart. In the event that any signature is delivered by facsimile transmission or any other form of electronic delivery, such signature shall create a valid and binding obligation of the party executing (or on whose behalf such signature is executed) with the same force and effect as if such signature page were an original thereof.
- 7. <u>Voting and Tender of Shares.</u> Subscriber agrees to vote the Shares in favor of an initial business combination that the Company negotiates and submits for approval to the Company's stockholders and shall not seek redemption with respect to such Shares. Additionally, the Subscriber agrees not to tender any Shares in connection with a tender offer presented to the Company's stockholders in connection with an initial business combination negotiated by the Company.

[Signature Page Follows]

If the foregoing accurately sets forth our understanding and agreement, please sign the enclosed copy of this Agreement and return it to us.

Very truly yours,

Aesther Healthcare Acquisition Corp.

By: /s/ Suren Ajjarapu

Name: Suren Ajjarapu
Title: Chief Executive Officer

Accepted and agreed as of the date first written above.

Aesther Healthcare Sponsor, LLC

By: /s/ Suren Ajjarapu

Name: Suren Ajjarapu
Title: Managing Member

[Signature Page to Securities Subscription Agreement]

PRIVATE PLACEMENT WARRANTS PURCHASE AGREEMENT

THIS PRIVATE PLACEMENT WARRANTS PURCHASE AGREEMENT, effective as of [], 2021 (as it may from time to time be amended, this "Agreement"), is entered into by and between Aesther Healthcare Acquisition Corp., a Delaware corporation (the "Company") and Aesther Healthcare Acquisition Sponsors LLC, a Delaware limited liability company (the "Purchaser").

WHEREAS:

The Company intends to consummate an initial public offering of the Company's units (the "**Public Offering**"), each unit consisting of one share of Class A common stock of the Company, par value \$0.0001 per share (each, a "**Share**"), and one half of one redeemable warrant;

Each whole warrant entitles the holder to purchase one Share at an exercise price of \$11.50 per Share; and

The Purchaser has agreed to purchase an aggregate of 5,261,000 warrants (or 5,711,000 warrants if the over-allotment option in connection with the Public Offering is exercised in full) at a price of \$1.00 per warrant (the "**Private Placement Warrants**"), each Private Placement Warrant entitling the holder to purchase one Share at an exercise price of \$11.50 per Share.

NOW THEREFORE, in consideration of the mutual promises contained in this Agreement and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties to this Agreement hereby, intending legally to be bound, agree as follows:

AGREEMENT

Section 1. Authorization, Purchase and Sale; Terms of the Private Placement Warrants.

A. <u>Authorization of the Private Placement Warrants</u>. The Company has duly authorized the issuance and sale of the Private Placement Warrants to the Purchaser.

B. Purchase and Sale of the Private Placement Warrants.

- (i) Simultaneously with the consummation of the Public Offering or on such earlier time and date as may be mutually agreed by the Purchaser and the Company (the "Initial Closing Date"), the Company shall issue and sell to the Purchaser, and the Purchaser shall purchase from the Company, an aggregate of 5,261,000 Private Placement Warrants at a price of \$1.00 per warrant for an aggregate purchase price of \$5,261,000 (the "Purchase Price"). Purchaser shall pay the Purchase Price by wire transfer of immediately available funds to the trust account (the "Trust Account") maintained by Continental Stock Transfer & Trust Company, acting as trustee ("Continental"), at least one (1) business day prior to the date of effectiveness (the "Effective Date") of the registration statement relating to the Public Offering (the "Registration Statement"). On the Initial Closing Date, upon the payment by the Purchaser of the Purchase Price, the Company, at its option, shall deliver a certificate evidencing the Private Placement Warrants purchased on such date duly registered in the Purchaser's name to the Purchaser or effect such delivery in book-entry form.
- (ii) Simultaneously with the consummation of the closing of the over-allotment option in connection with the Public Offering (the "Over-Allotment Option") or on such earlier time and date as may be mutually agreed by the Purchaser and the Company (each such date, an "Over-Allotment Closing Date," and each Over-Allotment Closing Date (if any) and the Initial Closing Date being sometimes referred to herein as a "Closing Date"), Purchaser shall purchase up to an additional 450,000 Private Placement Warrants (the "Additional Warrants"), in the same proportion as the amount of the option that is so exercised, and simultaneously with such purchase of Additional Warrants, as payment in full for the Additional Warrants being purchased hereunder, and at least one (1) business day prior to the Over-Allotment Closing Date, Purchaser shall pay \$1.00 per Additional Warrant, up to an aggregate amount of \$450,000 (the "Over-Allotment Purchase Price"), by wire transfer of immediately available funds or by such other method as may be reasonably acceptable to the Company, to the Trust Account. On the Over-Allotment Closing Date, upon the payment by the Purchaser of the Over-Allotment Purchase Price, the Company, at its option, shall deliver a certificate evidencing the Additional Warrants purchased on such date duly registered in the Purchaser's name to the Purchaser or effect such delivery in book-entry form.

C. Terms of the Private Placement Warrants.

- (i) Each Private Placement Warrant shall have the terms set forth in a Warrant Agreement to be entered into by the Company and Continental in connection with the Public Offering (the "Warrant Agreement"). Such terms include the fact that the Private Placement Warrants shall not be transferable, assignable or salable until 30 days after the completion of an initial business combination, subject to certain exceptions set forth in the Warrant Agreement.
- (ii) On or prior to the Effective Date, the Company and the Purchaser shall enter into a registration rights agreement (the "**Registration Rights Agreement**") pursuant to which the Company will grant certain registration rights to the Purchaser relating to the Private Placement Warrants and the Shares underlying the Private Placement Warrants.
- (iii) **Representations and Warranties of the Company.** As a material inducement to the Purchaser to enter into this Agreement and purchase the Private Placement Warrants, the Company hereby represents and warrants to the Purchaser (which representations and warranties shall survive the Closing Date) that:
- D. <u>Incorporation and Corporate Power</u>. The Company is a corporation duly incorporated, validly existing and in good standing under the laws of the State of Delaware and is qualified to do business in every jurisdiction in which the failure to so qualify would reasonably be expected to have a material adverse effect on the financial condition, operating results or assets of the Company. The Company possesses all requisite corporate power and authority necessary to carry out the transactions contemplated by this Agreement and the Warrant Agreement.

E. Authorization; No Breach.

- (i) The execution, delivery and performance of this Agreement and the Warrant Agreement have been duly authorized by the Company as of the Closing Date. This Agreement constitutes the valid and binding obligation of the Company, enforceable in accordance with its terms, subject to bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other laws of general applicability relating to or affecting creditors' rights and to general equitable principles (whether considered in a proceeding in equity or law). Upon issuance in accordance with, and payment pursuant to, the terms of the Warrant Agreement and this Agreement, the Private Placement Warrants will constitute valid and binding obligations of the Company, enforceable in accordance with their terms.
- (ii) The execution and delivery by the Company of this Agreement and the Warrant Agreement, the issuance and sale of the Private Placement Warrants, the issuance of the Shares upon exercise of the Private Placement Warrants and the fulfillment, of and compliance with, the respective terms hereof and thereof by the Company, do not and will not as of the Closing Date (a) conflict with or result in a breach of the terms, conditions or provisions of, (b) constitute a default under, (c) result in the creation of any lien, security interest, charge or encumbrance upon the Company's share capital or assets under, (d) result in a violation of, or (e) require any authorization, consent, approval, exemption or other action by or notice or declaration to, or filing with, any court or administrative or governmental body or agency pursuant to the amended and restated certificate of incorporation of the Company (in effect on the date hereof or as may be amended prior to completion of the contemplated Public Offering), or any material law, statute, rule or regulation to which the Company is subject, or any agreement, order, judgment or decree to which the Company is subject, except for any filings required after the date hereof under federal or state securities laws.

- F. <u>Title to Securities</u>. Upon issuance in accordance with, and payment pursuant to, the terms hereof, the Warrant Agreement, the Shares issuable upon exercise of the Private Placement Warrants will be duly and validly issued as fully paid and nonassessable. On the date of issuance of the Private Placement Warrants, the Shares issuable upon exercise of the Private Placement Warrants shall have been reserved for issuance. Upon issuance in accordance with, and payment pursuant to, the terms hereof and the Warrant Agreement, the Purchaser will have good title to the Private Placement Warrants and the Shares issuable upon exercise of such Private Placement Warrants, free and clear of all liens, claims and encumbrances of any kind, other than (i) transfer restrictions hereunder and under the other agreements contemplated hereby, (ii) transfer restrictions under federal and state securities laws, and (iii) liens, claims or encumbrances imposed due to the actions of the Purchaser.
- G. <u>Governmental Consents</u>. No permit, consent, approval or authorization of, or declaration to or filing with, any governmental authority is required in connection with the execution, delivery and performance by the Company of this Agreement or the consummation by the Company of any other transactions contemplated hereby.
- **Section 2. Representations and Warranties of the Purchaser.** As a material inducement to the Company to enter into this Agreement and issue and sell the Private Placement Warrants to the Purchaser, the Purchaser hereby represents and warrants to the Company (which representations and warranties shall survive the Closing Date) that:
- A. <u>Organization and Requisite Authority</u>. The Purchaser possesses all requisite power and authority necessary to carry out the transactions contemplated by this Agreement.

B. Authorization; No Breach.

- (i) This Agreement constitutes a valid and binding obligation of the Purchaser, enforceable in accordance with its terms, subject to bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other laws of general applicability relating to or affecting creditors' rights and to general equitable principles (whether considered in a proceeding in equity or law).
- (ii) The execution and delivery by the Purchaser of this Agreement and the fulfillment of and compliance with the terms hereof by the Purchaser does not and shall not as of each Closing Date (a) conflict with or result in a breach by the Purchaser of the terms, conditions or provisions of, (b) constitute a default under, (c) result in the creation of any lien, security interest, charge or encumbrance upon the Purchaser's equity or assets under, (d) result in a violation of, or (e) require authorization, consent, approval, exemption or other action by or notice or declaration to, or filing with, any court or administrative or governmental body or agency pursuant to the Purchaser's organizational documents in effect on the date hereof or as may be amended prior to completion of the contemplated Public Offering, or any material law, statute, rule or regulation to which the Purchaser is subject, or any agreement, instrument, order, judgment or decree to which the Purchaser is subject, except for any filings required after the date hereof under federal or state securities laws.

C. Investment Representations.

- (i) The Purchaser is acquiring the Private Placement Warrants and, upon exercise of the Private Placement Warrants, the Shares issuable upon such exercise (collectively, the "Securities"), for the Purchaser's own account, for investment purposes only and not with a view towards, or for resale in connection with, any public sale or distribution thereof.
- (ii) The Purchaser is an "accredited investor" as such term is defined in Rule 501(a)(3) of Regulation D, and the Purchaser has not experienced a disqualifying event as enumerated pursuant to Rule 506(d) of Regulation D under the Securities Act of 1933, as amended (the "Securities Act").
- (iii) The Purchaser understands that the Securities are being offered and will be sold to it in reliance on specific exemptions from the registration requirements of the United States federal and state securities laws and that the Company is relying upon the truth and accuracy of, and the Purchaser's compliance with, the representations and warranties of the Purchaser set forth herein in order to determine the availability of such exemptions and the eligibility of the Purchaser to acquire such Securities.

- (iv) The Purchaser did not decide to enter into this Agreement as a result of any general solicitation or general advertising within the meaning of Rule 502(c) under the Securities Act.
- (v) The Purchaser has been furnished with all materials relating to the business, finances and operations of the Company and materials relating to the offer and sale of the Securities which have been requested by the Purchaser. The Purchaser has been afforded the opportunity to ask questions of the executive officers and directors of the Company. The Purchaser understands that its investment in the Securities involves a high degree of risk and it has sought such accounting, legal and tax advice as it has considered necessary to make an informed investment decision with respect to the acquisition of the Securities.
- (vi) The Purchaser understands that no United States federal or state agency or any other government or governmental agency has passed on or made any recommendation or endorsement of the Securities or the fairness or suitability of the investment in the Securities by the Purchaser nor have such authorities passed upon or endorsed the merits of the offering of the Securities.
- (vii) The Purchaser understands that: (a) the Securities have not been and are not being registered under the Securities Act or any state securities laws, and may not be offered for sale, sold, assigned or transferred unless (1) subsequently registered thereunder or (2) sold in reliance on an exemption therefrom; and (b) except as specifically set forth in the Registration Rights Agreement, neither the Company nor any other person is under any obligation to register the Securities under the Securities Act or any state securities laws or to comply with the terms and conditions of any exemption thereunder. The Private Placement Warrants will bear a legend and appropriate "stop transfer" instructions (or an appropriate notation if the warrants are issued in book entry form) relating to the foregoing. The Purchaser further understands that the Securities and Exchange Commission (the "SEC") has taken the position that promoters or affiliates of a blank check company and their transferees, both before and after an initial business combination, are deemed to be "underwriters" under the Securities Act when reselling the securities of a blank check company. Based on that position, Rule 144 adopted pursuant to the Securities Act would not be available for resale transactions of the Securities until the one-year anniversary following consummation of an initial business combination despite technical compliance with the requirements of such Rule.
- (viii) The Purchaser has such knowledge and experience in financial and business matters, knows of the high degree of risk associated with investments in the securities of companies in the development stage such as the Company, is capable of evaluating the merits and risks of an investment in the Securities and is able to bear the economic risk of an investment in the Securities in the amount contemplated hereunder for an indefinite period of time. The Purchaser has adequate means of providing for its current financial needs and contingencies and will have no current or anticipated future needs for liquidity which would be jeopardized by the investment in the Securities. The Purchaser can afford a complete loss of its investment in the Securities.
- (ix) The Purchaser understands that the Private Placement Warrants shall bear the legend substantially in the form set forth in the Warrant Agreement.
- **Section 3. Conditions of the Purchaser's Obligations.** The obligations of the Purchaser to purchase and pay for the Private Placement Warrants are subject to the fulfillment, on or before the Closing Date, of each of the following conditions:
- A. <u>Representations and Warranties</u>. The representations and warranties of the Company contained in Section 2 shall be true and correct at and as of such Closing Date as though then made.
- B. <u>Performance</u>. The Company shall have performed and complied with all agreements, obligations and conditions contained in this Agreement that are required to be performed or complied with by it on or before such Closing Date.

- C. <u>No Injunction</u>. No litigation, statute, rule, regulation, executive order, decree, ruling or injunction shall have been enacted, entered, promulgated or endorsed by or in any court or governmental authority of competent jurisdiction or any self-regulatory organization having authority over the matters contemplated hereby, which prohibits the consummation of any of the transactions contemplated by this Agreement or the Warrant Agreement.
- D. <u>Warrant Agreement and Registration Rights Agreement</u>. The Company shall have entered into the Warrant Agreement, in the form of Exhibit A hereto, and the Registration Rights Agreement, in the form of Exhibit B hereto, in each case on terms satisfactory to the Purchaser.
- **Section 4. Conditions of the Company's Obligations.** The obligations of the Company to the Purchaser under this Agreement are subject to the fulfillment, on or before the Closing Date, of each of the following conditions:
- A. <u>Representations and Warranties</u>. The representations and warranties of the Purchaser contained in Section 3 shall be true and correct at and as of such Closing Date as though then made.
- B. <u>Performance</u>. The Purchaser shall have performed and complied with all agreements, obligations and conditions contained in this Agreement that are required to be performed or complied with by the Purchaser on or before such Closing Date.
- C. <u>No Injunction</u>. No litigation, statute, rule, regulation, executive order, decree, ruling or injunction shall have been enacted, entered, promulgated or endorsed by or in any court or governmental authority of competent jurisdiction or any self-regulatory organization having authority over the matters contemplated hereby, which prohibits the consummation of any of the transactions contemplated by this Agreement or the Warrant Agreement.
 - D. Warrant Agreement. The Company shall have entered into the Warrant Agreement.
- **Section 5. Termination.** This Agreement may be terminated at any time after [______], 20____ upon the election by either the Company or the Purchaser solely as to itself upon written notice to the other party if the closing of the Public Offering does not occur prior to such date.
- Section 6. Survival of Representations and Warranties. All of the representations and warranties contained herein shall survive the Closing Date.
- Section 7. Definitions. Terms used but not otherwise defined in this Agreement shall have the meaning assigned to such terms in the Registration Statement.

Section 8. Miscellaneous.

- A. <u>Successors and Assigns</u>. Except as otherwise expressly provided herein, all covenants and agreements contained in this Agreement by or on behalf of any of the parties hereto shall bind and inure to the benefit of the respective successors of the parties hereto whether so expressed or not. Notwithstanding the foregoing or anything to the contrary herein, the parties may not assign this Agreement without the prior written consent of the other party hereto, other than assignments by the Purchaser to affiliates thereof (including, without limitation, one or more of its members).
- B. <u>Severability</u>. Whenever possible, each provision of this Agreement shall be interpreted in such manner as to be effective and valid under applicable law, but if any provision of this Agreement is held to be prohibited by or invalid under applicable law, such provision shall be ineffective only to the extent of such prohibition or invalidity, without invalidating the remainder of this Agreement.

- C. <u>Counterparts</u>. This Agreement may be executed simultaneously in two or more counterparts, none of which need contain the signatures of more than one party, but all such counterparts taken together shall constitute one and the same agreement. Signatures to this Agreement transmitted via facsimile or e-mail shall be valid and effective to bind the party so signing (including any electronic signature covered by the U.S. federal ESIGN Act of 2000, Uniform Electronic Transactions Act, the Electronic Signatures and Records Act or other applicable law, e.g., www.docusign.com).
- D. <u>Descriptive Headings</u>; <u>Interpretation</u>. The descriptive headings of this Agreement are inserted for convenience only and do not constitute a substantive part of this Agreement. The use of the word "**including**" in this Agreement shall be by way of example rather than by limitation.
- E. <u>Governing Law</u>. This Agreement shall be deemed to be a contract made under the laws of the State of New York and for all purposes shall be construed in accordance with the internal laws of the State of New York, without regard to the conflicts of laws principles thereof.
- F. <u>Amendments</u>. This Agreement may not be amended, modified or waived as to any particular provision, except by a written instrument executed by the parties hereto.

[Signature page follows]

IN WITNESS WHEREOF , the parties hereto have executed this Agreement to be effective as of the date first set forth above.	
	COMPANY:
	Aesther Healthcare Acquisition Corp.
	Ву:
	Name: Suren Ajjarapu
	Title: Chief Executive Officer
	PURCHASER:
	Aesther Healthcare Sponsor, LLC
	By:
	Name: Suren Ajjarapu
	Title: Authorized Signatory
	7

EXHIBIT A

Warrant Agreement

EXHIBIT B

Registration and Shareholder Rights Agreement

INDEMNIFICATION AGREEMENT

THIS INDEMNIFICATION AGREEMENT (the "Agreement") is made and entered into as of _	, 2021, between Aesther
Healthcare Acquisition Corp., a Delaware corporation (the "Company"), and	_ ("Indemnitee").

WITNESSETH THAT:

WHEREAS, highly competent persons have become more reluctant to serve corporations as directors or in other capacities unless they are provided with adequate protection through insurance or adequate indemnification against inordinate risks of claims and actions against them arising out of their service to and activities on behalf of the corporation;

WHEREAS, the Board of Directors of the Company (the "Board") has determined that, in order to attract and retain qualified individuals, the Company will attempt to maintain on an ongoing basis, at its sole expense, liability insurance to protect persons serving the Company and its subsidiaries from certain liabilities. Although the furnishing of such insurance has been a customary and widespread practice among United States-based corporations and other business enterprises, the Company believes that, given current market conditions and trends, such insurance may be available to it in the future only at higher premiums and with more exclusions. At the same time, directors, officers, and other persons in service to corporations or business enterprises are being increasingly subjected to expensive and time-consuming litigation relating to, among other things, matters that traditionally would have been brought only against the Company or business enterprise itself. The Bylaws and Certificate of Incorporation of the Company contemplate indemnification of the officers and directors of the Company. Indemnitee may also be entitled to indemnification pursuant to the General Corporation Law of the State of Delaware ("DGCL"). The Bylaws, Certificate of Incorporation and the DGCL expressly provide that the indemnification provisions set forth therein are not exclusive, and thereby contemplate that contracts may be entered into between the Company and members of the board of directors, officers and other persons with respect to indemnification;

WHEREAS, the uncertainties relating to such insurance and to indemnification have increased the difficulty of attracting and retaining such persons;

WHEREAS, the Board has determined that the increased difficulty in attracting and retaining such persons is detrimental to the best interests of the Company's stockholders and that the Company should act to assure such persons that there will be increased certainty of such protection in the future;

WHEREAS, it is reasonable, prudent and necessary for the Company contractually to obligate itself to indemnify, and to advance expenses on behalf of, such persons to the fullest extent permitted by applicable law so that they will serve or continue to serve the Company free from undue concern that they will not be so indemnified;

WHEREAS, this Agreement is a supplement to and in furtherance of the Bylaws and Certificate of Incorporation of the Company and any resolutions adopted pursuant thereto or pursuant to the DGCL, and shall not be deemed a substitute therefor, nor to diminish or abrogate any rights of Indemnitee thereunder;

WHEREAS, Indemnitee does not regard the protection available under the Company's Bylaws and Certificate of Incorporation and insurance as adequate in the present circumstances, and may not be willing to serve as an officer or director without adequate protection, and the Company desires Indemnitee to serve in such capacity. Indemnitee is willing to serve, continue to serve and to take on additional service for or on behalf of the Company on the condition that he be so indemnified; and

NOW, THEREFORE, in consideration of Indemnitee's agreement to serve as a director or officer from and after the date hereof, the parties hereto agree as follows:

1. <u>Indemnity of Indemnitee</u>. The Company hereby agrees to hold harmless and indemnify Indemnitee to the fullest extent permitted by law, as such may be amended from time to time. In furtherance of the foregoing indemnification, and without limiting the generality thereof:

- (a) <u>Proceedings Other Than Proceedings by or in the Right of the Company.</u> Indemnitee shall be entitled to the rights of indemnification provided in this <u>Section I(a)</u> if, by reason of his or her Corporate Status (as hereinafter defined), the Indemnitee is, or is threatened to be made, a party to or participant in any Proceeding (as hereinafter defined) other than a Proceeding by or in the right of the Company. Pursuant to this <u>Section 1(a)</u>, Indemnitee shall be indemnified against all Expenses (as hereinafter defined), judgments, penalties, fines and amounts paid in settlement actually and reasonably incurred by him or her, or on his or her behalf, in connection with such Proceeding or any claim, issue or matter therein, if the Indemnitee acted in good faith and in a manner the Indemnitee reasonably believed to be in or not opposed to the best interests of the Company, and with respect to any criminal Proceeding, had no reasonable cause to believe the Indemnitee's conduct was unlawful.
- (b) <u>Proceedings by or in the Right of the Company</u>. Indemnitee shall be entitled to the rights of indemnification provided in this <u>Section 1(b)</u> if, by reason of his or her Corporate Status, the Indemnitee is, or is threatened to be made, a party to or participant in any Proceeding brought by or in the right of the Company. Pursuant to this <u>Section 1(b)</u>, Indemnitee shall be indemnified against all Expenses actually and reasonably incurred by the Indemnitee, or on the Indemnitee's behalf, in connection with such Proceeding if the Indemnitee acted in good faith and in a manner the Indemnitee reasonably believed to be in or not opposed to the best interests of the Company; provided, however, if applicable law so provides, no indemnification against such Expenses shall be made in respect of any claim, issue or matter in such Proceeding as to which Indemnitee shall have been adjudged to be liable to the Company unless and to the extent that the Court of Chancery of the State of Delaware shall determine that such indemnification may be made.
- (c) <u>Indemnification for Expenses of a Party Who is Wholly or Partly Successful</u>. Notwithstanding any other provision of this Agreement, to the extent that Indemnitee is, by reason of his or her Corporate Status, a party to and is successful, on the merits or otherwise, in any Proceeding, he shall be indemnified to the maximum extent permitted by law, as such may be amended from time to time, against all Expenses actually and reasonably incurred by him or her or on his or her behalf in connection therewith. If Indemnitee is not wholly successful in such Proceeding but is successful, on the merits or otherwise, as to one or more but less than all claims, issues or matters in such Proceeding, the Company shall indemnify Indemnitee against all Expenses actually and reasonably incurred by him or her or on his or her behalf in connection with each successfully resolved claim, issue or matter. For purposes of this Section and without limitation, the termination of any claim, issue or matter in such a Proceeding by dismissal, with or without prejudice, shall be deemed to be a successful result as to such claim, issue or matter.
- 2. <u>Additional Indemnity</u>. In addition to, and without regard to any limitations on, the indemnification provided for in <u>Section 1</u> of this Agreement, the Company shall and hereby does indemnify and hold harmless Indemnitee against all Expenses, judgments, penalties, fines and amounts paid in settlement actually and reasonably incurred by him or her or on his or her behalf if, by reason of his or her Corporate Status, he or she is, or is threatened to be made, a party to or participant in any Proceeding (including a Proceeding by or in the right of the Company), including, without limitation, all liability arising out of the negligence or active or passive wrongdoing of Indemnitee. The only limitation that shall exist upon the Company's obligations pursuant to this Agreement shall be that the Company shall not be obligated to make any payment to Indemnitee that is finally determined (under the procedures, and subject to the presumptions, set forth in <u>Sections 6</u> and <u>7</u> hereof) to be unlawful.

3. Contribution.

(a) Whether or not the indemnification provided in Sections 1 and $\frac{1}{2}$ hereof is available, in respect of any threatened, pending or completed action, suit or proceeding in which the Company is jointly liable with Indemnitee (or would be if joined in such action, suit or proceeding), the Company shall pay, in the first instance, the entire amount of any judgment or settlement of such action, suit or proceeding without requiring Indemnitee to contribute to such payment and the Company hereby waives and relinquishes any right of contribution it may have against Indemnitee. The Company shall not enter into any settlement of any action, suit or proceeding in which the Company is jointly liable with Indemnitee (or would be if joined in such action, suit or proceeding) unless such settlement provides for a full and final release of all claims asserted against Indemnitee.

- (b) Without diminishing or impairing the obligations of the Company set forth in the preceding subparagraph, if, for any reason, Indemnitee shall elect or be required to pay all or any portion of any judgment or settlement in any threatened, pending or completed action, suit or proceeding in which the Company is jointly liable with Indemnitee (or would be if joined in such action, suit or proceeding), the Company shall contribute to the amount of Expenses, judgments, fines and amounts paid in settlement actually and reasonably incurred and paid or payable by Indemnitee in proportion to the relative benefits received by the Company and all officers, directors or employees of the Company, other than Indemnitee, who are jointly liable with Indemnitee (or would be if joined in such action, suit or proceeding), on the one hand, and Indemnitee, on the other hand, from the transaction from which such action, suit or proceeding arose; provided, however, that the proportion determined on the basis of relative benefit may, to the extent necessary to conform to law, be further adjusted by reference to the relative fault of the Company and all officers, directors or employees of the Company other than Indemnitee who are jointly liable with Indemnitee (or would be if joined in such action, suit or proceeding), on the one hand, and Indemnitee, on the other hand, in connection with the events that resulted in such expenses, judgments, fines or settlement amounts, as well as any other equitable considerations which the Law may require to be considered. The relative fault of the Company and all officers, directors or employees of the Company, other than Indemnitee, who are jointly liable with Indemnitee (or would be if joined in such action, suit or proceeding), on the one hand, and Indemnitee, on the other hand, shall be determined by reference to, among other things, the degree to which their actions were motivated by intent to gain personal profit or advantage, the degree to which their conduct is active or passive.
- (c) The Company hereby agrees to fully indemnify and hold Indemnitee harmless from any claims of contribution which may be brought by officers, directors or employees of the Company, other than Indemnitee, who may be jointly liable with Indemnitee.
- (d) To the fullest extent permissible under applicable law, if the indemnification provided for in this Agreement is unavailable to Indemnitee for any reason whatsoever, the Company, in lieu of indemnifying Indemnitee, shall contribute to the amount incurred by Indemnitee, whether for judgments, fines, penalties, excise taxes, amounts paid or to be paid in settlement and/or for Expenses, in connection with any claim relating to an indemnifiable event under this Agreement, in such proportion as is deemed fair and reasonable in light of all of the circumstances of such Proceeding in order to reflect (i) the relative benefits received by the Company and Indemnitee as a result of the event(s) and/or transaction(s) giving cause to such Proceeding; and/or (ii) the relative fault of the Company (and its directors, officers, employees and agents) and Indemnitee in connection with such event(s) and/or transaction(s).
- 4. <u>Indemnification for Expenses of a Witness</u>. Notwithstanding any other provision of this Agreement, to the extent that Indemnitee is, by reason of his or her Corporate Status, a witness, or is made (or asked to) respond to discovery requests, in any Proceeding to which Indemnitee is not a party, he or she shall be indemnified against all Expenses actually and reasonably incurred by him or her or on his or her behalf in connection therewith.
- 5. <u>Advancement of Expenses</u>. Notwithstanding any other provision of this Agreement, the Company shall advance all Expenses incurred by or on behalf of Indemnitee in connection with any Proceeding by reason of Indemnitee's Corporate Status within thirty (30) days after the receipt by the Company of a statement or statements from Indemnitee requesting such advance or advances from time to time, whether prior to or after final disposition of such Proceeding. Such statement or statements shall reasonably evidence the Expenses incurred by Indemnitee and shall include or be preceded or accompanied by a written undertaking by or on behalf of Indemnitee to repay any Expenses advanced if it shall ultimately be determined that Indemnitee is not entitled to be indemnified against such Expenses. Any advances and undertakings to repay pursuant to this <u>Section 5</u> shall be unsecured and interest free.
- 6. <u>Procedures and Presumptions for Determination of Entitlement to Indemnification</u>. It is the intent of this Agreement to secure for Indemnitee rights of indemnity that are as favorable as may be permitted under the DGCL and public policy of the State of Delaware. Accordingly, the parties agree that the following procedures and presumptions shall apply in the event of any question as to whether Indemnitee is entitled to indemnification under this Agreement:
- (a) To obtain indemnification under this Agreement, Indemnitee shall submit to the Company a written request, including therein or therewith such documentation and information as is reasonably available to Indemnitee and is reasonably necessary to determine whether and to what extent Indemnitee is entitled to indemnification. The Secretary of the Company shall, promptly upon receipt of such a request for indemnification, advise the Board of Directors in writing that Indemnitee has requested indemnification. Notwithstanding the foregoing, any failure of Indemnitee to provide such a request to the Company, or to provide such a request in a timely fashion, shall not relieve the Company of any liability that it may have to Indemnitee unless, and to the extent that, such failure actually and materially prejudices the interests of the Company.

- (b) Upon written request by Indemnitee for indemnification pursuant to the first sentence of Section 6(a) hereof, a determination with respect to Indemnitee's entitlement thereto shall be made in the specific case by one of the following four methods, which shall be at the election of the board: (1) by a majority vote of the disinterested directors, even though less than a quorum, (2) by a committee of disinterested directors designated by a majority vote of the disinterested directors, even though less than a quorum, (3) if there are no disinterested directors or if the disinterested directors so direct, by independent legal counsel in a written opinion to the Board of Directors, a copy of which shall be delivered to the Indemnitee, or (4) if so directed by the Board of Directors, by the stockholders of the Company. For purposes hereof, disinterested directors are those members of the board of directors of the Company who are not parties to the action, suit or proceeding in respect of which indemnification is sought by Indemnitee.
- (c) If the determination of entitlement to indemnification is to be made by Independent Counsel pursuant to Section 6(b) hereof, the Independent Counsel shall be selected as provided in this Section 6(c). The Independent Counsel shall be selected by the Board of Directors. Indemnitee may, within 10 days after such written notice of selection shall have been given, deliver to the Company a written objection to such selection; provided, however, that such objection may be asserted only on the ground that the Independent Counsel so selected does not meet the requirements of "Independent Counsel" as defined in Section 13 of this Agreement, and the objection shall set forth with particularity the factual basis of such assertion. Absent a proper and timely objection, the person so selected shall act as Independent Counsel. If a written objection is made and substantiated, the Independent Counsel selected may not serve as Independent Counsel unless and until such objection is withdrawn or a court has determined that such objection is without merit. If, within 20 days after submission by Indemnitee of a written request for indemnification pursuant to Section 6(a) hereof, no Independent Counsel shall have been selected and not objected to, either the Company or Indemnitee may petition the Court of Chancery of the State of Delaware or other court of competent jurisdiction for resolution of any objection which shall have been made by the Indemnitee to the Company's selection of Independent Counsel and/or for the appointment as Independent Counsel of a person selected by the court or by such other person as the court shall designate, and the person with respect to whom all objections are so resolved or the person so appointed shall act as Independent Counsel under Section 6(b) hereof. The Company shall pay any and all reasonable fees and expenses of Independent Counsel incurred by such Independent Counsel in connection with acting pursuant to Section 6(b) hereof, and the Company shall pay all reasonable fees and e
- (d) In making a determination with respect to entitlement to indemnification hereunder, the person or persons or entity making such determination shall presume that Indemnitee is entitled to indemnification under this Agreement. Anyone seeking to overcome this presumption shall have the burden of proof and the burden of persuasion by clear and convincing evidence. Neither the failure of the Company (including by its directors or independent legal counsel) to have made a determination prior to the commencement of any action pursuant to this Agreement that indemnification is proper in the circumstances because Indemnitee has met the applicable standard of conduct, nor an actual determination by the Company (including by its directors or independent legal counsel) that Indemnitee has not met such applicable standard of conduct, shall be a defense to the action or create a presumption that Indemnitee has not met the applicable standard of conduct.
- (e) Indemnitee shall be deemed to have acted in good faith if Indemnitee's action is based on the records or books of account of the Enterprise, including financial statements, or on information supplied to Indemnitee by the officers of the Enterprise (as hereinafter defined) in the course of their duties, or on the advice of legal counsel for the Enterprise or on information or records given or reports made to the Enterprise by an independent certified public accountant or by an appraiser or other expert selected with reasonable care by the Enterprise. In addition, the knowledge and/or actions, or failure to act, of any director, officer, agent or employee of the Enterprise shall not be imputed to Indemnitee for purposes of determining the right to indemnification under this Agreement. Whether or not the foregoing provisions of this Section 6(e) are satisfied, it shall in any event be presumed that Indemnitee has at all times acted in good faith and in a manner he reasonably believed to be in or not opposed to the best interests of the Company. Anyone seeking to overcome this presumption shall have the burden of proof and the burden of persuasion by clear and convincing evidence.

- (f) If the person, persons or entity empowered or selected under Section 6 to determine whether Indemnitee is entitled to indemnification shall not have made a determination within sixty (60) days after receipt by the Company of the request therefor, the requisite determination of entitlement to indemnification shall be deemed to have been made and Indemnitee shall be entitled to such indemnification absent (i) a misstatement by Indemnitee of a material fact, or an omission of a material fact necessary to make Indemnitee's statement not materially misleading, in connection with the request for indemnification, or (ii) a prohibition of such indemnification under applicable law; provided, however, that such 60-day period may be extended for a reasonable time, not to exceed an additional thirty (30) days, if the person, persons or entity making such determination with respect to entitlement to indemnification in good faith requires such additional time to obtain or evaluate documentation and/or information relating thereto; and provided, further, that the foregoing provisions of this Section 6(g) shall not apply if the determination of entitlement to indemnification is to be made by the stockholders pursuant to Section 6(b) of this Agreement and if (A) within fifteen (15) days after receipt by the Company of the request for such determination, the Board of Directors or the Disinterested Directors, if appropriate, resolve to submit such determination to the stockholders for their consideration at an annual meeting thereof to be held within seventy-five (75) days after such receipt and such determination, such meeting is held for such purpose within sixty (60) days after having been so called and such determination is made thereat.
- (g) Indemnitee shall cooperate with the person, persons or entity making such determination with respect to Indemnitee's entitlement to indemnification, including providing to such person, persons or entity upon reasonable advance request any documentation or information which is not privileged or otherwise protected from disclosure and which is reasonably available to Indemnitee and reasonably necessary to such determination. Any Independent Counsel, member of the Board of Directors or stockholder of the Company shall act reasonably and in good faith in making a determination regarding the Indemnitee's entitlement to indemnification under this Agreement. Any costs or expenses (including attorneys' fees and disbursements) incurred by Indemnitee in so cooperating with the person, persons or entity making such determination shall be borne by the Company (irrespective of the determination as to Indemnitee's entitlement to indemnification) and the Company hereby indemnifies and agrees to hold Indemnitee harmless therefrom.
- (h) The Company acknowledges that a settlement or other disposition short of final judgment may be successful if it permits a party to avoid expense, delay, distraction, disruption and uncertainty. In the event that any action, claim or proceeding to which Indemnitee is a party is resolved in any manner other than by adverse judgment against Indemnitee (including, without limitation, settlement of such action, claim or proceeding with or without payment of money or other consideration) it shall be presumed that Indemnitee has been successful on the merits or otherwise in such action, suit or proceeding. Anyone seeking to overcome this presumption shall have the burden of proof and the burden of persuasion by clear and convincing evidence.
- (i) The termination of any Proceeding or of any claim, issue or matter therein, by judgment, order, settlement or conviction, or upon a plea of nolo contendere or its equivalent, shall not (except as otherwise expressly provided in this Agreement) of itself adversely affect the right of Indemnitee to indemnification or create a presumption that Indemnitee did not act in good faith and in a manner which he reasonably believed to be in or not opposed to the best interests of the Company or, with respect to any criminal Proceeding, that Indemnitee had reasonable cause to believe that his conduct was unlawful.

7. Remedies of Indemnitee.

(a) In the event that (i) a determination is made pursuant to Section 6 of this Agreement that Indemnitee is not entitled to indemnification under this Agreement, (ii) advancement of Expenses is not timely made pursuant to Section 5 of this Agreement, (iii) no determination of entitlement to indemnification is made pursuant to Section 6(b) of this Agreement within 90 days after receipt by the Company of the request for indemnification, (iv) payment of indemnification is not made pursuant to this Agreement within ten (10) days after receipt by the Company of a written request therefor or (v) payment of indemnification is not made within ten (10) days after a determination has been made that Indemnitee is entitled to indemnification or such determination is deemed to have been made pursuant to Section 6 of this Agreement, Indemnitee shall be entitled to an adjudication in an appropriate court of the State of Delaware, or in any other court of competent jurisdiction, of Indemnitee's entitlement to such indemnification. Indemnitee shall commence such proceeding seeking an adjudication within 180 days following the date on which Indemnitee first has the right to commence such proceeding pursuant to this Section 7(a). The Company shall not oppose Indemnitee's right to seek any such adjudication.

- (b) In the event that a determination shall have been made pursuant to <u>Section 6(b)</u> of this Agreement that Indemnitee is not entitled to indemnification, any judicial proceeding commenced pursuant to this <u>Section 7</u> shall be conducted in all respects as a de novo trial on the merits, and Indemnitee shall not be prejudiced by reason of the adverse determination under <u>Section 6(b)</u>.
- (c) If a determination shall have been made pursuant to <u>Section 6(b)</u> of this Agreement that Indemnitee is entitled to indemnification, the Company shall be bound by such determination in any judicial proceeding commenced pursuant to this <u>Section 7</u>, absent (i) a misstatement by Indemnitee of a material fact, or an omission of a material fact necessary to make Indemnitee's misstatement not materially misleading in connection with the application for indemnification, or (ii) a prohibition of such indemnification under applicable law.
- (d) In the event that Indemnitee, pursuant to this Section 7, seeks a judicial adjudication of his or her rights under, or to recover damages for breach of, this Agreement, or to recover under any directors' and officers' liability insurance policies maintained by the Company, the Company shall pay on his or her behalf, in advance, any and all expenses (of the types described in the definition of Expenses in Section 13 of this Agreement) actually and reasonably incurred by him or her in such judicial adjudication, regardless of whether Indemnitee ultimately is determined to be entitled to such indemnification, advancement of expenses or insurance recovery.
- (e) The Company shall be precluded from asserting in any judicial proceeding commenced pursuant to this Section 7 that the procedures and presumptions of this Agreement are not valid, binding and enforceable and shall stipulate in any such court that the Company is bound by all the provisions of this Agreement. The Company shall indemnify Indemnitee against any and all Expenses and, if requested by Indemnitee, shall (within ten (10) days after receipt by the Company of a written request therefore) advance, to the extent not prohibited by law, such expenses to Indemnitee, which are incurred by Indemnitee in connection with any action brought by Indemnitee for indemnification or advance of Expenses from the Company under this Agreement or under any directors' and officers' liability insurance policies maintained by the Company, regardless of whether Indemnitee ultimately is determined to be entitled to such indemnification, advancement of Expenses or insurance recovery, as the case may be.
- (f) Notwithstanding anything in this Agreement to the contrary, no determination as to entitlement to indemnification under this Agreement shall be required to be made prior to the final disposition of the Proceeding.
 - 8. Non-Exclusivity; Survival of Rights; Insurance; Primacy of Indemnification; Subrogation.
- (a) The rights of indemnification as provided by this Agreement shall not be deemed exclusive of any other rights to which Indemnitee may at any time be entitled under applicable law, the Certificate of Incorporation, the Bylaws, any agreement, a vote of stockholders, a resolution of directors or otherwise, of the Company. No amendment, alteration or repeal of this Agreement or of any provision hereof shall limit or restrict any right of Indemnitee under this Agreement in respect of any action taken or omitted by such Indemnitee in his Corporate Status prior to such amendment, alteration or repeal. To the extent that a change in the DGCL, whether by statute or judicial decision, permits greater indemnification than would be afforded currently under the Certificate of Incorporation, Bylaws and this Agreement, it is the intent of the parties hereto that Indemnitee shall enjoy by this Agreement the greater benefits so afforded by such change. No right or remedy herein conferred is intended to be exclusive of any other right or remedy, and every other right and remedy shall be cumulative and in addition to every other right and remedy given hereunder or now or hereafter existing at law or in equity or otherwise. The assertion or employment of any right or remedy hereunder, or otherwise, shall not prevent the concurrent assertion or employment of any other right or remedy.
- (b) To the extent that the Company maintains an insurance policy or policies providing liability insurance for directors, officers, employees, or agents or fiduciaries of the Company or of any other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise that such person serves at the request of the Company, Indemnitee shall be covered by such policy or policies in accordance with its or their terms to the maximum extent of the coverage available for any director, officer, employee, agent or fiduciary under such policy or policies. If, at the time of the receipt of a notice of a claim pursuant to the terms hereof, the Company has director and officer liability insurance in effect, the Company shall give prompt notice of the commencement of such proceeding to the insurers in accordance with the procedures set forth in the respective policies. The Company shall thereafter take all necessary or desirable action to cause such insurers to pay, on behalf of the Indemnitee, all amounts payable as a result of such proceeding in accordance with the terms of such policies.

- (c) In the event of any payment under this Agreement, the Company shall be subrogated to the extent of such payment to all of the rights of recovery of Indemnitee (other than against the Fund Indemnitors), who shall execute all papers required and take all action necessary to secure such rights, including execution of such documents as are necessary to enable the Company to bring suit to enforce such rights.
- (d) The Company shall not be liable under this Agreement to make any payment of amounts otherwise indemnifiable hereunder if and to the extent that Indemnitee has otherwise actually received such payment under any insurance policy, contract, agreement or otherwise.
- (e) The Company's obligation to indemnify or advance Expenses hereunder to Indemnitee who is or was serving at the request of the Company as a director, officer, employee or agent of any other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise shall be reduced by any amount Indemnitee has actually received as indemnification or advancement of expenses from such other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise.
- 9. Exception to Right of Indemnification. Notwithstanding any provision in this Agreement, the Company shall not be obligated under this Agreement to make any indemnity in connection with any claim made against Indemnitee:
- (a) for which payment has actually been made to or on behalf of Indemnitee under any insurance policy or other indemnity provision, except with respect to any excess beyond the amount paid under any insurance policy or other indemnity provision; or
- (b) for an accounting of profits made from the purchase and sale (or sale and purchase) by Indemnitee of securities of the Company within the meaning of Section 16(b) of the Securities Exchange Act of 1934, as amended, or similar provisions of state statutory law or common law; or
- (c) in connection with any Proceeding (or any part of any Proceeding) initiated by Indemnitee, including any Proceeding (or any part of any Proceeding) initiated by Indemnitee against the Company or its directors, officers, employees or other indemnitees, unless (i) the Board of Directors of the Company authorized the Proceeding (or any part of any Proceeding) prior to its initiation or (ii) the Company provides the indemnification, in its sole discretion, pursuant to the powers vested in the Company under applicable law.
- 10. <u>Duration of Agreement</u>. All agreements and obligations of the Company contained herein shall continue during the period Indemnitee is an officer or director of the Company (or is or was serving at the request of the Company as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise) and shall continue for five (5) years thereafter or, if longer, so long as Indemnitee shall be subject to any Proceeding (or any proceeding commenced under <u>Section 7</u> hereof) by reason of his or her Corporate Status, whether or not he or she is acting or serving in any such capacity at the time any liability or expense is incurred for which indemnification can be provided under this Agreement. This Agreement shall be binding upon and inure to the benefit of and be enforceable by the parties hereto and their respective successors (including any direct or indirect successor by purchase, merger, consolidation or otherwise to all or substantially all of the business or assets of the Company), assigns, spouses, heirs, executors and personal and legal representatives.
- 11. <u>Security</u>. To the extent requested by Indemnitee and approved by the Board of Directors of the Company may at any time and from time to time provide security to Indemnitee for the Company's obligations hereunder through an irrevocable bank line of credit, funded trust or other collateral. Any such security, once provided to Indemnitee, may not be revoked or released without the prior written consent of the Indemnitee.

12. Enforcement.

(a) The Company expressly confirms and agrees that it has entered into this Agreement and assumes the obligations imposed on it hereby in order to induce Indemnitee to serve as an officer or director of the Company, and the Company acknowledges that Indemnitee is relying upon this Agreement in serving as an officer or director of the Company.

- (b) This Agreement constitutes the entire agreement between the parties hereto with respect to the subject matter hereof and supersedes all prior agreements and understandings, oral, written and implied, between the parties hereto with respect to the subject matter hereof.
 - 13. <u>Definitions</u>. For purposes of this Agreement:
- (a) "Corporate Status" describes the status of a person who is or was a director, officer, employee, agent or fiduciary of the Company or of any other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise that such person is or was serving at the express written request of the Company.
- (b) "Disinterested Director" means a director of the Company who is not and was not a party to the Proceeding in respect of which indemnification is sought by Indemnitee.
- (c) "Enterprise" shall mean the Company and any other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise that Indemnitee is or was serving at the express written request of the Company as a director, officer, employee, agent or fiduciary.
- (d) "Expenses" shall include all reasonable attorneys' fees, retainers, court costs, transcript costs, fees of experts, witness fees, travel expenses, duplicating costs, printing and binding costs, telephone charges, postage, delivery service fees and all other disbursements or expenses of the types customarily incurred in connection with prosecuting, defending, preparing to prosecute or defend, investigating, participating, or being or preparing to be a witness in a Proceeding, or responding to, or objecting to, a request to provide discovery in any Proceeding. Expenses also shall include Expenses incurred in connection with any appeal resulting from any Proceeding and any federal, state, local or foreign taxes imposed on the Indemnitee as a result of the actual or deemed receipt of any payments under this Agreement, including without limitation the premium, security for, and other costs relating to any cost bond, supersede as bond, or other appeal bond or its equivalent. Expenses, however, shall not include amounts paid in settlement by Indemnitee or the amount of judgments or fines against Indemnitee.
- (e) "Independent Counsel" means a law firm, or a member of a law firm, that is experienced in matters of corporation law and neither presently is, nor in the past five years has been, retained to represent: (i) the Company or Indemnitee in any matter material to either such party (other than with respect to matters concerning Indemnitee under this Agreement, or of other indemnitees under similar indemnification agreements), or (ii) any other party to the Proceeding giving rise to a claim for indemnification hereunder. Notwithstanding the foregoing, the term "Independent Counsel" shall not include any person who, under the applicable standards of professional conduct then prevailing, would have a conflict of interest in representing either the Company or Indemnitee in an action to determine Indemnitee's rights under this Agreement. The Company agrees to pay the reasonable fees of the Independent Counsel referred to above and to fully indemnify such counsel against any and all Expenses, claims, liabilities and damages arising out of or relating to this Agreement or its engagement pursuant hereto.
- (f) "**Proceeding**" includes any threatened, pending or completed action, suit, arbitration, alternate dispute resolution mechanism, investigation, inquiry, administrative hearing or any other actual, threatened or completed proceeding, whether brought by or in the right of the Company or otherwise and whether civil, criminal, administrative or investigative, in which Indemnitee was, is or will be involved as a party or otherwise, by reason of the fact that Indemnitee is or was an officer or director of the Company, by reason of any action taken by him or her or of any inaction on his or her part while acting as an officer or director of the Company, or by reason of the fact that he is or was serving at the request of the Company as a director, officer, employee, agent or fiduciary of another corporation, partnership, joint venture, trust or other Enterprise; in each case whether or not he is acting or serving in any such capacity at the time any liability or expense is incurred for which indemnification can be provided under this Agreement; including one pending on or before the date of this Agreement, but excluding one initiated by an Indemnitee pursuant to Section 7 of this Agreement to enforce his rights under this Agreement.
- 14. <u>Severability</u>. The invalidity or unenforceability of any provision hereof shall in no way affect the validity or enforceability of any other provision. Without limiting the generality of the foregoing, this Agreement is intended to confer upon Indemnitee indemnification rights to the fullest extent permitted by applicable laws. In the event any provision hereof conflicts with any applicable law, such provision shall be deemed modified, consistent with the aforementioned intent, to the extent necessary to resolve such conflict.

- 15. <u>Modification and Waiver</u>. No supplement, modification, termination or amendment of this Agreement shall be binding unless executed in writing by both of the parties hereto. No waiver of any of the provisions of this Agreement shall be deemed or shall constitute a waiver of any other provisions hereof (whether or not similar) nor shall such waiver constitute a continuing waiver.
- 16. <u>Notice by Indemnitee</u>. Indemnitee agrees promptly to notify the Company in writing upon being served with or otherwise receiving any summons, citation, subpoena, complaint, indictment, information or other document relating to any Proceeding or matter which may be subject to indemnification covered hereunder. The failure to so notify the Company shall not relieve the Company of any obligation which it may have to Indemnitee under this Agreement or otherwise unless and only to the extent that such failure or delay materially prejudices the Company.
- 17. Notices. All notices and other communications given or made pursuant to this Agreement shall be in writing and shall be deemed effectively given: (a) upon personal delivery to the party to be notified, (b) when sent by confirmed electronic mail or facsimile if sent during normal business hours of the recipient, and if not so confirmed, then on the next business day, (c) five (5) days after having been sent by registered or certified mail, return receipt requested, postage prepaid, or (d) one (1) day after deposit with a nationally recognized overnight courier, specifying next day delivery, with written verification of receipt. All communications shall be sent:
 - (a) To Indemnitee at the address set forth below Indemnitee signature hereto.
 - (b) To the Company at:

Aesther Healthcare Acquisition Corp. 515 Madison Avenue, Suite 8078 New York, New York 10022

or to such other address as may have been furnished to Indemnitee by the Company or to the Company by Indemnitee, as the case may be.

- 18. <u>Counterparts</u>. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same Agreement. This Agreement may also be executed and delivered by facsimile signature and in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 19. <u>Headings</u>. The headings of the paragraphs of this Agreement are inserted for convenience only and shall not be deemed to constitute part of this Agreement or to affect the construction thereof.
- 20. <u>Governing Law and Consent to Jurisdiction</u>. This Agreement and the legal relations among the parties shall be governed by, and construed and enforced in accordance with, the laws of the State of Delaware, without regard to its conflict of laws rules. The Company and Indemnitee hereby irrevocably and unconditionally (i) agree that any action or proceeding arising out of or in connection with this Agreement shall be brought only in the Chancery Court of the State of Delaware (the "**Delaware Court**"), and not in any other state or federal court in the United States of America or any court in any other country, (ii) consent to submit to the exclusive jurisdiction of the Delaware Court for purposes of any action or proceeding arising out of or in connection with this Agreement, (iii) waive any objection to the laying of venue of any such action or proceeding in the Delaware Court, and (iv) waive, and agree not to plead or to make, any claim that any such action or proceeding brought in the Delaware Court has been brought in an improper or inconvenient forum.

SIGNATURE PAGE TO FOLLOW

By:
Name: Suren Ajjarapu
Title: Chief Executive Officer

INDEMNITEE

Name:

Address:

Signature Page To Indemnification Agreement

IN WITNESS WHEREOF, the parties hereto have executed this Agreement on and as of the day and year first above written.

Aesther Healthcare Acquisition Corp.

[__], 2021

Aesther Healthcare Sponsor, LLC

Re: Administrative Support Agreement

Ladies and Gentlemen:

This letter agreement by and between Aesther Healthcare Acquisition Corp. (the "Company") and Aesther Healthcare Sponsor, LLC ("AHS"), dated as of the date hereof, will confirm our agreement that, commencing on the date the securities of the Company are first listed on The Nasdaq Global Market (the "Listing Date"), pursuant to a Registration Statement on Form S-1 and prospectus filed with the U.S. Securities and Exchange Commission (the "Registration Statement") and continuing until the earlier of the consummation by the Company of an initial business combination or the Company's liquidation (in each case as described in the Registration Statement) (such earlier date hereinafter referred to as the "Termination Date"):

- (i) AHS shall make available, or cause to be made available, to the Company, or any successor location of AHS, certain office space, utilities and secretarial and administrative support as may be reasonably required by the Company. In exchange therefor, the Company shall pay AHS the sum of \$10,000 per month on the Listing Date and continuing monthly thereafter until the Termination Date; and
- (ii) AHS hereby irrevocably waives any and all right, title, interest, causes of action and claims of any kind as a result of, or arising out of, this letter agreement (each, a "Claim") in or to, and any and all right to seek payment of any amounts due to it out of, the trust account established for the benefit of the public stockholders of the Company and into which substantially all of the proceeds of the Company's initial public offering will be deposited (the "Trust Account") as a result of, or arising out of, this letter agreement, and hereby irrevocably waives any Claim it may have in the future, which Claim would reduce, encumber or otherwise adversely affect the Trust Account or any monies or other assets in the Trust Account, and further agrees not to seek recourse, reimbursement, payment or satisfaction of any Claim against the Trust Account or any monies or other assets in the Trust Account for any reason whatsoever.

This letter agreement constitutes the entire agreement and understanding of the parties hereto in respect of its subject matter and supersedes all prior understandings, agreements, or representations by or among the parties hereto, written or oral, to the extent they relate in any way to the subject matter hereof or the transactions contemplated hereby.

This letter agreement may not be amended, modified or waived as to any particular provision, except by a written instrument executed by the parties hereto.

No party hereto may assign either this letter agreement or any of its rights, interests, or obligations hereunder without the prior written approval of the other party. Any purported assignment in violation of this paragraph shall be void and ineffectual and shall not operate to transfer or assign any interest or title to the purported assignee.

This letter agreement constitutes the entire relationship of the parties hereto, and any litigation between the parties (whether grounded in contract, tort, statute, law or equity) shall be governed by, construed in accordance with, and interpreted pursuant to the laws of the State of New York, without giving effect to its choice of law principles.

[Signature Page Follows]

AESTHER HEALTHCARE ACQUISITION CORP.

By:
Name: Suren Ajjarapu
Title: Chief Executive Officer

AGREED TO AND ACCEPTED BY:
AESTHER HEALTHCARE SPONSOR, LLC

By:
Name: Suren Ajjarapu
Title: Chief Executive Officer

Very truly yours,

[Signature Page to Administrative Support Agreement]

CODE OF ETHICAL BUSINESS CONDUCT

Aesther Healthcare Acquisition Corp. (the "<u>Company</u>") has enjoyed a reputation as a company of high integrity. The Company has worked hard to earn the respect of customers, suppliers, and the public. This Code of Ethical Business Conduct ("<u>Code</u>") embodies the Company's commitment to continue to enjoy this fine reputation into the future. For that reason, the Company expects its directors, officers and employees to share the commitment to comply with all the provisions of the Code and the spirit in which it is intended.

This Code describes the general principles and guidelines applicable to all directors, officers and employees of the Company. Although the general principles outlined in this Code apply to the conduct of all of the Company's business transactions, the Company's directors, officers and employees are also bound by other specific Company policies. All managers are responsible for the enforcement of, and compliance with, all policies of the Company, including distribution and communications to ensure employee knowledge of and compliance with these policies.

APPLICATION OF THE CODE

Every director, officer, and employee ("employees") of the Company is required to comply with the Code and all Company policies. We also expect those agents, consultants and other representatives ("associates") working on the Company's behalf will adhere to high ethical standards. Accordingly, no director, officer or employee of the Company should ask an agent, consultant or other representative to engage in conduct that would be prohibited by the Code or any Company policy or applicable law.

Directors, officers and employees of the Company are expected to maintain high ethical standards in their actions and working relationships with customers, suppliers, fellow employees, competitors, representatives of government, and others. All members of the Company are expected to act in business matters with dual responsibility to the public interest and the Company's interest, above their own. Employees must use sound business practices to maintain their integrity and that of the Company.

COMPLIANCE WITH LAWS

It is the Company's policy to comply with all applicable federal, state and local laws and regulations in the conduct of its business. The Company, its associates and employees are prohibited by law from influencing or inducing favorable government action through bribery or collusion. Accordingly, no associates or employee shall make any payment or offer anything of value in the form of compensation, gift, contribution or otherwise to any government agent, employee or official, whether appointed or elected, for the purpose of inducing favorable governmental action. Should any associate or employee receive a solicitation for a payment, bribe, gift, or contribution from any government agent, employee or official, whether appointed or elected, it should be reported to the General Counsel immediately.

Any requests for information from a governmental or regulatory body should be immediately referred to the General Counsel's office for review. No associate or employee of the Company shall knowingly withhold or conceal information legally requested by any governmental or regulatory body, or knowingly furnish incorrect or misleading information to such body. Any associate or employee of the Company who either knows or has reason to believe that the Company itself, or another Company associate or employee has knowingly withheld or concealed, or is knowingly withholding or concealing information legally requested, or has knowingly furnished, or is knowingly furnishing materially incorrect or misleading information to any governmental or regulatory body, shall immediately report that good faith belief to the General Counsel.

The General Counsel will promptly review any such reports and make the determination whether any material requested by any governmental body is subject to any legal privilege and may be lawfully withheld. In no instance, will the Company or any of its employees knowingly and intentionally provide materially incorrect or misleading information to any government body.

USE OF CORPORATE FUNDS AND RESOURCES

No director, officer or employee will use Company funds, resources or property for his or her personal benefit unless such use is consistent with Company policy or has been properly approved by appropriate Company personnel. Company property must not be sold, loaned, given away, or otherwise disposed of-regardless of condition or value-without proper authorization.

POLITICAL ACTIVITIES AND CONTRIBUTIONS

Corporate funds shall not be used for direct or indirect contributions to political parties, candidates or campaigns. The Company does not prohibit directors, officers or employees from making personal contributions of their time and funds to political candidates, causes or parties of their choice. However, the decision to make such a contribution is personal and imposes no responsibility or obligation on the Company. Company employees may not use work time to assist any party or campaign, and may not be reimbursed for personal political activity.

PAYMENTS TO GOVERNMENT OFFICIALS

It is a violation of Company policy, to give or offer, either directly or indirectly, anything of value to government officials in order to influence their actions or decisions. Company funds or assets will not be used to make gifts to, provide entertainment for, or furnish assistance or other services to, government employees or public officials to induce them to do business with the Company. The U.S. Foreign Corrupt Practices Act applies globally and makes it illegal to offer or give money or anything of value, either directly or indirectly, to foreign government officials in order to obtain, retain or direct business, or to acquire any improper advantage. Nothing of value may be given to a government official, even if deemed nominal, without prior written approval of the Company's General Counsel. Employees are expected to report any request by a government official for payment of money or anything of value, and to report any circumstances that calls into question the integrity of the Company's dealings with government officials.

FINANCIAL ACCOUNTING AND REPORTING

Every director, officer and employee of the Company, and particularly the Chief Executive Officer and the Chief Financial Officer, are required to comply in all respects with all applicable laws, rules and regulations regarding financial accounting and reporting. This includes, but is not limited to, the laws, rules and regulations of the Securities and Exchange Commission ("SEC") and the Financial Accounting Standards Board ("FASB").

Good financial reporting starts with good recordkeeping, and the Company and its management rely on its records to prepare financial statements that present its results of operations and financial position in a full, fair, accurate, timely and understandable manner. These financial statements are relied on by stockholders, creditors, government authorities, and the public. It is therefore critical that all employees involved with recording, summarizing and maintaining business and accounting records do so in accordance with the following:

- All assets, liabilities, revenues and expenses will be recorded in the financial reports of the Company;
- No undisclosed or unrecorded funds or accounts will be established for any purpose;
- No false or artificial entries will be made for any reason; and
- No payments will be approved or made with the intention or understanding that any part of the payments are to be used for any purpose other than that described by the documentation supporting the payment.

Persons involved in preparing and finalizing the Company's financial information, whether for internal or external reporting purposes, should do so in accordance with the following:

- Assist in maintaining internal control over financial reporting.
- Communicate openly and honestly with the Company's external public accountants with respect to quarterly and annual financial reporting and related disclosures.
- Ensure the financial statements and related disclosures include all information deemed necessary to achieve an appropriate degree of transparency of business transactions.

The Chief Executive Officer and the Chief Financial Officer must assure that financial information disclosed in public communications and in the Company's periodic reports filed with the SEC is reported fully, fairly and accurately and in a timely and understandable manner. Every director, officer and employee of the Company, and particularly, the Chief Executive Officer and the Chief Financial Officer must promptly report (confidentially, if desired) to the Company's Board of Directors or to the Company's General Counsel:

- Any material violation of any applicable law, rule or regulation;
- Any incidence of fraud, whether material or not, by management or other persons responsible for recording, processing, summarizing or reporting information required to by disclosed by the Company in reports and statements filed with the SEC; and
- Any material information, fact or circumstance, including any deficiency in any internal control over financial reporting, that could affect or render untrue the information contained in any periodic report that the Company is required to file with the SEC or other regulatory body or that is disclosed in other public communications.

CONFLICT OF INTEREST RELATIONS WITH EMPLOYEES

It is the policy of the Company to provide employment opportunity, wages, and opportunities for advancement, training, and growth to all employees on the basis of merit. It is also the policy of the Company to comply with all existing legislation and established regulations of the various applicable governmental bodies concerned with prohibiting discrimination. The Company will not tolerate discrimination, harassment or other inappropriate treatment of employees on the basis of race, religion, sex, age, national origin, veteran status, disability, sexual orientation, gender identity and/or expression or other legally protected status. It is the Company's practice to deal fairly and equitably with all employees.

The Company is committed to providing a safe and healthy workplace, and shall maintain and, when appropriate, improve its plants, equipment, and methods to that end.

The Company encourages expression by employees about their work, including their ideas for continuous improvement.

ENVIRONMENTAL PROTECTION

The Company conducts its operations with the highest regard for the quality of the environment, including water, air and general land usage. The objective is to comply with standards established by appropriate local, state, or federal agencies at every operating location where emissions into water sources, the atmosphere or solid waste disposal are present. Directors, officers and employees must conduct the business of the Company in an environmentally sound manner, and must comply with applicable environmental laws and regulations.

PROTECTION AND INFORMATION

All directors, officers and employees must be in compliance with the following:

- All confidential information about the Company, including inventions, discoveries, formulas, trade secrets, customer lists and employee data, as well as confidential information acquired by the Company from another company, individual or entity subject to a secrecy and proprietary rights agreement, will be kept confidential. Employees must maintain the confidentiality of such information during and subsequent to the period of employment with the Company.
- Information gathered on competitors, customers, suppliers and other entities with which the Company does business, must be acquired legally and in a manner consistent with the Company's high level of ethics and proper business conduct. Directors, officers and employees who inadvertently obtain confidential information belonging to another company should contact the General Counsel prior to use or disclosure of such information.
- Directors, officers and employees of the Company should recognize that the business records and communications that they create have the potential to become public in the future. Therefore, the Company's directors, officers and employees should avoid exaggeration, derogatory remarks, guesswork or inappropriate characterizations of people and companies in any and all of their work-related communications. This applies equally to e-mail, internal memos and formal reports. Furthermore, the Company's directors, officers and employees are required to comply with the terms of the Company's document retention policies at all times, to avoid even the appearance of impropriety.

FAIR DEALING

Each director, officer and employee of the Company shall deal fairly with the Company's customers, suppliers, competitors and employees. No director, officer or employee may take unfair advantage of anyone through manipulation, concealment, abuse or privileged information, misrepresentation of material facts or any other unfair dealing practice.

ENFORCEMENT

The Code is important to the Company and must be taken seriously by all employees. Accordingly, violations of the Code will not be tolerated and will result in disciplinary action, which can include oral or written reprimand, probation, suspension or termination, in accordance with Company policy.

HOW TO HANDLE SUSPECTED VIOLATIONS OF THE CODE

All directors, officers and employees are expected to seek advice from appropriate personnel if they have any questions about the application of the Code to a specific situation. In addition, to help the Company achieve full compliance, directors, officers and employees are encouraged to raise questions and good faith concerns, and to cooperate fully in any investigation. Known or suspected violations are expected to be reported immediately.

Officers and employees should address their questions and concerns first to their managers, if appropriate. Directors should address their concerns to the Board of Directors unless the concern deals with an accounting, internal accounting control or auditing concern, in which case the concern should be referred to the Chair of the Audit Committee (if any).

HEDGING OF THE COMPANY'S SECURITIES

Section 955 of the Dodd-Frank Wall Street Reform and Consumer Protection Act added Section 14(j) to the Securities Exchange Act of 1934 which requires each issuer (including the Company) to disclose whether any employee, consultant, officer or member of the Board of Directors, or any designee of any employee, consultant, officer or board member, is permitted to purchase hedges on the Company's securities—that is, financial instruments that are designed to hedge or offset against any decrease in the market price for the Company's securities. The Board of Directors has concluded that it is inappropriate for employees, consultants, officers or members of the Board of Directors, or any designee of such persons, to purchase hedges.

AMENDMENTS AND WAIVERS OF THE CODE

Only the Board of Directors may amend or waive a provision of the Code for directors and executive officers of the Company, including the Chief Executive Officer and the Chief Financial Officer. Any such amendment or waiver must be disclosed publicly if and as required by law or stock exchange listing standard.



CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We consent to the inclusion in this Registration Statement on Form S-1 [Amendment No. 1] of our report dated July 16, 2021, except for footnotes 1 and 4, as to which the date is September 2, 2021, with respect to the audited financial statements of Aesther Healthcare Acquisition Corp. as of July 12, 2021 and the related statements of operations, changes in stockholders' equity and cash flow statement for the period from June 17, 2021 (inception) through July 12, 2021. Our report contains an explanatory paragraph regarding the Company's ability to continue as a going concern.

We also consent to the references to us under the heading "Experts" in such Registration Statement.

/s/ MaloneBailey, LLP www.malonebailey.com Houston, Texas September 2, 2021

10375 Richmond Avenue, Suite 710, Houston, Texas 77042 713.343.4286

Zhongzhou Holdings Financial Center (Tower B) #2205 No. 88, Haide Yi Road, Nanshan District, Shenzhen, P.R. China 518054 86.755.86278659

Jintai Guoyi Tower #2007-2008 No. 103, Chaoyang North Road, Chaoyang District, Beijing, P.R. China 100123 86.010.85563995

www.malonebailey.com

Public Company Accounting Oversight Board Registered AICPA

An Independently Owned and Operated Member of Nexia International



AESTHER HEALTHCARE ACQUISITION CORP.

AUDIT COMMITTEE CHARTER

Effective August 4, 2021

I. PURPOSES

The Audit Committee (the "Committee") is appointed by the Board of Directors (the "Board") of Aesther Healthcare Acquisition Corp. (the "Company") to assist the Board in its oversight of the accounting and financial reporting processes of the Company and the Company's compliance with legal and regulatory requirements. To assist the Board in fulfilling its responsibilities, the Committee shall: (A) oversee: (i) audits of the financial statements of the Company; (ii) the integrity of the Company's financial statements; (iii) the Company's processes relating to risk management and the conduct and systems of internal control over financial reporting and disclosure controls and procedures; (iv) the qualifications, engagement, compensation, independence and performance of the Company's independent auditor, and the auditor's conduct of the annual audit of the Company's financial statements and any other services provided to the Company; and (v) the performance of the Company's internal audit function, if any; and (B) produce the annual report of the Committee required by the rules of the U.S. Securities and Exchange Commission (the "SEC").

II. COMMITTEE MEMBERSHIP

A. Composition

The Committee shall consist of at least three members of the Board, subject to applicable phase-in rules of The Nasdaq Stock Market, LLC (the "NASDAQ"). Except as otherwise directed by the Board, a director selected as a Committee member shall continue to be a member for as long as he or she remains a director or until his or her earlier resignation or removal from the Committee. Any member may be removed from the Committee by the Board, with or without cause, at any time. Any vacancy on the Committee shall be filled by majority vote of the Board. No member of the Committee shall be removed except by majority vote of the Board.

B. Chair

The Chair of the Committee shall be appointed from among the Committee members by, and serve at the pleasure of, the Board, shall preside at meetings of the Committee and shall have authority to convene meetings, set agendas for meetings, and determine the Committee's information needs, except as otherwise provided by the Board or the Committee, provided that if the Board does not so designate a chairperson, the members of the Committee, by a majority vote, may designate a chairperson. In the absence of the Chair at a duly convened meeting, the Committee shall select a temporary substitute from among its members to serve as chair of the meeting.

C. Independence

Each member of the Committee shall be an "independent" director in accordance with applicable listing standards of the NASDAQ and Rule 10A-3(b)(1) under the United States Securities Exchange Act of 1934, as amended (the "*Exchange Act*"), subject to any exceptions or cure periods that are applicable pursuant to the foregoing requirements and the phase-in periods permitted under the rules of the NASDAQ under which the Committee is required to have only one independent member at the time of listing, a majority of independent members within 90 days of listing and all independent members within one year of listing. Any action duly taken by the Committee shall be valid and effective, whether or not the members of the Committee at the time of such action are later determined not to have satisfied the requirements for membership provided herein.

D. Financial Literacy

Each member of the Committee shall in the business judgment of the Board have the ability to read and understand fundamental financial statements and otherwise meet the financial literacy requirements of the NASDAQ. At least one member shall be an "audit committee financial expert" as such term is defined under applicable SEC rules.

E. Service on Multiple Audit Committees

No member of the Committee may serve on the audit committee of more than three public companies, including the Company, unless the Board has determined that such simultaneous service would not impair the ability of such member to effectively serve on the Committee.

III. AUTHORITY

In discharging its role, the Committee is empowered to inquire into any matter that it considers appropriate to carry out its responsibilities, with access to all books, records, facilities and personnel of the Company, and, subject to the direction of the Board, the Committee is authorized and delegated the authority to act on behalf of the Board with respect to any matter it determines to be necessary or appropriate to the accomplishment of its purposes.

The Committee shall have authority to retain, direct and oversee the activities of, and to terminate the engagement of, the Company's independent auditor and any other accounting firm retained by the Committee to prepare or issue any other audit report or to perform any other audit, review or attest services and any legal counsel, accounting or other advisor or consultant hired to assist the Committee, all of whom shall be accountable to the Committee.

The Company shall provide the Committee with appropriate funding, as determined by the Committee, for the payment of (a) compensation to any registered public accounting firm engaged for the purpose of preparing or issuing an audit report or performing other audit, review or attest services for the Company; (b) compensation to any independent counsel or other advisers retained by the Committee in carrying out its duties; and (c) ordinary administrative expenses of the Committee that are necessary or appropriate in carrying out its duties.

IV. COMMITTEE MEETINGS

The Committee shall meet on a regularly scheduled basis at least four times per year and additionally as circumstances dictate.

The Committee shall establish its own schedule of meetings. The Committee may also act by unanimous written consent of its members.

Notice of meetings shall be given to all Committee members or may be waived, in the same manner as required for meetings of the Board. Meetings of the Committee may be held by means of conference telephone or other communications equipment by means of which all persons participating in the meeting can hear and speak with each other. A majority of the members of the Committee shall constitute a quorum for a meeting and the affirmative vote of a majority of members present at a meeting at which a quorum is present shall constitute the action of the Committee. The Committee shall otherwise establish its own rules of procedure.

The Committee shall meet in executive session separately with each of the independent auditor, the internal auditor, if any, and with senior management, at least quarterly. At the end of each of the Committee's regularly scheduled meetings, and more frequently as deemed necessary, the Committee shall meet in private session with only the Committee members.

V. DELEGATION

The Committee, by resolution approved by a majority of the Committee, may form and delegate any of its responsibilities to a subcommittee so long as such subcommittee is solely comprised of one or more members of the Committee and such delegation is not otherwise inconsistent with law and applicable rules and regulations of the SEC and the NASDAQ.

VI. KEY RESPONSIBILITIES

The Committee relies on the expertise and knowledge of management, the internal auditors, if any, and the independent auditor in carrying out its oversight responsibilities. Management is responsible for the preparation, presentation, and integrity of the Company's financial statements, for the appropriateness of the accounting principles and reporting policies that are used by the Company, and for establishing and maintaining effective internal control over financial reporting. The independent auditor is responsible for auditing the Company's financial statements and, if applicable, the Company's internal control over financial reporting, and for reviewing the Company's unaudited interim financial statements.

The responsibilities set forth in this charter do not reflect or create any duty or obligation of the Committee to plan or conduct any audit; to determine or certify that the Company's financial statements are complete, accurate, fairly presented or in accordance with generally accepted accounting principles ("*GAAP*") or applicable law; to guarantee or otherwise certify as to the independent auditor's reports; to conduct investigations; or to assure compliance with laws and regulations or the Company's code of business conduct and ethics, internal policies, procedures and controls. The following responsibilities are set forth as a guide for fulfilling the Committee's purposes in such manner as the Committee determines is appropriate.

A. Oversight of the Independent Auditor

- (i) *Independent Auditor Retention*. The Committee is solely and directly responsible for the appointment, evaluation, compensation, retention and, if appropriate, replacement of the independent auditor. The Committee may, in its discretion, seek stockholder ratification of the public accounting firm selected to be the Company's independent auditor.
- (ii) *Independence*. The Committee shall assess at least annually the independent auditor's independence. In connection with this assessment, the Committee shall ensure the receipt of and review formal written statements from the independent auditor delineating all relationships between the auditor and the Company, consistent with applicable requirements of the Public Company Accounting Oversight Board ("*PCAOB*") regarding the independent auditor's communications with the Committee concerning independence. The Committee shall engage in an active dialogue with the independent auditor concerning any disclosed relationships or services that may impact the objectivity and independence of the auditor and take, or recommend that the Board take, appropriate action to oversee and ensure the independence of the auditor.
- (iii) *Quality and Performance*. The Committee shall evaluate at least annually the qualifications and performance of the independent auditor, including the lead partner. The evaluation will include obtaining a written report from the independent auditor describing the firm's internal quality control procedures; any material issues raised by the most recent internal quality control review, PCAOB inspection, or other PCAOB review of the firm, by a peer review of the firm or by any inquiry or investigation by governmental or professional authorities within the past five years, concerning an independent audit or audits carried out by the firm, and any steps taken to address any such issues.
- (iv) *General Oversight*. The independent auditor reports directly to the Committee. The Committee is responsible for oversight of the work of the independent auditor, including resolution of disagreements between management and the independent auditor regarding financial reporting. In connection with its oversight responsibility, the Committee shall consider the independent auditor's communications regarding, among other things, critical accounting policies and practices, all alternative accounting treatments within GAAP related to items material to the financial statements that have been discussed with management, including the ramifications of the alternative treatments and the treatment preferred by the independent auditor, and all material written communications between the independent auditor and management, and shall review the effect or potential effect of any regulatory regime, accounting initiatives or off-balance sheet structures on the Company's financial statements.

(v) *Audit Oversight*. The Committee shall establish with the independent auditor an understanding of the terms of the audit engagement, the role of the auditor with respect to the Company's financial statements and coordination of audit efforts to ensure completeness of coverage, reduction of redundant efforts, the effective use of audit resources, and the use of accounting firms other than the appointed auditors of the Company.

The Committee shall review the scope of the annual audit or interim review (including the level of involvement with unaudited quarterly or other interim-period information), and discuss the results, including, without limitation, the independent auditor's report and all matters required to be communicated to the Committee by the independent auditor in accordance with applicable auditing standards.

The Committee shall discuss with the independent auditor, before the issuance of the audit report, the overall audit strategy, including the timing of the audit, significant risks the auditor identified and significant changes to the planned audit strategy or identified risks. The Committee shall review with the independent auditor any audit problems or difficulties encountered during the course of the audit work and management's response, including any restrictions on the scope of the independent auditor's activities or access to required records, data and information, any difficult or contentious matters for which the auditor consulted outside the engagement team (for example, the audit firm's national office), any significant disagreements with management, and any other matters arising from the audit that are significant to the oversight of the Company's financial reporting process.

- (vi) *Auditor Rotation*. The Committee shall consider whether, in addition to assuring the regular rotation of the lead audit partner as required by law, in the interest of assuring continuing independence of an independent auditor, the Company should regularly rotate the firm appointed as the Company's independent auditor.
- (vii) *Pre-Approval of Auditor Services*. The Committee is exclusively authorized and directed to consider and, in its discretion, approve in advance any services (including the fees and material terms thereof) proposed to be carried out for the Company by the independent auditor or by any other firm proposed to be engaged by the Company as its independent auditor. In connection with approval of any permissible tax services and services related to internal control over financial reporting, the Committee shall discuss with the independent auditor the potential effects of such services on the independence of the auditor.

B. Financial Statements and Other Financial Disclosures

(i) *Quality and Integrity of Financial Statements*. The Committee shall review and discuss with management and the independent auditor: the critical accounting policies and practices used by the Company, and any significant changes in the selection or application of the Company's accounting and auditing principles and practices as suggested by the Company's independent auditor, internal auditors, if any, or management; the accounting treatment to be applied in respect of significant new transactions or other significant events not in the ordinary course of the Company's business; other policies and procedures adopted by the Company to fulfill its responsibilities regarding the presentation of financial statements in accordance with GAAP and applicable rules and regulations of the SEC, including the proper explanation and reconciliation of any non-GAAP measures presented; and any issues that arise with respect to the quality or integrity of the Company's financial statements.

- (ii) *Audited Financial Statements*. The Committee shall review and discuss with management and the independent auditor, before the issuance of the audit report, the financial statements and related notes and the "Management's Discussion and Analysis of Financial Condition and Results of Operations" proposed to be included in the Company's Annual Report on Form 10-K. In this connection, the Committee shall review and discuss with management and the independent auditor the analyses prepared by management setting forth significant financial reporting issues and judgments made in connection with the preparation of the financial statements (including analyses of the effects of alternative GAAP methods on the financial statements), and such other matters for which discussion shall be required by applicable auditing and related PCAOB standards. The Committee shall make a recommendation to the Board as to whether such financial statements should be included in the Company's Annual Report on Form 10-K.
- (iii) *Audit Committee Report*. The Committee shall annually prepare an audit committee report for inclusion where necessary in the proxy statement relating to the annual meeting of stockholders and/or annual report of the Company.
- (iv) *Quarterly Financial Statements*. The Committee shall review and discuss with management and the independent auditor the quarterly financial statements and related notes and the "Management's Discussion and Analysis of Financial Condition and Results of Operations" proposed to be included in the Company's Quarterly Reports on Form 10-Q, together with the analyses prepared by management setting forth significant financial reporting issues and judgments made in connection with the preparation of the financial statements, and such other matters for which discussion shall be required by applicable auditing standards and related PCAOB standards.
- (v) *Earnings Releases and Other Financial Information*. The Committee shall discuss with management and the independent auditor and, prior to issuance, review and approve the Company's earnings releases, including the financial information, use of any "pro forma" or "adjusted" non-GAAP information, and earnings guidance (if such is provided) to be disclosed in such releases. The Committee shall also discuss with management other significant financial information to be provided to analysts or rating agencies.
- (vi) *Payments*. The Committee shall review on a quarterly basis all payments made to the Company's sponsor, officers or directors, or to the Company's or their affiliates.

C. Controls and Procedures

- (i) *Oversight*. The Committee shall provide oversight of management's design and maintenance of the Company's internal control over financial reporting and disclosure controls and procedures. Prior to the filing of the Company's Annual Report on Form 10-K, the Committee shall review with the independent auditor, management and the head of the internal audit function, if any: the Company's annual assessment and report and the independent auditor's report on the effectiveness of the Company's internal control over financial reporting, to the extent then applicable; any "material weakness" or "significant deficiency" in the design or operation of internal control over financial reporting, any steps taken to resolve any such control deficiencies and the adequacy of disclosures about changes in internal control over financial reporting; and any related significant findings and recommendations of the independent auditor or internal audit function, if any, together with management's responses (including, in the case of the independent auditor, any concerns regarding matters within the scope of, and compliance with, Section 10A of the Exchange Act).
- (ii) *Certifications*. The Committee shall review and discuss with management and the independent auditor the certifications and any related disclosures made by the Company's Chief Executive Officer and Chief Financial Officer in the Company's periodic reports about the results of their evaluation of the effectiveness of disclosure controls and procedures and any significant deficiencies or material weaknesses in the design or operation of internal control over financial reporting, and any fraud involving management or other employees who have a significant role in the Company's internal control over financial reporting, prior to the filing of the Company's Annual Reports on Form 10-K and Quarterly Reports on Form 10-Q.

(iii) *Internal Audit Function*. At least annually, the Committee shall review with the independent auditor the responsibilities, budget, staffing, effectiveness and performance of the internal audit function, if any, including the structure, qualification and activities of the internal audit function and the scope of internal audit responsibilities in relation to the independent auditor's duties. The Committee shall review and assess the annual internal audit plan, if any, the process used to develop the plan, and the status of activities, significant findings, recommendations and management's response. The Committee shall recommend for Board approval all matters related to responsibilities, budget and staffing of the internal audit function, if any. The Committee shall recommend for Board approval the appointment and, if appropriate, replacement of the senior internal audit executive.

(iv) *Hiring Policies*. The Committee shall establish clear policies regarding the hiring of employees and former employees of the Company's independent auditor.

D. Risk Management, Compliance and Ethics

- (i) *Risk Management*. The Committee shall review and discuss with management, the head of the internal audit function, if any, and the independent auditor any significant risks or exposures and the Company's policies and processes with respect to risk assessment and risk management, and shall assess the steps management has taken to monitor and control such risks, except with respect to those risks for which oversight has been assigned to other committees of the Board or retained by the Board. The Committee shall review the Company's annual disclosures concerning the role of the Board in the risk oversight of the Company.
- (ii) *Legal and Regulatory Compliance*. The Committee shall review and assess with the Chairman, Co-Chairman or Co-Executive Chairman of the Board or outside counsel, as appropriate, legal and regulatory matters that may have a material impact on the Company's financial statements. The Committee shall also review and recommend for Board approval the code of business conduct and ethics and any other appropriate compliance policies, and will review requests for waivers under the code of business conduct and ethics sought with respect to any executive officer or director. The Committee shall review annually with the Chairman, Co-Chairman or Co-Executive Chairman of the Board or outside counsel, as appropriate, the scope, implementation and effectiveness of the ethics and compliance program, and any significant deviations by officers and employees from the code of business conduct and ethics or other compliance policies, and other matters pertaining to the integrity of management.
- (iii) *Procedures for Complaints*. The Committee shall establish "whistleblowing" procedures for (a) the receipt, retention and treatment of complaints received by the Company regarding accounting, internal accounting controls or auditing matters and (b) the confidential, anonymous submission by the Company's employees of concerns regarding questionable accounting or auditing matters. The Committee shall review any such significant complaints or concerns.
- (iv) *Review and Approval of Swap Transactions*. The Committee shall at least annually review and approve the Company's decision to enter into swaps and other derivative transactions that are exempt from exchange-execution and clearance requirements under "end-user exception" regulations, and review and discuss with management applicable Company policies governing the Company's use of swaps subject to the end-user exception.
- (v) *Compliance with Terms of Initial Public Offering.* The Committee shall monitor compliance with the terms of the Company's initial public offering and promptly take all action necessary to rectify any noncompliance or otherwise to cause compliance with the terms of such offering.

E. Related Party Transactions:

- (a) The Committee will review any issues relating to conflicts of interests and all related party transactions of the Company ("Related Party Transactions").
- (b) The Committee will analyze the following factors, in addition to any other factors the Committee deems appropriate, in determining whether to approve a Related Party Transaction:
- (1) fairness of the terms for the Company (including fairness from a financial point of view);
- (2) materiality of the transaction;
- (3) bids / terms for such transaction from unrelated parties;
- (4) structure of the transaction;
- (5) the policies, rules and regulations of the U.S. federal and state securities laws;
- (6) the policies of the Committee; and
- (7) interests of each related party in the transaction.
- (c) The Committee will only approve a Related Party Transaction if the Committee determines that the terms of the Related Party Transaction are beneficial and fair (including fair from a financial point of view) to the Company and are lawful under the laws of the United States. In the event multiple members of the Committee are deemed a related party, the Related Party Transaction will be considered by the disinterested members of the Board of Directors in place of the Committee.
 - (d) The following transactions will be exempted from the Policy and will be governed by the Company's other applicable policies:
 - (1) payment of compensation by the Company to its officers or directors for service to the Company in their stated capacity;
 - (2) transactions available to all employees or all stockholders of the Company on the same terms; and
 - (3) transactions which, when aggregated for any related party, involve less than \$120,000 and are approved by the Chief Executive Officer, who is not a related party in the transaction.
- (e) Approval of a Related Party Transaction may be conditioned upon the Company and the related party taking any or all of the following additional actions, or any other actions that the Committee deems appropriate:
 - (1) requiring the related party to resign from, or change position within, an entity that is involved in the Related Party Transaction with the Company;
 - (2) assuring that the related party will not be directly involved in negotiating the terms of the Related Party Transaction:
 - (3) limiting the duration or magnitude of the Related Party Transaction;
 - (4) requiring that information about the Related Party Transaction be documented and that reports reflecting the nature and amount of the Related Party Transaction be delivered to the Committee on a regular basis;
 - (5) requiring that the Company have the right to terminate the Related Party Transaction by giving a specified period of advance notice; or
 - (6) appointing a Company representative to monitor various aspects of the Related Party Transaction.

- (f) If the Company or a related party becomes aware that any Related Party Transaction exists that has not been previously approved or ratified under this policy, it will promptly submit the transaction to the Committee or Chair of the Committee or disinterested members of the Board of Directors for consideration. The Committee or Chair of the Committee or Board will evaluate the transaction under this policy and will consider all options, including ratification, amendment or termination of the Related Party Transaction.
- (g) All Related Party Transactions are to be disclosed in the Company's applicable filings with the SEC, as required by the Securities Act of 1933, as amended, and the Securities Exchange Act of 1934, as amended, and related rules and regulations. All Related Party Transactions will be disclosed to the Committee and any material Related Party Transaction will be disclosed to the Board of Directors.
- (h) The Committee is prohibited from approving or ratifying any Related Party Transaction whereby the Company directly or indirectly, including through any subsidiary, extends or maintains credit, arranges for the extension of credit, or renews an extension of credit, in the form of a personal loan to or for any director or executive officer (or equivalent thereof) of the Company.

F. Self-Evaluation and Reporting

- (i) *Self-Evaluation and Charter Review*. The Committee shall conduct an annual self-evaluation of the performance of the Committee, including its effectiveness and compliance with this charter, and recommend to the Board such amendments of this charter as the Committee deems appropriate.
- (ii) *Reporting*. The Committee shall report regularly to the Board on Committee findings and recommendations and any other matters the Committee deems appropriate or the Board requests, and maintain minutes or other records of Committee meetings and activities.

The Committee shall undertake such other responsibilities or tasks as the Board may delegate or assign to the Committee from time to time.

AESTHER HEALTHCARE ACQUISITION CORP.

COMPENSATION COMMITTEE CHARTER Effective August 4, 2021

The responsibilities and powers of the Compensation Committee of the Board of Directors (the "*Board*") of Aesther Healthcare Acquisition Corp. (the "*Company*"), as delegated by the Board, are set forth in this charter (this "*Charter*"). Whenever the Compensation Committee takes an action, it shall exercise its independent judgment on an informed basis that the action is in the best interests of the Company and its stockholders.

I. PURPOSE

The purpose of the Compensation Committee shall be to assist the Board in determining the compensation of the Chief Executive Officer, the chairman of the Board, the Chief Financial Officer and other executive officers of the Company (collectively, the "*Executives*") and make recommendations to the Board with respect to the compensation of the non-Executive officers of the Company and the independent directors.

II. COMMITTEE MEMBERSHIP

The Compensation Committee shall consist of at least three members of the Board. Each member shall meet the independence and experience requirements and standards established from time to time by the Securities and Exchange Commission (the "SEC") and any securities exchange on which the Company's securities are listed or quoted for trading, in each case as amended from time to time. In addition, each member must qualify as a "Non-Employee Director" under Rule 16b-3 under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and satisfy the requirements of Section 162(m) of the Internal Revenue Code for "outside directors," and any other regulatory requirements.

The Board shall elect the members of the Compensation Committee at the first Board meeting practicable and may make changes from time to time pursuant to the provisions below. The members of the Compensation Committee shall serve until their successors are appointed and qualify. Unless a chairman of the Compensation Committee (the "*Chairman*") is elected by the Board or by a majority of the members of the Compensation Committee, no chairman of the Compensation Committee shall be designated. If appointed by the Board or the members of the Compensation Committee, the Chairman shall be a member of the Compensation Committee and, if present, shall preside at each meeting of the Compensation Committee. The Chairman shall perform such duties as may from time to time be assigned to the Chairman by the Compensation Committee or the Board.

A Compensation Committee member may resign by delivering his or her written resignation to the chairman of the Board, or may be removed by majority vote of the Board by delivery to such member of written notice of removal, to take effect at a date specified therein, or upon delivery of such written notice to such member if no date is specified. The Board shall have the power at any time to fill vacancies in the Compensation Committee, subject to such new member(s) satisfying the above requirements.

III. MEETINGS AND COMMITTEE ACTION

The Compensation Committee shall meet at such times as it deems necessary to fulfill its responsibilities, but not less frequently than annually. Meetings of the Compensation Committee shall be called by a majority of the members of the Compensation Committee upon such notice as is provided for in the Company's charter documents with respect to meetings of the Board. A majority of the Compensation Committee members shall constitute a quorum. Actions of the Compensation Committee may be taken in person at a meeting or in writing without a meeting. Actions taken at a meeting, to be valid, shall require the approval of a majority of the members of the Compensation Committee present and voting. Actions taken in writing, to be valid, shall be signed by all members of the Compensation Committee. The Compensation Committee shall report its minutes from each meeting to the Board.

A majority of the members of the Compensation Committee may establish, consistent with the requirements of this Charter, such rules as may from time to time be necessary or appropriate for the conduct of the business of the Compensation Committee. At each meeting, a majority of the members of the Compensation Committee shall appoint as secretary a person who may, but need not, be a member of the Compensation Committee. A certificate of the secretary of the Compensation Committee or minutes of a meeting of the Compensation Committee executed by the secretary setting forth the names of the members of the Compensation Committee present at the meeting or actions taken by the Compensation Committee at the meeting shall be sufficient evidence at all times as to the members of the Compensation Committee who were present, or such actions taken.

The Compensation Committee shall have the authority to delegate any of its responsibilities to subcommittees as it may deem appropriate in its sole discretion. The Chief Executive Officer of the Company may not be present during voting or deliberations of the Compensation Committee with respect to his compensation.

IV. COMMITTEE AUTHORITY AND RESPONSIBILITIES

The Compensation Committee shall have the following authority and responsibilities:

- At least annually review the Company's corporate goals and objectives relevant to the Executives' compensation; evaluate the Executives' performance in light of such goals and objectives; and, either as a Compensation Committee or, together with the other independent directors (as directed by the Board), determine and approve the Executives' compensation level based on this evaluation (and Chief Executive Officer may not be present during voting or deliberations on his or her compensation). In determining the long-term incentive component of the Executives' compensation, the Compensation Committee will consider the Company's performance, the value of similar incentive awards to the Executives at comparable companies, the awards given to the Executives in past years and any relevant legal requirements and associated guidance of the applicable law.
- At least annually review and make recommendations to the Board with respect to director compensation to assist the Board in making the final determination as to director compensation.
- Attempt to ensure that the Company's compensation program is effective in attracting and retaining key employees, reinforce business strategies
 and objectives for enhanced stockholder value, and administer the compensation program in a fair and equitable manner consistent with
 established policies and guidelines.
- Administer the Company's incentive-compensation plans and equity-based plans, insofar as provided therein.
- Make recommendations to the Board regarding approval, disapproval, modification, or termination of existing or proposed employee benefit
 plans.
- Approve any special perquisites, special cash payments and other special compensation and benefit arrangements for the Executives and the Company's employees.
- Approve any share option award or any other type of award as may be required for complying with any tax, securities, or other regulatory requirement, or otherwise determined to be appropriate or desirable by the Compensation Committee or Board.
- Approve the policy for authorizing claims for expenses from the Executives.
- Retain or obtain the advice of a compensation consultant, legal counsel or other adviser, in the sole discretion of the Compensation Committee.
 The Compensation Committee shall be directly responsible for the appointment, compensation and oversight of the work of any compensation consultant, legal counsel and other adviser retained by the Compensation Committee. The Company shall provide for appropriate funding, as determined by the compensation committee, for payment of reasonable compensation to a compensation consultant, legal counsel or any other adviser retained by the Compensation Committee. The Compensation Committee shall have sole authority to approve related fees and retention terms.
- Review and approve the compensation disclosure and analysis prepared by the Company's management, as required to be included in the Company's proxy statement or annual report, or equivalent, filed with the SEC.
- Review and assess the adequacy of this charter annually.

AESTHER HEALTHCARE ACQUISITION CORP.

NOMINATING AND CORPORATE GOVERNANCE COMMITTEE CHARTER

Effective August 4, 2021

The responsibilities and powers of the Nominating and Corporate Governance Committee (the "Committee") of the Board of Directors (the "Board") of Aesther Healthcare Acquisition Corp. (the "Company"), as delegated by the Board, are set forth in this charter (this "Charter"). Whenever the Committee takes action, it shall exercise its independent judgment on an informed basis that the action is in the best interests of the Company and its stockholders.

I. PURPOSE

The Committee is established to help ensure that the Board is properly constituted to meet its fiduciary obligations to stockholders and the Company and that the Company has and follows appropriate corporate governance practices and standards.

II. COMMITTEE MEMBERSHIP

The Committee will consist of at least three members of the Board. The members of the Committee shall be appointed by and serve at the discretion of the Board.

Each member shall meet the independence and experience requirements and standards established from time to time to time by the Securities and Exchange Commission (the "SEC") and any securities exchange on which the Company's securities are listed or quoted for trading, in each case as amended from time to time.

Unless a chairman of the Committee (the "*Chairman*") is elected by the Board or by a majority of the members of the Committee, no chairman of the Committee shall be designated. If appointed by the Board or the members of the Committee, the Chairman shall be a member of the Committee and, if present, shall preside at each meeting of the Committee. The Chairman shall perform such duties as may from time to time be assigned to the Chairman by the Committee or the Board.

A Committee member may resign by delivering his or her written resignation to the chairman of the Board, or may be removed by majority vote of the Board by delivery to such member of written notice of removal, to take effect at a date specified therein, or upon delivery of such written notice to such member if no date is specified. The Board shall have the power at any time to fill vacancies in the Committee, subject to such new member(s) satisfying the above requirements.

III. MEETINGS AND PROCEDURES

The Committee shall meet at such times as it deems necessary to fulfill its responsibilities, but not less frequently than annually. Meetings of the Committee shall be called by a majority of the members of the Committee upon such notice as is provided for in the Company's charter documents with respect to meetings of the Board. A majority of the Committee members shall constitute a quorum. Actions of the Committee may be taken in person at a meeting or in writing without a meeting. Actions taken at a meeting, to be valid, shall require the approval of a majority of the members of the Committee present and voting. Actions taken in writing, to be valid, shall be signed by all members of the Committee. The Committee will maintain written minutes of its meetings, which minutes will be filed with the minutes of the Board.

The Committee may form subcommittees for any purpose that the Committee deems appropriate and may delegate to such subcommittees such power and authority as the Committee deems appropriate. The Committee shall not delegate to a subcommittee any power or authority required by law, regulation or listing standard to be exercised by the Committee as a whole.

IV. COMMITTEE AUTHORITY AND RESPONSIBILITIES

To the extent it deems necessary or appropriate, the Committee shall perform the following:

Board Composition, Evaluation and Nominating Activities

- Making recommendations to the Board regarding the size and composition of the Board, establishing procedures for the nomination process and screen and recommending candidates for election to the Board.
- Recommending for approval by the Board on an annual basis desired qualifications and characteristics for Board membership and with corresponding attributes. Generally, persons to be nominated should (i) have demonstrated notable or significant achievements in business, education or public service; (ii) possess the requisite intelligence, education and experience to make a significant contribution to the Board and bring a range of skills, diverse perspectives and backgrounds to its deliberations; and (iii) have the highest ethical standards, a strong sense of professionalism and intense dedication to serving the interests of the stockholders.
- Establishing and administering a periodic assessment procedure relating to the performance of the Board as a whole and its individual members.

Corporate Governance Generally

- Developing and recommending to the Board a set of corporate governance principles and practices.
- Reviewing periodically the Company's corporate governance principles and practices, the Company's compliance with these principles and practices, and recommend changes, as appropriate.
- Overseeing the evaluation of the Company's management.
- Overseeing, reviewing and reporting to the Board regarding the Company's succession planning for the Board, senior management and other key employees.
- Periodically reviewing and reassessing the adequacy and scope of this Charter and the Committee's established processes and procedures and recommending any proposed changes to the Board for approval.

In performing its responsibilities, the Committee shall have the authority to hire and obtain advice, reports or opinions from internal or external counsel and expert advisors, including search firms, and to set the terms and fees for any such counsel and advisors.